Technical Specification Group Terminals Meeting #18, New Orleans, USA, 4-6 December 2002

Source:	T1
Title:	CR's to TS 34.123-1 v5.1.1 related to package 1 test cases
Agenda item:	5.1.3
Document for:	Approval

This document contains 24 CRs to TS 34.123-1 v5.1.1 related to package 1 test cases. These CRs have been agreed by T1 and are put forward to TSG T for approval.

NOTE: TS 34.123-1 R99, Rel-4 and Rel-5 are all merged into the Rel-5 specification. This means that test cases for the three releases are included in TS 34.123-1 Rel-5 and therefore this is the only release being maintained.

CR related to corrections to idle mode test cases:

Spec	CR	Rev	Release	Subject	Cat	Version Current	Version New	Doc-2nd- Level	Work item	Releases affected
34.123-1	323	-	Rel-5	General corrections for clause 6	F	5.1.1	5.2.0	T1-020691	TEI	R99, Rel- 4, Rel-5
34.123-1	334	-	Rel-5	Corrections to package 1 & 2 idle mode test cases	F	5.1.1	5.2.0	T1-020707	TEI	R99, Rel- 4, Rel-5

CR related to corrections to MAC test cases:

Spec	CR	Rev	Release	Subject	Cat	Version Current	Version New	Doc-2nd- Level	Work item	Releases affected
34.123-1	321	-		Corrections to MAC Package 1 test cases 7.1.1.2, 7.1.1.3, 7.1.1.4,7.1.1.5 and 7.1.1.8	F	5.1.1	5.2.0	T1-020689	TEI	R99, Rel- 4, Rel-5

CR related to corrections to RLC test cases:

Spec	CR	Rev	Release	Subject	Cat	Version Current	Version New	Doc-2nd- Level	Work item	Releases affected
34.123-1	310	-	Rel-5	Correction to package 1 test case 7.2.3.22	F	5.1.1	5.2.0	T1-020659	TEI	R99, Rel- 4, Rel-5
34.123-1	311	-	Rel-5	Correction to package 1 test case 7.2.3.23	F	5.1.1	5.2.0	T1-020660	TEI	R99, Rel- 4, Rel-5
34.123-1	319	-	Rel-5	Modifications to package 1 RLC Test Cases	F	5.1.1	5.2.0	T1-020685	TEI	R99, Rel- 4, Rel-5
34.123-1	331	-	Rel-5	Correction to RLC P1 7.2.3.12 Correct use of Sequence Numbering	F	5.1.1	5.2.0	T1-020704	TEI	R99, Rel- 4, Rel-5
34.123-1	332	-	Rel-5	Correction to package 1 test case 7.2.3.13 and 7.2.3.14	F	5.1.1	5.2.0	T1-020705	TEI	R99, Rel- 4, Rel-5
34.123-1	364	-	Rel-5	Update to TC7.2.3.19(RLC PDU Continuous Transmission)	F	5.1.1	5.2.0	T1-020714	TEI	R99, Rel- 4, Rel-5

CR related to corrections to RRC test cases:

Spec	CR	Rev	Release	Subject	Cat	Version Current	Version New	Doc-2nd- Level	Work item	Releases affected
34.123-1	317	-	Rel-5	Correction of package 1 test case 8.1.1.7	F	5.1.1	5.2.0	T1-020668	TEI	R99, Rel- 4, Rel-5
34.123-1	333	-	Rel-5	Correction to P1 TC8.1.9 SIGNALLING CONNECTION RELEASE INDICATION test case as T1S020674rev1	F	5.1.1	5.2.0	T1-020706	TEI	R99, Rel- 4, Rel-5
34.123-1	335	-	Rel-5	Correction to Package 1 test cases (revision of T1S-020677)	F	5.1.1	5.2.0	T1-020710	TEI	R99, Rel- 4, Rel-5
34.123-1	337	-	Rel-5	Clause 8.1 (Package 1) Rel-5: Correction from CRs approved in RP17meeting	F	5.1.1	5.2.0	T1-020784	TEI	R99, Rel- 4, Rel-5
34.123-1	338	-	Rel-5	CR to Package 1 TC 8.4.1.1: Correction from CRs approved in RP17meeting and T1S020726/727 (revision to T1S020750, T1S020856)	F	5.1.1	5.2.0	T1-020786	TEI	R99, Rel- 4, Rel-5
34.123-1	339	-	Rel-5	Clause 8.2 (Package 1) Rel-5: Correction from CRs approved in RP17meeting	F	5.1.1	5.2.0	T1-020787	TEI	R99, Rel- 4, Rel-5
34.123-1	340	-	Rel-5	Clause 8.3 (Package 1) Rel-5: Correction from CRs approved in RP17meeting	F	5.1.1	5.2.0	T1-020788	TEI	R99, Rel- 4, Rel-5
34.123-1	341	-	Rel-5	Clause 8.3 (Package 1) Rel-5: Correction from CRs approved in RP17meeting (Revision to T1S020737)	F	5.1.1	5.2.0	T1-020789	TEI	R99, Rel- 4, Rel-5
34.123-1	370	-	Rel-5	Corrections to 8.1.2 RRC Connection Establishment and 8.1.3 RRC Connection Release, TDD tests	F	5.1.1	5.2.0	T1-020825	TEI	R99, Rel- 4, Rel-5

CR related to corrections to CS and PS NAS test cases:

Spec	CR	Rev	Release	Subject	Cat	Version Current	Version New	Doc-2nd- Level	Workitem	Releases affected
34.123-1	342	-	Rel-5	Update to clause 10 Circuit Switched Call Control tests as revision of T1S-020584	F	5.1.1	5.2.0	T1-020790	TEI	R99, Rel- 4, Rel-5
34.123-1	343	-	Rel-5	Editorial corrections in test cases 11.1.1.1, 11.3.2 (Package 1) and 11.1.1.2.1 (Package 3).	F	5.1.1	5.2.0	T1-020792	TEI	R99, Rel- 4, Rel-5
34.123-1	344	-	Rel-5	Extension of 'Test purpose' in test case 11.3.1 (Package 1 test case).	F	5.1.1	5.2.0	T1-020793	TEI	R99, Rel- 4, Rel-5
34.123-1	345	-	Rel-5	Modifications and corrections of GMM test cases	F	5.1.1	5.2.0	T1-020794	TEI	R99, Rel- 4, Rel-5

CR related to corrections to Radio Bearer test cases:

Spec	CR	Rev	Release	Subject		Version Current	Version New	Doc-2nd- Level	Workitem	Releases affected
34.123-1	314	-		Corrections to generic setup procedure for radio bearer testing	F	5.1.1	5.2.0	T1-020664	TEI	R99, Rel- 4, Rel-5
34.123-1	382	-	Rel-5	Corrections to radio bearer test cases in clause 14.2	F	5.1.1	5.2.0	T1-020840	TEI	R99, Rel- 4, Rel-5

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

Tdoc # T1-020659

Tdoc # T1S-020559

3GPP TSG- T1 SIG Meeting #2	
Singapore, 18 th – 20 th Sept 200)2

		CHAN	GE REQ	UES	ST			CR-Form-v7
ж	34.123-1	CR <mark>310</mark>	ж rev	-	Ħ	Current version:	5.1.0	ж
For <mark>H</mark>	ELP on using this for	m, see bottom o	f this page or	look a	t th	e pop-up text over	r the	nbols.

 Proposed change affects:
 UICC apps
 ME X
 Radio Access Network
 Core Network

Title:	ж	Correction to package 1 test case 7.2.3.22		
Source:	ж	Ericsson		
Work item code:	:#	TEI	<i>Date:</i>	11/09/2002
Category:	ж	F	Release: ೫	
		Use <u>one</u> of the following categories:	Use <u>one</u> of	the following releases:
		F (correction)	2	(GSM Phase 2)
		A (corresponds to a correction in an earlier release) R96	(Release 1996)
		B (addition of feature),	R97	(Release 1997)
		C (functional modification of feature)	R98	(Release 1998)
		D (editorial modification)	R99	(Release 1999)
		Detailed explanations of the above categories can	Rel-4	(Release 4)
		be found in 3GPP TR 21.900.	Rel-5	(Release 5)
			Rel-6	(Release 6)

Reason for change: ೫	Some PDUs sent may be lost in UL due to window size is to small.					
Summary of change: ₩	Added specific value for Tx Window Size parameter. Value is set to 256 to avoid PDU loss which otherwise could happen if the default value (128) is used. In the test case 2 * T / TTI SDUs of size AM_7_PayloadSize - 1 bytes is sent by the SS. As T = 2000 ms and TTI=20 ms then 200 SDUs (=200 RLC AM PDUs) will be sent by the SS. Tx window size = 256 guarantees that all SDUs will be returned and the test purpose can be achieved.					
Consequences if 🛛 🕷	Good UE may fail the test case					
not approved:						
Clauses affected: 🕺	7.2.3.22					
Other specs % Affected:	Y N X Other core specifications # X Test specifications # X O&M Specifications #					
Other comments: #	Affects R99, REL-4 and REL-5 test cases.					

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at http://www.3gpp.org/specs/CR.htm. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

7.2.3.22 Polling for status / Operation of Timer_Poll timer / Stopping Timer_Poll timer

7.2.3.22.1 Definition

This case tests that the UE will stop the Timer_Poll timer if it receives a STATUS PDU within Timer_Poll ms after a poll for status is transmitted. Incorrect operation of polling will cause degradation of service, or possible service failure.

7.2.3.22.2 Conformance requirement

Timer_Poll.

This timer shall only be used when so configured by upper layers. The value of the timer is signalled by upper layers. In the UE this timer shall be started when the successful or unsuccessful transmission of an AMD PDU containing a poll is indicated by lower layer. In UTRAN it should be started when an AMD PDU containing a poll is submitted to lower layer. If x is the value of the state variable VT(S) after the poll was submitted to lower layer, the timer shall be stopped upon receiving:

- positive acknowledgements for all the AMD PDUs with "Sequence Number" up to and including x 1; or
- a negative acknowledgement for the AMD PDU with "Sequence Number" = x 1.

If the timer expires and no STATUS PDU fulfilling the criteria above has been received:

- the Receiver shall be polled once more;
- the timer shall be restarted; and
- the new value of VT(S) shall be saved.

If a new poll is sent when the timer is active, the timer shall be restarted at the time specified above, and the value of VT(S) shall be saved.

[...]

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see subclause 9.7.1):
 - if polling is not prohibited, see subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clause 9.5.

7.2.3.22.3 Test purpose

To verify that the Timer_Poll timer is stopped when receiving a STATUS PDU that acknowledges all AMD PDUs with SN up to and including VT(S)-1 at the time the poll was transmitted.

7.2.3.22.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	First run	Second run
Tx_Window_Size	<u>256</u>	<u>256</u>
Polling info		
Last transmission PDU poll	FALSE	FALSE
Timer_poll	500	1000
Timer_Poll_Periodic	2000	2000

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let T be the value of the Timer_Poll_Periodic timer.

- a) The SS transmits at least 2 * T / TTI SDUs of size AM_7_PayloadSize 1 bytes.
- b) The SS receives PDUs from the UE, and notes the time on receiving the first PDU with the P bit set. This time will be recorded as $T_{1.}$
- c) The SS sends a STATUS PDU acknowledging all the PDUs up to and including the PDU carrying the poll request.
- d) The SS continues to receive PDUs from the UE and notes the time on receipt of the next PDU with the P bit set. This time will be recorded as T_2 .
- e) The SS may optionally release the radio bearer.

The test case is run once for each set of initial RLC parameters.

Expected sequence

Step	Direction		Message	Comments				
_	UE SS		_					
1	÷	-	DOWNLINK RLC PDU	SDU 1				
2	÷	-		SS continues to transmit RLC SDUs				
3	÷	-	DOWNLINK RLC PDU	SDU ceil(2T/TTI)				
4	÷	>	UPLINK RLC PDU	SDU 1				
5	÷	>	UPLINK RLC PDU	SDU 2				
6	5 →			SS continues to receive RLC PDUs				
7	->	>	UPLINK RLC PDU	SN = ceil(T/TTI), Poll: Note T_1				
8	÷	-	STATUS PDU	ACK SN 0 to SN ceil(T/TTI)				
9	÷	>	UPLINK RLC PDU	SN = ceil(T/TTI)+1				
10	10 →			SS continues to receive RLC PDUs				
11	\rightarrow		UPLINK RLC PDU	SN = ceil(2T/TTI), Poll: Note T_2				
12			RB RELEASE	Optional step				
NOTE	NOTE 1: The Expected Sequence shown is infomative.							
			INK and DOWNLINK PDU flows may over					
	Inf	format	ion such as SDU, PDU or Sequence numb	pers given in the comments column shall be				
	CO	nside	red informative only, for test case developm	nent purposes.				

7.2.3.22.5 Test requirements

For both execution runs, the measured time $T_2 - T_1$ shall be 2000 ms.

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

Tdoc # T1-020660

3GPP TSG- T1 SIG Meeting #25 Singapore, 18th – 20th Sept 2002

51	,	•								CR-Form-v7
				CHANGE		UF	ST			0
ж		34.123-1	СD	211	жrev		ж	Current version:	510	ж
		34.123-1	UK	311	жIev	-			5.1.0	
For <mark>H</mark>	ELP or	using this for	m, see	e bottom of this	s page or l	look	at th	e pop-up text over	the X syn	nbols.

Proposed change affects: UICC apps#

ME 🗙 Radio Access Network 📃 Core Network 🦲

Title:	ж	Correction to package 1 test case 7.2.3.23					
Source:	ж	Ericsson					
Work item code	:#	TEI		Date:	11/09/2002		
		-		.			
Category:	ж			Release: ೫			
		lse <u>one</u> of the following categ	gories:	Use <u>one</u> of	the following releases:		
		F (correction)		2	(GSM Phase 2)		
		A (corresponds to a corr	rection in an earlier rele	ease) R96	(Release 1996)		
		B (addition of feature),		R97	(Release 1997)		
		C (functional modificatio	n of feature)	R98	(Release 1998)		
		D (editorial modification)		R99	(Release 1999)		
		etailed explanations of the a	bove categories can	Rel-4	(Release 4)		
		e found in 3GPP TR 21.900.		Rel-5	(Release 5)		
				Rel-6	(Release 6)		

Reason for change: ℜ	The amount of PDUs may cause overflow in the UE RLC Tx buffer. The test purpose can still be achieved with decreased number of sent PDUs. The test procedure can be simplified
Summary of change: ^{भ्र}	 RLC parameter "Last retransmission PDU poll" is set to FALSE to make sure no polls are caused by UE retransmitting any PDUs. Test procedure:
	 a. Step a): The number of PDUs sent by SS are reduced from 2*Poll_PDU + ceil(T / TTI) to ceil(Transmission Window Size * Poll Window / 100), i.e. from 42 (2*16+ceil(200ms/20ms)) to 20 (ceil(32*60%))
	b. Step b): Added that time T1 is noted
	c. Step c): marked as void as the simplified test procedure does not require and STATUS PDU to be sent by SS.
	d. Step d): marked as void as this step does not serve any purpose
	 Step e): changed to state state that SS waits until a PDU with poll bit set is received and note the time as T2.
	 Expected sequence is updated according to the changes to the test procedure

Consequences if	Good UE may fail due to RLC Tx buffer overflow.
not approved:	Sood DE may fail due to REC 1X buller overnow.
Clauses affected:	¥ 7.2.3.23
	YN
Other specs	# X Other core specifications
Affected:	X Test specifications
	X O&M Specifications
Other comments:	X Affects R99, REL-4 and REL-5 test cases.

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

7.2.3.23 Polling for status / Operation of Timer_Poll timer / Restart of the Timer_Poll timer

7.2.3.23.1 Definition

This case tests that the UE will restart the Timer_Poll timer if another poll request is transmitted whilst the timer is running. Incorrect operation of polling will cause degradation of service, or possible service failure.

This test applies to all UE.

7.2.3.23.2 Conformance requirement

Timer_Poll.

This timer shall only be used when so configured by upper layers. The value of the timer is signalled by upper layers. In the UE this timer shall be started when the successful or unsuccessful transmission of an AMD PDU containing a poll is indicated by lower layer. In UTRAN it should be started when an AMD PDU containing a poll is submitted to lower layer. If x is the value of the state variable VT(S) after the poll was submitted to lower layer, the timer shall be stopped upon receiving:

- positive acknowledgements for all the AMD PDUs with "Sequence Number" up to and including x 1; or
- a negative acknowledgement for the AMD PDU with "Sequence Number" = x 1.

If the timer expires and no STATUS PDU fulfilling the criteria above has been received:

- the Receiver shall be polled once more;
- the timer shall be restarted; and
- the new value of VT(S) shall be saved.

If a new poll is sent when the timer is active, the timer shall be restarted at the time specified above, and the value of VT(S) shall be saved.

[...]

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see subclause 9.7.1):
 - if polling is not prohibited, see subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clause 9.5.

7.2.3.23.3 Test purpose

To verify that if a new poll is sent when the timer is running it is restarted.

198

7.2.3.23.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	
Polling info	
Last retransmission PDU poll	FALSE
Last transmission PDU poll	FALSE
Timer_poll	200
Poll_Window	60
Poll_PDU	16
Transmit window size	32

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let T be the value of the Timer_Poll timer.

- a) The SS starts transmission of <u>ceil(Tx_Window_Size * 60%)</u> <u>2*Poll_PDU + ceil(T / TTI)</u> numbers of SDUs of size AM_7_PayloadSize 1bytes.
- b) Whilst transmitting, the SS receives PDUs from the UE, until it receives the second PDU with the P bit set. This time is recorded as T₁. (Note: poll due to Poll_Window).
- c) VoidThe SS sends a STATUS PDU acknowledging all the PDUs received so far.
- d) <u>Void</u>The SS continues to receive consecutive PDUs with the poll bit set until a PDU is received without the poll bit set. The time the last PDU with the poll bit set was received is recorded as T₁.
- e) The SS continues to receive PDUs from the UE and notes the time on receiption of the next PDU with the P bit set. The SS waits until a PDU is received with the poll bit set and notes the time when it was received. This time is will be recorded as T₂.
- f) The SS may optionally release the radio bearer.

Expected sequence

Step	Direction	Message	Comments				
-	UE SS	_					
1	÷	DOWNLINK RLC PDU	SDU 1				
2	÷		SS continues to transmit RLC SDUs				
3	÷	DOWNLINK RLC PDU	SDU ceil(Tx_Window_Size * 60%) ^{2*} Poll_PDU + ceil(T / TTI) numbers of SDUs				
4 5	\rightarrow \rightarrow	UPLINK RLC PDU UPLINK RLC PDU	are sent. SDU 1 SDU 2				
6	\rightarrow		SS continues to receive RLC PDUs				
7	\rightarrow	UPLINK RLC PDU	SN = poll_PDU - 1, 1 st Poll, Timer_Poll started				
8 9	\rightarrow	Void Void					
10	\rightarrow	UPLINK RLC PDU	SS continues to receive RLC PDUs				
11	\rightarrow	UPLINK RLC PDU	SN= ceil(Tx_Window_Size * 60%)-1, 2 nd Poll, Timer_Poll restarted <u>.</u> ÷ Note T1				
12	÷	STATUS PDU <u>Void</u>	ACK SN 0 to SN = coil(Tx_Window_Size * 60%)-1				
13	\rightarrow	UPLINK RLC PDUVoid	SN = ceil(Tx_Window_Size * 60%), Poll, Timer_Poll restarted: Note T1				
14	÷	<u>Void</u>	SS continues to receive RLC PDUs. If poll bit is set in consecutive PDUs then note new value of T1 until a PDU without a poll bit set is received.				
15	<i>→</i>	UPLINK RLC PDU	SN = ceil(Tx_Window_Size * 60%) + ceil(T / TTI) -1, 3 rd Poll, Timer_Poll expired: <u>SS</u> waits for reception of PDU with poll bit set, 3^{rd} Poll, Timer_Poll expired. Note T ₂				
16		RB RELEASE	Optional step				
NOTE	16 Optional step NOTE: The Expected Sequence shown is infomative. The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity. Information such as SDU, PDU or Sequence numbers given in the comments column shall be considered informative only, for test case development purposes.						

7.2.3.23.5 Test requirements

The measured time $T_2 - T_1$ shall be 200 ms.

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

Tdoc # T1-020664

Tdoc **#***T*1S-020570

3GPP TSG-	T1 SIG Me	eting #25
Singapore,	18 th – 20 th	Sept 2002

	CHANGE REQUEST							
ж <mark>3</mark>	<mark>84.123-1</mark>	CR <mark>314</mark>	ж rev	- * (Current version	^{n:} 5.1.0	ж	
For <u>HELP</u> on u	ising this fo	rm, see bottom of	this page or l	look at the	pop-up text o	ver the X sym	bols.	
Proposed change affects: UICC apps # ME X Radio Access Network Core Network								
Title: ೫	Correctio	ns to generic setu	p procedure	for radio be	earer testing			
Source: ೫	Ericsson	, ETSI TF160						
Work item code: ℜ	TEI				Date: ೫	17/09/2002		
Category: अ Reason for change	F (con A (co B (ad C (fur D (ed Detailed ex be found in		ction in an ear of feature) ove categories nations havin	<i>lier release)</i> can g both CS	Use <u>one</u> of the 2 (G R96 (F R97 (F R98 (F R99 (F R99 (F Rel-4 (F Rel-5 (F Rel-6 (F and PS radio		rder of	
Summary of chang	PS ra CS. I Howe that C to be ge: # In 14 simul	 lishment can be be adio bearer in a see n CR (T1-020545) ever, current TTCN CS is established b performed. 1.2 (Generic test p taneous signalling) In test procedur radio bearer is on CS radio bearer 	cond RADIO the order wa implementa before PS and procedure for) following ch re step a) the established b ce table, step	BEARER s is proposed tion for rad d a change r testing mu hanges hav text have before PS r 0 9 and 10a	setup messag d to be PS firs lio bearer test would require ulti-RB combir re been introdu been changed adio bearer	e; or first PS a t and then CS cases have a e considerable nations and uced: d to reflect tha pdated to refle	t CS	
Consequences if not approved:	ж Dela	ayed availablility of	TTCN for m	ulti-RB con	nbination test	cases		

 Clauses affected:
 #
 14.1.2

 Other specs affected:
 #
 Y
 N

 X
 Other core specifications
 #

 X
 Test specifications
 #

	[X O&M Specifications	
Other comments:	ж	Affects R99, REL-4 and REL-5	

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

14 Interoperability Radio Bearer Tests

14.1 General information for interoperability radio bearer tests

The purpose of the interoperability radio bearer test cases are to ensure interoperability of UE's in different regions and networks. For this purpose representative radio bearer configurations that will be used in real network implementations have been defined in TS 34.108 [9], clause 6.10.

The applicability of radio bearer tests is dependent on the UE uplink and downlink radio access capabilities and UE support tele- and bearer-services. See TS 34.123-2, annex B for applicability of the specific test cases.

14.1.1 Generic radio bearer test procedure for single RB configurations

This procedure is used to test single radio bearer configurations and speech only radio bearers. For testing of multiple radio bearer combinations as well as for testing simultaneous transmission and reception of user data and signalling data then the procedure as specified in 14.1.2 should be used.

Initial conditions

UE in idle mode

Test procedure

- a) The SS establish setup the reference radio bearer configuration as specified in TS 34.108, clause 6.10 for the actual radio bearer test.
- b) The SS limits the UE allowed uplink transport format combinations according to the "Restricted UL TFCIs", as specified for the sub-test of the actual radio bearer test, using the RRC transport format combination control procedure. See note 1.
- c) The SS closes the test loop using UE test loop mode 1 and setting the UL RLC SDU size parameter, for all radio bearers under test, according to the "UL RLC SDU size" value as specified for the sub-test of the actual radio bearer test. See note 2.
- d) The SS transmits, for all radio bearers under test, one or more RLC SDUs having the size equal to the "Test data size" as specified for the sub-test of the actual radio bearer test. See note 3.
- e) The SS checks that, for all radio bearers under test, the content of the received RLC SDU has the correct content and is received having the correct transport format. See TS 34.109 [10] clause 5.3.2.6.2 for details regarding the UE loopback of RLC SDUs.
- f) The SS opens the UE test loop.
- g) Steps b) to f) are repeated for all sub-tests
- h) The SS may optionally release the radio bearer.
- i) The SS may optionally deactivate the radio bearer test mode.
- NOTE 1: The restricted set of TFCIs shall contain all possible TFCI that could happen in a sub-test. The actual TTI of the different radio bearers and signaling radio bearers as well as the possible UE processing delays shall be taken into consideration.

Release 5

1998

NOTE 2: Selection of UL RLC SDU size parameter:

For the case when the reference radio bearer configuration under test uses RLC transperant mode in downlink and is not configured for segmented operation then the radio bearer test case shall set the UL RLC SDU size equal to the UL RLC PDU size. See [7] TS 25.322 for details regarding UE operation in RLC transperent mode. In case the reference radio bearer configuration under test does not use RLC transparent mode then the UL RLC SDU size parameter shall be selected to achieve loop back of all test data received in the DL RLC SDU, i.e. the UL RLC SDU size is set to the nearest multiple of the payload size of the UL TF under test minus the size of the length indicator and expansion bit which is equal or bigger than the test data size. For some reference radio bearer configurations this may cause the UE to return the UL RLC SDU in more than one TTI, i.e. in case no UL TF is available to cover the UL RLC SDU size. However, as the test procedure only send downlink test data once there is no risk for the UE transmission buffer to become full even if the returned RLC SDUs need to be transmitted in more than one TTI.

NOTE 3: Selection of test data size:

For the case when the reference radio bearer configuration under test uses RLC transperant mode in downlink and is not configured for segmented operation then the radio bearer test case shall use a DL RLC SDU size (defined by the "Test data size" parameter) equal to the DL RLC PDU size. See [7] TS 25.322 for details regarding UE operation in RLC transperent mode. In case the reference radio bearer configuration under test does not use RLC transparent mode in downlink, the DL RLC SDU size/ test data size shall be set equal to the payload size of the DL TF under test minus the size of the length indicator and the expansion bit.

Expected sequence

CS paging procedure

Step	Direction		Message	Comments
	UE SS			
1	<		SYSTEM INFORMATION (BCCH)	Broadcast
2	<-	-	PAGING TYPE 1 (PCCH)	Paging (CS domain, TMSI)
3	>	>	RRC CONNECTION REQUEST (CCCH)	RRC
4	< RR0		RRC CONNECTION SETUP (CCCH)	RRC
5	> RRC CONNECTION SETUP COM		RRC CONNECTION SETUP COMPLETE (DCCH)	RRC
6	>	>	PAGING RESPONSE (DCCH)	RR

PS paging procedure

Step	Direction		p Direction		Message	Comments
	UE SS					
1	<-	-	SYSTEM INFORMATION (BCCH)	Broadcast		
2	<'	-	PAGING TYPE 1 (PCCH)	Paging (PS domain, P-TMSI)		
3	>		RRC CONNECTION REQUEST (CCCH)	RRC		
4	<-	-	RRC CONNECTION SETUP (CCCH)	RRC		
5	>	>	RRC CONNECTION SETUP COMPLETE (DCCH)	RRC		
6a	>		SERVICE REQUEST (DCCH)	GMM		
6b	<		SECURITY MODE COMMAND	RRC see note 1		
6c	>	>	SECURITY MODE COMPLETE	RRC see note 1		

Note 1 Step 6b and Step 6c are inserted in order to stop T3317 timer in the UE, which starts after transmitting SERVICE REQUEST message.

Step	Direction		Message	Comments
	UE	SS		
16	< >		Paging	Use the CS paging procedure for testing of CS and combined CS/PS reference radio bearer configurations.
				Use the PS paging procedure for testing of PS reference radio bearer configurations.
7	<-	-	ACTIVATE RB TEST MODE (DCCH)	TC
8	>	>	ACTIVATE RB TEST MODE COMPLETE (DCCH)	TC
9	<-	-	RADIO BEARER SETUP (DCCH)	RRC
10	>	>	RADIO BEARER SETUP COMPLETE (DCCH)	RRC
11	<-	-	TRANSPORT FORMAT COMBINATION CONTROL (DCCH)	RRC Transport format combinations is limited to "Restricted UL TFCIs", as specified for the sub-test
12	<-	-	CLOSE UE TEST LOOP (DCCH)	TC UE test mode 1 RLC SDU size is for every active radio bearer set to "UL RLC SDU size", as specified for the sub-test.
13	>	>	CLOSE UE TEST LOOP COMPLETE (DCCH)	TC
14	<-	-	DOWNLINK RLC SDU	Send test data using the downlink transport format combination under test
15	;	>	UPLINK RLC SDU	
16	<-	-	OPEN UE TEST LOOP (DCCH)	TC
17	>	>	OPEN UE TEST LOOP COMPLETE (DCCH)	TC
18			Repeat steps 11 to 17 for every sub-test.	
19			RB RELEASE	RRC Optional step
20	<-	-	DEACTIVATE RB TEST MODE	TC Optional step
21	;	>	DEACTIVATE RB TEST MODE COMPLETE	TC Optional step

14.1.2 Generic test procedure for testing multi-RB combinations and simultaneous signalling

This procedure is used to test multiple radio bearer combinations. This procedure is also used to verify simultaneous transmission and reception of user data and signalling data.

Initial conditions

UE in idle mode

Test procedure

a) The SS establish the reference radio bearer configuration as specified in TS 34.108, clause 6.10 for the actual radio bearer test. For the case when the reference radio bearer configuration includes radio bearers for both CS and PS domain then the radio bearer setup procedure has to be performed once per domain. The first radio bearer setup procedure shall perform configuration of the physical channel for the radio bearer combination under test as well as the transport channels for the <u>CS radio bearer(s), also the transport format combination set for only CS radio bearers has to provided. The second radio bearer procedure shall perform the configuration for the transport channel for the PS radio bearers. The Physical channel configuration shall be done for both CS and PS radio bearers combined. Here the transport format combination set for both CS and PS radio bearer(s). If only one physical channel is used then the second radio bearer setup procedure shall not perform any physical channel configuration, but only perform the transport channel configuration of the additional CS radio bearer(s) as well as map these transport channels to the existing physical channel.</u>

2000

- b) The SS limits the UE allowed uplink transport format combinations according to the "Restricted UL TFCIs", as specified for the sub-test of the actual radio bearer test, using the RRC transport format combination control procedure. See note 1.
- c) The SS closes the test loop using UE test loop mode 1 and setting the UL RLC SDU size parameter, for all radio bearers under test, according to the "UL RLC SDU size" value as specified for the sub-test of the actual radio bearer test. See note 2.
- d) The SS starts transmitting continous test data for all radio bearers under test. The number of RLC SDUs to transmit every TTI and the size "Test data size" is specified for each sub-test of the actual radio bearer test. See note 3.
- e) The SS waits the time T1 equal to 12 times the largest TTI. See note 4
- f) SS transmit a MEASUREMENT CONTROL message requesting periodic reporting with a period of T2.
- g) SS waits the time equal to 2 times T2
- h) During step e) to g) the SS checks that, for all radio bearers under test, the content of the received RLC SDUs have the correct content and is received having the correct transport format. See TS 34.109 [10] clause 5.3.2.6.2 for details regarding the UE loopback of RLC SDUs.
- i) The SS opens the UE test loop.
- j) Steps b) to i) are repeated for all sub-tests
- h) The SS may optionally release the radio bearer.
- i) The SS may optionally deactivate the radio bearer test mode.
- NOTE 1: The restricted set of TFCIs shall contain all possible TFCI that could happen in a sub-test. The actual TTI of the different radio bearers and signaling radio bearers as well as the possible UE processing delays shall be taken into consideration.
- NOTE 2: Selection of UL RLC SDU size parameter:

For the case when the reference radio bearer configuration under test uses RLC transperant mode in downlink and is not configured for segmented operation then the radio bearer test case shall set the UL RLC SDU size equal to the UL RLC PDU size. See [7] TS 25.322 for details regarding UE operation in RLC transperent mode. In case the reference radio bearer configuration under test does not use RLC transparent mode then, as the test procedure is based on continous downlink transmission of test data in sub-sequent TTIs, the UL RLC SDU size parameter shall be selected to adopt to the uplink data rate and to the uplink/downlink TTI ratio. Selection of UL RLC SDU size for the different radio bearers under test should be such that the UE returns data in sub-sequent TTIs without causing the UE transmission buffer to become full. To achieve this the UL RLC SDU size shall be set to UL TF payload size under test, minus the size of length indicator and expansion bit, and divided by the ratio between downlink and uplink TTI equal to 10 ms, and the uplink TTI equal to 20 ms, then for the transport format 4x336 (TF payload size = 4x320=1280 bits) the UL RLC SDU size parameter should be set to 632 bits (=1280bits/(20ms/10ms)- 8 bits).

NOTE 3: Selection of test data size:

For the case when the reference radio bearer configuration under test uses RLC transperant mode in downlink and is not configured for segmented operation then the radio bearer test case shall use a DL RLC SDU size (defined by the "Test data size" parameter) equal to the DL RLC PDU size. See [7] TS 25.322 for details regarding UE operation in RLC transperent mode. In case the reference radio bearer configuration under test does not use RLC transparent mode in downlink, the DL RLC SDU size/ test data size shall be set equal to the payload size of the DL TF under test minus the size of the length indicator and the expansion bit.

NOTE 4: [10] TS 34.109 clause 5.3.2.9 defines the loopback delay requirement for UE test loop mode 1 to be max 10 times actual TTI of a radio bearer when RLC and MAC is operated in transparent mode. As RLC/MAC may be operated in non-transparent modes depending on the actual reference radio bearer configuration to be tested an additional 2 TTI have been added to secure that UE starts transmitting data in uplink before SS transmit the MEASUREMENT CONTROL message.

Expected sequence

CS paging procedure

Step	Direction		Message	Comments
	UE SS			
1	<		SYSTEM INFORMATION (BCCH)	Broadcast
2	<		PAGING TYPE 1 (PCCH)	Paging (CS domain, TMSI)
3	>		RRC CONNECTION REQUEST (CCCH)	RRC
4	<		RRC CONNECTION SETUP (CCCH)	RRC
5	>		RRC CONNECTION SETUP COMPLETE (DCCH)	RRC
6	>	>	PAGING RESPONSE (DCCH)	RR

PS paging procedure

Step	Direction		Message	Comments
	UE	SS		
1	<		SYSTEM INFORMATION (BCCH)	Broadcast
2	<-	-	PAGING TYPE 1 (PCCH)	Paging (PS domain, P-TMSI)
3	>	>	RRC CONNECTION REQUEST (CCCH)	RRC
4	<-	-	RRC CONNECTION SETUP (CCCH)	RRC
5	>		RRC CONNECTION SETUP COMPLETE (DCCH)	RRC
6a	>		SERVICE REQUEST (DCCH)	GMM
6b	<		SECURITY MODE COMMAND	RRC see note 1
6c	>	>	SECURITY MODE COMPLETE	RRC see note 1

Note 1 Step 6b and Step 6c are inserted in order to stop T3317 timer in the UE, which starts after transmitting SERVICE REQUEST message.

l

Step	Direction		Message	Comments
	UE	SS		
16	>	-	Paging	Use the CS paging procedure for testing of CS and combined CS/PS reference radio bearer configurations.
				Use the PS paging procedure for testing of PS reference radio bearer configurations.
7	<-		ACTIVATE RB TEST MODE (DCCH)	TC
8	>		ACTIVATE RB TEST MODE COMPLETE (DCCH)	TC
9	<		RADIO BEARER SETUP (DCCH)	RRC In case the reference radio bearer configuration includes radio bearers for both PS and CS domain then the physical channel and PSCS radio bearer(s) are configured in the first RADIO BEARER SETUP message. CS-PS radio bearer(s) are configured in a second RADIO BEARER SETUP message, see step 10a.
10	>	>	RADIO BEARER SETUP COMPLETE (DCCH)	RRC
10a				In case the reference radio bearer configuration includes radio bearers for both PS and CS domain then repeat steps 9 and 10 to configure the <u>CS-PS</u> radio bearer(s)
11	<-	-	TRANSPORT FORMAT COMBINATION CONTROL (DCCH)	RRC Transport format combinations is limited to "Restricted UL TFCIs", as specified for the sub-test
12	<-	-	CLOSE UE TEST LOOP (DCCH)	TC UE test mode 1 RLC SDU size is for every active radio bearer set to "UL RLC SDU size", as specified for the sub-test.
13	>		CLOSE UE TEST LOOP COMPLETE (DCCH)	TC
14a	<- ;		Test data	SS sends continues test data in every TTI using the downlink transport format combination under test. The number of RLC SDUs and their sizes are specified in the actual test case. SS checks returned data
14b			Wait T1	SS continue to send data every TTI and check the returned data for time T1 T1 = 12 times the max TTI in the actual radio bearer combination under test
15a	<- ;	>	Test data (DTCH) +	SS continues sending test data in every TTI. SS sends a MEASUREMENT CONTROL
4.51	<-			message simultaneously to the test data requesting periodic reporting at interval T2
15b	< >		Test data (DTCH) +	SS continue to send data in every TTI and check the returned data for time 2xT2
	>		MEASUREMENT REPORT (DCCH)	SS checks that at least one MEASUREMENT REPORT message is received
16	<		OPEN UE TEST LOOP (DCCH)	TC
17	>	>	OPEN UE TEST LOOP COMPLETE (DCCH)	TC
18			Repeat steps 11 to 17 for every sub-test.	
19			RB RELEASE (DCCH)	RRC Optional step
20	<-	-	DEACTIVATE RB TEST MODE (DCCH)	TC Optional step
21	>		DEACTIVATE RB TEST MODE COMPLETE (DCCH)	TC Optional step

Tdoc **#***T1-020668*

3GPP TSG- T1 Meeting #17 Luton, England 4th – 8th Nov 2002

3GPP TSG- T1 SIG Meeting #25 Singapore, 18th – 20th September 2002

T1S-020582

CR-Form-v6.1						
CHANGE REQUEST						
^ж Т;	S 34.123-1	CR 317	ж геv - ^{ж (}	Current version: 5.1.1	ж	
	Spec Title:	User Equipment (UE) conformance spec	ification;	ж	
	-	Part 1: Protocol conf	ormance specificatio	on		
For <u>MELP</u> (on using this to	rm, see dottom of this	page or look at the	pop-up text over the X sy	MDOIS.	
Proposed chan	nge affects: ೫	(U)SIM ME/	UE X Radio Acc	ess Network Core No	etwork	
Title:	策 CR to 34.1	23-1 R5; Correction of	f package 1 test cas	se 8.1.1.7		
Source:	# Ericsson					
Work item code	e:# TEI			<i>Date:</i> ೫ <u>11/9/2002</u>		
WORK Rem COde						
Category: Reason for cha	F (cor A (con B (add C (fun D (edd Detailed ex be found in	the following categories rection) responds to a correctior dition of feature), actional modification of fe itorial modification) planations of the above 3GPP <u>TR 21.900</u> .	n in an earlier release) eature) categories can	R97 (Release 1997) R98 (Release 1998) R99 (Release 1999) REL-4 (Release 4) REL-5 (Release 5)		
Summary of change: # In Step 2 in expected sequence a clarifying text is added. In the corresponding specific message contents the CN domain is changed from CS to the the text: Set to a new CN Domain.						
Consequences not approved:	sif [#] If ch	anges are not approve	ed, UE might not be	properly tested.		
Clauses affecte	ed: ¥ Clau	ise 8.1.1.7				
Other specs affected:	Т	ther core specificatior est specifications &M Specifications	is ¥			
Other commen	ts: ೫ Affe	cts R99, REL-4, REL-	5			

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at: <u>http://www.3gpp.org/3G_Specs/CRs.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

8.1.1.7 Paging for Connection in connected mode (CELL_DCH)

8.1.1.7.1 Definition

8.1.1.7.2 Conformance requirement

When the UE receives a PAGING TYPE 2 message, it shall not affect the state of any other ongoing RRC procedures, when not stated otherwise elsewhere.

The UE shall:

- 1> indicate reception of paging; and
- 1> forward the IE "Paging cause" and the IE "Paging record type identifier" to upper layers.

...

In the UE, the initial direct transfer procedure shall be initiated, when the upper layers request establishment of a signalling connection. This request also includes a request for the transfer of a NAS message.

The UE shall, in the INITIAL DIRECT TRANSFER message:

•••

The UE shall:

- 1> transmit the INITIAL DIRECT TRANSFER message on the uplink DCCH using AM RLC on signalling radio bearer RB3;
- 1> when the INITIAL DIRECT TRANSFER message has been submitted to lower layers for transmission:
 - 2> confirm the establishment of a signalling connection to upper layers; and
 - 2> add the signalling connection with the identity indicated by the IE "CN domain identity" in the variable ESTABLISHED_SIGNALLING_CONNECTIONS.
- 1> when the successful delivery of the INITIAL DIRECT TRANSFER message has been confirmed by RLC:
 - 2> the procedure ends.

Reference

3GPP TS 25.331 clause 8.1.8.2, 8.1.11.

8.1.1.7.3 Test purpose

To confirm that the UE responds to a PAGING TYPE 2 message which includes the IE "Paging Cause" and the IE "Paging Record Type Identifier".

To confirm that the UE responds with a RRC STATUS message after it has received an invalid PAGING TYPE 2 message.

8.1.1.7.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: CELL_DCH state (state 6-9 or state 6-10) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE, after executing a location registration or attach procedure followed by the release of the TMSI of P-TMSI allocated. The UE has been registered in both CS and PS domains.

Test Procedure

The SS transmits an invalid PAGING TYPE 2 message. UE shall respond by transmitting a RRC STATUS message on the uplink DCCH using RLC-AM mode. Finally, SS transmits a PAGING TYPE 2 message, which includes a matched Paging Record Type Identifier. Then the UE shall responds to this message by the transmission of an INITIAL DIRECT TRANSFER message.

Expected sequence

Step	Direction		Message	Comment
-	UE	SS	_	
1			Void	
2	*	<u></u>	PAGING TYPE 2	SS pages UE from a new CN domain, see specific message contents. See message content.
3	-	>	RRC STATUS	The UE shall respond by reporting the protocol error to the SS.
4	÷	<u>,</u>	PAGING TYPE 2	SS pages the UE with a matched identifier and with a valid "paging cause" IE from a new CN Domain.
5	-	>	INITIAL DIRECT TRANSFER	The UE shall respond to the paging message sent in step 4.

Specific Message Contents

PAGING TYPE 2 (Step 2)

SS sends a message containing a protocol error causing the UE to perform procedure specific error handling.

Information Element	Value/remark
Message Type	
RRC transaction identifier	Arbitrarily selects an integer between 0 and 3
Integrity check info	The presence of this IE is dependent on IXIT
	statements in TS 34.123-2. If integrity protection is
	indicated to be active, this IE is present with the values
	of the sub IEs as stated below. Else, this IE and the
	sub-IEs are omitted.
- Message authentication code	SS calculates the value of MAC-I for this message and writes to this IE.
- RRC Message sequence number	SS provides the value of this IE, from its internal counter.
Paging Cause	Set to value "Spare"
CN Domain Identity	Set to a new CN Domain CS
Paging Record Type Identifier	Set to "IMSI (GSM-MAP)" for UEs supporting GSM-
	MAP core network type or "IMSI (DS-41)" for UEs
	supporting ANSI-41 core network type.

RRC STATUS (Step 3)

Use the same message type found in TS 34.108, clause 9, with the following exception.

Information Element	Value/remark
Identification of received message	
- Received message type	PAGING TYPE 2
- RRC transaction identifier	Checked to see if the value is identical to the same IE in the PAGING TYPE 2 message.
Protocol error information	
- Protocol Error Cause	Information element value not comprehended

PAGING TYPE 2 (Step 4)

Use the same message type found in TS 34.108, clause 9, with the following exception.

Information Element	Value/remark
Paging cause	Terminating Call supported by the UE
CN domain identity	Domain supported by the UE
Paging record type identifier	Set to "IMSI (GSM-MAP)" for UEs supporting GSM-
	MAP core network type or "IMSI (DS-41)" for UEs
	supporting ANSI-41 core network type.

INITIAL DIRECT TRANSFER (Step 5) - for UEs supporting GSM-MAP core networks

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
CN domain identity	CS domain or PS domain
Intra Domain NAS Node Selector	
- CHOICE version	R99
CHOICE CN type	GSM
CHOICE Routing basis	IMSI (response to IMSI paging)
Routing parameter	Bit string (10) consisting of DecimalToBinary [(IMSI div 10) mod 1000]. The bits of the result are numbered from b0 to b9, with bit b0 being the least significant.
Entered parameter	FALSE
NAS message	Not checked
START	Not checked
Measured results on RACH	Not checked

INITIAL DIRECT TRANSFER (Step 5) - for UEs supporting ANSI-41 core networks

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE is present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs are omitted.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
CN domain identity	CS domain or PS domain
Intra Domain NAS Node Selector	
- CHOICE version	ANSI-41 : Bitstring(14), all bits set to 0
NAS message	Not checked
START	Not checked
Measured results on RACH	Not checked

8.1.1.7.5 Test requirement

After step 2 the UE shall respond to the paging message by transmitting RRC STATUS on the DCCH, stating the protocol error as " Information element value not comprehended ".

After step 4 the UE shall respond to the paging message by transmitting an INITIAL DIRECT TRANSFER message on the uplink DCCH.

3GPP TSG-T1/SIG Meeting #25 Singapore, September 18th-20th, 2002

Tdoc T1S-020613

		CR-Form-v5.1						
	CHANGE REQUEST							
ж	34.123-1 CR 319 #r	ev - ^{# Current version:} 5.1.0 [#]						
For <u>HELP</u> or	For <u>HELP</u> on using this form, see bottom of this page or look at the pop-up text over the $#$ symbols.							
Proposed chang	ge affects: ೫ (U)SIM ME/UE	X Radio Access Network Core Network						
Title:	# CR to 34.123-1 REL-5; Modification	ns to package 1 RLC Test Cases						
Source:	# Anite Telecoms							
Work item code:	:# TEI	<i>Date:</i> ೫ <mark>11/9/2002</mark>						
Category:	 F Use <u>one</u> of the following categories: F (correction) A (corresponds to a correction in a B (addition of feature), C (functional modification of feature) D (editorial modification) Detailed explanations of the above categories be found in 3GPP <u>TR 21.900</u>. 	re) R97 (Release 1997) R98 (Release 1998) R99 (Release 1999)						
Reason for chan	of magnitude to the TTI of the	ner values to be tested which are of a similar order transport channels used does not permit the timer UE to be tested with sufficient confidence.						

Summary of change: ₩	Cummary of change: # In clause 7.2.3.19 use Timer_poll_periodic timer values of 500ms and 2000ms (instead of 100ms and 2000ms)		
	In clause 7.2.3.23 user Timer_poll_timer value of 600ms (instead of 200ms)		
	In clause 7.2.3.26 user Timer_status_periodic value of 400ms (instead of 100ms)		
Consequences if अ not approved:	The test prose cannot test the UE behaviour adequately.		
Clauses affected: #	7.2.3.19, 7.2.3.23, 7.2.3.26		
Other specs % affected:	Other core specifications # Test specifications # O&M Specifications •		
Other comments: #	Affects R99, REL-4 and REL-5 test cases.		

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.

- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

Expected sequence

Step	Direction	ion Message Comments		
-	UE SS			
1	÷	DOWNLINK RLC PDU	SDU 1	
2	\leftarrow	DOWNLINK RLC PDU	SDU 2	
3	÷		SS continues to transmit RLC SDUs	
4	÷	DOWNLINK RLC PDU	SDU 2P	
5	\rightarrow	UPLINK RLC PDU	SDU 1 Expanded to (4 *	
6	\rightarrow	UPLINK RLC PDU	AM_7_PayloadSize) - 1 bytes by test function	
7	\rightarrow		SS continues to receive RLC SDUs	
8	\rightarrow	UPLINK RLC PDU	SDU P, Poll	
9	\leftarrow	STATUS PDU		
10	\rightarrow	UPLINK RLC PDU	SDU P+1 Expanded to (4 *	
11	\rightarrow	UPLINK RLC PDU	AM_7_PayloadSize) - 1 bytes by test function	
12				
	\rightarrow		SS continues to receive RLC SDUs	
13	\rightarrow	UPLINK RLC PDU	SDU 2P, Poll	
14	RB RELEASE Optional step			
NOTE	NOTE 1: The Expected Sequence shown is infomative. The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity. Information such as SDU, PDU or Sequence numbers given in the comments column shall be considered informative only, for test case development purposes.			

7.2.3.18.5 Test requirements

The UE shall return uplink PDUs that contain polls for status in sequence numbers 4 * P - 1 and 8 * P - 1. No other PDUs shall poll for status.

7.2.3.19 Polling for status / Timer triggered polling (Timer_Poll_Periodic)

7.2.3.19.1 Definition

This case tests that the UE will poll for a status request every Timer_Poll_Periodic ms when that mode is enabled. Incorrect operation of polling will cause degradation of service, or at worst service failure.

7.2.3.19.2 Conformance requirement

This timer shall only be used when "timer based polling" is configured by upper layers. The value of the timer is signalled by upper layers. The timer shall be started when the RLC entity is created. When the timer expires, the RLC entity shall:

- restart the timer;
- if AMD PDUs are available for transmission or retransmission (not yet acknowledged):
 - trigger a poll.

[...]

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see TS 25.322 subclause 9.7.1):
 - if polling is not prohibited, see TS 25.322 subclause 9.5:

- set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clauses 9.5, 9.7.1 and 11.3.2.1.1.

7.2.3.19.3 Test purpose

- 1. To verify that the UE polls the SS in the next PDU to be transmitted or retransmitted each time the Timer_Poll_Periodic timer expires.
- 2. To verify that if there is no PDU to be transmitted, and all the PDUs have already been acknowledged, the timer is restarted, but no poll is sent.

7.2.3.19.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	First run	Second run
Polling info		
Last retransmission PDU poll	FALSE	FALSE
Last transmission PDU poll	FALSE	FALSE
Timer_poll_periodic	100 500	2000

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let T be the value of Timer_Poll_Periodic:

- a) The SS waits for at least 2*T ms before starting any transmissions, and monitors the uplink.
- b) The SS sends 4 * (T/TTI) RLC SDUs of size (AM_7_PayloadSize/2) 1 bytes to the UE. The SDUs are packed 2 SDUs to one PDU.
- c) The SS waits for the first PDU to be received with the P bit set, records the arrival time (T_1) and responds with a STATUS PDU normally.
- d) The SS waits for the reception of the next PDU with the P bit set, records the arrival time (T₂), and then transmits a STATUS PDU reporting that none of the uplink PDUs were correctly received, except for the last PDU containing the poll bit which is acknowledged.
- e) The SS waits for the next PDU received with the P bit set, and records the arrival time (T_3) .
- f) The SS waits for the reception of the next PDU with the P bit set and records the arrival time (T_4) .
- g) The SS may optionally release the radio bearer.

The Test is repeated using the parameters specified for the second run.

Step	Direction		Message	Comments	
	UE	SS			
1	÷	÷	DOWNLINK RLC PDU	SDU 1, SDU2	
2	•	÷		SS continues to transmit RLC SDUs	
3		÷	DOWNLINK RLC PDU	SDU 4*(T/TTI)-1SDU 4*(T/TTI)	
4		>	UPLINK RLC PDU	SDU 1, SN=0	
5	-	>	UPLINK RLC PDU	SDU 2, SN=1	
6	-	>		SS continues to receive RLC PDUs	
7	-)	UPLINK RLC PDU	$SN = x$, Poll: Note T_1	
8	•	÷	STATUS PDU	ACK SN 0 to SN x	
9	-	>	UPLINK RLC PDU	SN = x+1	
10	_	>		SS continues to receive RLC PDUs	
11		>	UPLINK RLC PDU	$SN = x + ceil(T/TTI)$, Poll: Note T_2	
12	•	÷	STATUS PDU	NAK SN x+1 to SN x + ceil(T/TTI)-1	
13		>	UPLINK RLC PDU	PDUs including some retransmissions	
14	\rightarrow		UPLINK RLC PDU		
15	-	>		SS continues to receive RLC PDUs	
16	_)	UPLINK RLC PDU	Poll: Note T ₃	
17		<u>.</u>	STATUS PDU	Normal	
18	-	>		SS continues to receive RLC PDUs	
19	-	>	UPLINK RLC PDU	Poll: Note T ₄	
20	RB RELEASE Optional step				
NOTE	NOTE 1: The Expected Sequence shown is infomative.				
	The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity.				
	Information such as SDU, PDU or Sequence numbers given in the comments column shall be				
NOTE	considered informative only, for test case development purposes.				
NOTE	IOTE 2: The value of x may be different for each iteration.				

7.2.3.19.5 Test requirements

No PDUs shall be received from the UE for 2*T ms before step 1.

 $x \ll ceil (T/TTI).$

Time $T_2 - T_1$ shall be T.

Time $T_3 - T_2$ shall be T.

Time $T_4 - T_3$ shall be T.

7.2.3.20 Polling for status / Polling on Poll_Window% of transmission window

7.2.3.20.1 Definition

This case tests that the UE will poll for a status request when it has reached Poll_Window% of the transmission window, when that mode is enabled. Incorrect operation of polling will cause degradation of service, or at worst service failure.

This test applies to all UE.

98

7.2.3.20.2 Conformance requirement

1. A poll is triggered for each AMD PDU when J≥Poll_Window, where J is the window transmission percentage defined by

$$J = \frac{(4096 + VT(S) + 1 - VT(A)) \mod 4096}{VT(WS)} * 100,$$

where the constant 4096 is the modulus for AM described in 3GPP TS 25.322 subclause 9.4 and VT(S) is the value of the variable before the AMD PDU is submitted to lower layer.

- 2. The Polling function is used by the Sender to request the peer RLC entity for a status report. The "Polling bit" in the AMD PDU indicates the poll request. There are several triggers for initiating the Polling function. Which of the triggers shall be used is configured by upper layers for each RLC entity. The following triggers can be configured:
 - •••••
 - 6) Window based.

The Sender triggers the Polling function when the condition described in subclause 9.6 d) ("Poll_Window") is fulfilled.

- 3. The Sender shall:
 - if a poll has been triggered by one or several poll triggers (see subclause 9.7.1):
 - if polling is not prohibited, see subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1".

Reference

25.322 clauses 9.6, 9.7.1 and 11.3.2.1.1.

7.2.3.20.3 Test purpose

To verify that the UE polls the SS when the window based polling condition J >= Poll_Window is fulfilled.

7.2.3.20.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	
Polling info	
Last transmission PDU poll	FALSE
Poll_Window	50
Transmission window size	8
Downlink RLC	
Receiving window size	8

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Release 5

Test procedure

Let W be the size of the transmission window.

- a) The SS transmits (W/2) + 2 RLC SDUs of size AM_7_PayloadSize 1 bytes.
- b) The SS checks the sequence number of the first three uplink PDUs received with the P bit set.
- c) The SS sends another RLC SDU of size AM_7_PayloadSize 1 bytes.
- d) The SS checks the sequence number of the next uplink PDU received with the P bit set.
- e) The SS waits until no more new PDUs are received.
- f) The SS sends a STATUS PDU acknowledging the received RLC PDUs with SN = 0 through W/2, followed by two further RLC SDUs.

100

- g) The SS checks the sequence number of the next uplink PDU received with the P bit set.
- h) The SS may optionally release the radio bearer.

Expected sequence

Step	Direction	Message	Comments	
	UE SS	_		
1	÷	DOWNLINK RLC PDU	SDU 1	
2	÷		SS continues to transmit RLC SDUs	
3 4 5	$\begin{array}{c} \leftarrow \\ \rightarrow \\ \rightarrow \end{array}$	DOWNLINK RLC PDU UPLINK RLC PDU UPLINK RLC PDU	SDU W/2+2 SDU 1, SN=0 SDU 2, SN=1	
6	\rightarrow		SS continues to receive RLC PDUs	
7	\rightarrow	UPLINK RLC PDU	SN = W/2-1, Poll	
7a	\rightarrow	UPLINK RLC PDU	SN = W/2, Poll	
7b	\rightarrow	UPLINK RLC PDU	SN = W/2 + 1, Poll	
8	(DOWNLINK RLC PDU	SDU W/2 + 3	
9	\rightarrow	UPLINK RLC PDU	SN = W/2 + 2, Poll	
9a			SS waits 10 TTI periodsto ensure no more	
10	,		new PDUs are received	
10	÷	STATUS PDU	ACK SN 0 to W/2 (UE sets VT(A) to W/2+1)	
11	÷	DOWNLINK RLC PDU	SDU W/2 + 4	
12	÷	DOWNLINK RLC PDU	SDU W/2 + 5	
13	\rightarrow	UPLINK RLC PDU	SN = W/2+3	
14	\rightarrow	UPLINK RLC PDU	SN = W/2+4, Poll	
15		RB RELEASE Optional step		
NOTE				
1	The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity.			
	Information such as SDU, PDU or Sequence numbers given in the comments column shall be			
	considered informative only, for test case development purposes.			

7.2.3.20.5 Test requirements

The SS shall receive RLC PDUs with the P bit set in PDUs with sequence numbers of 3, 4, 5, 6 and 8. No other PDUs shall have their P bits set.

7.2.3.21 Polling for status / Operation of Timer_Poll timer / Timer expiry

7.2.3.21.1 Definition

This case tests that the UE will retransmit a poll for status if it does not receive a STATUS PDU within Timer_Poll ms after a poll for status is transmitted. Incorrect operation of polling will cause degradation of service, or possible service failure.

7.2.3.21.2 Conformance requirement

Timer_Poll.

This timer shall only be used when so configured by upper layers. The value of the timer is signalled by upper layers. In the UE this timer shall be started when the successful or unsuccessful transmission of an AMD PDU containing a poll is indicated by lower layer. In UTRAN it should be started when an AMD PDU containing a poll is submitted to lower layer. If x is the value of the state variable VT(S) after the poll was submitted to lower layer, the timer shall be stopped upon receiving:

- positive acknowledgements for all the AMD PDUs with "Sequence Number" up to and including x 1; or
- a negative acknowledgement for the AMD PDU with "Sequence Number" = x 1.

If the timer expires and no STATUS PDU fulfilling the criteria above has been received:

- the Receiver shall be polled once more;
- the timer shall be restarted; and
- the new value of VT(S) shall be saved.

If a new poll is sent when the timer is active, the timer shall be restarted at the time specified above, and the value of VT(S) shall be saved.

[...]

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see subclause 9.7.1):
 - if polling is not prohibited, see subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clauses 11.3.2.1.1 and 11.3.4.1.

7.2.3.21.3 Test purpose

To verify that if the timer expires and no STATUS PDU containing an acknowledgement or negative acknowledgement of the AMD PDUs up to that which triggered the timer has been received, the receiver is polled once more.

7.2.3.21.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	First run	Second run
Polling info		
Last transmission PDU poll	FALSE	FALSE
Timer_poll	500	1000
Timer_Poll_Periodic	2000	2000

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let T be the value of the Timer_Poll_Periodic timer.

- a) The SS transmits at least 2 * T / TTI SDUs of size AM_7_PayloadSize 1 bytes.
- b) The SS receives PDUs from the UE, and notes the time on receiving the first PDU with the P bit set, but does not respond. This time will be recorded as T_{1.}
- c) The SS continues to receive PDUs from the UE and notes the time on receipt of the next PDU with the P bit set. This time will be recorded as T_2 .
- d) The SS may optionally release the radio bearer.

The test case is run once for each set of initial RLC parameters.

Expected sequence

Step	Direction		Message	Comments	
-	UE	SS			
1	÷	-	DOWNLINK RLC PDU	SDU 1	
2	÷	-		SS continues to transmit RLC SDUs	
3	÷	-	DOWNLINK RLC PDU	SDU ceil(2T/TTI)	
4		>	UPLINK RLC PDU	SDU 1	
5		>	UPLINK RLC PDU	SDU 2	
6	÷	>		SS continues to receive RLC PDUs	
7		>	UPLINK RLC PDU	SN = ceil(T/TTI), Poll: Note T ₁	
8		>	UPLINK RLC PDU	SN = ceil(T/TTI)+1	
9	÷	>		SS continues to receive RLC PDUs	
10	\rightarrow UPLINK RLC PDU Poll: Note T ₂		Poll: Note T ₂		
11			RB RELEASE Optional step		
NOTE	NOTE: The Expected Sequence shown is infomative.				
	The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity.				
	Information such as SDU, PDU or Sequence numbers given in the comments column shall be				
	considered informative only, for test case development purposes.				

7.2.3.21.5 Test requirements

For the first run, the measured time $T_2 - T_1$ shall be 500 ms.

For the second run, the measured time $T_2 - T_1$ shall be 1000 ms.

7.2.3.22 Polling for status / Operation of Timer_Poll timer / Stopping Timer_Poll timer

7.2.3.22.1 Definition

This case tests that the UE will stop the Timer_Poll timer if it receives a STATUS PDU within Timer_Poll ms after a poll for status is transmitted. Incorrect operation of polling will cause degradation of service, or possible service failure.

7.2.3.22.2 Conformance requirement

Timer_Poll.

This timer shall only be used when so configured by upper layers. The value of the timer is signalled by upper layers. In the UE this timer shall be started when the successful or unsuccessful transmission of an AMD PDU containing a poll is indicated by lower layer. In UTRAN it should be started when an AMD PDU containing a poll is submitted to lower layer. If x is the value of the state variable VT(S) after the poll was submitted to lower layer, the timer shall be stopped upon receiving:

- positive acknowledgements for all the AMD PDUs with "Sequence Number" up to and including x 1; or
- a negative acknowledgement for the AMD PDU with "Sequence Number" = x 1.

If the timer expires and no STATUS PDU fulfilling the criteria above has been received:

- the Receiver shall be polled once more;
- the timer shall be restarted; and
- the new value of VT(S) shall be saved.

If a new poll is sent when the timer is active, the timer shall be restarted at the time specified above, and the value of VT(S) shall be saved.

[...]

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see subclause 9.7.1):
 - if polling is not prohibited, see subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clause 9.5.

7.2.3.22.3 Test purpose

To verify that the Timer_Poll timer is stopped when receiving a STATUS PDU that acknowledges all AMD PDUs with SN up to and including VT(S)-1 at the time the poll was transmitted.

7.2.3.22.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	First run	Second run
Polling info		
Last transmission PDU poll	FALSE	FALSE
Timer_poll	500	1000
Timer_Poll_Periodic	2000	2000

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let T be the value of the Timer_Poll_Periodic timer.

- a) The SS transmits at least 2 * T / TTI SDUs of size AM_7_PayloadSize 1 bytes.
- b) The SS receives PDUs from the UE, and notes the time on receiving the first PDU with the P bit set. This time will be recorded as $T_{1.}$
- c) The SS sends a STATUS PDU acknowledging all the PDUs up to and including the PDU carrying the poll request.
- d) The SS continues to receive PDUs from the UE and notes the time on receipt of the next PDU with the P bit set. This time will be recorded as T_2 .
- e) The SS may optionally release the radio bearer.

The test case is run once for each set of initial RLC parameters.

Expected sequence

Step	p Direction		Message	Comments
-	UE	SS		
1	÷	-	DOWNLINK RLC PDU	SDU 1
2	÷	-		SS continues to transmit RLC SDUs
3	÷	-	DOWNLINK RLC PDU	SDU ceil(2T/TTI)
4		>	UPLINK RLC PDU	SDU 1
5		>	UPLINK RLC PDU	SDU 2
6	÷	>		SS continues to receive RLC PDUs
7		>	UPLINK RLC PDU	SN = ceil(T/TTI), Poll: Note T ₁
8	÷	-	STATUS PDU	ACK SN 0 to SN ceil(T/TTI)
9	7	>	UPLINK RLC PDU	SN = ceil(T/TTI)+1
10	÷	>		SS continues to receive RLC PDUs
11		>	UPLINK RLC PDU	SN = ceil(2T/TTI), Poll: Note T_2
12				
NOTE	NOTE 1: The Expected Sequence shown is infomative.			
	The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity.			
	Information such as SDU, PDU or Sequence numbers given in the comments column shall be			
	considered informative only, for test case development purposes.			

7.2.3.22.5 Test requirements

For both execution runs, the measured time $T_2 - T_1$ shall be 2000 ms.

7.2.3.23 Polling for status / Operation of Timer_Poll timer / Restart of the Timer_Poll timer

7.2.3.23.1 Definition

This case tests that the UE will restart the Timer_Poll timer if another poll request is transmitted whilst the timer is running. Incorrect operation of polling will cause degradation of service, or possible service failure.

This test applies to all UE.

7.2.3.23.2 Conformance requirement

Timer_Poll.

This timer shall only be used when so configured by upper layers. The value of the timer is signalled by upper layers. In the UE this timer shall be started when the successful or unsuccessful transmission of an AMD PDU containing a poll is indicated by lower layer. In UTRAN it should be started when an AMD PDU containing a poll is submitted to lower layer. If x is the value of the state variable VT(S) after the poll was submitted to lower layer, the timer shall be stopped upon receiving:

- positive acknowledgements for all the AMD PDUs with "Sequence Number" up to and including x 1; or
- a negative acknowledgement for the AMD PDU with "Sequence Number" = x 1.

If the timer expires and no STATUS PDU fulfilling the criteria above has been received:

- the Receiver shall be polled once more;
- the timer shall be restarted; and
- the new value of VT(S) shall be saved.

If a new poll is sent when the timer is active, the timer shall be restarted at the time specified above, and the value of VT(S) shall be saved.

[...]

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see subclause 9.7.1):
 - if polling is not prohibited, see subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clause 9.5.

7.2.3.23.3 Test purpose

To verify that if a new poll is sent when the timer is running it is restarted.

7.2.3.23.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

1

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	
Polling info	
Last transmission PDU poll	FALSE
Timer_poll	200<u>600</u>
Poll_Window	60
Poll_PDU	16
Transmit window size	32

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let T be the value of the Timer_Poll timer.

- a) The SS starts transmission of $2*Poll_PDU + ceil(T / TTI)$ numbers of SDUs of size AM_7_PayloadSize 1bytes.
- b) Whilst transmitting, the SS receives PDUs from the UE, until it receives the second PDU with the P bit set. (Note: poll due to Poll_Window).
- c) The SS sends a STATUS PDU acknowledging all the PDUs received so far.
- d) The SS continues to receive consecutive PDUs with the poll bit set until a PDU is received without the poll bit set. The time the last PDU with the poll bit set was received is recorded as T₁.
- e) The SS continues to receive PDUs from the UE and notes the time on receiption of the next PDU with the P bit set. This time will be recorded as T₂.
- f) The SS may optionally release the radio bearer.

Step	Direction Message Comments						
	UE SS						
1	÷	DOWNLINK RLC PDU	SDU 1				
0	/						
2	÷		SS continues to transmit RLC SDUs				
3	÷	DOWNLINK RLC PDU	2* Poll_PDU + ceil(T / TTI) numbers of				
			SDUs are sent.				
4	\rightarrow	UPLINK RLC PDU	SDU 1				
5	\rightarrow	UPLINK RLC PDU	SDU 2				
6	\rightarrow		SS continues to receive RLC PDUs				
_							
7	\rightarrow	UPLINK RLC PDU	SN = poll_PDU - 1, 1 st Poll, Timer_Poll				
			started				
8	\rightarrow	Void					
9 10	\rightarrow	Void UPLINK RLC PDU	SS continues to receive RLC PDUs				
10	7	UPLINK REC PDU					
11	\rightarrow	UPLINK RLC PDU	SN= ceil(Tx_Window_Size * 60%)-1, 2 nd				
			Poll, Timer_Poll restarted: Note T1				
12	÷	STATUS PDU	ACK SN 0 to SN = ceil(Tx_Window_Size *				
4.0	``		60%)-1				
13	\rightarrow	UPLINK RLC PDU	SN = ceil(Tx_Window_Size * 60%), Poll,				
14			Timer_Poll restarted: Note T1				
14	\rightarrow		SS continues to receive RLC PDUs. If poll				
	,		bit is set in consecutive PDUs then note new				
			value of T1 until a PDU without a poll bit set				
			is received.				
45							
15	\rightarrow	UPLINK RLC PDU	SN = ceil(Tx_Window_Size * 60%) + ceil(T / TTI) -1, 3^{rd} Poll, Timer_Poll expired: Note T ₂				
16		RB RELEASE	Optional step				
NOTE	· The Eyr	ected Sequence shown is infomative.					
		LINK and DOWNLINK PDU flows may over	rlap, but are shown separate for clarity.				
			pers given in the comments column shall be				
	considered informative only, for test case development purposes.						

7.2.3.23.5 Test requirements

The measured time $T_2 - T_1$ shall be $\frac{200}{600}$ ms.

7.2.3.24 Polling for status / Operation of timer Timer_Poll_Prohibit

7.2.3.24.1 Definition

This case tests that the UE will not send a poll request within Timer_Poll_Prohibit ms of a previous poll request when this mode of operation is enabled. Incorrect operation of polling will cause degradation of service, or possible service failure.

7.2.3.24.2 Conformance requirement

The timers defined in this subclause are normative. The timers shall be considered active from the time they are started until the time they either expire or are stopped.

b) Timer_Poll_Prohibit.

This timer shall only be used when so configured by upper layers. It is used to prohibit transmission of polls within a certain period. The value of the timer is signalled by upper layers.

In the UE this timer shall be started when the successful or unsuccessful transmission of an AMD PDU containing a poll is indicated by lower layer. In UTRAN it should be started when an AMD PDU containing a poll is submitted to lower layer.

From the time a poll is triggered until the timer expires, polling is prohibited. If another poll is triggered while polling is prohibited, its transmission shall be delayed until the timer expires (see subclause 9.7.1). Only one poll shall be transmitted when Timer_Poll_Prohibit expires even if several polls were triggered in the meantime. This timer shall not be affected by the reception of STATUS PDUs.

When Timer_Poll_Prohibit is not configured by upper layers, polling is never prohibited.

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see subclause 9.7.1):
 - if polling is not prohibited, see subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clauses 9.5, 9.7.1 and 11.3.2.1.1.

7.2.3.24.3 Test purpose

- 1. To verify that no poll is transmitted if one or several polls are triggered when the Timer_Poll_Prohibit timer is active and has not expired.
- 2. To verify that the UE polls only once after Timer_Poll_Prohibit expires even though triggered several times during the prohibit time.

7.2.3.24.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	
Polling info	
Timer_poll_prohibit	500
Last transmission PDU poll	FALSE
Poll_PDU	4
Poll_Window	50
Transmission window size	32
Downlink RLC	
Receiving window size	128

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let T be the value of the Timer_Poll_Prohibit timer.

a) The SS starts transmission of at least $(2*Poll_PDU) + ceil(T / TTI)$ SDUs of size AM_7_PayloadSize - 1 bytes.

- b) Whilst transmitting, the SS receives PDUs from the UE, and notes the time on receiving the first PDU with the P bit set. This time will be recorded as T₁.
- c) The SS does not respond to the poll request.
- d) The SS continues to receive PDUs from the UE and notes the time on receipt of the next PDU with the P bit set. This time will be recorded as T_2 .
- e) The SS may optionally release the radio bearer.

Step	Direction	Message	Comments			
	UE SS					
1	÷	DOWNLINK RLC PDU	SDU 1			
2	(SS continues to transmit RLC SDUs			
3	÷	DOWNLINK RLC PDU	SDU (2*Poll_PDU)+ ceil(T/TTI)			
4	\rightarrow	UPLINK RLC PDU	SDU 1			
5	\rightarrow	UPLINK RLC PDU	SDU 2			
6	\rightarrow		SS continues to receive RLC PDUs			
7	\rightarrow	UPLINK RLC PDU	SN = Poll_PDU - 1, Poll: Note T ₁			
8	\rightarrow		SS continues to receive RLC PDUs			
9 10	→	Void UPLINK RLC PDU	SN = (Transmission Window Size / 2) – 1, No Poll			
11	\rightarrow		SS continues to receive RLC PDUs			
12	\rightarrow	UPLINK RLC PDU $SN = Poll_PDU + ceil(T/TTI) - 1, P$ T ₂				
12a	\rightarrow	→ SS continues to receive R				
13		RB RELEASE	Optional step			
NOTE	NOTE: The Expected Sequence shown is infomative.					
		LINK and DOWNLINK PDU flows may over				
			pers given in the comments column shall be			
	considered informative only, for test case development purposes.					

7.2.3.24.5 Test requirements

- 1. The measured time $T_2 T_1$ shall be Timer_poll_prohibit ms.
- 2. Only one poll shall be received from the UE after step 7, the poll in step 12.
- 3. After step 12 no further poll shall be received from the UE for the next Timer_poll_prohibit ms.

7.2.3.25 Receiver Status Triggers / Detection of missing PDUs

7.2.3.25.1 Definition

This case tests that the UE transmits a status report whenever it detects that a PDU is missing, if this mode of operation is enabled. Incorrect operation of status reporting will cause degradation of service, or possible service failure.

7.2.3.25.2 Conformance requirement

Detection of missing PDU(s).

If the Receiver detects one or several missing AMD PDUs it shall trigger the transmission of a status report to the Sender.

Reference

TS 25.322 clause 9.7.2.

7.2.3.25.3 Test purpose

To verify that a status report is transmitted if there are one or more missing PDUs.

7.2.3.25.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

These settings apply to both the uplink and downlink DTCH.

Test procedure

- a) The SS transmits 7 SDUs, each of size AM_7_PayloadSize 1 bytes, in PDUs with consecutive sequence numbers starting from 0, followed by 5 SDUs in PDUs with consecutive sequence numbers starting from 8, followed by an SDU in a PDU with a sequence number of 15.
- b) While transmitting, the SS monitors the uplink for STATUS PDUs.
- c) The SS may optionally release the radio bearer

Expected sequence

Step	Direction	on	Message	Comments		
_	UE S	SS	-			
1	÷		DOWNLINK RLC PDU	SN = 0		
2	÷			SS continues to transmit RLC SDUs		
3	÷	\leftarrow DOWNLINK RLC PDU SN = 6		SN = 6		
4	←		DOWNLINK RLC PDU	SN = 8		
5	\rightarrow		STATUS PDU	SN = 7 missing		
6	÷		DOWNLINK RLC PDU			
7	÷		SS continues to transmit RLC SDUs			
8	÷	DOWNLINK RLC PDU SN = 12		SN = 12		
9	÷		DOWNLINK RLC PDU SN = 15			
10	\rightarrow		STATUS PDU SN = 7, 13, 14 missing			
11			RB RELEASE Optional step			
NOTE	NOTE: The Expected Sequence shown is infomative.					
	The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity.					
			· · · · ·	ers given in the comments column shall be		
	cons	sider	ed informative only, for test case developm	nent purposes.		

7.2.3.25.5 Test requirements

A STATUS PDU shall be received from the UE after step 4, indicating that the PDU with sequence number 7 was missing.

A STATUS PDU shall be received from the UE after step 9, indicating that the PDUs with sequence numbers 7, 13 and 14 were missing.

7.2.3.26 Receiver Status Triggers / Operation of timer Timer_Status_Periodic

7.2.3.26.1 Definition

This case tests that the UE transmits a status report every Timer_Status_Periodic ms when this mode of operation is enabled. Incorrect operation of status reporting will cause degradation of service, or possible service failure.

7.2.3.26.2 Conformance requirement

This timer shall only be used when timer based status reporting is configured by upper layers.

This timer shall be started when the RLC entity is created. When the timer expires the transmission of a status report shall be triggered and the timer shall be restarted.

Reference

TS 25.322 clauses 9.5, 9.7.2 and 11.5.2.

7.2.3.26.3 Test purpose

To verify that a status report is transmitted each time the Timer_Status_Periodic timer expires.

7.2.3.26.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Downlink RLC	
Timer_STATUS_periodic	100400

These settings apply to both the uplink and downlink DTCH.

Test procedure

Let T be the value of the Timer_STATUS_periodic timer.

- a) The SS starts transmission of at least ceil(2 * T / TTI) SDUs of size AM_7_PayloadSize 1 bytes.
- b) The SS waits to receive a STATUS PDU and notes the time. This time will be recorded as T_1 .
- c) The SS waits to receive a second STATUS PDU and notes the time. This time will be recorded as T_2 .
- d) The SS may optionally release the radio bearer.

Step	Direction		Message	Comments		
	UE	SS				
1	÷	<u>.</u>	DOWNLINK RLC PDU	SDU 1		
2	•	<u>-</u>		SS continues to transmit RLC SDUs		
3	•	<u>.</u>	DOWNLINK RLC PDU	SDU m		
4	-	>	STATUS PDU	Note T ₁		
5	•	<u></u>	DOWNLINK RLC PDU			
6	•	<u>,</u>		SS continues to transmit RLC SDUs		
7	•	<u>.</u>	DOWNLINK RLC PDU	SDU m + ceil(T/TTI)		
8	-	>	STATUS PDU	Note T ₂		
9		RB RELEASE Optional step		Optional step		
NOTE	1: TI	ne Exp	ected Sequence shown is infomative.			
	The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity.					
				pers given in the comments column shall be		
	CC	onside	red informative only, for test case developn	nent purposes.		

7.2.3.26.5 Test requirements

The measured time $T_2 - T_1$ shall be $\frac{100-400}{2}$ ms.

Tdoc **#**T1S-020638

			CHANGE	REQ	UE	ST	•		CR-Form-v7
ж	34.123-1	CR	321	жrev	-	ж	Current version:	5.1.0	ж
Far H	ELP on using this for		- h - ((f ()- :-		1	<u></u>		460 99 01 11	ah ala

For <u>HELP</u> on using this form, see bottom of this page or look at the pop-up text over the **#** symbols.

ME X Radio Access Network Core Network

Title:	ж	Corrections to MAC Package 1 test cases 7.1.1.2, 7	7.1.1.3, 7.1.1	.4,7.1.1.5 and 7.1.1.8
Source:	ж	Motorola and MCC Task 160		
Work item code.	: X	TEI	<i>Date:</i> ೫	12/09/2002
Category:	ж	F	Release: ೫	REL-5
		Use <u>one</u> of the following categories:	Use <u>one</u> of	the following releases:
		F (correction)	2	(GSM Phase 2)
		A (corresponds to a correction in an earlier release)	R96	(Release 1996)
		B (addition of feature),	R97	(Release 1997)
		C (functional modification of feature)	R98	(Release 1998)
		D (editorial modification)	R99	(Release 1999)
		Detailed explanations of the above categories can	Rel-4	(Release 4)
		be found in 3GPP TR 21.900.	Rel-5	(Release 5)
			Rel-6	(Release 6)

Reason for change: # 7.1.1.2:

TCTF fields for BCCH and CCCH cannot be used, as the UE MAC layer will not reject the PDU with them, but rather forward it to the BCCH and CCCH RLC entities, respectively.

7.1.1.2,7.1.1.3, 7.1.1.4, 7.1.1.5 & 7.1.1.8

Expiry of NAS Timers (MM T3240 or GMM T3317) will cause UE to transmit Signalling Connection Release Indication.

With RRC Delays comparable to that of RLC Delays, and further SRB 2 having higher Priority over SRB 3, RRC Status message and RLC status PDU can be received in any order.

Summary of change: # 7.1.1.2

Invalid combinations of TCTF corrected

7.1.1.2, 7.1.1.3, 7.1.1.4, 7.1.1.5 & 7.1.1.8

	Updated test procedure and comments in step 2a of the test sequence to verify that the UE MAC Layer has discarded the MAC PDU with an invalid header. Added note making it clear that the RLC status PDU and RRC Status message can be received in any order.
Consequences if not approved:	* Good UE will fail the test
Clauses affected:	# 7.1.1.2, 7.1.1.3, 7.1.1.4, 7.1.1.5, 7.1.1.8
Other specs affected:	Y N % X Test specifications X O&M Specifications
Other comments:	# Affects R99, REL-4 and REL-5 test cases.

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.

3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

7.1.1.2 DTCH or DCCH mapped to RACH/FACH / Invalid TCTF

7.1.1.2.1 Definition

This tests that the MAC applies the correct header to the MAC PDU according to the type of logical channel carried on the RACH/FACH transport channel. Incorrect application of MAC headers would result in inoperation of the UE.

7.1.1.2.2 Conformance requirement

DTCH or DCCH mapped to RACH/FACH:

TCTF field, C/T field, UE-Id type field and UE-Id are included in the MAC header. The following fields are defined for the MAC header:

- Target Channel Type Field

...

Coding of the Target Channel Type Field on FACH for FDD

TCTF	Designation
00	BCCH
01000000	СССН
01000001-	Reserved
01111111	(PDUs with this coding
	will be discarded by this
	version of the protocol)
1000000	СТСН
1000001-	Reserved
10111111	(PDUs with this coding
	will be discarded by this
	version of the protocol)
11	DCCH or DTCH
	over FACH

Reference(s)

TS 25.321 clauses 9.2.1 and 9.2.1.1 c).

7.1.1.2.3 Test purpose

- 1. To verify that the UE discards PDUs with reserved or incorrect values in the TCTF field.
- 2. To verify that the TCTF field, C/T field, UE-Id type and UE-Id field are correctly applied when a DTCH or DCCH is mapped to the RACH/FACH.

7.1.1.2.4 Method of test

Initial conditions

System Simulator:

1 cell, default parameters, Ciphering Off.

The SCCPCH is configured as specified in TS 34.108 clause 6.10.2.4.3.3 (Interactive/Background 32 kbps RAB + SRB for PCCH + SRB for CCCH + SRB for DCCH + SRB for BCCH) with the following exceptions for the FACH:

Higher	RAB/signalli	RB#3 (SRB#3)	
layer	User of Radio Bearer		Test
RLC	Logical channel type		DCCH
	RLC mode		TM
	Payload size	es, bit	168
	Max data rat	e, bps	33600 (alt.
			50400)
	RLC header,	bit	0
MAC	MAC header, bit		0 (note)
MAC	MAC multiple	exing	Simulated by SS
Layer 1	TrCH type		FACH
	TB sizes, bit		168
		TF0, bits	0 x 168
	TES	TF1, bits	1 x 168
	115	TF2, bits	2 x 168
		TF3, bits	N/A (alt. 3 x 168)
	TTI, ms		10
	Coding type		CC 1/2
	CRC, bit		16
	Max number of bits/TTI		752 (alt. 1136)
	before rate matching		
	RM attribute		200-240
NOTE:	NOTE: The SS MAC layer must be configured not to add a MAC header so that the header can be added by the test case in order to create the necessary invalid values.		

and using the configuration in TS 34.108 clause 6.10.2.4.3.3 for the PCH.

The TFCS should be configured as specified in clause 6.10.2.4.3.3.1.4. User Equipment:

The UE shall operate under normal test conditions, Ciphering Off.

The Test-USIM shall be inserted.

The SS starts broadcasting the System Information as specified in TS 34.108 clause 6.1, using the configuration for the PRACH and SCCPCH (signalled in SYSTEM INFORMATION 5) as follows:

- The SCCPCH is configured as specified in TS 34.108 clause 6.10.2.4.3.3 (Interactive/Background 32 kbps RAB + SRB for PCCH + SRB for CCCH + SRB for DCCH + SRB for BCCH).
- 2. The PRACH is configured as specified in TS 34.108 clause 6.10.2.4.4.1.

The SS follows the procedure in TS 34.108 clause 7.4.2.1 (Mobile Terminated) so that the UE shall be in state BGP 6-2 (CS-CELL_FACH_INITIAL).

Test procedure

- a) The SS receives the PAGING RESPONSE message from the UE and checks the TCTF field.
- b) The SS transmits MAC PDUs containing RLC AM PDUs containing a DIRECT TRANSFER message containing an AUTHENTICATION REQUEST message.
 - 1. Dummy octet string for NAS Message, of size sufficient enough to fit in one RLC PDU of 144 bits, including the correct RLC AM header.
 - 2. The IE CN Domain Identity is Set to PS Domain (no signalling connection for this domain exists).
 - 3. The polling bit in RLC header is set for transmission of RLC STATUS PDU.

Field	Value	
TCTF	<u>01000001</u> 00 'B	
UE ID Type	C-RNTI	
UE ID	As set in RRC CONNECTION	
	SETUP message.	
C/T	Logical Channel ID for SRB #3 (AM-	
	DCCH NAS High Priority)	

Where a TCTF size of 8-bits is used, 6-bits from the RLC payload shall be discarded.

- c) The SS_-checks that UE shall neither transmit RRC Status message on SRB2 nor RLC Status PDU on SRB3monitors the RACH for 10 s to ensure that no RACH access occurs.
- d) The SS again transmits MAC PDUs as in b) above, but this time uses the correct TCTF of 11'B. The sequence numbers in the RLC headers shall be identical with those sent in b).
- e) SS Receives RLC Status PDU on SRB #3 acknowledging the receipt of the above RLC PDU.
- f) The SS receives a RRC STATUS message on the uplink DCCH using AM RLC on SRB # 2.
- g) The SS repeats steps b), c), d) e) and f), with the TCTF field set as follows in step b):

Iteration	TCTF Value
2	01000000001111111<u>'</u>BB
3	01000001'B1000000'B
4	10000000'<u>1000001'</u>B
5	10000001<u>10111111</u>'В

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	\rightarrow	PAGING RESPONSE	Check TCTF
2	\leftarrow	MAC PDU(TCTF, UE-ID, C/T, RLC AM	Sent with incorrect TCTF =
		PDU(SN=x, DIRECT TRANSFER))	<u>01000001'B, 01111111'B,</u>
			1000000'B, 10000001'B, or
			<u>10111111'B00'B, 0100 0000'B,</u>
			0100 0001'B, 1000 0000'B, or 100
			0001'B,
2a		wait for T = 10 s	SS monitors for RACH access
			attempts checks that UE shall
			neither transmit RRC-Status
			message on SRB 2 nor RLC
			Status PDU on SRB 3
			See note 1 below
3	÷	MAC PDU(TCTF, UE-ID, C/T, RLC AM	Sent with correct TCTF = 11'B
		PDU(SN=x, DIRECT TRANSFER))	
4	\rightarrow	RLC-STATUS-PDU	ACK PDUs with SN = x and TCTF
			Field is recognised as correct for
			the DCCH.
			See note 2 below
5	\rightarrow	RRC Status PDUmessage	
1: UE v	will Transı	nit Signalling Connection Release Indication on	n expiry of MM Timer T3240

Note 1: UE will Transmit Signalling Connection Release Indication on expiry of MM Timer T3240 or GMM Timer T3317

Note 2: RRC Status message may be received before RLC Status PDU.

Steps 2-5 of above expected sequence are repeated for iterations 2 to 5. Note: For iteration k the SN in steps 2 and 4 starts with x + (k-1).

Specific Message Contents

None

7.1.1.2.5 Test Requirement

In step a) the TCTF field should have the value 00'B. Note that this may be implied from receipt of the PAGING RESPONSE message correctly by the SS test script. During the test the SS shall request an RLC status report with every transmitted PDU by setting of the Polling Bit. The UE shall not send any STATUS PDUs indicating missing PDUs.

At the end of each iteration (steps 4 and 5 of expected sequence) the SS shall receive an RRC Status <u>PDU message</u> on SRB # 2, and RLC Status PDU on SRB # 3 with TCTF field set to value '01' B.

7.1.1.3 DTCH or DCCH mapped to RACH/FACH / Invalid C/T Field

7.1.1.3.1 Definition

This tests that the MAC applies the correct header to the MAC PDU according to the type of logical channel carried on the RACH/FACH transport channel. Incorrect application of MAC headers would result in inoperation of the UE.

7.1.1.3.2 Conformance requirement

DTCH or DCCH mapped to RACH/FACH:

TCTF field, C/T field, UE-Id type field and UE-Id are included in the MAC header. The following fields are defined for the MAC header:

- C/T field

The C/T field provides identification of the logical channel instance when multiple logical channels are carried on the same transport channel...

C/T field	Designation
0000	Logical channel 1
0001	Logical channel 2
1110	Logical channel 15
1111	Reserved
	(PDUs with this coding will be
	discarded by this version of
	the protocol)

Structure of the C/T field

Reference(s)

TS 25.321 clauses 9.2.1 and 9.2.1.1 c).

7.1.1.3.3 Test purpose

- 1. To verify that the UE discards PDUs with reserved or incorrect values in C/T field.
- 2. To verify that the TCTF field, C/T field, UE-Id type and UE-Id field are correctly applied when a DTCH or DCCH is mapped to the RACH/FACH.

7.1.1.3.4 Method of test

Initial conditions

System Simulator: See clause 7.1.1.2.4.

User Equipment: See clause 7.1.1.2.4.

Test procedure

- a) The SS receives the PAGING RESPONSE message from the UE and checks the C/T field.
- b) The SS transmits MAC PDUs containing RLC AM PDUs containing a DIRECT TRANSFER message containing.
 - 1. Dummy octet string for NAS Message, of size sufficient enough to fit in one RLC PDU of 144 bits, including the correct RLC AM header.
 - 2. The IE CN Domain Identity is Set to PS Domain (no signalling connection for this domain exists).
 - 3. The polling bit in RLC header is set for Transmission of RLC STATUS PDU.

The MAC header shall be set as follows:

Field	Value
TCTF	11'B
UE ID Type	C-RNTI
UE ID	As set in RRC CONNECTION
	SETUP message.
C/T	0111'B

- c) The SS <u>checks that UE shall neither transmit RRC Status message on SRB2 nor RLC Status</u> <u>PDU onmonitors the RACH for 10 s to ensure that no RACH access occurs</u>.
- d) The SS again transmits MAC PDUs as in b) above, but this time uses the correct C/T value for AM-DCCH NAS High Priority of 0010'B. The sequence numbers in the RLC headers shall be identical with those sent in b).
- e) SS receives RLC Status PDU on SRB #3 acknowledging the receipt of the above RLC PDU.
- f) The SS receives a RRC STATUS message on the uplink DCCH using AM RLC on SRB # 2.
- g) The SS repeats steps b), c), d), e) and f), with the C/T field set as follows:

Iteration	C/T Value
2	1111'B

Step	Direction		Direction Message		Message	Comments	
	UE	SS	-				
1	-	>	PAGING RESPONSE	Check C/T field			
2	÷	-	MAC PDU(TCTF, UE-ID, C/T, RLC AM PDU(SN=x, DIRECT TRANSFER))	Sent with incorrect C/T = 0111'B, c 1111'B.			
2a			wait for T = 10 s	SS checks that UE shall neither transmit RRC-Status message on SRB 2 nor RLC Status PDU on SRB 3monitors for RACH access attempts See note 1 below			
3	÷	-	MAC PDU(TCTF, UE-ID, C/T, RLC AM PDU(SN=x, DIRECT TRANSFER))	Sent with correct C/T = 0010'B			
4	-	>	RLC-STATUS-PDU	ACK PDUs with SN = x C/T Field is recognised as correct for the DCCH See note 2 below			
5	_	>	RRC Status PDU message				

Note 1: UE will Transmit Signalling Connection Release Indication on expiry of MM Timer T3240 or GMM Timer T3317

Note 2: RRC Status message may be received before RLC Status PDU.

Steps 2 to 5 of the expected sequence are repeated for iteration 2. Note: For iteration k the SN in steps 2 and 4 starts with x + (k-1).

Specific Message Contents

None

7.1.1.3.5 Test Requirement

In step a) the C/T field should be set to the Logical Channel ID for SRB #3 (0010'B). Note that this may be implied from receipt of the PAGING RESPONSE message correctly by the SS test script.

During the test the SS shall request RLC status report with every transmitted PDU by setting of the Polling Bit. The UE shall not send any STATUS PDUs indicating missing PDUs.

At the end of each iteration (steps 4 and 5 of expected sequence) the SS shall receive a RLC Status PDU on SRB # 3, with C/T field set to value '0010'B and RRC Status PDUmessage on SRB # 2.

7.1.1.4 DTCH or DCCH mapped to RACH/FACH / Invalid UE ID Type Field

7.1.1.4.1 Definition

This tests that the MAC applies the correct header to the MAC PDU according to the type of logical channel carried on the RACH/FACH transport channel. Incorrect application of MAC headers would result in inoperation of the UE.

7.1.1.4.2 Conformance requirement

DTCH or DCCH mapped to RACH/FACH:

TCTF field, C/T field, UE-Id type field and UE-Id are included in the MAC header. The following fields are defined for the MAC header:

- UE-Id Type

The UE-Id Type field is needed to ensure correct decoding of the UE-Id field in MAC Headers.

UE-Id Type field 2 bits	UE-Id Type
00	U-RNTI
01	C-RNTI
10	Reserved (PDUs with this coding will be discarded by this version of the protocol)
11	Reserved (PDUs with this coding will be discarded by this version of the protocol)

Table 9.2.1.7: UE-Id Type field definition

Reference(s)

TS 25.321 clauses 9.2.1 and 9.2.1.1 c).

7.1.1.4.3 Test purpose

- 1. To verify that the UE discards PDUs with reserved values in UE-Id type field.
- 2. To verify that the TCTF field, C/T field, UE-Id type and UE-Id field are correctly applied when a DTCH or DCCH is mapped to the RACH/FACH.

7.1.1.4.4 Method of test

Initial conditions

System Simulator: See clause 7.1.1.2.4.

User Equipment:

See clause 7.1.1.2.4.

Test procedure

- a) The SS receives the PAGING RESPONSE message from the UE and checks the UE-Id Type field.
- b) The SS transmits MAC PDUs containing RLC AM PDUs containing a DIRECT TRANSFER message containing.
 - 1. Dummy octet string for NAS Message, of size sufficient enough to fit in one RLC PDU of 144 bits, including the correct RLC AM header.
 - 2. The IE CN Domain Identity is Set to PS Domain (no signalling connection for this domain exists)
 - 3. The polling bit in RLC header is set for transmission of RLC STATUS PDU.

The MAC header shall be set as follows:

Field	Value
TCTF	11'B
UE ID Type	10'B
UE ID	As set in RRC CONNECTION
	SETUP message.
C/T	Logical Channel ID for SRB #3 (AM-
	DCCH NAS High Priority): 0010'B

- c) The SS <u>checks that UE shall neither transmit RRC Status message on SRB2 nor RLC Status</u> <u>PDU on SRB3.monitors the RACH for 10 s to ensure that no RACH access occurs.</u>
- d) The SS again transmits MAC PDUs as in b) above, but this time uses the correct UE-Id type value for C-RNTI of 01'B. The sequence numbers in the RLC headers shall be identical with those sent in b).
- e) SS Receives RLC Status PDU on SRB #3 acknowledging the receipt of the above RLC PDU.
- f) The SS receives a RRC STATUS message on the uplink DCCH using AM RLC on SRB # 2
- g) The SS repeats steps b), c), d), e) and f), with the UE-Id type field set as follows in step b):

Iteration	UE-Id type Value	
2	11'B	

Step	Direction	Message	Comments
	UE SS		
1	\rightarrow	PAGING RESPONSE	Check UE-Id Type
2	÷	MAC PDU(TCTF, UE-ID, C/T, RLC AM	Sent with incorrect UE-Id Type =
		PDU(SN=x, DIRECT TRANSFER))	10'B, or 11'B.
2a		wait for $T = 10 s$	SS checks that UE shall neither
			transmit RRC-Status message on
			SRB 2 nor RLC Status PDU on
			SRB 3monitors for RACH access
			attempts
			See note 1 below
3	\leftarrow	MAC PDU(TCTF, UE-ID, C/T, RLC AM	Sent with correct UE-Id Type = 01
		PDU(SN=x, DIRECT TRANSFER))	
4	\rightarrow	RLC-STATUS-PDU	ACK PDU with $SN = x$
			UE-Id is recognised as correct for
			the UE
			See note 2 below
5	\rightarrow	RRC Status PDU message	RRC
$1: U\overline{E}$	will Transmi	t Signalling Connection Release Indication on	expiry of MM Timer T3240

or GMM Timer T3317

Note 2: RRC Status message may be received before RLC Status PDU.

Steps 2 to 5 of the expected sequence are repeated for iteration 2. Note: For iteration k the SN in step 2 and 4 starts with x + (k-1).

Specific Message Contents

None

7.1.1.4.5 Test Requirement

In step a) the UE-Id Type field should be set to 01'B. Note that this may be implied from receipt of the PAGING RESPONSE message correctly by the SS test script. During the test the SS request an RLC status report with every transmitted PDU by setting of the Polling Bit. The UE shall not send any STATUS PDUs indicating missing PDUs.

At the end of each iteration (steps 4 and 5 of expected sequence) the SS shall receive a RLC Status PDU on SRB # 3, with UE Id type correctly set to '01'B and RRC Status <u>PDUmessage</u> on SRB # 2.

7.1.1.5 DTCH or DCCH mapped to RACH/FACH / Incorrect UE ID

7.1.1.5.1 Definition

This tests that the MAC applies the correct header to the MAC PDU according to the type of logical channel carried on the RACH/FACH transport channel. Incorrect application of MAC headers would result in inoperation of the UE.

7.1.1.5.2 Conformance requirement

DTCH or DCCH mapped to RACH/FACH:

TCTF field, C/T field, UE-Id type field and UE-Id are included in the MAC header. The following fields are defined for the MAC header:

UE-Id

The UE-Id field provides an identifier of the UE on common transport channels...

Lengths of UE Id field

UE Id type	Length of UE Id field
U-RNTI	32 bits
C-RNTI	16 bits

Reference(s)

TS 25.321 clauses 9.2.1 and 9.2.1.1 c).

7.1.1.5.3 Test purpose

- 1. To verify that the UE ignores PDUs with UE-Ids that do not match the Id allocated to it.
- 2. To verify that the TCTF field, C/T field, UE-Id type and UE-Id field are correctly applied when a DTCH or DCCH is mapped to the RACH/FACH.

7.1.1.5.4 Method of test

Initial conditions

System Simulator: See clause 7.1.1.2.4.

User Equipment: See clause 7.1.1.2.4.

Test procedure

- a) The SS receives the PAGING RESPONSE message from the UE and checks the UE-Id field.
- b) The SS transmits MAC PDUs containing RLC AM PDUs containing a DIRECT TRANSFER message containing.
 - 1. Dummy Octet String for NAS Message, of size sufficient enough to fit in one RLC PDU of 144 bits, including the correct RLC AM header.
 - 2. The IE CN Domain Identity is Sset to PS Domain (no signalling connection for this domain exists)
 - 3. The polling bit in RLC header is set for transmission of RLC STATUS PDU.

The MAC header shall be set as follows:

Field	Value
TCTF	11'B
UE ID Type	C-RNTI
UE ID	Address allocated in RRC CONNECTION SETUP message + 1.
C/T	Logical Channel ID for SRB #3 (AM- DCCH NAS High Priority): 0010'B

- c) The SS <u>checks that UE shall neither transmit RRC Status message on SRB2 nor RLC Status</u> <u>PDU on SRB3</u>monitors the RACH for 10 s to ensure that no RACH access occurs.
- d) The SS again transmits MAC PDUs as in b) above, but this time uses the correct UE-Id value of the address allocated in the RRC CONNECTION SETUP message. The sequence numbers in the RLC headers shall be identical with those sent in b).
- e) SS Receives RLC Status PDU on SRB #3 acknowledging the receipt of the above RLC PDU
- f) The SS receives a RRC STATUS message on the uplink DCCH using AM RLC on SRB # 2.

Step	Direc	tion	Message	Comments
	UE	SS		
1	Ŷ	•	PAGING RESPONSE	Check UE-Id
2	÷	-	MAC PDU(TCTF, UE-ID, C/T, RLC AM PDU(SN=x, DIRECT TRANSFER))	Sent with incorrect UE-Id = C- RNTI+1
2a			wait for $T = 10 s$	SS checks that UE shall neither transmit RRC-Status message on SRB 2 nor RLC Status PDU on SRB 3monitors for RACH access attempts See note 1 below
3	÷	-	MAC PDU(TCTF, UE-ID, C/T, RLC AM PDU(SN=x, DIRECT TRANSFER))	Sent with correct UE-Id = C-RNTI
4	÷	•	RLC-STATUS-PDU	ACK PDUs with SN = x UE-Id is recognised as correct for the UE See note 2 below
5	\rightarrow	•	RRC Status PDUmessage	
1: UE v	will Tr	ansmi	t Signalling Connection Release Indication on expi	ry of MM Timer T3240

or GMM Timer T3317

Note 2: RRC Status message may be received before RLC Status PDU.

Specific Message Contents

None

7.1.1.5.5 Test Requirement

In step a) the UE-Id field should be set to the C-RNTI allocated in the RRC CONNECTION SETUP message. Note that this may be implied from receipt of the PAGING RESPONSE message correctly by the SS test script.

During the test the SS shall request an RLC status report with every transmitted PDU by setting of the Polling Bit. The UE shall not send any STATUS PDUs indicating missing PDUs.

At the end of the expected sequence (steps 4 and 5) the SS shall receive RLC Status PDU on SRB # 3 with correct C-RNTI and RRC Status PDUmessage on SRB # 2.

7.1.1.8 DTCH or DCCH mapped to DCH / Invalid C/T Field

7.1.1.8.1 Definition

This tests that the MAC applies the correct header to the MAC PDU according to the type of logical channel carried on the DCH transport channel. Incorrect application of MAC headers would result in inoperation of the UE.

7.1.1.8.2 Conformance requirement

DTCH or DCCH mapped to DCH, no multiplexing of dedicated channels on MAC: - no MAC header is required.

DTCH or DCCH mapped to DCH, with multiplexing of dedicated channels on MAC: -C/T field is included in MAC header.

The following fields are defined for the MAC header:

- C/T field
 - The C/T field provides identification of the logical channel instance when multiple logical channels are carried on the same transport channel...

C/T field	d Designation	
0000	Logical channel 1	
0001	Logical channel 2	
1110	Logical channel 15	
1111	Reserved	
	(PDUs with this coding will be	
	discarded by this version of	
	the protocol)	

Structure of the C/T field

Reference(s)

TS 25.321 clauses 9.2.1 and 9.2.1.1 b).

7.1.1.8.3 Test purpose

- 1. To verify that the UE discards PDUs with reserved or incorrect values in C/T field.
- 2. To verify that the C/T field is correctly applied when a DTCH or DCCH is mapped to a DCH.

7.1.1.8.4 Method of test

Initial conditions

System Simulator:

- 1 cell, default parameters, Ciphering Off.

Lligharla		DAD/signalling DD		
Higher lay	yer	RAB/signalling RB		RB#3 (SRB#3)
		User of Radio Bea	rer	NAS_DT
				High prio
RLC		Logical channel typ	be	DCCH
		RLC mode		ТМ
		Payload sizes, bit		148
		Max data rate, bps		3700
		RLC header, bit		0
MAC		MAC header, bit		0 (note)
		MAC multiplexing		Simulated by SS
Layer 1		TrCH type		DCH
		TB sizes, bit		148
		TFS	TF0, bits	0 x 148
			TF1, bits	1 x 148
		TTI, ms		40
		Coding type		CC 1/3
		CRC, bit		16
		Max number of bits/TTI before rate		516
		matching		
		Uplink: Max numbe	er of bits/radio	129
		frame before rate matching		
	RM attribute			155-165
NOTE:	the head	SS MAC layer must be configured not to add a MAC header so that eader can be added by the test case in order to create the ssary invalid values.		

The DCH/DPCH is configured as specified in TS 34.108 clause 6.10.2.4.1.2: Stand-alone UL:3.4 DL:3.4 kbps SRBs for DCCH) with the following exception:

The TFCS should be configured as specified in clause 6.10.2.4.1.2.1.1.2.

User Equipment:

The UE shall operate under normal test conditions, Ciphering Off.

The Test-USIM shall be inserted.

The SS starts broadcasting the System Information as specified in TS 34.108 clause 6.1, using the configuration for the PRACH and SCCPCH (signalled in SYSTEM INFORMATION 5) as follows:

- The SCCPCH is configured as specified in TS 34.108 clause 6.10.2.4.3.3 (Interactive/Background 32 kbps RAB + SRB for PCCH + SRB for CCCH + SRB for DCCH + SRB for BCCH).
- 2. The PRACH is configured as specified in TS 34.108 clause 6.10.2.4.4.1.

The SS follows the procedure in TS 34.108 clause 7.4.2.1 (Mobile Terminated) so that the UE shall be in state BGP 6-1 (CS-CELL_DCH_INITIAL). During this procedure the RRC CONNECTION SETUP message shall allocate a DCH to carry the signalling radio bearers as follows:

1. The DCH/DPCH is configured as specified in TS 34.108 clause 6.10.2.4.1.2: Stand-alone UL:3.4 DL:3.4 kbps SRBs for DCCH).

Test procedure

- a) The SS receives the PAGING RESPONSE message from the UE and checks the C/T field.
- b) The SS transmits MAC PDUs containing RLC AM PDUs containing a DIRECT TRANSFER message containing

- 1. Dummy octet string for NAS Message, of size sufficient enough to fit in one RLC PDU of 144 bits, including the correct RLC AM header.
- 2. The IE CN Domain Identity is Set to PS Domain (no signalling connection for this Domain exists).
- 3. The polling bit in RLC header is set for transmission of RLC STATUS PDU.

The MAC header shall be set as follows:

Field	Value
C/T	0100'B

- c) The SS <u>checks that UE shall neither transmit RRC Status message on SRB2 nor RLC Status</u> <u>PDU on SRB3</u>monitors the DCH (DCCH/SRB#3) for 10 s to ensure that no transmissions occur.
- d) The SS again transmits MAC PDUs as in b) above, but this time uses the correct C/T value for AM-DCCH NAS High Priority of 0010'B. The sequence numbers in the RLC headers shall be identical with those sent in b).
- e) SS Receives RLC Status PDU on SRB #3 acknowledging the receipt of the above RLC PDU.
- f) The SS receives a RRC STATUS message on the uplink DCCH using AM RLC on SRB # 2.
- g) The SS repeats steps b), c), d), e) and f), with the C/T field set as follows in step b):

Iteration	C/T Value
2	1111'B

Expected sequence

Step	Direc	tion	Message	Comments
-	UE	SS		
1	-	>	PAGING RESPONSE	Check C/T field
2	÷	_	MAC PDU(C/T, RLC AM PDU(SN=x, DIRECT TRANSFER))	Sent with incorrect C/T = 0100'B, 1111'B
2a			wait for T = 10 s	SS checks that UE shall neither transmit RRC-Status message or SRB 2 nor RLC Status PDU on SRB 3monitors for DCH (SRB#3) transmissions See note 1 below
3	÷	-	MAC PDU(C/T, RLC AM PDU(SN=x, DIRECT TRANSFER))	Sent with correct C/T = 0010'B
4	-	>	RLC-STATUS-PDU	ACK PDUs with SN = x C/T Field is recognised as correct for the DCCH See note 2 below
5	-	>	RRC Status PDUmessage	
1: UE •	will Tr	ansmi	t Signalling Connection Release Indication on ex	piry of MM Timer T3240
AM Tir				

Note 2: RRC Status message may be received before RLC Status PDU.

Steps 2 to 5 of the expected sequence are repeated for iteration 2. Note: For iteration 2 the SN in steps 2 and 4 starts with x+1.

Specific Message Contents

None

7.1.1.8.5 Test Requirement

In step a) the C/T field should be set to the Logical Channel ID for SRB #3 (0010'B). Note that this may be implied from receipt of the PAGING RESPONSE message correctly by the SS test script.

During the test the SS shall request RLC status reports with every transmitted PDU by setting of the Polling Bit. The UE shall not send any STATUS PDUs indicating missing PDUs.

At the end of each iteration (steps 4 and 5 of expected sequence) the SS shall receive a RLC Status PDU on SRB # 3 with C/T field set to '0010'B and RRC Status PDU message on SRB # 2.

3GPP TSG–T1/SIG Meeting #25 Singapore, 18th – 20th September 2002

Tdoc: T1-020691

	CHANGE REQUEST	CR-Form-v4
ж	<mark>34.123-1</mark> CR <mark>323 ^ℋev -</mark> ^ℋ	Current version: 5.1.0 [%]
For <u>HELP</u> on	using this form, see bottom of this page or look at the	e pop-up text over the # symbols.
Proposed change	e affects: ೫ (U)SIM ME/UE X Radio Ac	ccess Network Core Network
Title:	General corrections for clause 6	
Source:	Siemens, Ericsson	
Work item code:	f TEI	Date: # 10 September 2002
Category:	 F Use <u>one</u> of the following categories: F (correction) A (corresponds to a correction in an earlier release B (addition of feature), C (functional modification of feature) D (editorial modification) Detailed explanations of the above categories can be found in 3GPP <u>TR 21.900</u>. 	Release: #REL-5Use one of the following releases: 2(GSM Phase 2)e)R96(Release 1996)R97(Release 1997)R98(Release 1998)R99(Release 1999)REL-4(Release 4)REL-5(Release 5)
Reason for chang	re: 第 - To avoid misunderstandings, references t	to ITLL regions are removed. The text
	 proposed is included as appeared in TS2 References to default RF signal levels in To not cause SS fail due to SS tolerances CPICH_Ec (FDD) and PCCPCH_RSCP (least 5 dB. Thus need tables 6.3 and 6.4 and TDD be updated accordingly. This changes in specific test cases as introduc (T1-020513) and CR262 (T1-020514) at T 	clause 6 not correct s the transmission power level TDD) between cells should differ at with default RF signal levels for FDD hange correspond to the approved ced by CR260 (T1-020512), CR261
Summary of char	 Text added to distinguish a detectable ce Table 6.3 is corrected, giving references to Table 6.4 is added with reference test free 25.102. Also this table is specified fot TDI Table 6.4a is included to specify properly 1.28 Mcps option From Ericsson document (T1S-020554) References to default RF signal le cells from 34.108 table 6.1.1 to 34 In table 6.3: changed power separatio CPICH_Ec(FDD) to be 5 dB. 	to Band I, Band II and Band III quencies for Band c, according to TS D 3.84 Mcps option. the frequencies to be used for TDD evels in clause 6 changed for FDD I.123-1 table 6.3
Consequences if not approved:	 Information will be incomplete and confusing Idle mode test cases not aligned with core specified 	
Clauses affected:	ж <mark>6</mark>	
Other specs affected:	#Other core specifications#Test specifications0&M Specifications	

Other comments:	ж	Affects R99, REL-4, REL-5
		Based on T1S-020593, including the changes from Ericsson document
		T1S-020554
		Approved at Singapore meeting, T1S-020643

6 Idle mode operations

In the following paragraphs some explanatory text is given concerning the nature of the tests in this clause and the general behaviour of the SS is described.

Since the conformance requirements of most of the tests in this clause cannot be tested explicitly, testing is done implicitly by testing the UE behaviour from its responses to the SS.

In some cases, a test is performed in multiple stages in order that the requirements can be tested within the above constraints.

For any UE all the carriers are in its supported band(s) of operation.

Unless otherwise stated in the method of test, in all of the tests of this clause:

- the default values of the system information data fields given in TS 34.108 are used;
- the UE is equipped with a USIM containing default values. The USIM is in the idle updated state in the default location area with a TMSI assigned at the beginning of each test;
- the cells shall be configured such that Squal>0 (FDD only) and Srxlev>0 while applying Qqualmin (FDD only) and Qrxlevmin in table 6.1. In addition, for an FDD cell, the measured primary CPICH RSCP value shall be greater than or equal to -95 dBm (definition of High Quality cell, see TS 25.304, clause 5.1.2.2). In addition, for a TDD cell, the measured P-CCPCH RSCP shall be greater than or equal to -84 dBm (definition of High Quality cell, see TS 25.304, clause 5.1.2.2).

Three different methods A, B and C are applied in the tests:

Method A:

- the SS is continuously paging the UE on all cells at the start of the test and does not respond to RACH requests from the UE (which causes a cell reselection). Where a test specifies that the UE is not paged on a particular cell, only idle paging is transmitted. This method is similar to the one used in TS 51.010-1, clause 20.

Method B:

- the SS is continuously paging the UE on all cells at the start of the test and responds to RACH requests from the UE with an IMMEDIATE ASSIGNMENT REJECT (GERAN cell) or RRC CONNECTION REJECT (UTRAN cell) message which causes the UE to return to Idle mode. Where a test specifies that the UE is not paged in a particular cell, only idle paging is transmitted.

Method C:

- no continuously paging as in method A or B. Normal response to RACH requests so Location Updating and Calls can be done.

In case a test specifies that UE shall read System Information on BCCH while camped on a UTRAN cell, SS shall notify UE on the BCCH modification by sending a PAGING TYPE 1 message to UE. This message shall contain IE BCCH Modification Info with the following settings:

Information Element	Value/remark
BCCH modification info	
MIB Value Tag	Set to the same value as the value tag of the MIB after
	the BCCH modification
BCCH Modification time	Not present

Parameter	Setting
IMSI attach/detach	Method A, B: Not allowed
	Method C: Allowed
Intra-frequency cell re-selection	Allowed
indicator	
Cell_selection_and_reselection_qu	CPICH RSCP (FDD)
ality_measure	
Qqualmin (FDD only)	-24 dB
Qrxlevmin (FDD)	-115 dBm
Qrxlevmin (TDD)	-103 dBm
DRX cycle length	1,28 s

For a UE camping in a FDD cell, CPICH_Ec/Io and SCH_Ec/Io shall fulfill requirements in TS 25.133, clause 8.1.2.2.1: The UE is able to identify a new detectable cell belonging to the monitored set within $T_{identify intra}$ when CPICH Ec/Io \geq -20 dB and SCH_Ec/Io \geq -20 dB.

For a UE camping in a TDD cell, the UE shall be able to identify a new detectable cell belonging to the monitored set within T_{identify intra} when fulfill requirements in TS 25.123, clause 8.1.2.2. for TDD 3.84 Mcps option and 8.1A.2.2 for TDD 1.28 Mcps option.

It is a UE option whether to indicate access technologies to the user (TS 23.122, clause 4.4.3.1.2). Therefore, for combined UTRAN/GSM tests, it is indicated in parentheses which access technology shall be indicated to the user if the UE has this capability.

If a parameter is indicated with a *, it means that the parameter is calculated internally in the UE and is only shown for clarification of the test procedure.

The PLMN numbers indicated in table 6.2 are used in test cases to associate a cell with an MCC and MNC for that cell. If no PLMN is explicitly specified, the default value is PLMN 1.

Table 6.2: Location Area Information (LAI) in System Information type 3 messages broadcast on the BCCH (GSM) or System Information Block Type 1 broadcast on the BCH (UMTS)

PLMN	MCC1	MCC2	MCC3	MNC1	MNC2	MNC3	LAC
1	0	0	1	0	1	Not present	Х
2	0	0	2	1	1	Not present	х
3	0	0	4	2	1	Not present	х
4	0	0	5	3	1	Not present	х
5	0	0	6	4	1	Not present	х
6	0	0	7	5	1	Not present	х
7	0	0	8	6	1	Not present	х
8	0	0	9	7	1	Not present	х
9	0	1	0	0	2	Not present	х
10	0	1	1	1	2	Not present	х
11	0	1	2	2	2	Not present	х
12	0	1	3	3	2	Not present	х
NOTE:	x' denotes a	ny value.					

References: TS 23.122, annex A and TS 23.003, clause 2.

The test channel numbers indicated in tables 6.3, 6.4, <u>6.4a</u> and 6.5 are used in test cases to associate a cell with a frequency for that cell. The frequencies for GSM and DCS cells in table 6.5 are identical to those used in TS 51.010-1, clause 26.3.1. The RF signal levels <u>are given in table 6.5</u> for GSM cells are given in table 6.5, in table 6.3 for UTRAN FDD cells, in table 6.4 for UTRAN TDD cells 3.84 Mcps option and in table 6.4a for UTRAN TDD cells 1.28 Mcps option. in TS 34.108, table 6.1.1 and for UTRAN TDD cells in TS 34.108, table 6.1.5. If no channel is explicitly specified, the default value is Test Channel 1.

	Band I ITU region 1		Band II ITU region 2		Band III ITU region 3	
Test Channel	CPICH_Ec dBm / 3.84 MHz	Uplink UARFCN	CPICH_Ec dBm / 3.84 MHz	Uplink UARFCN	CPICH_Ec dBm / 3.84 MHz	Uplink UARFCN
1	- <u>6072</u>	9 613	- <u>60</u> 72	9 263	- <u>6072</u>	8 563
2	- <mark>65</mark> 75	9 663	- <mark>65</mark> 75	9 313	- <mark>65</mark> 75	8 613
3	- <mark>70</mark> 78	9 713	- <mark>70</mark> 78	9 363	- <mark>70</mark> 78	8 663
4	- <mark>75</mark> 81	9 763	- <mark>75</mark> 81	9 413	- <mark>75</mark> 81	8 713
5	- <mark>80</mark> 84	9 813	- <mark>80</mark> 84	9 463	- <mark>80</mark> 84	8 763
6	- <u>85</u> 87	9 863	- <u>85</u> 87	9 513	- <mark>85</mark> 87	8 813

Table 6.3: UTRA (FDD) test frequencies

References: TS 34.108, clause 5.1.1 and TS 34.121, clause 4.

Table 6.4: UTRA TDD test frequencies (3.84 Mcps option)

	Band a		Band b ITU	region 2	Band c	
Test Channel	P-CCPCH_ RSCP [dBm]	UARFCN	P-CCPCH_ RSCP [dBm]	UARFCN	<u>P-CCPCH</u> RSCP [dBm]	<u>UARFCN</u>
1	- <u>54</u> 61	9 513	- <u>54</u> 61	9 263	<u>-54</u>	<u>9563</u>
2	- <u>59</u> 64	9 550	- <u>59</u> 64	9 <u>400</u> 537	<u>-59</u>	<u>9577</u>
3	- <u>64</u> 67	9 587	-6 <mark>4</mark> 7	9 <u>537</u> 663	<u>-64</u>	<u>9591</u>
4	- <u>69</u> 70	10 063	- <u>69</u> 70	9 <u>663</u> 937	<u>-69</u>	<u>9605</u>
5	- <u>74</u> 73	10 087	-7 <u>4</u> 3	9 <u>800<mark>563</mark></u>	<u>-74</u>	<u>9619</u>
6	- <u>79</u> 76	10 112	-7 <mark>96</mark>	9 <u>937<mark>637</mark></u>	<u>-79</u>	<u>9637</u>

Table 6.4a: UTRA TDD test frequencies (1.28 Mcps option)

	Band	Band a		Band b		C
<u>Test</u> Channel	<u>P-CCPCH</u> RSCP [dBm]	UARFCN	P-CCPCH RSCP [dBm]	<u>UARFCN</u>	P-CCPCH RSCP [dBm]	<u>UARFCN</u>
<u>1</u>	<u>-54</u>	<u>9 505</u>	<u>-54</u>	<u>9 255</u>	<u>-54</u>	<u>9555</u>
<u>2</u>	<u>-59</u>	<u>9 550</u>	<u>-59</u>	<u>9 400</u>	<u>-59</u>	<u>9573</u>
<u>3</u>	<u>-64</u>	<u>9 595</u>	<u>-64</u>	<u>9 545</u>	-64	<u>9591</u>
4	<u>-69</u>	<u>10 055</u>	<u>-69</u>	<u>9 655</u>	<u>-69</u>	<u>9608</u>
5	-74	10 088	-74	9 800	-74	<u>9626</u>
<u>6</u>	<u>-79</u>	10 120	-79	<u>9 945</u>	-79	<u>9645</u>

References: TS 34.108, clause 5.1.2 and TS 34.122, clause 4.

	G	SM 900	DC	S 1 800
Test Channel	level BCCH ARFCN dBμVemf() / dBm		level dBµVemf() / dBm	BCCH ARFCN
1	+65 / -48	1	+65 / -48	520
2	+63 / -50	7	+63 / -50	580
3	+61 / -52	39	+61 / -52	610
4	+55 / -58	65	+55 / -58	702
5	+59 / -54	66	+59 / -54	703
6	+57 / -56	85	+57 / -56	830
7	+55 / -58	97	+55 / -58	885
8	+53 / -60	124		

Table 6.5: GSM/DCS test frequencies and levels

	GS	SM 450	DCS 480		
Test Channel	level dBμVemf() / dBm	BCCH ARFCN	level dBµVemf() / dBm	BCCH ARFCN	
1	+65 / -48	259	+65 / -48	306	
2	+63 / -50	261	+63 / -50	308	
3	+61 / -52	267	+61 / -52	314	
4	+55 / -58	268	+55 / -58	315	
5	+59 / -54	281	+59 / -54	328	
6	+57 / -56	288	+57 / -56	335	
7	+55 / -58	291	+55 / -58	338	
8	+53 / -60	293	+53 / -60	340	

	Multiba	nd 900/1800	PCS 1900		
Test Channel	level dBμVemf() / dBm	BCCH ARFCN	level dBmVemf() / dBm	BCCH ARFCN	
1	+65 / -48	520	+65 / -48	512	
2	+63 / -50	7	+63 / -50	520	
3	+61 / -52	39	+61 / -52	580	
4	+55 / -58	702	+55 / -58	610	
5	+59 / -54	66	+59 / -54	702	
6	+57 / -56	85	+57 / -56	703	
7	+55 / -58	885	+55 / -58	800	
8	+53 / -60	124			

	Multiba	and 450/900	Multiba	nd 480/900
Test Channel	level BCCH ARFCN dBμVemf() / dBm		level dBµVemf() / dBm	BCCH ARFCN
1	+65 / -48	1	+65 / -48	1
2	+63 / -50	261	+63 / -50	308
3	+61 / -52	267	+61 / -52	314
4	+55 / -58	65	+55 / -58	65
5	+59 / -54	281	+59 / -54	328
6	+57 / -56	288	+57 / -56	335
7	+55 / -58	124	+55 / -58	124
8	+53 / -60	293	+53 / -60	340

	Multiba	nd 450/1800	Multibar	nd 480/1800
Test	level	BCCH ARFCN	level	BCCH ARFCN
Channel	dBµVemf()		dBµVemf()	
	/ dBm		/ dBm	
1	+65 / -48	520	+65 / -48	520
2	+63 / -50	261	+63 / -50	308
3	+61 / -52	267	+61 / -52	314
4	+55 / -58	702	+55 / -58	702
5	+59 / -54	281	+59 / -54	328
6	+57 / -56	288	+57 / -56	335
7	+55 / -58	885	+55 / -58	885
8	+53 / -60	293	+53 / -60	340

For testing an E-GSM Mobile station, the BCCH ARFCN of GSM Test Channel 7 at GSM 900 column shall be 985 (instead of 97). For testing an R-GSM Mobile station, the BCCH ARFCN of GSM Test Channel 7 at GSM 900 column shall be 965 (instead of 97).

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

Tdoc #T1-020704

Tdoc #T1S-020692

3GPP TSG- T1 SIG Meeting #26 Luton, UK, 4th – 8th November 2002

		(CHANGE	REQ	UE	ST			CR-Form-v7
ж	34.123-1	CR	331	ж rev	-	Ħ	Current version:	5.1.0	ж
For <mark>H</mark>	ELP on using this for	m, see	e bottom of this	s page or	look	at th	e pop-up text over	r the X syr	nbols.

Proposed change affects: UICC apps#

ME X Radio Access Network Core Network

Title:	Correction to RLC P1 7.2.3.12 Correct use of Sequence Numbering							
Source:	ж	Motorola, Ericsson						
Work item code	:X	TEI	<i>Date:</i>	03/10/2002				
Category:	ж	F	Release: ೫	REL-5				
		Use one of the following categories:	Use <u>one</u> of	the following releases:				
		F (correction)	2	(GSM Phase 2)				
		A (corresponds to a correction in an earlier release)	R96	(Release 1996)				
		B (addition of feature),	R97	(Release 1997)				
		C (functional modification of feature)	R98	(Release 1998)				
		D (editorial modification)	R99	(Release 1999)				
		Detailed explanations of the above categories can	Rel-4	(Release 4)				
		be found in 3GPP <u>TR 21.900</u> .	Rel-5	(Release 5)				
			Rel-6	(Release 6)				

Reason for change: #	The window size that a UE can support depends on the RLC AM buffer size which is reported in the UE capability. It is proposed to reduce the transmission window size from 2047 to 128, as 128 is the default value used for transmission window size in RLC test cases, and as the test puppose can still be achieved using window size of 128.				
Summary of change: #	1. Uplink and Downlink window size changed from 2047 to 128				
	 Added use of polling mechanism "Every Poll_PDU PDU" (poll every 64th UPLINK PDU) to ensure that UE transmission buffer will not become full during test. 				
	 Changed SS polling to every 64th PDU (50% of UE receiving window size) to ensure UE will not discard any PDU received outside UE receive window 				
	4. Updated expected sequence for UE polling of every 64 th PDU.				
	 Renumbered PDUs in expected sequence (starting from 1 instead of 0) to correspond to how PDUs are referenced in the test requirement. 				
Consequences if % not approved:	Good UE will fail the test				

Clauses affected: # 7.2.3.12

		Y	Ν			
Other specs affected:	Ħ		Χ	Other core specifications % Test specifications O&M Specifications		
Other comments:	ж	A		ts R99, REL-4 and REL-5 test cas	ISES.	

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

7.2.3.12 Correct use of Sequence Numbering

7.2.3.12.1 Definition

Peer RLC entities use sequence numbering to detect missing PDUs, and for flow control purposes. Incorrect operation of sequence numbering will result in failure of the UE to communicate.

7.2.3.12.2 Conformance requirement

This sub-clause describes the state variables used in AM and UM in order to specify the peer-to-peer protocol. All state variables are non-negative integers. UMD and AMD PDUs are numbered by modulo integer sequence numbers (SN) cycling through the field: 0 to $2^{12} - 1$ for AM ... All arithmetic operations contained in the present document on VT(S), VT(A), VT(MS), VR(R), VR(H) and VR(MR) are affected by the AM modulus. ... When performing arithmetic comparisons of state variables or Sequence number values a modulus base shall be used. This modulus base is subtracted (within the appropriate field) from all the values involved and then an absolute comparison is performed. At the Sender, VT(A) ... shall be assumed to be the modulus base in AM ... At the Receiver, VR(R) ... shall be assumed to be the modulus base in AM ... At the Receiver, VR(R) ... shall be assumed to be the modulus base in AM

The RLC shall maintain the following state variables in the Sender.

a) VT(S) - Send state variable.

This state variable contains the "Sequence Number" of the next AMD PDU to be transmitted for the first time (i.e. excluding retransmitted PDUs). It shall be updated after the aforementioned AMD PDU is transmitted or after transmission of a MRW SUFI which includes $SN_MRW_{LENGTH} > VT(S)$ (see subclause 11.6). The initial value of this variable is 0.

[...]

If the AMD PDU is transmitted for the first time, the Sender shall:

- set the "Sequence Number" field equal to VT(S);

Reference(s)

TS 25.322, clauses 9.4 and 11.3.2.1.

7.2.3.12.3 Test purpose

- 1. To verify that the UE transmits the first PDU with the Sequence Number field equal to 0.
- 2. To verify that the UE increments the Sequence Number field according to the number of PDUs transmitted.
- 3. To verify that the UE wraps the Sequence Number after transmitting the 2^{12} -1th PDU.
- 4. To verify that the UE receiver accepts PDUs with SNs that wrap around every 2^{12} -1th PDU.

7.2.3.12.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	
Polling info	
Poll_PDU	64
Transmission window size	1282047
Downlink RLC	
Receiving window size	<u>128</u> 2047

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to (2 * AM_7_PayloadSize) - 1 bytes.

Test procedure

a) The SS sends 2049 RLC SDUs to the UE, each of $(2 * AM_7_PayloadSize) - 1$ bytes. The SS polls for status on each $\frac{12864}{12864}$ th RLC PDU and the last PDU transmitted

- b) When the SS received an uplink PDU with the P bit set to 1, the SS transmits a STATUS PDU acknowledging all the RLC PDUs received so far.
- c) The SS checks the sequence numbers of the RLC PDUs it receives in the uplink
- d) The SS checks the content of the SDUs it receives from the UE.
- e) The SS may optionally release the radio bearer.

Step	Direction	Message	Comments				
	UE SS						
1		RB ESTABLISHMENT	See generic procedures				
2	←	DOWNLINK RLC PDU # <mark>19</mark>					
3	÷	DOWNLINK RLC PDU #24					
			Transmission of DOWNLINK PDUs				
			continues				
4	\rightarrow	UPLINK RLC PDU <u>#1</u>	SN should be set to 0				
5	\rightarrow	UPLINK RLC PDU <u>#2</u>	SN should be set to 1				
	÷		Transmission of DOWNLINK PDUs				
			continues to SN = 63127				
6	\leftarrow	DOWNLINK RLC PDU # <u>64</u> 127	Poll				
7	\rightarrow	UPLINK STATUS PDU					
	\leftarrow		Transmission of DOWNLINK PDUs				
			continues, polling every <u>64</u> 128 th PDU, to				
-	,		SN = <u>4094</u> 4095				
8	÷	DOWNLINK RLC PDU #40965					
9	÷	DOWNLINK RLC PDU #40976	SN=0				
10	÷	DOWNLINK RLC PDU #40987	SN=1, Poll				
10a	\rightarrow	UPLINK RLC PDUs	Transfer of RLC PDUs continues to				
			SN = 2046, pollUE transmission of UPLINK				
			RLC PDUs continues, polling every 64 th PDU, to SN=4094.				
			On poll, SS acknowledge all received PDUs				
10b	÷	STATUS PDUVoid	ACK SN 0 to 2046				
100 10c	÷	UPLINK RLC PDU <u>#4096</u> 8	Transfer of RLC PDUs continues to				
100	/	01 EINIX IXEC 1 D0 <u>#4030</u> 5	SN = 4095 3 , Poll (cause: Poll_PDU=64)				
10d	÷	STATUS PDU	ACK SN 2047 -upto 4095 3				
10a	\rightarrow	VoidUPLINK RLC PDU	SN should be set to 4094				
11	\rightarrow	VoidUPLINK RLC PDU	SN should be set to 4095				
12	\rightarrow	UPLINK RLC PDU#4097	SN should be set to 0				
13	\rightarrow	UPLINK RLC PDU#4098	SN should be set to 1, Poll				
14	\leftarrow	DOWNLINK STATUS PDU	,				
15		RB RELEASE	Optional step				
NOTE	NOTE: The Expected Sequence shown is infomative.						
The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity.							
Information such as SDU, PDU or Sequence numbers given in the comments column shall be							
considered informative only, for test case development purposes.							

30

7.2.3.12.5 Test requirements

- 1. The first PDU received shall have the SN field set to 0. The second PDU shall have the SN field set to 1, and the 4097^{th} PDU shall have the SN field set to 0.
- 2. The size and data content of the received SDUs shall match those of the transmitted SDUs.

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

Tdoc #T1-020705

3GPP TSG- T1 SIG Meeting #26 Luton, UK, 4th – 7th Sept 2002

Tdoc **#***T*1*S*-020693

										CR-Form-v7
		СН	ANGE	REQ	UE	ST	•			-
ж <mark>3</mark>	<mark>4.123-1</mark>	CR 33	2	жrev	-	ж	Current vers	sion:	5.1.0	ж
For <u>HELP</u> on u	sina this fo	rm see bot	tom of this	nage or	look	at th	e non-un tex	t over	the # sv	mbols
		111, 000 001		page of	1001	at th				
					_					
Proposed change a	affects:	UICC apps	Ж	ME X	Rad	dio A	ccess Netwo	ork	Core Ne	etwork
Title: #	Correctio	n to packag	e 1 test ca	ase 7.2.3	.13 a	nd 7	.2.3.14			
Source: #	Ericsson									
									4.0./00.00	
Work item code: %	TEI						Date: #	8 02/	10/2002	
Category: ೫	F	the fellowing	ootoocris-				Release: #		-	000001
		the following rection)	categories	2			Use <u>one</u> oi 2		llowing rei 1 Phase 2)	
	A (coi	rresponds to		n in an eal	rlier re	eleas			ase 1996)	
		dition of featu Actional modi		eature)			R97 R98	•	ease 1997) ease 1998)	
	D (edi	itorial modific	ation)				R99	(Rele	ase 1999)	
		planations of 3GPP TR 2 ⁻		categories	s can		Rel-4 Rel-5		ease 4) ease 5)	
	be lound in	<u>1112</u>	<u></u> .				Rel-6	•	ase 6)	
Reason for change	v 92 1	Window	eiza (1536) used in		ond	run of test ca	507	2 3 13 and	4
Reason for change	, . .		•				able the test of			
							ity. The test p		se will still	be
		achieved	l if window	size is r	educ	ed fr	om 1536 to 1	28.		
	2						ehaviour whe			
							entation, the u ived from upp			
							buffer is full.		· •	
				est proce	dure	and	test requiren	nent o	f test cas	е
		7.2.3.13.								
Summary of chang	e:ж C	hanges to t	est case 7	.2.3.13:						
		Ũ			RLC	para	meters to sh	ow va	lues for fi	rst run
			and secon							
			Changed t I 28	ransmiss	ion w	/indc	w size for se	cond	run from <i>'</i>	1536 to
		t		LC PDUs	s takiı		expected sec le actual trans			
			fro	m 3*W to	o W+	1 SE	er of transmitt DUs. W+1 SD rn the first W	Us is	enough to	o verify

	size)
	 ii. step c) have been changed such that additional W+1 SDUs are sent after that SS have acknowleded the first W SDUs
	iii. step e) is changed in accordance with step c), but the number of sent SDUs are W/2+1.
	iv. Step g) is changed in accordance to step e)
	 v. Expected sequence step 7 is changed, and new steps 13a to 13c, 19a to 19c and 25a to 25 c have been added according to the changes in the test procedure
	 Test requirement have been changed to also cover the case when a UE discard SDUs when the transmission buffer is full.
	Changes to test case 7.2.3.14:
	 Updated table with RLC parameters to show values for first run and second run.
	 f. Changed UE transmitting and receiving window size for second run from 1536 to 128
Consequences if # not approved:	Good UE may fail
Clauses affected: #	7.2.3.13, 7.2.3.14
	YN
Other specs #	
Affected:	X Test specifications X O&M Specifications

Other comments: # Affects R99, REL-4 and REL-5 test cases.

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

7.2.3.13 Control of Transmit Window

7.2.3.13.1 Definition

This test is to check that the UE is able to correctly control its RLC transmission window. Correct operation of RLC windowing is critical for acknowledged mode operation.

7.2.3.13.2 Conformance requirement

VT(MS) - Maximum Send state variable.

This state variable contains the "Sequence Number" of the first AMD PDU that can be rejected by the peer Receiver, VT(MS) = VT(A) + VT(WS). This value represents the upper edge of the transmission window. The transmitter shall not transmit AMD PDUs with "Sequence Number" $\geq VT(MS)$ unless $VT(S) \geq VT(MS)$. In that case, the AMD PDU with "Sequence Number" = VT(S) - 1 can also be transmitted. VT(MS) shall be updated when VT(A) or VT(WS) is updated.

The initial value of this variable is Configured_Tx_Window_size.

... The receiver is always allowed to change the Tx window size of the peer entity during a connection, but the minimum and the maximum allowed value is given by RRC configuration. The Rx window of the receiver is not changed.

Reference(s)

TS 25.322, clauses 9.2.2.11.3 and 9.4.

7.2.3.13.3 Test purpose

To verify that the UE does not transmit PDUs with sequence numbers outside of the transmit window, except the PDU with SN=VT(S)-1, even when the transmit window size is changed by the receiver.

7.2.3.13.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	First run	Second run
Transmission window size	8	<u>128</u>

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let W be the size of the transmit window.

The length of all transmitted SDUs is set to AM_7_PayloadSize - 1 bytes.

- a) The SS transmits $\frac{3*}{M+1}$ RLC SDUs to the UE.
- b) The SS checks the RLC SDUs received on the uplink, but does not reply to poll requests from the UE, or transmit STATUS PDUs for any other reason.

Release 5

- c) After confirming that the UE has stopped transmitting new RLC SDUs for at least (2*W*TTI) ms, the SS transmits a STATUS PDU acknowledging all the RLC PDUs received so far. <u>The SS transmits W+1 additional</u> RLC SDUs to the UE.
- d) The SS again checks the RLC SDUs received on the uplink, but does not reply to poll requests from the UE, or transmit further STATUS PDUs for any other reason.
- e) After confirming that the UE has again stopped transmitting new RLC SDUs for at least (2*W*TTI) ms, the SS transmits a STATUS PDU acknowledging all the RLC PDUs received so far, and containing a WINDOW command to reduce the UE transmit window size (W) to half its initial size. <u>The SS transmits W/2+1 additional RLC SDUs to the UE (where W is the original window size)</u>.
- f) The SS checks the RLC SDUs received on the uplink, but does not reply to poll requests from the UE, or transmit STATUS PDUs for any other reason.
- g) After confirming that the UE has stopped transmitting new RLC SDUs for at least (2*W*TTI) ms, the SS transmits a STATUS PDU acknowledging all the RLC PDUs received so far. <u>The SS transmits W/2+1 additional RLC SDUs to the UE (where W is the original window size).</u>
- h) The SS checks the RLC SDUs received on the uplink.
- i) The SS may optionally release the radio bearer.
- NOTE: Window arithmetic is carried out modulo 4096.

The test procedure is run with the window transmit window size set to the default (8), and the repeated with the transmit window size set to $\frac{1536128}{1536128}$.

Expected sequence

Step	Direction UE SS	Message	Comments			
1		RB ESTABLISHMENT	See generic procedures			
2	÷		SDU 1			
		DOWNLINK RLC PDU				
3	\leftarrow	DOWNLINK RLC PDU	SDU 2			
4	\leftarrow	DOWNLINK RLC PDU	SDU 3			
5	÷	DOWNLINK RLC PDU	SDU 4			
	,					
6	÷		SS continues to transmit RLC SDUs			
7	÷	DOWNLINK RLC PDU	SDU <mark>3</mark> W <u>+1</u>			
8	\rightarrow	UPLINK RLC PDU	SDU 1			
9	\rightarrow	UPLINK RLC PDU	SDU 2			
3		of Elivic Reo 1 Do				
10	\rightarrow		SS continues to receive RLC SDUs			
11	\rightarrow	UPLINK RLC PDU	SDU W + poll			
12			No new transmissions from UE			
13	÷	STATUS PDU				
<u>13a</u>	<u>←</u>	DOWNLINK RLC PDUs	<u>SDU W+2</u>			
13b		<u></u>	SS continues to transmit RLC SDUs			
<u>13c</u>	\leftarrow	DOWNLINK RLC PDUs	<u>SDU 2W+2</u>			
14	\rightarrow	UPLINK RLC PDU	SDU W+1			
15	\rightarrow	UPLINK RLC PDU	SDU W+2			
15	7	OFLINK REC FDU	SDU W+2			
16	\rightarrow		SS continues to receive RLC SDUs			
17	\rightarrow	UPLINK RLC PDU	SDU 2W + poll			
18			No new transmissions from UE			
19	÷	STATUS PDU	WINDOW = W/2			
<u>19a</u>	<u> </u>	DOWNLINK RLC PDUs	SDU 2W+3			
	<u>~</u>	DOWNLINK REC PD05				
<u>19b</u>		<u></u>	SS continues to transmit RLC SDUs			
<u>19c</u>	←	DOWNLINK RLC PDUs	<u>SDU 2W + W/2 + 3</u>			
20	د	UPLINK RLC PDU	SDU 2W/1			
20	\rightarrow		SDU 2W+1			
21	\rightarrow	UPLINK RLC PDU	SDU 2W+2			
22	÷		SS continues to receive RLC SDUs			
23	\rightarrow	UPLINK RLC PDU	SDU 2W + W/2 + poll			
24			No new transmissions from UE			
<u>0</u> -	/					
25	÷	STATUS PDU				
<u>25a</u>	<u>←</u>	DOWNLINK RLC PDUs	<u>SDU 2W + W/2 + 4</u>			
<u>25b</u>			SS continues to transmit RLC SDUs			
<u>25c</u>	←	DOWNLINK RLC PDUs	<u>SDU 3W + 4</u>			
26	\rightarrow	UPLINK RLC PDU	SDU 2W+W/2+1			
27	\rightarrow	UPLINK RLC PDU	SDU 2W+W/2+2			
20	÷		SS continues to receive RLC SDUs			
28	$\overline{\mathbf{r}}$					
29	\rightarrow	UPLINK RLC PDU	SDU 3W + poll			
29		RB RELEASE	Optional step			

NOTE 1:	The Expected Sequence shown is infomative.
	The UPLINK and DOWNLINK PDU flows may overlap, but are shown separate for clarity. Information such as SDU, PDU or Sequence numbers given in the comments column shall be
	considered informative only, for test case development purposes.
NOTE 2:	The polls in step 11, 17, 23 and 29 are transmitted as the last PDU in buffer trigger is set to
	TRUE and the transmitted PDU is the last PDU in the transmitter window, see TS 25.322 clause
	9.7.1.

7.2.3.13.5 Test requirements

From steps 8 to 11, the SDU contents reassembled from the uplink shall match those of the first W transmitted SDUs.

At step 12 there shall be no further transmission on the uplink DTCH whilst the SS is waiting, except for any repeats of PDUs with sequence numbers from 0 to W-1.

After step 13, the UE shall resume transmission of the next W SDUs. The contents of these SDUs shall match those of SDUs W+1 to $2*W_{\star}$ or W+2 to 2*W+1 (see note), sent on the downlink.

At step 18 there shall be no further transmission on the uplink DTCH whilst the SS is waiting, except for any repeats of PDUs with sequence numbers from W to 2W-1.

After step 19, the UE shall resume transmission of the next W/2 SDUs. The contents of these SDUs shall match those of SDUs 2*W+1 to 2*W+W/2, or 2*W+3 to 2*W+W/2+4 (see note), sent on the downlink.

At step 24 there shall be no further transmission on the uplink DTCH whilst the SS is waiting, except for any repeats of PDUs with sequence number from 2W to W/2-1.

After step 25, the UE shall resume transmission of the next W/2 SDUs. The contents of these SDUs shall match those of SDUs 2*W+W/2+1 to 3*W, or 2*W+W/2+1 to 3*W (see note), sent on the downlink.

NOTETS 25.322 does not specify the UE behaviour when transmission buffer is full. Thus, depending on UEimplementation, the uplink AM entity may or may not discard AM RLC PDUs received from upper layer(e.g. from UE test loop function) when transmission buffer is full. The SS need to take this into
consideration when comparing SS sent SDUs in downlink with the UE returned SDUs in uplink.

7.2.3.14 Control of Receive Window

7.2.3.14.1 Definition

This test is to check that the UE is able to correctly control its RLC receive window. Correct operation of RLC windowing is critical for acknowledged mode operation.

This test applies to all UE.

7.2.3.14.2 Conformance requirement

Upon reception of an AMD PDU with "Sequence Number" outside the interval $VR(R) \leq SN < VR(MR)$, the Receiver shall:

- discard the AMD PDU;
- if the "polling bit" in the discarded AMD PDU is set to "1":
 - initiate the STATUS PDU transfer procedure.

Reference(s)

TS 25.322, clause 11.3.4.2.

7.2.3.14.3 Test purpose

To verify that the UE discards PDUs with sequence numbers outside the upper boundary of the receive window.

180

7.2.3.14.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	First run	Second run
Transmission window size	8	<u>128</u>
Downlink RLC		
Missing PDU Indicator	FALSE	FALSE
Receiving window size	8	<u>128</u>

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to AM_7_PayloadSize - 1 bytes.

Test procedure

Let W be the size of the receive window.

The length of all transmitted SDUs is set to AM_7_PayloadSize - 1 bytes.

- a) The SS transmits W RLC SDUs to the UE, polling only on the last RLC PDU.
- b) The SS checks the RLC SDUs received on the uplink, responding to poll requests with acknowledgements. Then after receiving the STATUS PDU from the UE it transmits a further RLC SDU with the poll bit set. The SS sets the sequence number for the associated RLC PDU above the top of the receive window, for example, 2*W.
- c) The SS receives a STATUS PDU from the UE.
- d) The SS transmits a further RLC SDU with the sequence number set to the value of the next sequence number within the receive window.
- e) The SS checks the RLC SDUs received on the uplink.
- f) The SS may optionally release the radio bearer.

This test case is run once for the default receive window size (8) and again with the receive window size set to 1281536.

Expected sequence

Step	Dire	rection Message Comments					
	UE	SS					
1			RB ESTABLISHMENT	See generic procedures			
2	€	-	DOWNLINK RLC PDU	SDU 1, SN = 0			
3	÷	<u>.</u>	DOWNLINK RLC PDU	SDU 2, SN = 1			
4	•	-		SS continues to transmit RLC SDUs			
5		<u>.</u>	DOWNLINK RLC PDU	SDU W + Poll, SN = W-1			
6 7		→ →	STATUS PDU UPLINK RLC PDU	SDU 1			
8		7 }	UPLINK RLC PDU	SDU 2			
0	-	7	OPLINK RLC PDU	500 2			
9				UE continues to transmit RLC SDUs			
10	_	>	UPLINK RLC PDU	SDU W, Poll			
10a	€	<u>.</u>	STATUS PDU				
11	÷	<u>.</u>	DOWNLINK RLC PDU	SDU W+1, SN = 2W, Poll			
11a	-	>	STATUS PDU	, , ,			
12	€	<u>.</u>	DOWNLINK RLC PDU	SDU W+2, SN = W			
13	-	>	UPLINK RLC PDU	SDU W+2			
14			RB RELEASE	Optional step			
NOTE			ected Sequence shown is infomative.				
	TI	ne UPL	INK and DOWNLINK PDU flows may over	rlap, but are shown separate for clarity.			
				ers given in the comments column shall be			
	CC	onside	red informative only, for test case developn	nent purposes.			

7.2.3.14.5 Test requirements

- 1. The SS shall receive back SDUs 1 to W, and SDU W + 2 only. No other SDUs shall be looped back.
- 2. The SS shall receive a STATUS PDU in step 11a. SN=2W shall not be indicated as received in the STATUS PDU. Negative acknowledgements shall not be indicated for SN=W to 2W-1 either.

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

3GPP TSG- T1 SIG Meeting #26 Luton, UK, 4th – 8th November 2002

			(CHANG	E RE	EQL	JE	ST				CR	P-Form-v6.1
æ		<mark>4.123-1</mark> bec Title:	User	333 Equipment (: Protocol co		forma	ance				^{n:} 5. '	1.1	ж Ж
Proposed	l change a	affects: ೫	(U)		ME/UE	<mark>X</mark> F	Radio	o Ac	cess Netv	work	Со	re Net	work
Title: Source:	ж ж	case as T1	S0206	TC8.1.9 SIG 574rev1	SNALLIN		NNC	ECT	TION REL	EASE	INDIC	ATION	l test
Work iten	n code: ೫	TEI							Date	: ¥ (09/10/20	002	
Category	. ೫	Use <u>one</u> of F (corr A (cor B (add C (fun D (edi	rection) respon- lition of ctional torial m planatic	ds to a correc f feature), modification (podification) ons of the abc	ction in ai of feature	<i>)</i>		lease	2	e of the (G (R (R (R (R (R -4 (R	REL-5 e followir SM Pha elease f elease f elease f elease f elease f	se 2) 1996) 1997) 1998) 1999) 1999)	ises:

Reason for change: ೫	In the test current procedure unexpected sequence is performed by the UE between step 6 and step 8 for the retry the registration a number of times.
Summary of change: ೫	 T3317 to be initiated by SERVICE REQUEST message for PS domain or T3230 (T3240 is initiated for the release RR connection after T3230 is expired) to be initiated by CM SERVICE REQUEST message for CS domain is used so that SIGNALLING CONNECTION RELEASE message is transmitted after these timers are expired.
	2. Initial condition is changed to idle state.
	The revision from T1S020674 is below
	3. Editorial corrections are added in conformance requirment with yellow marker.
Consequences if ॥ ॥ not approved:	Actual test is not performed correctly.
Clauses affected: #	8.1.9
Other specs अ	Other core specifications #

T1S020696

affected:		Test specifications O&M Specifications	
Other comments:	ж	Affects R99, REL-4, REL-5	

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at: <u>http://www.3gpp.org/3G_Specs/CRs.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

8.1.9 Signalling Connection Release Indication

8.1.9.1 Definition

8.1.9.2 Conformance requirement

The UE shall, on receiving a request to release (abort) the signalling connection from upper layers for a specific CN domain:

1> if a signalling connection in the variable ESTABLISHED_SIGNALLING_CONNECTIONS for the specific CN domain identified with the IE "CN domain identity" exists:

2> initiate the signalling connection release indication procedure.

1> otherwise:

The UE shall:

. . .

- 1> set the IE "CN Domain Identity" to the value indicated by the upper layers. The value of the IE indicates the CN domain whose associated signalling connection the upper layers are indicating to be released;
- 1> remove the signalling connection with the identity indicated by upper layers from the variable ESTABLISHED_SIGNALLING_CONNECTIONS;
- 1> transmit a SIGNALLING CONNECTION RELEASE INDICATION message on DCCH using AM RLC.

When the SIGNALLING CONNECTION RELEASE INDICATION message has been submitted to lower layers for transmission the procedure ends.

In order to establish an MM connection, the mobile station proceeds as follows (TS24 008 clause 4.5.1.1, 4.5.1.2, 4.5.3.1)

If no RR connection exists, the MM sublayer requests the RR sublayer to establish an RR connection and enters MM sublayer state WAIT FOR RR CONNECTION (MM CONNECTION). This request contains an establishment cause and a CM SERVICE REQUEST message. When the establishment of an RR connection is indicated by the RR sublayer, the MM sublayer of the mobile station starts timer T3230, gives an indication to the CM entity that requested the MM connection establishment, and enters MM sublayer state WAIT FOR OUTGOING MM <u>CONNECTION.</u>

If T3230 expires (i.e. no response is given but a RR connection is available) the MM connection establishment is aborted and the requesting CM sublayer is informed. If no other MM connection exists then the mobile station shall proceed as described in clause 4.5.3.1 for release of the RR connection. Otherwise the mobile station shall return to the MM sublayer state where the request of an MM connection was received, i.e. to MM sublayer state MM connection active. Other ongoing MM connections (if any) shall not be affected.

If all MM connections are released by their CM entities, and no RRLP procedure (see 3GPP TS 04.31 [23b]) and no LCS procedure over RRC (see 3GPP TS 25.331 [23c]) is ongoing, the mobile station shall set timer T3240 and enter the state WAIT FOR NETWORK COMMAND, expecting the release of the RR connection.

The UE initiates the Service request procedure by sending a SERVICE REQUEST message. The timer T3317 shall be started after the SERVICE REQUEST message has been sent and the UE enters the GMM-SERVICE-REQUEST-INITIATED is entered. The SERVICE REQUEST message shall contain the valid P-TMSI and the Service type shall indicate either signalling or paging response(TS24.008 clause 4.7.13.1). The following abnormal cases can be identified(TS24.008 clause 4.7.13.5).

c) T3317 expired

The UE shall enter GMM-REGISTERED state.

Release 5

If the UE is in PMM-IDLE state then the procedure shall be aborted and the UE shall initiate a PS signalling connection release.

Reference

3GPP TS 25.331 clause 8.1.14, TS24.008 clause 4.5.1.1, 4.5.1.2, 4.5.3.1, 4.7.13.1, 4.7.13.5

8.1.9.3 Test purpose

To confirm that the UE transmits a SIGNALLING CONNECTION RELEASE INDICATION message after <u>upper</u> <u>layer requests to release its signalling connection</u>it fails to receive a response for the LOCATION UPDATING REQUEST message.

8.1.9.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: : Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE. Switched off (state 1) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is-transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by attempting to make an outgoing callturned on and it shall find a suitable cell to camp on. The UE shall initiate a location updating procedure. Then Tthe UE shall establish an RRC connection and transmit a <u>SERVICE ATTACH</u> REQUEST message or a <u>CM SERVICE LOCATION UPDATING</u> REQUEST message using the INITIAL DIRECT TRANSFER message depending on supported CN domain. The SS does not respond to this message, and the UE shall send a SIGNALLING CONNECTION RELEASE INDICATION message which includes the CN domain identity with the same value as that in the INITIAL DIRECT TRANSFER message.

Expected sequence

Step Direction		ction	Message	Comment
-	UE	SS		
1				The UE <u>initiates an outgoing</u> <u>callis powered on</u> .
2		\rightarrow	RRC CONNECTION REQUEST	UE shall initiate the location updating procedure.
3		÷	RRC CONNECTION SETUP	
4				The UE configures the layer 2 and layer 1.
5		\rightarrow	RRC CONNECTION SETUP COMPLETE	
6		→	INITIAL DIRECT TRANSFER	Depending on supported CN domain, includes <u>SERVICE</u> <u>REQUESTATTACH REQUEST</u> message (PS domain) or <u>CM</u> <u>SERVICE</u> <u>REQUESTLOCATION UPDATE</u> <u>REQUEST message (</u> CS domain) <u>message</u> is emdedded in INITIAL DIRECT TRANSFER message.
7				The SS does not respond and waits <u>for T3317 (PS domain) or</u> <u>T3230+T3240 (CS domain</u> <u>Juntil the timer for location</u> update procedure or attach procedure expires.
8		\rightarrow	SIGNALLING CONNECTION RELEASE INDICATION	

Specific Message Content

SIGNALLING CONNECTION RELEASE INDICATION (Step 8)

Information Element	Value/remark
CN domain identity	Check to see if this value is the same as in the uplink
	INITIAL DIRECT TRANSFER message.

8.1.9.5 Test requirement

After step 1 the UE shall initiate the LOCATION UPDATING procedure or ATTACH procedure and establish an RRC connection.

After step 7 the UE shall transmit a SIGNALLING CONNECTION RELEASE INDICATION message which includes the same CN domain identity as that found in the INITIAL DIRECT TRANSFER message.

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

Tdoc **#***T1-020707*

3GPP TSG- T1 SIG Meeting #25 Singapore, 18th – 20th Sept 2002

Tdoc # T1S-020703

		CHANGE	EREQ	UES	Т		CR-Form-v7
ж	34.123-1 CR	<mark>334</mark>	ж rev	- *	Current version:	5.1.1	ж
For <mark>H</mark>	ELP on using this form, se	e bottom of th	is page or	look at t	he pop-up text ove	r the X syr	nbols.

Proposed change affects: UICC apps#

ME X Radio Access Network Core Network

Title:	ж	Corrections to package 1 & 2 idle mode test cases	S	
Source:	ж	Ericsson, Siemens		
Work item code:	:¥	TEI	Date: ೫	15/10/2002
Category:	¥	F	Release: ೫	REL-5
outegory.		Use <u>one</u> of the following categories: F (correction) A (corresponds to a correction in an earlier release B (addition of feature), C (functional modification of feature)	Use <u>one</u> of 2	the following releases: (GSM Phase 2) (Release 1996) (Release 1997) (Release 1998)
		D (editorial modification) Detailed explanations of the above categories can be found in 3GPP TR 21.900.	R99 Rel-4 Rel-5 Rel-6	(Release 1999) (Release 4) (Release 5) (Release 6)

Reason for change: #	1.	For test cases 6.2.1.1, 6.2.1.6, 6.2.1.7, 6.2.1.8 and 6.2.1.9 all cells
Ū		belong to different PLMNs. This means that UE will trigger a location
		registration procedure each time a new cell is selected. However,
		currently Method B (see 34.123-1, section 6) is used in these test cases.
		Method B means that SS is continuously paging the UE on all cells at the
		start of the test and responds to RACH requests from the UE with an
		RRC CONNECTION REJECT message, which causes the UE to return
		to Idle mode. This will cause that the UE will never get registered, and
		will neither select the cell (PLMN) nor display the PLMN. Instead,
		method C should be used. Method C means that SS responds 'normally'
		RACH requests so Location Updating and Calls can be done. This is
		also in alignment with e.g. 6.1.1.3 and 6.1.1.4.
	2.	For some idle mode test cases Qrxlevmin have been specified as even
		numbers. According to 25.331 Qrxlevmin is of type Integer (-11525 by
		step of 2). Srxlev values have also been updated accordingly (Srxlev =
		CPICH_Ec – Qrxlevmin)
	3	Correction of power levels for GSM cells in the TDD case in test case
	0.	6.2.1.9 according to approved changes to the FDD case in CR274 (T1-
		020528) at T#17 (to increase separation between cells).
	4.	Corrections of unit for CPICH_Ec in test case 6.2.2.1
	5.	CPICH_Ec should be used to specify power level of UTRAN cell in test

	case 6.2.2.3.
	 6. For inter-frequency idle mode test cases the cell numbers need to be aligned to the default cell numbers as defined in 34.108, clause 6.1 in SIB11/SIB 12 (cell 1,2 and 3 used to refer to intra-frequency cells; cell 4,5 and 6 used for inter-frequency cells; and cell 7 and 8 used for inter-RAT cells). For test cases using more than 3 inter-frequency cells specific definitions of SIB11 and SIB12 need to be used and the cell number in those test cases will not be associated with the default intra-, inter- and inter-RAT cell numbers as defined in 34.108 clause 6.1. 7. Alignment of conformance requirements for September 02 release of
	core specification (03.22, 05.08, 23.122 and 25.304).
Summary of change: #	1. For test cases 6.2.1.1, 6.2.1.6, 6.2.1.7, 6.2.1.8 and 6.2.1.9: Change from method B to method C
	2. For test cases 6.1.2.1, 6.1.2.8, 6.2.2.1, 6.2.2.2 and 6.2.2.3: Changed Qrxlevmin to be specified as odd numbers.
	 Changed RF signal level for GSM cell 5 and cell6 for the TDD case in cell definition table in clause 6.2.1.9.4 according to the RF signal levels specified for the FDD case.
	 Changed unit of CPICH_Ec from dBm to dBm/3.84 MHz in table for step a-c in clause 6.2.2.1.4.
	 Changed from CPICH_RSCP to CPICH_Ec parameter in table for Cell 3 and step a-c in clause 6.2.2.3.4.
	6. To align with default cell numbers in 34.108 clause 6.1
	a. For test case 6.1.2.1 and 6.1.2.8: Cell 3 changed to Cell 4 (inter-frequency cell).
	 For test cases: 6.1.2.1, 6.1.2.2, 6.1.2.8, 6.2.2.1: Conformance requirements have been aligned to the september-02 release of core specifications (23.122, 25.304 and 05.08)
	 For test cases 6.2.1.6 and 6.2.2.2: Corrections to references to core specifications in conformance requirements.
	Changes from T1S-020591 (Siemens):
	- 6.1.1.4 Method of test: values for TDD updated according to TS 25.123, (clause 9.1) range (-9450) and idle mode tolerancies (clause 4.2)
	- 6.1.2.1.4 Test procedure:
	 values for TDD updated according to TS 25.123, (clause 9.1) range (-94 50) and idle mode tolerances (clause 4.2)
	o correction in Step i)
	\circ Qrxlevmin changed to -79 to clarify the value for S negative.
	- 6.1.2.2.4, 6.1.2.8.4, Method of test: values for TDD updated according to TS 25.123, (clause 9.1) range (-9450) and idle mode tolerances (clause 4.2)
	Changes from T1S-020592 (Siemens):
	Method of test: values for TDD updated according to TS 25.123, (clause 9.1) range (-9450) and idle mode tolerancies (clause 4.2) in clauses 6.2.1.1, 6.2.1.1.4,

	<u>6.2.1.66.2.1.9, 6.2.2.3.4</u>
	Changes based on comments during e-mail approval introduced in T1S-020700 ((revision of T1S-020655):
	Comments from Siemens:
	<mark>- 6.1.2.1.4</mark>
	• Step i-k are for FDD and TDD as well. Deleting of (FDD) is needed.
	 Step I-m (TDD)> Qrxlevmin should be changed to -59.
	- 6.2.1.1.4, 6.2.1.6.4, 6.2.1.7.4, 6.2.1.8.4
	 Power levels for FDD chnaged in accordance to changes as been introduced in table 6.3 and 6.4 (5 dB separation between UTRA cells)
	Changes based on comments during e-mail approval introduced in T1S-020703 ((revision of T1S-020700):
	Comments from Siemens:
	<mark>- 6.1.2.1.4</mark>
	 Test procedure, step I), changed Qrxlevmin value for TDD to -59
Consequences if	육 Good UE will fail
not approved:	Idle mode test cases not aligned with core specification
Clauses affected:	% 6.1.1.4, 6.1.2.1, 6.1.2.2, 6.1.2.8, 6.2.1.1, 6.2.1.6, 6.2.1.7, 6.2.1.8, 6.2.1.9, 6.2.2.1, 6.2.2.2 and 6.2.2.3
	YN
Other specs affected:	X Other core specifications # X Test specifications # X O&M Specifications •
Other comments:	# Affects R99, REL-4 and REL-5 test cases.
	T1S-020655 is a revision of T1S-020640 where changes to generic clause 6.1 is moved (merged into T1S-020643); and changes merged from TDD CRs T1S-020591 and T1S-020592.
	No impact on backwards compatibility to March 02 UE/SS identified.

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.

3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

<Start of modified section>

6.1.1.4 PLMN selection of RPLMN, HPLMN, UPLMN and OPLMN; Automatic mode

6.1.1.4.1 Definition

Test to verify that in Automatic Network Selection Mode, the UE selects PLMNs in a prioritized order. Forbidden PLMNs shall not be selected. If available, the RPLMN shall be selected at switch-on, otherwise the list shall include in priority order HPLMN, User-PLMN and Operator-PLMN. The last priority in the list is "Other PLMN/access technology combinations" which is not included in this test.

Only UTRAN cells and a UE equipped with a USIM with Radio Access Technology fields set to UTRAN are considered.

6.1.1.4.2 Conformance requirement

1. At switch on, or following recovery from lack of coverage, the MS selects the registered PLMN or equivalent PLMN (if it is available) using all access technologies that the MS is capable of and if necessary (in the case of recovery from lack of coverage, see TS 23.122, clause 4.5.2) attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

2. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 2.1 HPLMN (if not previously selected);
- 2.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 2.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 2.5 Other PLMN/access technology combinations in order of decreasing signal quality.

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in the "forbidden LAs for regional provision of service" list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

3. If a "PLMN not allowed" message is received by an MS in response to an LR request from a VPLMN, that VPLMN is added to a list of "forbidden PLMNs" in the SIM and thereafter that VPLMN will not be accessed by the MS when in automatic mode. A PLMN is removed from the "forbidden PLMNs" list if, after a subsequent manual selection of that PLMN, there is a successful LR. This list is retained when the MS is switched off or the SIM is removed. The HPLMN shall not be stored on the list of "forbidden PLMNs".

References

- 1. TS 23.122, clause 4.4.3.1.
- 2. TS 23.122, clause 4.4.3.1.1.
- 3. TS 23.122, clause 3.1.

NOTE: TS 31.102 defines the USIM fields.

6.1.1.4.3 Test purpose

- 1. To verify that if available, the RPLMN is selected at switch-on.
- 2. To verify that in Automatic Network Selection Mode Procedure, the UE selects the RPLMN, HPLMN, UPLMN and OPLMN in a prioritized order.
- 3. To verify that forbidden PLMNs are not selected.

6.1.1.4.4 Method of test

Initial conditions

The UE is in automatic PLMN selection mode.

"IMSI attach" flag in the BCCH is set to allowed.

Cell levels are from table 6.3 (FDD).

All Radio Access Technology USIM fields and cells are UTRAN.

Cell	CPICH_ Ec [dBm/3.84 MHz] (FDD)	P-CCPCH_ RSCP [dBm] (TDD)	Test Channel	PLMN
Cell 1	-60	<mark>-5<u>4</u>0</mark>	1	PLMN 1
Cell 2	-65	-5 <u>9</u> 5	2	PLMN 2
Cell 3	-70	<mark>-6<u>4</u>0</mark>	3	PLMN 3
Cell 4	-75	<mark>-6<u>9</u>5</mark>	4	PLMN 4
Cell 5	-80	<mark>-7<u>4</u>0</mark>	5	PLMN 5
Cell 6	-85	<mark>-7<u>9</u>5</mark>	6	PLMN 6

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	
EFLOCI		PLMN 1	
EFHPLMNWACT	1 st	PLMN 2	
	1 st	PLMN 3	
	2 nd	PLMN 4	
EFOPLMNWACT	1 st	PLMN 5	
	2 nd	PLMN 6	
	PLMN 3		

Test procedure

Method C is applied.

- a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE.
- d) Cell 1 is switched off.
- e) The SS waits for random access requests from the UE.
- f) Cell 2 is switched off.
- g) The SS waits for random access requests from the UE.

- i) Cell 4 is switched off.
- j) The SS waits for random access requests from the UE.
- k) Cell 5 is switched off.
- 1) The SS waits for random access requests from the UE.
- m) Cell 6 is switched off.

6.1.1.4.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1. The displayed PLMN shall be PLMN 1.
- 2) In step e), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN 2.
- 3) In step g), the response from the UE shall be on Cell 4. The displayed PLMN shall be PLMN 4.
- 4) In step j), the response from the UE shall be on Cell 5. The displayed PLMN shall be PLMN 5.
- 5) In step 1), the response from the UE shall be on Cell 6. The displayed PLMN shall be PLMN 6.
- 6) After step m), the UE shall inform that only limited service is possible

<End of modified section>

<Start of next modified section>

6.1.2 Cell selection and reselection

- 6.1.2.1 Cell reselection
- 6.1.2.1.1 Definition

Test to verify that the UE performs the cell reselection correctly for intra/inter-frequency cells if the serving cell becomes barred or S<0.

6.1.2.1.2 Conformance requirement

- 1. When camped on a cell, the UE shall regularly search for a better cell according to the cell reselection criteria. If a better cell is found, that cell is selected. The change of cell may imply a change of RAT.
- 2. A "suitable cell" is a cell on which the UE may camp on to obtain normal service. Such a cell shall fulfil all the following requirements.
 - 2.1 The cell is part of the selected PLMN or of a PLMN considered as equivalent by the UE according to the information provided by the NAS.
 - 2.2 The cell is not barred.
 - 2.3 The cell is not part of the list of "forbidden LAs for roaming".
 - 2.4 The cell selection criteria are fulfilled.
- 3. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 3.1 UE internal triggers, so as to meet performance as specified in TS 25.133 for FDD mode and in TS 25.123 for TDD mode.
 - 3.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified.
- 4. Cell Reselection Criteria:
 - 4.1 The UE shall perform ranking of all cells that fulfil the S criterion.
 - 4.2 The cells shall be ranked according to the R criteria, <u>deriving Qmeas,n and Qmeas,s and calculating the R</u> values using CPICH RSCP, P-CCPCH RSCP and the averaged received signal level as specified in TS 25.133 and TS 25.123 for FDD, TDD and GSM cells, respectively.

The offset Qoffset1_{s,n} is used for Qoffset_{s,n} to calculate R_n , the hysteresis Qhyst1_s is used for Qhyst_s to calculate R_s .

If the usage of HCS is indicated in system information, TEMP_OFFSET1_n is used for TEMP_OFFSET_n to calculate TO_n . If it is indicated in system information that HCS is not used, TEMP_OFFSET_n is not applied when calculating R_n . The best ranked cell is the cell with the highest R value.

If a TDD cell is ranked as the best cell, the UE shall perform cell re-selection to that TDD cell.

If an FDD cell is ranked as the best cell and the quality measure for cell selection and re-selection is set to CPICH RSCP, the UE shall perform cell re-selection to that FDD cell. If a TDD cell is ranked as the best cell, the UE shall perform cell re-selection to that TDD cell.

- 4.3 In all cases, the UE shall reselect the new cell, only if the the following conditions are met:
 - <u>cell reselection criteria are fulfilled the new cell is better ranked than the serving cell</u> during a time interval Treselection.
 - more than 1 second has elapsed since the UE camped on the current serving cell.

- 4.4 The cell ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME. However, TEMP_OFFSET_n and PENALTY_TIME_n are only applicable if the usage of HCS is indicated in system information.
- 5. When cell status "barred" is indicated, the UE shall select another cell according to the following rule:
 - 5.1 If the "Intra-frequency cell re-selection indicator" IE in Cell Access Restriction IE is set to value "allowed", the UE may select another cell on the same frequency if selection/re-selection criteria are fulfilled.
 - If the UE is camping on another cell, the UE shall exclude the barred cell from the neighbouring cell list until the expiry of a time interval T_{barred}. The time interval T_{barred} is sent via system information in a barred cell together with Cell status information in the Cell Access Restriction IE.
 - If the UE does not select another cell, and the barred cell remains to be the "best" one, the UE shall after expiry of the time interval T_{barred} again check whether the status of the barred cell has changed.
 - 5.2 If the "Intra-frequency cell re-selection indicator" IE is set to "not allowed" the UE shall not re-select a cell on the same frequency as the barred cell. For emergency call, the Intra-frequency cell re-selection indicator IE" shall be ignored, i.e. even if it is set to "not allowed" the UE may select another intra-frequency cell.

References

- 1. TS 25.304, clause 5.2.1.
- 2. TS 25.304, clause 4.3.
- 3. TS 25.304, clause 5.2.5.1.
- 4. TS 25.304, clause 5.2.6.1.4.
- 5. TS 25.304, clause 5.3.1.1.

6.1.2.1.3 Test purpose

- 1. To verify that the UE performs cell reselection on the following occasions:
 - 1.1 Serving cell becomes barred;
 - 1.2 S<0 for serving cell.
- 2. To verify conformance requirement 5.
- NOTE: Reselection triggered by the cell becoming a part of a forbidden registration area is tested in clause 9.4.2.3 and clause 9.4.2.4.

6.1.2.1.4 Method of test

Initial conditions

Treselection, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME are not used, so the cell-ranking criterion R equals CPICH_RSCP for FDD cells, and P-CCPCH RSCP for TDD cells.

Step a-c (FDD):

Parameter	Unit	Cell 1	Cell 2	Cell 4
Test Channel		1	1	2
CPICH_Ec	dBm/3.84 MHz	-60	-70	-80
Qrxlevmin	dBm	-115	-115	-115
Srxlev*	dBm	55	45	35
CellBarred		Not barred	Not barred	Not barred

Step a-c (TDD):

Release 5

Parameter	Unit	Cell 1	Cell 2	Cell 43
P-CCPCH RSCP	dBm	-69	<mark>-7<u>4</u>1</mark>	<mark>-7<u>9</u>3</mark>
Qrxlevmin	dBm	-103	-103	-103
Srxlev*	dB	34	29 32	<mark>2430</mark>

Step d-f:

CellBarred	Not barred - > Barred	Not barred	Not barred
Intra-frequency cell re-selection indicator	Not allowed		
Tbarred	10s		

Step g-h:

Intra-frequency cell re-selection indicator		Not Allowed -> Allowed		
---	--	---------------------------	--	--

Step i-k-(FDD):

Parameter	Unit	Cell 1	Cell 2	Cell 🛂
CellBarred		Barred ->	Not barred	Not barred
Celibalieu		Not barred	Not balled	Not barreu

Step 1-m (FDD):

Qrxlevmin	dBm	-115 -> - <u>51<mark>50</mark></u>	-115	-115
Srxlev*	dBm	55 -> - <mark>9</mark> 10	45	35

Step 1-m (TDD):

Qrxlevmin	-103 -> 5 <u>9</u> 68	-103	-103
Srxlev*	<mark>34 -> -<u>10</u>6</mark>	<mark>29<mark>32</mark></mark>	<mark>24</mark> 30

Test procedure

Method B is applied.

- a) The SS activates Cell 12 and 4-3 and monitors them for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE.
- d) The SS sets Cell 1 to be barred. The SS notifies UE of the BCCH modification.
- e) The SS waits for random access requests from the UE.
- f) The SS sets "Intra-frequency cell re-selection indicator" to "Allowed". The SS notifies UE of the BCCH modification.
- g) The SS waits for random access requests from the UE.
- h) The UE is switched off.

- i) The SS activates Cell 1 3 and monitors them for random access requests from the UE. The SS sets Cell 1 to be not barred.
- j) The UE is switched on.
- k) The SS waits for random access requests from the UE.
- For FDD cell, Qrxlevmin is increased to -<u>5150</u> dBm, so S will become negative. For TDD cell, Qrxlevmin is increased to -<u>5968</u>dBm, so S will become negative. The SS notifies UE of the BCCH modification
- m) The SS waits for random access requests from the UE.

6.1.2.1.5 Test requirements

- 1) In step c), after the UE has responded on Cell 1, it shall not respond on any other cell within 1 min.
- 2) In step e), the UE shall respond on Cell **4**.
- 3) In step g), the UE shall respond on Cell 2.
- 4) In step k), after the UE has responded on Cell 1, it shall not respond on any other cell within 1 min.
- 5) In step m), the UE shall respond on Cell 2.

6.1.2.2 Cell reselection using Qhyst, Qoffset and Treselection

6.1.2.2.1 Definition

Test to verify that the UE performs the cell reselection correctly if system information parameters Qoffset, Qhyst and Treselection are applied for non-hierarchical cell structures. TEMP_OFFSET and PENALTY_TIME are only applicable when HCS is applied and are tested in clauses 6.1.2.4 and 6.1.2.5.

6.1.2.2.2 Conformance requirement

- 1. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 1.1 UE internal triggers, so as to meet performance as specified in TS 25.133 for FDD mode and in TS 25.123 for TDD mode.
 - 1.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified.
- 2. Cell Reselection Criteria:
 - 2.1 The UE shall perform ranking of all cells that fulfil the S criterion.
 - 2.2 The cells shall be ranked according to the R criteria, <u>deriving Qmeas,n and Qmeas,s and calculating the R</u> values using CPICH RSCP, P-CCPCH RSCP and the averaged received signal level as specified in TS 25.133 and TS 25.123 for FDD, TDD and GSM cells, respectively.

The offset $Qoffset1_{s,n}$ is used for $Qoffset_{s,n}$ to calculate R_n , the hysteresis $Qhyst1_s$ is used for $Qhyst_s$ to calculate R_s .

If the usage of HCS is indicated in system information, $\text{TEMP}_OFFSET1_n$ is used for TEMP_OFFSET_n to calculate TO_n . If it is indicated in system information that HCS is not used, TEMP_OFFSET_n is not applied when calculating R_{n_n} . The best ranked cell is the cell with the highest R value.

If a TDD cell is ranked as the best cell, the UE shall perform cell re-selection to that TDD cell.

If an FDD cell is ranked as the best cell and the quality measure for cell selection and re-selection is set to CPICH RSCP, the UE shall perform cell re-selection to that FDD cell. If a TDD cell is ranked as the best cell, the UE shall perform cell re-selection to that TDD cell.

- 2.3 In all cases, the UE shall reselect the new cell, only if the the following conditions are met:
 - <u>cell reselection criteria are fulfilled</u>the new cell is better ranked than the serving cell during a time interval Treselection.
 - more than 1 second has elapsed since the UE camped on the current serving cell.
- 2.4 The cell ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME. However, TEMP_OFFSETn and PENALTY_TIMEn are only applicable if the usage of HCS is indicated in system information.

References

- 1. TS 25.304, clause 5.2.5.1.
- 2. TS 25.304, clause 5.2.6.1.4.

6.1.2.2.3 Test purpose

- 1. To verify that the UE calculates R from Qhyst and Qoffset and that the modification of these parameters on the BCCH triggers the cell reselection evaluation process. TEMP_OFFSET and PENALTY_TIME are not applied.
- 2. To verify that the UE reselects the new cell, if the cell reselection criteria are fulfilled during a time interval Treselection.

6.1.2.2.4 Method of test

Initial conditions

For FDD only:

Step a-c:

Parameter	Unit	Cell 1	Cell 2
Test channel		1	1
CPICH_Ec	dBm/3.84 MHz	-60	-70
Qhyst1 _s	dB	20	0
Rs*	dB	-40	-70
R _n *	dB	-70	-60

Step d-e:

CPICH_Ec	dBm/3.84 MHz	-60 -> -70	-70 -> -60	
R _s *	dB	-40 -> -50	-70 -> -60	
R _n *	dB	-70 -> -60	-60 -> -70	

Step f-g:

Qhyst1 _s	dB	20 -> 0	0
R _s *	dB	-50 -> -70	-60
R _n *	dB	-60	-70

Step h-j:

CPICH_Ec	dBm/3.84 MHz	-70 -> -60	-60 -> -70
Qoffset1 _{s,n}	dBm	0 -> 20	0
Rs*	dBm	-70 -> -60	-60 -> -70
R _n *	dBm	-60 -> -90	-70 -> -60

Step k-l:

Release 5

	CPICH_Ec	dBm/3.84 MHz	-60 -> -70	-70 -> -60
	Rs* Rn*	dB	-60 -> -70	-70 -> -60
	R _n *	dB	-90 -> -80	-60 -> -70
Step m-n:				
	Qoffset1 _{s,n}	dB	20 -> 0	0
	R _s *	dB	-70	-60
	R _n *	dB	-80 -> -60	-70
Step o-p:				
	Treselection _s	S	30	0
For TDD only: Step a-c:				
	Parameter	Unit	Cell 1	Cell 2
	P-CCPCH RSCP		-69 8	-794
	Qhyst1 _s	dB	10	0
	R _s *	dB	<mark>-598</mark>	<mark>-794</mark>
	R _n *	dB	<mark>-794</mark>	-698
Step d-e:				
•				
	P-CCPCH RSCP		-6 <u>9</u> 8 -> -7 <u>9</u> 1	-7 <u>9</u> 4 -> -6 <u>9</u> 8
	Rs*	dB	-5 <u>9</u> 8 -> -6 <u>9</u> 1	<u>-79 ->-69</u> 8
~ ^	R _n *	dB	-7 <u>9</u> 4 -> -6 <u>9</u> 8	<u>-69 ->-79</u> 4
Step f-g:				
	Qhyst1 _s	dB	10 -> 0	0
	Rs*	dB	-6 <u>9</u> 1 -> - <u>79</u> 71	-6 <u>9</u> 8
	R _n *	dB	<mark>-6<u>9</u>8</mark>	<mark>-7<u>9</u>4</mark>
Step h-j:				
	P-CCPCH RSCP	dBm	-7 <u>9</u> 4 -> -6 <u>9</u> 8	-6 <u>9</u> 8 ->-7 <u>9</u> 1
	Qoffset1 _{s,n}	dB	0 -> 10	0
	Rs*	dB	- 7 <u>9</u> 4 ->-6 <u>9</u> 8	<u>-69 -> -79</u> 4
	R _n *	dB	- 6 <u>9</u> 8 -> - <u>79</u> 81	<u>-79 -> -69</u> 8
Step k-1:				
	P-CCPCH RSCP	dBm	-6 <u>9</u> 8 -> -7 <u>9</u> 4	-7 <u>9</u> 4 -> -6 <u>9</u> 8
	Rs*	dB	-6 <u>9</u> 8 -> -7 <u>9</u> 1	-7 <u>9</u> 4 -> -6 <u>9</u> 8
	R _n *	dB	- <u>79</u> 81 -> - <u>89</u> 78	-6 <u>9</u> 8 -> -7 <u>9</u> 1
Step m-n:				
	Qoffset1 _{s,n}	dB	10 -> 0	0
	Rs*	dB	-794	-6 <mark>98</mark>
	R _n *	dB	- <u>89</u> 78 -> - <u>79</u> 68	-7 <u>9</u> 1
Step o-p:	. <u></u>	·		
	Treselections	S	0 -> 30	0
	<u>`</u>			

33

Test procedure

Method B is applied.

- a) The SS activates Cell 1 and 2 and monitors them for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits to see if there is any random access requests from the UE.
- d) The SS changes the level of Cell 1 and 2 and waits for 10 s (TS 25.133, A.4.2.1.2 for FDD mode and TS 25.123, A.4.2.1.2 for TDD mode).

- e) The SS waits for random access requests from the UE.
- f) The SS resets Qhyst for Cell 1 and notifies UE of the BCCH modification.
- g) The SS waits for random access requests from the UE.
- h) The UE is switched off. The SS changes the level of Cell 1 and Cell 2. The SS changes Qoffset in Cell 1.
- i) The UE is switched on.
- j) The SS waits to see if there is any random access requests from the UE.
- k) The SS changes the level of Cell 1 and 2 and waits for 10 s (TS 25.133, clause A.4.2.1.2 for FDD mode and TS 25.123, clause A.4.2.1.2 for TDD mode).
- 1) The SS waits for random access requests from the UE.
- m) The SS resets Qoffset for Cell 1, and notifies UE of the BCCH modification.
- n) The SS waits for random access requests from the UE.
- o) Step h-n) is repeated except that Treselection is 30 s

6.1.2.2.5 Test Requirements

- 1) In step c), after the UE has responded on Cell 1, it shall not respond on any other cell within 1 min.
- 2) In step e), the UE shall keep responding on Cell 1.
- 3) In step g), the UE shall respond on Cell 2.
- 4) In step j), the UE shall select a cell to camp on and eventually make a reselection to Cell 1.
- 5) In step l), the UE shall keep responding on Cell 1.
- 6) In step n), the UE shall respond on Cell 2.
- 7) In step o), the UE shall respond as in previous steps except that when reselecting to Cell 2, there shall be no response from the UE on Cell 2 within 28 s of notifying UE on the BCCH modification, but the UE shall respond on Cell 2 within 35 s.
- NOTE: Minimum time set by Treselection 2 s tolerance. Maximum time set by Treselection + 1 280 msec. for DRX cycle + 1280 ms for system information block type scheduling + 2 s tolerance.

<End of modified section>

<Start of next modified section>

6.1.2.8 Cell reselection: Equivalent PLMN

6.1.2.8.1 Definition

Test to verify that the UE performs the cell reselection correctly to a cell belonging to a PLMN Equivalent to the registered PLMN, if the serving cell of registered PLMN becomes barred or S<0.

6.1.2.8.2 Conformance requirement

- 1. When camped on a cell, the UE shall regularly search for a better cell according to the cell reselection criteria. If a better cell is found, that cell is selected. The change of cell may imply a change of RAT.
- 2. A "suitable cell" is a cell on which the UE may camp on to obtain normal service. Such a cell shall fulfil all the following requirements.
 - 2.1 The cell is part of the selected PLMN or of a PLMN considered as equivalent by the UE according to the information provided by the NAS
 - 2.2 The cell is not barred
 - 2.3 The cell is not part of the list of "forbidden LAs for roaming"
 - 2.4 The cell selection criteria are fulfilled
- 3. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 3.1 UE internal triggers, so as to meet performance as specified in TS 25.133 for FDD mode and in TS 25.123 for TDD mode.
 - 3.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified
- 4. Cell Reselection Criteria:
 - 4.1 The UE shall perform ranking of all cells that fulfil the S criterion
 - 4.2 The cells shall be ranked according to the R criteria, <u>deriving Qmeas,n and Qmeas,s and calculating the R values using CPICH RSCP, P-CCPCH RSCP and the averaged received signal level as specified in TS 25.133 and TS 25.123 for FDD, TDD and GSM cells, respectively.</u>

The offset $Qoffset1_{s,n}$ is used for $Qoffset_{s,n}$ to calculate R_n , the hysteresis $Qhyst1_s$ is used for $Qhyst_s$ to calculate R_s .

If the usage of HCS is indicated in system information, TEMP_OFFSET1_n is used for TEMP_OFFSET_n to calculate TO_n . If it is indicated in system information that HCS is not used, TEMP_OFFSET_n is not applied when calculating R_{n} . The best ranked cell is the cell with the highest R value.

If a TDD cell is ranked as the best cell, the UE shall perform cell re-selection to that TDD cell.

If an FDD cell is ranked as the best cell and the quality measure for cell selection and re-selection is set to CPICH RSCP, the UE shall perform cell re-selection to that FDD cell. If a TDD cell is ranked as the best cell, the UE shall perform cell re-selection to that TDD cell.

4.3 In all cases, the UE shall reselect the new cell, only if the the following conditions are met:

cell reselection criteria are fulfilled the new cell is better ranked than the serving cell during a time interval Treselection.

more than 1 second has elapsed since the UE camped on the current serving cell.

- 4.4 The cell ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME. However, TEMP_OFFSET_n and PENALTY_TIME_n are only applicable if the usage of HCS is indicated in system information.
- 5. When cell status "barred" is indicated, the UE shall select another cell according to the following rule:
 - 5.1 If the "Intra-frequency cell re-selection indicator" IE in Cell Access Restriction IE is set to value "allowed", the UE may select another cell on the same frequency if selection/re-selection criteria are fulfilled.
 - If the UE is camping on another cell, the UE shall exclude the barred cell from the neighbouring cell list until the expiry of a time interval T_{barred}. The time interval T_{barred} is sent via system information in a barred cell together with Cell status information in the Cell Access Restriction IE.
 - If the UE does not select another cell, and the barred cell remains to be the "best" one, the UE shall after expiry of the time interval T_{barred} again check whether the status of the barred cell has changed.
 - 5.2 If the "Intra-frequency cell re-selection indicator" IE is set to "not allowed" the UE shall not re-select a cell on the same frequency as the barred cell. For emergency call, the Intra-frequency cell re-selection indicator IE" shall be ignored, i.e. even if it is set to "not allowed" the UE may select another intra-frequency cell.

References

- 1. TS 25.304, 5.2.1
- 2. TS 25.304, 4.3
- 3. TS 25.304, 5.2.5.1
- 4. TS 25.304, 5.2.6.1.4
- 5. TS 25.304, 5.3.1.1

6.1.2.8.3 Test purpose

- 1. To confirm that the UE treats the cell of the equivalent PLMN as a cell of the current PLMN.
- 2. To verify that the UE performs cell reselection on the following occasions:
 - 2.1 Serving cell becomes barred.
 - 2.2 S<0 for serving cell.
- 3. To verify conformance requirement 5.
- NOTE: Reselection triggered by the cell becoming a part of a forbidden registration area is tested in clause 9.4.2.3 "Location updating / rejected / location area not allowed" and 9.4.2.4 "Location updating / rejected / roaming not allowed in this LA".

6.1.2.8.4 Method of test

Initial conditions

Treselection, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME are not used, so the cell-ranking criterion R equals CPICH_RSCP for FDD cells, and P-CCPCH RSCP for TDD cells.

The UE is Idle Updated on PLMN1 in cell 1, and The SS includes PLMN 2 and PLMN 3 under IE 'Equivalent PLMN' during Idle Update Procedure.

Cell 1 indicates in System Information Block Type 11 that Cell 2 and Cell 14 are neighbouring cells.

Cell 2 indicates in System Information Block Type 11 that Cell 1 and Cell 1 are neighbouring cells.

Cell 🖶 indicates in System Information Block Type 11 that Cell 1 and Cell 2 are neighbouring cells.

Step a-c (FDD):

Parameter	Unit	Cell 1	Cell 2	Cell 🛂
Test Channel		1	1	2
PLMN		PLMN 1	PLMN 2	PLMN 3
CPICH_Ec	dBm/ 3.84 MHz	-60	-70	-80
Qrxlevmin	dBm	-115	-115	-115
Srxlev*	dB	55	45	35
CellBarred		Not barred	Not barred	Not barred

Step a-c (TDD):

Parameter	Unit	Cell 1	Cell 2	Cell 4
Test Channel		1	<mark>1</mark>	2
PLMN		PLMN 1	PLMN 2	PLMN 3
P-CCPCH RSCP	dBm	-69	<mark>-7<u>4</u>1</mark>	<mark>-7<u>9</u>3</mark>
Qrxlevmin	dBm	-103	-103	-103
Srxlev*	dB	34	<mark>2932</mark>	<u>24</u> 30

Step d-f:

CellBarred		Not barred- >Barred	
Intra-frequency cell re-selection indicator		Not Allowed	
Tbarred	S	10	

Step g-h:

Intra-frequency cell re-selection indicator		Not Allowed -> Allowed			
---	--	---------------------------	--	--	--

Step i-k:

Parameter	Unit	Cell 1	Cell 2	Cell 4
CellBarred		Barred ->	Not barred	Not barred
Celibarieu		Not barred	Not balled	Not balled

Step 1-m (FDD):

Qrxlevmin	dBm	-115 -> - <u>51<mark>50</mark></u>	-115	-115
Srxlev*	dBm	55 -> - <u>9</u> 10	45	35

Step l-m (TDD):

Qrxlevmin	DBm	-103 -> - 60 <u>59</u>	-103	-103
Srxlev*	dBm	34 -> - <u>10</u> 9	<mark>29</mark> 32	<mark>24</mark> 30

Test procedure

Method C is applied.

- a) The SS activates Cell 1 2 and 4 and monitors them for random access requests from the UE.
- b) The UE is switched on.

30

- c) The SS waits for random access requests from the UE. A complete Location Update is done. SS specifies PLMN 2 and 3 are Equivalent to PLMN 1 in Location Update Accept Message.
- d) The SS sets Cell 1 to be barred. The SS notifies UE of the BCCH modification.
- e) The SS waits for random access requests from the UE. A complete Location Update is done. SS specifies PLMN 1 and 2 are Equivalent to PLMN 3 in Location Update Accept Message.
- f) The SS sets "Intra-frequency cell re-selection indicator" to "Allowed".
- g) The SS waits for random access requests from the UE. A complete Location Update is done.
- h) The UE is switched off.
- i) The SS activates Cell 1 2 and 4 2 and monitors them for random access requests from the UE.
- j) The UE is switched on.
- k) The SS waits for random access requests from the UE. A complete Location Update is done. SS specifies PLMN 2 and 3 are Equivalent to PLMN 1 in Location Update Accept Message.
- 1) For FDD cell, Qrxlevmin is increased to -5150 dBm, so S will become negative. For TDD cell, Qrxlevmin is increased to -5960 dBm, so S will become negative. The SS notifies UE of the BCCH modification
- m) The SS waits for random access requests from the UE.

6.1.2.8.5 Test requirements

- 1) In step c), after the UE has responded on Cell 1, it shall not respond on any other cell within 1 min.
- 2) In step e), the UE shall respond on Cell
- 3) In step g), the UE shall respond on Cell 2.
- 4) In step k), after the UE has responded on Cell 1, it shall not respond on any other cell within 1 min.
- 5) In step m), the UE shall respond on Cell 2.

6.2 Multi-mode environment (2G/3G case)

6.2.1 PLMN and RAT selection and reselection

- 6.2.1.1 Selection of the correct PLMN and associated RAT
- 6.2.1.1.1 Definition

Test to verify that the UE selects the correct combination of PLMN and associated access technology according to the fields on the USIM.

6.2.1.1.2 Conformance requirement

1. At switch on, or following recovery from lack of coverage, the MS selects the registered PLMN or equivalent PLMN (if it is available) using all access technologies that the MS is capable of and if necessary (in the case of recovery from lack of coverage, see TS 23.122, clause 4.5.2) attempts to perform a Location Registration.

If successful registration is achieved, the MS indicates the selected PLMN.

If there is no registered PLMN, or if registration is not possible due to the PLMN being unavailable or registration failure, the MS follows either Automatic or Manual Network Selection Mode Procedure depending on its operating mode.

Release 5

- 2. The "HPLMN Selector with Access Technology", "User Controlled PLMN Selector with Access Technology" and "Operator Controlled PLMN Selector with Access Technology" data fields in the SIM include associated access technologies for each PLMN entry. The PLMN/access technology combinations are listed in priority order. If an entry includes more than one access technology, then no priority is defined for the preferred access technology and the priority is an implementation issue.
- 3. To allow provision for multiple HPLMN codes, the HPLMN access technologies are stored on the SIM together with PLMN codes. This version of the specification does not support multiple HLPMN codes and the "HPLMN Selector with Access Technology" data field is only used by the MS to get the HPLMN access technologies. The HPLMN code is the PLMN code included in the IMSI.
- 4. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 4.1 HPLMN (if not previously selected);
- 4.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 4.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 4.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 4.5 Other PLMN/access technology combinations in order of decreasing signal quality.

References

- 1. TS 23.122, clause 4.4.3.1.
- 2. TS 23.122, clause 4.4.3
- 3. TS 23.122, clause 4.4.3
- 4. TS 23.122, clause 4.4.3.1.1

NOTE: TS 31.102 defines the USIM fields.

6.2.1.1.3 Test purpose

1. To verify that the UE selects the correct combination of HPLMN/access technology combination according to the fields on the USIM.

6.2.1.1.4 Method of test

Initial conditions

The UE is in automatic PLMN selection mode.

Cell levels are from tables 6.3 and 6.4.

Cell	CPICH_Ec / RF signal level [dBm/3.84 MHz]	P-CCPCH_RSCP/ RF signal level [dBm] (TDD)	Test Channel	PLMN	Radio Access Technology
Cell 1	-48	-48	1	PLMN 1	GSM
Cell 2	- <mark>7072</mark>	- <u>59</u> 61	1	PLMN 1	UTRAN
Cell 3	-75	-64	2	PLMN 2	UTRAN
Cell 4	-50	-50	2	PLMN 2	GSM

The UE is equipped with a USIM containing default values except for those listed below.

USIM A

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI			
EF _{HPLMNwAcT}	1 st	PLMN 1	GSM
	2 nd	PLMN 1	UTRAN

USIM B

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI			
EFHPLMNWACT	1 st	PLMN 2	UTRAN
	2 nd	PLMN 2	GSM

Test procedure

Method \underline{CB} is applied.

- a) The SS activates cells 1-4 and monitors the cells for random access requests from the UE. The UE shall have a USIM with settings according to USIM A.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE.
- d) The UE is switched off and a USIM with settings according to USIM B is inserted.
- e) The UE is switched on.
- f) The SS waits for random access requests from the UE.

6.2.1.1.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1. The displayed PLMN shall be PLMN1 (GSM).
- 2) In step f), the response from the UE shall be on Cell 3. The displayed PLMN shall be PLMN2 (UTRAN).

<End of modified section>

<Start of next modified section>

6.2.1.6 Selection of RAT for HPLMN; Automatic mode

6.2.1.6.1 Definition

Test to verify that the UE selects the HPLMN RAT according to the HPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall try to obtain registration on the same PLMN using other UE-supported RATs.

6.2.1.6.2 Conformance requirement

- To allow provision for multiple HPLMN codes, the HPLMN access technologies are stored on the SIM together with PLMN codes. This version of the specification does not support multiple HLPMN codes and the "HPLMN Selector with Access Technology" data field is only used by the MS to get the HPLMN access technologies. The HPLMN code is the PLMN code included in the IMSI.
- 2. The "HPLMN Selector with Access Technology", "User Controlled PLMN Selector with Access Technology" and "Operator Controlled PLMN Selector with Access Technology" data fields in the SIM include associated access technologies for each PLMN entry. The PLMN/access technology combinations are listed in priority order. If an entry includes more than one access technology, then no priority is defined for the preferred access technology and the priority is an implementation issue.
- 3. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 3.1 HPLMN (if not previously selected);
- 3.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 3.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 3.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 3.5 Other PLMN/access technology combinations in order of decreasing signal quality.

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in the "forbidden LAs for regional provision of service" list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

References

- 1. TS 23.122, clause 4.4.3.
- 2. TS 23.122, clause 4.4.3.1.1 (f).
- 3. TS 23.122, clause 4.4.3.1.1.

NOTE: TS 31.102 defines the USIM fields.

6.2.1.6.3 Test purpose

- 1. To verify that:
 - 1.1 the UE searches for a HPLMN RAT according to the HPLMN Selector with Access Technology data field on the USIM in priority order.

1.2 If no RAT on the priority list is available, the UE tries to obtain registration on the same PLMN using other UE-supported RATs.

6.2.1.6.4 Method of test

Initial conditions

The UE is in automatic PLMN selection mode.

Cell levels are from tables 6.3 and 6.4.

Ce) 	CPICH_Ec / RF signal level [dBm/3.84 MHz] (FDD)	P-CCPCH_RSCP / RF signal level [dBm] (TDD)	Test Channel	PLMN	Radio Access Technology
Cel	11	- <mark>70</mark> 72	- <u>59</u> 61	1	PLMN 2	UTRAN
Cel	12	-48	-48	1	PLMN 2	GSM
Cel	13	-75	-64	2	PLMN 3	UTRAN
Cel	14	-50	-50	2	PLMN 3	GSM

The UE is equipped with a USIM containing default values except for those listed below.

USIM A

USIM field	Priority	PLMN	Access Technology Identifier
		PLMN 1	
EFHPLMNWACT	1 st	PLMN 2	UTRAN
	2 nd	PLMN 2	GSM

USIM B

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI		PLMN 1	
EFHPLMNWACT	1 st	PLMN 2	UTRAN
	2 nd		

Test procedure

Method \underline{CB} is applied.

- a) The SS activates cells 1-4 and monitors the cells for random access requests from the UE. The UE shall have a USIM with settings according to USIM A.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE.
- d) The UE is switched off and a USIM with settings according to USIM A is again inserted. All cells except Cell 1 are active.
- e) The SS waits for random access requests from the UE.
- f) The UE is switched off and a USIM with settings according to USIM B is inserted. All cells except Cell 1 are active.
- g) The UE is switched on.
- h) The SS waits for random access requests from the UE.

6.2.1.6.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{HPLMNwAcT}). The displayed PLMN shall be PLMN2 (UTRAN).
- 2) In step e), the response from the UE shall be on Cell 2 (2nd priority RAT for EF_{HPLMNwAcT}). The displayed PLMN shall be PLMN2 (GSM).
- 3) In step h), the response from the UE shall be on Cell 2. The displayed PLMN shall be PLMN2 (GSM). (PLMN2 is not available on UTRAN so registration on the same PLMN is attempted using other UE-supported RATs).

6.2.1.7 Selection of RAT for UPLMN; Automatic mode

6.2.1.7.1 Definition

Test to verify that the UE selects the UPLMN RAT according to the UPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall not try to obtain registration on the same PLMN(s) with other RAT(s) but instead search for PLMNs in the OPLMN list.

6.2.1.7.2 Conformance requirement

1. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 1.1 HPLMN (if not previously selected);
- 1.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality.

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in the "forbidden LAs for regional provision of service" list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

References

- 1. TS 23.122, clause 4.4.3.1.1.
- NOTE: TS 31.102 defines the USIM fields.

6.2.1.7.3 Test purpose

- 1. To verify that:
 - 1.1 the UE selects the UPLMN RAT according to the UPLMN RAT priority list on the USIM.
 - 1.2 If no RAT on the list is available, the UE does not try to obtain registration on the same PLMN with another RAT but instead searches for PLMNs in the OPLMN list.

6.2.1.7.4 Method of test

Initial conditions

The UE is in automatic PLMN selection mode.

Cell levels are from tables 6.3 and 6.4.

In system information broadcast in each cell, the neighbouring cell list does not contain any other cell belonging to the same PLMN.

31

Cell	CPICH_Ec / RF signal level [dBm/3.84 MHz] (FDD)	P-CCPCH / RF signal level [dBm] (TDD)	Test Channel	PLMN	Radio Access Technology
Cell 1	- <mark>7072</mark>	- <mark>59</mark> 61	1	PLMN 3	UTRAN
Cell 2	-48	-48	1	PLMN 3	GSM
Cell 3	-75	-64	2	PLMN 4	UTRAN
Cell 4	-50	-50	2	PLMN 4	GSM
Cell 5	- <mark>80</mark> 78	- <mark>6<u>9</u>7</mark>	3	PLMN 5	UTRAN

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI		PLMN 1	
EF _{HPLMNwAcT}	1 st	PLMN 2	UTRAN
	2 nd	PLMN 2	GSM
EFPLMNwAcT	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EFOPLMNWACT	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM

Test procedure

Method \underline{CB} is applied.

- a) The SS activates cells 1-5 and monitors the cells for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE.
- d) Cell 1 is switched off.
- e) The SS waits for random access requests from the UE.
- f) Cell 4 is switched off.
- g) The SS waits for random access requests from the UE.

6.2.1.7.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{PLMNwAcT}). The displayed PLMN shall be PLMN3 (UTRAN).
- 2) In step e), the response from the UE shall be on Cell 4 (2nd priority RAT for EF_{PLMNwAcT}). The displayed PLMN shall be PLMN4 (GSM).
- 3) In step g), the response from the UE shall be on Cell 5 (1st priority RAT for EF_{OPLMNwAcT}). The displayed PLMN shall be PLMN5 (UTRAN).

6.2.1.8 Selection of RAT for OPLMN; Automatic mode

6.2.1.8.1 Definition

Test to verify that the UE selects the OPLMN RAT according to the OPLMN RAT priority list on the USIM. If no RAT on the list is available, the UE shall not try to obtain registration on the same PLMN(s) with other RAT(s) but instead search for other PLMN/access technology combinations with received high quality signal in random order.

6.2.1.8.2 Conformance requirement

1. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 1.1 HPLMN (if not previously selected);
- 1.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality.

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in the "forbidden LAs for regional provision of service" list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

References

- 1. TS 23.122, clause 4.4.3.1.1.
- NOTE: TS 31.102 defines the USIM fields.

6.2.1.8.3 Test purpose

- 1. To verify that:
 - 1.1 the UE selects the OPLMN RAT according to the OPLMN RAT priority list on the USIM.
 - 1.2 If no RAT on the list is available, the UE does not try to obtain registration on the same PLMN(s) with other RAT(s) but instead searches for "other PLMN/access technology combinations with received high quality signal in random order".

6.2.1.8.4 Method of test

Initial conditions

The UE is in automatic PLMN selection mode.

Cell levels are from tables 6.3 and 6.4.

In system information broadcast in each cell, the neighbouring cell list does not contain any other cell belonging to the same PLMN.

Cell	CPICH_Ec / RF signal level [dBm/3.84 MHz] (FDD)	P-CCPCH_RSCP / RF signal level [dBm] (TDD)	Test Channel	PLMN	Radio Access Technology
Cell 1	- <mark>70</mark> 72	- <mark>59</mark> 61	1	PLMN 5	UTRAN
Cell 2	-48	-48	1	PLMN 5	GSM
Cell 3	-75	-64	2	PLMN 6	UTRAN
Cell 4	-50	-50	2	PLMN 6	GSM
Cell 5	- <mark>80</mark> 78	- <mark>6<u>9</u>7</mark>	3	PLMN 7	UTRAN

33

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI		PLMN 1	
EFHPLMNWACT	1 st	PLMN 2	UTRAN
	2 nd	PLMN 2	GSM
EFPLMNWACT	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EFOPLMNWACT	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM

Test procedure

Method \underline{CB} is applied.

- a) The SS activates cells 1-5 and monitors the cells for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE.
- d) Cell 1 is switched off.
- e) The SS waits for random access requests from the UE.
- f) Cell 4 is switched off.
- g) The SS waits for random access requests from the UE.

6.2.1.8.5 Test Requirements

- 1) In step c), the response from the UE shall be on Cell 1 (1st priority RAT for EF_{OPLMNwAcT}). The displayed PLMN shall be PLMN5 (UTRAN).
- 2) In step e), the response from the UE shall be on Cell 4 (2nd priority RAT for EF_{OPLMNwAcT}). The displayed PLMN shall be PLMN6 (GSM).
- 3) In step g), the response from the UE shall be on either Cell 2, 3 or 5 (other PLMN/access technology combination) with associated PLMN5 (GSM), PLMN6 (UTRAN) or PLMN7 (UTRAN) shown.

6.2.1.9 Selection of "Other PLMN / access technology combinations"; Automatic mode

6.2.1.9.1 Definition

Test to verify that if neither RPLMN, HPLMN, UPLMN nor OPLMN is available, the UE first tries to obtain registration on "Other PLMN/access technology combinations with received high quality signal in random order" and secondly on "Other PLMN/access technology combinations in order of decreasing signal quality".

6.2.1.9.2 Conformance requirement

1. Automatic Network Selection Mode Procedure:

The MS selects and attempts registration on other PLMNs, if available and allowable in the following order:

- 1.1 HPLMN (if not previously selected);
- 1.2 Each PLMN in the "User Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.3 Each PLMN in the "Operator Controlled PLMN Selector with Access Technology" data field in the SIM (in priority order);
- 1.4 Other PLMN/access technology combinations with received high quality signal in random order;
- 1.5 Other PLMN/access technology combinations in order of decreasing signal quality.

In 1.5, the MS shall order the PLMN/access technology combinations in order of decreasing signal quality within each access technology. The order between PLMN/access technology combinations with different access technologies is an MS implementation issue.

If successful registration is achieved, the MS indicates the selected PLMN.

If registration cannot be achieved because no PLMNs are available and allowable, the MS indicates "no service" to the user, waits until a new PLMN is available and allowable and then repeats the procedure.

If there were one or more PLMNs which were available and allowable, but an LR failure made registration on those PLMNs unsuccessful or an entry in the "forbidden LAs for regional provision of service" list prevented a registration attempt, the MS selects the first such PLMN again and enters a limited service state.

- 2. UTRA case: The UE shall scan all RF channels in the UTRA band according to its capabilities to find available PLMNs. On each carrier, the UE shall search for the strongest cell according to the cell search procedures (for FDD, see TS 25.214, and TDD, see TS 25.224) and read its system information, in order to find out which PLMN the cell belongs to. If the UE can read the PLMN identity, the found PLMN shall be reported to the NAS as a high quality PLMN (but without the RSCP value), provided that the following high quality criterion is fulfilled:
 - For an FDD cell, the measured primary CPICH RSCP value shall be greater than or equal to -95 dBm.
 - For a TDD cell, the measured P-CCPCH RSCP shall be greater than or equal to -84 dBm.

Found PLMNs that do not satisfy the high quality criterion, but for which the UE has been able to read the PLMN identities are reported to the NAS together with the CPICH RSCP value for UTRA FDD cells and P-CCPCH RSCP for UTRA TDD cells.

3. GSM case: A PLMN shall be understood to be received with high quality signal if the signal level is above -85 dBm.

References

- 1. TS 23.122, clause 4.4.3.1.1.
- 2. TS 25.304, clause 5.1.2.2.
- 3. TS 03.22, clause 4.4.3.
- NOTE: TS 31.102 defines the USIM fields.

6.2.1.9.3 Test purpose

- 1. To verify that:
 - 1.1 If neither RPLMN, HPLMN, UPLMN nor OPLMN is available, the UE tries to obtain registration on "Other PLMN/access technology combinations with received high quality signal in random order".
 - 1.2 If no PLMN is available in test purpose 1.1, the UE tries to obtain registration on "Other PLMN/access technology combinations in order of decreasing signal quality".
- 2. The "random order" in test purpose 1.1 is not verified.

6.2.1.9.4 Method of test

Initial conditions

The UE is in automatic PLMN selection mode.

Cell	CPICH_Ec /RF signal level [dBm/3.84 MHz] (FDD)	P- CCPCH_RSCP / RF signal level [dBm] (TDD)	"High Quality signal"	Test Channel	PLMN	Radio Access Technology
Cell 1	-80	-69	Yes	1	PLMN 7	UTRAN
Cell 2	-65	-65	Yes	1	PLMN 8	GSM
Cell 3	-103	<mark>-9<u>4</u>3</mark>	No	2	PLMN 9	UTRAN
Cell 4	-110	<mark>-9<u>9</u>0100</mark>	No	3	PLMN 10	UTRAN
Cell 5	-90	<mark>-88</mark>	No	3	PLMN 11	GSM
Cell 6	-95	<mark>-91</mark>	No	4	PLMN 12	GSM

The UE is equipped with a USIM containing default values except for those listed below.

USIM field	Priority	PLMN	Access Technology Identifier
EFLOCI		PLMN 1	
EFHPLMNWACT	1 st	PLMN 2	UTRAN
	2 nd	PLMN 2	GSM
EFPLMNWACT	1 st	PLMN 3	UTRAN
	2 nd	PLMN 4	GSM
EFOPLMNWACT	1 st	PLMN 5	UTRAN
	2 nd	PLMN 6	GSM

Test procedure

Method \underline{CB} is applied.

- a) The SS activates cells 1-6 and monitors the cells for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access requests from the UE.
- d) The cell on which a response was received, is switched off.
- e) Step c-d) is repeated until the UE informs that no network is available.

6.2.1.9.5 Test Requirements

- 1) In step c), the displayed PLMN is noted.
- 2) When the test procedure has finished, the noted PLMNs shall have appeared in the following order: PLMN7, PLMN8 in random order followed by the other PLMNs. PLMN9 shall come before PLMN10 and PLMN11 shall come before PLMN12.

6.2.2 Cell selection and reselection

6.2.2.1 Cell reselection if cell becomes barred or S<0; UTRAN to GSM

6.2.2.1.1 Definition

Test to verify that if both a GSM and UTRAN network is available, the UE performs cell reselection from UTRAN to GSM if the UTRAN cell becomes barred or S falls below zero.

6.2.2.1.2 Conformance requirement

- 1. When camped on a cell, the UE shall regularly search for a better cell according to the cell reselection criteria. If a better cell is found, that cell is selected. The change of cell may imply a change of RAT.
- 2. A "suitable cell" is a cell on which the UE may camp on to obtain normal service. Such a cell shall fulfil all the following requirements.
 - 2.1 The cell is part of the selected PLMN or, of a PLMN considered as equivalent by the UE according to the information provided by the NAS.
 - 2.2 The cell is not barred.
 - 2.3 The cell is not part of the list of "forbidden LAs for roaming".
 - 2.4 The cell selection criteria are fulfilled.
- 3. When camped normally, the UE shall execute the cell reselection evaluation process on the following occasions/triggers:
 - 3.1 UE internal triggers, so as to meet performance as specified in TS 25.133 for FDD and TS 25.123 for TDD.
 - 3.2 When information on the BCCH used for the cell reselection evaluation procedure has been modified.
- 4. Cell Reselection Criteria:
 - 4.1 The UE shall perform ranking of all cells that fulfil the S criterion.
 - 4.2 The cells shall be ranked according to the R criteria-specified above, deriving Qmeas,n and Qmeas,s and calculating the R values using CPICH RSCP, P-CCPCH RSCP and the averaged received signal level as specified in TS 25.133 and TS 25.123 for FDD, TDD and GSM cells, respectively.

<u>The offset Qoffset1_{s,n} is used for Qoffset_{s,n} to calculate R_n , the hysteresis Qhyst1_s is used for Qhysts to calculate R_s .</u>

If the usage of HCS is indicated in system information, TEMP_OFFSET1_n is used for TEMP_OFFSET_n to calculate TO_n . If it is indicated in system information that HCS is not used, TEMP_OFFSET_n is not applied when calculating R_{n_n} . The best ranked cell is the cell with the highest R value.

If a TDD or GSM cell is ranked as the best cell, then the UE shall perform cell re-selection to that TDD or GSM cell.

4.3 In all cases, the UE shall reselect the new cell, only if the the following conditions are met:

<u>cell reselection criteria are fulfilled the new cell is better ranked than the serving cell</u> during a time interval Treselection.

- more than 1 second has elapsed since the UE camped on the current serving cell.
- 4.4 The cell ranking criterion R is derived from Q, Qhyst, Qoffset, TEMP_OFFSET and PENALTY_TIME. However, TEMP_OFFSET and PENALTY_TIME are only applicable if the usage of HCS is indicated in system information.

- 1. TS 25.304, clause 5.2.1.
- 2. TS 25.304, clause 4.3.
- 3. TS 25.304, clause 5.2.5.1.
- 4. TS 25.304, clause 5.2.6.1.4.

6.2.2.1.3 Test purpose

1. To verify that the UE performs reselection from UTRAN to GSM on the following occasions:

37

- 1.1 Serving cell becomes barred.
- 1.2 S<0 for serving cell.

6.2.2.1.4 Method of test

Initial conditions

All cells belong to the same PLMN.

The Inter-RAT Cell Info List of Cell 1 (UTRAN) refers to Cell 2 (GSM) and Cell 3 (GSM).

The 3G Neighbour Cell Description of Cell 2 (GSM) and Cell 3 (GSM) refers to Cell 1 (UTRAN) Step a-c:

Parameter	Unit	Cell 1 (UTRAN)
Test Channel		1
CPICHEc (FDD)	dBm <u>/</u> <u>3.84</u> <u>MHz</u>	-60
P-CCPCHRSCP (TDD)	dBm	-60
Qrxlevmin	dBm	- <u>101<mark>100</mark></u>
Srxlev*	dBm	<u>41</u> 40
CellBarred		Not barred

Parameter	Unit	Cell 2 (GSM)	Cell 3 (GSM)
Test Channel		1	2
RF Signal Level	dBm	-80	-85
RXLEV_ACCESS _MIN	dBm	-100	-100
C1*	dBm	20	15
FDD_Qmin	dB	-20	-20
FDD_Qoffset	dBm	0	0

Step d-f:

Parameter	Unit	Cell 1 (UTRAN)
CellBarred		Not barred -> Barred
Tbarred	S	80

Step g:

Parameter	Unit	Cell 1 (UTRAN)
Qrxlevmin	dB	- <u>101<mark>400</mark>-> -<u>41</u>40</u>
Srxlev*	dB	40 -> - <u>19</u> 20

38

Test procedure

Method B is applied.

- a) The SS activates cells 1, 2, and 3. The SS monitors cells 1, 2 and 3 for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access request from the UE.
- d) The SS sets Cell 1 to be barred.
- e) The SS waits for random access request from the UE.
- f) The UE is switched off.
- g) Step a-e) is repeated except that in step d), Qrxlevmin is increased, so S will become negative instead of being barred.

6.2.2.1.5 Test Requirements

- 1) In step c), after the UE has responded on Cell 1, it shall not respond on any other cell within 1 min.
- 2) In step e), the UE shall respond on Cell 2.
- 3) In step g), the UE shall respond on Cell 2 after Qrxlevmin is increased.

6.2.2.2 Cell reselection if cell becomes barred or C1<0; GSM to UTRAN

6.2.2.2.1 Definition

Test to verify that if both a GSM and UTRAN network is available, the UE performs cell reselection from GSM to UTRAN if the GSM cell becomes barred or the path loss criterion C1 falls below zero for a period of 5 s.

6.2.2.2.2 Conformance requirement

- 1. At least every 5 s the MS shall calculate the value of C1 and C2 for the serving cell and re-calculate C1 and C2 values for non serving cells (if necessary). The MS shall then check whether:
 - 1.1 The path loss criterion (C1) for current serving cell falls below zero for a period of 5 s. This indicates that the path loss to the cell has become too high.
- 2. While camped on a cell of the registered PLMN ("camped normally"), the MS may need to select a different cell ("normal cell reselection" state). The following events trigger a cell reselection:
 - 2.1 The path loss criterion parameter C1 (see TS 03.22, clause 3.6) indicates that the path loss to the cell has become too high;
 - 2.2 The cell camped on (current serving cell) has become barred.

References

- 1. TS 05.08, clause 6.6.2.
- 2. TS <u>23.122</u>03.22, clause 4.5.

6.2.2.2.3 Test purpose

- 1. To verify that the UE performs reselection from GSM to UTRAN on the following occasions:
 - 1.1 Serving cell becomes barred.
 - 1.2 The path loss criterion C1 for serving cell falls below zero for a period of 5 s.

6.2.2.2.4 Method of test

Initial conditions

The USIM does not contain any preferred RAT.

The 3G Neighbour Cell Description of Cell 1 (GSM) refers Cell 2 (UTRAN) and Cell 3 (UTRAN).

The Inter-RAT Cell Info List of Cell 2 (UTRAN) and Cell 3 (UTRAN) refers to Cell 1 (GSM).

Step a-c:

Parameter	Unit	Cell 1 (GSM)
Test Channel		1
RF Signal Level	dBm	-50
RXLEV_ACCESS_ MIN	dBm	-70
MS_TXPWR_MAX_ CCH	dBm	Max. output power of UE
FDD_Qmin	dB	-20
FDD_Qoffset	dBm	0
CELL_BAR_ACCES S		Not barred
C1*	dBm	20

Parameter	Unit	Cell 2 (UTRAN)	Cell 3 (UTRAN)
P-CCPCH_RSCP (TDD)	dBm	-60	-70
CPICH_Ec (FDD)	dBm/3.8 4 MHz	-60	-70
Qrxlevmin	dBm	- <u>101</u> 100	- <u>101</u> 100
Srxlev*	dBm	<u>41</u> 40	<u>31</u> 30

Step d-e:

Parameter	meter Unit Cell 1 (GSM)	
CELL_BAR_ACCE SS		Not barred -> Barred

Step f-g:

Parameter	Unit	Cell 1 (GSM)
RF Signal Level	dBm	-50 -> -80 (4sec) -> -50
C1*	dBm	20 -> -10 (4sec) -> 20

Step h:

Parameter	Unit	Cell 1 (GSM)
RF Signal Level	dBm	-50 -> -80
C1*	dBm	20 -> -10

Test procedure

Method B is applied.

- a) The SS activates cells 1, 2, and 3. The SS monitors cells 1, 2 and 3 for random access requests from the UE.
- b) The UE is switched on.
- c) The SS waits for random access request from the UE.
- d) The SS sets Cell 1 to be barred.
- e) The SS waits for random access request from the UE.
- f) The UE is switched off.
- g) Step a-e) is repeated except that in step d), the SS reduces signal level on Cell 1 to -80 dBm for 4 s and then raises the level back to -50 dBm (C1 becomes -10 dBm during this period).
- h) The SS reduces signal level on Cell 1 to -80 dBm.

6.2.2.2.5 Test Requirements

- 1) In step c), after the UE has responded on Cell 1, it shall not respond on any other cell within 1 min.
- 2) In step e), the UE shall respond on Cell 2.
- 3) In step g), there shall be no access on Cell 2 within 30 s, after having reduced the signal level on Cell 1.
- 4) In step h), the UE shall respond on Cell 2.

6.2.2.3 Cell reselection timings; GSM to UTRAN

6.2.2.3.1 Definition

Test to verify that the UE meets the cell reselection timing requirements when both a GSM and UTRAN network is available.

6.2.2.3.2 Conformance requirement

- 1. If the 3G Cell Reselection list (see TS 04.18) includes UTRAN frequencies, the MS shall, at least every 5 s update the value RLA_C for the serving cell and each of the at least 6 strongest non-serving GSM cells.
 - 1.1 The MS shall then reselect a suitable UTRAN cell if its measured RSCP value exceeds the value of RLA_C for the serving cell and all of the suitable non-serving GSM cells by the value XXX_Qoffset for a period of 5 s and, for FDD, the UTRAN cells measured Ec/No value is equal or greater than the value FDD_Qmin.
 - Ec/No and RSCP are the measured quantities.
 - FDD_Qmin and XXX_Qoffset are broadcast on BCCH of the serving cell. XXX indicates other radio access technology/mode.
 - 1.2 In case of a cell reselection occurring within the previous 15 s, XXX_Qoffset is increased by 5 dB.
 - 1.3 Cell reselection to UTRAN shall not occur within 5 s after the MS has reselected a GSM cell from an UTRAN cell if a suitable GSM cell can be found.
 - 1.4 If more than one UTRAN cell fulfils the above criteria, the UE shall select the cell with the greatest RSCP value.

2. The MS shall be able to identify and select a new best UTRAN cell on a frequency, which is part of the 3G Cell Reselection list, within 30 s after it has been activated under the condition that there is only one UTRAN frequency in the list and under good radio conditions.

The allowed time is increased by 30 s for each additional UTRAN frequency in the 3G Cell Reselection list. However, multiple UTRAN cells on the same frequency in the neighbour cell list does not increase the allowed time.

NOTE: Definitions of measurements are in TS 25.215 and TS 25.101 for FDD mode, in TS 25.225 and TS 25.102 for TDD mode, clause 3.2 and TS 05.08, clause 6.1.

References

- 1. TS 05.08, clause 6.6.5.
- 2. TS 05.08, clause 6.6.4.

6.2.2.3.3 Test purpose

- 1. To verify that:
 - 1.1 The UE meets conformance requirement 1.1 and additionally, that no reselection is performed if the period is less than 5 s
 - 1.2 The UE meets conformance requirement 1.2.
 - 1.3 The UE meets conformance requirement 1.3.

6.2.2.3.4 Method of test

Initial conditions

The 3G Neighbour Cell Description of Cell 1 (GSM) refers Cell 2 (UTRAN) and Cell 3 (UTRAN).

The Inter-RAT Cell Info List of Cell 2 (UTRAN) and Cell 3 (UTRAN) refers to Cell 1 (GSM).

Step a-c:

Parameter	Unit	Cell 1 (GSM)	Cell 2 (GSM)
Test Channel		1	2
RF Signal Level	dBm	-70	-85
RXLEV_ACCESS_ MIN	dBm	-100	-100
MS_TXPWR_MAX_ CCH	dBm	Max. output power of UE	Max. output power of UE
FDD_Qmin	dBm	-20	-20
FDD_Qoffset	dBm	5	5

Parameter	Unit	Cell 3 (UTRAN)
Test Channel		1
CPICH_Ec_CPICH_RSCP (FDD)	dBm <u>/</u> <u>3.84</u> <u>MHz</u>	-74
P-CCPCH_RSCP (TDD)	dBm	- <mark>7463</mark>
Qrxlevmin	dBm	- <u>101100</u>
Srxlev*	dBm	<u>27</u> 26

Step d-g:

Parameter	Unit	Cell 1 (GSM)	Cell 2 (GSM)
RF Signal Level	dBm	-70 -> -82 (4 s) -> -70	OFF

42

Step h-j:

Parameter	Unit	Cell 1 (GSM)	Cell 2 (GSM)
RF Signal Level	dBm	-82 -> -70	OFF

Step k-m:

Parameter	Unit	Cell 1 (GSM)	Cell 2 (GSM)
		-82 ->	OFF
RF Signal Level	dBm	-70 -> -82	
		-02	

Test procedure

NOTE: Step a-c): Test purpose 1.3. Step d-g): test purpose 1.1. Step h-k): test purpose 1.2.

Method B is applied.

- a) The SS activates the channels. The UE is not paged on any of the cells.
- b) The UE is switched on.
- c) After 50 s, the SS starts paging continuously on cells 1 and 3 for 20 s. The SS monitors cells 1 and 3 for random access requests from the UE.
- d) Cell 2 is switched off. The SS stops paging on the cells and waits for 20 s. (The UE should revert to Cell 1 due to cell reselection).
- e) The SS starts paging continuously on Cell 3.
- f) The SS decreases the transmit level of Cell 1 to -82 dBm for a period of 4 s (RSCP will then exceed RLA_C value of Cell 1 by more than XXX_Qoffset) and then changes the level back to -70 dBm.
- g) The SS waits to see if there is any random access requests from the UE on Cell 3.
- h) The SS stops paging on all cells and sets the transmit level of Cell 1 to -82 dBm.
- i) The SS waits 20 s and then starts paging continously on Cell 1. (The UE should revert to Cell 3 due to cell reselection).
- j) The SS increases the transmit level of Cell 1 to -70 dBm and waits for the UE to access on Cell 1. The SS records the time t from the increase in the level of Cell 1 to the first response from the UE.
- k) The SS stops paging on all cells and sets the transmit level of Cell 1 back to -82 dBm.
- 1) The SS waits 20 s (The UE should revert to Cell 3 due to cell reselection).
- m) The SS increases the transmit level of Cell 1 to -70 dBm. After t+2 s (i.e. 2 s after reselection to Cell 1), the SS starts paging continuously on Cell 3, changes the level of Cell 1 back to -82 dBm and waits to see if there is any random access request on Cell 3. (Within 15 sec after reselection to GSM, the level of Cell 1 is -82 + 10 dBm=-72 dBm. After the 15 s period, the level of Cell 1 is -82 + 5 dBm=-77 dBm. The level of Cell 3 is -74 dBm, thus leading to reselection to Cell 3 after 15 s).

6.2.2.3.5 Test Requirements

- 1) In step c), after the UE has reselected Cell 1 from Cell 3 as indicated by random access requests, any random access requests on Cell 3 shall not occur within 4,5 s of the last random access request on Cell 1.
- 2) In step g), there shall be no access on Cell 3 within 34 s of decreasing the level of Cell 1.

- 3) In step j), the UE shall respond on Cell 1.
- 4) In step m), there shall be no response on Cell 3 within 11 s after the level of Cell 1 is changed back to -82 dBm.
- NOTE: The 11 s is derived from (t+15) s minimum cell reselection timer minus (t+2) s from the start of step m) up to the decrease of the level of Cell 1. A further 2 s are subtracted to cover for any uncertainty introduced by the random access process occurring after step g).

<End of modified section>

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

3GPP TSG- T1 SIG Meeting #26 Luton, UK, 4th – 8th November 2002

T1-020710

T1S-020716

	CHANGE REQUEST								R-Form-v6.1		
ж	-	84.123-1			жrev	-	ж	Current vers	^{ion:} 5.	1.1	ж
	S	pec Title:		Equipment (U : Protocol cor							ж
For <mark>H</mark>	ELP on u	ising this for	m, see	bottom of thi	s page or	look	at th	e pop-up text	over the	ж sym	bols.
Propose	d change	affects: अ	(U)	SIM ME	E/UE X	Rad	io Ac	cess Network	K Co	ore Net	work
Title:	ж	CR to 34.1	23-1 R	el-5: Correcti	on to Pac	kage	1 tes	<mark>st cases (revis</mark>	sion of T1	S-0206	677)
Source:	ж	Panasoni	С								
Work ite	m code:೫	TEI						Date: ೫	22/10/2	2002	
Category	<i>):</i> ¥	Use <u>one</u> of f F (con A (con B (add C (fun D (edit	rection) respond lition of ctional i torial m planatio	owing categorie ds to a correctio feature), modification of odification) ns of the above <u>TR 21.900</u> .	on in an ea feature)		eleas	R97 R98 R99		ase 2) 1996) 1997) 1998) 1999) 4)	ases:

Reason for change: ३	€ <mark>1.</mark>	In clause 8.3.1.1, physical channel parameter should be provided in the CELL UPDATE CONFIRM message when the SS requests the UE to move from CELL_FACH to CELL_DCH.
	Modifi	cations from T1S-020677
	1.	The condition in which the IE "Protocol Error Indicator" in RRC CONNECTION REQUEST message should be set to TRUE is not stated in the core spec. Therefore, this IE should not be checked.
	Modifi	cations from T1S-020630
	1.	In clause 8.3.1.1, CELL UPDATE CONFIRM message which includes the IE "Transport channel information elements" will cause the UE to transit to CELL_DCH state, and not CELL_FACH state.
	2.	In clause 8.4.1.5, the UE will not read the modified MIB and SIB 11 broadcasted in step 1 since it has entered CELL_DCH state.
	3.	Editorial changes.
Cummon of changes	e la alou	
Summary of change: \$	€ <u>In Clau</u> ●	In step 8 k=2, physical channel parameter is added to allow UE to

	establish the dedicated physical channel when UE moves into CELL_DCH state.
	Modifications from T1S-020677
	 In clause 8.1.2.9 The IE "Protocol Error Indicator" in RRC CONNECTION REQUEST message is set to "Not Checked"
	Modifications from T1S-020630
	 In clause 8.1.1.1 Editorial: In Test Requirement, description of events after step 2 and 3 are modified and added respectively.
	 In clause 8.3.1.1 In step 8 k=2, when the IE "Transport channel information elements" is included in the CELL UPDATE CONFIRM message, the UE must transit to CELL_DCH state. Step 11a and 11b are to bring the UE back to CELL_FACH state in order to continue testing in CELL_FACH. The Specific Message Content of CELL UPDATE CONFIRM (step 8, k=2) is edited. The IE "Added or Reconfigured uplink TrCH information" and "Added or Reconfigured downlink TrCH information" are not included in SIB 5. Editorial: Comments in Expected Sequence are edited to avoid ambiguity.
	 In clause 8.3.1.3 Editorial: 2 copies of the Specific Message Content of CELL UPDATE CONFIRM in step 12, 18 are found. The incorrect Message Content is deleted.
	 In clause 8.4.1.5 Modified MIB and SIB 11 should be transmitted before idle update preamble. Current test case broadcast these messages only in step 1 after the UE has entered CELL_DCH, when the UE will not read the broadcasted messages. SIB 11 is added into the Expected Sequence in step 9 because it is described in the Test Procedure. In the Specific Message Content of the MEASUREMENT REPORT in step 6, the IE "CPICH RSCP" is set to present since in the MEASUREMENT CONTROL in step 5, the IE "CPICH RSCP reporting indicator" is set to TRUE for active set cells.
Consequences if not approved:	The test prose cannot test UE correctly. Some test proses will be ambiguous.
Clauses affected: #	8 8.1.1.1, <mark>8.1.2.9,</mark> 8.3.1.1, 8.3.1.3, 8.4.1.5
Other specs \$ affected:	Ø Other core specifications # Test specifications Ø Ø Ø Ø Ø Ø Ø Ø Ø Ø
Other comments: \$	Affects R99, REL-4, REL-5

How to create CRs using this form: Comprehensive information and tips about how to create CRs can be found at: <u>http://www.3gpp.org/3G_Specs/CRs.htm</u>. Below is a brief summary:

Release 5

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

START OF MODIFICATIONS

8.1.1.1 Paging for Connection in idle mode

8.1.1.1.1 Definition

8.1.1.1.2 Conformance requirement

A UE in idle mode, CELL_PCH state or URA_PCH state shall receive the paging information for all its monitored paging occasions. For an UE in idle mode, the paging occasions are specified in [25.304] and depend on the IE "CN domain specific DRX cycle length coefficient", as specified in subclause 8.6.3.1a. For a UE in CELL_PCH state or URA_PCH state, the paging occasions depend also on the IE "UTRAN DRX cycle length coefficient" and the IE "RRC State Indicator", as specified in subclauses 8.6.3.2 and 8.6.3.3 respectively.

When the UE receives a PAGING TYPE 1 message, it shall perform the actions as specified below.

If the UE is in idle mode, for each occurrence of the IE "Paging record" included in the message the UE shall:

- 1> if the IE "Used paging identity" is a CN identity:
 - 2> compare the IE "UE identity" with all of its allocated CN UE identities:

2> if one match is found:

- 3> indicate reception of paging; and
- 3> forward the IE "CN domain identity", the IE "UE identity" and the IE "Paging cause" to the upper layers.
- 1> otherwise:

:

2> ignore that paging record.

In the UE, the initial direct transfer procedure shall be initiated, when the upper layers request establishment of a signalling connection. This request also includes a request for the transfer of a NAS message.

Upon initiation of the initial direct transfer procedure when the UE is in idle mode, the UE shall:

- 1> set the variable ESTABLISHMENT_CAUSE to the cause for establishment indicated by upper layers;
- 1> perform an RRC connection establishment procedure, according to subclause 8.1.3;
- 1> if the RRC connection establishment procedure was not successful:

2> indicate failure to establish the signalling connection to upper layers and end the procedure.

- 1> when the RRC connection establishment procedure is completed successfully:
 - 2> continue with the initial direct transfer procedure as below.

Upon initiation of the initial direct transfer procedure when the UE is in CELL_PCH or URA_PCH state, the UE shall:

1> perform a cell update procedure, according to subclause 8.3.1, using the cause "uplink data transmission";

1> when the cell update procedure completed successfully:

2> continue with the initial direct transfer procedure as below.

- The UE shall, in the INITIAL DIRECT TRANSFER message:
 - 1> set the IE "NAS message" as received from upper layers; and
 - 1> set the IE "CN domain identity" as indicated by the upper layers; and

1> set the IE "Intra Domain NAS Node Selector" as follows:

- 2> derive the IE "Intra Domain NAS Node Selector" from TMSI/PMTSI, IMSI, or IMEI; and
- 2> provide the coding of the IE "Intra Domain NAS Node Selector" according to the following priorities:

1. derive the routing parameter for IDNNS from TMSI (CS domain) or PTMSI (PS domain) whenever a valid TMSI/PTMSI is available;

- 2. base the routing parameter for IDNNS on IMSI when no valid TMSI/PTMSI is available;
- 3. base the routing parameter for IDNNS on IMEI only if no (U)SIM is inserted in the UE.
- 1> calculate the START according to subclause 8.5.9 for the CN domain as set in the IE "CN Domain Identity"; and
- 1> include the calculated START value for that CN domain in the IE "START".

In CELL_FACH state, the UE shall:

- 1> include a measurement report in the IE "Measured results on RACH", as specified in the IE "Intra-frequency reporting quantity for RACH reporting" and the IE "Maximum number of reported cells on RACH" in System Information Block type 12 (or "System Information Block Type 11" if "System Information Block Type 12" is not being broadcast);
- 1> include in the IE "Measured results on RACH" all requested reporting quantities for cells for which measurements are reported.

The UE shall:

- 1> transmit the INITIAL DIRECT TRANSFER message on the uplink DCCH using AM RLC on signalling radio bearer RB3;
- 1> when the INITIAL DIRECT TRANSFER message has been submitted to lower layers for transmission:
 - 2> confirm the establishment of a signalling connection to upper layers; and
 - 2> add the signalling connection with the identity indicated by the IE "CN domain identity" in the variable ESTABLISHED_SIGNALLING_CONNECTIONS.
- 1> when the successful delivery of the INITIAL DIRECT TRANSFER message has been confirmed by RLC:

2> the procedure ends.

Reference

3GPP TS 25.331 clause 8.1.2 and 8.1.8, 3GPP TS 25.211 clause 5.3.3.10 (FDD), 3GPP TS 25.221 (TDD), 3GPP TS 25.304 clause 8.

8.1.1.1.3 Test purpose

To confirm that the UE establishes an RRC connection after it receives a PAGING TYPE 1 message which includes IE "UE identity" (in IE "Paging Record") set to the IMSI of the UE, and responds with a correct INITIAL DIRECT TRANSFER message.

8.1.1.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108 with a CN UE identity (set to IMSI), depending on the CN domain(s) supported by the UE.

Test Procedure

SS transmits SYSTEM INFORMATION BLOCK TYPE 1 or 13 messages, depending on the CN type supported by the UE. The SS transmits a PAGING TYPE 1 message, which includes an unmatched CN UE identity for the UE in the idle state. The UE shall not change its state. The SS transmits a PAGING TYPE 1 message, which includes a matched CN UE identity for the UE in the idle state. During transmission of PAGING TYPE 1 messages, SS selects the correct paging indicator on the PICH in order to allow the UE to respond to paging. Then the UE transmits an RRC CONNECTION REQUEST to the SS, the SS transmits an RRC CONNECTION SETUP to the UE. When the UE receives this message, the UE establishes an RRC connection and transmits an RRC CONNECTION SETUP COMPLETE message and an INITIAL DIRECT TRANSFER message on the uplink DCCH.

NOTE: For UEs supporting GSM-MAP CN type only, SYSTEM INFORMATION TYPE 1 messages are to be sent by SS in this test case. On the other hand, SS transmits SYSTEM INFORMATION TYPE 13 messages if the UE under test supports only ANSI-41 CN type.

Expected sequence

Step	Direct	ion	Message	Comment
_	UE	SS		
1 ←			SYSTEM INFORMATION BLOCK TYPE 13 or SYSTEM INFORMATION BLOCK TYPE 1	Transmit these messages on the BCCH, in addition to the normal BCCH transmissions. See specific message contents.
2	2 ←		PAGING TYPE 1	The SS transmits the message, which includes an unmatched identity (incorrect IMSI), and the UE does not change its state.
3	3 ←		PAGING TYPE 1	The SS transmits the message, which includes a matched identity (test-SIM IMSI).
4	\rightarrow		RRC CONNECTION REQUEST	
5	÷		RRC CONNECTION SETUP	SS assigns DPCH resources to allow UE to establish an RRC connection.
6	\rightarrow		RRC CONNECTION SETUP COMPLETE	
7	\rightarrow		INITIAL DIRECT TRANSFER	

Specific Message Contents

SYSTEM INFORMATION BLOCK TYPE 1 (Step 1) – for UEs supporting GSM-MAP core networks

Information Element	Value/remark		
CN common GSM-MAP NAS system information	LAC		
CN domain system information list	Only 1 entry		
 CN domain system information 			
- CN domain identity	Supported Domain (PS Domain or CS Domain)		
- CHOICE CN Type	GSM-MAP		
 CN domain specific NAS system information 	1E 01(CS) or 00 00(PS)		
- CN domain specific DRX cycle length coefficient	7		
UE Timers and constants in idle mode			
- T300	4000 milliseconds		
- N300	7		
- T312	10 seconds		
- N312	1		

SYSTEM INFORMATION TYPE 13 (Step 1) - for UEs supporting ANSI-41 core networks

Information Element	Value/remark
CN domain system information list	Only 1 entry
CN domain system information	
- CN domain identity	Supported Domain (PS Domain or CS Domain)
- CHOICE CN Type	ANSI-41
- CN domain specific NAS system information	Default
- CN domain specific DRX cycle length coefficient	7
UE Timers and constants in idle mode	
- T300	4000 milliseconds
- N300	7
- T312	10 sec
- N312	1

PAGING TYPE 1 (Step 2)

Information Element	Value/remark		
Message Type			
Paging record list	Only 1 entry		
Paging record			
CHOICE Used paging identity	CN identity		
- Paging cause	Terminating Call with one of the supported services		
- CN domain identity	Registered Domain (PS Domain or CS Domain)		
- CHOICE UE Identity	IMŠI		
- IMSI	Set to an arbitrary octet string of length 7 bytes which is		
	different from the IMSI value stored in the TEST USIM		
	card.		
BCCH modification info	Not Present		

PAGING TYPE 1 (Step 3)

Information Element	Value/remark
Message Type	
Paging record list	Only 1 entry
Paging record	
CHOICE Used paging identity	CN identity
- Paging cause Terminating Call with one of the supported service	
- CN domain identity Registered Domain (PS Domain or CS Domain	
- CHOICE UE Identity	IMŠI
- IMSI	Set to the same octet string as in the IMSI stored in the
	TEST USIM card
BCCH modification info	Not Present

RRC CONNECTION REQUEST (Step 4)

Information Element	Value/remark
Message type	
Initial UE identity	Same as the IMSI stored in the TEST USIM card, or the registered TMSI or P-TMSI
Establishment Cause	Check to see if it is set to the same value as "Paging Cause" IE in the PAGING TYPE 1 message transmitted on step 3.
Protocol Error Indicator	Check to see if it is set to FALSE
Measured results on RACH	Not checked.

Release 5

INITIAL DIRECT TRANSFER (Step 7) – for UEs supporting GSM-MAP core networks

Information Element	Value/remark		
Message Type			
Integrity check info	Not present		
CN domain identity	CS domain or PS domain		
Intra Domain NAS Node Selector			
- CHOICE version	R99		
CHOICE CN type	GSM-MAP		
CHOICE Routing basis	IMSI (response to IMSI paging)		
Routing parameter	Bit string (10) consisting of DecimalToBinary [(IMSI div		
	10) mod 1000]. The bits of the result are numbered from		
	b0 to b9, with bit b0 being the least significant		
Entered parameter	False		
NAS message	Not checked		
START	Not checked		
Measured results on RACH	Not checked		

INITIAL DIRECT TRANSFER (Step 7) - for UEs supporting ANSI-41 core networks

Information Element	Value/remark		
Message Type			
Integrity check info	Not present		
CN domain identity	CS domain or PS domain		
Intra Domain NAS Node Selector			
- CHOICE version	ANSI-41 : Bitstring(14), all bits set to 0		
NAS message	Not checked		
START	Not checked		
Measured results on RACH	Not checked		

8.1.1.1.5 Test requirement

After step 2 the UE shall not <u>respond to PAGING TYPE 1 message sent in step 2</u>transmit on the uplink CCCH in order to establish a RRC connection.

After step 3 the UE shall transmit RRC CONNECTION REQUEST message on the uplink CCCH.

After step 5 the UE shall have an RRC connection based on dedicated physical channel resources and transmit an RRC CONNECTION SETUP COMPLETE message and INITIAL DIRECT TRANSFER message on the uplink DCCH.

END OF MODIFICATIONS

<START OF MODIFICATIONS>

- 8.1.2.9 RRC Connection Establishment: Success after Physical channel failure and Failure after Invalid configuration
- 8.1.2.9.1 Definition
- 8.1.2.9.2 Conformance requirement

If the UE failed to establish, per TS 25.331 subclause 8.5.4, the physical channel(s) indicated in the RRC CONNECTION SETUP message.

After having received an RRC CONNECTION SETUP message with the value of the IE "Initial UE identity" equal to the value of the variable INITIAL_UE_IDENTITY.

Before the RRC CONNECTION SETUP COMPLETE message is delivered to lower layers for transmission,

Release 5

the UE shall:

- 1> clear the entry for the RRC CONNECTION SETUP message in the table "Accepted transactions" in the variable TRANSACTIONS;
- 1> check the value of V300, and:
 - 2> if V300 is equal to or smaller than N300:
 - 3> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;
 - 3> set the IEs in the RRC CONNECTION REQUEST message according to TS 25.331 subclause 8.1.3.3;
 - 3> perform the mapping of the Access Class to an Access Service Class as specified in TS 25.331 subclause 8.5.13, and apply the given Access Service Class when accessing the RACH;
 - 3> submit a new RRC CONNECTION REQUEST message to the lower layers for transmission on the uplink CCCH;
 - 3> increment counter V300; and
 - 3> restart timer T300 when the MAC layer indicates success or failure in transmitting the message.
 - 2> if V300 is greater than N300:

•••

If the UE receives an RRC CONNECTION SETUP message which contains an IE "Initial UE identity" with a value which is identical to the value of the variable INITIAL_UE_IDENTITY; and

the variable INVALID_CONFIGURATION becomes set to TRUE due to the received RRC CONNECTION SETUP message:

the UE shall:

- 1> clear the entry for the RRC CONNECTION SETUP message in the table "Accepted transactions" in the variable TRANSACTIONS and proceed as below;
- 1> if V300 is equal to or smaller than N300:
 - 2> set the variable PROTOCOL_ERROR_INDICATOR to TRUE;
 - 2> set the IEs in the RRC CONNECTION REQUEST message according to TS 25.331 subclause 8.1.3.3;
 - 2> perform the mapping of the Access Class to an Access Service Class as specified in TS 25.331 subclause 8.5.13; and
 - 2> apply the given Access Service Class when accessing the RACH;
 - 2> submit a new RRC CONNECTION REQUEST message to the lower layers for transmission on the uplink CCCH;
 - 2> increment counter V300; and
 - 2> restart timer T300 when the MAC layer indicates success or failure in transmitting the message.
- 1> if V300 is greater than N300:

•••

Reference

3GPP TS 25.331 clause 8.1.3.

8.1.2.9.3 Test purpose

- 1. To confirm that the UE retries to establish the RRC connection until V300 is greater than N300 when a physical channel failure occurs because SS does not configure the physical channel that is specified in the transmitted RRC CONNECTION SETUP message.
- 2. To confirm that the UE retries to establish the RRC connection until V300 is greater than N300 when the transmitted RRC CONNECTION SETUP message causes invalid configuration in the UE.

8.1.2.9.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE

Test Procedure

Before the test starts, an internal counter K in SS is initialised to a value = 0. Following this, the UE shall transmit an RRC CONNECTION REQUEST message to the SS on the uplink CCCH, after the operator attempts to make an outgoing call. SS increments K every time such a message is received. Then, SS shall send a RRC CONNECTION SETUP message that contains an invalid configuration. UE shall then send RRC CONNECTION REQUEST message to SS again. This cycle is repeated until K reaches N300+1. When K is equal to N300+1, the SS again transmits an RRC CONNECTION SETUP message including an invalid configuration. Upon receiving this message the UE shall not send another RRC CONNECTION REQUEST message.

Next the SS re-initialises the internal counter K to value = 0, after which the operator attempts to make another outgoing call. Following this, the UE shall transmit an RRC CONNECTION REQUEST message to the SS on the uplink CCCH. SS increments K every time such a message is received. SS transmits an RRC CONNECTION SETUP message to make the UE configure the physical channel in order to communicate on the DCCH but SS does not configure the physical channel. Then the UE detects the physical channel failure and transmits an RRC CONNECTION REQUEST message. This cycle is repeated until K reaches N300+1. When K is equal to N300+1, the SS transmits the RRC CONNECTION SETUP message and configures the physical channel. The UE shall detect "in-sync" from physical layer and then acknowledge the establishment of RRC connection by sending the RRC CONNECTION SETUP COMPLETE message on uplink DCCH.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			SS initialises counter K to 0.
			Operator is asked to make
			an outgoing call and SS
			starts to wait for RRC
			CONNECTION REQUEST
			on uplink CCCH.
2	\rightarrow	RRC CONNECTION REQUEST	See specific message
			contents.
2a			SS increments K by 1 for
			every RRC CONNECTION
			REQUEST message
			received in step 2
2b	÷	RRC CONNECTION SETUP	See specific message
			contents.
3			SS checks to see if K is
			equal to N300+1. If so, goes
			to step 3a. Else, continues
			to execute step 2.
3a			SS waits to verify that the
			UE does not send any
			further RRC CONNECTION
			REQUEST message
3b			SS re-initialises counter K to
0.0			0. Operator is asked to
			make another outgoing call
			and SS starts to wait for
			RRC CONNECTION
			REQUEST on uplink CCCH.
3c	\rightarrow	RRC CONNECTION REQUEST	See specific message
00	,		contents.
3d			SS increments K by 1 for
ou			every RRC CONNECTION
			REQUEST message
			received in step 3c
3e			SS checks to see if K is
26			equal to N300+1. If so, goes
			to step 6. Else, continues to
			execute step 4
1	\	RRC CONNECTION SETUP	Use the default message
4	~		Ũ
			with the same message sub-
			type specified in clause 9 in
			TS 34.108. SS does not
			configure the physical
_			channel.
5	,		The next step is step 3c.
6	÷	RRC CONNECTION SETUP	Use the default message
			with the same message sub-
			type specified in clause 9 in
			TS 34.108. SS configures
			the physical channel.
7			The UE configures the layer
			1 and layer 2.
8	\rightarrow	RRC CONNECTION SETUP COMPLETE	Use the default message
			with the same message sub-
			type specified in clause 9 in

Specific Message Contents

RRC CONNECTION REQUEST (Step 2 & step 3c, K=0)

The same message sub-type found in clause 9 of TS 34.108 applies, with the following exceptions:

Information Element	Value/remark	
Initial UE identity	Same as the IMSI stored in the TEST USIM card, or the registered TMSI or P-TMSI	
Establishment Cause	Originating Interactive Call or Originating Background Call or Originating Streaming Call	

RRC CONNECTION REQUEST (Step 2 & step 3c, K>0)

The same message sub-type found in clause 9 of TS 34.108 applies, with the following exceptions:

Information Element	Value/remark		
Initial UE identity	Same as the IMSI stored in the TEST USIM card, or the		
	registered TMSI or P-TMSI		
Establishment Cause	Originating Interactive Call or Originating Background		
	Call or Originating Streaming Call		
Protocol error indicator	TRUENot Checked		

RRC CONNECTION SETUP (Step 2b)

Use the same message sub-type found in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
RRC State Indicator	CELL_DCH
Uplink DPCH info	Not present

8.1.2.9.5 Test requirement

After step 3a the UE shall not send any further RRC CONNECTION REQUEST message.

After step 8 the UE shall transmit an RRC CONNECTION SETUP COMPLETE message and establish an RRC connection.

<END OF MODIFICATIONS>

START OF MODIFICATIONS

- 8.3.1.1 Cell Update: cell reselection in CELL_FACH
- 8.3.1.1.1 Definition
- 8.3.1.1.2 Conformance requirement

A UE shall initiate the cell update procedure in the following cases:

1> Uplink data transmission:

• • •

- 1> Paging response:
 - •••

1> Radio link failure:

...

1> Re-entering service area:

...

1> RLC unrecoverable error:

•••

- 1> Cell reselection:
 - 2> if none of the criteria for performing cell update with the causes specified above in the current subclause is met:
 - 3> if the UE is in CELL_FACH or CELL_PCH state and the UE performs cell re-selection; or
 - 3> if the UE is in CELL_FACH state and the variable C_RNTI is empty:
 - 4> perform cell update using the cause "cell reselection".

...

When initiating cell update procedure, the UE shall:

1> stop timer T305;

1> if the UE is in CELL_DCH state:

...

1> move to CELL_FACH state, if not already in that state;

1> if the UE performs cell re-selection:

2> clear the variable C_RNTI; and

- 2> stop using that C_RNTI just cleared from the variable C_RNTI in MAC.
- 1> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;
- 1> in case of a cell update procedure:
 - 2> set the contents of the CELL UPDATE message according to TS 25.331 subclause 8.3.1.3;

2> submit the CELL UPDATE message for transmission on the uplink CCCH.

•••

1> set counter V302 to 1;

1> start timer T302 when the MAC layer indicates success or failure in transmitting the message.

. . .

In case of cell update procedure the UE shall transmit a CELL UPDATE message.

The UE shall set the IEs in the CELL UPDATE message as follows:

- 1> set the IE "Cell update cause" corresponding to the cause specified in TS 25.331 subclause 8.3.1.2 that is valid when the CELL UPDATE message is submitted to lower layers for transmission;
- NOTE: During the time period starting from when a cell update procedure is initiated by the UE until when the procedure ends, additional CELL UPDATE messages may be transmitted by the UE with different causes.

- 1> set the IE "U-RNTI" to the value of the variable U_RNTI;
- 1> if the value of the variable PROTOCOL_ERROR_INDICATOR is TRUE:

...

1> if the value of the variable FAILURE_INDICATOR is TRUE:

•••

...

When the UE receives a CELL UPDATE CONFIRM message; and

- if the message is received on the CCCH, and IE "U-RNTI" is present and has the same value as the variable U_RNTI; or
- if the message is received on DCCH:

the UE shall:

- 1> stop timer T302;
- 1> in case of a cell update procedure and the CELL UPDATE CONFIRM message:
 - 2> includes "RB information elements"; and/or
 - 2> includes "Transport channel information elements"; and/or
 - 2> includes "Physical channel information elements"; and
 - 2> if the variable ORDERED_RECONFIGURATION is set to FALSE:
 - 3> set the variable ORDERED_RECONFIGURATION to TRUE.
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following:
 - • •
 - 2> perform the physical layer synchronisation procedure as specified in TS 25.214;
 - ...
- •••
- 1> enter a state according to subclause 8.6.3.3 applied on the CELL UPDATE CONFIRM / URA UPDATE CONFIRM message.

• • •

If the UE after state transition remains in CELL_FACH state, it shall

- 1> start the timer T305 using its initial value if timer T305 is not running and periodical cell update has been configured by T305 in the IE "UE Timers and constants in connected mode" set to any other value than "infinity";
- 1> select PRACH according to TS 25.331 subclause 8.5.17;
- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> not prohibit periodical status transmission in RLC;
- 1> if the IE "UTRAN DRX cycle length coefficient" is included in the same message:
 - 2> ignore that IE and stop using DRX.

If the UE after the state transition remains in CELL_FACH state; and

Release 5

- a C-RNTI is stored in the variable C_RNTI;

•••

the UE shall:

•••

- 1> in case of a cell update procedure:
 - 2> set the IE "RRC transaction identifier" in any response message transmitted below to the value of "RRC transaction identifier" in the entry for the CELL UPDATE CONFIRM message in the table "Accepted transactions" in the variable TRANSACTIONS; and
 - 2> clear that entry.

•••

1> transmit a response message as specified in TS 25.331 subclause 8.3.1.7;

•••

If the CELL UPDATE CONFIRM message:

- includes the IE "RB information to release list":

the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include the IE "RB information to release list"; and
- includes the IE "RB information to reconfigure list"; or
- includes the IE "RB information to be affected list":

the UE shall:

1> transmit a RADIO BEARER RECONFIGURATION COMPLETE as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- includes "Transport channel information elements":

the UE shall:

1> transmit a TRANSPORT CHANNEL RECONFIGURATION COMPLETE as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- includes "Physical channel information elements":

the UE shall:

```
1> transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE as response message using AM RLC.
```

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and

Release 5

- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- includes "CN information elements"; or
- includes the IE "Ciphering mode info"; or
- includes the IE "Integrity protection mode info"; or
- includes the IE "New C-RNTI"; or
- includes the IE "New U-RNTI":

the UE shall:

1> transmit a UTRAN MOBILITY INFORMATION CONFIRM as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- does not include "CN information elements"; and
- does not include the IE "Ciphering mode info"; and
- does not include the IE "Integrity protection mode info"; and
- does not include the IE "New C-RNTI"; and
- does not include the IE "New U-RNTI":

the UE shall:

1> transmit no response message.

If the new state is CELL_FACH, the response message shall be transmitted using the new configuration after the state transition., and the UE shall:

•••

1> if the variable PDCP_SN_INFO is empty:

•••

- 2> if the CELL UPDATE CONFIRM or URA UPDATE CONFIRM message did not contain the IE "Ciphering mode info":
 - 3> when RLC has been requested to transmit the response message,

4> continue with the remainder of the procedure.

•••

If any or several of the following conditions are true:

...;

- reselection to another UTRA cell (including the previously serving cell) before completion of the cell update or URA update procedure;

the UE shall:

^{1&}gt; stop T302 if it is running;

```
•••
```

1> check whether it is still in "in service area";

```
•••
```

- 1> in case of a cell update procedure:
 - 2> clear any entry for the CELL UPDATE CONFIRM message in the table "Accepted transactions" in the variable TRANSACTIONS.

If the UE detects "in service area" if it has not entered idle mode, and:

- 1> if V302 is equal to or smaller than N302, the UE shall:
 - 2> if the UE performed cell re-selection:
 - 3> delete its C-RNTI.
 - 2> in case of a cell update procedure:
 - 3> set the contents of the CELL UPDATE message according to TS 25.331 subclause 8.3.1.3;
 - 3> submit the CELL UPDATE message for transmission on the uplink CCCH.
 - 2> increment counter V302;
 - 2> restart timer T302 when the MAC layer indicates success or failure to transmit the message.
- 1> if V302 is greater than N302, the UE shall:

• • •

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.1.3 Test purpose

- 1. To confirm that the UE executes a cell update procedure after the successful reselection of another UTRA cell.
- 2. To confirm that the UE sends the correct uplink response message when executing cell update procedure due to cell reselection.
- 8.3.1.1.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 and 2 are active.

UE: PS-DCCH+DTCH_FACH (state 6-11) in cell 1 as specified in clause 7.4 of TS 34.108.

Test Procedure

Parameter	Unit	Cell 1		Cell 2	
		T0	T1	T0	T1
UTRA RF		Ch. 1		Ch. 1	
Channel					
Number					
CPICH Ec	dBm/3.84MHz	-60	-75	-75	-60
(FDD)					
P-CCPCH	dBm	-60	-75	-75	-60
RSCP (TDD)					

Table 8.3.1.1

Table 8.3.1.1 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. Columns marked "T0" denote the initial conditions. SS switches the power settings repeatedly between columns "T1" and "T0", whenever the description below specifies that the transmission power settings for cell 1 and cell 2 be reversed.

The UE is in the CELL_FACH state, camping onto cell 1. SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall find cell 2 to be more suitable for service and hence perform a cell reselection. After the completion of cell reselection, the UE shall transmits a CELL UPDATE message to the SS on the uplink CCCH of cell 2 and set IE "Cell update cause" to "Cell Reselection". After the SS receives this message, it transmits a CELL UPDATE CONFIRM message, which includes the IE "RRC State Indicator" set to "CELL_FACH", to the UE on the downlink DCCH. UE shall verify that IE "New C-RNTI" is not included in the downlink message and shall send a CELL UPDATE message to SS again. SS shall then send a CELL UPDATE CONFIRM message which includes a valid IE "New C-RNTI". SS verifies that the UE send UTRAN MOBILITY INFORMATION CONFIRM message.. UE shall stay in CELL_FACH state. SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with CELL UPDATE CONFIRM message and allocates new C-RNTI and U-RNTI identities to the UE. The IE "RRC State Indicator" is set to "CELL_FACH" in this message. The UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message. Following this, SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall initiate a cell update procedure by transmitting a CELL UPDATE message and stating the cause as 'cell re-selection'. SS replies with a CELL UPDATE CONFIRM message which contains IE "Physical channel information elements". The UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to acknowledge the change in physical resources. Then, SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains IE "Transport channel information elements" and IE "RRC State Indicator" is set to "CELL DCH". The UE shall move to CELL DCH state and send TRANSPORT CHANNEL RECONFIGURATION COMPLETE message. The SS shall transmit PHYSICAL CHANNEL RECONFIGURATION message, asking the UE to move to CELL_FACH state. The UE shall send a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message after transiting to CELL_FACH state. Following this, SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 2. SS replies with a CELL UPDATE CONFIRM message which contains IE "RB information to be affected list". The UE shall send RADIO BEARER RECONFIGURATION COMPLETE message. Then, SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains IE "RB information to release list". The UE shall send RADIO BEARER RELEASE COMPLETE message. Finally, SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 2. SS shall not respond to this message but SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS shall then send CELL UPDATE CONFIRM message to UE. UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message. SS calls for generic procedure C.2 to check that UE is in CELL FACH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the CELL_FACH state in cell 1
2			Void	SS applies the downlink transmission power settings, according to the values in columns "T1" of table 8.3.1.1. The UE shall find that the cell 2 is better for service and perform a reselection. SS waits for the maximum duration required for the UE to camp to cell 2.
3	\rightarrow	•	CELL UPDATE	Value "cell reselection" shall be indicated in IE "Cell update cause"
4	÷	-	CELL UPDATE CONFIRM	IE "RRC State Indicator" is set to "CELL_FACH". SS set k=0.
4a	\rightarrow	•	CELL UPDATE	Value "cell reselection" shall be indicated in IE "Cell update cause"
4b	÷	-	CELL UPDATE CONFIRM	See message content. SS set k=0.
5	\rightarrow	•	UTRAN MOBILITY INFORMATION CONFIRM	

			SS reverses the transmission power level of cell 1 and cell 2
7	\rightarrow	CELL UPDATE	
8	÷	CELL UPDATE CONFIRM	IE "RRC State Indicator" is set to "CELL_FACH". If k ≥ 0, new C-RNTI and U-RNTI identities are assigned to the UE. If k>0 IE "Physical channel information elements" is included in this message. If k>1, IE "Transport channel information elements" is included in this message and IE "RRC State Indicator" is set to "CELL DCH". If k>2, IE "RE information to be affected list" is included in this message. If k>3, IE "RB information to release list" is included in this message. Increment k by 1.
9	<i>→</i>	UTRAN MOBILITY INFORMATION CONFIRM	If k=1 when SS received this message, go to step 6. Elself k=1 and this message is not received, test fails. If this message is not received, proceed to next step.
10	<i>→</i>	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	If k=2 when SS received this message, go to step 6. Elself k=2 and this message is not received, test fails. If this message is not received, proceed to next step.
11	<i>→</i>	TRANSPORT CHANNEL RECONFIGURATION COMPLETE	If k=3 when SS received this message, proceed to next step. If k=3 and this message is not received, test fails. go to step 6. Else test fails. If this message is not received, proceed to next step.
<u>11a</u>	£	PHYSICAL CHANNEL RECONFIGURATION	The UE is in CELL_DCH now. The SS shall send PHYSICAL CHANNEL RECONFIGURATION message to the UE asking the UE to transit to CELL_FACH state.
<u>11b</u>	≥	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	
<u>11c</u>			The SS reverses the transmission power level of cell 1 and cell 2.
<u>11d</u>	<u>→</u>	CELL UPDATE	
<u>11e</u>	Ŧ	CELL UPDATE CONFIRM	IE "RRC State Indicator" is set to "CELL_FACH". If $k \ge 0$, new C-RNTI and U-RNTI identities are assigned to the UE. If $k>0$ IE "Physical channel information elements" is included in this message. If $k>2$, IE "RB information to be affected list" is included in this message. If $k>3$, IE "RB information to release list" is included in this message. If $k>3$, IE "RB information to release list" is included in this message. Increment k by 1.
12	<i>></i>	RADIO BEARER RECONFIGURATION COMPLETE	If k=4 when SS received this message, go to step <u>11c6</u> . Elself k=4 and this message is not received, test fails. If this
			message is not received, proceed to next step.

1	I	1	1
13	<i>→</i>	RADIO BEARER RELEASE COMPLETE	If k=5 when SS received this message, proceed to next step. Elself k=5 and this message is not received, test fails. If this message is not received, test fails.
14			SS reverses the transmission power level of cell 1 and cell 2.
15	\rightarrow	CELL UPDATE	
16			SS reverses the transmission power level of cell 1 and cell 2.
17	\rightarrow	CELL UPDATE	
18	←	CELL UPDATE CONFIRM	
19	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
20	\leftrightarrow	CALL C.2	If the test result of C.2 indicates that UE is in CELL_FACH state, the test passes, otherwise it fails.

Specific Message Contents

CELL UPDATE (Step 3, 7, <u>11d,</u> 15 and 17)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	In step 3, check to see if set to '0000 0000 0000 0000
	0001'. In step 7 and when k<1, check to see if set to
	'0000 0000 0000 0000 0001'. In step 7 and when k>0,
	check to see if set to same string in IE "S-RNTI" in IE
	"New U-RNTI" of CELL UPDATE CONFIRM message in
	previous step 8. In step 15 and 17, check to see if set to
	same string in IE "S-RNTI" in IE "New U-RNTI" of CELL
	UPDATE CONFIRM message in previous step 8.
Cell Update Cause	Check to see if set to 'Cell Re-selection'

CELL UPDATE CONFIRM (Step 4)

Use the same message sub-type found in TS 34.108, clause 9.

CELL UPDATE CONFIRM (Step 4b and 18)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
RRC State Indicator	CELL_FACH
New C-RNTI	'1010 1010 1010 1010'

CELL UPDATE CONFIRM (Step 8 and k = 0)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
- SRNC Identity	'0000 0000 0001'
- S-RNTI	An arbitrary 20-bits string which is different from original
	S-RNTI
New C-RNTI	An arbitrary 16-bits string which is different from original
	C-RNTI assigned in RRC connection establishment
	procedure.

CELL UPDATE CONFIRM (Step 8 and k=1)

Use the same message sub-type found in step 8 and k=0, with the following exceptions:

Information Element	Value/remark
Maximum allowed uplink TX power	3 dB below the following value: Minimum of {33 dBm for FDD and 30 dBm for TDD, maximum uplink power allowed under the UE power class}

CELL UPDATE CONFIRM (Step 8 and k=2)

Use the same message sub-type found in step 8 and k=1, with the following exceptions:

Information Element	Value/remark
RRC State indicator	CELL_DCH
UL Transport channel information for all transport	Same as the IE in RADIO BEARER SETUP (Packet to
<u>channels</u>	CELL_DCH from CELL_FACH in PS)
Added or Reconfigured uplink TrCH information	Same as the IE in RADIO BEARER SETUP (Packet to
	CELL_DCH from CELL_FACH in PS)the system
	information block type 5
DL Transport channel information for all transport	Same as the IE in RADIO BEARER SETUP (Packet to
<u>channels</u>	CELL DCH from CELL FACH in PS)
Added or Reconfigured downlink TrCH information	Same as RADIO BEARER SETUP (Packet to
	CELL_DCH from CELL_FACH in PS)the system
	information block type 5
CHOICE channel requirement	
Uplink DPCH info	Same as the IE in RADIO BEARER SETUP (Packet to
	CELL DCH from CELL FACH in PS)
Downlink information common for all radio links	Same as the IE in RADIO BEARER SETUP (Packet to
	CELL_DCH from CELL_FACH in PS)
Downlink information per radio link list	Same as the IE in RADIO BEARER SETUP (Packet to
	CELL DCH from CELL FACH in PS)

CELL UPDATE CONFIRM (Step 11e8-and k=3)

Use the same message sub-type found in step 8 and k=21, with the following exceptions:

Information Element	Value/remark
RB information to be reconfigure	
- RB identity	20
- PDCP info	Not Present
- PDCP SN info	Not Present
- RLC info	Not Present
- RB mapping info	Not Present
- RB stop/continue	Stop

CELL UPDATE CONFIRM (Step 11e8 and k=4)

Use the same message sub-type found in step $11e^{8}$ and k=3, with the following exceptions:

Information Element	Value/remark
RB information to release	
-RB identity	4

PHYSICAL CHANNEL RECONFIGURATION (Step 11a)

Use the same message sub-type found in [9] TS 34.108 clause 9, which is entitled "Packet to CELL FACH from CELL DCH in PS":

PHYSICAL CHANNEL RECONFIGURATION COMPLETE (Step 11b)

Use the same message sub-type found in [9] TS 34.108 clause 9.

8.3.1.1.5 Test requirement

After step 2 the UE shall reselect to cell 2 and then it shall transmit a CELL UPDATE message which, sets the value "cell reselection" in IE "Cell update cause".

After step 3 the UE shall transmit CELL UPDATE message which sets the value "cell reselection" in IE "Cell update cause".

After step 4a, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message to acknowledge that it has started to use the new RNTI identities allocated.

After step 6 the UE shall sent a CELL UPDATE message to the cell with stronger transmitting power, in order to indicate that a cell reselection has taken place.

After step 8, if k=1, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message to acknowledge that it has started to use the new RNTI identities allocated.

If k=2, the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the new physical channel assigned.

If k=3, the UE shall transmit a TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to acknowledge that it has reconfigured the transport channels.

<u>After step 11e</u>, If k=4, the UE shall transmit a RADIO BEARER RECONFIGURATION COMPLETE message to acknowledge that it has reconfigured the radio bearers.

If k=5, the UE shall transmit a RADIO BEARER RELEASE COMPLETE message to acknowledge that it has release its radio bearers.

After step 14 the UE shall transmit a CELL UPDATE message to the cell with stronger transmitting power, in order to indicate that a cell reselection has taken place.

After step 16 the UE shall transmit a CELL UPDATE message to the cell with stronger transmitting power, in order to indicate that a cell reselection has taken place.

After step 18, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message to acknowledge that it has started to use the new RNTI identities allocated.

END OF MODIFICATIONS

START OF MODIFICATIONS

- 8.3.1.3 Cell Update: periodical cell update in CELL_FACH
- 8.3.1.3.1 Definition

8.3.1.3.2 Conformance requirement

UE shall initiate the cell update procedure in the following cases:

- 1> Uplink data transmission:
 - ...
- 1> Paging response:
 - ...
- 1> Radio link failure:
 - ...
- 1> Re-entering service area:
 - • •
- 1> RLC unrecoverable error:
 - ...
- 1> Cell reselection:
 - •••
- 1> Periodical cell update:
 - 2> if none of the criteria for performing cell update with the causes specified above in the current subclause is met; and
 - 2> if the UE is in CELL_FACH or CELL_PCH state; and
 - 2> if the timer T305 expires; and
 - 2> if the criteria for "in service area" as specified in TS 25.331 subclause 8.5.5.2 is fulfilled; and
 - 2> if periodic updating has been configured by T305 in the IE "UE Timers and constants in connected mode" set to any other value than "infinity":
 - 3> perform cell update using the cause "periodical cell update".

When initiating the cell update procedure, the UE shall:

1> stop timer T305;

•••

1> move to CELL_FACH state, if not already in that state;

•••

- 1> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;
- 1> in case of a cell update procedure:

2> set the contents of the CELL UPDATE message according to TS 25.331 subclause 8.3.1.3;

2> submit the CELL UPDATE message for transmission on the uplink CCCH.

1> set counter V302 to 1;

1> start timer T302 when the MAC layer indicates success or failure in transmitting the message.

•••

In case of cell update procedure the UE shall transmit a CELL UPDATE message.

The UE shall set the IEs in the CELL UPDATE message as follows:

- 1> set the IE "Cell update cause" corresponding to the cause specified in TS 25.331 subclause 8.3.1.2 that is valid when the CELL UPDATE message is submitted to lower layers for transmission;
- NOTE: During the time period starting from when a cell update procedure is initiated by the UE until when the procedure ends, additional CELL UPDATE messages may be transmitted by the UE with different causes.
- 1> set the IE "U-RNTI" to the value of the variable U_RNTI;

•••

When the UE receives a CELL UPDATE CONFIRM message; and

- if the message is received on the CCCH, and IE "U-RNTI" is present and has the same value as the variable U_RNTI; or
- if the message is received on DCCH:

the UE shall:

1> stop timer T302;

•••

1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following:

2> perform the physical layer synchronisation procedure as specified in TS 25.214;

1> enter a state according to TS 25.331 subclause 8.6.3.3 applied on the CELL UPDATE CONFIRM / URA UPDATE CONFIRM message.

If the UE after state transition remains in CELL_FACH state, it shall

- 1> start the timer T305 using its initial value if timer T305 is not running and periodical cell update has been configured by T305 in the IE "UE Timers and constants in connected mode" set to any other value than "infinity";
- 1> select PRACH according to TS 25.331 subclause 8.5.17;
- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> not prohibit periodical status transmission in RLC;
- 1> if the IE "UTRAN DRX cycle length coefficient" is included in the same message:

2> ignore that IE and stop using DRX.

If the UE after the state transition remains in CELL_FACH state; and

- a C-RNTI is stored in the variable C_RNTI;

or

- the UE after the state transition moves to another state than the CELL_FACH state:

Release 5

the UE shall:

- 1> in case of a cell update procedure:
 - 2> set the IE "RRC transaction identifier" in any response message transmitted below to the value of "RRC transaction identifier" in the entry for the CELL UPDATE CONFIRM message in the table "Accepted transactions" in the variable TRANSACTIONS; and
 - 2> clear that entry.
- 1> transmit a response message as specified in TS 25.331 subclause 8.3.1.7;
- 1> in case of a cell update procedure:
 - 2> clear the entry for the CELL UPDATE CONFIRM message in the table "Rejected transactions" in the variable TRANSACTIONS.
- 1> set the variable CELL_UPDATE_STARTED to FALSE;
- 1> clear the variable SECURITY_MODIFICATION.
- •••

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- includes "CN information elements"; or
- includes the IE "Ciphering mode info"; or
- includes the IE "Integrity protection mode info"; or
- includes the IE "New C-RNTI"; or
- includes the IE "New U-RNTI":

the UE shall:

1> transmit a UTRAN MOBILITY INFORMATION CONFIRM as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- does not include "CN information elements"; and
- does not include the IE "Ciphering mode info"; and
- does not include the IE "Integrity protection mode info"; and
- does not include the IE "New C-RNTI"; and
- does not include the IE "New U-RNTI":

the UE shall:

1> transmit no response message.

•••

Release 5

When the UE receives a UTRAN MOBILITY INFORMATION message, it shall:

- 1> act on received information elements as specified in TS 25.331 subclause 8.6;
- 1> if the IE "UE Timers and constants in connected mode" is present:
 - 2> store the values of the IE "UE Timers and constants in connected mode" in the variable TIMERS_AND_CONSTANTS, replacing any previously stored value for each timer and constant; and
 - 2> for each updated timer value:
 - 3> start using the new value next time the timer is started;
 - 2> for each updated constant value:
 - 3> start using the new value directly;
- 1> set the IE "RRC transaction identifier" in the UTRAN MOBILITY INFORMATION CONFIRM message to the value of "RRC transaction identifier" in the entry for the UTRAN MOBILITY INFORMATION message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

1> transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC;

•••

Reference

3GPP TS 25.331 clause 8.3.1, 8.3.3.3.

8.3.1.3.3 Test purpose

1. To confirm that the UE executes a periodical cell update procedure following the expiry of timer T305.

8.3.1.3.4 Method of test

Initial Condition

System Simulator: 2 cells – Cell 1 and 2 are active.

UE: PS-DCCH+DTCH_FACH (state 6-11) in cell 1 as specified in clause 7.4 of TS 34.108.

Test Procedure

Table 8.3.1.3

Parameter	Unit	Cell 1		Ce	ll 2
		T0	T1	T0	T1
UTRA RF Channel Number		Ch. 1		Ch. 1	
CPICH Ec (FDD)	dBm/ 3.84 MHz	-60	-75	-75	-60
P-CCPCH RSCP (TDD)	dBm	-60	-75	-75	-60

Table 8.3.1.3 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution.

The UE is in CELL_FACH state. When the UE detects the expiry of timer T305 according to the settings in system information, the UE transmits a CELL UPDATE message to the SS on the uplink CCCH with a cause indicating periodical cell updating. SS replies with a CELL UPDATE CONFIRM message, and IE "RRC State Indicator" is set to

"CELL_FACH". SS verifies that the UE does not transmit any uplink message. SS then waits for T305 to expire again. The UE shall send another CELL UPDATE message to report periodic cell updating. After the SS receives this message, it transmits a CELL UPDATE CONFIRM message which includes the IEs "new C-RNTI", "new U-RNTI" to the UE on the downlink DCCH. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities. Next, SS transmits UTRAN MOBILITY INFORMATION message, which includes IE "T305" set to "infinity", to UE. UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message. SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.3, causing the UE to enter CELL FACH state in cell 2 and transmit a CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". Then SS shall transmit CELL UPDATE CONFIRM. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities. SS then monitors the uplink CCCH for a period of 60 minutes (ideally the SS should monitor this up to the maximum possible value for timer T305 (720 minutes), but for practical reasons 60 minutes (twice default timer of 30 minutes) is regarded as being sufficient) and verifies that no CELL UPDATE message is received. After this, the SS transmits UTRAN MOBILITY INFORMATION message, which includes IE "T305" set to '5', to UE. UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message. SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.3, causing the UE to enter CELL FACH state in cell 1 and transmit a CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". Then SS shall transmit CELL UPDATE CONFIRM. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities. UE shall resume periodic cell updating procedure and transmit CELL UPDATE message after T305 (5 minutes) expires.

Expected sequence

Step	Direc	Direction Message		Comment
	UE	SS		
1				The UE is in the CELL_FACH
				state. SS waits until T305 has
				expired.
2	-	>	CELL UPDATE	IE "Cell update cause" shall be
				set to "periodical cell updating"
3	<		CELL UPDATE CONFIRM	No RNTI identities are given.
				No information on PRACH and
				S-CCPCH are provided.
4				SS verifies that no uplink
				message is received from UE.
				SS waits for another period to
				allow T305 to expire.
5		>	CELL UPDATE	Set to "periodical cell update"
				in IE "Cell update cause" upon
				the expiry of timer T305.
6	€	-	CELL UPDATE CONFIRM	Including IEs "new C-RNTI",
				"new U-RNTI" and IE "RRC
				State Indicator" is set to
1				"CELL_FACH"
7		>	UTRAN MOBILITY INFORMATION	
			CONFIRM	

8	~	UTRAN MOBILITY INFORMATION	IE "T305" is set to 'infintiy'.
9	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
10			SS applies the downlink transmission power settings, according to the values in columns "T1" of table 8.3.1.3
11	\rightarrow	CELL UPDATE	IE "Cell update cause" shall be set to "cell reselection".
12	\leftarrow	CELL UPDATE CONFIRM	
12a	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
13			SS waits for 60 minutes and checks that no CELL UPDATE message is transmitted on uplink PRACH channel.
14	÷	UTRAN MOBILITY INFORMATION	IE "T305" is set to '5.
15	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
16			SS applies the downlink transmission power settings, according to the values in columns "T0" of table 8.3.1.3
17	*	CELL UPDATE	IE "Cell update cause" shall be set to "cell reselection".
18	÷	CELL UPDATE CONFIRM	
18a	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
19	→	CELL UPDATE	UE shall transmit this message with "cell update cause" set to "periodical cell updating" after T305 expires.
20	÷	CELL UPDATE CONFIRM	

Specific Message Contents

CELL UPDATE (Step 2 and 5)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark		
U-RNTI			
- SRNC Identity	Check to see if set to '0000 0000 0001'		
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'		
Cell Update Cause	Check to see if set to 'periodical cell updating'		

CELL UPDATE (Step 11 and 17)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark			
U-RNTI				
- SRNC Identity	Check to see if set to '0000 0000 0001'			
- S-RNTI	Check to see if set to same bit string as in IE "S-RNTI"			
	in IE "U-RNTI" of the CELL UPDATE CONFIRM			
	message sent in step 6.			
Cell Update Cause	Check to see if set to "cell reselection"			

CELL UPDATE CONFIRM (Step 3, 12, 18 and 20)

Use the same message sub-type found in TS 34.108, clause 9.

CELL UPDATE CONFIRM (Step 6, 12 and 18)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark		
New U-RNTI			
- SRNC Identity	Set to '0000 0000 0001'		
- S-RNTI	Set to an arbitrary string different from '0000 0000 0000		
	0000 0001'		
New C-RNTI	'1010 1010 1010 1010'		

CELL UPDATE (Step 19)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to same bit string as in IE "S-RNTI"
	in IE "U-RNTI" of the CELL UPDATE CONFIRM
	message sent in step 6.
Cell Update Cause	Check to see if set to 'periodical cell updating'

UTRAN MOBILITY INFORMATION (Step 8)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
New U-RNTI	Not Present
New C-RNTI	Not Present
UE Timers and constants in connected mode	
- T305	infinity

UTRAN MOBILITY INFORMATION (Step 14)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark		
New U-RNTI	Not Present		
New C-RNTI	Not Present		
UE Timers and constants in connected mode			
- T305	5		

Release 5

8.3.1.3.5 Test requirement

After step 1 the UE shall detect the expiry of timer T305 and then transmits a CELL UPDATE message setting value "periodical cell update" into IE "Cell update cause".

After step 3 the UE shall not send any uplink message as a response to CELL UPDATE CONFIRM message sent in step 3.

After step 4 the UE shall send a CELL UPDATE message, specifying the cell updating cause to be "periodical cell update".

After step 6 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

After step 8, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 10, the UE shall transmit a CELL UPDATE message with IE "cell update cause" set to "cell reselection" on the uplink CCCH.

After step 12, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

Between step 12a and 14, the UE shall not transmit any CELL UPDATE message.

After step 14, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 16, the UE shall transmit a CELL UPDATE message with IE "cell update cause" set to "cell reselection" on the uplink CCCH.

After step 18, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 18a, the UE shall transmit a CELL UPDATE message with IE "cell update cause" set to "periodical cell update" on the uplink CCCH.

END OF MODIFICATIONS

START OF MODIFICATIONS

- 8.4.1.5 Measurement Control and Report: Intra-frequency measurement for transition from CELL_DCH to CELL_FACH state
- 8.4.1.5.1 Definition
- 8.4.1.5.2 Conformance requirement

Upon transition from CELL_DCH to CELL_FACH/CELL_PCH/URA_PCH state, the UE shall:

- 1> stop intra-frequency type measurement reporting;
- 1> if the transition is due to a reconfiguration message which included the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD), and the UE selects a cell other than that indicated by this IE; or
- 1> if the transition is due to a reconfiguration message which does not include the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD); or
- 1> if the transition is not due to a reconfiguration message:

2> delete the measurements of type intra-frequency associated with the variable MEASUREMENT_IDENTITY.

1> begin monitoring cells listed in the IE "intra-frequency cell info list" received in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11 in [8] TS 25.331).

Upon transition from CELL_FACH to CELL_DCH state, the UE shall:

- 1> retrieve each set of measurement control information of measurement type "intra-frequency" stored in the variable MEASUREMENT_IDENTITY;
- 1> if the IE "measurement validity" for a measurement has been assigned the value "CELL_DCH:

2> resume the measurement reporting.

- 1> if no intra-frequency measurements applicable to CELL_DCH state are stored in the variable MEASUREMENT_IDENTITY:
 - 2> continue monitoring the list of neighbouring cells assigned in the IE "intra-frequency cell info list" in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11 in [8] TS 25.331);
 - 2> if the IE "intra-frequency measurement reporting criteria" was included in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11 in [8] TS 25.331):
 - 3> send the MEASUREMENT REPORT message when reporting criteria in IE "Reporting information for state CELL_DCH" are fulfilled.

Reference

3GPP TS 25.331, clause 8.4.1.6.1, 8.4.1.7.1

8.4.1.5.3 Test Purpose

- 1. To confirm that the UE stops performing intra-frequency measurement reporting specified in a MEASUREMENT CONTROL message, when it moves from CELL_DCH state to CELL_FACH state.
- 2. To confirm that the UE reads the System Information Block type 11 or 12 messages when it enters CELL_FACH state from CELL_DCH state, and starts to monitor the cells listed in the IE "intra-frequency cell info list".
- 3 To confirm that the UE performs measurements on uplink RACH transmissions and appends the measured results in RACH messages, when it receives IE "intra-frequency reporting quantity for RACH reporting" and IE "Maximum number of reported cells on RACH" in the System Information Block type 11 or 12 messages.
- 4. To confirm that the UE applies the reporting criteria in IE "intra-frequency reporting criteria" in System Information Block Type 11 or 12 messages following a state transition from CELL_FACH to CELL_DCH, if no intra-frequency measurements applicable to CELL_DCH are stored.

8.4.1.5.4 Method of test

Initial Condition

System Simulator: 3 cells - Cell 1 and cell 2 are active, while cell 3 is switched off..

UE: PS-DCCH+DTCH_DCH (state 6-10) in cell 1 as specified in clause 7.4 of TS 34.108.

Specific Message Contents

For MASTER IFORMATION BLOCK and system information block 11 of Cell 1 (gives IE's which are different from defaults given in 34.108 subclause 6.1) to be transmitted before idle update preamble.

MASTER INFORMATION BLOCK

Use the same message sub-type found in clause 6.1 of TS 34.108, with the following exception:

Information Element	Value/Remarks		
MIB Value Tag	1		

System Information Block type 11

Use the same message sub-type found in clause 6.1 of TS 34.108, with the following exception:

Information Element	Value/remark
SIB12 indicator	FALSE
FACH measurement occasion info	Not Present
Measurement control system information	
- Use of HCS	Not used
 Cell selection and reselection quality measure 	CPICH RSCP
 Intra-frequency measurement system information 	
 Intra-frequency measurement identity 	Not present
 Intra-frequency cell info list 	
 CHOICE intra-frequency cell removal 	Remove no intra-frequency cells
 New intra-frequency cells 	
 Intra-frequency cell id 	1
<u> </u>	
- Cell individual offset	<u>0 dB</u>
 Reference time difference to cell 	Not present
- Read SFN Indicator	FALSE
- CHOICE mode	<u>FDD</u>
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	Not present
- Cells for measurement	Not Present
-Intra-frequency measurement quantity	Not Present
-Intra-frequency reporting quantity for RACH	Not Present
reporting	
-Maximum number of reported cells on RACH	Not Present
-Reporting information for state CELL_DCH	Not Present
 Inter-frequency measurement system information 	Not Present
 Inter-RAT measurement system information 	Not Present
 Traffic volume measurement system information 	Not Present
 UE internal measurement system information 	Not Present

Test Procedure

Table 8.4.1.5-1 illustrates the downlink power to be applied for the 3 cells at various time instants of the test execution. Columns marked "T0" denote the initial conditions, while columns marked "T1 are to be applied subsequently. The exact instants on which these values shall be applied are described in the texts in this clause.

Parameter	Unit	Cell 1		Cell 2		Cell 3	
		T0	T1	T0	T1	T0	T1
UTRA RF Channel Number		Ch. 1		Ch. 1		Ch. 1	
CPICH Ec	dBm/ 3.84 MHz	-60	-60	-75	-85	-122	-70

Table 8.4.1.5-1

The UE is initially in CELL_DCH state. The System Information Block type 11 message is modified compared to the default message contents, in order to prevent the reporting of "Cell synchronisation information". No measurement to be applied by the UE in CELL_DCH state is specified in any of the System Information Block type 11 or 12 messages.

Release 5

SS sends a MEASUREMENT CONTROL message to UE. In this message, the SS requests the establishment of an intra-frequency measurement for the measurement of cell 2's CPICH RSCP. At the same time, reporting of CPICH RSCP values of active set cells and monitored set cells are requested with the reporting criteria set to "periodic reporting" and "reporting interval" set to 16 seconds. The UE shall start transmitting MEASUREMENT REPORT messages at 16 seconds interval corresponding to the requested reporting event.

SS transmits PHYSICAL CHANNEL RECONFIGURATION message to move the UE to CELL_FACH. After receiving this message, the UE shall reconfigure itself and reply with a PHYSICAL CHANNEL

RECONFIGURATION COMPLETE message on RACH. SS monitors the uplink channels to verify that no MEASUREMENT REPORT messages are received.

SS reconfigures itself according to the settings in columns marked "T1" in table 8.4.1.5-1. SS transmits System Information Block type 12 messages in cell 1, which include cell 3 into the IE "intra-frequency cell info list" and modifies SIB11 to indicate that SIB12 is now being broadcast. IEs "Intra-frequency reporting quantity for RACH Reporting" and IE "Maximum number of Reported cells on RACH" are also specified in the System Information Type 12 messages. Event type 1a reporting criterion is specified for intra-frequency measurements. SS transmit SYSTEM INFORMATION CHANGE INDICATION message to UE. SS waits until T305 has expired. The UE shall respond with a CELL UPDATE message, which comprises IE "Measured results on RACH" to report the readings of CPICH RSCP for cell 1 and cell 3. SS replies with CELL UPDATE CONFIRM message on the downlink DCCH. This message does not change the physical resources nor allocate any new RNTI identities. SS transmits PHYSICAL CHANNEL RECONFIGURATION message again, and configures dedicated physical channel for both uplink and downlink directions. The UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and return to CELL_DCH state. SS listens to the uplink DCCH for MEASUREMENT REPORT messages.

SS shall receive the MEASUREMENT REPORT messages at 500 milliseconds interval.

SS verifies that it includes CPICH RSCP values of the cells 1, 2 and 3 in IE "Cell measured results" and the triggering of event '1a' on cell 3 in IE "Event results".

Step	Direction	Message	Comment
•	UE SS		
1	÷	Master Information Block System Information Block type 11	UE is in PS- DCCH+DTCH_DCH (state 6- 10) in cell 1. System Information Block Type 11 to be broadcast does not specify any measurement type to be configured in the UE in CELL_DCH.
2		Void	
3		Void	
4		Void	
5	÷	MEASUREMENT CONTROL	SS requests for measurement of cell 2's CPICH RSCP value and reporting of CPICH RSCP values of active cells and monitored set cells.
6	\rightarrow	MEASUREMENT REPORT	UE shall send periodic report at 16 seconds interval.
7	\leftarrow	PHYSICAL CHANNEL RECONFIGURATION	SS moves the UE to CELL_FACH state.
8	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall move to CELL_FACH state.
9	÷	Master Information Block System Information Block type <u>11,</u> 12	SS reconfigures itself according to the settings stated in column "T1" of table 8.4.1.5-1. SIB 11 is modified to indicate that SIB12 is now broadcast and to add cell 2 as a neighbour cell. SIB 12 indicates that cell 3 is included in the IE "intra-frequency cell info list". SS waits for 1 minute and verifies that no MEASUREMENT REPORT messages are detected on the uplink.
10	÷	SYSTEM INFORMATION CHANGE INDICATION	SS waits until T305 has expired.
11	→	CELL UPDATE	UE shall transmit this message with measured results on RACH channels for cell 1 and cell 3 present in this message.

Expected Sequence

Step	Direction	Message	Comment
	UE SS		
12	÷	CELL UPDATE CONFIRM	No changes in physical resource allocation and RNTI identities.
13	←	PHYSICAL CHANNEL	SS configures dedicated
		RECONFIGURATION	physical channels.
14	\rightarrow	PHYSICAL CHANNEL	UE shall transit to CELL_DCH
		RECONFIGURATION COMPLETE	state.
15	\rightarrow	MEASUREMENT REPORT	Repeated at 500 milliseconds
			interval

Specific Message Content

MASTER INFORMATION BLOCK (Step 1)

Use the same message sub-type found in clause 6.1 of TS 34.108, with the following exception:

Information Element	Value/Remarks
MIB-Value Tag	4

System Information Block type 11 (Step 1)

Information Element	Value/remark
SIB12 indicator	FALSE
FACH measurement occasion info	Not Present
Measurement control system information	
	Not used
 Cell selection and reselection quality measure 	CPICH RSCP
Intra-frequency measurement identity	Not present
- Intra-frequency cell info list	
	Remove no intra-frequency cells
- Intra-frequency cell id	4
	0 dB
 Reference time difference to cell 	Not present
- Read SFN Indicator	FALSE
	FDD
- Primary Scrambling Code	Set to same code as used for cell 1
	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	Not present
	Not Present
 Intra-frequency measurement quantity 	Not Present
——Intra-frequency reporting quantity for RACH	Not Present
reporting	
	Not Present
	Not Present
 Inter-frequency measurement system information 	Not Present
Inter-RAT measurement system information	Not Present
	Not Present
 UE internal measurement system information 	Not Present

MEASUREMENT CONTROL (Step 5)

Information Element	Value/remark
Measurement Identity	5
Measurement Command	Setup
Measurement Reporting Mode	
- Measurement Reporting Transfer Mode	Acknowledged Mode RLC
- Periodic Reporting / Event Trigger Reporting Mode	Periodical Reporting
Additional measurements list	Not Present
CHOICE measurement type	Intra-frequency measurement
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Remove no intra-frequency cells
- New intra-frequency info list	
- Intra-frequency cell id	2
- Cell info	
- Cell individual offset	0 dB
 Reference time difference to cell 	Not Present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 2
 Primary CPICH TX power 	Not Present
- TX Diversity Indicator	FALSE
- Cells for measurement	Not Present
 Intra-frequency measurement quantity 	
- Filter Coefficient	Not Present (Default is 0)
- Measurement quantity	CPICH RSCP
 Intra-frequency reporting quantity 	
 Reporting quantities for active set cells 	
 SFN-SFN observed time difference reporting 	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator	
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- Reporting quantities for monitored set cells	No report
- SFN-SFN observed time difference reporting	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator - Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
- Reporting quantities for detected cells	Not present
- Reporting cell status	
- CHOICE reported cell	Report cells within active and/or monitored set on used
	frequency or within active and/or monitored set on used
	used frequency
- Maximum number of reported cells	2
- Measurement validity	Not present
- CHOICE report criteria	Periodical reporting criteria
- Amount of reporting	Infinity
- Reporting interval	16 seconds
DPCH compressed mode status info	Not Present
	notriodolit

MEASUREMENT REPORT (Step 6)

Check to see if set to 5
Check to see if set to "Intra-frequency measured results list"
Check to see if it is absent
Check to see if this IE is absent
Check to see if this IE is absent
Check to see if it's the same code for cell 1
Check to see if this IE is absent
Check to see if this IE is <u>absentpresent</u> Check to see if this IE is absent
Check to see if it is absent
Check to see if this IF is absent
Check to see if this IE is absent
Check to see if it's the same code for cell 2
Check to see if this IE is absent
Check to see if this IE is present
Check to see if this IE is absent
Check to see if this IE is absent
Check to see if this IE is absent
Check to see if this IE is absent

PHYSICAL CHANNEL RECONFIGURATION (Step 7)

Use the same message sub-type found in [9] TS 34.108 clause 9, which is entitled "(Packet to CELL_FACH from CELL_DCH in PS)"

MASTER INFORMATION BLOCK (Step 9)

Use the same message sub-type found in clause 6.1 of TS 34.108, with the following exception:

Information Element	Value/Remarks
MIB Value Tag	2

System Information Block type 11 (Step 9)

Information Element	Value/remark
SIB12 indicator	TRUE
FACH measurement occasion info	Not Present
Measurement control system information	
- Intra-frequency measurement system information	
- Intra-frequency measurement identity	Not present
- Intra-frequency cell info list	'
- CHOICE intra-frequency cell removal	Remove no intra-frequency cells
- New intra-frequency cells	
- Intra-frequency cell id	1
- Cell info	
- Cell individual offset	0 dB
 Reference time difference to cell 	Not present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
 Cell selection and Re-selection info 	Not present
 Intra-frequency cell id 	2
- Cell info	
 Cell individual offset 	0 dB
 Reference time difference to cell 	Not present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
 Primary Scrambling Code 	Set to same code as used for cell 2
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
 Cell selection and Re-selection info 	
- Qoffset _{s,n}	0 dB
 Maximum allowed UL TX power 	0 dBm
- HCS neighbouring cell information	Not Present
- CHOICE Mode	FDD
- Qqualmin	-20dB
- Qrxlevmin	-115dBm
- Cells for measurement	Not Present
-Intra-frequency measurement quantity	Not Present
-Intra-frequency reporting quantity for RACH	Not Present
reporting	
-Maximum number of reported cells on RACH	Not Present
-Reporting information for state CELL_DCH	Not Present
- Inter-frequency measurement system information	Not Present
- Inter-RAT measurement system information	Not Present
- Traffic volume measurement system information	Not Present
- UE internal measurement system information	Not Present

System Information Block type 12 (Step 9)

Information Element	Value/remark
FACH measurement occasion info	Not Present
Measurement control system information	
- Use of HCS	Not used
- Cell selection and reselection quality measure	CPICH RSCP
- Intra-frequency measurement system information	
 Intra-frequency measurement identity 	6
 Intra-frequency cell cells 	
- CHOICE intra-frequency cell removal	Remove no intra-frequency cells
- New intra-frequency cells	
- Intra-frequency cell id	3
- Cell info	
 Cell individual offset Reference time difference to cell 	0 dB Not Present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 3
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	
- Qoffset _{s,n}	OdB
- Maximum allowed UL TX power	0dBm
- HCS neighbouring cell information	Not Present
- CHOICE Mode	FDD
- Qqualmin, Qrxlevmin	-20dB, -115dBm
 Intra-frequency measurement quantity Filter Coefficient 	Not Present (Default is 0)
- Measurement quantity	CPICH RSCP
- Intra-frequency measurement for RACH reporting	
- SFN-SFN observed time difference	No report
- Reporting quantity	CPICH RSCP
- Maximum number of reported cells on RACH	Current cell + best neighbour
 Reporting information for state CELL_DCH 	
 Intra-frequency reporting quantity 	
- Reporting quantities for active set cells	
- SFN-SFN observed time difference reporting	No report
indicator - Cell synchronisation information reporting	FALSE
indicator	FALSE
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
 Reporting quantities for monitored set cells 	
- SFN-SFN observed time difference reporting	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator	
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator	FALSE TRUE
- CPICH RSCP reporting indicator	FALSE
- Reporting quantities for detected cells	Not present
- CHOICE report criteria	Intra-frequency measurement reporting criteria
- Parameter required for each event	
- Intra-frequency event identity	1a
- Triggering condition 1	Not Present
- Triggering condition 2	Monitored set cells
- Reporting range constant	20.0 dB
- Cells forbidden to affect reporting	Not present
- W	
- Hysteresis	1.0 dB Not Present
 Threshold used frequency Reporting deactivation threshold 	7
- Replacement activation threshold	Not Present

- Time to trigger	60 ms
 Amount of reporting 	Infinity
- Reporting Interval	500 milliseconds
- Reporting cell status	
- CHOICE reported cell	Report cells within active and/or monitored set on used frequency or within active and/or monitored set on non- used frequency
 Maximum number of reported cells 	3
- Inter-frequency measurement system information	Not present
 Inter-RAT measurement system information 	Not present
- Traffic volume measurement system information	Not present
- UE internal measurement system information	Not present

SYSTEM INFORMATION CHANGE INDICATION (Step 10)

Information Element	Value/Remarks
BCCH modification info	
- MIB Value tag	2

CELL UPDATE (Step 11)

Information Element	Value/remark
U-RNTI	Check to see if set to the same value assigned during
	the execution of procedure P3 or P5.
START list	Checked to see if this IE is present
AM_RLC error indication(RB2, RB3 or RB4)	FALSE
AM_RLC error indication(RB>4)	FALSE
Cell update cause	Check to see if it is set to "Periodical cell update"
Failure case	Check to see if it is absent
Measured results on RACH	
 Measurement result for current cell 	
 CHOICE measurement quantity 	Check to see if set to "CPICH RSCP"
- CPICH RSCP	Check to see if it is present
 Measurement results for monitored cells 	
 SFN-SFN observed time difference 	Not Checked
- Primary CPICH info	
- Primary scrambling code	Check to see if the same as cell 3's code.
- CHOICE measurement quantity	Check to see if set to "CPICH RSCP"
- CPICH RSCP	Check to see if it is present

PHYSICAL CHANNEL RECONFIGURATION (Step 13)

Use the same message sub-type found in [9] TS 34.108 clause 9, which is entitled "(Packet to CELL_DCH from CELL_FACH in PS)".

MEASUREMENT REPORT (Step 15)

Information Element	Value/remark
Measurement identity	Check to see if set to 6
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measurement results list 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
- SFN-SFN observed time difference	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 1
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	
- Cell Identity	Check to see if it is absent
 SFN-SFN observed time difference 	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 3
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event results	Check to see if this set to 'Intra-frequency measurement
	event results'
 Intra-frequency event identity 	Check to see if set to '1a'
 Cell measurement event results 	
- CHOICE Mode	Check to see if set to 'FDD'
- Primary CPICH info	
- Primary Scrambling Code	Check to see if set to the same code for cell 3

8.4.1.5.5 Test Requirement

After step 5, the UE shall start to transmit MEASUREMENT REPORT messages at 16 seconds interval. The message shall contain IE "measured result" to report cell 2's CPICH RSCP value.

After step 8, the UE shall not send any MEASUREMENT REPORT messages containing reporting quantities requested in MEASUREMENT CONTROL messages in step 5.

After step 10, the UE shall perform a cell update procedure and transmit a CELL UPDATE message. In this message, measured values CPICH RSCP for cell 1 and cell 3 shall be included in the IE "measured results on RACH".

After step 15, the UE shall apply the intra-frequency measurement reporting criteria" received in System Information Block type 12 messages of step 9. It shall send MEASUREMENT REPORT messages at 500 milliseconds interval. In these messages, triggering of event '1a' shall be reported in IE "Event results" with IE "Primary CPICH info" containing the primary scrambling code for cell 3.

The message shall contain IE "measured result" to report CPICH RSCP values of cell 1, 2 and 3.

END OF MODIFICATIONS

3GPP TSG-T1/SIG Meeting #26 Singapore, 4 – 8 November 2002

Tdoc	ж Т	1S-	020	658
------	------------	------------	-----	-----

							CR-Form-v7
	CHANGE REQUEST						
[#] 34	<mark>.123-1</mark>	CR <mark>364</mark>	ж rev	- * C	Current vers	^{ion:} 5.1.1	ж
For <u>HELP</u> on usi	ng this for	m, see bottom o	of this page or	look at the p	pop-up text	over the # syr	nbols.
Proposed change af	fects: l	JICC apps #	ME X	Radio Acc	ess Networ	k Core Ne	etwork
				-			
Title: ೫	CR to TS	<mark>34.123-1 R5; U</mark> r	date to TC7 2	3 19(RI C	PDU Contir	nuous Transmi	ssion)
		04.120 1 1(0, 0)		.0.15(1120			001011)
	Anritsu						
Work item code:	TEI				<i>Date:</i>	11/10/2002	
Category: ೫				F	Release: ೫		
C	F (con A (cor B (add C (fun D (edi Detailed exp	the following cate rection) responds to a cor dition of feature), ctional modification torial modification blanations of the 3GPP <u>TR 21.900</u>	- rection in an ear on of feature)) above categories		Use <u>one</u> of 2 R96 R97 R98 R99 Rel-4 Rel-5 Rel-6	the following rele (GSM Phase 2) (Release 1996) (Release 1997) (Release 1998) (Release 1999) (Release 4) (Release 5) (Release 6)	eases:
Reason for change:	¥ Curr	ent test procedu	re does not gu	larantee cor	ntinuous up	link transmissio	on, and
	the to	est purpose is n	ot completely a	achieved.			
Summary of change	refle	CR supersedes ctor as T1S-020 osed during offli	658 during T1	/SIG#25 in S	Singapore t		
	chan	ated value of tim ge is identical to iple during T1/S	the change p				
	in ma Upda	ated loopback si any UL PDUs. T ated number of s alling from the U	his ensures th	at the UE w	vill transmit	data in every T	TI.
	1. Th - a - a	ated test proced ne P bit is set wh poll is triggered poll is triggered the UE shall re	nen: during normal and there is da	transmissio ata 'availabl	on. le' (i.e. unac		in which
		PDUs are sent owledged.	when timer_p	oll_periodic	expires an	d all PDUs hav	ve been

	3. When the timer expires in case 2 above, the timer is restarted, and the P bit is set the next time a poll is triggered during normal transmission.Updated expected sequence and notes to reflect these changes.Updated test requirements to reflect these changes.		
Consequences if # not approved:	Test purpose is not achieved.		
Clauses affected: #	7.2.3.19		
Other specs % affected:			

How to create CRs using this form:

Other comments:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

Affects R99, REL-4 and REL-5 test cases.

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

7.2.3.19 Polling for status / Timer triggered polling (Timer_Poll_Periodic)

7.2.3.19.1 Definition

This case tests that the UE will poll for a status request every Timer_Poll_Periodic ms when that mode is enabled. Incorrect operation of polling will cause degradation of service, or at worst service failure.

7.2.3.19.2 Conformance requirement

This timer shall only be used when "timer based polling" is configured by upper layers. The value of the timer is signalled by upper layers. The timer shall be started when the RLC entity is created. When the timer expires, the RLC entity shall:

- restart the timer;
- if AMD PDUs are available for transmission or retransmission (not yet acknowledged):
 - trigger a poll.

[...]

The Sender shall:

- if a poll has been triggered by one or several poll triggers (see TS 25.322 subclause 9.7.1):
 - if polling is not prohibited, see TS 25.322 subclause 9.5:
 - set the "Polling bit" in the AMD PDU header to "1";
- otherwise:
 - set the "Polling bit" in the AMD PDU header to "0".

Reference

TS 25.322 clauses 9.5, 9.7.1 and 11.3.2.1.1.

7.2.3.19.3 Test purpose

- 1. To verify that the UE polls the SS in the next PDU to be transmitted or retransmitted each time the Timer_Poll_Periodic timer expires.
- 2. To verify that if there is no PDU to be transmitted, and all the PDUs have already been acknowledged, the timer is restarted, but no poll is sent.

7.2.3.19.4 Method of test

Initial conditions

The generic procedure for Radio Bearer establishment (clause 7.1.3 of TS 34.108) is executed, with all the parameters as specified in the procedure, with the exception that the default Radio Access Bearer is replaced with the RAB defined for AM 7-bit "Length Indicator" tests in clause 7.2.3.1.

The following RLC parameter values are used in place of the values in clause 7.2.3.1:

Uplink RLC	First run	Second run
Polling info		
Last retransmission PDU poll	FALSE	FALSE
Last transmission PDU poll	FALSE	FALSE
Timer_poll_periodic	<u>5</u> 400	2000

These settings apply to both the uplink and downlink DTCH.

The Radio Bearer is placed in UE test loop mode 1 with the UL SDU size set to $(AM_7_PayloadSize * ceil(T/(5*TTI)))$ - 1 bytes.

Test procedure

Let T be the value of Timer_Poll_Periodic:

- a) The SS waits for at least 2*T ms before starting any transmissions, and monitors the uplink.
- b) The SS sends 4 * (T/TTI) five RLC SDUs of size floor(AM_7_PayloadSize/52) 1 bytes to the UE. The SDUs are packed concatenated five2 SDUs to one PDU. The UE is expected to loop this data back in five RLC SDUs, segmented into a total of at least ceil(T/TTI) RLC PDUs.
- c) The SS checks that at least one RLC PDU is received on the uplink with the P bit set and records the arrival time of the last RLC PDU received with the P bit set (T_1) . The SS does not send any STATUS PDUs in response to these poll requests. The SS waits for the first PDU to be received with the P bit set, records the arrival time (T_1) and responds with a STATUS PDU normally.
- d) The SS continues to receive RLC PDUs until all of the data has been received. The SS waits for the reception of the next PDU with the P bit set, records the arrival time (T_2) , and then transmits a STATUS PDU reporting that none of the uplink PDUs were correctly received, except for the last PDU containing the poll bit which is acknowledged.
- e) The SS waits for the UE to retransmit an RLC PDU in order to transmit a poll (this may be the PDU with SN VT(S) 1, or a PDU that has not been acknowledged). The SS checks that The SS waits for the next PDU received with the P bit is set, and records the arrival time $(\underline{T_3T_2})$.
- f) The SS responds with a STATUS PDU acknowledging all received PDUs. The SS waits for the reception of the next PDU with the P bit set and records the arrival time (T_4) .
- g) The SS waits for 2*T ms to ensure that no further polls are received from the UE.
- h) The SS sends five RLC SDUs of size floor(AM_7_PayloadSize/5) 1 bytes to the UE. The SDUs are concatenated five SDUs to one PDU. The UE is expected to loop this data back in five RLC SDUs, segmented into a total of at least ceil(T/TTI) RLC PDUs.
- i) The SS checks that at least one RLC PDU is received on the uplink with the poll bit set and records the arrival time of the first RLC PDU received with the poll bit set (T₃). The SS does not send any STATUS PDUs in response to these poll requests.
- j) The SS continues to receive RLC PDUs until all of the data has been received.
- k) The SS responds with a STATUS PDU acknowledging all received PDUs.
- lg) The SS may optionally release the radio bearer.

The Test is repeated using the parameters specified for the second run.

Expected sequence

Step Direction		irection Message		Comments	
	UE	SS			
1	÷	-	DOWNLINK RLC PDU	SDU 1, SDU2 <u>, SDU 3, SDU 4, SDU 5, SN=0</u>	
2 ←		-		SS continues to transmit RLC SDUs	
3	+		DOWNLINK RLC PDU	SDU 4*(T/TTI)-1SDU 4*(T/TTI)	
<u>2</u> 4			UPLINK RLC PDU	PSDU 1, SN=0	
<u>3</u> 5		>	UPLINK RLC PDU	P <mark>S</mark> DU 2, SN=1	
<u>4</u> 6	÷	>		SS continues to receive RLC PDUs	
<u>5</u> 7			UPLINK RLC PDU	$\frac{SN = x}{T_1}$ Poll: Note T ₁	
8			STATUS PDU	ACK SN 0 to SN x	
9		>	UPLINK RLC PDU	SN = x+1	
<u>6</u> 10		>		SS continues to receive RLC PDUs	
<u>7</u> 44			UPLINK RLC PDU	SN = x + ceil(T/TTI) - 1, Poll: Note T2	
12	←		STATUS PDU	NAK SN x+1 to SN x + coil(T/TTI)-1	
<u>8</u> 13	~	*	UPLINK RLC PDU	PDUs including some retransmissionsRetransmission of VT(S)-1	
				or unacknowledged PDU in order to transmit	
				a poll. Poll: Note T ₂ .	
14		>	UPLINK RLC PDU		
15	5 →			SS continues to receive RLC PDUs	
16	\rightarrow		UPLINK RLC PDU	Poll: Note T ₃	
<u>9</u> 17	7 ←		STATUS PDU	Normal	
1 <mark>0</mark> 8	\rightarrow	<u></u>		SS continues to receive RLC PDUsmonitors	
				uplink for 2*T ms	
<u>11</u>	<u> </u>	-	DOWNLINK RLC PDU	<u>SDU 6, SDU 7, SDU 8, SDU 9, SDU 10,</u> SN=1	
<u>12</u>	<u> </u>	<u>></u>	UPLINK RLC PDU	$\frac{SIV=1}{SN = ceil(T/TTI)}$	
<u>13</u>	3	<u>></u>	<u></u>	SS continues to receive RLC PDUs	
<u>14</u>	E	<u>></u>	UPLINK RLC PDU	Poll: Note T ₃	
<u>15</u>	<u> –</u>	<u>></u>	<u></u>	SS continues to receive RLC PDUs	
1 <u>6</u> 9	÷	>	UPLINK RLC PDU	Poll: Note T ₄ SN = 2*ceil(T/TTI)-1	
<u>17</u>	<u> </u>		STATUS PDU	<u>Normal</u>	
1820 RB RELEASE Optional step					
NOTE			bected Sequence shown is infomative.		
			-INK and DOWNLINK PDU flows may ove		
			red informative only, for test case develop	bers given in the comments column shall be	
			e of x may be different for each iteration.		
tore 2. The value of A may be different for each relation:					

7.2.3.19.5

Test requirements

<u>1.</u> No PDUs shall be received from the UE for 2*T ms before step 1.

<u>x <= ceil (T/TTI).</u>

<u>2.</u> Time $T_2 - T_1$ shall be T.

3. No PDUs shall be received from the UE for 2*T ms after step 9

<u>4.</u> Time $T_3 - T_2$ shall be <u>n*T for some integer $n \ge 0$.</u>

99

3GPP TSG-T1 N Luton, UK, 4 th –		0784
3GPP TSG-T1/S	-	0734
Luton, UK, 4 th –		Form-v7
	CHANGE REQUEST	-
ж 3	4.123-1 CR 337 # rev - ^{# Current version:} 5.1.1 [#]	
For <u>HELP</u> on u	sing this form, see bottom of this page or look at the pop-up text over the st symbo	ols.
Proposed change Title: अ	Affects: UICC apps # ME X Radio Access Network Core Network Core Network Core Network Core Network CR to 34.123-1 clause 8.1 (Package 1) Rel-5: Correction from CRs approved in	ork
ா பட். ூ	RP17meeting	
Source: ೫	Panasonic	
Work item code: ଞ	TEI Date: ₩ 28/10/2002	
Category: ₩	FRelease: %REL-5Use one of the following categories:Use one of the following releaseF (correction)2(GSM Phase 2)A (corresponds to a correction in an earlier release)R96(Release 1996)B (addition of feature),R97(Release 1997)C (functional modification of feature)R98(Release 1998)D (editorial modification)R99(Release 1999)Detailed explanations of the above categories canRel-4(Release 4)be found in 3GPP TR 21.900.Rel-5(Release 5)Rel-6(Release 6)Rel-6	əs:
Reason for change	 1. From CR1680 The use of synchronization procedures A and B is clarified wherever applicable. 	
Summary of chang	e: # 1. Change to 8.1.2.1 The conformance requirement was revised	
Consequences if not approved:	# The test specifications are not aligned with the core specification	
Clauses affected:	¥ 8.1.2.1	
Other specs Affected:	Y N % X Other core specifications % X Test specifications X O&M Specifications	
Other comments:	# Affects R99, REL-4, REL-5	

How to create CRs using this form:

Release 5

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

8.1.2 RRC Connection Establishment

8.1.2.1 RRC Connection Establishment in CELL_DCH state: Success

8.1.2.1.1 Definition

8.1.2.1.2 Conformance requirement

The UE shall initiate the procedure when upper layers in the UE requests the establishment of a signalling connection and the UE is in idle mode (no RRC connection exists).

Upon initiation of the procedure, the UE shall:

•••

- 1> set the contents of the RRC CONNECTION REQUEST message according to TS 25.331 subclause 8.1.3.3;
- 1> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;
- 1> perform the mapping of the Access Class to an Access Service Class as specified in TS 25.331 subclause 8.5.13, and apply the given Access Service Class when accessing the RACH;
- 1> submit the RRC CONNECTION REQUEST message for transmission on the uplink CCCH;
- 1> set counter V300 to 1; and
- 1> start timer T300 when the MAC layer indicates success or failure to transmit the message;
- 1> select a Secondary CCPCH according to TS 25.304;
- 1> start receiving all FACH transport channels mapped on the selected Secondary CCPCH.

The UE shall, in the transmitted RRC CONNECTION REQUEST message:

- 1> set the IE "Establishment cause" to the value of the variable ESTABLISHMENT_CAUSE;
- 1> set the IE "Initial UE identity" to the value of the variable INITIAL_UE_IDENTITY;

•••

The UE shall compare the value of the IE "Initial UE identity" in the received RRC CONNECTION SETUP message with the value of the variable INITIAL_UE_IDENTITY.

If the values are different, the UE shall:

1> ignore the rest of the message.

If the values are identical, the UE shall:

- 1> stop timer T300, and act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following:
 - 2> if the UE will be in the CELL_FACH state at the conclusion of this procedure:

...

1> if the UE will be in the CELL_DCH state at the conclusion of this procedure:

<u>2</u>+> perform the physical layer synchronization procedure <u>A</u> as specified in TS 25.214;

1> enter UTRA RRC connected mode, in a state according to TS 25.331 subclause 8.6.3.3;

1> submit an RRC CONNECTION SETUP COMPLETE message to the lower layers on the uplink DCCH after successful state transition per TS 25.331 subclause 8.6.3.3, with the contents set as specified below:

2> set the IE "RRC transaction identifier" to:

- 3> the value of "RRC transaction identifier" in the entry for the RRC CONNECTION SETUP message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 3> clear that entry.

•••

- 2> retrieve its UTRA UE radio access capability information elements from variable UE_CAPABILITY_REQUESTED; and then
- 2> include this in IE "UE radio access capability" and IE "UE radio access capability extension", provided this IE is included in variable UE_CAPABILITY_REQUESTED;
- 2> retrieve its inter-RAT-specific UE radio access capability information elements from variable UE_CAPABILITY_REQUESTED; and then
- 2> include this in IE "UE system specific capability".

When the RRC CONNECTION SETUP COMPLETE message has been submitted to lower layers for transmission the UE shall:

1> consider the procedure to be successful;

And the procedure ends.

Reference

3GPP TS 25.331 clause 8.1.3.

8.1.2.1.3 Test purpose

- 1. To confirm that the UE leaves the Idle Mode and correctly establishes signalling radio bearers on the DCCH.
- 2. To confirm that the UE indicates the requested UE radio access capabilities and UE system specific capabilities (may be used by UTRAN e.g. to configure inter RAT- measurements).

8.1.2.1.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE.

Test Procedure

The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by attempting to make an outgoing call. After SS receives this message, it assigns the necessary radio resources and U-RNTI to be used by the UE. SS then transmits an RRC CONNECTION SETUP message containing an IE "Initial UE Identity" that does not match the IE "Initial UE Identity" in the most recent RRC CONNECTION REQUEST message sent by the UE. UE receives the RRC CONNECTION SETUP message before timer T300 expires but discards it due to a IE "Initial UE Identity" mismatch. UE shall wait for timer T300 to time out before re-transmitting a RRC CONNECTION REQUEST message to the SS. SS again assigns the necessary radio resources and U-RNTI. SS then transmits a RRC CONNECTION SETUP message containing an IE "Initial UE Identity" that matches the IE "Initial UE Identity" in the most recent RRC CONNECTION REQUEST sent by the UE. SS then waits for the UE to transmit an RRC CONNECTION REQUEST sent by the UE. SS then waits for the UE to transmit an RRC CONNECTION SETUP COMPLETE message on the DCCH. SS calls for generic procedure C.3 to check that UE is in CELL DCH state.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	\rightarrow	RRC CONNECTION REQUEST	By outgoing call operation. See specific message contents.
2	÷	RRC CONNECTION SETUP	This message is not addressed to the UE. See specific message contents.
3	<i>→</i>	RRC CONNECTION REQUEST	UE shall re-transmit the request message again after a time out of T300 from step 1.
4	÷	RRC CONNECTION SETUP	See specific message contents.
5			The UE configures the layer 2 and layer 1.
6	\rightarrow	RRC CONNECTION SETUP COMPLETE	See specific message contents.
7	\leftrightarrow	CALL C.3	If the test result of C.3 indicates that UE is in CELL_DCH state, the test passes, otherwise it fails.

Specific Message Content

System Information Block type 11 (FDD)

Use the default system information block with the same type specified in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
- Intra-frequency reporting quantity for RACH Reporting	
- SFN-SFN observed time difference reporting	No report
indicator	
- CHOICE mode	
- FDD	
- Reporting quantity	CPICH Ec/N0
 Maximum number of reported cells on RACH 	current cell

RRC CONNECTION REQUEST (Step 1)

Use the default message with the same message type specified in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
Measured results on RACH	Check to see if set in accordance with the IE "Intra-
	frequency reporting quantity for RACH Reporting"
	included in SYSTEM INFORMATION BLOCK Type 11
 Measurement result for current cell 	
- CHOICE mode	
- FDD	
- CHOICE measurement quantity	
- CPICH Ec/N0	The actual reported value is not checked

RRC CONNECTION SETUP (Step 2)

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark
Initial UE Identity	Set to the same type as in the RRC CONNECTION REQUEST message but with a different value

RRC CONNECTION SETUP (Step 4)

Use the default message with the same message type and covering the scenario used in this test (Transition to CELL_DCH) specified in clause 9 of TS 34.108.

RRC CONNECTION SETUP COMPLETE (Step 6)

Use the default message with the same message type specified in clause 9 of TS 34.108 with the following exception.

Information Element	Value/remark
UE Radio Access Capability	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.
UE radio access capability extension	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.
UE system specific Capability	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.

8.1.2.1.5 Test requirement

After step 2 the UE shall re-transmit the RRC CONNECTION REQUEST message again in order to continue the RRC connection establishment procedure.

After step 6 the UE shall establish an RRC connection and continue the procedure of the outgoing call on the DCCH.

T1020786

T1S020869

3GPP TSG-T1 N Luton, UK, 4 th – 3GPP TSG-T1/S Luton, UK, 4 th –	8 th Nov 2002 IG Meeting #26	T1020786 T1S020869					
CHANGE REQUEST							
ж <mark>а</mark>	34.123-1 CR <mark>338</mark> ೫ rev 	urrent version: 5.1.1 [#]					
For <u>HELP</u> on using this form, see bottom of this page or look at the pop-up text over the % symbols. Proposed change affects: UICC apps% ME X Radio Access Network Core Network							
Title: #	CR to Package 1 TC 8.4.1.1: Correction from CRs a T1S020726/727 (revision to T1S020750, T1S020856						
Source: ೫	Panasonic						
Work item code: %	TEI	Date: ₩ 6/11/2002					
Category:		elease: % REL-5Use one of the following releases:2(GSM Phase 2)R96(Release 1996)R97(Release 1997)R98(Release 1998)R99(Release 1999)Rel-4(Release 4)Rel-5(Release 5)Rel-6(Release 6)					
Reason for change	 e: # 1. From CR1573 UE reports of "SFN-SFN observed time of and Inter-frequency measurement report point of view. 2. From CR1558 It's clarified that the UE behavior is unspect triggering condition "Active set cells" or "/ cells" for the intra-frequency events 1a of behavior is unspecified in Rel-99 UE whe other than "Active set cells" for the intra-f 3. From CR 1541 The handling of the "UE internal measure which the UE may receive in SIB11/SIB1 specifications. 4. Other technical corrections. The revision from T1S020724 is below. 5. To align with T1S020726/727 default SIB corrections 	is not needed from functional ecified in Rel-99 UE when using a Active set cells and monitored set r 1e. It's clarified that the UE en using a tiggering condition frequency events 1b or 1f. ement system information" IE 2 is currently unclear in the					

Summary of change: #	 Changes from RP17 1. IE "SFN-SFN observed time difference" and IE "SFN-SFN observed time difference reporting indicator" are removed. 	
	 Event 1f and 1b for monitored set cells are changed to event 1e and 1a for monitored set cells. Cell power settings are changed to trigger the new events. 	
	 IE "UE internal Measurement System Information" from IE" Measurement control system information" is removed. 	
	Other corrections4. In Specific Message Content (step 7), the primary scrambling code of intra-freq cell id 3 is corrected.	
	 In Specific Message Content (step 7), the IE "Time to trigger" and "Amount of reporting" are corrected. 	
	The revision from T1S020724 is below and shown with yellow marker.	
	 In expected sequence step 4, in order to align with the values in IE"Reference time difference to cell" of SIB11 and MEASUREMENT CONTROL messages, IE"Default DPCH Offset Value " and IE" DPCH frame offset " is set to 0. 	
	 7. In Specific Message Content SIB 11(step 1) is modified. Aligned with T1S020726/727 Set 1024 in IE"Reference time difference to cell" 	
	 8. In Specific Message Content MEASUREMENT CONTROL (step 7) is modified. Set 0 in IE"Reference time difference to cell" 	
	 In Table 8.4.1.1-1, correction to CPICH Ec value in cell 2 "T0" column is removed. It is reversed to -70dBm. In Specific Message Content, IE "Reporting Range" is set to 15dB in 	
	MEASUREMENT CONTROL (Step 12) in order for event 1a to be triggered.	
	 Revision to T1S020856 1. In the Expected Sequence, step 5 and 6 are modified. This is because in step 5, it is difficult to determine the arrival time of the first MEASUREMENT REPORT message at the SS. 2. In the Test Procedure, a statement is added to clarified that in the RADIO BEARER SETUP message in P11 or P13, the IE"Default DPCH Offset 	
	Value" and IE "DPCH frame offset" should be set to "0".	
Consequences if % not approved:	The test specifications are not aligned while the core specification	
Clauses affected: #	8.4.1.1	
	YN	

Other specs Affected:	ж	Χ	Other core specifications Test specifications O&M Specifications	ж	
Other comments:	ж	Affec	ts R99, REL-4, REL-5		

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

<Start of Modifications>

- 8.4.1.1 Measurement Control and Report: Intra-frequency measurement for transition from idle mode to CELL_DCH state
- 8.4.1.1.1 Definition

8.4.1.1.2 Conformance requirement

Upon transition from idle mode to CELL_DCH state, the UE shall:

- 1> begin or continue monitoring the list of cells assigned in the IE "intra-frequency cell info list" in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11);
- 1> if the "intra-frequency measurement reporting criteria" IE was included in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11):
 - 2> begin measurement reporting according to the IE.

In CELL_DCH state, the UE shall:

1> transmit a MEASUREMENT REPORT message on the uplink DCCH when the reporting criteria stored in variable MEASUREMENT_IDENTITY are met for any ongoing measurements that are being performed in the UE.

•••

The reporting criteria are fulfilled if either:

- the first measurement has been completed for a newly initiated measurement with periodic reporting; or
- the time period indicated in the stored IE "Periodical reporting criteria" has elapsed since the last measurement report was submitted to lower layers for a given measurement; or
- an event in stored IE "Measurement reporting criteria" was triggered.

For the measurement, which triggered the MEASUREMENT REPORT message, the UE shall:

- 1> set the IE "measurement identity" to the measurement identity, which is associated with that measurement in variable MEASUREMENT_IDENTITY;
- 1> set the IE "measured results" to include measurements according to the IE "reporting quantity" of that measurement stored in variable MEASUREMENT_IDENTITY; and
 - 2> if all the reporting quantities are set to "false":

3> not set the IE "measured results".

- 1> set the IE "Measured results" in the IE "Additional measured results" according to the IE "reporting quantity" for all measurements associated with the measurement identities included in the "Additional measurements list" stored in variable MEASUREMENT_IDENTITY of the measurement that triggered the measurement report; and
 - 2> if more than one additional measured results are to be included:
 - 3> sort them in ascending order according to their IE "measurement identity" in the MEASUREMENT REPORT message.
- 1> if the MEASUREMENT REPORT message was triggered by an event (i.e. not a periodical report):

•••

The UE shall:

Release 5

1> transmit the MEASUREMENT REPORT message on the uplink DCCH using either AM or UM RLC according to the stored IE "measurement reporting mode" associated with the measurement identity that triggered the report.

When the MEASUREMENT REPORT message has been submitted to lower layers for transmission:

1> the procedure ends.

•••

Upon reception of a MEASUREMENT CONTROL message the UE shall perform actions specified in TS 25.331 subclause 8.6 unless otherwise specified below.

The UE shall:

- 1> read the IE "Measurement command";
- 1> if the IE "Measurement command" has the value "setup":
 - 2> store this measurement in the variable MEASUREMENT_IDENTITY according to the IE "measurement identity", first releasing any previously stored measurement with that identity if that exists;
 - 2> for measurement types "inter-RAT measurement" or "inter-frequency measurement":

...

2> for measurement type "UE positioning measurement":

•••

- 2> for any other measurement type:
 - 3> if the measurement is valid in the current RRC state of the UE:
 - 4> begin measurements according to the stored control information for this measurement identity.
- 1> if the IE "Measurement command" has the value "modify":
 - 2> for all IEs present in the MEASUREMENT CONTROL message:
 - 3> if a measurement was stored in the variable MEASUREMENT_IDENTITY associated to the identity by the IE "measurement identity":
 - 4> for measurement types "inter-frequency measurement" that require measurements on a frequency other than the actually used frequency, or that require measurements on another RAT:

•••

- 4> for any other measurement type:
 - 5> replace the corresponding information stored in variable MEASUREMENT_IDENTITY associated to the identity indicated by the IE "measurement identity" with the one received in the MEASUREMENT CONTROL message;
 - 5> resume the measurements according to the new stored measurement control information.
- 3> otherwise:
 - 4> set the variable CONFIGURATION_INCOMPLETE to TRUE.
- 2> for all optional IEs that are not present in the MEASUREMENT CONTROL message:
 - 3> leave the currently stored information elements unchanged in the variable MEASUREMENT_IDENTITY if not stated otherwise for that IE.
- 1> if the IE "measurement command" has the value "release":

•••

1> clear the entry for the MEASUREMENT CONTROL message in the table "Accepted transactions" in the variable TRANSACTIONS;

Reference

3GPP TS 25.331 clause 8.4.1.8.1, 8.4.1.3, 8.4.2.2.

8.4.1.1.3 Test Purpose

- 1. To confirm that the UE continues to monitor intra-frequency measurement quantity of the cells listed in System Information Block type 11 or 12 messages, after it has entered CELL_DCH state from idle mode. When the intra-frequency measurement reporting criteria specified in System Information Block type 11 or 12 messages have been met, it shall report the measurements using MEASUREMENT REPORT message(s).
- 2. To confirm that the UE terminates monitoring and reporting activities for the cells listed in "intra-frequency cell info list" IE in System Information Block type 11 or 12 messages, after it has received a MEASUREMENT CONTROL message that specifies the measurement type to be "intra-frequency measurement" with the same measurement identity as in System Information Block Type 11 or 12 messages. To confirm that the UE reconfigures the monitoring and reporting activities based on the last MEASUREMENT CONTROL message received.

8.4.1.1.4 Method of test

Initial Condition

System Simulator: 3 cells – Cell 1, Cell 2 and Cell 3 are active.

UE: "Registered idle mode on CS" (state 2) or "Registered idle mode on PS" (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE If the UE supports both CS and PS domains, the initial UE state shall be "Registered idle mode on CS/PS" (state 7).

Test Procedure

Table 8.4.1.1-1 illustrates the downlink power to be applied for the 3 cells at various time instants of the test execution. Column marked "T0" denotes the initial conditions, while columns marked "T1" are to be applied subsequently. The exact instants on which these values shall be applied are described in the texts in this clause.

Parameter	Unit	Cell 1			Cell 2		Cell 3			
		T0	T1	T2	T0	T1	T2	T0	T1	T2
UTRA RF Channel Number			Ch. 1			Ch. 1			Ch.1	
CPICH Ec	dBm/3.84 MHz	-60	-60	-60	<mark>-670</mark> -70	- <u>6</u> 80	- <mark>8</mark> 60	- <u>80</u> 60	- <u>6</u> 80	- <mark>6</mark> 80

Table 8.4.1.1-1

The UE is initially in idle mode and has selected cell 1 for camping. The System Information Block type 11 messages are modified with respect to the default settings to prevent reporting of "Cell synchronisation information" and also to include cell 2 into the monitored neighbour cell list. The key measurement parameters in the modified System Information Block message are as follow: measurement type = "intra-frequency measurement", measurement quantity = "CPICH RSCP", report criteria = "periodic reporting criteria", reporting interval = "64 seconds".

SS prompts the operator to make an outgoing call of a supported traffic class. SS and UE shall execute procedure P3 (for CS service) or P5 (for PS service). Next SS and UE shall execute procedure P7 (for CS service) or P9 (for PS service). Then SS and UE shall execute procedure P11 (for CS service) or P13 (for PS service). The UE shall send a MEASUREMENT REPORT message after reaching CELL_DCH state, reporting cell 2's CPICH RSCP value. After 64 seconds has passed since SS receives the first MEASUREMENT REPORT message, the UE shall transmit a second MEASUREMENT REPORT message.

Note: In P11 or P13 in step 4, in RADIO BEARER SETUP message. IE "Default DPCH Offset Value" and IE "DPCH frame offset" are set to "0".

SS sends a MEASUREMENT CONTROL message on the downlink DCCH. In this message, SS configures an intrafrequency measurement based on the measurement quantity CPICH RSCP. Parameters used in this message are: measurement identity = "1", report criteria = "event-trigger", event identity = "1 \underline{e} f", reporting threshold = "-70 dBm". SS checks to see that no MEASUREMENT REPORT messages are sent within the next 64 seconds (which is due to periodic reporting). SS reconfigures the downlink transmission power settings according to values in column "T1" in table 8.4.1.1-1. The UE shall transmit a MEASUREMENT REPORT message when it detects that the CPICH RSCP of cell 3 has <u>risendropped above</u> below the threshold value specified in the previous MEASUREMENT CONTROL message.

SS sends then a new MEASUREMENT CONTROL message to add cells 1 and 2 to the list of the cells the UE shall measure. Since the RSCP for cell 2 is <u>above</u> the threshold for event 1 ef to be triggered, a MEASUREMENT REPORT triggered by cell 2 shall be sent by the UE.

SS reconfigures the downlink transmission power settings according to values in column "T2" in table 8.4.1.1-1. SS sends a new MEASUREMENT CONTROL message on the downlink DCCH. In this message, SS configures an intrafrequency measurement based on the measurement quantity CPICH RSCP. Parameters used in this message are: measurement identity = "1", report criteria = "event-trigger", event identity = "1aB", Reporting range 8db. SS reconfigures the downlink transmission power settings according to values in column "T0" in table 8.4.1.1-1. The UE shall transmit a MEASUREMENT REPORT message when it detects that the condition for event 1ab is fulfilled. SS calls for generic procedure C.3 to check that UE is in CELL_DCH state.

Expected Sequence

Step	Direc	tion	Message	Comment
-	UE	SS	-	
1	÷		System Information Block type 11	The UE is in idle mode and camped onto cell 1. The System Information Block type 11 messages to be transmitted are different from the default settings (see specific message contents)
2	\leftrightarrow	>	SS executes procedure P3 (clause 7.4.2.1.2) or P5 (clause 7.4.2.2.2) specified in TS 34.108.	
3	$\stackrel{(+)}{\leftarrow}$	>	SS executes procedure P7 (clause 7.4.2.3.2) or P9 (clause 7.4.2.4.2) specified in TS 34.108.	
4	\leftrightarrow	>	SS executes procedure P11 (clause 7.4.2.5.2) or P13 (clause 7.4.2.6.2) specified in TS 34.108.	IE"Default DPCH Offset Value " and IE" DPCH frame offset " in RADIO BEARER SETUP message is set to "0".
5	<mark>→</mark> 9	S	Void	SS shall wait for a MEASUREMENT REPORT message
6	\rightarrow	•	MEASUREMENT REPORT	After receiving this message, SS shall expect to receive the next MEASUREMENT REPORT message after waits 64 seconds.
6a	\rightarrow		MEASUREMENT REPORT	SS shall receive consecutive MEASUREMENT REPORT messages at 64 seconds interval.
7	÷	-	MEASUREMENT CONTROL	A measurement with "measurement identity" IE set to "1" is assigned, with the IE "CHOICE reporting criteria" set to "intra-frequency measurement reporting criteria". See specific message content for the rest of the message.

Step	Direction	Message	Comment
-	UE SS		
8			SS waits for 64 seconds and verifies that no further MEASUREMENT REPORT
			messages are detected on the uplink DCCH.
9			SS re-adjusts the downlink transmission power settings according to columns "T1" in table 8.4.1.1-1.
10	\rightarrow	MEASUREMENT REPORT	SS verifies that UE transmits a MEASUREMENT REPORT message triggered by cell 3 and containing report the measured CPICH RSCP value of cell 3.
10a	÷	MEASUREMENT CONTROL	A MEASUREMENT CONTROL is sent to the UE to modify the list of the cells the UE shall monitor.
10b	\rightarrow	MEASUREMENT REPORT	SS verifies that UE transmits a MEASUREMENT REPORT message triggered by cell 2.
11			SS re-adjusts the downlink transmission power settings according to columns "T2" in table 8.4.1.1-2.
12	÷	MEASUREMENT CONTROL	A measurement with "measurement identity" IE set to "1" is assigned, with the IE "CHOICE reporting criteria" set to "intra-frequency measurement reporting criteria". See specific message content for the rest of the message.
13			SS re-adjusts the downlink transmission power settings according to columns "T0" in table 8.4.1.1-3 and waits 5 seconds.
14	<i>→</i>	MEASUREMENT REPORT	SS verifies that UE transmits a MEASUREMENT REPORT message to report occurrence of event 1ab.
15	\leftrightarrow	CALL C.3	If the test result of C.3 indicates that UE is in CELL_DCH state, the test passes, otherwise it fails.

Specific Message Contents

All messages indicated below shall use the same content as described in default message content, with the following exceptions:

System Information Block type 11 (Step 1)

Information Element	Value/remark
SIB12 indicator	FALSE
FACH measurement occasion info	Not Present
Measurement control system information	
- Use of HCS	Not used
- Cell selection and reselection quality measure	CPICH RSCP
 Intra-frequency measurement system information Intra-frequency measurement identity 	Not Present
- Intra-frequency measurement identity	Absence of this IE is equivalent to default value 14
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Not present
	(This IE shall be ignored by the UE for SIB11)Remove
	no intra-frequency cells
- New intra-frequency cells	1
- Intra-frequency cell id - Cell info	
- Cell individual offset	Not present
	Absence of this IE is equivalent to default value 0dB0
	<mark>d₿</mark>
- Reference time difference to cell	Not Present
- Read SFN Indicator	TRUE
- CHOICE Mode - Primary CPICH Info	FDD
- Primary Scrambling Code	Refer to clause titled "Default settings for cell No.1
	(FDD)" in clause 6.1.4 of TS34.108 Set to same code as
	used for cell 1
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection	Not Present <u>(The IE shall be absent as this is the serving</u> cell)
- Intra-frequency cell id	2
- Cell info	2
- Cell individual offset	Not present
	Absence of this IE is equivalent to default value 0dB0
Reference time difference to cell Read SFN Indicator	2 56 chips<u>1024</u> TRUE
- CHOICE Mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Refer to clause titled "Default settings for cell No.2
	(FDD)" in clause 6.1.4 of TS34.108Set to same code as
	used for cell 2
 Primary CPICH TX power TX Diversity Indicator 	Not Present FALSE
- Cell selection and Re-selection info	Not present
	For neigbouring cell, if HCS is not used and all the
	parameters in cell selection and re-selection info are
	Default value, this IE is absent.
	Not Present (Default is 0dB) Not Present
	HOLFISSIN 133dBm
- HCS neighbouring cell information	Not Present
	FDD
	-20dB
	-115dBm
 Intra-frequency measurement quantity Filter Coefficient 	Not Procent (Default is 0)
- CHOICE Mode	Not Present (Default is 0) FDD
- Measurement quantity	CPICH RSCP
- Intra-frequency measurement for RACH reporting	Not Present
- Maximum number of reported cells on RACH	Not Present
- Reporting information for state CELL_DCH	
- Intra-frequency reporting quantity	
 Reporting quantities for active set cells SFN-SFN observed time difference reporting 	No report
indicator	

- Cell synchronisation information reporting	FALSE
indicator	
 Cell identity reporting indicator 	FALSE
- CHOICE mode	FDD
- CPICH Ec/No reporting indicator	FALSE
 CPICH RSCP reporting indicator 	FALSE
- Pathloss reporting indicator	FALSE
 Reporting quantities for monitored set cells 	
- SFN-SFN observed time difference reporting	No report
indicator	
 Cell synchronisation information reporting 	FALSE
indicator	
- Cell identity reporting indicator	FALSE
- CHOICE mode	FDD
 CPICH Ec/No reporting indicator 	FALSE
 CPICH RSCP reporting indicator 	TRUE
 Pathloss reporting indicator 	FALSE
 Reporting quantities for detected cells 	Not present
 Measurement Reporting Mode 	
 Measurement Report Transfer Mode 	Acknowledged mode RLC
 Periodical Reporting / Event Trigger Reporting 	Periodical reporting
Mode	
- CHOICE report criteria	Periodic reporting criteria
 Amount of reporting 	Infinity
- Reporting interval	64 seconds
 Inter-frequency measurement system information 	Not present
 Inter-RAT measurement system information 	Not Present
- Traffic volume measurement system information	Not Present
—- UE internal measurement system information	Not Present

MEASUREMENT REPORT (Step 6 and 6a)

Information Element	Value/remark
Measurement identity	Check to see if set to 1
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measurement results 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
- SFN-SFN observed time difference	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	"Checked to see if set to within an acceptable range"
- Pathloss	Check to see if this IE is present
Measured Results on RACH	Check to see if this IE is absent
Additional Measured results	Check to see if this IE is absent
Event Results	Check to see if this IE is absent

1

MEASUREMENT CONTROL (Step 7)

Information Element	Value/remark
Measurement Identity Measurement Command	1 Sotup
Measurement Command Measurement Reporting Mode	Setup
- Measurement Reporting Transfer Mode	Acknowledged Mode RLC
- Periodic Reporting / Event Trigger Reporting Mode	Event Trigger
Additional measurements list	Not Present
CHOICE measurement type	Intra-frequency measurement
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Remove all intra-frequency cells
 New intra-frequency cells 	2 new intra-frequency cells
- Intra-frequency cell id	3
- Cell info	
- Cell individual offset	0 dB
 Reference time difference to cell Read SFN Indicator 	<mark>256-0</mark> chips TRUE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 32
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Intra-frequency cell id	1
- Cell info	
- Cell individual offset	0 dB
 Reference time difference to cell 	Not Present
- Read SFN Indicator	TRUE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1 Not Present
 Primary CPICH TX power TX Diversity Indicator 	FALSE
- Cell for measurement	Not Present
- Intra-frequency measurement quantity	
- Filter Coefficient	Not Present (Default is 0)
- CHOICE Mode	FDD
- Measurement quantity	CPICH RSCP
 Intra-frequency reporting quantity 	
 Reporting quantities for active set cells 	
- SFN-SFN observed time difference reporting	No report
indicator	
- Cell synchronisation information reporting indicator	FALSE
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	TRUE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
- Reporting quantities for monitored set cells	
- SFN-SFN observed time difference reporting	No report
indicator	
- Cell synchronisation information reporting	TRUE
indicator	
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	
 Pathloss reporting indicator Reporting quantities for detected cells 	FALSE Not present
- Reporting quantities for detected cens	Not Present
- Measurement validity	Not present
- CHOICE report criteria	Intra-frequency measurement reporting criteria
- Parameters required for each events	
- Intra-frequency event identity	<u>1e</u>
- Triggering condition 1	4fNot present
- Triggering condition 2	Monitored set cells
- Reporting range	Not Present
- Cells forbidden to affect reporting range	Not Present
- W	Not Present

 Hysteresis Threshold used frequency Reporting deactivation threshold Replacement activation threshold Time to trigger Amount of reporting Reporting interval Reporting cell status CHOICE reported cell 	1 dBNot Present -70dBm1 dB Not Present-70 dBm Not Present 0 msecNot Present 0 msecInfinity Not Present Not Present Not Present Report cells within active and/or monitored set on used
	frequency or within active and/or monitored set on non- used frequency
 Maximum number of reported cells 	3
DPCH compressed mode status info	Not Present

MEASUREMENT REPORT (Step 10)

Information Element	Value/remark
Measurement identity	Check to see if set to 1
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
- Intra-frequency measurement results	Check to see if measurement results for 2 cells are included (the order in which the different cells are reported is not important)
 Cell measured results 	(for cell 1)
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
 Cell synchronisation information Primary CPICH Info 	Check to see if this IE is absent
- Primary Scrambling Code	Check to see if it's the same code for cell 1
- CPICH Ec/No	Check to see if this IE is present
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	(for cell 3)
- Cell Identity	Check to see if it is absent
- SFN-SFN observed time difference	Check to see if this IE is absent
- Cell synchronisation information	Check to see if this IE is present and that the COUNT-
	C-SFN frame difference is included in it.
- Primary CPICH Info	
 Primary Scrambling Code 	Check to see if it's the same code for cell 3
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Additional Measured Results Event Results	Check to see if this IE is absent
- CHOICE event result	Check to see if this IE is set to "Intra-frequency
	measurement event results"
- Intra-frequency event identity	Check to see if this IE is set to "1 <u>ef</u> "
- Cell measured event results	
- CHOICE mode	Check to see if this IE is set to "FDD"
- Primary CPICH info	
- Primary Scrambling Code	Check to see if it's the same code for cell 3

MEASUREMENT CONTROL (Step 10a)

Information Element	Value/remark
Measurement Identity	1
Measurement Command	Modify
Measurement Reporting Mode	Not Present
Additional measurements list	Not Present
CHOICE measurement type	Intra-frequency measurement
 Intra-frequency cell info list 	
 CHOICE intra-frequency cell removal 	Remove no intra-frequency cells
 New intra-frequency info list 	1 new intra-frequency cells
 Intra-frequency cell id 	2
- Cell info	
 Cell individual offset 	0 dB
 Reference time difference to cell 	Not Present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 2
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell for measurement	Not Present
 Intra-frequency measurement quantity 	Not Present
 Intra-frequency reporting quantity 	Not Present
- Reporting cell status	Not Present
 Measurement validity 	Not Present
- CHOICE report criteria	Not Present

l

l

MEASUREMENT REPORT (Step 10b)

Information Element	Value/remark
Measurement identity	Check to see if set to 1
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
- Intra-frequency measurement results	Check to see if measurement results for 3 cells are included (the order in which the different cells are reported is not important)
- Cell measured results	(for cell 1)
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
 Cell synchronisation information Primary CPICH Info 	Check to see if this IE is absent
- Primary Scrambling Code	Check to see if it's the same code for cell 1
- CPICH Ec/No	Check to see if this IE is present
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	(for cell 2)
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
- Cell synchronisation information	Check to see if this IE is present and that the COUNT-
	C-SFN frame difference is included in it.
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	(for cell 3)
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
- Cell synchronisation information	Check to see if this IE is present and that the COUNT-
	C-SFN frame difference is included in it.
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 3
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Additional Measured Results	Check to see if this IE is absent
Event Results	
- CHOICE event result	Check to see if this IE is set to "Intra-frequency
	measurement event results"
 Intra-frequency event identity 	Check to see if this IE is set to "1 <u>ef</u> "
- Cell measured event results	
- CHOICE mode	Check to see if this IE is set to "FDD"
- Primary CPICH info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2

MEASUREMENT CONTROL (Step 12)

Information Element	Value/remark
Measurement Identity	
Measurement Command	Setup
Measurement Reporting Mode	
- Measurement Reporting Transfer Mode	Acknowledged Mode RLC
- Periodic Reporting / Event Trigger Reporting Mode	Event Trigger
Additional measurements list CHOICE measurement type	Not Present Intra-frequency measurement
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Remove all intra-frequency cells
- New intra-frequency cells	2 new intra-frequency cells
- Intra-frequency cell id	1
- Cell info	
 Cell individual offset Reference time difference to cell 	0 dB Not Present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
 Intra-frequency cell id Cell info 	2
- Cell individual offset	0 dB
- Reference time difference to cell	256 chips
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 2
 Primary CPICH TX power TX Diversity Indicator 	Not Present FALSE
- Cells for measurement	Not Present
- Intra-frequency cell id	
- Intra-frequency measurement quantity	
- Filter Coefficient	Not Present (Default is 0)
- Measurement quantity	CPICH RSCP
 Intra-frequency reporting quantity Reporting quantities for active set cells 	
- SFN-SFN observed time difference reporting	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator	
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator - Pathloss reporting indicator	FALSE FALSE
- Reporting quantities for monitored set cells	
- SFN-SFN observed time difference reporting	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator	
- Cell identity reporting indicator	FALSE FALSE
- CPICH Ec/No reporting indicator - CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- Reporting quantities for detected cells	Not present
- Reporting cell status	Not Present
- Measurement validity	Not present
- CHOICE report criteria	Intra-frequency measurement reporting criteria
- Parameters required for each events	1.0
 Intra-frequency event identity Triggering condition 1 	1 <u>ab</u> Not presentMonitored Cells
- Triggering condition 2	Monitored set cells
- Reporting range	158 dB
- Cells forbidden to affect reporting range	Not Present
- W	0

- Hysteresis	0 dB
- Threshold used frequency	Not Present
 Reporting deactivation threshold 	1Not Present
- Replacement activation threshold	Not Present
- Time to trigger	5000 msec
- Amount of reporting	Infinity Not Present
- Reporting interval	64secNot Present
- Reporting cell status	Not Present
DPCH compressed mode status info	Not Present

MEASUREMENT REPORT (Step 14)

Information Element	Value/remark
Measurement identity	Check to see if set to 1
Measured Results	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Additional Measured Results	Check to see if this IE is absent
Event Results	
 CHOICE event result 	
 Intra-frequency event identity 	Check to see if this IE is set to "1ab"
- Cell measured event results	
- CHOICE mode	Check to see if this IE is set to "FDD"
- Primary CPICH info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2

8.4.1.1.5 Test Requirement

After step 5 the UE shall start to transmit 2 MEASUREMENT REPORT messages at 64 seconds interval. The measurement quantity "CPICH RSCP" of cell 2 shall be reported in these messages.

After step 7 the UE shall not transmit any MEASUREMENT REPORT messages within 64 seconds after SS has transmitted the MEASUREMENT CONTROL message in step 7.

After step 9 the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH, to report that the CPICH RSCP value for cell <u>3</u>² has <u>risen above</u><u>dropped below</u> the threshold stated in the MEASUREMENT CONTROL message transmitted by the SS in step 7. This MEASUREMENT REPORT message shall also contain IE "Event results", indicating the triggering of event 'lef' by cell 3. It shall also contain the measured CPICH RSCP value and cell synchronisation information for cell 3, and the measured CPICH Ec/No and RSCP values for cell 1.

After step 10a, the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH to report that the CPICH RSCP value for cell 2 has <u>risen above</u> dropped below the threshold stated in the MEASUREMENT CONTROL message transmitted by the SS in step <u>10a</u>⁷. The MEASUREMENT REPORT message shall contain the measured CPICH RSCP value and cell synchronisation information for cell 2 and cell 3, as well as the measured CPICH Ec/No and RSCP for cell 1. The IE "Event results" in this message shall indicate that cell 2 has triggered the event.

After step 13, the UE shall transmit a MEASUREMENT REPORT message containing IE "Event results", indicating the triggering of event '1ab' by cell 2. The MEASUREMENT REPORT message shall not contain any measured results.

<End of Modifications>

3GPP TSG-T1 M Luton, UK, 4 th – 3GPP TSG-T1/S Luton, UK, 4 th –	8 th Nov 2002 G Meeting #26 T1S0207	T1-020787 T1S020735	
		m-v7	
^ж 3	4.123-1 CR 339 * rev - ^{* Current version:} 5.1.1 [*]		
For <u>HELP</u> on u Proposed change	sing this form, see bottom of this page or look at the pop-up text over the of symbols.		
Title: #	CR to 34.123-1 clause 8.2 (Package 1) Rel-5: Correction from CRs approved in RP17meeting		
Source: अ	Panasonic		
Work item code: ೫	TEI Date: # 28/10/2002		
Category: ₩	FRelease: \$ REL-5Use one of the following categories:Use one of the following releases:F (correction)2A (corresponds to a correction in an earlier release)R96B (addition of feature),R97C (functional modification of feature)R98D (editorial modification)R99D (editorial modification)R99D tetailed explanations of the above categories canRel-4be found in 3GPP TR 21.900.Rel-5Rel-6(Release 6)		
Reason for change	: # From CR1680 The use of synchronization procedures A and B is clarified wherever applicable	-	
Summary of chang	e: # The conformance requirement was revised.		
Consequences if not approved:	# The test specifications are not aligned with the core specification		
Clauses affected:	8.2.1.1, 8.2.1.10, 8.2.3.7, 8.2.3.8, 8.2.3.9, 8.2.3.15, 8.2.3.18, 8.2.3.19		
Other specs Affected:	Y N % X Other core specifications % X Test specifications X O&M Specifications		
Other comments:	# Affects R99, REL-4, REL-5		

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.

- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

<End of Modifications>

- 8.2.1.1 Radio Bearer Establishment for transition from CELL_DCH to CELL_DCH: Success
- 8.2.1.1.1 Definition

8.2.1.1.2 Conformance requirement

If the UE receives:

- a RADIO BEARER SETUP message; or

it shall:

- 1> if the UE will enter the CELL_DCH state from any state other than CELL_DCH state at the conclusion of this procedure:
 - $_$ perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If the UE was in CELL_DCH state upon reception of the reconfiguration message and remains in CELL_DCH state, the UE shall:

- 1> if the IE "Uplink DPCH Info" is absent, not change its current UL Physical channel configuration;
- 1> if the IE "Downlink information for each radio link" is absent, not change its current DL Physical channel configuration.

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

- 1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER SETUP message, the UE shall:

1> transmit a RADIO BEARER SETUP COMPLETE as response message on the uplink DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4.

8.2.1.1.3 Test purpose

To confirm that the UE establishes a new radio bearer according to a RADIO BEARER SETUP message.

8.2.1.1.4 Method of test

Initial Condition

System Simulator: 1 cell

UE: CS-DCCH_DCH (state 6-5) or PS_DCCH_DCH (state 6-7) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE.

Test Procedure

The UE is in CELL_DCH state, after the test operator is prompted to make an out-going call. Before step 1, only signalling radio bearers have been established. The SS transmits a RADIO BEARER SETUP message to the UE. This message requests the establishment of radio access bearer. After the UE receives this message, it configures them and establishes a radio access bearer. Finally the UE transmits a RADIO BEARER SETUP COMPLETE message using AM RLC. SS calls for generic procedure C.3 to check that UE is in CELL_DCH state.

Expected sequence

Step	Direc	ction	Message	Comment
	UE	SS		
1	€	-	RADIO BEARER SETUP	
2	-	>	RADIO BEARER SETUP COMPLETE	
3	$3 \leftrightarrow$		CALL C.3	If the test result of C.3 indicates that UE is in CELL_DCH state, the test passes, otherwise it fails.

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is identical to the message sub-type indicated by "Non speech from CELL_DCH to CELL_DCH in CS" or "Speech from CELL_DCH to CELL_DCH in CS" or "Packet to CELL_DCH from CELL_DCH in PS" in [9] TS 34.108 clause 9.

8.2.1.1.5 Test requirement

After step 1 the UE shall transmit a RADIO BEARER SETUP COMPLETE message.

<End of Modifications>

<Start of Modifications>

- 8.2.1.10 Radio Bearer Establishment for transition from CELL_FACH to CELL_DCH: Success
- 8.2.1.10.1 Definition
- 8.2.1.10.2 Conformance requirement

If the UE receives:

- a RADIO BEARER SETUP message; or

it shall:

- 1> if the UE will enter the CELL DCH state from any state other than CELL DCH state at the conclusion of this procedure:
 - 2+ perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If after state transition the UE enters CELL_DCH state, the UE shall, after the state transition:

- 1> remove any C-RNTI from MAC;
- 1> clear the variable C_RNTI.

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

- 1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER SETUP message, the UE shall:

1> transmit a RADIO BEARER SETUP COMPLETE as response message on the uplink DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4.

8.2.1.10.3 Test purpose

To confirm that the UE establishes a new radio bearer according to a RADIO BEARER SETUP message.

8.2.1.10.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: PS-DCCH_FACH (state 6-8) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is in CELL_FACH state, after SS prompts the test operator to initiate a packet-switched data call. The SS transmits a RADIO BEARER SETUP message to the UE. After the UE receives this message, it configures them and establishes the required radio bearers. Finally the UE transmits a RADIO BEARER SETUP COMPLETE message using AM RLC. SS calls for generic procedure C.3 to check that UE is in CELL_DCH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	€		RADIO BEARER SETUP	
2	-	>	RADIO BEARER SETUP COMPLETE	
3	÷	→	CALL C.3	If the test result of C.3 indicates that UE is in CELL_DCH state, the test passes, otherwise it fails.

Specific Message Contents

RADIO BEARER SETUP

The contents of RADIO BEARER SETUP message in this test case is identical the message sub-type indicated by "Packet to CELL_DCH from CELL_FACH in PS" in [9] TS 34.108 clause 9.

8.2.1.10.5 Test requirement

After step 1 the UE shall transmit a RADIO BEARER SETUP COMPLETE message on the DCCH using AM RLC.

<End of Modifications>

<Start of Modifications>

- 8.2.3.7 Radio Bearer Release for transition from CELL_DCH to CELL_FACH: Success
- 8.2.3.7.1 Definition
- 8.2.3.7.2 Conformance requirement

If the UE receives:

- a RADIO BEARER RELEASE message; or

it shall:

1> if the UE will enter the CELL_DCH state from any state other than CELL_DCH state at the conclusion of this procedure:

- $\underline{24}$ > perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If after state transition the UE enters CELL_FACH state, the UE shall, after the state transition:

1> if the IE "Frequency info" is included in the received reconfiguration message:

2> select a suitable UTRA cell according to TS 25.304 on that frequency.

1> if the IE "Frequency info" is not included in the received reconfiguration message:

2> select a suitable UTRA cell according to TS 25.304.

1> if the received reconfiguration message included the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD), and the UE selects another cell than indicated by this IE or the received reconfiguration message did not include the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD):

2> initiate a cell update procedure according to TS 25.331 subclause 8.3.1 using the cause "Cell reselection";

2> when the cell update procedure completed successfully:

3> if the UE is in CELL_PCH or URA_PCH state:

- 4> initiate a cell update procedure according to TS 25.331 subclause 8.3.1 using the cause "Uplink data transmission";
- 4> proceed as below.
- 1> select PRACH according to TS 25.331 subclause 8.5.17;
- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> if the contents of the variable C_RNTI is empty:
 - 2> perform a cell update procedure according to TS 25.331 subclause 8.3.1 using the cause "Cell reselection";
 - 2> when the cell update procedure completed successfully:
 - 3> if the UE is in CELL_PCH or URA_PCH state:
 - 4> initiate a cell update procedure according to TS 25.331 subclause 8.3.1 using the cause "Uplink data transmission";
 - 4> proceed as below.

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

- 1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER RELEASE message, the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message on the uplink DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4.

8.2.3.7.3 Test purpose

To confirm that the UE release the existing the radio bearer according to a RADIO BEARER RELEASE message.

8.2.3.7.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: PS-DTCH+DCCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is in CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message to the UE. The UE releases the radio access bearer and transmits a RADIO BEARER RELEASE COMPLETE message on the uplink DCCH using AM RLC. SS calls for generic procedure C.2 to check that UE is in CELL_FACH state.

Expected sequence

Step	Direc	ction	Message	Comment
	UE	SS		
1	¥	<u>-</u>	RADIO BEARER RELEASE	SS releases the radio bearer in the fashion specified in the message and allocate common channel resources to carry the remaining radio bearers.
2				The UE selects PRACH and S- CCPCH indicated in SIB5 and SIB6 after entering CELL FACH state. The UE shall release dedicated channels, and reconfigure the remaining radio bearers using the common channel.
3	1.	>	RADIO BEARER RELEASE COMPLETE	
4	¢	\rightarrow	CALL C.2	If the test result of C.2 indicates that UE is in CELL_FACH state, the test passes, otherwise it fails.

Specific Message Contents

RADIO BEARER RELEASE (Step 1)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in [9] TS 34.108 clause 9 with the following exception:

Information Element	Value/remark
New C-RNTI	0000 0000 0000 0001B

8.2.3.7.5 Test requirement

After step 2 the UE shall transmit a RADIO BEARER RELEASE COMPLETE message.

- 8.2.3.8 Radio Bearer Release for transition from CELL_DCH to CELL_FACH: Success (Cell re-selection)
- 8.2.3.8.1 Definition
- 8.2.3.8.2 Conformance requirement

If the UE receives:

- a RADIO BEARER RELEASE message; or

it shall:

1> if the UE will enter the CELL DCH state from any state other than CELL DCH state at the conclusion of this procedure:

<u>24></u> perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;

1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If after state transition the UE enters CELL_FACH state, the UE shall, after the state transition:

1> if the IE "Frequency info" is included in the received reconfiguration message:

2> select a suitable UTRA cell according to TS 25.304 on that frequency.

1> if the IE "Frequency info" is not included in the received reconfiguration message:

2> select a suitable UTRA cell according to TS 25.304

1> if the received reconfiguration message included the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD), and the UE selects another cell than indicated by this IE or the received reconfiguration message did not include the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD):

2> initiate a cell update procedure according to TS 25.331 subclause 8.3.1 using the cause "Cell reselection";

- 1> select PRACH according to TS 25.331 subclause 8.5.17;
- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> if the contents of the variable C_RNTI is empty:

2> perform a cell update procedure according to TS 25.331 subclause 8.3.1 using the cause "Cell reselection";

- 2> when the cell update procedure completed successfully:
 - 3> if the UE is in CELL_PCH or URA_PCH state:
 - 4> initiate a cell update procedure according to TS 25.331 subclause 8.3.1 using the cause "Uplink data transmission";

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- includes "CN information elements"; or
- includes the IE "Ciphering mode info"; or
- includes the IE "Integrity protection mode info"; or
- includes the IE "New C-RNTI"; or
- includes the IE "New U-RNTI":

the UE shall:

1> transmit a UTRAN MOBILITY INFORMATION CONFIRM as response message using AM RLC.

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

- 1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER RELEASE message, the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message on the uplink DCCH using AM RLC. Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4, 8.3.1.7.

8.2.3.8.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RELEASE COMPLETE message after the UE completes a cell update procedure.

8.2.3.8.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is in CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message to request the UE to transit from CELL_DCH to CELL_FACH. The UE initiates the cell update procedure because the UE cannot detect the specified cell in this message. The UE transmits a CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". The SS shall transmit a CELL UPDATE CONFIRM message on downlink CCCH after receiving CELL UPDATE message. The UE then transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the DCCH using AM RLC. The UE transmits a RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC. SS calls for generic procedure C.2 to check that UE is in CELL_FACH state.

Expected sequence

Step	Dire	ction	Message	Comment
-	UE	SS		
1			Void	
2			Void	
3	•	.	RADIO BEARER RELEASE	Assigned the transition from CELL_DCH to CELL_FACH
4	-	>	CELL UPDATE	The value "cell reselection" shall be set in IE "cell update cause".
5	•	÷	CELL UPDATE CONFIRM	See message content.
6	\rightarrow		UTRAN MOBILITY INFORMATION CONFIRM	
7	\rightarrow		RADIO BEARER RELEASE COMPLETE	
8		\rightarrow	CALL C.2	If the test result of C.2 indicates that UE is in CELL_FACH state, the test passes, otherwise it fails.

Specific Message Contents

RADIO BEARER RELEASE (Step 3) (FDD)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in [9] TS 34.108 clause 9 with the following exceptions:

Information Element	Value/remark
Downlink information for each radio links	
- Primary CPICH info	
- Primary scrambling code	150

RADIO BEARER RELEASE (Step 3) (TDD)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in [9] TS 34.108 clause 9 with the following exceptions:

Information Element	Value/remark
Downlink information for each radio links	
- Primary CCPCH info	
- Cell parameters ID	4

CELL UPDATE (Step 4)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in [9] TS 34.108 clause 9 with the following exceptions:

Information Element	Value/remark
Cell Update Cause	"cell reselection"

CELL UPDATE CONFIRM (Step 5)

Use the same message type found in clause [9] TS 34.108 clause 9, with the following exceptions.

Info	ormation Element	Value/Remarks
New C-RNTI		'1010 1010 1010 1010'

UTRAN MOBILITY INFORMATION CONFIRM (Step 6)

Only the message type is checked.

8.2.3.8.5 Test requirement

After step 3 the UE shall transmit a CELL UPDATE message on the CCCH with IE "Cell update cause" set to "cell reselection".

After step 5 UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the DCCH using AM RLC.

After step 6 UE shall transmit a RADIO BEARER RELEASE COMPLETE message on the DCCH using AM RLC.

- 8.2.3.9 Radio Bearer Release for transition from CELL_FACH to CELL_DCH: Success
- 8.2.3.9.1 Definition
- 8.2.3.9.2 Conformance requirement

If the UE receives:

- a RADIO BEARER RELEASE message; or

it shall:

- 1> if the UE will enter the CELL DCH state from any state other than CELL DCH state at the conclusion of this procedure:
 - $\underline{24}$ > perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If after state transition the UE enters CELL_DCH state, the UE shall, after the state transition:

- 1> remove any C-RNTI from MAC;
- 1> clear the variable C_RNTI.

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

- 1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER RELEASE message, the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message on the uplink DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4.

8.2.3.9.3 Test purpose

To confirm that an UE, in state CELL_FACH, releases the radio access bearers using common physical channel. After the release, it shall access the affected radio bearers on the DPCH.

8.2.3.9.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is in CELL_FACH state. The SS transmits a RADIO BEARER RELEASE message to the UE. In this message, SS commands the UE to release radio access bearers on common physical channel. At the same time, SS allocates DPCH to support the affected radio bearers. The UE shall release the indicated radio access bearers and transmit a RADIO BEARER RELEASE COMPLETE message on the uplink DCCH using AM RLC. SS calls for generic procedure C.3 to check that UE is in CELL_DCH state.

Expected sequence

Step	Direction		Message	Comment			
	UE	SS					
1	•	ť	RADIO BEARER RELEASE				
2				UE shall release the radio access bearers carried by common physical channel.			
3	-)	RADIO BEARER RELEASE COMPLETE				
4	÷	\rightarrow	CALL C.3	If the test result of C.3 indicates that UE is in CELL_DCH state, the test passes, otherwise it fails.			

Specific Message Contents

RADIO BEARER RELEASE (Step 1)

Use the same message sub-type titled "Packet to CELL_DCH from CELL_FACH in PS" in [9] TS 34.108 clause 9.

8.2.3.9.5 Test requirement

After step 2 the UE shall transmit a RADIO BEARER RELEASE COMPLETE message using the dedicated physical channel allocated.

<End of Modifications>

<Start of Modifications>

- 8.2.3.15 Radio Bearer Release for transition from CELL_FACH to CELL_FACH: Success
- 8.2.3.15.1 Definition
- 8.2.3.15.2 Conformance requirement

If the UE receives:

- a RADIO BEARER RELEASE message; or

it shall:

1> if the UE will enter the CELL_DCH state from any state other than CELL_DCH state at the conclusion of this procedure:

- $\underline{24}$ > perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If the UE was in CELL_FACH state upon reception of the reconfiguration message and remains in CELL_FACH state, the UE shall:

1> if the IE "Frequency info" is included in the received reconfiguration message:

2> select a suitable UTRA cell according to TS 25.304 on that frequency;

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

- 1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER RELEASE message, the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message on the uplink DCCH using AM RLC.

If the IE "RAB information to reconfigure" is included then the UE shall:

1> if the entry for the radio access bearer identified by the IE "CN domain identity" together with the IE "RAB Identity" in the variable ESTABLISHED_RABS already exists:

•••

1> else:

2> set the variable INVALID_CONFIGURATION to TRUE.

If the variable INVALID_CONFIGURATION is set to TRUE the UE shall:

- 1> keep the configuration existing before the reception of the message;
- 1> transmit a failure response message, setting the information elements as specified below:
 - 2> include the IE "RRC transaction identifier"; and
 - 3> set it to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
 - 3> clear that entry.
 - 2> set the IE "failure cause" to "invalid configuration".
- 1> set the variable INVALID_CONFIGURATION to FALSE;
- 1> continue with any ongoing processes and procedures as if the reconfiguration message was not received.

The procedure ends.

• • •

- 1> in case of reception of a RADIO BEARER RECONFIGURATION message:
 - 2> if the radio bearer reconfiguration procedure affects several radio bearers:
 - 3> (may) include the identities of the radio bearers for which the procedure would have been successful into the RADIO BEARER RECONFIGURATION FAILURE message.
 - 2> transmit a RADIO BEARER RECONFIGURATION FAILURE as response message on the DCCH using AM RLC.

Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4, 8.6.4.2a, 8.2.2.11, 8.2.2.9.

8.2.3.15.3 Test purpose

To confirm that the UE releases the existing the radio bearer(s) according to the RADIO BEARER RELEASE message.

8.2.3.15.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: PS-DCCH+DTCH_FACH (state 6-11) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is in CELL_FACH state. The SS transmits a RADIO BEARER RELEASE message to the UE. The UE releases the radio access bearer and transmits a RADIO BEARER RELEASE COMPLETE message on the uplink DCCH using AM RLC. SS calls for generic procedure C.2 to check that UE is in CELL_FACH state.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	←	RADIO BEARER RELEASE	
2		Void	
3	\rightarrow	RADIO BEARER RELEASE COMPLETE	
4	÷	RADIO BEARER RECONFIGURATION	The IE "RAB information to reconfigure" is included with the same RAB identity as was released with the RADIO BEARER RELEASE message.
5	\rightarrow	RADIO BEARER RECONFIGURATION FAILURE	The UE responds with failure, in case the RB is properly removed
6	\leftrightarrow	CALL C.2	If the test result of C.2 indicates that UE is in CELL_FACH state, the test passes, otherwise it fails.

Specific Message Contents

RADIO BEARER RELEASE

Use the same message sub-type titled "Packet to CELL_FACH from CELL_FACH in PS" in [9] TS 34.108 clause 9.

RADIO BEARER RECONFIGURATION (Step 4)

The contents of RADIO BEARER RECONFIGURATION message in this test case is specified below:

Information Element	Condition	Value/remark
Message Type		
RRC transaction identifier		Arbitrarily selects an integer between 0 and 3
Integrity check info		The presence of this IE is dependent on IXIT
5,		statements in TS 34.123-2. If integrity
		protection is indicated to be active, this IE is
		present with the values of the sub IEs as
		stated below. Else, this IE and the sub-IEs are
		omitted.
 message authentication code 		SS calculates the value of MAC-I for this
5		message and writes to this IE.
- RRC message sequence number		SS provides the value of this IE, from its
		internal counter.
Integrity protection mode info		Not Present
Ciphering mode info		Not Present
Activation time		Now
New U-RNTI		Not Present
New C-RNTI		Not Present
RRC State indicator		CELL_FACH
UTRAN DRX cycle length coefficient		Not Present
CN information info		Not Present
URA identity		Not Present
RAB information to reconfigure list		
- RAB information to reconfigure		(AM DTCH for PS domain)
- RAB identity		0000 0101B
- CN domain identity		PS domain
 NAS Synchronization Indicator 		Not Present
RB information to reconfigure list		TS25.331 specifies that "Although this IE is not
		always required, need is MP to align with
		ASN.1".
 RB information to reconfigure 		(Dummy)
- RB identity		1
- PDCP info		Not Present
- PDCP SN info		Not Present
- RLC info		Not Present
- RB mapping info		Not Present
- RB stop/continue		Not Present
RB information to be affected list		Not Present
UL Transport channel information for all transport		Not Present
channels		
Deleted TrCH information list		Not Present
Added or Reconfigured TrCH information list		Not Present
CHOICE mode		Not Present
Deleted DL TrCH information list		Not Present
Added or Reconfigured DL TrCH information list		Not Present
Frequency info	_	Not Present
Maximum allowed UL TX power	_	Not Present
CHOICE channel requirement	_	Not Present
CHOICE Mode		FDD
- Downlink PDSCH information		Not Present
Downlink information common for all radio links		Not Present
Downlink information per radio link list		Not Present

RADIO BEARER RECONFIGURATION FAILURE (step 5)

The contents of RADIO BEARER RECONFIGURATION FAILURE message in this test case is the same as the RADIO BEARER RECONFIGURATION FAILURE message as found in [9] TS 34.108 clause 9, with the following exceptions:

Information Element	Value/remark		
Failure cause	Invalid configuration		

8.2.3.15.5 Test requirement

After step 1 the UE shall transmit a RADIO BEARER RELEASE COMPLETE message using AM RLC on the common physical channel.

After step 4, UE shall transmit a RADIO BEARER RECONFIGURATION FAILURE message to verify that the RAB is properly removed.

<End of Modifications>

<Start of Modifications>

- 8.2.3.18 Radio Bearer Release from CELL_DCH to CELL_PCH: Success
- 8.2.3.18.1 Definition
- 8.2.3.18.2 Conformance requirement

If the UE receives:

- a RADIO BEARER RELEASE message; or

it shall:

- 1> if the UE will enter the CELL DCH state from any state other than CELL DCH state at the conclusion of this procedure:
 - <u>42</u>> perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If after state transition the UE enters CELL_PCH or URA_PCH state, the UE shall, after the state transition and transmission of the response message:

1> if the IE "Frequency info" is included in the received reconfiguration message:

2> select a suitable UTRA cell according to TS 25.304 on that frequency.

1> if the IE "Frequency info" is not included in the received reconfiguration message:

2> select a suitable UTRA cell according to TS 25.304.

- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> if the IE "UTRAN DRX cycle length coefficient" is not included in the same message:

2> set the variable INVALID_CONFIGURATION to TRUE.

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and

1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER RELEASE message, the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message on the uplink DCCH using AM RLC.

If the new state is CELL_PCH, the response message shall be transmitted using the old configuration before the state transition, but the new C-RNTI shall be used if the IE "New C-RNTI" was included in the received reconfiguration message, and the UE shall:

1> when RLC has confirmed the successful transmission of the response message:

•••

2> enter the new state (CELL_PCH);

...

Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4.

8.2.3.18.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RELEASE COMPLETE before entering CELL_PCH state after it received a RADIO BEARER RELEASE message and released its radio access bearers.

8.2.3.18.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is in CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message. The UE transmits a RADIO BEARER RELEASE COMPLETE message using AM RLC and enters into CELL_PCH state. SS calls for generic procedure C.4 to check that UE is in CELL_PCH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	÷	<u>.</u>	RADIO BEARER RELEASE	
2	<i>→</i>		RADIO BEARER RELEASE COMPLETE	The UE sends this message before it completes state transition.
2a			Void	SS waits 5 seconds to allow the UE to read system information before the next step.
3	\leftrightarrow		CALL C.4	If the test result of C.4 indicates that UE is in CELL_PCH state, the test passes, otherwise it fails.

Specific Message Contents

RADIO BEARER RELEASE (Step 1) (FDD)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in [9] TS 34.108 clause 9 with following exceptions:

Information Element	Value/remark
RRC State Indicator	CELL_PCH
UTRAN DRX cycle length coefficient	3
Downlink information for each radio link	
- Primary CPICH info	
- Primary scrambling code	100

RADIO BEARER RELEASE (Step 1) (TDD)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in [9] TS 34.108 clause 9 with following exceptions:

Information Element	Value/remark
RRC State Indicator	CELL_PCH
UTRAN DRX cycle length coefficient	3
Downlink information for each radio links	
- Primary CCPCH info	
-Cell parameters ID	4

8.2.3.18.5 Test requirement

After step 1 the UE transmits a RADIO BEARER RELEASE COMPLETE message on uplink DCCH using AM RLC.

8.2.3.19 Radio Bearer Release from CELL_DCH to URA_PCH: Success

- 8.2.3.19.1 Definition
- 8.2.3.19.2 Conformance requirement

If the UE receives:

- a RADIO BEARER RELEASE message; or

it shall:

- 1> if the UE will enter the CELL DCH state from any state other than CELL DCH state at the conclusion of this procedure:
 - $\underline{24}$ > perform the physical layer synchronisation procedure <u>A</u> as specified in TS 25.214;
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified in the following and perform the actions below.

The UE shall then:

1> enter a state according to TS 25.331 subclause 8.6.3.3.

If after state transition the UE enters CELL_PCH or URA_PCH state, the UE shall, after the state transition and transmission of the response message:

- 1> if the IE "Frequency info" is included in the received reconfiguration message:
 - 2> select a suitable UTRA cell according to TS 25.304 on that frequency.

- 1> if the IE "Frequency info" is not included in the received reconfiguration message:
 - 2> select a suitable UTRA cell according to TS 25.304.
- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> if the IE "UTRAN DRX cycle length coefficient" is included in the same message:
 - 2> use the value in the IE "UTRAN DRX Cycle length coefficient" for calculating Paging occasion and PICH Monitoring Occasion as specified in TS 25.331 subclause 8.6.3.2.
- 1> if the IE "UTRAN DRX cycle length coefficient" is not included in the same message:
 - 2> set the variable INVALID_CONFIGURATION to TRUE.

•••

- 1> if the UE enters URA_PCH state, and after cell selection the criteria for URA update caused by "URA reselection" according to TS 25.331 subclause 8.3.1 is fulfilled:
 - 2> initiate a URA update procedure according to TS 25.331 subclause 8.3.1 using the cause "URA reselection";
 - 2> when the URA update procedure is successfully completed:
 - 3> the procedure ends.

The UE shall transmit a response message as specified in TS 25.331 subclause 8.2.2.4, setting the information elements as specified below. The UE shall:

- 1> set the IE "RRC transaction identifier" to the value of "RRC transaction identifier" in the entry for the received message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

In case the procedure was triggered by reception of a RADIO BEARER RELEASE message, the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message on the uplink DCCH using AM RLC.

If the new state is URA_PCH, the response message shall be transmitted using the old configuration before the state transition, but the new C-RNTI shall be used if the IE "New C-RNTI" was included in the received reconfiguration message, and the UE shall:

1> when RLC has confirmed the successful transmission of the response message:

•••

- 2> enter the new state (URA_PCH);
- •••

Reference

3GPP TS 25.331 clause 8.2.2.3, 8.2.2.4.

8.2.3.19.3 Test purpose

To confirm that the UE transmits a RADIO BEARER RELEASE COMPLETE before entering URA_PCH state after it received a RADIO BEARER RELEASE message and released its radio bearers.

8.2.3.19.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108.

Test Procedure

The UE is in CELL_DCH state. The SS transmits a RADIO BEARER RELEASE message. The UE transmits a RADIO BEARER RELEASE COMPLETE message using AM RLC and enters into URA_PCH state. SS calls for generic procedure C.5 to check that UE is in URA_PCH state.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1	÷		RADIO BEARER RELEASE	
2	<i>→</i>		RADIO BEARER RELEASE COMPLETE	The UE sends this message before it completes state transition.
2a			Void	SS waits 5 seconds to allow the UE to read system information before the next step.
3	\leftrightarrow		CALL C.5	If the test result of C.5 indicates that UE is in URA_PCH state, the test passes, otherwise it fails.

Specific Message Contents

RADIO BEARER RELEASE (Step 1) (FDD)

Use the same message sub-type titled "Packet to CELL_FACH from CELL_DCH in PS" in [9] TS 34.108 clause 9 with following exceptions:

Information Element	Value/remark
RRC State Indicator	URA_PCH
UTRAN DRX cycle length coefficient	3
Downlink information for each radio link	
- Primary CPICH info	
- Primary scrambling code	100

RADIO BEARER RELEASE (Step 1) (TDD)

Information Element	Value/remark
RRC State Indicator	CELL_PCH
UTRAN DRX cycle length coefficient	3
Downlink information for each radio links	
- Primary CCPCH info	
-Cell parameters ID	4

8.2.3.19.5 Test requirement

After step 1 the UE transmits a RADIO BEARER RELEASE COMPLETE message to the UE on uplink DCCH using AM RLC.

<End of Modifications>

3GPP TSG-T1 Meeting #17 Luton, UK, 4 th – 8 th Nov 2002						T1-020788				
3GPP TSG-T1/SI	3GPP TSG-T1/SIG Meeting #26 Luton, UK, 4 th – 8 th Nov 2002						T1S020736			
			CHA	NGE F	REQI	JEST	Г			CR-Form-v7
^ж ТS 3	<mark>4.1</mark> 2	<mark>23-1</mark>	CR <mark>340</mark>	ж	rev	- *	Current v	ersion	5.1.1	ж
For <u>HELP</u> on us	sing t	this for	m, see botto	m of this pa	age or lo	ook at tl	he pop-up t	ext ove	er the X sy	mbols.
Proposed change a	affect	<i>ts:</i> ι	JICC apps ≇		MEX	Radio A	Access Net	work	Core No	etwork
Title: ¥		to 34. 17mee	123-1 clause eting	e 8.3 (Pack	age 1) F	Rel-5: C	orrection fr	om CR	s approved	lin
Source: ೫	Par	nasonio	С							
Work item code: %	TE	l					Date	;	8/10/2002	
Category: ₩	Deta	F (corr A (corr B (add C (fund D (edit iled exp	the following or rection) responds to a lition of feature ctional modific torial modifica blanations of t 3GPP <u>TR 21.9</u>	correction ir e), cation of feat tion) he above ca	ture)		2	of the (GS (Re (Re (Re (Re (Re 5 (Re	EL-5 following rel SM Phase 2) blease 1996) blease 1997) blease 1998) blease 1999) blease 4) blease 5) blease 6)	
Passon for change	. .	1	From CR1	680						
Reason for change	. њ		The use of applicable. From CR1 UE report	synchroniz 573 of "SFN-SF requency n	-N obse	rved tim	ne differenc	e" mea	ified where asurement d from a fui	in Intra-
Summary of chang	ie:	1.	Change to The confor					4.3		
		2.			ed time	differer		"SFN-	SFN observ	ved time
Consequences if not approved:	ж	The t	test specifica	ations are n	ot aligne	ed with	the core sp	ecifica	tion	
Clauses affected:	ж	8.3.1	.1, 8.3.1.3, 8	3 <mark>.3.4.1, 8.3</mark> .	<mark>.4.2, 8.3</mark>	.4.3				
Other specs	¥	YN X	Other core	specificatio	ons	ж				

Affected:	XTest specificationsXO&M Specifications	
Other comments:	# Affects R99, REL-4, REL-5	

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

----<Start of Modifications>---

8.3 RRC connection mobility procedure

- 8.3.1 Cell Update
- 8.3.1.1 Cell Update: cell reselection in CELL_FACH
- 8.3.1.1.1 Definition

8.3.1.1.2 Conformance requirement

A UE shall initiate the cell update procedure in the following cases:

1> Uplink data transmission:

• • •

1> Paging response:

• • •

1> Radio link failure:

...

1> Re-entering service area:

•••

1> RLC unrecoverable error:

• • •

- 1> Cell reselection:
 - 2> if none of the criteria for performing cell update with the causes specified above in the current subclause is met:
 - 3> if the UE is in CELL_FACH or CELL_PCH state and the UE performs cell re-selection; or
 - 3> if the UE is in CELL_FACH state and the variable C_RNTI is empty:

4> perform cell update using the cause "cell reselection".

•••

When initiating cell update procedure, the UE shall:

- 1> stop timer T305;
- 1> if the UE is in CELL_DCH state:
 - •••

•••

- 1> move to CELL_FACH state, if not already in that state;
- 1> if the UE performs cell re-selection:
 - 2> clear the variable C_RNTI; and

2> stop using that C_RNTI just cleared from the variable C_RNTI in MAC.

1> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;

1> in case of a cell update procedure:

2> set the contents of the CELL UPDATE message according to TS 25.331 subclause 8.3.1.3;

2> submit the CELL UPDATE message for transmission on the uplink CCCH.

•••

1> set counter V302 to 1;

1> start timer T302 when the MAC layer indicates success or failure in transmitting the message.

• • •

In case of cell update procedure the UE shall transmit a CELL UPDATE message.

The UE shall set the IEs in the CELL UPDATE message as follows:

- 1> set the IE "Cell update cause" corresponding to the cause specified in TS 25.331 subclause 8.3.1.2 that is valid when the CELL UPDATE message is submitted to lower layers for transmission;
- NOTE: During the time period starting from when a cell update procedure is initiated by the UE until when the procedure ends, additional CELL UPDATE messages may be transmitted by the UE with different causes.
- 1> set the IE "U-RNTI" to the value of the variable U_RNTI;

1> if the value of the variable PROTOCOL_ERROR_INDICATOR is TRUE:

•••

1> if the value of the variable FAILURE_INDICATOR is TRUE:

•••

•••

When the UE receives a CELL UPDATE CONFIRM message; and

- if the message is received on the CCCH, and IE "U-RNTI" is present and has the same value as the variable U_RNTI; or
- if the message is received on DCCH:

the UE shall:

- 1> stop timer T302;
- 1> in case of a cell update procedure and the CELL UPDATE CONFIRM message:
 - 2> includes "RB information elements"; and/or
 - 2> includes "Transport channel information elements"; and/or
 - 2> includes "Physical channel information elements"; and
 - 2> if the variable ORDERED_RECONFIGURATION is set to FALSE:
 - 3> set the variable ORDERED_RECONFIGURATION to TRUE.
- 1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following:

```
____
```

. . .

...

1> enter a state according to subclause 8.6.3.3 applied on the CELL UPDATE CONFIRM / URA UPDATE CONFIRM message.

perform the physical layer synchronisation procedure as specified in TS 25.214;

•••

If the UE after state transition remains in CELL_FACH state, it shall

- 1> start the timer T305 using its initial value if timer T305 is not running and periodical cell update has been configured by T305 in the IE "UE Timers and constants in connected mode" set to any other value than "infinity";
- 1> select PRACH according to TS 25.331 subclause 8.5.17;
- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> not prohibit periodical status transmission in RLC;
- 1> if the IE "UTRAN DRX cycle length coefficient" is included in the same message:

2> ignore that IE and stop using DRX.

If the UE after the state transition remains in CELL_FACH state; and

```
- a C-RNTI is stored in the variable C_RNTI;
```

• • •

the UE shall:

•••

- 1> in case of a cell update procedure:
 - 2> set the IE "RRC transaction identifier" in any response message transmitted below to the value of "RRC transaction identifier" in the entry for the CELL UPDATE CONFIRM message in the table "Accepted transactions" in the variable TRANSACTIONS; and
 - 2> clear that entry.

• • •

1> transmit a response message as specified in TS 25.331 subclause 8.3.1.7;

•••

If the CELL UPDATE CONFIRM message:

- includes the IE "RB information to release list":

the UE shall:

1> transmit a RADIO BEARER RELEASE COMPLETE as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include the IE "RB information to release list"; and
- includes the IE "RB information to reconfigure list"; or
- includes the IE "RB information to be affected list":

the UE shall:

1> transmit a RADIO BEARER RECONFIGURATION COMPLETE as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- includes "Transport channel information elements":

the UE shall:

1> transmit a TRANSPORT CHANNEL RECONFIGURATION COMPLETE as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- includes "Physical channel information elements":

the UE shall:

1> transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- includes "CN information elements"; or
- includes the IE "Ciphering mode info"; or
- includes the IE "Integrity protection mode info"; or
- includes the IE "New C-RNTI"; or
- includes the IE "New U-RNTI":

the UE shall:

1> transmit a UTRAN MOBILITY INFORMATION CONFIRM as response message using AM RLC.

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- does not include "CN information elements"; and
- does not include the IE "Ciphering mode info"; and
- does not include the IE "Integrity protection mode info"; and
- does not include the IE "New C-RNTI"; and
- does not include the IE "New U-RNTI":

the UE shall:

1> transmit no response message.

If the new state is CELL_FACH, the response message shall be transmitted using the new configuration after the state transition., and the UE shall:

•••

1> if the variable PDCP_SN_INFO is empty:

.

2> if the CELL UPDATE CONFIRM or URA UPDATE CONFIRM message did not contain the IE "Ciphering mode info":

3> when RLC has been requested to transmit the response message,

```
4> continue with the remainder of the procedure.
```

•••

If any or several of the following conditions are true:

...;

 reselection to another UTRA cell (including the previously serving cell) before completion of the cell update or URA update procedure;

the UE shall:

1> stop T302 if it is running;

•••

1> check whether it is still in "in service area";

•••

- 1> in case of a cell update procedure:
 - 2> clear any entry for the CELL UPDATE CONFIRM message in the table "Accepted transactions" in the variable TRANSACTIONS.

If the UE detects "in service area" if it has not entered idle mode, and:

- 1> if V302 is equal to or smaller than N302, the UE shall:
 - 2> if the UE performed cell re-selection:
 - 3> delete its C-RNTI.
 - 2> in case of a cell update procedure:
 - 3> set the contents of the CELL UPDATE message according to TS 25.331 subclause 8.3.1.3;
 - 3> submit the CELL UPDATE message for transmission on the uplink CCCH.
 - 2> increment counter V302;
 - 2> restart timer T302 when the MAC layer indicates success or failure to transmit the message.
- 1> if V302 is greater than N302, the UE shall:

•••

Reference

3GPP TS 25.331 clause 8.3.1

8.3.1.1.3 Test purpose

- 1. To confirm that the UE executes a cell update procedure after the successful reselection of another UTRA cell.
- 2. To confirm that the UE sends the correct uplink response message when executing cell update procedure due to cell reselection.

8.3.1.1.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 and 2 are active.

UE: PS-DCCH+DTCH_FACH (state 6-11) in cell 1 as specified in clause 7.4 of TS 34.108.

Test Procedure

Parameter	Unit	Cell 1		Cell 2	
		T0	T1	T0	T1
UTRA RF		Ch. 1		Ch. 1	
Channel					
Number					
CPICH Ec	dBm/3.84MHz	-60	-75	-75	-60
(FDD)					
P-CCPCH	dBm	-60	-75	-75	-60
RSCP (TDD)					

Table 8.3.1.1

Table 8.3.1.1 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. Columns marked "T0" denote the initial conditions. SS switches the power settings repeatedly between columns "T1" and "T0", whenever the description below specifies that the transmission power settings for cell 1 and cell 2 be reversed.

The UE is in the CELL_FACH state, camping onto cell 1. SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall find cell 2 to be more suitable for service and hence perform a cell reselection. After the completion of cell reselection, the UE shall transmits a CELL UPDATE message to the SS on the uplink CCCH of cell 2 and set IE "Cell update cause" to "Cell Reselection". After the SS receives this message, it transmits a CELL UPDATE CONFIRM message, which includes the IE "RRC State Indicator" set to "CELL_FACH", to the UE on the downlink DCCH. UE shall verify that IE "New C-RNTI" is not included in the downlink message and shall send a CELL UPDATE message to SS again. SS shall then send a CELL UPDATE CONFIRM message which includes a valid IE "New C-RNTI". SS verifies that the UE send UTRAN MOBILITY INFORMATION CONFIRM message.. UE shall stay in CELL_FACH state. SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with CELL UPDATE CONFIRM message and allocates new C-RNTI and U-RNTI identities to the UE. The IE "RRC State Indicator" is set to "CELL_FACH" in this message. The UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message. Following this, SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall initiate a cell update procedure by transmitting a CELL UPDATE message and stating the cause as 'cell re-selection'. SS replies with a CELL UPDATE CONFIRM message which contains IE "Physical channel information elements". The UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message to acknowledge the change in physical resources. Then, SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains IE "Transport channel information elements". The UE shall send TRANSPORT CHANNEL RECONFIGURATION COMPLETE message. Following this, SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 2. SS replies with a CELL UPDATE CONFIRM message which contains IE "RB information to be affected list". The UE shall send RADIO BEARER RECONFIGURATION COMPLETE message. Then, SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS replies with a CELL UPDATE CONFIRM message which contains IE "RB information to release list". The UE shall send RADIO BEARER RELEASE COMPLETE message. Finally, SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.1. The UE shall send a CELL UPDATE message on the uplink CCCH of cell 2. SS shall not respond to this message but SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.1. UE shall send a CELL UPDATE message on the uplink CCCH of cell 1. SS shall then send CELL UPDATE CONFIRM message to UE. UE shall reply with UTRAN MOBILITY INFORMATION CONFIRM message. SS calls for generic procedure C.2 to check that UE is in CELL FACH state.

Step	Direction Message		Message	Comment
	UE	SS	7	
1				The UE is in the CELL_FACH state in cell 1
2			Void	SS applies the downlink transmission power settings, according to the values in columns "T1" of table 8.3.1.1. The UE shall find that the cell 2 is better for service and perform a reselection. SS waits for the maximum duration required for the UE to camp to cell 2.
3		→	CELL UPDATE	Value "cell reselection" shall be indicated in IE "Cell update cause"
4		÷	CELL UPDATE CONFIRM	IE "RRC State Indicator" is set to "CELL_FACH". SS set k=0.
4a		>	CELL UPDATE	Value "cell reselection" shall be indicated in IE "Cell update cause"
4b	•	÷	CELL UPDATE CONFIRM	See message content. SS set k=0.
5	-	〉	UTRAN MOBILITY INFORMATION CONFIRM	
6				SS reverses the transmission power level of cell 1 and cell 2.
7	· ·	>	CELL UPDATE	

Expected sequence

8	←	CELL UPDATE CONFIRM	IE "RRC State Indicator" is set
0			
			to "CELL_FACH". If $k \ge 0$, new
			C-RNTI and U-RNTI identities
			are assigned to the UE. If k>0,
			IE "Physical channel
			information elements" is
			included in this message. If
			k>1, IE "Transport channel
			information elements" is
			included in this message. If
			k>2, IE "RB information to be
			affected list" is included in this
			message. If k>3, IE "RB
			information to release list" is
			included in this message.
			Increment k by 1.
9	\rightarrow	UTRAN MOBILITY INFORMATION	If k=1 when SS received this
		CONFIRM	message, go to step 6. Else
			test fails. If this message is not
40			received, proceed to next step.
10	\rightarrow	PHYSICAL CHANNEL	If k=2 when SS received this
		RECONFIGURATION COMPLETE	message, go to step 6. Else
			test fails. If this message is not
			received, proceed to next step.
11	\rightarrow	TRANSPORT CHANNEL	If k=3 when SS received this
		RECONFIGURATION COMPLETE	message, go to step 6. Else
			test fails. If this message is not
40			received, proceed to next step.
12	\rightarrow	RADIO BEARER RECONFIGURATION	If k=4 when SS received this
		COMPLETE	message, go to step 6. Else
			test fails. If this message is not
10			received, proceed to next step. If k=5 when SS received this
13	\rightarrow	RADIO BEARER RELEASE COMPLETE	
			message, proceed to next
			step. Else test fails. If this
			message is not received, test fails.
14			SS reverses the transmission
14			power level of cell 1 and cell 2.
15	\rightarrow	CELL UPDATE	
16	,		SS reverses the transmission
10			power level of cell 1 and cell 2.
17	\rightarrow	CELL UPDATE	
18	←	CELL UPDATE CONFIRM	
19	\rightarrow	UTRAN MOBILITY INFORMATION	
-		CONFIRM	
20	\leftrightarrow	CALL C.2	If the test result of C.2
			indicates that UE is in
			CELL_FACH state, the test
			passes, otherwise it fails.

Specific Message Contents

CELL UPDATE (Step 3, 7, 15 and 17)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	In step 3, check to see if set to '0000 0000 0000 0000
	0001'. In step 7 and when k<1, check to see if set to
	'0000 0000 0000 0000 0001'. In step 7 and when k>0,
	check to see if set to same string in IE "S-RNTI" in IE
	"New U-RNTI" of CELL UPDATE CONFIRM message in
	previous step 8. In step 15 and 17, check to see if set to
	same string in IE "S-RNTI" in IE "New U-RNTI" of CELL
	UPDATE CONFIRM message in previous step 8.
Cell Update Cause	Check to see if set to 'Cell Re-selection'

CELL UPDATE CONFIRM (Step 4)

Use the same message sub-type found in TS 34.108, clause 9.

CELL UPDATE CONFIRM (Step 4b and 18)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark	
RRC State Indicator	CELL_FACH	
New C-RNTI	'1010 1010 1010 1010'	

CELL UPDATE CONFIRM (Step 8 and k = 0)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark		
New U-RNTI			
- SRNC Identity	'0000 0000 0001'		
- S-RNTI	An arbitrary 20-bits string which is different from original		
	S-RNTI		
New C-RNTI	An arbitrary 16-bits string which is different from original		
	C-RNTI assigned in RRC connection establishment		
	procedure.		

CELL UPDATE CONFIRM (Step 8 and k=1)

Use the same message sub-type found in step 8 and k=0, with the following exceptions:

Information Element	Value/remark
Maximum allowed uplink TX power	3 dB below the following value: Minimum of {33 dBm for FDD and 30 dBm for TDD,
	maximum uplink power allowed under the UE power class}

CELL UPDATE CONFIRM (Step 8 and k=2)

Use the same message sub-type found in step 8 and k=1, with the following exceptions:

Information Element	Value/remark
Added or Reconfigured uplink TrCH information	Same as the system information block type 5
Added or Reconfigured downlink TrCH information	Same as the system information block type 5

CELL UPDATE CONFIRM (Step 8 and k=3)

Use the same message sub-type found in step 8 and k=2, with the following exceptions:

Information Element	Value/remark	
RB information to be reconfigure		
- RB identity	20	
- PDCP info	Not Present	
- PDCP SN info	Not Present	
- RLC info	Not Present	
- RB mapping info	Not Present	
- RB stop/continue	Stop	

CELL UPDATE CONFIRM (Step 8 and k=4)

Use the same message sub-type found in step 8 and k=3, with the following exceptions:

Information Element	Value/remark
RB information to release	
-RB identity	4

8.3.1.1.5 Test requirement

After step 2 the UE shall reselect to cell 2 and then it shall transmit a CELL UPDATE message which, sets the value "cell reselection" in IE "Cell update cause".

After step 3 the UE shall transmit CELL UPDATE message which sets the value "cell reselection" in IE "Cell update cause".

After step 4a, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message to acknowledge that it has started to use the new RNTI identities allocated.

After step 6 the UE shall sent a CELL UPDATE message to the cell with stronger transmitting power, in order to indicate that a cell reselection has taken place.

After step 8, if k=1, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message to acknowledge that it has started to use the new RNTI identities allocated.

If k=2, the UE shall transmit a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the new physical channel assigned.

If k=3, the UE shall transmit a TRANSPORT CHANNEL RECONFIGURATION COMPLETE message to acknowledge that it has reconfigured the transport channels.

If k=4, the UE shall transmit a RADIO BEARER RECONFIGURATION COMPLETE message to acknowledge that it has reconfigured the radio bearers.

If k=5, the UE shall transmit a RADIO BEARER RELEASE COMPLETE message to acknowledge that it has release its radio bearers.

After step 14 the UE shall transmit a CELL UPDATE message to the cell with stronger transmitting power, in order to indicate that a cell reselection has taken place.

After step 16 the UE shall transmit a CELL UPDATE message to the cell with stronger transmitting power, in order to indicate that a cell reselection has taken place.

After step 18, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message to acknowledge that it has started to use the new RNTI identities allocated.

---<End of Modifications>---

----<Start of Modifications>----

- 8.3.1.3 Cell Update: periodical cell update in CELL_FACH
- 8.3.1.3.1 Definition
- 8.3.1.3.2 Conformance requirement

UE shall initiate the cell update procedure in the following cases:

1> Uplink data transmission:

•••

1> Paging response:

•••

1> Radio link failure:

•••

1> Re-entering service area:

•••

1> RLC unrecoverable error:

• • •

1> Cell reselection:

•••

- 1> Periodical cell update:
 - 2> if none of the criteria for performing cell update with the causes specified above in the current subclause is met; and
 - 2> if the UE is in CELL_FACH or CELL_PCH state; and
 - 2> if the timer T305 expires; and
 - 2> if the criteria for "in service area" as specified in TS 25.331 subclause 8.5.5.2 is fulfilled; and
 - 2> if periodic updating has been configured by T305 in the IE "UE Timers and constants in connected mode" set to any other value than "infinity":

3> perform cell update using the cause "periodical cell update".

When initiating the cell update procedure, the UE shall:

```
1> stop timer T305;
```

•••

1> move to CELL_FACH state, if not already in that state;

•••

- 1> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;
- 1> in case of a cell update procedure:
 - 2> set the contents of the CELL UPDATE message according to TS 25.331 subclause 8.3.1.3;

2> submit the CELL UPDATE message for transmission on the uplink CCCH.

1> set counter V302 to 1;

1> start timer T302 when the MAC layer indicates success or failure in transmitting the message.

• • •

In case of cell update procedure the UE shall transmit a CELL UPDATE message.

The UE shall set the IEs in the CELL UPDATE message as follows:

- 1> set the IE "Cell update cause" corresponding to the cause specified in TS 25.331 subclause 8.3.1.2 that is valid when the CELL UPDATE message is submitted to lower layers for transmission;
- NOTE: During the time period starting from when a cell update procedure is initiated by the UE until when the procedure ends, additional CELL UPDATE messages may be transmitted by the UE with different causes.
- 1> set the IE "U-RNTI" to the value of the variable U_RNTI;

•••

When the UE receives a CELL UPDATE CONFIRM message; and

- if the message is received on the CCCH, and IE "U-RNTI" is present and has the same value as the variable U_RNTI; or
- if the message is received on DCCH:

the UE shall:

1> stop timer T302;

•••

1> act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following:

2> perform the physical layer synchronisation procedure as specified in TS 25.214;

1> enter a state according to TS 25.331 subclause 8.6.3.3 applied on the CELL UPDATE CONFIRM / URA UPDATE CONFIRM message.

If the UE after state transition remains in CELL_FACH state, it shall

- 1> start the timer T305 using its initial value if timer T305 is not running and periodical cell update has been configured by T305 in the IE "UE Timers and constants in connected mode" set to any other value than "infinity";
- 1> select PRACH according to TS 25.331 subclause 8.5.17;
- 1> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
- 1> not prohibit periodical status transmission in RLC;
- 1> if the IE "UTRAN DRX cycle length coefficient" is included in the same message:

2> ignore that IE and stop using DRX.

If the UE after the state transition remains in CELL_FACH state; and

- a C-RNTI is stored in the variable C_RNTI;

or

- the UE after the state transition moves to another state than the CELL_FACH state:

the UE shall:

- 1> in case of a cell update procedure:
 - 2> set the IE "RRC transaction identifier" in any response message transmitted below to the value of "RRC transaction identifier" in the entry for the CELL UPDATE CONFIRM message in the table "Accepted transactions" in the variable TRANSACTIONS; and
 - 2> clear that entry.
- 1> transmit a response message as specified in TS 25.331 subclause 8.3.1.7;
- 1> in case of a cell update procedure:
 - 2> clear the entry for the CELL UPDATE CONFIRM message in the table "Rejected transactions" in the variable TRANSACTIONS.
- 1> set the variable CELL_UPDATE_STARTED to FALSE;
- 1> clear the variable SECURITY_MODIFICATION.

•••

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- includes "CN information elements"; or
- includes the IE "Ciphering mode info"; or
- includes the IE "Integrity protection mode info"; or
- includes the IE "New C-RNTI"; or
- includes the IE "New U-RNTI":

the UE shall:

```
1> transmit a UTRAN MOBILITY INFORMATION CONFIRM as response message using AM RLC.
```

If the CELL UPDATE CONFIRM message:

- does not include "RB information elements"; and
- does not include "Transport channel information elements"; and
- does not include "Physical channel information elements"; and
- does not include "CN information elements"; and
- does not include the IE "Ciphering mode info"; and
- does not include the IE "Integrity protection mode info"; and
- does not include the IE "New C-RNTI"; and
- does not include the IE "New U-RNTI":

the UE shall:

1> transmit no response message.

•••

When the UE receives a UTRAN MOBILITY INFORMATION message, it shall:

- 1> act on received information elements as specified in TS 25.331 subclause 8.6;
- 1> if the IE "UE Timers and constants in connected mode" is present:
 - 2> store the values of the IE "UE Timers and constants in connected mode" in the variable TIMERS_AND_CONSTANTS, replacing any previously stored value for each timer and constant; and
 - 2> for each updated timer value:
 - 3> start using the new value next time the timer is started;
 - 2> for each updated constant value:
 - 3> start using the new value directly;
- 1> set the IE "RRC transaction identifier" in the UTRAN MOBILITY INFORMATION CONFIRM message to the value of "RRC transaction identifier" in the entry for the UTRAN MOBILITY INFORMATION message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;

1> transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC;

•••

Reference

3GPP TS 25.331 clause 8.3.1, 8.3.3.3.

8.3.1.3.3 Test purpose

1. To confirm that the UE executes a periodical cell update procedure following the expiry of timer T305.

8.3.1.3.4 Method of test

Initial Condition

System Simulator: 2 cells – Cell 1 and 2 are active.

UE: PS-DCCH+DTCH_FACH (state 6-11) in cell 1 as specified in clause 7.4 of TS 34.108.

Test Procedure

Table 8.3.1.3

Parameter	Unit	Cell 1		Cell 2	
		T0	T1	T0	T1
UTRA RF Channel Number		Ch. 1		Ch. 1	
CPICH Ec (FDD)	dBm/ 3.84 MHz	-60	-75	-75	-60
P-CCPCH RSCP (TDD)	dBm	-60	-75	-75	-60

Table 8.3.1.3 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution.

The UE is in CELL_FACH state. When the UE detects the expiry of timer T305 according to the settings in system information, the UE transmits a CELL UPDATE message to the SS on the uplink CCCH with a cause indicating periodical cell updating. SS replies with a CELL UPDATE CONFIRM message, and IE "RRC State Indicator" is set to

"CELL_FACH". SS verifies that the UE does not transmit any uplink message. SS then waits for T305 to expire again. The UE shall send another CELL UPDATE message to report periodic cell updating. After the SS receives this message, it transmits a CELL UPDATE CONFIRM message which includes the IEs "new C-RNTI", "new U-RNTI" to the UE on the downlink DCCH. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities. Next, SS transmits UTRAN MOBILITY INFORMATION message, which includes IE "T305" set to "infinity", to UE. UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message. SS configures its downlink transmission power settings according to columns "T1" in table 8.3.1.3, causing the UE to enter CELL FACH state in cell 2 and transmit a CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". Then SS shall transmit CELL UPDATE CONFIRM. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities. SS then monitors the uplink CCCH for a period of 60 minutes (ideally the SS should monitor this up to the maximum possible value for timer T305 (720 minutes), but for practical reasons 60 minutes (twice default timer of 30 minutes) is regarded as being sufficient) and verifies that no CELL UPDATE message is received. After this, the SS transmits UTRAN MOBILITY INFORMATION message, which includes IE "T305" set to '5', to UE. UE shall transmit UTRAN MOBILITY INFORMATION CONFIRM message. SS configures its downlink transmission power settings according to columns "T0" in table 8.3.1.3, causing the UE to enter CELL FACH state in cell 1 and transmit a CELL UPDATE message on uplink CCCH with IE "Cell update cause" set to "cell reselection". Then SS shall transmit CELL UPDATE CONFIRM. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities. UE shall resume periodic cell updating procedure and transmit CELL UPDATE message after T305 (5 minutes) expires.

Expected sequence

Step	Direction		Message	Comment
	UE	SS		
1				The UE is in the CELL_FACH state. SS waits until T305 has expired.
2		>	CELL UPDATE	IE "Cell update cause" shall be set to "periodical cell updating"
3	÷	_	CELL UPDATE CONFIRM	No RNTI identities are given. No information on PRACH and S-CCPCH are provided.
4				SS verifies that no uplink message is received from UE. SS waits for another period to allow T305 to expire.
5		>	CELL UPDATE	Set to "periodical cell update" in IE "Cell update cause" upon the expiry of timer T305.
6		-	CELL UPDATE CONFIRM	Including IEs "new C-RNTI", "new U-RNTI" and IE "RRC State Indicator" is set to "CELL_FACH"
7		>	UTRAN MOBILITY INFORMATION CONFIRM	

8	\	UTRAN MOBILITY INFORMATION	IE "T305" is set to 'infintiy'.
<u> </u>	\rightarrow	UTRAN MOBILITY INFORMATION	
-	7	CONFIRM	
10			SS applies the downlink
			transmission power settings,
			according to the values in
			columns "T1" of table 8.3.1.3
11	\rightarrow	CELL UPDATE	IE "Cell update cause" shall be
- 10			set to "cell reselection".
12	(CELL UPDATE CONFIRM	
12a	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
13			SS waits for 60 minutes and
			checks that no CELL UPDATE
			message is transmitted on
			uplink PRACH channel.
14	(UTRAN MOBILITY INFORMATION	IE "T305" is set to '5.
15	\rightarrow	UTRAN MOBILITY INFORMATION CONFIRM	
16			SS applies the downlink
			transmission power settings,
			according to the values in
			columns "T0" of table 8.3.1.3
17	\rightarrow	CELL UPDATE	IE "Cell update cause" shall be
			set to "cell reselection".
18	÷	CELL UPDATE CONFIRM	
18a	\rightarrow	UTRAN MOBILITY INFORMATION	
		CONFIRM	
19	\rightarrow	CELL UPDATE	UE shall transmit this
			message with "cell update
			cause" set to "periodical cell
			updating" after T305 expires.
20	÷	CELL UPDATE CONFIRM	

Specific Message Contents

CELL UPDATE (Step 2 and 5)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to '0000 0000 0000 0000 0001'
Cell Update Cause	Check to see if set to 'periodical cell updating'

CELL UPDATE (Step 11 and 17)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to same bit string as in IE "S-RNTI"
	in IE "U-RNTI" of the CELL UPDATE CONFIRM
	message sent in step 6.
Cell Update Cause	Check to see if set to "cell reselection"

CELL UPDATE CONFIRM (Step 3, 12, 18 and 20)

Use the same message sub-type found in TS 34.108, clause 9.

CELL UPDATE CONFIRM (Step 6, 12 and 18)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
New U-RNTI	
- SRNC Identity	Set to '0000 0000 0001'
- S-RNTI	Set to an arbitrary string different from '0000 0000 0000
	0000 0001'
New C-RNTI	'1010 1010 1010 1010'

CELL UPDATE (Step 19)

The same message found in TS 34.108, clause 9 shall be transmitted by the UE on the uplink CCCH, with the exception of the following IEs:

Information Element	Value/remark
U-RNTI	
- SRNC Identity	Check to see if set to '0000 0000 0001'
- S-RNTI	Check to see if set to same bit string as in IE "S-RNTI"
	in IE "U-RNTI" of the CELL UPDATE CONFIRM
	message sent in step 6.
Cell Update Cause	Check to see if set to 'periodical cell updating'

UTRAN MOBILITY INFORMATION (Step 8)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
New U-RNTI	Not Present
New C-RNTI	Not Present
UE Timers and constants in connected mode	
- T305	infinity

UTRAN MOBILITY INFORMATION (Step 14)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
New U-RNTI	Not Present
New C-RNTI	Not Present
UE Timers and constants in connected mode	
- T305	5

Release 5

8.3.1.3.5 Test requirement

After step 1 the UE shall detect the expiry of timer T305 and then transmits a CELL UPDATE message setting value "periodical cell update" into IE "Cell update cause".

After step 3 the UE shall not send any uplink message as a response to CELL UPDATE CONFIRM message sent in step 3.

After step 4 the UE shall send a CELL UPDATE message, specifying the cell updating cause to be "periodical cell update".

After step 6 the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH.

After step 8, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 10, the UE shall transmit a CELL UPDATE message with IE "cell update cause" set to "cell reselection" on the uplink CCCH.

After step 12, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

Between step 12a and 14, the UE shall not transmit any CELL UPDATE message.

After step 14, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 16, the UE shall transmit a CELL UPDATE message with IE "cell update cause" set to "cell reselection" on the uplink CCCH.

After step 18, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

After step 18a, the UE shall transmit a CELL UPDATE message with IE "cell update cause" set to "periodical cell update" on the uplink CCCH.

----<End of Modifications>----

----<Start of Modifications>----

- 8.3.4 Active set update in soft handover (FDD)
- 8.3.4.1 Active set update in soft handover: Radio Link addition
- 8.3.4.1.1 Definition

8.3.4.1.2 Conformance requirement

Upon reception of an ACTIVE SET UPDATE message the UE shall act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following. The UE shall:

- 1> first add the RLs indicated in the IE "Radio Link Addition Information";
- 1> perform the physical layer synchronisation procedure **B** as specified in TS 25.214;
- 1> set the IE "RRC transaction identifier" in the ACTIVE SET UPDATE COMPLETE message to the value of "RRC transaction identifier" in the entry for the ACTIVE SET UPDATE message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;
- 1> transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC without waiting for the <u>completion of the</u> Physical Layer synchronization <u>B</u>, specified in TS25.214;

...

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.1.3 Test purpose

1. To confirm that the UE continues to communicate with the SS on both the additional radio link and an already existing radio link after the radio link addition.

8.3.4.1.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 and 2 are active

UE: CS-DCCH+DTCH_DCH (state 6-9) or PS-DCCH+DTCH_DCH (state 6-10) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

Parameter	Unit	Cell 1				Cell 2			
		T0	T1	T2	T3	T0	T1	T2	T3
UTRA RF Channel Number		Ch. 1				Ch. 1			
CPICH Ec	dBm/ 3.84 MHz	-60	-60	OFF	-60	-75	-60	-60	OFF

Table 8.3.4.1

Table 8.3.4.1 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution.

Initially, the UE goes to connected mode and establishes a radio access bearer in CELL_DCH state in cell 1.

SS configures its downlink transmission power settings according to columns "T1" in table 8.3.4.1. UE shall be triggered to transmit a MEASUREMENT REPORT message which includes the primary scrambling code for cell 2 according to IE "Intra-frequency event identity", which is set to '1a' in the SYSTEM INFORMATION BLOCK TYPE 11. After the MEASUREMENT REPORT message is received, the SS configures the new radio link to be added from cell 2 and then the SS transmits to the UE an ACTIVE SET UPDATE message in cell 1 on DCCH using AM RLC which includes the IE "Radio Link Addition Information" (e.g. Downlink DPCH information and other optional parameters relevant for the additional radio links with Primary CPICH info used for the reference ID).

When the UE receives this message, the UE shall configure layer 1 to begin reception without affecting the current uplink and downlink activities of existing radio links. The UE shall transmit an ACTIVE SET UPDATE COMPLETE message to the SS on the uplink DCCH using AM RLC without waiting for the physical channel synchronisation <u>B</u>.

SS configures its downlink transmission power settings according to columns "T2" in table 8.3.4.1. UE shall not detect the DPCH from cell 1 but continue to communicate through the another DPCH from cell 2. The UE shall transmit a MEASUREMENT REPORT message which indicates the event '1b' for cell 1.

SS shall transmit a UE CAPABILITY ENQUIRY message to confirm that the UE can respond this message through the DPCH in cell 2. The UE shall transmit a UE CAPABILITY ENQUIRY INFORMATION message. Then SS transmits a UE CAPABILITY INFORMATION CONFIRM message.

SS configures its downlink transmission power settings according to columns "T1" in table 8.3.4.1. UE shall detect DPCH from cell 1 and 2 and transmit a MEASUREMENT REPORT message which indicates the event '1a' for cell 1.

The SS configures its downlink transmission power settings according to columns "T3" in table 8.3.4.1. UE shall not detect the DPCH from cell 2 but continue to communicate through another DPCH from cell 1. The UE shall transmit a MEASUREMENT REPORT message which indicates the event '1b' for cell 2.

SS shall transmit a UE CAPABILITY ENQUIRY message to confirm that the UE can respond this message through the DPCH in cell 1. The UE shall transmit a UE CAPABILITY INFORMATION message. Then SS transmits a UE CAPABILITY INFORMATION CONFIRM message. SS calls for generic procedure C.3 to check that UE is in CELL_DCH state.

Step	Direction		Message	Comment		
	UE SS		UE SS		-	
1				SS configures its downlink transmission power settings according to columns "T1" in table 8.3.4.1.		
2	\rightarrow		MEASUREMENT REPORT	See specific message contents for this message		
3	÷		ACTIVE SET UPDATE	SS transmits this message in cell 1 on downlink DCCH using AM RLC. The message includes IE "Radio Link Addition Information". (e.g. Downlink DPCH information and other optional parameters relevant for the additional radio links with Primary CPICH info used for the reference ID in cell 2)		
4	→		ACTIVE SET UPDATE COMPLETE	The UE shall configure a new radio link to cell 2, without interfering with existing connections on the radio link in cell 1.		
5				SS configures its downlink transmission power settings according to columns "T2" in table 8.3.4.1		
5a	<i>→</i>		MEASUREMENT REPORT	See specific message contents for this message		

Expected sequence

6	~	UE CAPABILITY ENQUIRY	Use default message.
7	\rightarrow	UE CAPABILITY INFORMATION	Use default message.
8	\leftarrow	UE CAPABILITY INFORMATION CONFIRM	Use default message.
9			SS configures its downlink
			transmission power settings
			according to columns "T1" in
			table 8.3.4.1
9a	\rightarrow	MEASUREMENT REPORT	See specific message
			contents for this message
10			Wait 15 seconds and SS
			configures its downlink
			transmission power settings
			according to columns "T3" in
			table 8.3.4.1
10a	\rightarrow	MEASUREMENT REPORT	See specific message
			contents for this message
11	\leftarrow	UE CAPABILITY ENQUIRY	Use default message.
12	\rightarrow	UE CAPABILITY INFORMATION	Use default message.
13	÷	UE CAPABILITY INFORMATION CONFIRM	Use default message.
14	\leftrightarrow	CALL C.3	If the test result of C.3
			indicates that UE is in
			CELL_DCH state, the test
			passes, otherwise it fails.

Specific Message Content

The contents of SIB11 broadcasted in cell 1 shall be in accordance with the default SIB11 as specified in section 6.1 of TS 34.108, with the following exceptions:

- New intra-frequency cells						
 Intra-frequency cell id 	1					
- Cell info						
 Cell individual offset 	0dB					
 Reference time difference to cell 	Not Present					
 Read SFN indicator 	TRUE					
- CHOICE mode	FDD					
- Primary CPICH info						
- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)"					
	in clause 6.1 of TS 34.108					
- Primary CPICH TX power	Not Present					
- TX Diversity indicator	FALSE					
- Intra-frequency cell id	2					
- Cell info						
- Cell individual offset	0dB					
- Reference time difference to cell	Not Present					
- Read SFN indicator	TRUE					
- CHOICE mode	FDD					
- Primary CPICH info						
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)"					
i mary coramoning oddo	in clause 6.1 of TS 34.108					
- Primary CPICH TX power	Not Present					
- TX Diversity indicator	FALSE					

The contents of SIB12 in cell 1, and SIB11 and SIB12 in cell 2 shall be in accordance with the detault SIBs as specified in TS 34.108.

MEASUREMENT REPORT (Step 2)

[Information Element	Value/remark
	Message Type	
	Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
	- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
	- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
	Measurement identity	1
	Measured Results	
	 Intra-frequency measured results 	
	- Cell measured results	
	- Cell Identity	Checked that this IE is absent
		Checked that this IE is absent
	 Cell synchronisation information Primary CPICH info 	Checked that this IE is absent
	- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)" in clause 6.1 of TS 34.108
	- CPICH Ec/N0	Checked that this IE is absent
	- CPICH RSCP	Checked that this IE is present
	- Pathloss	Checked that this IE is absent
	- Cell measured results	
	- Cell Identity	Checked that this IE is absent
		Checked that this IE is absent
	- Cell synchronisation information	Checked that this IE is present and includes IE COUNT- C-SFN frame difference
	- Primary CPICH info	
	- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108
	- CPICH Ec/N0	Checked that this IE is absent
	- CPICH RSCP	Checked that this IE is present
	- Pathloss	Checked that this IE is absent
	Measured results on RACH	Checked that this IE is absent
	Additional measured results	Checked that this IE is absent
	Event results	
	 Intra-frequency measurement event results 	
	 Intra-frequency event identity 	1a
	- Cell measurement event results	
	- Primary CPICH info	
	- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108

ACTIVE SET UPDATE (Step 3)

The message to be used in this test is defined in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
Radio link addition information	
- Primary CPICH Info	
- Primary Scrambling Code	Refer to clause titled "Default settings for cell
	No.2 (FDD)" in clause 6.1 of TS 34.108
 Downlink DPCH info for each RL 	
- CHOICE mode	FDD
 Primary CPICH usage for channel estimation 	P-CPICH can be used.
- DPCH frame offset	Calculated value from Cell synchronisation
	information
 Secondary CPICH info 	Not Present
- DL channelisation code	This IE is repeated for all existing downlink
	DPCHs allocated to the UE
 Secondary scrambling code 	Not Present
- Spreading factor	Refer to TS 34.108 clause 6.10.2.4 "Typical
	radio parameter sets"
- Code Number	For each DPCH, assign the same code
-	number in the current code given in cell 1.
- Scrambling code change	Not Present
- TPC Combination Index	0
- SSDT Cell Identity	Not Present
- Close loop timing adjustment mode	Not Present
- TFCI Combining Indicator	Not Present
- SCCPCH information for FACH	Not Present

MEASUREMENT REPORT (Step 5a)

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
Measurement identity	1
Measured Results	
 Intra-frequency measured results 	
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
- Cell synchronisation information	Checked that this IE is absent
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.1
	(FDD)" in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
 Cell synchronisation information 	Checked that this IE is absent
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)"
	in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
Measured results on RACH	Checked that this IE is absent
Additional measured results Event results	Checked that this IE is absent
- Intra-frequency measurement event results	
- Intra-frequency event identity	1b
- Cell measurement event results	
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)" in clause 6.1 of TS 34.108

MEASUREMENT REPORT (Step 9a)

The received message at this step should have the same contents as the message received in Step 6, with the following exceptions:

Information Element	Value/remark
Event results - Intra-frequency measurement event results - Intra-frequency event identity - Cell measurement event results - Primary CPICH info - Primary scrambling code	1a Refer to clause titled "Default settings for cell No.1 (FDD)" in clause 6.1 of TS 34.108

MEASUREMENT REPORT (Step 10a)

The received message at this step should have the same contents as the message received in Step 6, with the following exceptions:

Information Element	Value/remark
Event results - Intra-frequency measurement event results	
 Intra-frequency event identity Cell measurement event results Primary CPICH info 	1b
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108

8.3.4.1.5 Test requirement

After step 1 the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC.

After step 3 the UE shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC to acknowledge the completion of the active set additional procedure.

After step 5a the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC.

After step 6 the UE shall transmit a UE CAPABILITY INFORMATION message.

After step 9a the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC.

After step 10a the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC.

After step 11 the UE shall transmit a UE CAPABILITY INFORMATION message.

8.3.4.2 Active set update in soft handover: Radio Link removal

8.3.4.2.1 Definition

8.3.4.2.2 Conformance requirement

Upon reception of an ACTIVE SET UPDATE message the UE shall act upon all received information elements as specified in 8.6, unless specified otherwise in the following. The UE shall:

- 1> first add the RLs indicated in the IE "Radio Link Addition Information";
- 1> remove the RLs indicated in the IE "Radio Link Removal Information". If the UE active set is full or becomes full, an RL, which is included in the IE "Radio Link Removal Information" for removal, shall be removed before adding RL, which is included in the IE "Radio Link Addition Information" for addition;
- 1> perform the physical layer synchronisation procedure **B** as specified in TS 25.214;
- 1> set the IE "RRC transaction identifier" in the ACTIVE SET UPDATE COMPLETE message to the value of "RRC transaction identifier" in the entry for the ACTIVE SET UPDATE message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;
- 1> transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC without waiting for the <u>completion of the</u> Physical Layer synchronization <u>B</u>, <u>specified in TS25.214</u>;
- ...

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.2.3 Test purpose

1. To confirm that the UE continues to communicate with the SS on the remaining radio link after radio link removal on the active set.

2. To confirm that the UE is not using the removed radio link to communicate with the SS.

8.3.4.2.4 Method of test

Initial Condition

System Simulator: 2 cells - both Cell 1 and Cell 2 are active

UE: CS-DCCH+DTCH_DCH (state 6-9) or PS-DCCH+DTCH_DCH (state 6-10) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE.

Test Procedure

Table 8.3.4.2

Parameter	Unit	Cell 1				Cell 2			
		T0	T1	T2	T3	T0	T1	T2	T3
UTRA RF Channel Number		Ch. 1				Ch. 1			
CPICH Ec	dBm/3. 84MHz	-60	-60	-75	-60	-75	-60	-60	OFF

Table 8.3.4.2 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution.

At the start of the test, the UE goes to connected mode and establishes a radio access bearer service in the CELL_DCH state in cell 1.

SS configures its downlink transmission power settings according to columns "T1" in table 8.3.4.2. UE shall be triggered to transmit a MEASUREMENT REPORT message which includes the primary scrambling code for cell 2 according to IE "Intra-frequency event identity", which is set to '1a' in the SYSTEM INFORMATION BLOCK TYPE 11. After the MEASUREMENT REPORT message is received, the SS configures the new radio link to be added from cell 2 and then the SS transmits to the UE an ACTIVE SET UPDATE message in cell 1 on DCCH using AM RLC which includes the IE "Radio Link Addition Information" (e.g. Downlink DPCH information and other optional parameters relevant for the additional radio links with Primary CPICH info used for the reference ID).

When the UE receives this message, the UE shall configure layer 1 to begin reception without affecting the current uplink and downlink activities of existing radio links. The UE shall transmit an ACTIVE SET UPDATE COMPLETE message to the SS on the uplink DCCH using AM RLC.

SS configures its downlink transmission power settings according to columns "T2" in table 8.3.4.2. UE shall transmit a MEASUREMENT REPORT message which includes the primary scrambling code for cell 1 according to IE "Intrafrequency event identity", which is set to '1b' in the SYSTEM INFORMATION BLOCK TYPE 11. After the MEASUREMENT REPORT message is received, the SS remove the radio link from cell 1 and then SS transmits an ACTIVE SET UPDATE message, which includes IE "Radio Link Removal Information" and specifying the P-CPICH information of the cell to be removed.

When the UE receives this message, the UE RRC entity shall request UE L1 entity to terminate transmission and reception of the radio link from cell 1. Then the UE transmits an ACTIVE SET UPDATE COMPLETE message to the SS on the uplink DCCH using AM RLC.

SS shall transmit a UE CAPABILITY ENQUIRY message to confirm that the UE can respond this message through the DPCH in cell 2. The UE shall transmit a UE CAPABILITY INFORMATION message. Then SS transmits a UE CAPABILITY INFORMATION CONFIRM message.

SS configures its downlink transmission power settings according to columns "T3" in table 8.3.4.2 so as to generate a radio link failure condition. The UE shall detect the radio link failure UE shall re-select to cell 1 and transmit a CELL UPDATE message. SS transmits a CELL UPDATE CONFIRM message after it receive CELL UPDATE message from UE. Then the UE shall transmit an UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH to acknowledge the receipt of the new UE identities.

Expected sequence

Step	Direction		Message	Comment	
	UE	SS			
1				SS configures its downlink transmission power settings according to columns "T1" in	
				table 8.3.4.2	
2	\rightarrow		MEASUREMENT REPORT	See specific message contents for this message	
3	(ACTIVE SET UPDATE	SS transmits this message in cell 1 on downlink DCCH using AM RLC. The message includes IE "Radio Link Addition Information". (e.g. Downlink DPCH information and other optional parameters relevant for the additional radio links with Primary CPICH info used for the reference ID in cell 2)	
4	<i>→</i>		ACTIVE SET UPDATE COMPLETE	The UE shall configure a new radio link to cell 2, without interfering with existing connections on the radio link in cell 1.	
5				SS configures its downlink transmission power settings according to columns "T2" in table 8.3.4.2	
6	\rightarrow		MEASUREMENT REPORT	See specific message contents for this message	
7	+		ACTIVE SET UPDATE	The SS transmits this message on downlink DCCH using AM RLC which includes IE "Radio Link Removal Information".	
8	\rightarrow		ACTIVE SET UPDATE COMPLETE	The UE shall remove the radio link associated with cell 1.	
9	÷		UE CAPABILITY ENQUIRY	Use default message.	
10	\rightarrow		UE CAPABILITY INFORMATION	Use default message.	
11	÷		UE CAPABILITY INFORMATION CONFIRM	Use default message.	
12				SS configures its downlink transmission power settings according to columns "T3" in table 8.3.4.2	
13	3 >		CELL UPDATE	UE sends this message in cell 1.	
14	←		CELL UPDATE CONFIRM	See message content.	
15	\rightarrow		UTRAN MOBILITY INFORMATION CONFIRM		

Specific Message Contents

The contents of SIB11 broadcasted in cell 1 shall be in accordance with the default SIB11 as specified in section 6.1 of TS 34.108, with the following exceptions:

Nou intro fraguanay colla	
- New intra-frequency cells	4
- Intra-frequency cell id	1
- Cell info	
 Cell individual offset 	0dB
 Reference time difference to cell 	Not Present
- Read SFN indicator	TRUE
- CHOICE mode	FDD
- Primary CPICH info	100
- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)"
- Fillinary scrainbilling code	5
	in clause 6.1 of TS 34.108
 Primary CPICH TX power 	Not Present
- TX Diversity indicator	FALSE
 Intra-frequency cell id 	2
- Cell info	
- Cell individual offset	0dB
- Reference time difference to cell	Not Present
- Read SFN indicator	TRUE
- CHOICE mode	FDD
	FUU
- Primary CPICH info	
 Primary scrambling code 	Refer to clause titled "Default settings for cell No.2 (FDD)"
	in clause 6.1 of TS 34.108
 Primary CPICH TX power 	Not Present
- TX Diversity indicator	FALSE

The contents of SIB12 in cell 1, and SIB11 and SIB12 in cell 2 shall be in accordance with the detault SIBs as specified in TS 34.108.

MEASUREMENT REPORT (Step 2)

Γ	Information Element	Value/remark
ľ	Message Type	
	Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
	- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
	- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
	Measurement identity	1
	Measured Results	
	 Intra-frequency measured results 	
	 Cell measured results 	
	- Cell Identity	Checked that this IE is absent
		Checked that this IE is absent
	 Cell synchronisation information Primary CPICH info 	Checked that this IE is absent
	- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)" in clause 6.1 of TS 34.108
	- CPICH Ec/N0	Checked that this IE is absent
	- CPICH RSCP	Checked that this IE is present
	- Pathloss	Checked that this IE is absent
	 Cell measured results 	
	- Cell Identity	Checked that this IE is absent
		Checked that this IE is absent
	- Cell synchronisation information	Checked that this IE is present and includes IE COUNT- C-SFN frame difference
	- Primary CPICH info	
	- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108
	- CPICH Ec/N0	Checked that this IE is absent
	- CPICH RSCP	Checked that this IE is present
	- Pathloss	Checked that this IE is absent
	Measured results on RACH	Checked that this IE is absent
	Additional measured results	Checked that this IE is absent
	Event results	
	 Intra-frequency measurement event results 	
	 Intra-frequency event identity 	1a
	- Cell measurement event results	
	- Primary CPICH info	
	- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108

ACTIVE SET UPDATE (Step 3)

The message to be used in this test is defined in Annex.A, with the following exceptions:

Information Element	Value/remark
Radio link addition information	Value/ICIIIdIK
- Primary CPICH Info	Defende elevere fille d''Defende entire referere all
- Primary Scrambling Code	Refer to clause titled "Default settings for cell
	No.2 (FDD)" in clause 6.1 of TS 34.108
- Downlink DPCH info for each RL	
- CHOICE mode	FDD
 Primary CPICH usage for channel estimation 	P-CPICH can be used.
- DPCH frame offset	Calculated value from Cell synchronisation
	information
 Secondary CPICH info 	Not Present
 DL channelisation code 	This IE is repeated for all existing downlink
	DPCHs allocated to the UE
 Secondary scrambling code 	Not Present
- Spreading factor	Refer to TS 34.108 clause 6.10.2.4 "Typical
	radio parameter sets"
- Code Number	For each DPCH, assign the same code
	number in the current code given in cell 1.
- Scrambling code change	Not Present
- TPC Combination Index	0
- SSDT Cell Identity	Not Present
- Close loop timing adjustment mode	Not Present
- TFCI Combining Indicator	Not Present
- SCCPCH information for FACH	Not Present

MEASUREMENT REPORT (Step 6)

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
Measurement identity	1
Measured Results	
 Intra-frequency measured results list 	
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
- Cell synchronisation information	Checked that this IE is absent
- CHOICE mode	FDD
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)"
	in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
- Cell synchronisation information	Checked that this IE is absent
- CHOICE mode	FDD
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)"
	in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
Measured results on RACH	Checked that this IE is absent
Additional measured results	Checked that this IE is absent
Event results	
- CHOICE event result	Intra-frequency measurement event results
- Intra-frequency event identity	1b
- Cell measurement event results	
- CHOICE mode	FDD
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)" in clause 6.1 of TS 34.108

ACTIVE SET UPDATE (Step 7)

The message to be used in this test is the same as the message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
Radio link removal information	1 radio link to be removed
- Primary CPICH info - Primary scrambling code	Set to the same P-CPICH scrambling code assigned for cell 1

CELL UPDATE (Step 13)

The contents of CELL UPDATE message is identical as "Contents of CELL UPDATE message" as found in TS 34.108, clause 9 with the following exceptions:

Information Element	Value/remark		
Cell Update Cause	"radio link failure"		

CELL UPDATE CONFIRM (Step 14)

Use the same message sub-type found in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark				
New C-RNTI	'1010 1010 1010 1010'				

8.3.4.2.5 Test requirement

After step 1 the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC.

After step 3 the UE shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC to acknowledge the completion of the active set additional procedure.

After step 5 the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC.

After step 7 the UE shall remove the radio link from cell 1 and it shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC.

After step 10 the UE shall transmit a UE CAPABILITY INFORMATION message.

After step 12 the UE shall transmit a CELL UPDATE message on the CCCH with IE "Cell update cause" set to "radio link failure".

After step 14, the UE shall transmit a UTRAN MOBILITY INFORMATION CONFIRM message on the uplink DCCH using AM RLC.

8.3.4.3 Active set update in soft handover: Combined radio link addition and removal

8.3.4.3.1 Definition

8.3.4.3.2 Conformance requirement

Upon reception of an ACTIVE SET UPDATE message the UE shall act upon all received information elements as specified in 8.6, unless specified otherwise in the following. The UE shall:

- 1> first add the RLs indicated in the IE "Radio Link Addition Information";
- 1> remove the RLs indicated in the IE "Radio Link Removal Information". If the UE active set is full or becomes full, an RL, which is included in the IE "Radio Link Removal Information" for removal, shall be removed before adding RL, which is included in the IE "Radio Link Addition Information" for addition;
- 1> perform the physical layer synchronisation procedure <u>B</u> as specified in TS 25.214;
- 1> set the IE "RRC transaction identifier" in the ACTIVE SET UPDATE COMPLETE message to the value of "RRC transaction identifier" in the entry for the ACTIVE SET UPDATE message in the table "Accepted transactions" in the variable TRANSACTIONS; and
- 1> clear that entry;
- 1> transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH using AM RLC without waiting for the <u>completion of the</u> Physical Layer synchronization <u>B</u>, specified in TS25.214;

•••

Reference

3GPP TS 25.331 clause 8.3.4

8.3.4.3.3 Test purpose

1. To confirm that the UE continues to communicate with the SS on the added radio link and removes radio link which exists prior to the execution of active set update procedure.

8.3.4.3.4 Method of test

Initial Condition

System Simulator: 3 cells- Cell 1, Cell 2 and Cell 3 are active, with downlink transmission power settings according to columns "T0" in table 8.3.4.3.

UE: CS-DCCH+DTCH_DCH (state 6-9) or PS-DCCH+DTCH_DCH (state 6-10) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE [Active set is not full.]

Table 8.3.4.3

Test Procedure

Parameter	Unit	Cell 1				Cell 2				Cell 3						
		T0	T1	T2	T3	T4	T0	T1	T2	T3	T4	T0	T1	T2	T3	T4
UTRA RF Channel Number		Ch. ′	1				Ch. 1					Ch. 1				
CPICH Ec	dBm/ 3.84 MHz	-60	-60	-60	OFF	-60	-80	-60	-60	OFF	-70	-80	-80	-60	-60	OFF

Table 8.3.4.3 illustrates the downlink power to be applied for the 3 cells at various time instants of the test execution.

The UE goes to connected mode and establishes a radio access bearer in the CELL_DCH state in cell 1.

SS configures its downlink transmission power settings according to columns "T1" in table 8.3.4.3. UE transmits a MEASUREMENT REPORT message which includes the primary scrambling code for cell 2 according to IE "Intrafrequency event identity", which is set to '1a' in the SYSTEM INFORMATION BLOCK TYPE 11. After the MEASUREMENT REPORT message is received, the SS configures the new radio link to be added from cell 2 and then the SS transmits to the UE in cell 1 an ACTIVE SET UPDATE message which includes IE "Radio Link Addition Information", indicating the addition of cell 2 into the active set, on DCCH using AM RLC.

When the UE receives this message, the UE shall configure layer 1 to begin reception without affecting the current uplink and downlink activities of existing radio links. The UE shall transmit an ACTIVE SET UPDATE COMPLETE message to the SS on the uplink DCCH using AM RLC.

SS configures its downlink transmission power settings according to columns "T2" in table 8.3.4.3. UE shall be triggered to transmit a MEASUREMENT REPORT message which includes the primary scrambling code for cell 3 according to IE "Intra-frequency event identity", which is set to '1a' in the SYSTEM INFORMATION BLOCK TYPE 11. After the MEASUREMENT REPORT message is received, the SS configures the new radio link to be added from cell 3 and then the SS transmits to the UE an ACTIVE SET UPDATE message which includes IE "Radio Link Addition Information" and IE "Radio Link Removal Information", indicating the removal of cell 2 and addition of cell 3 into the active set, on DCCH using AM RLC.

When the UE receives this message, the UE shall configure layer 1 to begin reception without affecting the current uplink and downlink activities of existing radio links and then the UE removes the radio link specified in an ACTIVE SET UPDATE message. The UE shall transmit an ACTIVE SET UPDATE COMPLETE message to the SS on the uplink DCCH using AM RLC.

SS configures its downlink transmission power settings according to columns "T3" in table 8.3.4.3.

SS shall transmit a UE CAPABILITY ENQUIRY message to confirm that the UE can respond to this message through the DPCH in cell 3. The UE shall transmit a UE CAPABILITY INFORMATION message. Then SS transmits a UE CAPABILITY INFORMATION CONFIRM message.

SS configures its downlink transmission power settings according to columns "T4" in table 8.3.4.3.

SS shall transmit a UE CAPABILITY ENQUIRY message to confirm that the UE can respond to this message through the DPCH in cell 1. The UE shall transmit a UE CAPABILITY INFORMATION message. Then SS transmits a UE CAPABILITY INFORMATION CONFIRM message.

Expected sequence

Step	Direction		Message	Comment			
_	UE	SS					
0a				SS configures the initial active set with only cell 1. SS configures its downlink transmission power settings according to columns "T1" in table 8.3.4.3			
0b	-		MEASUREMENT REPORT	See specific message contents for this message			
0c	0c ←		ACTIVE SET UPDATE	The SS transmit this message on downlink DCCH using AM RLC which includes IE "Radio Link Addition Information" for cell 2.			
0d		>	ACTIVE SET UPDATE COMPLETE	The UE adds the radio link in cell 2.			
1				SS configures its downlink transmission power settings according to columns "T2" in table 8.3.4.3			
2	-	>	MEASUREMENT REPORT	See specific message contents for this message			
3	÷	-	ACTIVE SET UPDATE	The SS transmit this message on downlink DCCH using AM RLC which includes IE "Radio Link Addition Information" for cell 3 and IE "Radio Link Removal Information" for cell 2.			
4		>	ACTIVE SET UPDATE COMPLETE	The UE shall configure a new radio link in cell 3 and removes the old radio link in cell 2.			
4a				SS configures its downlink transmission power settings according to columns "T3" in table 8.3.4.3			
5	←		UE CAPABILITY ENQUIRY	Use default message.			
6	\rightarrow		UE CAPABILITY INFORMATION	Use default message.			
7	(UE CAPABILITY INFORMATION CONFIRM	Use default message.			
8				SS configures its downlink transmission power settings according to columns "T4" in table 8.3.4.3			
9		-	UE CAPABILITY ENQUIRY	Use default message.			
10	-		UE CAPABILITY INFORMATION	Use default message.			
11	€		UE CAPABILITY INFORMATION CONFIRM	Use default message.			

Specific Message Content

The contents of SIB11 broadcast- in cell 1 and cell 2 shall be in accordance with the default SIB11 as specified in section 6.1 of TS 34.108, with the following exception:

- New intra-frequency cells	
- Intra-frequency cell id	1
- Cell info	•
- Cell individual offset	0dB
- Reference time difference to cell	Not Present
- Read SFN indicator	TRUE
- CHOICE mode	FDD
- Primary CPICH info	FDD
5	Poter to clauge titled "Default acttings for call No.1 (EDD)"
- Primary scrambling code	Refer to clause titled "Default settings for cell No.1 (FDD)" in clause 6.1 of TS 34.108
 Primary CPICH TX power 	Not Present
 TX Diversity indicator 	FALSE
 Intra-frequency cell id 	2
- Cell info	
 Cell individual offset 	0dB
 Reference time difference to cell 	Not Present
 Read SFN indicator 	TRUE
- CHOICE mode	FDD
- Primary CPICH info	
 Primary scrambling code 	Refer to clause titled "Default settings for cell No.2 (FDD)"
	in clause 6.1 of TS 34.108
 Primary CPICH TX power 	Not Present
 TX Diversity indicator 	FALSE
 Intra-frequency cell id 	3
- Cell info	
 Cell individual offset 	0dB
 Reference time difference to cell 	Not Present
 Read SFN indicator 	TRUE
- CHOICE mode	FDD
- Primary CPICH info	
 Primary scrambling code 	Refer to clause titled "Default settings for cell No.3 (FDD)"
	in clause 6.1 of TS 34.108
 Primary CPICH TX power 	Not Present
- TX Diversity indicator	FALSE

The contents of SIB12 in cell 1 and cell 2, and SIB11 and SIB12 in cell 23 shall be in accordance with the detault SIBs as specified in TS 34.108.

MEASUREMENT REPORT (Step 0b)

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
Measurement identity	1
Measured Results	
 Intra-frequency measured results 	
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
 Cell synchronisation information Primary CPICH info 	Checked that this IE is absent
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
- Cell synchronisation information	Checked that this IE is present and includes IE COUNT- C-SFN frame difference
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
Measured results on RACH	Checked that this IE is absent
Additional measured results	Checked that this IE is absent
Event results	Checked that this IE is absent
- Intra-frequency measurement event results	
- Intra-frequency event identity	1a
- Cell measurement event results	
- Primary CPICH info	Refer to clause titled "Default pattings for call No.2 (CDD)"
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108

ACTIVE SET UPDATE (Step 0c)

The message to be used in this test is defined in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
Radio link addition information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 2
- Downlink DPCH info for each RL	Calculated value from Cell synchronisation information
- DPCH frame offset	

MEASUREMENT REPORT (Step 2)

Information Element	Value/remark
Message Type	
Integrity check info	The presence of this IE is dependent on IXIT statements in TS 34.123-2. If integrity protection is indicated to be active, this IE shall be present with the values of the sub IEs as stated below. Else, this IE and the sub-IEs shall be absent.
- Message authentication code	This IE is checked to see if it is present. The value is compared against the XMAC-I value computed by SS.
- RRC Message sequence number	This IE is checked to see if it is present. The value is used by SS to compute the XMAC-I value.
Measurement identity	1
Measured Results	
 Intra-frequency measured results 	
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
 Cell synchronisation information Primary CPICH info 	Checked that this IE is absent
- Primary scrambling code	Refer to clause titled "Default settings for cell No.3 (FDD)" in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
- Cell measured results	
- Cell Identity	Checked that this IE is absent
	Checked that this IE is absent
- Cell synchronisation information	Checked that this IE is present and includes IE COUNT- C-SFN frame difference
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.2 (FDD)" in clause 6.1 of TS 34.108
- CPICH Ec/N0	Checked that this IE is absent
- CPICH RSCP	Checked that this IE is present
- Pathloss	Checked that this IE is absent
Measured results on RACH	Checked that this IE is absent
Additional measured results	Checked that this IE is absent
Event results	Checked that this IE is absent
- Intra-frequency measurement event results	
- Intra-frequency event identity	1a
 Cell measurement event results 	
- Primary CPICH info	
- Primary scrambling code	Refer to clause titled "Default settings for cell No.3 (FDD)" in clause 6.1 of TS 34.108

ACTIVE SET UPDATE (Step 3)

The message to be used in this test is defined in TS 34.108, clause 9, with the following exceptions:

Information Element	Value/remark
Radio link addition information	Valuo/romark
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 3
- Downlink DPCH info for each RI	Out to same code as assigned for cents
- CHOICE mode	FDD
- Primary CPICH usage for channel estimation	P-CPICH can be used.
- DPCH frame offset	Calculated value from Cell synchronisation
	information
- Secondary CPICH info	Not Present
- DL channelisation code	This IE is repeated for all existing downlink
	DPCHs allocated to the UE
- Secondary scrambling code	Not Present
- Spreading factor	Refer to TS 34.108 clause 6.10.2.4 "Typical
	radio parameter sets"
- Code Number	For each DPCH, assign the same code
	number in the current code given in cell 1.
 Scrambling code change 	Not Present
- TPC Combination Index	0
- SSDT Cell Identity	Not Present
 Close loop timing adjustment mode 	Not Present
- TFCI Combining Indicator	Not Present
 SCCPCH information for FACH 	Not Present
Radio link removal information	
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as assigned for cell 2

8.3.4.3.5 Test requirement

At step 0a the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC. After step 0c the UE shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH. After step 1 the UE shall transmit a MEASUREMENT REPORT message on the uplink DCCH using AM RLC. After step 3 the UE shall transmit an ACTIVE SET UPDATE COMPLETE message on the uplink DCCH. After step 5 the UE shall transmit a UE CAPABILITY INFORMATION message on the uplink DCCH in cell 3. After step 9 the UE shall transmit a UE CAPABILITY INFORMATION message on the uplink DCCH in cell 1.

----<End of Modifications>---

3GPP TSG-T1 Meeting #17 T1-020789 Luton, UK, 4th – 8th Nov 2002 3GPP TSG-T1/SIG Meeting #26 Luton, UK, 4th – 8th Nov 2002 T1S020870 CR-Form-v7 **CHANGE REQUEST** # Current version: **5.1.1** ж 34.123-1 CR 341 ж жrev For **HELP** on using this form, see bottom of this page or look at the pop-up text over the **#** symbols. ME X Radio Access Network Core Network **Proposed change affects:** UICC apps# # CR to 34.123-1 clause 8.3 (Package 1) Rel-5: Correction from CRs approved in Title: RP17meeting (Revision to T1S020737) Source: **#** Panasonic Date: # 28/10/2002 Work item code: # TEI ਸ਼ F Category: Release: # REL-5

Use <u>one</u> of the following categories:	Use <u>one</u> o	f the following releases:
F (correction)	2	(GSM Phase 2)
A (corresponds to a correction in an earlier release) R96	(Release 1996)
B (addition of feature),	R97	(Release 1997)
C (functional modification of feature)	R98	(Release 1998)
D (editorial modification)	R99	(Release 1999)
Detailed explanations of the above categories can	Rel-4	(Release 4)
be found in 3GPP <u>TR 21.900</u> .	Rel-5	(Release 5)
	Rel-6	(Release 6)

Reason for change: ೫	 From CR 1624 The current semantics description states that under only one transmission gap in the transmission gap pattern, TGD should be set to "0", but this is not a possible value for this parameter.
	 From CR1573 UE report of "SFN-SFN observed time difference" measurement in Intra- and Inter-frequency measurement report is not needed from a functional point of view.
	 From CR 1541 The handling of the "UE internal measurement system information" IE which the UE may receive in SIB11/SIB12 is currently unclear in the specifications.
	4. Alignments with T1S020726/727.
Summary of change: ೫	1.Change to 8.4.1.3
	The value of TGD shall be revised from "0" to "undefined" under this situation.
	<mark>2.1. C</mark> hange to 8.4.1.3, 8.4.1.5

	IE "SFN-SFN observed time difference" and IE "SFN-SFN observed time difference reporting indicator" is deleted.	
	3.2. Change to 8.4.1.3,8.4.1.5 Remove IE "UE internal Measurement System Information" from IE" Measurement control system information"	
	Revision to T1S020737	
	1. Specific Message Contents of SIB 11/12 is aligned with T1S020726/727	
	in several instances.2. Correction to cover page: correction to IE "TGD" does not exist in this CR,	
	but was mentioned in the cover page. Such descriptions are removed.	
Consequences if #	The test specifications are not aligned with the core specification	
not approved:		
Clauses affected: #	8.4.1.3, 8.4.1.5	
Other specs #	Y N X Other core specifications	
Affected:	X Test specifications	
	X O&M Specifications	
Other commonto		
Other comments: #	Affects R99, REL-4, REL-5	

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

---<Start of Modifications>---

- 8.4.1.3 Measurement Control and Report: Intra-frequency measurement for transition from idle mode to CELL_FACH state
- 8.4.1.3.1 Definition

8.4.1.3.2 Conformance requirement

Upon transition from idle mode to CELL_FACH state, the UE shall:

1> begin or continue monitoring cells listed in the IE "intra-frequency cell info list" received in System Information Block type 12 (or System Information Block type 11).

In CELL_FACH state, the UE shall:

- 1> include a measurement report in the IE "Measured results on RACH", as specified in the IE "Intra-frequency reporting quantity for RACH reporting" and the IE "Maximum number of reported cells on RACH" in System Information Block type 12 (or "System Information Block Type 11" if "System Information Block Type 12" is not being broadcast);
- 1> include in the IE "Measured results on RACH" all requested reporting quantities for cells for which measurements are reported.

Upon transition from CELL_FACH to CELL_DCH state, the UE shall:

- 1> retrieve each set of measurement control information of measurement type "intra-frequency" stored in the variable MEASUREMENT_IDENTITY;
- 1> if the IE "measurement validity" for a measurement has been assigned the value "CELL_DCH:

2> resume the measurement reporting.

- 1> if no intra-frequency measurements applicable to CELL_DCH state are stored in the variable MEASUREMENT_IDENTITY:
 - 2> continue monitoring the list of neighbouring cells assigned in the IE "intra-frequency cell info list" in System Information Block type 12 (or System Information Block type 11);
 - 2> if the IE "intra-frequency measurement reporting criteria" was included in System Information Block type 12 (or System Information Block type 11):
 - 3> send the MEASUREMENT REPORT message when reporting criteria in IE "Reporting information for state CELL_DCH" are fulfilled.

Reference

3GPP TS 25.331, clause 8.4.1.9.1, 8.4.1.7.1, 8.4.2.2.

8.4.1.3.3 Test Purpose

- 1. To confirm that the UE begins or continues to monitor cells listed in IE "intra-frequency cell info list" of System Information Block type 11 or 12 messages after it has entered CELL_FACH state from idle mode.
- 2. To confirm that the UE applies the reporting criteria stated in "intra-frequency measurement reporting criteria" IE in System Information Block Type 11 or 12 in a subsequent transition to CELL_DCH state.
- 3. To confirm that the UE reports measured results on RACH messages, if it receives IE "Intra-frequency reporting quantity for RACH reporting" and IE "Maximum number of reported cells on RACH" from System Information Block Type 11 or 12 upon a transition from idle mode to CELL_FACH state.

8.4.1.3.4 Method of test

Initial Condition

System Simulator: 2 cells. Cell 1 and cell 2 are active.

UE: "Registered idle mode on PS" (state 3) in cell 1 as specified in clause 7.4 of TS 34.108. If the UE supports both CS and PS domains, the initial UE state shall be "Registered idle mode on CS/PS" (state 7).

Test Procedure

Table 8.4.1.3-1 illustrates the downlink power to be applied for the 2 cells in this test case.

Parameter	Unit	Cell 1	Cell 2
UTRA RF		Ch. 1	Ch. 1
Channel Number			
CPICH Ec	dBm/	-60	-70
	3.84		
	MHz		

Table 8.4.1.3-1

The UE is initially in idle mode and camps on cell 1. The System Information Block type 11 are modified compared to the default settings to prevent reporting of "Cell synchronisation information" and also to include cell 2 into the IE "intra-frequency cell info list". The key measurement parameters are as follow: measurement type = "intra-frequency measurement", measurement quantity = "CPICH RSCP", reporting mode = "event reporting". In the System Information Block type 11 messages, reporting of CPICH RSCP is also required for intra-frequency reporting when transmitting RACH messages on cell 1.

SS prompts the operator to make an outgoing call for one of the traffic classes supported by the UE. SS and UE shall execute procedure P6. Next SS and UE shall execute procedure P10. Then SS and UE shall execute procedure P14. SS starts timer T305 and waits until timer T305 expires, the UE shall send a CELL UPDATE message on the CCCH which includes the measured value of cell 1's CPICH RSCP in IE "Measured results on RACH". SS then replies with CELL UPDATE CONFIRM message on the downlink DCCH, without changing the physical channel resources.

SS transmits PHYSICAL CHANNEL RECONFIGURATION message, and allocates dedicated physical channels to the UE. The UE shall transit to CELL_DCH state and then send a MEASUREMENT REPORT message, correctly stating the measurement identity. The measurement identity indicated shall match the value that was previously broadcast on System Information Block type 11 messages when the UE was still in idle mode. The IE "Measured results" in the MEASUREMENT REPORT messages shall contain measured values of cell 2's CPICH RSCP.

Expected Sequence

Step	Direction	Message	Comment
-	UE SS	7	
1	÷	System Information Block type 1, System Information Block type 11	The UE is in idle mode and camps onto cell 1. System Information Block type 1 and 11 to be transmitted are different from the default settings (see specific message contents)
2	\leftrightarrow	SS executes procedure P6 (clause 7.4.2.2.2) specified in TS 34.108.	SS prompts the test operator to make an outgoing call.
3	\leftrightarrow	SS executes procedure P10 (clause 7.4.2.4.2) specified in TS 34.108.	
4	\leftrightarrow	SS executes procedure P14 (clause 7.4.2.6.2) specified in TS 34.108.	
5		Void	

			1
6			SS monitors the uplink DCCH to confirm that no
			MEASUREMENT REPORT
			messages are detected. SS
			waits for 5 minutes (for the
			expiry of T305 timer).
7	\rightarrow	CELL UPDATE	This message shall contain IE
			"Measured results on RACH"
			reporting the measured CPICH RSCP for cell 1.
8	←	CELL UPDATE CONFIRM	SS does not change the
0	, ,		physical channel
			configurations.
9	~	PHYSICAL CHANNEL	SS assigns dedicated physical
		RECONFIGURATION	resources.
10	\rightarrow	PHYSICAL CHANNEL	UE shall transit to CELL_DCH
		RECONFIGURATION COMPLETE	state.
11	\rightarrow	MEASUREMENT REPORT	LIF abolt bogin to report call
11	7		UE shall begin to report cell 2's CPICH RSCP value
			periodically at 16 seconds
			interval. The measurement
			identity shall match the one
			that is broadcast for use in
			CELL_DCH in SIB11 in step 1.

Specific Message Content

System Information Block type 1 (Step 1)

Information Element	Value/Remarks
UE Timers and constants in connected mode	
- T305	5 minutes.

System Information Block type 11 (Step 1)

	· · · · ·
Information Element	Value/remark
SIB12 indicator	FALSE
FACH measurement occasion info	Not Present
Measurement control system information	
- Use of HCS	Not used
- Cell selection and reselection quality measure	CPICH RSCP
- Intra-frequency measurement system information	-
- Intra-frequency measurement identity	5
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Remove no intra-frequency cellsNot Present
- New intra-frequency cells	
- Intra-frequency cell id	1
- Cell info	
- Cell individual offset	0 dBNot Present
- Reference time difference to cell	Not present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1 Refer to clause
	titled "Default settings for cell No.1 (FDD)" in clause
	6.1.4 of TS34.108
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	Not present
- Intra-frequency cell id	2
- Cell info	
- Cell individual offset	0 dBNot Present
- Reference time difference to cell	Not Present
- Read SFN Indicator	TRUE
- CHOICE mode	FDD
- Primary CPICH Info	Oat to some code on wood for call 0. Defeate de
- Primary Scrambling Code	Set to same code as used for cell 2 Refer to clause
	titled "Default settings for cell No.2 (FDD)" in clause
	6.1.4 of TS34.108
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	
- Qoffset1 _{s,n}	Not Present (Default is 0 dB)
- Qoffset2 _{s,n}	Not Present
- Maximum allowed UL TX power	0 dBm
- HCS neighbouring cell information	Not Present
- CHOICE Mode	FDD
- Qqualmin	-20dB
- Qrxlevmin	-115dBm
- Cells for measurement	Not Present
 Intra-frequency Measurement quantity Filter Coefficient 	ONot Procent
- Fliter Coefficient - CHOICE Mode	<mark>9Not Present</mark> FDD
 Measurement quantity Intra-frequency reporting quantity measurement for 	CPICH RSCP
RACH reporting - SFN-SFN observed time difference reporting	No report
indicator	No report
- CHOICE mode	FDD
Reporting quantity	CPICH RSCP
- Maximum number of reported cells on RACH	Current cell
- Reporting information for state CELL_DCH	
- Reporting information for state CELL_DCH - Intra-frequency reporting quantity	
- Reporting quantities for active set cells	
- Reporting quantities for active set cells SFN-SFN observed time difference reporting	No report
SEN-SEN observed time difference reporting	No report
	FALSE
- Cell synchronisation information reporting indicator	
	FALSE
- Cell identity reporting indicator - CHOICE mode	FDD
- CPICH Ec/No reporting indicator	FALSE

Release 5

- CPICH RSCP reporting indicator	FALSE
 Pathloss reporting indicator 	FALSE
 Reporting quantities for monitored set cells 	
 SFN-SFN observed time difference reporting 	No report
indicator	
 Cell synchronisation information reporting 	TRUE
indicator	
 Cell identity reporting indicator 	FALSE
- CHOICE mode	FDD
 CPICH Ec/No reporting indicator 	FALSE
 CPICH RSCP reporting indicator 	TRUE
 Pathloss reporting indicator 	FALSE
 Reporting quantities for detected set cells 	Not present
- Measurement Reporting Mode	
- Measurement Reporting Transfer Mode	Acknowledged mode RLC
- Periodic Reporting/Event Trigger Reporting Mode	Event trigger
- CHOICE report criteria	Intra-frequency measurement reporting criteria
- Parameters required for each event	
- Intra-frequency event identity	1a
- Triggering condition 1	Not Present
- Triggering condition 2	Monitored set cells
- Reporting Range Constant	15 dB
- Cells forbidden to affect reporting range	Not Present
- W	0.0
- Hysteresis	1.0 dB
- Threshold used frequency	Not Present
- Reporting deactivation threshold	0
- Replacement activation threshold	Not Present
- Time to trigger	60 ms
- Amount of reporting	Infinity
- Reporting interval	16 seconds
- Reporting Cell Status	
- CHOICE reported cell	Report cells within active and/or monitored set on used
	frequency or within active and/or monitored set on non-
	used frequency
- Maximum number of reported cells	2
- Inter-frequency measurement system information	Not Present
- Traffic volume measurement system information	Not Present
UE internal measurement system information	Not Present

CELL UPDATE (Step 7)

Information Element	Value/remark
U-RNTI	Check to see if set to same U-RNTI value assigned in
	the execution of procedure P6.
START list	Checked to see if this IE is present
AM_RLC error indication(RB2, RB3 or RB4)	FALSE
AM_RLC error indication(RB>4)	FALSE
Cell update cause	Check to see if set to 'Periodical cell update'
Failure cause	Check to see if this IE is absent
Measured results on RACH	
 Measurement result for current cell 	
 CHOICE measurement quantity 	Check to see if set to 'CPICH RSCP'
- CPICH RSCP	Checked to see if set to within an acceptable range.
- Measurement results for monitored cells	Checked to see if this IE is absent.

PHYSICAL CHANNEL RECONFIGURATION (Step 9)

Use the same message sub-type found in [9] TS 34.108 clause 9, which is entitled "Packet to CELL_DCH from CELL_FACH".

MEASUREMENT REPORT (Step 11)

Information Element	Value/remark
Measurement identity	Check to see if set to 5
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results
	list"
- Intra-frequency measurement results	
- Cell measured results	
- Cell Identity	Check to see if it is absent
- SEN-SEN observed time difference	Check to see if this IF is absent
- Cell synchronisation information	Check to see if this IE is present and if the reported cell
	synchronisation information is correct
- Primary CPICH Info	-,
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
- Cell synchronisation information	Check to see if this IE is absent
- Primary CPICH Info	Check to see if it's the same code for cell 1
- Primary Scrambling Code	Check to see if this IE is absent
- CPICH Ec/No	Check to see if this IE is present
- CPICH RSCP	Check to see if this IE is absent
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event Results	Charly to one if not to lintre frequency measurement
- CHOICE event result	Check to see if set to "Intra-frequency measurement event results"
- Intra-frequency event identity	Check to see if set to "1a"
- Cell measurement event results - CHOICE Mode	Check to see if set to "FDD"
- Primary CPICH info	Check to see if set to the scrambling code of cell 2
- Primary Scrambling Code	Check to see if set to the scrambling code of cell 2

8.4.1.3.5 Test Requirement

After step 5 the UE shall not transmit any MEASUREMENT REPORT messages on the uplink DCCH.

After step 6 the UE shall initiate cell update procedure by transmitting CELL UPDATE message on CCCH. In this message, IE "cell update cause" shall be set to "periodic cell update". It shall include IE "measured results on RACH", containing the measurement value for cell 1's CPICH RSCP.

After step 10 the UE shall transmit MEASUREMENT REPORT messages at 16 seconds interval. In these messages, cell 2's CPICH RSCP value shall be reported in IE "Measured results". The IE "measurement identity" in this message shall match the IE "Intra-frequency measurement identity" found in System Information Block type 11 messages transmitted in step 1. The MEASUREMENT REPORT messages shall also contain IE "Event results", indicating that intra-frequency event "1a" has triggered in the UE.

----<End of Modifications>---

----<Start of Modifications>----

- 8.4.1.5 Measurement Control and Report: Intra-frequency measurement for transition from CELL_DCH to CELL_FACH state
- 8.4.1.5.1 Definition

8.4.1.5.2 Conformance requirement

Upon transition from CELL_DCH to CELL_FACH/CELL_PCH/URA_PCH state, the UE shall:

- 1> stop intra-frequency type measurement reporting;
- 1> if the transition is due to a reconfiguration message which included the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD), and the UE selects a cell other than that indicated by this IE; or
- 1> if the transition is due to a reconfiguration message which does not include the IE "Primary CPICH info" (for FDD) or "Primary CCPCH info" (for TDD); or
- 1> if the transition is not due to a reconfiguration message:
 - 2> delete the measurements of type intra-frequency associated with the variable MEASUREMENT_IDENTITY.
- 1> begin monitoring cells listed in the IE "intra-frequency cell info list" received in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11 in [8] TS 25.331).

Upon transition from CELL_FACH to CELL_DCH state, the UE shall:

- 1> retrieve each set of measurement control information of measurement type "intra-frequency" stored in the variable MEASUREMENT_IDENTITY;
- 1> if the IE "measurement validity" for a measurement has been assigned the value "CELL_DCH:

2> resume the measurement reporting.

- 1> if no intra-frequency measurements applicable to CELL_DCH state are stored in the variable MEASUREMENT_IDENTITY:
 - 2> continue monitoring the list of neighbouring cells assigned in the IE "intra-frequency cell info list" in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11 in [8] TS 25.331);
 - 2> if the IE "intra-frequency measurement reporting criteria" was included in System Information Block type 12 (or System Information Block type 11, according to subclause 8.1.1.6.11 in [8] TS 25.331):
 - 3> send the MEASUREMENT REPORT message when reporting criteria in IE "Reporting information for state CELL_DCH" are fulfilled.

Reference

3GPP TS 25.331, clause 8.4.1.6.1, 8.4.1.7.1

8.4.1.5.3 Test Purpose

- 1. To confirm that the UE stops performing intra-frequency measurement reporting specified in a MEASUREMENT CONTROL message, when it moves from CELL_DCH state to CELL_FACH state.
- 2. To confirm that the UE reads the System Information Block type 11 or 12 messages when it enters CELL_FACH state from CELL_DCH state, and starts to monitor the cells listed in the IE "intra-frequency cell info list".

- 3 To confirm that the UE performs measurements on uplink RACH transmissions and appends the measured results in RACH messages, when it receives IE "intra-frequency reporting quantity for RACH reporting" and IE "Maximum number of reported cells on RACH" in the System Information Block type 11 or 12 messages.
- 4. To confirm that the UE applies the reporting criteria in IE "intra-frequency reporting criteria" in System Information Block Type 11 or 12 messages following a state transition from CELL_FACH to CELL_DCH, if no intra-frequency measurements applicable to CELL_DCH are stored.

8.4.1.5.4 Method of test

Initial Condition

System Simulator: 3 cells - Cell 1 and cell 2 are active, while cell 3 is switched off..

UE: PS-DCCH+DTCH_DCH (state 6-10) in cell 1 as specified in clause 7.4 of TS 34.108.

Test Procedure

Table 8.4.1.5-1 illustrates the downlink power to be applied for the 3 cells at various time instants of the test execution. Columns marked "T0" denote the initial conditions, while columns marked "T1 are to be applied subsequently. The exact instants on which these values shall be applied are described in the texts in this clause.

Parameter	Unit	Ce	ll 1	Ce	ll 2	Ce	II 3
		T0	T1	T0	T1	T0	T1
UTRA RF Channel Number		Ch	i. 1	Ch	i. 1	Ch	i. 1
CPICH Ec	dBm/ 3.84 MHz	-60	-60	-75	-85	-122	-70

Table 8.4.1.5-1

The UE is initially in CELL_DCH state. The System Information Block type 11 message is modified compared to the default message contents, in order to prevent the reporting of "Cell synchronisation information". No measurement to be applied by the UE in CELL_DCH state is specified in any of the System Information Block type 11 or 12 messages.

SS sends a MEASUREMENT CONTROL message to UE. In this message, the SS requests the establishment of an intra-frequency measurement for the measurement of cell 2's CPICH RSCP. At the same time, reporting of CPICH RSCP values of active set cells and monitored set cells are requested with the reporting criteria set to "periodic reporting" and "reporting interval" set to 16 seconds. The UE shall start transmitting MEASUREMENT REPORT messages at 16 seconds interval corresponding to the requested reporting event.

SS transmits PHYSICAL CHANNEL RECONFIGURATION message to move the UE to CELL_FACH. After receiving this message, the UE shall reconfigure itself and reply with a PHYSICAL CHANNEL

RECONFIGURATION COMPLETE message on RACH. SS monitors the uplink channels to verify that no MEASUREMENT REPORT messages are received.

SS reconfigures itself according to the settings in columns marked "T1" in table 8.4.1.5-1. SS transmits System Information Block type 12 messages in cell 1, which include cell 3 into the IE "intra-frequency cell info list" and modifies SIB11 to indicate that SIB12 is now being broadcast. IEs "Intra-frequency reporting quantity for RACH Reporting" and IE "Maximum number of Reported cells on RACH" are also specified in the System Information Type 12 messages. Event type 1a reporting criterion is specified for intra-frequency measurements. SS transmit SYSTEM INFORMATION CHANGE INDICATION message to UE. SS waits until T305 has expired. The UE shall respond with a CELL UPDATE message, which comprises IE "Measured results on RACH" to report the readings of CPICH RSCP for cell 1 and cell 3. SS replies with CELL UPDATE CONFIRM message on the downlink DCCH. This message does not change the physical resources nor allocate any new RNTI identities. SS transmits PHYSICAL CHANNEL RECONFIGURATION message again, and configures dedicated physical channel for both uplink and downlink directions. The UE shall send PHYSICAL CHANNEL RECONFIGURATION COMPLETE message and return to CELL_DCH state. SS listens to the uplink DCCH for MEASUREMENT REPORT messages.

SS shall receive the MEASUREMENT REPORT messages at 500 milliseconds interval.

SS verifies that it includes CPICH RSCP values of the cells 1, 2 and 3 in IE "Cell measured results" and the triggering of event '1a' on cell 3 in IE "Event results".

Step	Direction	Message	Comment
	UE SS		
1	÷	Master Information Block System Information Block type 11	UE is in PS- DCCH+DTCH_DCH (state 6- 10) in cell 1. System Information Block Type 11 to be broadcast does not specify any measurement type to be configured in the UE in CELL_DCH.
2		Void	
3		Void	
4		Void	
5	÷	MEASUREMENT CONTROL	SS requests for measurement of cell 2's CPICH RSCP value and reporting of CPICH RSCP values of active cells and monitored set cells.
6	\rightarrow	MEASUREMENT REPORT	UE shall send periodic report at 16 seconds interval.
7	÷	PHYSICAL CHANNEL RECONFIGURATION	SS moves the UE to CELL_FACH state.
8	\rightarrow	PHYSICAL CHANNEL RECONFIGURATION COMPLETE	UE shall move to CELL_FACH state.
9	÷	Master Information Block System Information Block type 12	SS reconfigures itself according to the settings stated in column "T1" of table 8.4.1.5-1. SIB 11 is modified to indicate that SIB12 is now broadcast and to add cell 2 as a neighbour cell. SIB 12 indicates that cell 3 is included in the IE "intra-frequency cell info list". SS waits for 1 minute and verifies that no MEASUREMENT REPORT messages are detected on the uplink.
10	÷	SYSTEM INFORMATION CHANGE	SS waits until T305 has
11	\rightarrow	INDICATION CELL UPDATE	expired. UE shall transmit this
			message with measured results on RACH channels for cell 1 and cell 3 present in this message.

Expected Sequence

Step	Direction	Message	Comment
	UE SS		
12	÷	CELL UPDATE CONFIRM	No changes in physical resource allocation and RNTI identities.
13	\leftarrow	PHYSICAL CHANNEL	SS configures dedicated
		RECONFIGURATION	physical channels.
14	\rightarrow	PHYSICAL CHANNEL	UE shall transit to CELL_DCH
		RECONFIGURATION COMPLETE	state.
15	\rightarrow	MEASUREMENT REPORT	Repeated at 500 milliseconds
			interval

Specific Message Content

MASTER INFORMATION BLOCK (Step 1)

Use the same message sub-type found in clause 6.1 of TS 34.108, with the following exception:

Information Element	Value/Remarks
MIB Value Tag	1

System Information Block type 11 (Step 1)

Information Element	Value/remark
SIB12 indicator	FALSE
FACH measurement occasion info	Not Present
Measurement control system information	
- Use of HCS	Not used
- Cell selection and reselection quality measure	CPICH RSCP
- Intra-frequency measurement system information	
- Intra-frequency measurement identity	Not present
- Intra-frequency cell info list	•
- CHOICE intra-frequency cell removal	Remove no intra-frequency cellsNot Present
- New intra-frequency cells	
- Intra-frequency cell id	1
- Cell info	
- Cell individual offset	0 dBNot Present
- Reference time difference to cell	Not present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1 Refer to clause
	titled "Default settings for cell No.1 (FDD)" in clause
	6.1.4 of TS34.108
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
 Cell selection and Re-selection info 	Not present
- Cells for measurement	Not Present
-Intra-frequency measurement quantity	Not Present
 Intra-frequency reporting quantity for RACH 	Not Present
reporting	
-Maximum number of reported cells on RACH	Not Present
 Reporting information for state CELL_DCH 	Not Present
 Inter-frequency measurement system information 	Not Present
 Inter-RAT measurement system information 	Not Present
- Traffic volume measurement system information	Not Present
- UE internal measurement system information	Not Present

MEASUREMENT CONTROL (Step 5)

Information Element	Value/remark
Measurement Identity	5
Measurement Command	Setup
Measurement Reporting Mode	
- Measurement Reporting Transfer Mode	Acknowledged Mode RLC
- Periodic Reporting / Event Trigger Reporting Mode	Periodical Reporting
Additional measurements list	Not Present
CHOICE measurement type	Intra-frequency measurement
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Remove no intra-frequency cells
- New intra-frequency info list	
- Intra-frequency cell id	2
- Cell info	
- Cell individual offset	0 dB
 Reference time difference to cell 	Not Present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 2
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cells for measurement	Not Present
 Intra-frequency measurement quantity 	
- Filter Coefficient	Not Present (Default is 0)
 Measurement quantity 	CPICH RSCP
 Intra-frequency reporting quantity 	
 Reporting quantities for active set cells 	
 SFN-SFN observed time difference reporting 	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator	
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	FALSE
- Pathloss reporting indicator	FALSE
- Reporting quantities for monitored set cells	No report
SFN-SFN observed time difference reporting	No-report
indicator	FALSE
 Cell synchronisation information reporting indicator 	FALSE
- Cell identity reporting indicator	FALSE
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
- Reporting quantities for detected cells	Not present
- Reporting cell status	Not present
- CHOICE reported cell	Report cells within active and/or monitored set on used
	frequency or within active and/or monitored set on non-
	used frequency
- Maximum number of reported cells	2
- Measurement validity	Not present
- CHOICE report criteria	Periodical reporting criteria
- Amount of reporting	Infinity
- Reporting interval	16 seconds
DPCH compressed mode status info	Not Present

I

MEASUREMENT REPORT (Step 6)

Information Element	Value/remark
Measurement identity	Check to see if set to 5
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measured results list 	
 Cell measured results 	
- Cell Identity	Check to see if it is absent
- SFN-SFN observed time difference	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 1
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
 Cell measured results 	
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
 Cell synchronisation information 	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Additional measured result list	Check to see if this IE is absent
Event results	Check to see if this IE is absent

PHYSICAL CHANNEL RECONFIGURATION (Step 7)

Use the same message sub-type found in [9] TS 34.108 clause 9, which is entitled "(Packet to CELL_FACH from CELL_DCH in PS)"

MASTER INFORMATION BLOCK (Step 9)

Use the same message sub-type found in clause 6.1 of TS 34.108, with the following exception:

Information Element	Value/Remarks
MIB Value Tag	2

ĺ

System Information Block type 11 (Step 9)

Information Element	Value/remark
SIB12 indicator	TRUE
FACH measurement occasion info	Not Present
Measurement control system information	
- Intra-frequency measurement system information	
- Intra-frequency measurement identity	Not present
- Intra-frequency cell info list	
- CHOICE intra-frequency cell removal	Remove no intra-frequency cellsNot Present
- New intra-frequency cells	
- Intra-frequency cell id	1
- Cell info	
- Cell individual offset	0 dBNot Present
- Reference time difference to cell	Not present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1 Refer to clause
	titled "Default settings for cell No.1 (FDD)" in clause
	6.1.4 of TS34.108
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	Not present
- Intra-frequency cell id	2
- Cell info	-
- Cell individual offset	0 dBNot Present
- Reference time difference to cell	Not present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 2 Refer to clause
	titled "Default settings for cell No.1 (FDD)" in clause
	6.1.4 of TS34.108
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection info	
- Qoffset _{s.n}	0 dB
- Maximum allowed UL TX power	0 dBm
- HCS neighbouring cell information	Not Present
- CHOICE Mode	FDD
- Qqualmin	-20dB
- Qrxlevmin	-115dBm
- Cells for measurement	Not Present
-Intra-frequency measurement quantity	Not Present
-Intra-frequency reporting quantity for RACH	Not Present
reporting	
-Maximum number of reported cells on RACH	Not Present
-Reporting information for state CELL_DCH	Not Present
- Inter-frequency measurement system information	Not Present
- Inter-RAT measurement system information	Not Present
- Traffic volume measurement system information	Not Present
	Not Present
o = internal measureme nt system mormation	norriodoni

1

System Information Block type 12 (Step 9)

Information Element	Value/remark
FACH measurement occasion info	Not Present
Measurement control system information	
- Use of HCS	Not used
- Cell selection and reselection quality measure	CPICH RSCP
- Intra-frequency measurement system information	e e e e e e e e e e e e e e e e e e e
 Intra-frequency measurement identity Intra-frequency cell cells 	6
- CHOICE intra-frequency cell removal	Remove no intra-frequency cellsNot Present
- New intra-frequency cells	Remove no inita-nequency cells Not Fresent
- Intra-frequency cell id	3
- Cell info	3
- Cell individual offset	0 dB Not Present
- Reference time difference to cell	Not Present
- Read SFN Indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 3 Refer to clause
	titled "Default settings for cell No.3 (FDD)" in clause
	6.1.4 of TS34.108
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
 Cell selection and Re-selection info 	
- Qoffset _{s,n}	0dB
- Maximum allowed UL TX power	0dBm
 HCS neighbouring cell information 	Not Present
- CHOICE Mode	FDD
- Qqualmin, Qrxlevmin	-20dB, -115dBm
- Intra-frequency measurement quantity	Net Deserve (Defectivity 0)
- Filter Coefficient	Not Present (Default is 0)
- Measurement quantity	CPICH RSCP
 Intra-frequency reporting quantity measurement for RACH reporting 	
- SFN-SFN observed time difference reporting	No report
indicator	
- CHOICE mode	FDD
- Reporting quantity	CPICH RSCP
- Maximum number of reported cells on RACH	Current cell + best neighbour
 Reporting information for state CELL_DCH 	
 Intra-frequency reporting quantity 	
 Reporting quantities for active set cells 	
 SFN-SFN observed time difference reporting 	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator	54.05
- Cell identity reporting indicator	FALSE
- CHOICE mode	
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE FALSE
 Pathloss reporting indicator Reporting quantities for monitored set cells 	FALSE
- Reporting quantities for monitored set cells - SFN-SFN observed time difference reporting	No report
indicator	No report
- Cell synchronisation information reporting	FALSE
indicator	
- Cell identity reporting indicator	FALSE
- CHOICE mode	FDD
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
- Reporting quantities for detected cells	Not present
- CHOICE report criteria	Intra-frequency measurement reporting criteria
- Parameter required for each event	
- Intra-frequency event identity	1a
- Triggering condition 1	Not Present
- Triggering condition 2	Monitored set cells
- Triggering condition 1	Not Present

Release 5

 Reporting range constant Cells forbidden to affect reporting W Hysteresis 	20.0 dB Not present 0.0 1.0 dB
 Threshold used frequency Reporting deactivation threshold 	Not Present
- Replacement activation threshold	Not Present
- Time to trigger	60 ms
 Amount of reporting 	Infinity
- Reporting Interval	500 milliseconds
- Reporting cell status	
- CHOICE reported cell	Report cells within active and/or monitored set on used frequency or within active and/or monitored set on non- used frequency
- Maximum number of reported cells	3
- Inter-frequency measurement system information	Not present
 Inter-RAT measurement system information 	Not present
- Traffic volume measurement system information	Not present
 - UE internal measurement system information 	Not present

SYSTEM INFORMATION CHANGE INDICATION (Step 10)

Information Element	Value/Remarks
BCCH modification info	
- MIB Value tag	2

CELL UPDATE (Step 11)

Information Element	Value/remark
U-RNTI	Check to see if set to the same value assigned during
	the execution of procedure P3 or P5.
START list	Checked to see if this IE is present
AM_RLC error indication(RB2, RB3 or RB4)	FALSE
AM_RLC error indication(RB>4)	FALSE
Cell update cause	Check to see if it is set to "Periodical cell update"
Failure case	Check to see if it is absent
Measured results on RACH	
 Measurement result for current cell 	
- CHOICE measurement quantity	Check to see if set to "CPICH RSCP"
- CPICH RSCP	Check to see if it is present
 Measurement results for monitored cells 	
- SFN-SFN observed time difference	Not Checked
- Primary CPICH info	
- Primary scrambling code	Check to see if the same as cell 3's code.
- CHOICE measurement quantity	Check to see if set to "CPICH RSCP"
- CPICH RSCP	Check to see if it is present

PHYSICAL CHANNEL RECONFIGURATION (Step 13)

Use the same message sub-type found in [9] TS 34.108 clause 9, which is entitled "(Packet to CELL_DCH from CELL_FACH in PS)".

MEASUREMENT REPORT (Step 15)

Information Element	Value/remark
Measurement identity	Check to see if set to 6
Measured Results	
- CHOICE measurement	Check to see if set to "Intra-frequency measured results list"
 Intra-frequency measurement results list 	
- Cell measured results	
- Cell Identity	Check to see if it is absent
- SFN-SFN observed time difference	Check to see if this IE is absent
- Cell synchronisation information	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 1
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
- Cell synchronisation information	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 2
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
- Cell measured results	
- Cell Identity	Check to see if it is absent
	Check to see if this IE is absent
- Cell synchronisation information	Check to see if this IE is absent
- Primary CPICH Info	
- Primary Scrambling Code	Check to see if it's the same code for cell 3
- CPICH Ec/No	Check to see if this IE is absent
- CPICH RSCP	Check to see if this IE is present
- Pathloss	Check to see if this IE is absent
Measured Results on RACH	Check to see if this IE is absent
Event results	Check to see if this set to 'Intra-frequency measurement
	event results'
- Intra-frequency event identity	Check to see if set to '1a'
- Cell measurement event results	
- CHOICE Mode	Check to see if set to 'FDD'
- Primary CPICH info	
- Primary Scrambling Code	Check to see if set to the same code for cell 3

8.4.1.5.5 Test Requirement

After step 5, the UE shall start to transmit MEASUREMENT REPORT messages at 16 seconds interval. The message shall contain IE "measured result" to report cell 2's CPICH RSCP value.

After step 8, the UE shall not send any MEASUREMENT REPORT messages containing reporting quantities requested in MEASUREMENT CONTROL messages in step 5.

After step 10, the UE shall perform a cell update procedure and transmit a CELL UPDATE message. In this message, measured values CPICH RSCP for cell 1 and cell 3 shall be included in the IE "measured results on RACH".

After step 15, the UE shall apply the intra-frequency measurement reporting criteria" received in System Information Block type 12 messages of step 9. It shall send MEASUREMENT REPORT messages at 500 milliseconds interval. In these messages, triggering of event '1a' shall be reported in IE "Event results" with IE "Primary CPICH info" containing the primary scrambling code for cell 3.

The message shall contain IE "measured result" to report CPICH RSCP values of cell 1, 2 and 3.

---<End of Modifications>---

Tdoc **#***T*1S020757

3GPP TSG-T1 SIG Meeting #26 Luton, UK, 5th – 7th November 2002

ж <mark>3</mark>	<mark>4.123-1</mark>	CR <mark>342</mark>	ж rev	- *	Current vers	^{sion:} 5.1.1	ж	
For <u>HELP</u> on u	ising this for	rm, see bottom of ti	his page or	look at th	e pop-up text	over the ¥ sy	rmbols.	
Proposed change	affects: (JICC apps#	MEX	Radio A	ccess Networ	rk 📃 Core N	letwork	
Title: ೫	CR to TS of T1S-02	34.123-1: Update t 20584	o clause 10	Circuit S	witched Call (Control tests a	s revision	
Source:	Nokia, Er	icsson						
Work item code: ℜ	TEI				Date: ೫	25/10/2002		
Category: ₩	F (con A (cor B (add C (fun D (edi Detailed exp	the following categor rection) responds to a correc dition of feature), ctional modification of torial modification) planations of the abo 3GPP <u>TR 21.900</u> .	tion in an eai ffeature)		2	Rel-5 the following re (GSM Phase 2) (Release 1996) (Release 1997) (Release 1999) (Release 1999) (Release 4) (Release 5) (Release 6))))	
Reason for change		cases updated according to Ericsson c					ions <mark>and</mark>	
Summary of chang	2. 3. 4. 5. 6. 7.	GSM related term Conformance req in TCs 10.1.2.1.1 10.1.2.7.2, 10.1.2 10.1.3.4.4, 10.1.3 Conformance req Test requirement Conformance req 10.1.2.3.6, 10.1.2 10.1.3.4.6 and 10 Conformance req and Specific mes Conformance req 10.1.2.3.5, 10.1.2 10.1.2.3.5, 10.1.2 10.1.2.3.5, 10.1.2 10.1.2.3.5, 10.1.2 10.1.2.3.5, 10.1.2 10.1.2.7.4, 10.1.2 10.1.3.3.3, 10.1.3 10.1.3.5.7 and 10 Subclause heade	uirement, F , 10.1.2.2.1 2.8.2, 10.1.2 3.4.8 and 10 juirement and c 10.1.2.2. juirement and c 10.1.2.2. juirement and 2.3.7, 10.1.2 0.1.3.5.9 juirement and 2.4.4, 10.1.2 2.5.4, 10.1.2 2.5.4, 10.1.2 2.3.6, 10.1.3 0.1.3.5.8	Reference , 10.1.2.4 2.9.2, 10.1 0.1.3.5.6 nd Refere 1.2.2.2 nd Refere 3 odated in 2.4.1, 10.1 Reference nts update nd Refere 2.4.9, 10.1 2.5.6, 10.1 2.8.3, 10.1 3.4.1, 10.1	s and Expecte .3, 10.1.2.4.6 I.2.9.4, 10.1.3 ences updated mces updated TCs 10.1.2.3. I.2.4.13, 10.1. s, Test purposed in TC 10.1 ences updated I.2.4.10, 10.1. I.2.5.7, 10.1.2 I.2.9.1, 10.1.2 I.3.4.5, 10.1.3	ed sequence (, 10.1.2.5.5, 1 3.1.1, 10.1.3.2. d, editorial corr d, editorial corr .1, 10.1.2.3.3, .2.7.5, 10.1.2.3 se, Expected s .2.3.2 d in TCs 10.1.2 .2.4.11, 10.1.2 2.7.1, 10.1.2.7 2.9.5, 10.1.3.3 3.4.7, 10.1.3.5	0.1.2.6.6, .1, rections in rections in 8.4, sequence 2.3.4, 2.4.12, .3, .2, .3,	

originating" in 10.1.2.4 and in relevant test cases.

- 9. Conformance requirement, References, Test purpose, Expected sequence and Test requirements updated in TC 10.1.2.4.2
- 10. Conformance requirement and References updated, editorial correction in Specific message contents in TCs 10.1.2.4.5 and 10.1.2.5.3
- 11. Conformance requirement, References, Test purpose and Expected sequence updated in TC 10.1.2.4.7
- 12. Conformance requirement and References updated, editorial correction in Initial conditions in TC 10.1.2.4.8
- 13. Conformance requirement and Expected sequence updated in TC 10.1.2.5.1
- 14. Conformance requirement updated, editorial correction in Test requirements in TC 10.1.2.5.8
- 15. Subclause header changed from "U10 call active" to "U10 active" in 10.1.2.6 and in relevant test cases.
- 16. Conformance requirement and References updated, editorial corrections in Test purpose, Expected sequence and Test requirements in TC 10.1.2.6.1
- 17. Conformance requirement, References and Expected sequence updated, editorial corrections in Test purpose and Test requirements in TC 10.1.2.6.2
- Conformance requirement and References updated, editorial corrections in Test purpose and Specific message contents in TC 10.1.2.6.3
- 19. Conformance requirement and References updated, editorial corrections in Test purpose and Test requirements in TC 10.1.2.6.4
- 20. Conformance requirement, References, Expected sequence and Specific message contents updated, editorial corrections in Test purpose in TC 10.1.2.6.5
- 21. Conformance requirement, References, Expected sequence and Specific message contents updated in TC 10.1.2.9.3
- 22. Conformance requirement and References updated, editorial corrections in Test procedure in TC 10.1.3.3.1
- 23. Conformance requirement and References updated, editorial corrections in Test requirements in TC 10.1.3.3.4
- 24. Conformance requirement, References and Expected sequence updated, editorial corrections in Test requirements in TC 10.1.3.3.5
- 25. Conformance requirement updated, editorial corrections in Test requirements in TC 10.1.3.3.7
- 26. Conformance requirement and References updated, editorial corrections in Initial conditions in TC 10.1.3.4.2
- 27. Conformance requirement, References and Specific message contents updated in TC 10.1.3.4.3
- 28. Conformance requirement updated, editorial corrections in Test purpose and Test requirements in TC 10.1.3.5.1
- 29. Conformance requirement, References and Test requirements updated in TC 10.1.3.5.2
- 30. Conformance requirement, References and Specific message contents updated in TC 10.1.3.5.4
- 31. Conformance requirement, References, Expected sequence and Specific message contents updated in TC 10.1.3.5.5
- 32. Conformance requirement, References, Test purpose, Initial conditions, Test procedure and Test requirements updated in TC 10.1.4.1.1
- 33. Text deleted in 10.1.4.2
- 34. Conformance requirement and Initial conditions updated, editorial corrections in Test procedure and Test requirements in TC 10.1.4.2.1
- 35. Initial conditions updated, editorial corrections in Conformance requirement Test purpose, Test procedure and Test requirements in TC 10.1.4.3.1
- 36. Initial conditions updated, editorial corrections in Conformance requirement Test purpose and Test requirements in TC 10.1.4.3.2
- Conformance requirement, References, Test procedure and Expected sequence updated, editorial corrections in Specific message contents in TC 10.2.1
- 38. References and Expected sequence updated, editorial corrections in TC

	10.2				
	39. RRC signaling details removed from Tables 10.1.2/1, 10.1.2/3, 10.1.2/4,				
	10.1.3/1, 10.1.3/2 and 10.1.3/3.				
	. Changed Table 10.1.2/3 so that authentication procedure is performed				
	prior to security mode control.				
	41. Added authentication procedure in Table 10.1.2/4 prior to security mode				
	control.				
	42. Test requirements in clauses 10.1.2.2.2.5, 10.1.2.3.1.5, 10.1.2.3.3.5,				
	10.1.2.3.5.5, 10.1.2.3.6.5, 10.1.2.3.7.5, 10.1.2.4.1.5, 10.1.2.4.2.5,				
	10.1.2.4.3.5, 10.1.2.4.4.5, 10.1.2.4.5.5, 10.1.2.4.6.5, 10.1.2.4.7.5,				
	<mark>10.1.2.4.8.5, 10.1.2.4.10.5, 10.1.2.4.12.5, 10.1.2.4.13.5, 10.1.2.5.1.5,</mark>				
	10.1.2.5.2.5, 10.1.2.5.3.5, 10.1.2.5.4.5, 10.1.2.5.5.5, 10.1.2.5.8.5,				
	10.1.2.6.1.5, 10.1.2.6.2.5, 10.1.2.6.3.5, 10.1.2.6.4.5, 10.1.2.6.6.5,				
	10.1.2.7.1.5, 10.1.2.7.2.5, 10.1.2.7.3.5, 10.1.2.7.5.5, 10.1.2.8.1.5,				
	10.1.2.8.2.5, 10.1.2.8.4.5, 10.1.2.9.1.5, 10.1.2.9.2.5, 10.1.3.1.1.5,				
	10.1.3.2.1.5, 10.1.3.3.1.5, 10.1.3.3.2.5, 10.1.3.3.3.5, 10.1.3.3.4.5,				
	10.1.3.3.5.5, 10.1.3.3.7.5, 10.1.3.4.1.5, 10.1.3.4.2.5, 10.1.3.4.3.5,				
	10.1.3.4.4.5, 10.1.3.4.6.5, 10.1.3.5.1.5, 10.1.3.5.2.5, 10.1.3.5.3.5,				
	10.1.3.5.4.5, 10.1.3.5.5.5, 10.1.3.5.6.5, 10.1.3.5.9.5, 10.1.4.1.1.5,				
	10.1.4.2.1.5, 10.1.4.3.1.5, 10.1.4.3.2.5 and 10.3.5 updated.				
	43. Step 2 in Expected sequence of TC 10.1.2.3.3 modified so that timer value				
	is set to 30s and reference to timer tolerances in 34.108 added.				
	44. Test purpose, Test procedure and Test requirement in TCs 10.1.2.5.5,				
	10.1.2.6.2 and 10.1.2.7.2 updated because of removed conformance				
	requirement.				
	. In TC 10.1.4.1.1 term FACH changed to FACCH in Conformance				
	requirement and Test requirement and text deleted from Test purpose.				
	46. Test case 10.2.1 (Call re-establishment/call present, re-establishment				
	allowed) removed as call re-establishment procedure is not used for				
	UMTS. Instead lower layers are used for re-establishment. Re-				
	establishment of lower layers is covered by test cases 8.3.1.8 (RRC/Cell				
	update: Radio Link Failure) and 9.3.2 (Handling of IMSI shorter than the				
	maximum length).				
Consequences if % not approved:	Test cases in clause 10 are not according to the latest core specification.				
Clauses affected: #	See "Summary of change"				
	YN				
Other specs #	X Other core specifications %				
affected:	X Test specifications 34.123-2				
	X O&M Specifications				

How to create CRs using this form:

Other comments:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

24.008 V3.12.0 (2002-06) used

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

10 Circuit Switched Call Control (CC)

10.1 Circuit switched Call Control (CC) state machine verification

10.1.1 General on CC state machine verification

The principle of checking the call control functions consists in the validation of each call control identified state.

State U0 as an initial state is not verified in the tests of 10.1.2 (establishment of an outgoing call).

State U0.1 is never verified.

The steps to be followed within each performed test are:

- bring the UE into the required state;
- trigger the tested event;
- check the UE response and new state.

In clauses 10.1.2 and 10.1.3 different tables are defined to bring the UE into the required initial state. The exact table to be chosen is specified individually in clause "Initial conditions" of "Method of test" for each test case.

For each test, unless otherwise specified, a circuit switched basic service among those supported by the UE but excluding the emergency call teleservice shall be chosen arbitrarily, and the test shall be performed according to that basic service. If the only circuit switched basic service supported by the mobile is emergency call, then the incoming call tests shall not be performed and the other call control tests shall be performed with the EMERGENCY SETUP message replacing the SETUP message.

The initial states are to be checked through STATUS ENQUIRY messages sent by the SS, when feasible. This is not explicitly stated in the tables of expected sequences of signalling messages. The checking of final states are explicitly included into the expected sequences of signalling messages.

The following postamble may be used by the SS to bring UE back to idle mode in those test cases, in which it is not already included into expected sequence of signalling messages:

Step	Direction		Message	Comments
	UE	SS		
Ν	<		RRC CONNECTION RELEASE	
n+1		>	RRC CONNECTION RELEASE COMPLETE	
n+2	U	E		the UE shall releases the RRC
				connection.the main signalling link

Table 10.1.1/1: A postamble to bring the UE back to idle mode.

The postamble has not been included into the all of the tests in order to leave an option to concatenate the procedures in the future by using a final state of a test case as an initial state to another one.

For the special case of U0, the state is checked by sending STATUS ENQUIRY message with all possible values of transaction identifier (seven values) as U0 is the only state in which for every TI the UE will answer with release complete with cause #81. If U0 is to be verified when no RRC connection exists, first a mobile terminating radio connection must be established.

The UE responses are either call management messages received by the SS or lower layers functions activated within the UE or MMI actions (e.g. the buzzing of an alerting tone).

A time-out within the UE is triggered by the SS when it does not answer back an UE expected response.

The test sequences may be split in 3 main groups:

- establishment and release of an outgoing call;
- establishment and release of an incoming call;
- in-call functions.

Some test cases use Basic Generic Procedures, "Mobile terminated establishment of Radio Resource Connection" and "Radio Bearer Setup Procedure" defined in TS34.108 clause 7.

General tolerance value on protocol timers defined in TS34.108 is used in some test cases if no specific tolerance on timer is defined in a test case.

Remark on verification of transient states

Some call control states of the user equipment may be transient, depending on implementation, configuration of the UE and previous messages.

If a test starts in a transient state, then the test is executed without verification of the starting state.

10.1.2 Establishment of an outgoing call

Initial conditions

As a minimum requirement the UE is updated and has been given a TMSI, a ciphering key and cipher key sequence number, and the layer 2, RRC and MM functionalities have been verified.

There are as many CM initial conditions as states to be checked.

The tables below describe message exchanges which bring the UE in the requested initial states.

A state may be taken as initial only when all the states which lead to this initial states have been validated. The order followed in the test procedure will be U0, U0.1, U1, U3, U4, U10, U12, U19, U11 as seen in the table underneath.

The UE is brought again in the initial state starting with U0 at each new test performed.

Table 10.1.2/1: Establishment of an outgoing call, procedure 1 (late assignment)	Table 10.1.2/1: Est	tablishment of an	outgoing call	, procedure 1	(late assignment)
--	---------------------	-------------------	---------------	---------------	-------------------

Γ	Step	Direc	tion	Message	Comments
	Ē	UE	SS		
	1	T.	•	RRC CONNECTION REQUEST Mobile Originated	Establishment cause: Originating
			_	establishment of Radio Resource Connection	Conversational Call Initiate outgoing call
	2	<mark><</mark>	-	RRC CONNECTION SETUPVoid	
	3	<mark>-</mark> ;	>	RRC CONNECTION SETUP COMPLETEVoid	
	4	-;	>	CM SERVICE REQUEST	U0.1
	5	<	-	AUTHENTICATION REQUEST	
	6	-:	>	AUTHENTICATION RESPONSE	
	7	<	-	SECURITY MODE COMMAND	
	8	-;	>	SECURITY MODE COMPLETE	
	9	-:	>	SETUP	U1
	10	<	-	CALL PROCEEDING	U3
	11	<	-	ALERTING	U4
	12			Radio Bearer Setup Procedure	DTCH, See TS 34.108 clause 7.1.3
	13	<	-	CONNECT	
	14	-:	>	CONNECT ACKNOWLEDGE	U10
	A15	<	-	DISCONNECT	U12 (note 1)
	B15	<	-	DISCONNECT	U12 (note 2)
	B16	-;	>	RELEASE	U19
	C15				MMI action, terminate call
	C16	-;	>	DISCONNECT	U11
1	NOTE 1:	The	Progre	ess Indicator IE with progress description #8 "in band	information or appropriate pattern now
		avai	lable" i	is included.	
1	NOTE 2:	The	Progre	ess Indicator IE is not included.	

Table 10.1.2/2: Void

Table 10.1.2/3: Establishment of an outgoing call, procedure 3

Step	Direc	tion	Message	Comments
	UE	SS		
1	-	٧	Mobile Originated establishment of Radio Resource	Establishment cause: Originating
			Connection RRC CONNECTION REQUEST	Conversational CallInitiate outgoing call
2	<mark><</mark>	-	RRC CONNECTION SETUPVoid	
3	∀ 7		RRC CONNECTION SETUP COMPLETEVoid	
4	->	>	CM SERVICE REQUEST	U0.1
<u>4a</u>	<u><</u>	-	AUTHENTICATION REQUEST	
4 <u>a</u> 4 <u>b</u> 5	-> < ->	>	AUTHENTICATION RESPONSE	
5	<	-	SECURITY MODE COMMAND	
6	->	>	SECURITY MODE COMPLETE	
7	->	>	SETUP	U1
8	<mark><</mark>	-	AUTHENTICATION REQUESTVoid	
9	v 7	<mark>≻</mark>	AUTHENTICATION RESPONSEVoid	
10	<		CALL PROCEEDING	U3
11			Radio Bearer Setup Procedure	DTCH, See TS 34.108 clause 7.1.3
12	<	-	ALERTING	U4
13	<	-	CONNECT	
14	->	>	CONNECT ACKNOWLEDGE	U10
A15	<	-	DISCONNECT	U12 (note 1)
B15	<	-	DISCONNECT	U12 (note 2)
B16	->	>	RELEASE	U19
C15				MMI action, terminate call
C16	->	>	DISCONNECT	U11
NOTE 1:	The	Progre	ess Indicator IE with progress description #8 "in band	information or appropriate pattern now
			is included.	
NOTE 2:	The	Progre	ess indicator IE is not included.	

Table 10.1.2/4: Establishment of an	outgoing call, procedure 4
-------------------------------------	----------------------------

Step	Direction Message		Message	Comments
-	UE	SS	C C	
1	-		Mobile Originated establishment of Radio Resource	Establishment cause: Originating
		-	ConnectionRRC CONNECTION REQUEST	Conversational CallInitiate outgoing call
2	<mark>-</mark>	_	RRC CONNECTION SETUPVoid	
3	V T	<mark>~</mark>	RRC CONNECTION SETUP COMPLETEVoid	
4	->		CM SERVICE REQUEST	U0.1
5	<	-	IDENTITY REQUEST	
6	->	>	IDENTITY RESPONSE	
<mark>6a</mark>	V	-	AUTHENTICATION REQUEST	
<mark>6a</mark> 6b 7		>	AUTHENTICATION RESPONSE	
	<	-	SECURITY MODE COMMAND	
8	->	>	SECURITY MODE COMPLETE	
9	->		SETUP	U1
10	¥	-	Radio Bearer Setup Procedure	DTCH (note 1), See TS_34.108 clause
				<u>7.1.3 (note 1)</u>
11	<	-	CALL PROCEEDING	U3
12	<	-	ALERTING	U4
13	<	-	CONNECT	
14	->	>	CONNECT ACKNOWLEDGE	U10
A15	<	-	DISCONNECT	U12 (note 2)
B15	<	-	DISCONNECT	U12 (note 3)
B16	->	>	RELEASE	U19
C15				MMI action, terminate call
C16	÷	>	DISCONNECT	U11
			hannel is appropriate for the chosen bearer capability	
NOTE 2			ess Indicator IE with progress description #8 "in band	information or appropriate pattern now
			s included.	
NOTE 3	The	Progre	ess Indicator IE is not included.	

10.1.2.1 Outgoing call / U0 null state

10.1.2.1.1 Outgoing call / U0 null state / MM connection requested

10.1.2.1.1.1 Definition

The call control entity of the User Equipment requests the MM-sublayer to establish a mobile originating MM-connection.

10.1.2.1.1.2 Conformance requirement

1)—If no RR connection exists, the MM sublayer requests the RR sublayer to establish an RR connection and enters MM sublayer state WAIT FOR RR CONNECTION (MM CONNECTION). This request contains an establishment cause and a CM SERVICE REQUEST message. Upon initiation of an outgoing basic call by user the UE shall initiate establishment of an MM connection, using as first MM message a CM SERVICE REQUEST message with CM service type "Mobile originating call establishment or packet mode connection establishment".

References

TS 24.008 clause 5.2.1.1 and clause 4.5.1.1, TS 25.331 clause 8.1.3.

10.1.2.1.1.3 Test purpose

To verify that upon initiation of an outgoing basic call by user the UE initiates establishment of an MM connection, using as first MM message a CM SERVICE REQUEST message with CM service type "Mobile originating call establishment or packet mode connection establishment".

10.1.2.1.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. When the SS receives CM SERVICE REQUEST, the contents of it shall be checked.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	->	RRC CONNECTION REQUEST Void	UE initiates outgoing call
2	4-	RRC CONNECTION SETUPVoid	
3	→	RRC CONNECTION SETUP COMPLETEVoid	
4	->	CM SERVICE REQUEST	SS shall verify the CM service type
			requested type of call which is asked for
			"basic" or "emergency" by the UE
5	<-	RRC CONNECTION RELEASE Void	The SS releases the RRC connection.
6	→	RRC CONNECTION RELEASE COMPLETEVoid	
7	UE		the UE shall release the main signalling
			link

Specific message contents:

None.

10.1.2.1.1.5 Test requirements

After step 3 the UE shall initiate establishment of an MM connection, using as first MM message a CM SERVICE REQUEST message with CM service type "Mobile originating call establishment or packet mode connection establishment".

10.1.2.2 Outgoing call / U0.1 MM connection pending

10.1.2.2.1 Outgoing call / U0.1 MM connection pending / CM service rejected

10.1.2.2.1.1 Definition

A request for MM connection is rejected by the SS.

10.1.2.2.1.2 Conformance requirement

If the service request cannot be accepted, the network returns a CM SERVICE REJECT message to the UE. Upon receiving indication of an MM connection establishment being rejected, CC entity should inform upper layer of this rejection.

If no other MM connection is active, the network may start the RR connection release when the CM SERVICE REJECT message is sent.

References

TS 24.008, clause 4.5.1.1, TS 24.007, clause 6.2.2.

10.1.2.2.1.3 Test purpose

To verify that a CC entity of the UE in CC-state U0.1, "MM-connection pending", upon the UE receiving a CM SERVICE REJECT message, returns to CC state U0, "Null".

10.1.2.2.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U0.1 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. When the SS receives CM SERVICE REQUEST, the contents of it shall be checked. The SS rejects it by CM SERVICE REJECT. Then the SS will check the state of the UE by using STATUS ENQUIRY with all the relevant transaction identifiers.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	CM SERVICE REJECT	
2	<-	STATUS ENQUIRY	
3	->	RELEASE COMPLETE	cause shall be #81 (invalid TI value)
4	SS		repeat steps 2-3 to cover all the
			transaction identifiers from 000 110
5	<-	RRC CONNECTION RELEASE Void	The SS releases the RRC connection.
6	->	RRC CONNECTION RELEASE COMPLETEVoid	
7	UE		the UE shall release the main signalling
			link

Specific message contents:

None.

10.1.2.2.1.5 Test requirements

After step 2 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.2.2 Outgoing call / U0.1 MM connection pending / CM service accepted

10.1.2.2.2.1 Definition

A CM request is accepted for the MM-connection by the SS.

10.1.2.2.2.2 Conformance requirement

Having entered the "MM connection pending" state, upon MM connection establishment, the call control entity of the UE sends a setup message to its peer entity. This setup message is

- a SETUP message, if the call to be established is a basic call A CC entity of the UE in CC state U0.1, "MM connection pending", upon the UE receiving a CM SERVICE ACCEPT message, shall send a SETUP message specifying the Called party BCD number that was entered into the UE and then enter CC state U1, "Call initiated".

It then enters the "call initiated" state.

References

TS 24.008, clause 4.5.1.1 and clause 5.2.1.1.

10.1.2.2.2.3 Test purpose

To verify that a CC entity of the UE in CC-state U0.1, "MM-connection pending", upon the UE receiving a CM SERVICE ACCEPT message, sends a SETUP message specifying the Called party BCD number that was entered into the UE and then enters CC state U1, "Call initiated".

10.1.2.2.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U0.1 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. When the UE is requesting a MM-connection, the SS will indicate acceptance by sending a CM SERVICE ACCEPT message. The UE shall respond with SETUP. Then the SS will check the state of the call control entity by STATUS ENQUIRY with the relevant transaction identifiers.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	CM SERVICE ACCEPT	
2	->	SETUP	with called party BCD number.
3	<-	STATUS ENQUIRY	
4	->	STATUS	cause shall be #30 (response to enq.) and state U1 call initiated.

Specific message contents:

None.

10.1.2.2.2.5 Test requirements

After step 1 a CC entity of the UE in CC state U0.1, "MM connection pending", shall send a SETUP message specifying the Called party BCD number that was entered into the UE and then enter CC state U1, "Call initiated".

10.1.2.2.3 Outgoing call / U0.1 MM connection pending / lower layer failure

10.1.2.2.3.1 Definition

The call control entity of the UE being in the state, U0.1, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.2.2.3.2 Conformance requirement

1) If an RR connection failure occurs or the IMSI is deactivated during the establishment of an MM connection, the MM connection establishment is aborted, timers T3230 is stopped, and an indication is given to the CM entity that requested the MM connection establishment. This shall be treated as a rejection for establishment of the new MM connection, and the MM sublayer shall release all active MM connections. Upon a lower layer failure the UE releases the MM connection in progress and returns to idle mode. In that state no call exists, and the CC entity is in state U0, "Null".

References

TS 24.008, clause 4.5.1.2, clause 5.2.1.1, clause 5.5.3.2 and clause 8.3, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.2.2.3.3 Test purpose

To verify that after the UE with a CC entity in state U0.1, "MM-connection pending", has detected a lower layer failure and has returned to idle mode, the CC entity is in state U0, "Null".

10.1.2.2.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U0.1 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. When the UE has sent a CM SERVICE REQUEST message, the SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS re-modifies the scrambling code of downlink transmission(DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	S	S		SS modifies the scrambling code of DPCH for generating lower layer failure
2	-:	>	CELL UPDATE	CCCH
3	<	:-	RRC CONNECTION RELEASE	CCCH
4	S	S		SS re-modifies the scrambling code of DPCH to the original one.
5	S	S		SS waits 60 s. UE shall send no message on DCCH

Specific message contents:

None.

10.1.2.2.3.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.2.3 Outgoing call / U1 call initiated

10.1.2.3.1 Outgoing call / U1 call initiated / receiving CALL PROCEEDING

10.1.2.3.1.1 Definition

The call control entity of the UE being in the state, U1, a CALL PROCEEDING message is sent by the SS.

10.1.2.3.1.2 Conformance requirement

1)-<u>Having entered the "call initiated" state, when the call control entity of the UE receives a CALL PROCEEDING</u> message, it shall enter the "mobile originating call proceeding" state. A CC entity of the UE in CC state U1, "Call initiated", upon receipt of a CALL PROCEEDING message, shall enter CC state U3, "Mobile originating call proceeding".

References

TS 24.008, clauses 5.2.1.1, 5.2.1.2 and 5.2.1.3.

10.1.2.3.1.3 Test purpose

To verify that a CC entity of the UE in CC-state U1, "Call initiated", upon receipt of a CALL PROCEEDING message, enters CC state U3, "Mobile originating call proceeding".

10.1.2.3.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U1 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U1. The SS sends a CALL PROCEEDING message to the UE. The SS checks by using the status enquiry procedure that the CC entity has entered the state U3.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	CALL PROCEEDING	tone generation not mandatory
2	<-	STATUS ENQUIRY	
3	->	STATUS	cause #30, state U3

Specific message contents:

None.

10.1.2.3.1.5 Test requirements

After step 1 a CC entity of the UE in CC state U1, "Call initiated", shall enter CC state U3, "Mobile originating call proceeding".

10.1.2.3.2 Outgoing call / U1 call initiated / rejecting with RELEASE COMPLETE

10.1.2.3.2.1 Definition

The call control entity of the UE being in the state, U1, the call is rejected by a RELEASE COMPLETE message sent by the SS.

10.1.2.3.2.2 Conformance requirement

1) A call control entity of the UE in any call control state shall, upon receipt of a RELEASE COMPLETE message from its peer entity in the network: stop all running call control timers ; release the MM connection; and return to the "null" state.

A CC entity of the UE in CC state U1, "Call initiated", upon receipt of a RELEASE COMPLETE message with valid cause value, shall enter CC state U0, "Null".

2) On returning to idle mode, the CC entities relating to the seven mobile originating transaction identifiers shall be in state U0, "Null".

3) On releasing the MM connection, the UE shall wait for MM layer release initiated by the network.

References

Conformance requirement 1: TS 24.008, clause 5.4.4.1.3 5.4.2 and clause 5.4.4.

Conformance requirement 2: TS 24.008, clause 5.5.3.2 and 8.3.1.

Conformance requirement 3: TS 24.008, clause 5.4.4.1.3 and clause 4.5.3.

10.1.2.3.2.3 Test purpose

- 1) To verify that a CC entity of the UE in CC-state U1, "Call initiated", upon receipt of a RELEASE COMPLETE message with valid cause value, enters CC state U0, "Null".
- 2) To verify that in returning to idle mode, the CC entities relating to the seven mobile originating transaction identifiers are in state U0, "Null".

3) To verify that in releasing the MM connection, the UE shall wait for MM layer release initiated by SS.

10.1.2.3.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U1 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U1. The SS sends a RELEASE COMPLETE message to the UE. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1	<	:-	RELEASE COMPLETE	See specific message content below.
2	<	:-	STATUS ENQUIRY	
3	->		RELEASE COMPLETE	cause #81 (invalid TI value)
4	S	S		repeat steps 2-3 to cover all the
				transaction identifiers from 000110
5	<-		RRC CONNECTION RELEASE Void	The SS releases the RRC
				connection.the main signalling link shall
				be released.
6	-	>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

RELEASE COMPLETE

- 1) With a valid cause value among:
 - related to numbering,
 - #1 Unassigned (unallocated) number
 - #3 Nno route to destination
 - #22 <u>N</u>number changed
 - #28 Linvalid number format (incomplete number)

related to bearer capabilities,

#8 Ooperator determined barring

#34 No circuit/channel available

- #57 **B**bearer capability not authorized
- #58 **B**bearer capability not presently available
- #63 Service or option not available, unspecified
- #65 **B**bearer service not implemented

#34-no-circuit/channel-available (call-queuing).

10.1.2.3.2.5 Test requirements

After step 2 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.3.3 Outgoing call / U1 call initiated / T303 expiry

10.1.2.3.3.1 Definition

The call control entity of the UE being in the state, U1, if no response is then received from the SS, timer T303 expires at the UE side.

10.1.2.3.3.2 Conformance requirement

1) If timer T303 elapses in the "call initiated" state before any of the CALL PROCEEDING, ALERTING, CONNECT or RELEASE COMPLETE messages has been received, the clearing procedure described in TS 24.008 clause 5.4 is performed. A CC entity of the UE in CC state U1, "Call initiated", upon expiry of T303 shall send a DISCONNECT message to its peer entity and enter state U11, "Disconnect request".

References

TS 24.008, clause 5.2.1.1 and clause 5.4.

10.1.2.3.3.3 Test purpose

1) To verify that a CC entity of the UE in CC-state U1, "Call initiated", upon expiry of T303 sends a DISCONNECT message to its peer entity and enters state U11, "Disconnect request".

10.1.2.3.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U1 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U1. When T303 expires at the UE, the UE shall send DISCONNECT. The SS checks by using the status enquiry procedure that the CC entity has entered the state U11, disconnect request.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	SS			SS waits for T303 expiry.
2	->		DISCONNECT	Shall be transmitted between 27 s and
				303 s after the CM SERVICE
				REQUEST, check the timer T303
				accuracy, see TS34.108 clause 4.2.3.
3	<-		STATUS ENQUIRY	
4	->		STATUS	cause #30, status U11

Specific message contents:

None.

10.1.2.3.3.5 Test requirements

After step 1 Uupon expiry of timer T303, a CC entity of the UE in CC state U1, "Call initiated", shall send a DISCONNECT message and enter state U11, "Disconnect request".

10.1.2.3.4 Outgoing call / U1 call initiated / lower layer failure

10.1.2.3.4.1 Definition

The call control entity of the UE being in the state, U1, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.2.3.4.2 Conformance requirement

The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. Upon a lower layer failure MM informs the relevant CM entities that the MM connection has been interrupted. As call re-establishment is not allowed, the CC entity must perform a local release. The UE returns to idle mode. In that state no call exists, and the CC entity is in state U0, "Null".

References

TS 24.008, clause 4.5.2.3, and clause 5.2.1.1 and clause 5.5.3.2, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.2.3.4.3 Test purpose

To verify that after the UE with a CC entity in state U1 "Call initiated", has detected a lower layer failure and has returned to idle mode, the CC entity is in state U0, "Null".

10.1.2.3.4.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U1 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The UE is brought to the state U1. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS re-modifies the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	S	S		SS modifies the scrambling code of DPCH for generating lower layer failure
2	->		CELL UPDATE	CCCH
3	<	-	RRC CONNECTION RELEASE	СССН
4	S	S		SS re-modifies the scrambling code of DPCH to the original one.
5	S	S		SS waits 60 s. UE shall send no message on DCCH

Specific message contents:

None.

10.1.2.3.4.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

1564

10.1.2.3.5 Outgoing call / U1 call initiated / receiving ALERTING

10.1.2.3.5.1 Definition

The call control entity of the UE being in the state, U1, an ALERTING message is sent to the UE as an indication that a call is being alerted at a called end.

10.1.2.3.5.2 Conformance requirement

1) When the call control entity of the UE in the "call initiated" state or "mobile originating call proceeding" state receives an ALERTING message then, the call control entity of the UE shall stop timer T303 and T310 (if running) and shall enter the "call delivered" state. A CC entity of the UE in CC state U1, "Call initiated", upon receipt of an ALERTING message, shall enter CC state U4, "Call delivered".

References

TS 24.008, clause 5.2.1.1 and clause 5.2.1.5.

10.1.2.3.5.3 Test purpose

To verify that a CC entity of the UE in CC-state U1, "Call initiated", upon receipt of an ALERTING message, enters CC state U4, "Call delivered".

10.1.2.3.5.4 Method of test

Related ICS/IXIT statements

supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U1 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U1. The SS sends an ALERTING message to the UE. The SS checks by using the status enquiry procedure that the CC entity has entered the state U4, call delivered.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1	<	:-	ALERTING	
2	<	:-	STATUS ENQUIRY	
3	<-		STATUS	cause #30, state U4

Specific message contents:

None.

10.1.2.3.5.5 Test requirements

After step 1 a CC entity of the UE in CC state U1, "Call initiated", shall enter CC state U4, "Call delivered".

- 10.1.2.3.6 Outgoing call / U1 call initiated / entering state U10
- 10.1.2.3.6.1 Definition

The call control entity of the UE being in the state, U1, a CONNECT message is received by the UE.

10.1.2.3.6.2 Conformance requirement

1) The call control entity of the UE in the "call initiated" state, in the "mobile originating call proceeding" state or in the "call delivered" state, shall, upon receipt of a CONNECT message:

- attach the user connection;
- return a CONNECT ACKNOWLEDGE message;
- stop any locally generated alerting indication (if applied);
- stop timer T303 and T310 (if running);
- <u>enter the "active" state.</u> A CC entity of the UE in CC state U1, "Call initiated", upon receipt of a CONNECT message, shall send a CONNECT ACKNOWLEDGE message to its peer entity and enter CC state U10, "Active".

References

TS 24.008, clause 5.2.1.1 and clause 5.2.1.6.

10.1.2.3.6.3 Test purpose

To verify that a CC entity of the UE in CC-state U1, "Call initiated", upon receipt of a CONNECT message, sends a CONNECT ACKNOWLEDGE message to its peer entity and enters CC state U10, "Active".

10.1.2.3.6.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U1 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U1. The SS sends a CONNECT message to the UE. The UE shall respond by sending a CONNECT ACKNOWLEDGE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U10, active.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1	<	<-	CONNECT	
2	->		CONNECT ACKNOWLEDGE	
3			STATUS ENQUIRY	
4			STATUS	cause #30, state U10

Specific message contents:

None.

10.1.2.3.6.5 Test requirements

After step 1 a CC entity of the UE in CC state U1, "Call initiated", shall send a CONNECT ACKNOWLEDGE message and shall enter CC state U10, "Active".

10.1.2.3.7 Outgoing call / U1 call initiated / unknown message received

10.1.2.3.7.1 Definition

The call control entity of the UE being in the state, U1, an unknown message is received by the UE.

10.1.2.3.7.2 Conformance requirement

1) If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC entity of the UE in CC-state U1, "Call initiated", upon receipt of a message with message type not defined for the protocol discriminator from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.4.

10.1.2.3.7.3 Test purpose

To verify that a CC entity of the UE in CC-state U1, "Call initiated", upon receipt of a message with message type not defined for the protocol discriminator from its peer entity returns a STATUS message.

10.1.2.3.7.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U1 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U1. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	<	:-	unknown message	message type not defined for PD
2	-:	>	STATUS	cause #97, state U1
3	<	:-	STATUS ENQUIRY	
4	-:	>	STATUS	cause #30, state U1

Specific message contents:

None.

10.1.2.3.7.5 Test requirements

After step 1 and step 3 a CC entity of the UE in CC state U1, "Call initiated", shall return a STATUS message.

10.1.2.4 Outgoing call / U3 UE-Mobile originating call proceeding

10.1.2.4.1 Outgoing call / U3 UE Mobile originating call proceeding / ALERTING received

10.1.2.4.1.1 Definition

The call control entity of the UE being in the state, U3, an ALERTING message is sent to the UE as an indication that a call is being alerted at a called end.

10.1.2.4.1.2 Conformance requirement

1) When the call control entity of the UE in the "call initiated" state or "mobile originating call proceeding" state receives an ALERTING message then, the call control entity of the UE shall stop timer T303 and T310 (if running) and shall enter the "call delivered" state. A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", upon receipt of an ALERTING message shall enter CC state U4, "Call Delivered".

References

TS 24.008 clause 5.2.1.5.

10.1.2.4.1.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon receipt of a ALERTING message enters CC-state U4, "Call Delivered".

10.1.2.4.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends an ALERTING message to the UE. The SS checks by using the status enquiry procedure that the CC entity has entered the state U4, call delivered.

Expected sequence

Step	Direc	tion	Message	Comments
	UE	SS		
1	<	-	ALERTING	
2			STATUS ENQUIRY	
3			STATUS	cause #30, state U4

Specific message contents:

None.

10.1.2.4.1.5 Test requirements

After step 1 a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", shall enter CC-state U4, "Call Delivered".

10.1.2.4.2 Outgoing call / U3 UE Mobile originating call proceeding / CONNECT received

10.1.2.4.2.1 Definition

The call control entity of the UE being in the state, U3, a CONNECT message is received by the UE.

10.1.2.4.2.2 Conformance requirement

1) The call control entity of the UE in the "call initiated" state, in the "mobile originating call proceeding" state or in the "call delivered" state, shall, upon receipt of a CONNECT message:

- attach the user connection;
- return a CONNECT ACKNOWLEDGE message;
- stop any locally generated alerting indication (if applied);
- stop timer T303 and T310 (if running);
- <u>enter the "active" state.</u> A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", upon receipt of a CONNECT message shall return a "CONNECT ACKNOWLEDGE" message to its peer entity and enter the CC state U10, "Active".

References

Conformance requirement 1: TS 24.008 clause 5.2.1.6.

Conformance requirement 2: TS 24.008 clause 5.2.1.6.

²⁾ The UE shall then stop any locally generated indication.

10.1.2.4.2.3 Test purpose

- To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon receipt of a CONNECT message returns a "CONNECT ACKNOWLEDGE" message to its peer entity and enters the CC state U10, "Active".
- 2) To verify that the UE stops locally generated <u>alerting</u> indication, if any.

10.1.2.4.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The SS sends a CONNECT message to the UE. The UE shall respond by sending a CONNECT ACKNOWLEDGE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U10, active.

Expected sequence

Ste	ер	Direction		Message	Comments
		UE	SS		
1	1			Radio Bearer Setup Procedure	(DTCH), See_TS_34.108_clause 7.1.3
2	2	<	-	CONNECT	the UE shall stop locally generated
					alerting indication tone generation, if any
3	3	-:	>	CONNECT ACKNOWLEDGE	
4	4	<	-	STATUS ENQUIRY	
5	5	-;	>	STATUS	cause #30, state U10

Specific message contents:

None.

10.1.2.4.2.5 Test requirements

After step 1 a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", shall return a "CONNECT ACKNOWLEDGE" message and enter the CC state U10, "Active".

The UE shall stop locally generated <u>alerting</u> indication.

10.1.2.4.3 Outgoing call / U3 <u>UE Mobile</u> originating call proceeding / PROGRESS received without in band information

10.1.2.4.3.1 Definition

The call control entity of the UE being in the state, U3, a PROGRESS message is received by the UE. The PROGRESS message does not contain indication of in-band information availability.

10.1.2.4.3.2 Conformance requirement

1) In order to inform the UE that the call is progressing in the PLMN/ISDN environment the network may send a progress indicator information element to the calling UE either:

a) in an appropriate call control message, if a state change is required (e.g., ALERTING or CONNECT); or

b) in the PROGRESS message, if no state change is appropriate.

This progress indicator information element shall contain progress description value #32 "Call is end-to-end <u>ISDN/PLMN".</u> A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", upon receipt of a PROGRESS message with valid cause values shall stay in CC-state U3.

2) At any time during the establishment or release of a call and during an active call the network may send a <u>PROGRESS message to the UE.</u>

On receipt of a PROGRESS message during the establishment or release of a call the UE shall stop all call control timers related to that call. After receipt of the PROGRESS message timer T310 shall be stopped.

References

Conformance requirement 1: TS 24.008 clause 5.2.1.4.2

Conformance requirement 2: TS 24.008 clause 5.5.611.3.

10.1.2.4.3.3 Test purpose

- 1) To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon receipt of a PROGRESS message with valid cause values stays in CC-state U3.
- 2) To verify that after receipt of the PROGRESS message timer T310 is stopped.

10.1.2.4.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a PROGRESS message not containing

Release 5

1571

indication of in-band information availability to the UE. The SS checks that the UE has stopped T310, i.e. at T310 timeout no DISCONNECT message is sent by the UE. Then the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	PROGRESS	(note)
2	<-	STATUS ENQUIRY	
3	->	STATUS	cause #30, state U3
4	SS		SS waits at least 45 s and checks no DISCONNECT is sent by the UE
5	<-	STATUS ENQUIRY	
6	->	STATUS	cause #30, state U3

NOTE: Tested with a valid Progress Indicator, Progress description value among:

#4 call has returned to PLMN/ISDN;

- #32 call is end-to-end PLMN/ISDN.

Specific message contents:

None.

10.1.2.4.3.5 Test requirements

After step 1 a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", shall stay in CC-state U3.

After step 3 SS waits at least 45 s and checks no DISCONNECT is sent by the UE.

- 10.1.2.4.4 Outgoing call / U3 <u>UE_Mobile</u> originating call proceeding / PROGRESS with in band information
- 10.1.2.4.4.1 Definition

The call control entity of the UE being in the state, U3, a PROGRESS message indicating availability of in band information is received by the UE.

10.1.2.4.4.2 Conformance requirement

- When the network wants to make the UE attach the user connection (e.g. in order to provide in-band tones/announcement) before the UE has reached the "active" state of a call, the network may include a progress indicator IE indicating user attachment in a suitable CC message:
 - Either it includes the IE in a SETUP, CALL PROCEEDING, ALERTING, or CONNECT message that is send during call establishment
 - it sends a PROGRESS message containing the IE. A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", upon receipt of a PROGRESS message indicating in-band announcement shall through connect the traffic channel for speech, if DTCH is in a speech mode. If DTCH is not in speech mode, the UE shall not through connect the DTCH.

On reception of a SETUP, CALL PROCEEDING, ALERTING, CONNECT, or PROGRESS message the UE shall proceed as specified elsewhere in TS 24.008 clause 5; if the progress indicator IE indicated user attachment and a speech mode traffic channel is appropriate for the call the UE shall in addition: attach the user connection for speech as soon as an appropriate channel in speech mode is available.

2) <u>At any time during the establishment or release of a call and during an active call the network may send a</u> <u>PROGRESS message to the UE.</u> On receipt of a PROGRESS message during the establishment or release of a call the UE shall stop all call control timers related to that call. After receipt of the PROGRESS message, T310 shall be stopped.

References

Conformance requirement 1: TS 25.331 clause 8.2.1, TS 24.008 clause 5.5.15.2.1.4, clause 5.2.1.9, clause 5.5.1 and clause 11.3.

Conformance requirement 2: TS 24.008 clause 5.5.6

10.1.2.4.4.3 Test purpose

- 1) To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon receipt of a PROGRESS message indicating in-band announcement through-connects the traffic channel for speech, if DTCH is in speech mode. If DTCH is not in a speech mode, the UE does not through-connect the DTCH.
- 2) To verify that after receipt of the PROGRESS message, T310 is stopped.

10.1.2.4.4.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The SS sends a PROGRESS message containing indication of in-band information availability to the UE. The SS checks that if channel mode is speech, the DTCH shall be through connected. If channel mode is not speech, the DTCH shall not be through connected. Also the SS checks that the UE has stopped T310, i.e. at T310 time-out no DISCONNECT message is sent by the UE. Then the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1			Radio Bearer Setup Procedure	(DTCH), See TS34.108 clause 7.1.3
2	<	:-	PROGRESS	(note)
				the UE shall stop all the CC timers , if
				channel mode is speech, the DTCH
				shall be through connected. If channel
				mode is not speech, the DTCH shall not
				be through connected.
3	<		STATUS ENQUIRY	
4		>	STATUS	cause #30, state U3
5	S	S		SS waits at least 45 s and checks no
				DISCONNECT is sent by the UE.
6	<	:-	STATUS ENQUIRY	
7		>	STATUS	cause #30, state U3
8	S	S		If the channel mode is speech the SS
				will check that the user connection for
				speech is attached (both downlink and
				uplink).

Specific message contents:

NOTE: Tested with a valid Progress Indicator, Progress description value among:

- #1 call is not end to end PLMN/ISDN;
- #2 destination address is non PLMN/ISDN;
- #3 originating address is non PLMN/ISDN;
- #8 in band information or appropriate pattern now available.

10.1.2.4.4.5 Test requirements

After step 2 a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", shall through-connect the traffic channel for speech, if DTCH is in a speech mode. If DTCH is not in speech mode, the UE shall not through-connect the DTCH.

After step 4 the SS waits at least 45 s and checks no DISCONNECT is sent by the UE.

After step 7 the SS checks that the user connection for speech is attached (both downlink and uplink), if the channel mode is speech.

10.1.2.4.5 Outgoing call / U3 UE Mobile originating call proceeding / DISCONNECT with in band tones

10.1.2.4.5.1 Definition

The call control entity of the UE being in the state, U3, a DISCONNECT message indicating availability of in band information is received by the UE.

10.1.2.4.5.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon receipt of a DISCONNECT message with progress indicator #8:

- i) if an appropriate speech traffic channel is not connected, continue clearing as defined in TS24.008 clause 5.4.4.1.2.1 without connecting to the in-band tone/announcement;
- ii) if an appropriate speech traffic channel is connected, attach the user connection for speech if it is not yet attached and enter the "disconnect indication" state. In that state, if upper layers request the clearing of the call, the call

control entity of the UE shall proceed as defined in TS24.008 clause 5.4.4.1.2.1.A CC entity of the UE in CCstate U3, "Mobile Originating Call Proceeding", upon receipt of a DISCONNECT with progress indicator #8, shall through connect the speech channel to make in band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE shall send a RELEASE message.

References

TS 24.008 clause 5.2.1.4 and clause 5.4.4.1.1.1

10.1.2.4.5.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon receipt of a DISCONNECT with progress indicator #8 through-connects the speech channel to make in-band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE sends a RELEASE message.

10.1.2.4.5.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The SS sends a DISCONNECT message containing indication of in-band information availability to the UE. The SS checks that if channel mode is speech, the DTCH shall be through connected and the UE enters state U12, disconnect indication. If channel mode is not speech, the DTCH shall not be through connected and the UE shall enter state U19, release request.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1		Radio Bearer Setup Procedure	(DTCH), See TS_34.108 clause 7.1.3
2	<-	DISCONNECT	(note)
A3	SS		DTCH in speech mode: the SS will check that the audio path for in band tones is attached.
A4	<-	STATUS ENQUIRY	
A5	->	STATUS	cause #30, state U12
			DTCH is not in speech mode:
B3	->	RELEASE	
B4	<-	STATUS ENQUIRY	
B5	->	STATUS	cause #30, state U19

Specific message contents:

- NOTE: the Progress Indicator, Progress description_value:
 - #8 in band information or appropriate pattern now available.

10.1.2.4.5.5 Test requirements

After step 2 a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", shall through-connect the speech channel to make in-band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE shall send a RELEASE message.

10.1.2.4.6 Outgoing call / U3 UE-Mobile originating call proceeding / DISCONNECT without in band tones

10.1.2.4.6.1 Definition

The call control entity of the UE being in the state, U3, a DISCONNECT message is received by the UE. The DISCONNECT message does not contain indication of in-band information availability.

10.1.2.4.6.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon the receipt of a DISCONNECT message without progress indicator information element or with progress indicator different from #8:

- stop all running call control timers;
- send a RELEASE message;
- start timer T308; and
- <u>enter the "release request" state.</u> A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", upon receipt of a DISCONNECT without progress indicator shall return a RELEASE message and enter the CCstate U19, "Release Request"

References

TS 24.008 clause 5.4.4.<u>1.2.1</u>

10.1.2.4.6.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon receipt of a DISCONNECT without progress indicator returns a RELEASE message and enters the CC-state U19, "Release Request".

10.1.2.4.6.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a DISCONNECT message not containing indication of in-band information availability to the UE. The UE shall respond with a RELEASE message. The SS checks by using the status enquiry procedure that the CC entity of the UE has entered the state U19, release request.

Expected sequence

ſ	Step	Direc	tion	Message	Comments
		UE	SS		
11	1	<	-	DISCONNECT	without progress indicator
	2	-;	>	RELEASE	
	3	<	-	STATUS ENQUIRY	
	4	<- ->		STATUS	cause #30, state U19

Specific message contents:

None.

10.1.2.4.6.5 Test requirements

After step 1 a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", shall send a RELEASE message and enter the CC-state U19, "Release Request".

10.1.2.4.7 Outgoing call / U3 Mobile UE originating call proceeding / RELEASE received

10.1.2.4.7.1 Definition

The call control entity of the UE being in the state, U3, a RELEASE message is received by the UE.

10.1.2.4.7.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release the MM connection; and return to the "null" state. A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", upon receipt of a RELEASE will return a RELEASE COMPLETE and enter the CC-state U0, "Null".

2) The UE on returning to the idle mode shall release the MM connection and the CC entities relating to the seven mobile originating transaction identifiers shall be in CC state U0, "Null".

3) On releasing the MM connection, the UE shall wait for MM layer release initiated by the network.

References

Conformance requirement 1: TS 24.008 clause 5.4.3.32 and clause 5.4.4.

Conformance requirement 2: TS 24.008 clause 4.5.3, clause 5.5.3.2 and 8.3.1.

Conformance requirement 3: TS 24.008 clause 5.4.4.1.3 and clause 4.5.3, TS 25.331 clause 8.1.4.

10.1.2.4.7.3 Test purpose

1) To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon receipt of a RELEASE will return a RELEASE COMPLETE and enter the CC-state U0, "Null".

2) To verify that the UE on returning to the idle mode releases the MM-connection and that the CC-entities relating to the seven mobile originating transaction identifiers are in CC-state U0, "Null".

3) To verify that in releasing the MM connection, the UE shall wait for MM layer release initiated by SS.

10.1.2.4.7.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a RELEASE message to the UE. The UE shall respond with a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	RELEASE	with cause "Normal, unspecified"
2	->	RELEASE COMPLETE	
3	<-	STATUS ENQUIRY	
4	->	RELEASE COMPLETE	cause #81 (invalid TI value)
5	SS		repeat steps 3-4 to cover all the transaction identifiers from 000110
6	<-	RRC CONNECTION RELEASE Void	The SS releases the RRC connection.the main signalling link shall
7	→	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.2.4.7.5 Test requirements

After step 1 a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", shall send a RELEASE COMPLETE message.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.4.8 Outgoing call / U3 Mobile UE originating call proceeding / termination requested by the user

10.1.2.4.8.1 Definition

The call control entity of the UE being in the state, U3, the user requests to terminate the call.

10.1.2.4.8.2 Conformance requirement

1) Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and entering the "disconnect request" state. A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", upon request by the user to terminate will send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

References

TS 24.007 clause 6.2.2., TS 24.008 clause 5.4.3.1

10.1.2.4.8.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", upon request by the user to terminate will send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.2.4.8.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

—The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The user requests termination of the call. The UE shall send a DISCONNECT message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U11, disconnect request.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1			MMI action, terminate call
2	->	DISCONNECT	
3	<-	STATUS ENQUIRY	
4	->	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.2.4.8.5 Test requirements

After step 1 a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", shall send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.2.4.9 Outgoing call / U3 Mobile UE originating call proceeding / traffic channel allocation

10.1.2.4.9.1 Definition

The call control entity of the UE being in the state, U3, a radio bearer establishment procedure is performed.

10.1.2.4.9.2 Conformance requirement

1) It is a network dependent decision when to initiate the assignment of an appropriate traffic channel during the mobile originating call establishment phase. Initiation of a suitable RR procedure to assign an appropriate traffic channel does neither change the state of a call control entity nor affect any call control timer. A CC-entity of the UE in CC state U3, "Mobile Originating Call Proceeding", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, shall stay in CC state U3.

References

TS 25.331 clause 8.2.1, TS 24.008 clause 5.2.1.9.

10.1.2.4.9.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, stays in CC-state U3.

10.1.2.4.9.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1			Radio Bearer Setup Procedure	(DTCH), See TS_34.108 clause 7.1.3
2	<	:-	STATUS ENQUIRY	
3			STATUS	cause #30, state U3

Specific message contents:

None.

10.1.2.4.9.5 Test requirements

After step 1 the CC state U3, "Mobile Originating Call Proceeding", shall remain unchanged.

10.1.2.4.10 Outgoing call / U3 Mobile UE originating call proceeding / timer T310 time-out

10.1.2.4.10.1 Definition

The call control entity of the UE being in the state, U3, if no response is then received from the SS, timer T310 expires at the UE side.

10.1.2.4.10.2 Conformance requirement

- 1) 1) If timer T310 elapses before any of the ALERTING, CONNECT or DISCONNECT messages has been received, the UE shall perform the clearing procedure described in TS 24.008 clause 5.4. A CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding" shall, upon expiry of timer T310, and not before, initiate call release by sending DISCONNECT and enter the CC state U11, "Disconnect Request".
- 2) Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and entering the "disconnect request" state.

References

Conformance requirement 1: TS 24.008 clause 5.2.1.3./Abnormal case,

Conformance requirement 2: TS 24.008 clause 5.4.3.1 and clause 11.3.

10.1.2.4.10.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding" will, upon expiry of timer T310, initiate call release by sending DISCONNECT and enter the CC-state U11, "Disconnect Request".

10.1.2.4.10.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The T310 expires at the UE and the UE shall send DISCONNECT. The SS checks timer T310 accuracy and that the CC entity has entered the state U11, disconnect request.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	SS		the SS waits for T310 time-out
2	->	DISCONNECT	check the timer T310 accuracy, see
			TS34.108 clause 4.2.3
3	<-	STATUS ENQUIRY	
4	->	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.2.4.10.5 Test requirements

After step 1 Uupon expiry of timer T310, a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding" shall initiate call release by sending a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.2.4.11 Outgoing call / U3 Mobile UE originating call proceeding / lower layer failure

10.1.2.4.11.1 Definition

The call control entity of the UE being in the state, U3, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.2.4.11.2 Conformance requirement

1) The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. If a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding" has detected a lower layer failure and has returned to idle mode, the CC entity is in state U0, "Null".

References

TS 24.008 clause 4.5.2.3, and clause 4.5.3 and clause 5.5.3.2, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.2.4.11.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding" having detected a lower layer failure and having returned to idle mode, the CC entity is in state U0, "Null".

10.1.2.4.11.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The UE is brought to the state U3. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS remodifies the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	S	S		SS modifies the scrambling code of DPCH for generating lower layer failure
2	->	>	CELL UPDATE	CCCH
3	<	-	RRC CONNECTION RELEASE	CCCH
4	S	S		SS re-modifies the scrambling code of DPCH to the original one.
5	S	S		SS waits 60 s. UE shall send no message on DCCH

Specific message contents:

None.

10.1.2.4.11.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.2.4.12 Outgoing call / U3 Mobile UE originating call proceeding / unknown message received

10.1.2.4.12.1 Definition

The call control entity of the UE being in the state, U3, an unknown message is received by the UE.

10.1.2.4.12.2 Conformance requirement

1)—If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding" having received an unknown message from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.54.

10.1.2.4.12.3 Test purpose

To verify that a CC-entity of the UE in CC-state U3, "Mobile Originating Call Proceeding" having received an unknown message from its peer entity returns a STATUS message.

10.1.2.4.12.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U3. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	<-		unknown message	message type not defined for PD
2	-;	>	STATUS	cause #97, state U3
3	<	:-	STATUS ENQUIRY	
4	-:	>	STATUS	cause #30, state U3

Specific message contents:

None.

10.1.2.4.12.5 Test requirements

After step 1 a CC entity of the UE in CC state U3, "Mobile Originating Call Proceeding", shall return a STATUS message.

10.1.2.4.13 Outgoing call / U3 Mobile UE originating call proceeding / Internal alerting indication

10.1.2.4.13.1 Definition

The call control entity of the UE being in the state, U3, an ALERTING message is sent to the UE when the user connection is not attached to the radio path.

10.1.2.4.13.2 Conformance requirement

1) When the call control entity of the UE in the "call initiated" state or "mobile originating call proceeding" state receives an ALERTING message then, the call control entity of the UE shall stop timer T303 and T310 (if running) and shall enter the "call delivered" state. In this state, for speech calls:

an alerting indication should be given to the user. If the UE has not attached the user connection then the UE shall internally generate an alerting indication. If the UE has attached the user connection then the network is responsible for generating the alerting indication and the UE need not generate one. When the call control entity of the UE in the "mobile originating call proceeding" state receives an ALERTING message then it shall enter "call delivered" state and, for speech calls, if the user connection is not attached to the radio path, the UE shall internally generate an alerting indication.

References

TS 24.008 clause 5.2.1.5.

10.1.2.4.13.3 Test purpose

When the call control entity of the UE in the "mobile originating call proceeding" state receives an ALERTING message then it enters "call delivered" state and, for speech calls, if the user connection is not attached to the radio path, the UE generates internally an alerting indication.

10.1.2.4.13.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.
- way to give internally generated alerting indication for outgoing calls.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U3 by using table 10.1.2/1.

Test procedure

The SS sends an ALERTING message to the UE. The SS checks by using the status enquiry procedure that the CC entity has entered the state U4, call delivered. Also it is checked that the UE generates internally alerting indication to the user in the way described in the ICS/IXIT statements.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	<	:-		the UE shall generate an alerting indication to the user in the way described in the ICS/IXIT statements
2	<	-	STATUS ENQUIRY	
3	->	>	STATUS	cause #30, state U4

Specific message contents:

None.

10.1.2.4.13.5 Test requirements

After step 1 CC entity of the UE in CC state U3, the "Mobile Originating Call Proceeding" shall enter "Call Delivered" state and, for speech calls, if the user connection is not attached to the radio path, the UE shall internally generate an alerting indication.

10.1.2.5 Outgoing call / U4 call delivered

10.1.2.5.1 Outgoing call / U4 call delivered / CONNECT received

10.1.2.5.1.1 Definition

The call control entity of the UE being in the state, U4, a CONNECT message is received by the UE.

10.1.2.5.1.2 Conformance requirement

1) The call control entity of the UE in the "call initiated" state, in the "mobile originating call proceeding" state or in the "call delivered" state, shall, upon receipt of a CONNECT message:

- attach the user connection;
- return a CONNECT ACKNOWLEDGE message:
- stop any locally generated alerting indication (if applied);
- stop timer T303 and T310 (if running);
- enter the "active" state. A CC entity of the UE in CC state U4, "Call Delivered", upon receipt of the CONNECT message shall return a CONNECT ACKNOWLEDGE to its peer entity and enter the CC-state U10, "Active".

References

TS 24.008 clause 5.2.1.6.

10.1.2.5.1.3 Test purpose

To verify that a CC-entity of the UE in CC-state U4, "Call Delivered", upon receipt of the CONNECT message returns a CONNECT ACKNOWLEDGE to its peer entity and enters the CC-state U10, "Active".

10.1.2.5.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U4. The SS sends a CONNECT message to the UE. The UE shall respond by sending a CONNECT ACKNOWLEDGE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U10, active.

Expected sequence

ſ	Step	Direc	ction	Message	Comments
		UE	SS		
	1	<-		CONNECT	
	2	-:	>		UE stops locally generated alerting indication, if applicable
	3	<	-	STATUS ENQUIRY	
	4	-;	>	STATUS	cause #30, state U10

Specific message contents:

None.

10.1.2.5.1.5 Test requirements

After step 1 a CC entity of the UE in CC state U4, "Call Delivered", shall return a CONNECT ACKNOWLEDGE message and enter the CC state U10, "Active".

10.1.2.5.2 Outgoing call / U4 call delivered / termination requested by the user

10.1.2.5.2.1 Definition

The call control entity of the UE being in the state, U4, the user requests to terminate the call.

10.1.2.5.2.2 Conformance requirement

1) <u>Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate</u> clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and <u>entering the "disconnect request" state.</u> A CC entity of the UE in CC state U4, "Call Delivered", upon request by the user to terminate shall send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.4.3.1

10.1.2.5.2.3 Test purpose

To verify that a CC-entity of the UE in CC-state U4, "Call Delivered", upon request by the user to terminate will send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.2.5.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U4. The user requests termination of the call. The UE shall send a DISCONNECT message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U11, disconnect request.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1				MMI action, terminate call
2	-:	>	DISCONNECT	
3	<	:-	STATUS ENQUIRY	
4	-:	>	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.2.5.2.5 Test requirements

After step 1 a CC entity of the UE in CC state U4, "Call Delivered", shall send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

10.1.2.5.3 Outgoing call / U4 call delivered / DISCONNECT with in band tones

10.1.2.5.3.1 Definition

The call control entity of the UE being in the state, U4, a DISCONNECT message indicating availability of in band information is received by the UE.

10.1.2.5.3.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon receipt of a DISCONNECT message with progress indicator #8:

- i) if an appropriate speech traffic channel is not connected, continue clearing as defined in TS 24.008 clause 5.4.4.1.2.1 without connecting to the in-band tone/announcement;
- ii) if an appropriate speech traffic channel is connected, attach the user connection for speech if it is not yet attached and enter the "disconnect indication" state. In that state, if upper layers request the clearing of the call, the call control entity of the UE shall proceed as defined in TS 24.008 clause 5.4.4.1.2.1.A CC entity of the UE in CCstate U4, "Call Delivered" shall, upon receipt of a DISCONNECT with a progress indicator indicating in band information, shall through connect the speech channel to make in band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE shall send a RELEASE message.

References

TS 25.331 clause 8.2.1., TS 24.008 clause 5.4.4.1.1.1, clause 5.5.1 and clause 5.2.1.9.

10.1.2.5.3.3 Test purpose

To verify that a CC-entity of the UE in CC-state U4, "Call Delivered", upon receipt of a DISCONNECT with a progress indicator indicating in-band information, through-connects the speech channel to make in-band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE shall send a RELEASE message.

10.1.2.5.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U4. The SS sends a DISCONNECT message containing indication of in-band information availability to the UE. The SS checks that if channel mode is MO telephony, the DTCH shall be through connected and the UE enters state U12, disconnect indication. If channel mode is not speech, the DTCH shall not be through connected and the UE shall enter state U19, release request.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	<	-	DISCONNECT	(note)
				DTCH in speech mode:
A2	SS			the SS will check that the audio path for
				in band tones is attached.
A3	<-		STATUS ENQUIRY	
A4	->		STATUS	cause #30, state U12
				DTCH is not in speech mode:
B2	-:	>	RELEASE	
B3	<	-	STATUS ENQUIRY	
B4	-:	>	STATUS	cause #30, state U19

Specific message contents:

NOTE: the Progress Indicator, Progress Description value:

- #8 in band information or appropriate pattern now available.

10.1.2.5.3.5 Test requirements

After step 1 a CC entity of the UE in CC state U4, "Call Delivered", shall through-connect the speech channel to make in-band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE shall send a RELEASE message.

10.1.2.5.4 Outgoing call / U4 call delivered / DISCONNECT without in band tones

10.1.2.5.4.1 Definition

The call control entity of the UE being in the state, U4, a DISCONNECT message is received by the UE. The DISCONNECT message does not contain indication of in-band information availability.

10.1.2.5.4.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon the receipt of a DISCONNECT message without progress indicator information element or with progress indicator different from #8:

- stop all running call control timers;

- send a RELEASE message;

- start timer T308; and

<u>enter the "release request" state.</u> A CC entity of the UE in CC state U4, "Call Delivered", upon receipt of a DISCONNECT without progress indicator, shall return a RELEASE message and enter the CC state U19, "Release Request".

References

TS 24.008 clause 5.4.4.<u>1.2.1</u>

10.1.2.5.4.3 Test purpose

To verify that a CC-entity of the UE in CC-state U4, "Call Delivered", upon receipt of a DISCONNECT without progress indicator, returns a RELEASE message and enters the CC-state U19, "Release Request".

10.1.2.5.4.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U4. The SS sends a DISCONNECT message not containing indication of in-band information availability to the UE. The UE shall respond with a RELEASE message. The SS checks by using the status enquiry procedure that the CC entity of the UE has entered the state U19, release request.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	<	:-	DISCONNECT	without progress indicator
2	-:	>	RELEASE	
3	<	-	STATUS ENQUIRY	
4	-:	>	STATUS	cause #30, state U19

Specific message contents:

None.

10.1.2.5.4.5 Test requirements

After step 1 a CC entity of the UE in CC state U4, "Call Delivered", shall return a RELEASE message and enter the CC-state U19, "Release Request".

10.1.2.5.5 Outgoing call / U4 call delivered / RELEASE received

10.1.2.5.5.1 Definition

The call control entity of the UE being in the state, U4, a RELEASE message is received by the UE.

10.1.2.5.5.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release the MM connection; and return to the "null" state. A CC-entity of the UE in CC-state U4, "Call Delivered", upon receipt of the RELEASE message shall respond with the RELEASE COMPLETE message and enter the CC state U0, "Null".

2) The UE on returning to idle mode shall release the MM connection and the CC entities relating to the seven mobile originating transaction identifiers shall be in CC state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.2-3.3 and clause 5.4.4.

Conformance requirement 2: TS 24.008 clause 4.5.3, clause 5.5.3.2 and 8.3.1.

10.1.2.5.5.3 Test purpose

1) To verify that a CC-entity of the UE in CC-state U4, "Call Delivered", upon receipt of the RELEASE message will respond with the RELEASE COMPLETE message and enter the CC-state U0, "Null".

2) To verify that the UE on returning the idle mode releases the MM connection and that the CC entities relating to the seven mobile originating transaction identifiers are in CC state U0, "Null".

10.1.2.5.5.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U4. The SS sends a RELEASE message to the UE. The UE shall respond with a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1 2	<	>	RELEASE RELEASE COMPLETE	with cause "Normal, unspecified"
3 4 5	<- -> <mark>SS</mark>		STATUS ENQUIRY RELEASE COMPLETE	cause #81 (invalid TI value) repeat steps 3-4 to cover all the
6	<mark>6</mark>		Void RRC CONNECTION RELEASE <u>Void</u>	transaction identifiers from 000110 The SS releases the RRC connection.the main signalling link shall
7	_	>	RRC CONNECTION RELEASE COMPLETEVoid	be released.

Specific message contents:

None.

10.1.2.5.5.5 Test requirements

After step 1 a CC entity of the UE in CC state U4, "Call Delivered", shall respond with the RELEASE COMPLETE message.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.5.6 Outgoing call / U4 call delivered / lower layer failure

10.1.2.5.6.1 Definition

The call control entity of the UE being in the state, U4, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.2.5.6.2 Conformance requirement

1) The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. When CC entity of the UE in CC state U4, "Call Delivered" has detected a lower layer failure and has returned to idle mode, the CC entity is in CC state U0, "Null".

References

TS 24.008 clause 4.5.2.3,- and clause 4.5.3 and clause 5.5.3.2., TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.2.5.6.3 Test purpose

To verify that a CC-entity of the UE in CC-state U4, "Call Delivered" having detected a lower layer failure and has returned to idle mode, the CC-entity is in CC-state U0, "Null".

10.1.2.5.6.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The UE is brought to the state U4. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS remodifies the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	S	S		SS modifies the scrambling code of DPCH for generating lower layer failure
2	-:	>	CELL UPDATE	CCCH
3	<	-	RRC CONNECTION RELEASE	СССН
4	S	S		SS re-modifies the scrambling code of DPCH to the original one.
5	SS			SS waits 60 s. UE shall send no message on the DCCH

Specific message contents:

None.

10.1.2.5.6.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.2.5.7 Outgoing call / U4 call delivered / traffic channel allocation

10.1.2.5.7.1 Definition

The call control entity of the UE being in the state, U4, a radio bearer establishment procedure is performed.

10.1.2.5.7.2 Conformance requirement

1) It is a network dependent decision when to initiate the assignment of an appropriate traffic channel during the mobile originating call establishment phase. Initiation of a suitable RR procedure to assign an appropriate traffic channel does neither change the state of a call control entity nor affect any call control timer. A CC entity of the UE in

CC state U4, "Call Delivered", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, shall stay in CC state U4.

References

TS 25.331 clause 8.2.1., TS 24.008 clause 5.2.1.9.

10.1.2.5.7.3 Test purpose

To verify that a CC-entity of the UE in CC-state U4, "Call Delivered", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, stays in CC-state U4.

10.1.2.5.7.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U4. The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

	Step	Direc	ction	Message	Comments
		UE	SS		
ĺ	1			Radio Bearer Setup Procedure	DTCH, See TS_34.108 clause 7.1.3
	2	<	-	STATUS ENQUIRY	
	3	-;	>	STATUS	cause #30, state U4

Specific message contents:

None.

10.1.2.5.7.5 Test requirements

After step 1 the CC state U4, "Call delivered", shall remain unchanged.

10.1.2.5.8 Outgoing call / U4 call delivered / unknown message received

10.1.2.5.8.1 Definition

The call control entity of the UE being in the state, U4, an unknown message is received by the UE.

10.1.2.5.8.2 Conformance requirement

1) <u>If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the</u> receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC entity of the UE in CC state U4, "Call Delivered", having received an unknown message from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.4.

10.1.2.5.8.3 Test purpose

To verify that a CC-entity of the UE in CC-state U4, "Call Delivered", having received an unknown message from its peer entity returns a STATUS message.

10.1.2.5.8.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U4 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U4. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direc	tion	Message	Comments
	UE	SS		
1	<-		unknown message	message type not defined for PD
2	-;	>	STATUS	cause #97, state U4
3	<-		STATUS ENQUIRY	
4			STATUS	cause #30, state U4

Specific message contents:

None.

10.1.2.5.8.5 Test requirements

After step 1 a CC entity of the UE in CC state U4, "Call Delivered",, shall return a STATUS message.

Release 5

10.1.2.6 U10 call active

10.1.2.6.1 U10 call active / termination requested by the user

10.1.2.6.1.1 Definition

The call control entity of the UE being in the state, U10, the user requests to terminate the call.

10.1.2.6.1.2 Conformance requirement

1) Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and entering the "disconnect request" state. A CC entity of the UE in CC state U10, "Call Active", upon request by the user to terminate shall send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.4.3.1

10.1.2.6.1.3 Test purpose

To verify that the a CC-entity of the UE in CC-state U10, "Call-Active", upon request by the user to terminate will send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.2.6.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U10. The user requests termination of the call. The UE shall send a DISCONNECT message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U11, disconnect request.

Expected sequence

Γ	Step	Direction		Message	Comments
		UE	SS		
ſ	1	•			MMI action, terminate call
	2	-:	>	DISCONNECT	U11
	3	<	:-	STATUS ENQUIRY	
	4	-:	>	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.2.6.1.5 Test requirements

After step 1 a CC entity of the UE in CC state U10, "Call Active", shall send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

10.1.2.6.2 U10 call active / RELEASE received

10.1.2.6.2.1 Definition

The call control entity of the UE being in the state, U10, a RELEASE message is received by the UE.

10.1.2.6.2.2 Conformance requirement

1)-The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release the MM connection; and return to the "null" state. A CC entity of the UE in CC state U10, "Call Active", upon receipt of the RELEASE shall respond with the RELEASE COMPLETE message and enter the CC state U0, "Null"

2) When the UE returns to the idle mode it shall release the MM connection and the CC entities relating to the seven mobile originating transaction identifiers shall be in CC state U0, "Null"

References

Conformance requirement 1: TS 24.008 clause 5.4.2-3.3 and clause 5.4.4.

Conformance requirement 2: TS 24.008 clauses 4.5.3, 5.5.3.2 and 8.3.1.

10.1.2.6.2.3 Test purpose

1) To verify that the a CC-entity of the UE in CC-state U10, "Call-Active", upon receive of the RELEASE will respond with the RELEASE COMPLETE message and enter the CC-state U0, "Null"

2) To verify that the UE on returning to the idle mode releases the MM connection and that the CC entities relating to the seven mobile originating transaction identifiers are in CC state U0, "Null"

10.1.2.6.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U10. The SS sends a RELEASE message to the UE. The

Release 5

UE shall respond with a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	RELEASE	with cause "Normal, unspecified"
2	->	RELEASE COMPLETE	the UE starts T3240
3	<-	STATUS ENQUIRY	
4	->	RELEASE COMPLETE	cause #81 (invalid TI value)
5	<mark>SS</mark>	Void	repeat steps 3-4 to cover all the
6	<-	RRC CONNECTION RELEASE Void	transaction identifiers from 000110 The SS releases the RRC connection.the main signalling link shall be released.
7	→	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.2.6.2.5 Test requirements

After step 1 a CC entity of the UE in CC state U10, "Call Active", shall return a RELEASE COMPLETE message.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.6.3 U10 call active / DISCONNECT with in band tones

10.1.2.6.3.1 Definition

The call control entity of the UE being in the state, U10, a DISCONNECT message indicating availability of in band information is received by the UE.

10.1.2.6.3.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon receipt of a DISCONNECT message with progress indicator #8:

- i) if an appropriate speech traffic channel is not connected, continue clearing as defined in TS 24.008 clause 5.4.4.1.2.1 without connecting to the in-band tone/announcement;
- ii) if an appropriate speech traffic channel is connected, attach the user connection for speech if it is not yet attached and enter the "disconnect indication" state. In that state, if upper layers request the clearing of the call, the call control entity of the UE shall proceed as defined in TS 24.008 clause 5.4.4.1.2.1.A CC entity of the UE in CCstate U10, "Call Active", upon receipt of a DISCONNECT message with a Progress Indicator indicating in band information, shall through connect the speech channel to make in band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE shall send a RELEASE message.

References

TS 24.008 clause 5.4.4.1.1.1.1 and clause 5.5.1.

10.1.2.6.3.3 Test purpose

To verify that a CC-entity of the UE in CC-state U10, "Call-Active", upon receipt of a DISCONNECT message with a Progress Indicator indicating in-band information, through-connects the speech channel to make in-band announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE sends a RELEASE message.

10.1.2.6.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U10. The SS sends a DISCONNECT message containing indication of in-band information availability to the UE. The SS checks that if channel mode is speech, the DTCH shall be through connected and the UE enters state U12, disconnect indication. If channel mode is not speech, the DTCH shall not be through connected and the UE enters state U19, release request.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	DISCONNECT	(note)
A2	SS		DTCH in speech mode: the SS will check that the audio path for in band tones is attached.
A3	<-	STATUS ENQUIRY	
A4	->	STATUS	cause #30, state U12
			DTCH is not in speech mode:
B2	->	RELEASE	
B3	<-	STATUS ENQUIRY	
B4	->	STATUS	cause #30, state U19

Specific message contents:

NOTE: the Progress Indicator, Progress Description<u>value</u>:

#8 in band information or appropriate pattern now available.

10.1.2.6.3.5 Test requirements

After step 1 a CC entity of the UE in CC state U10, "Call Active", shall through-connect the speech channel to make inband announcements available, if traffic channel is in speech mode. If DTCH is not in speech mode, the UE shall send a RELEASE message.

10.1.2.6.4 U10 call-active / DISCONNECT without in band tones

10.1.2.6.4.1 Definition

The call control entity of the UE being in the state, U10, a DISCONNECT message is received by the UE. The DISCONNECT message does not contain indication of in-band information availability.

10.1.2.6.4.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon the receipt of a DISCONNECT message without progress indicator information element or with progress indicator different from #8:

stop all running call control timers;

- send a RELEASE message;

- start timer T308; and

<u>enter the "release request" state.</u> A CC entity of the UE in CC state U10, "Call Active", upon receipt of a DISCONNECT message without progress indicator, shall return a RELEASE message and enter the CC-state U19, "Release Request".

References

TS 24.008 clause 5.4.4.<u>1.2.1</u>

10.1.2.6.4.3 Test purpose

To verify that the a CC-entity of the UE in CC-state U10, "Call-Active", upon receipt of a DISCONNECT message without progress indicator, returns a RELEASE message and enters the CC-state U19, "Release Request".

10.1.2.6.4.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U10. The SS sends a DISCONNECT message not containing indication of in-band information availability to the UE. The UE shall respond with a RELEASE message. The SS checks by using the status enquiry procedure that the CC entity of the UE has entered the state U19, release request.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	DISCONNECT	without progress indicator
2	->	RELEASE	
3	<-	STATUS ENQUIRY	
4	->	STATUS	cause #30, state U19

Specific message contents:

None.

10.1.2.6.4.5 Test requirements

After step 1 A CC entity of the UE in CC state U10, "Call Active", shall return a RELEASE message and enter the CCstate U19, "Release Request".

10.1.2.6.5 U10 call-active / RELEASE COMPLETE received

10.1.2.6.5.1 Definition

The call control entity of the UE being in the state, U10, the call is cleared by a RELEASE COMPLETE message sent by the SS.

10.1.2.6.5.2 Conformance requirement

1) <u>A call control entity shall accept an incoming RELEASE COMPLETE message used to initiate the call clearing even though the cause information element is not included. A CC entity of the UE in CC state U10, "active", upon receipt of a RELEASE COMPLETE message with valid cause value, shall enter CC state U0, "Null".</u>

2) A call control entity of the UE in any call control state shall, upon receipt of a RELEASE COMPLETE message from its peer entity in the network: stop all running call control timers ; release the MM connection; and return to the "null" state. On returning to idle mode, the CC entities relating to the seven mobile originating transaction identifiers shall be in state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.2 and clause 5.4.4.

Conformance requirement 2: TS 24.008 clause 5.4.4.1.3 and clause 8.3.1.

10.1.2.6.5.3 Test purpose

- 1) To verify that a CC entity of the UE in CC-state U10, "Call-active" upon receipt of a RELEASE COMPLETE message with valid cause value, enters CC state U0, "Null".
- 2) To verify that in returning to idle mode, the CC entities relating to the seven mobile originating transaction identifiers are in state U0, "Null".

10.1.2.6.5.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 by using table 10.1.2/1.

Test procedure

The SS sends a RELEASE COMPLETE message to the UE. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	RELEASE COMPLETE	note 1
2	<-	STATUS ENQUIRY	note 2
3	->	RELEASE COMPLETE	cause #81 (invalid TI value),
4	SS		repeat steps 2-3 to cover all the
			transaction identifiers from 000110
5	<-	RRC CONNECTION RELEASE Void	The SS releases the RRC
			connection.the main signalling link shall
			be released.
6	→	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

NOTE 1: With the cause value chosen arbitrarily or cause value not included.

NOTE 2: TI flag has the value indicating the UE as an originator of the call.

10.1.2.6.5.5 Test requirements

After step 2 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.6.6 U10 call active / SETUP received

10.1.2.6.6.1 Definition

If the UE does not react correctly when receiving a SETUP message on a new Transaction Identifier during an active call, the active call may be lost.

10.1.2.6.6.2 Conformance requirement

 A busy UE which satisfies the compatibility requirements indicated in the SETUP message shall respond either with a CALL CONFIRMED message if the call setup is allowed to continue or a RELEASE COMPLETE message if the call setup is not allowed to continue, both with cause #17 "user busy". A UE that has a call established when receiving a SETUP message shall respond either with a CALL CONFIRMED message or a RELEASE COMPLETE message, both with cause #17 "user busy".

2) The call control state of the existing transaction shall not be affected by the incoming SETUP message.

Reference(s):

Conformance requirement 1: TS 24.008 clause 5.2.2.3.1.

Conformance requirement 2: TS 24.008 clause 5.1.1.

10.1.2.6.6.3 Test purpose

- To verify that a User Equipment that has a call established and receives a SETUP message answers either with a CALL CONFIRMED message with cause "user busy" if it supports call waiting, or with a RELEASE COMPLETE message with cause "user busy" otherwise.
- 2) To verify that after having sent this message, the UE is still in state U10 for the established call.

10.1.2.6.6.4 Method of test

Related ICS/IXIT statement(s)

- supported MO circuit switched basic services.

- support of call waiting Y/N.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is idle updated with valid TMSI and CKSN.

The UE is brought into the state U10 by using table 10.1.2/1.

Test Procedure

The UE has a mobile originated call in the U10 state. When UE sends a SETUP message and SS receives it in the first call establishment, SS sends a CALL PROCEEDING message without Network Call Control Capabilities IE.

The SS sends a SETUP message to the UE (with signal IE indicating "call waiting tone on" and without Network Call Control Capabilities IE).

If the UE does not support call waiting it shall answer by a RELEASE COMPLETE message.

If the UE supports call waiting it shall answer by a CALL CONFIRMED message followed by an ALERTING. The second transaction is then released by the SS with a RELEASE COMPLETE message.

In both cases the SS checks by using the status enquiry procedure that the CC entity of the UE is still in state U10, active call for the original call.

Step	Direc	ction	Message	Comments
	UE	SS		
1	<	:-	SETUP	this message establishes a second transaction The TI value shall be the same as the one that is in use for the MO call. The TI flag shall have the value specified for an MT call.
A2	-:	>	RELEASE COMPLETE	if the UE does not support call waiting with cause "user busy" with the TI of the second transaction
B2	-:	>	CALL CONFIRMED	if the UE supports call waiting with cause "user busy" with the TI of the second transaction
B3	-:	>	ALERTING	with the TI of the second transaction
B4	<	-	RELEASE COMPLETE	with the TI of the second transaction
5	<	-	STATUS ENQUIRY	with the TI of the original transaction
6	->	>	STATUS	cause #30, state U10 with the TI of the original transaction

Expected sequence

NOTE: The Transaction Identifier of the second transaction shall be different from the one of the already established transaction.

Specific message contents

SETUP message contains a Signal IE with value "call waiting tone on" (H'07).

10.1.2.6.6.5 Test requirements

After step 1 a UE that has a call established shall answer either with a CALL CONFIRMED message with cause "user busy" if it supports call waiting, or with a RELEASE COMPLETE message with cause "user busy" otherwise.

After step A2 or B2-B4 the UE is stillshall be in state U10 for the established call.

- 10.1.2.7 U11 disconnect request
- 10.1.2.7.1 U11 disconnect request / clear collision
- 10.1.2.7.1.1 Definition

The call control entity of the UE being in the state, U11, a DISCONNECT message is received by the UE.

10.1.2.7.1.2 Conformance requirement

1)-The call control entity of the UE in the "disconnect request" state, shall, upon receipt of a DISCONNECT message:

- stop all running call control timers;
- send a RELEASE message;
- start timer T308; and
- <u>enter the "release request" state.</u> A CC entity of the UE in CC state U11, "Disconnect Request", upon receipt of a DISCONNECT message, shall return to its peer entity the RELEASE message and enter the CC state U19, "Release Request".

References

TS 24.008 clause 5.4.4.2.<u>52</u>.1.

10.1.2.7.1.3 Test purpose

To verify that the a CC-entity of the UE in CC-state U11, "Disconnect Request", upon receipt of a DISCONNECT message, returns to its peer entity the RELEASE message and enters the CC-state U19, "Release Request".

10.1.2.7.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U11 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U11. The SS sends a DISCONNECT message to the UE.

Release 5

1604

The UE shall respond with a RELEASE message. The SS checks by using the status enquiry procedure that the CC entity of the UE has entered the state U19, release request.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	DISCONNECT	
2	->	RELEASE	
3	<-	STATUS ENQUIRY	
4	->	STATUS	cause #30, state U19

Specific message contents:

None.

10.1.2.7.1.5 Test requirements

After step 1 a CC entity of the UE in CC state U11, "Disconnect Request", shall return the RELEASE message and enter the CC-state U19, "Release Request".

10.1.2.7.2 U11 disconnect request / RELEASE received

10.1.2.7.2.1 Definition

The call control entity of the UE being in the state, U11, a RELEASE message is received by the UE.

10.1.2.7.2.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release the MM connection; and return to the "null" state. A CC entity of the UE in CC state U11, "Disconnect Request", upon receipt of the RELEASE message shall return RELEASE COMPLETE and enter the CC state U0, "Null".

2) On returning to the idle mode the UE shall release the MM connection and the CC entities relating to the seven mobile originating transaction identifiers shall be in CC state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.3.3

Conformance requirement 2: TS 24.008 clause 4.5.3, clause 5.5.3.2 and clause 8.3.1.

10.1.2.7.2.3 Test purpose

1) To verify that the a CC-entity of the UE in CC-state U11, "Disconnect Request", upon receipt of the RELEASE message shall return RELEASE COMPLETE and enter the CC-state U0, "Null".

 To verify that the UE on returning to the idle mode releases the MM connection and that the CC entities relating to the seven mobile originating transaction identifiers are in CC state U0, "Null".

10.1.2.7.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U11 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U11. The SS sends a RELEASE message to the UE. The UE shall respond with a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	RELEASE	
2	->	RELEASE COMPLETE	
3	<-	STATUS ENQUIRY	
4	->	RELEASE COMPLETE	cause #81 (invalid TI value)
5	<mark>SS</mark>	Void	repeat steps 3-4 to cover all the
			transaction identifiers from 000110
6	<-	RRC CONNECTION RELEASE Void	The SS releases the RRC
			connection.the main signalling link shall
			be released.
7	→	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.2.7.2.5 Test requirements

After step 1 a CC entity of the UE in CC state U11, "Disconnect Request", shall return the RELEASE COMPLETE.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.7.3 U11 disconnect request / timer T305 time-out

10.1.2.7.3.1 Definition

The call control entity of the UE being in the state, U11, if no response is then received from the SS, timer T305 expires at the UE side.

10.1.2.7.3.2 Conformance requirement

1) The call control entity of the UE in the "disconnect request" state, shall upon expiry of timer T305: send a RELEASE message to the network with the cause number originally contained in the DISCONNECT message and optionally, a second cause information element with cause #102 "recovery on timer expiry", start timer T308, and enter the "release request" state. A CC entity of the UE in CC state U11, "Disconnect Request" shall on expiry of T305, proceed with the connection release procedure by sending the RELEASE message to its peer entity and shall enter the CC state U19, "Release Request".

References

TS 24.008 clause 5.4.3.5 and clause 11.3.

10.1.2.7.3.3 Test purpose

To verify that the CC-entity of the UE in CC-state U11, "Disconnect Request" shall on expiry of T305, proceed with the connection release procedure by sending the RELEASE message to its peer entity and enters the CC-state U19, "Release Request".

10.1.2.7.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U11 by using table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U11. Then T305 expires at the UE and the UE shall send a RELEASE message. The SS checks timer T305 accuracy and that the CC entity has entered the state U19, release request.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	S	S		SS waits until T305 expires at the UE
2	-:	>	RELEASE	SS checks the time between
				DISCONNECT and RELEASE (note),
				check the timer T305 accuracy, see
				TS34.108 clause 4.2.3
3	<	:-	STATUS ENQUIRY	
4	-:	>	STATUS	cause #30, state U19

Specific message contents:

NOTE: With the same cause value as originally contained in the DISCONNECT message. An additional cause information element (#102 recovery on timer expiry) may be included.

10.1.2.7.3.5 Test requirements

<u>After step 1 Uupon expiry of timer T305 a CC entity of the UE in CC state U11, "Disconnect Request", shall proceed</u> with the connection release procedure by sending the RELEASE message and enter the CC-state U19, "Release Request".

10.1.2.7.4 U11 disconnect request / lower layer failure

10.1.2.7.4.1 Definition

The call control entity of the UE being in the state, U11, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.2.7.4.2 Conformance requirement

1) The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. A CC entity of the UE in CC state U11, "Disconnect Request" having detected a lower layer failure shall return to the idle mode. The CC entity is in state U0, "Null".

References

TS 24.008 clause 4.5.2.3, and clause 4.5.3, clause 5.5.3.2 and clause 8.3, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.2.7.4.3 Test purpose

To verify that the a CC-entity of the UE in CC-state U11, "Disconnect Request" having detected a lower layer failure returns to the idle mode. The CC entity is thus in state U0, "Null".

10.1.2.7.4.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U11 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The UE is brought to the state U11. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS re-modifies the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	S	S		SS modifies the scrambling code of DPCH for generating lower layer failure
2	-:	>	CELL UPDATE	СССН
3	<	-	RRC CONNECTION RELEASE	CCCH
4	S	S		SS re-modifies the scrambling code of DPCH to the original one.
5	S	S		SS waits 60 s.

Specific message contents:

None.

Release 5

1608

10.1.2.7.4.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.2.7.5 U11 disconnect request / unknown message received

10.1.2.7.5.1 Definition

The call control entity of the UE being in the state, U11, an unknown message is received by the UE.

10.1.2.7.5.2 Conformance requirement

1) If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC entity of the UE in CC state U11, "Disconnect Request", having received an unknown message from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.4.

10.1.2.7.5.3 Test purpose

To verify that a CC-entity of the UE in CC-state U11, "Disconnect Request", having received an unknown message from its peer entity returns a STATUS message.

10.1.2.7.5.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U11 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U11. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	<	:-	unknown message	message type not defined for PD
2	-;	>	STATUS	cause #97, state U11
3	<	-	STATUS ENQUIRY	
4	-;	>	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.2.7.5.5 Test requirements

After step 1 a CC entity of the UE in CC state U11, "Disconnect Request", shall return a STATUS message.

10.1.2.8 U12 disconnect indication

10.1.2.8.1 U12 disconnect indication / call releasing requested by the user

10.1.2.8.1.1 Definition

The call control entity of the UE being in the state, U12, the user requests to terminate the call.

10.1.2.8.1.2 Conformance requirement

1) <u>Response from the upper layers:</u>

i) If the upper layers request the clearing of the call, the call control entity of the UE shall:

- stop all running call control timers;
- send a RELEASE message;
- start timer T308; and
- <u>enter the "release request" state.</u> A CC entity of the UE in CC state U12, "Disconnect Indication" being in network initiated call release phase, shall, upon receiving a call release request from the user send a RELEASE to its peer entity and enter CC state U19, "Release Request".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.4.4.2.2.1

10.1.2.8.1.3 Test purpose

To verify that a CC-entity of the UE in CC-state U12, "Disconnect Indication" being in network initiated call release phase, shall, upon receiving a call release request from the user sends a RELEASE to its peer entity and enters CC-state U19, "Release Request"

10.1.2.8.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U12 by using Option A of table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U12. The user requests termination of the call. The UE shall send a RELEASE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U19, release request.

Expected sequence

Step	Direc	tion	Message	Comments
	UE	SS		
1				MMI action, "on hook"
2	->	•	RELEASE	
3	<-	-	STATUS ENQUIRY	
4	->	•	STATUS	cause #30, state U19

Specific message contents:

None.

10.1.2.8.1.5 Test requirements

After step 1 a CC entity of the UE in CC state U12, "Disconnect Indication" being in network initiated call release phase, shall send a RELEASE message and enter CC-state U19, "Release Request".

10.1.2.8.2 U12 disconnect indication / RELEASE received

10.1.2.8.2.1 Definition

The call control entity of the UE being in the state, U12, a RELEASE message is received by the UE.

10.1.2.8.2.2 Conformance requirement

1)—The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release the MM connection; and return to the "null" state. A CC entity of the UE in CC state U12, "Disconnect Indication", upon receipt of a RELEASE message shall return to its peer entity the RELEASE COMPLETE message and enter the CC state U0, "Null".

2) On returning to the idle mode the UE shall release the MM-connection and the CC-entities relating to the seven mobile originating transaction identifiers shall be in CC state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.23.3

Conformance requirement 2: TS 24.008 clause 4.5.3, clause 5.5.3.2 and clause 8.3.1.

10.1.2.8.2.3 Test purpose

- 1) To verify that a CC-entity of the UE in CC-state U12, "Disconnect Indication", upon receipt of a RELEASE message returns to its peer entity the RELEASE COMPLETE message and enters the CC-state U0, "Null".
- 2) To verify that the UE on returning to the idle mode releases the MM-connection and that the CC-entities relating to the seven mobile originating transaction identifiers are in CC-state U0, "Null".

10.1.2.8.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U12 by using Option A of table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U12. The SS sends a RELEASE message to the UE. The UE shall respond with a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Direc	ction	Message	Comments
	UE	SS		
1	<	:-	RELEASE	
2	-:	>	RELEASE COMPLETE	
3	<	-	STATUS ENQUIRY	
4	-:	>	RELEASE COMPLETE	cause #81 (invalid TI value)
5	S	S		repeat steps 3-4 to cover all the
6	<	:-	RRC CONNECTION RELEASE Void	transaction identifiers from 000110 <u>The SS releases the RRC</u> <u>connection.the main signalling link shall</u> be released.
7		>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.2.8.2.5 Test requirements

After step 1 a CC entity of the UE in CC state U12, "Disconnect Indication", shall return the RELEASE COMPLETE message.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.8.3 U12 disconnect indication / lower layer failure

10.1.2.8.3.1 Definition

The call control entity of the UE being in the state, U12, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.2.8.3.2 Conformance requirement

1) The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. A CC entity of the UE in CC state U12, "Disconnect Indication" having detected a lower layer failure shall return to idle mode. The CC entity is in state U0, "Null".

References

TS 24.008 clause <u>4.5.2.3</u>5.4.4.2.5.1, and clause 4.5.3, clause 5.5.3.2 and clause 8.3, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.2.8.3.3 Test purpose

To verify that a CC-entity of the UE in CC-state U12, "Disconnect Indication" having detected a lower layer failure returns to idle mode. The CC-entity is thus in state U0, "Null".

10.1.2.8.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U12 by using Option A of table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The UE is brought to the state U12. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS re-modifies the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direc	tion	Message	Comments
	UE	SS		
1	S	S		SS modifies the scrambling code of
				DPCH for generating lower layer failure
2	-:	>	CELL UPDATE	CCCH
3	<	-	RRC CONNECTION RELEASE	CCCH
4	S	S		SS re-modifies the scrambling code of
				DPCH to the original one.
5	S	S		SS waits 60 s.
				UE shall send no message on the
				DCCH

Specific message contents:

None.

10.1.2.8.3.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.2.8.4 U12 disconnect indication / unknown message received

10.1.2.8.4.1 Definition

The call control entity of the UE being in the state, U12, an unknown message is received by the UE.

10.1.2.8.4.2 Conformance requirement

If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC entity of the UE in CC state U12, "Disconnect Indication" having received an unknown message from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.4.

10.1.2.8.4.3 Test purpose

To verify that a CC-entity of the UE in CC-state U12, "Disconnect Indication" having received an unknown message from its peer entity returns a STATUS message.

10.1.2.8.4.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U12 by using Option A of table 10.1.2/3.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U12. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direc	tion	Message	Comments
	UE	SS		
1	<-		unknown message	message type not defined for PD
2	->		STATUS	cause #97, state U12
3	<-		STATUS ENQUIRY	
4	->	•	STATUS	cause #30, state U12

Specific message contents:

None.

10.1.2.8.4.5 Test requirements

After step 1 a CC entity of the UE in CC state U12, "Disconnect Indication", shall return a STATUS message.

10.1.2.9 Outgoing call / U19 release request

10.1.2.9.1 Outgoing call / U19 release request / timer T308 time-out

10.1.2.9.1.1 Definition

The call control entity of the UE being in the state, U19, if no response is then received from the SS, timer T308 expires at the UE side.

10.1.2.9.1.2 Conformance requirement

1)—The call control entity of the UE in the "release request" state shall at first expiry of timer T308 retransmit the <u>RELEASE message and restart timer T308.A CC entity of the UE in CC state U19, "Release Request" will, upon</u> the first expiry of timer T308 send the RELEASE message to its peer entity and remain in the CC state U19.

References

TS 24.008 clause 5.4.4.1.3.1 and clause 11.3.

10.1.2.9.1.3 Test purpose

To verify that a CC-entity of the UE in CC-state U19, "Release Request" will, upon the first expiry of timer T308 send the RELEASE message to its peer entity and remain in the CC-state U19.

10.1.2.9.1.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U19 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U19. When T308 expires at the UE, the UE shall send a RELEASE message. The SS checks timer T308 accuracy and that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	S	S		SS waits until T308 at the UE
2	-:	>	RELEASE	SS checks the time between the two
				RELEASE messages
				check the timer T308 accuracy, see
				TS34.108 clause 4.2.3
3	<	-	STATUS ENQUIRY	
4	-:	>	STATUS	cause #30, state U19

Specific message contents:

None.

10.1.2.9.1.5 Test requirements

After step 1 Uupon the first expiry of timer T308 (after step 1) a CC entity of the UE in CC state U19, "Release Request", shall send the RELEASE message and remain in the CC-state U19.

10.1.2.9.2 Outgoing call / U19 release request / 2nd timer T308 time-out

10.1.2.9.2.1 Definition

The call control entity of the UE being in the state, U19, if no response is then received after timer T308 has expired two times in success at the UE.

10.1.2.9.2.2 Conformance requirement

1) <u>At second expiry of timer T308, the call control entity of the UE shall: release the MM connection; and return to the</u> <u>"null" state.</u> <u>A CC entity of the UE in CC state U19, "Release Request", upon the 2nd expiry of the timer T308, shall</u> enter the CC-state U0, "Null".

2) Subsequently the UE shall proceed with releasing the MM connection and enter the idle mode with the CC entities relating to the seven mobile originating transaction identifiers in state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.4.1.3.1 and clause 11.3.

Conformance requirement 2: TS 24.008 clause 4.5.3, clause 5.5.3.2 and 8.3.1.

10.1.2.9.2.3 Test purpose

- 1) To verify that a CC-entity of the UE in CC-state U19, "Release Request", upon the 2nd expiry of the timer T308, enters the CC-state U0, "Null".
- 2) To verify that subsequently the UE proceeds with releasing the MM-connection and enters the idle mode with the CC entities relating to the seven mobile originating transaction identifiers in state U0, "Null".

10.1.2.9.2.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U19 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U19. The SS allows T308 expiry at the UE, and the UE shall repeat sending the RELEASE message and start timer T308 again. The SS allows again T308 expiry at the UE. The UE shall abort the RRC connection. The SS waits long enough to enable the UE to return to idle state listening to paging, and then pages UE to create RRC connection. Finally, the SS will check the state of the UE by using STATUS ENQUIRY with the relevant transaction identifiers.

Expected sequence

Γ	Step	Direc	ction	Message	Comments
		UE	SS		
Ī	1	SS			SS waits until T308 expiry at the UE
	2	-;	>	RELEASE	
	3	<	-	STATUS ENQUIRY	
	4	-:		STATUS	cause #30, state U19
	5	S	S		SS waits until the second T308 expiry at
					the UE
.	6	S	-		SS waits T3240 expiry at the UE
	7	U	E		The SS releases the RRC
					connection.the main signalling link shall
	0	0	~		be released.
	8	S	S		SS waits 10 s for the UE to return to
	0			Mabile terminated establishment of Dadia Descures	listening to paging
	9			Mobile terminated establishment of Radio Resource Connection	See TS34.108
	9a	-3		PAGING RESPONSE	
	9a 10	<		STATUS ENQUIRY	
	10	-		RELEASE COMPLETE	cause #81 (invalid TI value)
	12	ss			repeat steps 10-11 to cover all the
	14	00			transaction identifiers from 000110
	13	<-		RRC CONNECTION RELEASEVoid	The SS releases the RRC
				·····	connection.the main signalling link shall
					be released.
	14	-;	>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.2.9.2.5 Test requirements

After step 5 Uupon the 2nd expiry of the timer T308 (after step 5) a CC entity of the UE in CC state U19, "Release Request", shall enter the CC-state U0, "Null".

After step 10 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.9.3 Outgoing call / U19 release request / RELEASE received

10.1.2.9.3.1 Definition

The call control entity of the UE being in the state, U19, a RELEASE message is received by the UE.

10.1.2.9.3.2 Conformance requirement

1) Clear collision can also occur when both sides simultaneously transfer RELEASE messages related to the same call. The entity receiving such a RELEASE message whilst within the "release request" state shall: stop timer T308; release the MM connection; and enter the "null" state (without sending a RELEASE COMPLETE message). A CC entity of the UE in CC state U19, "Release Request", upon receipt of a RELEASE, shall release the MM connection and enter the CC state U0, "Null" with the CC entities relating to the seven mobile originating transaction identifiers in state U0, "Null".

References

TS 24.008 clause 5.4.4.2.5.1, clause 11.3, clause 5.5.3.2 and 8.3.1.

10.1.2.9.3.3 Test purpose

To verify that a CC-entity of the UE in CC-state U19, "Release Request", upon receipt of a RELEASE, shall release the MM-connection and enters the CC-state U0, "Null" with the CC entities relating to the seven mobile originating transaction identifiers in state U0, "Null".

10.1.2.9.3.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U19 by using table 10.1.2/4.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U19. The SS sends a RELEASE message to the UE. The UE shall release the MM-connection. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

ſ	Step	Direction	Message	Comments
		UE SS		
1	1	<-	RELEASE	(note)
	2	<-	STATUS ENQUIRY	
	3	->	RELEASE COMPLETE	cause #81 (invalid TI value)
	4	SS		repeat steps 2-3 to cover all the
	5	<-	RRC CONNECTION RELEASE Void	transaction identifiers from 000110 <u>The SS releases the RRC</u> <u>connection.the main signalling link shall</u> <u>be released.</u>
	6	→	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

NOTE: With the same cause number as originally contained in DISC and optional cause #102 recovery on timer expiry.None

10.1.2.9.3.5 Test requirements

After step 2 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.9.4 Outgoing call / U19 release request / RELEASE COMPLETE received

10.1.2.9.4.1 Definition

The call control entity of the UE being in the state, U19, a RELEASE COMPLETE message is received by the UE.

10.1.2.9.4.2 Conformance requirement

- A call control entity shall accept an incoming RELEASE COMPLETE message used to initiate the call clearing even though the cause information element is not included. A CC-entity of the UE in CC-state U19, "Release Request", upon receipt of a RELEASE COMPLETE, shall release the MM connection and enter the CC state U0, "Null" with the CC entities relating to the seven mobile originating transaction identifiers in state U0, "Null".
- 2) A call control entity of the UE in any call control state shall, upon receipt of a RELEASE COMPLETE message from its peer entity in the network: stop all running call control timers ; release the MM connection; and return to the "null" state.

References

Conformance requirement 1: TS 24.008 clause 5.4.2

Conformance requirement 2: TS 24.008 clause 5.4.4.1.3, clause 4.5.3 and clause 8.3.1.

10.1.2.9.4.3 Test purpose

To verify that a CC-entity of the UE in CC-state U19, "Release Request", upon receipt of a RELEASE COMPLETE, shall release the MM-connection and enters the CC-state U0, "Null" with the CC entities relating to the seven mobile originating transaction identifiers in state U0, "Null".

10.1.2.9.4.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U19 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The CC entity of the UE is brought to the state U19. The SS sends a RELEASE COMPLETE message to the UE. The UE shall release the MM-connection. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<	:-	RELEASE COMPLETE	
2	<-		STATUS ENQUIRY	
3	-:	>	RELEASE COMPLETE	cause #81 (invalid TI value)
4	S	S		repeat steps 2-3 to cover all the
5	<	-	RRC CONNECTION RELEASE Void	transaction identifiers from 000110 the main signalling link shall be released. The SS releases the RRC connection.
6	Ŧ	>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.2.9.4.5 Test requirements

After step 2 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.2.9.5 Outgoing call / U19 release request / lower layer failure

10.1.2.9.5.1 Definition

The call control entity of the UE being in the state, U19, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.2.9.5.2 Conformance requirement

The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. A CC entity of the UE in CC state U19, "Release Request", having detected a lower layer failure, shall return to the idle mode, the CC entity is in state U0, "Null".

References

TS 24.008 clause 4.5.2.3, and clause 4.5.3, clause 5.5.3.2 and clause 8.3, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.2.9.5.3 Test purpose

To verify that a CC-entity of the UE in CC-state U19, "Release Request", having detected a lower layer failure, returns to the idle mode, the CC entity is in state U0, "Null".

10.1.2.9.5.4 Method of test

Related ICS/IXIT statements

- supported MO circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U19 by using table 10.1.2/1.

Test procedure

An MO circuit switched basic service is selected that is supported by the UE; if the UE supports MO telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then, the UE is made to initiate a call. The UE is brought to the state U19. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS re-modifies the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	SS			SS modifies the scrambling code of DPCH for generating lower layer failure
2	->		CELL UPDATE	CCCH
3	<-		RRC CONNECTION RELEASE	СССН
4	S	S		SS re-modifies the scrambling code of DPCH to the original one.
5	SS			SS waits 60 s. UE shall send no message on the DCCH

Specific message contents:

None.

10.1.2.9.5.5 Test requirements

After step 4 CC the UE shall not send any message to the SS during 60 s.

10.1.3 Establishment of an incoming call / Initial conditions

The tables below describe message exchanges which bring the UE in the requested initial states in case of an incoming call.

A state may be taken as initial only when all the states which lead to this initial states have been validated. The order will be U0, U6, U9, U7, U8, U10, U26 etc. as in the following tables.

Step	Dire	ction	Message	Comments	
	UE	SS			
1			Mobile terminated establishment of Radio Resource	See TS 34.108 <u>clause 7.1.2</u>	
			Connection	Establishment cause: Terminating Conversational Call.	
2	-	>	PAGING RESPONSE		
3		<-	AUTHENTICATION REQUEST		
4	-	·>	AUTHENTICATION RESPONSE		
5		<-	SECURITY MODE COMMAND		
6	-	·>	SECURITY MODE COMPLETE		
7		<-	SETUP	U6, (note 1)	
8	-	·>	CALL CONFIRMED	U9	
A9	-	·>	CONNECT	U8, p = Y, (note 2)	
B9	-	·>	ALERTING	U7, p = N, (note 2)	
B10	ι	JE		(note 3)	
B11	-	·>	CONNECT	U8	
12			Radio Bearer Setup Procedure	DTCH, See TS 34.108 clause 7.1.3	
13		<-	CONNECT ACKNOWLEDGE	U10	
NOTE '	1: Wit	h signa			
NOTE 2	The UE is supporting immediate connect (p = Y/N). See ICS/IXIT statement.				
NOTE 3	3: If n	ecessa	ry (see ICS/IXIT statement), the UE is made to accept	the call in the way described in a	
	ICS	S/IXIT s	tatement.		

Table 10.1.3/2: Establishment of an incoming call, procedure 2

Step	Dire	ction	Message	Comments
	UE	SS		
1			Mobile terminated establishment of Radio Resource Connection	See TS34.108 <u>clause 7.1.2</u> Establishment cause: Terminating Conversational Call.
2	-	·>	PAGING RESPONSE	
3		<-	SECURITY MODE COMMAND	
4		·>	SECURITY MODE COMPLETE	
5		<-	SETUP	U6, (note 1)
6		·>		
A7	-	·>	CONNECT	U8, $p = Y$, (note 2)
A8 B7			Radio Bearer Setup Procedure	DTCH, See TS34.108 <u>clause 7.1.3</u>
B8	•	·>	Radio Bearer Setup Procedure	U7, p = N, (note 2) DTCH, See TS34.108 clause 7.1.3
B9	ı	JE	Radio Bearer Setup Flocedure	(note 3)
B10	-	·>	CONNECT	U8
11		<-	AUTHENTICATION REQUEST	
12	-	·>	AUTHENTICATION RESPONSE	
13		<-	CONNECT ACKNOWLEDGE	U10
NOTE 1			al information included in the SETUP message.	
NOTE 2			supporting immediate connect (p = Y/N). See ICS/IXI	T statement.
NOTE 3			ary (see ICS/IXIT statement), the UE is made to accept	t the call in the way described in a
	ICS	S/IXIT s	statement.	

Table 10.1.3/3: Void

Table 10.1.3/4: Establishment of an incoming call, procedure 4

Step	Dire	ction	Message	Comments
	UE	SS		
1			Mobile terminated establishment of Radio Resource	See TS 34.108 clause 7.1.2
			Connection	Establishment cause: Terminating
				Conversational Call.
2	-	·>	PAGING RESPONSE	
3		<-	SECURITY MODE COMMAND	
4	-	·>	SECURITY MODE COMPLETE	
5		<-	SETUP	U6, (note 1)
6	-	·>	CALL CONFIRMED	U9
7			Radio Bearer Setup Procedure	DTCH, See TS 34.108 clause 7.1.3
A8		·>	CONNECT	U8, p = Y, (note 2)
B8		·>	ALERTING	U7, p = N, (note 2)
B9	ι	JE		(note 3)
B10		·>	CONNECT	U8
11		<-	AUTHENTICATION REQUEST	
12		·>	AUTHENTICATION RESPONSE	
13		<-	CONNECT ACKNOWLEDGE	U10
NOTE 1	NOTE 1: The signa		al information element is not included in the SETUP message.	
NOTE 2	TE 2: The UE is		s supporting immediate connect (p = Y/N). See ICS/IXIT statement.	
NOTE 3	B: Ifn	ecessa	ary (see ICS/IXIT statement), the UE is made to accept	the call in the way described in a
	ICS	S/IXIT s	statement.	

10.1.3.1 Incoming call / U0 null state

10.1.3.1.1 Incoming call / U0 null state / SETUP received with a non supported bearer capability

10.1.3.1.1.1 Definition

The call control entity of the UE being in the state, U0, a SETUP message is received with only one bearer capability and this bearer capability is not supported by the UE.

10.1.3.1.1.2 Conformance requirement

1) When the network is providing a basic service at the called side, the UE shall check that the basic service(s) offered by the network in the Bearer Capability information element(s) match(es) the basic services that the UE is able to support. If a mismatch is detected, then the UE shall proceed as follows:

- if the SETUP message contained two bearer capability information elements for only one of which a mismatch is detected, the UE shall either:
 - under the conditions specified in 3GPP TS 27.001 (e.g. TS 61 and TS 62), accept the SETUP message with a CALL CONFIRMED message containing the, possibly negotiated, bearer capability information element for which no mismatch is detected, or
 - reject the call using cause No. 88 "incompatible destination".
- otherwise the UE shall reject the offered call using a RELEASE COMPLETE message with cause No. 88
 <u>"incompatible destination"</u>. A CC entity of the UE, upon receipt of SETUP containing one bearer capability and this bearer capability is not supported, shall return a RELEASE COMPLETE with correct cause value to its peer entity and return to the idle mode. The CC entities relating to the seven mobile terminating transaction identifiers shall be in the state U0,"Null".

References

TS 24.008 clause 5.2.2.2, clause 8.3.1 and annex B.<u>3.2</u>

10.1.3.1.1.3 Test purpose

To verify that a CC entity of the UE, upon receipt of SETUP containing one bearer capability and this bearer capability is not supported, returns a RELEASE COMPLETE with correct cause value to its peer entity, and returns to the idle mode. To verify that the CC-entities relating to the seven mobile terminating transaction identifiers are then in the state U0, "Null".

10.1.3.1.1.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

Test procedure

A mobile terminated call is initiated. The UE receives a SETUP message that contains a bearer capability not supported by the UE. The UE returns a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity is still in the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1			Mobile terminated establishment of Radio Resource	SS sends paging, See TS34.108
			Connection	
2	->	PAGING RESPONSE		
3	<-		AUTHENTICATION REQUEST	
4	->	AUTHENTICATION RESPONSE		
5	<	-	SECURITY MODE COMMANDVoid	The SS starts integrity protection.
6	-	>	SECURITY MODE COMPLETEVoid	
7	<	-	SETUP	(note 1)
8	-:	>	RELEASE COMPLETE	(note 2)
9	<	-	STATUS ENQUIRY	
10	-:	>	RELEASE COMPLETE	Cause #81 (invalid TI value).
11	S	S		Repeat steps 9-10 to cover all the transaction identifiers from 000 110.

Specific message contents:

- NOTE 1: With one bearer capability and that bearer capability is not supported by the UE.
- NOTE 2: With cause #88 incompatible destination.

10.1.3.1.1.5 Test requirements

After step 7 a CC entity of the UE shall return a RELEASE COMPLETE message with cause value #88 (incompatible destination) and return to the idle mode.

After step 9 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.3.2 Incoming call / U6 call present

10.1.3.2.1 Incoming call / U6 call present / automatic call rejection

10.1.3.2.1.1 Definition

Although the state U6 is transient, the ability to refuse a call (automatically) in this state is tested, if it is implemented at the UE.

10.1.3.2.1.2 Conformance requirement

1)—If the mobile user wishes to refuse the call, a RELEASE COMPLETE message shall be sent with the cause #21 "call rejected". A CC entity of the UE in CC state U6, "Call Present", upon receipt of a rejection indication of the incoming call from the user, send RELEASE COMPLETE with the appropriate cause value to its peer entity and enter the CC-state U0, "Null". The CC entities relating to the seven mobile terminating transaction identifiers shall be in state U0, "Null".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.2.2.3.1, clause 5.5.3.2 and clause 8.3.1.

10.1.3.2.1.3 Test purpose

To verify that a CC entity of the UE in CC-state U6, "Call Present", shall upon receipt of a rejection indication of the incoming call from the user, shall send RELEASE COMPLETE with the appropriate cause value to its peer entity and enter the CC-state U0, "Null". The CC entities relating to the seven mobile terminating transaction identifiers are then in state U0, "Null".

10.1.3.2.1.4 Method of test

Related ICS/IXIT statements

- supported teleservices;
- the UE supports an ability to refuse a call after receipt of a SETUP message.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U6 by using table 10.1.3/2.

Test procedure

A teleservice is selected that is supported by the UE; if the UE supports speech, the selected teleservice is speech. If necessary, the UE is configured for that teleservice. Then a mobile terminated call is initiated. The call control entire of the UE is brought to the state U6 (Note: The state U6 is not checked, since it is not stable). The UE is made to refuse the call (the refusal may require some preliminary preparations in order to achieve refusal at this point). The UE shall send a RELEASE COMPLETE message and enter a call control state U0. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1 2 3	->		RELEASE COMPLETE STATUS ENQUIRY	the UE is made to refuse the call (note)
3 4 5	-	<- -> SS	RELEASE COMPLETE	cause #81 (invalid TI value) repeat steps 3-4 to cover all the
6		<-	RRC CONNECTION RELEASE Void	transaction identifiers from 000110 the main signalling link shall be released. <u>The SS releases the RRC</u> connection.
7		→	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

NOTE: With cause value #21 call rejected.

10.1.3.2.1.5 Test requirements

After step 1 a CC entity of the UE shall return a RELEASE COMPLETE message with cause value #21 (call rejected) and return to the idle mode.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.3.3 Incoming call / U9 mobile terminating call confirmed

- 10.1.3.3.1 Incoming call / U9 mobile terminating call confirmed / alerting or immediate connecting
- 10.1.3.3.1.1 Definition

The call control entity of the UE having entered the state, U9, with signal information received in the preceding SETUP message, the subsequent behaviour of the UE is tested.

10.1.3.3.1.2 Conformance requirement

 The call control entity of the UE having entered the "mobile terminating call confirmed" state, if the call is accepted at the called user side, the UE proceeds as described in TS 24.008 clause 5.2.2.5. Otherwise, if the signal information element was present in the SETUP message user alerting is initiated at the UE side; if the signal information element was not present in the SETUP message, user alerting is initiated when an appropriate channel is available.

Here, initiation of user alerting means:

- the generation of an appropriate tone or indication at the UE; and
- <u>sending of an ALERTING message by the call control entity of the MS to its peer entity in the network and entering the "call received" state.</u> A CC entity in CC state U9, "Mobile Terminating Call Confirmed", (if signalled by the network in previous SETUP message that it may alert) shall either send a ALERTING message to its peer entity and enter state U7, or send a CONNECT message to its peer entity and enter U8.
- 2) In the "mobile terminating call confirmed" state or the "call received" state, the call control entity in the UE indicates acceptance of a mobile terminating call by:
 - sending a CONNECT message to its peer entity in the network;
 - starting Timer T313; and

- entering the "connect request" state.

References

Conformance requirement 1: TS 24.008 clause 5.2.2.3.2 and

Conformance requirement 2: TS 24.008 clause 5.2.2.5.

10.1.3.3.1.3 Test purpose

To verify that a CC entity in CC-state U9, "Mobile Terminating Call Confirmed", (if signalled by the network in previous SETUP message that it may alert) will either send a ALERTING message to its peer entity and enter state U7, or send a CONNECT message to its peer entity and enter U8.

10.1.3.3.1.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U9 by using table 10.1.3/2.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U9 by using a SETUP message containing signalling information element. (The state U9 is not a stable state in this case, and consequently it is not checked as an initial state.) If the UE supports immediate connect for the selected basic service (p = Y), it sends a CONNECT message and enters the state U8, connect request. Otherwise (p = N) the UE sends an ALERTING message and enters the state U7, call receiving the status enquiry procedure that the CC entity has entered its state as described.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
A1	->		CONNECT	p = Y
A2	<-		STATUS ENQUIRY	
A3	->		STATUS	cause #30, state U8
B1	->		ALERTING	p = N
B2	<-		STATUS ENQUIRY	
B3	-	>	STATUS	cause #30, state U7

Specific message contents:

None.

10.1.3.3.1.5 Test requirements

At step A1 the UE shall send a CONNECT message and enter U8 if the network has signalled in previous SETUP message that UE may not alert. CC entity in CC state U9, "Mobile Terminating Call Confirmed", (if signalled by the network in previous SETUP message that it may alert)

At step B1 the UE shall either send an ALERTING message and enter state U7 if the network has signalled in previous SETUP message that UE may alert. , or send a CONNECT message and enter U8,

10.1.3.3.2 Incoming call / U9 mobile terminating call confirmed / DTCH assignment

10.1.3.3.2.1 Definition

The call control entity of the UE being in the state, U9, a radio bearer establishment procedure is performed for traffic channel.

10.1.3.3.2.2 Conformance requirement

1) It is a network dependent decision when to initiate the assignment of a traffic channel during the mobile terminating call establishment phase.

Initiation of the assignment phase does not directly change the state of a CC entity nor affect any call control timer, but may have some secondary effects (see e.g. TS 24.008 clause 5.2.2.3.2). A CC entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, shall send an ALERTING message and enter state U7.

2) The call control entity of the UE having entered the "mobile terminating call confirmed" state, if the call is accepted at the called user side, the UE proceeds as described in TS24.008 clause 5.2.2.5. Otherwise, if the signal information element was present in the SETUP message user alerting is initiated at the UE side; if the signal information element was not present in the SETUP message, user alerting is initiated when an appropriate channel is available.

Here, initiation of user alerting means:

- the generation of an appropriate tone or indication at the UE; and
- sending of an ALERTING message by the call control entity of the MS to its peer entity in the network and entering the "call received" state.

References

TS 25.331 clause 8.2.1, Conformance requirement 1: TS 24.008 clause 5.2.2.7 and

Conformance requirement 2: TS 24.008 clause 5.2.2.3.2.

10.1.3.3.2.3 Test purpose

To verify that a CC-entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, shall sends an ALERTING message and enters state U7.

10.1.3.3.2.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U9 by using table 10.1.3/4.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U9 (by using a SETUP message not containing the signal information element). The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The UE sends an ALERTING message and enters state U7, call received. The SS verifies by using the status enquiry procedure that the UE has entered the correct state.

Expected sequence

	Step	Direction		Message	Comments
		UE	SS		
	1			Radio Bearer Setup Procedure	DTCH, See TS 34.108 clause 7.1.3
	2	->		ALERTING	
	3	<-		STATUS ENQUIRY	
	4	->		STATUS	cause #30, state U7

Specific message contents:

None.

10.1.3.3.2.5 Test requirements

After step 1 a CC entity of the UE in CC state U9, "Mobile Terminating Call Confirmed" shall send an ALERTING message and enter state U7.

- 10.1.3.3.3 Incoming call / U9 mobile terminating call confirmed / termination requested by the user
- 10.1.3.3.3.1 Definition

The call control entity of the UE being in the state, U9, the user requests for releasing of the call.

10.1.3.3.3.2 Conformance requirement

1)—Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and entering the "disconnect request" state. A CC-entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed", upon request by the user to terminate shall send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.4.3.1

10.1.3.3.3.3 Test purpose

To verify that a CC-entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed", upon request by the user to terminate will send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.3.3.3.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used;
- the UE supports user requested call clearing in the state U9.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U9 by using table 10.1.3/4.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U9 (by using a SETUP message not containing the signal information element). Then the user requests termination of the call, if possible. The UE sends a DISCONNECT message and enters state U11, disconnect request. The SS verifies by using the status enquiry procedure that the UE has entered the correct state.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1				the UE is made to clear the call
2	->		DISCONNECT	
3	<-		STATUS ENQUIRY	
4	->		STATUS	cause #30, state U11

Specific message contents:

None.

10.1.3.3.3.5 Test requirements

After step 1 a CC entity of the UE in CC state U9, "Mobile Terminating Call Confirmed", shall send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.3.3.4 Incoming call / U9 mobile terminating call confirmed / DISCONNECT received

10.1.3.3.4.1 Definition

The call control entity of the UE being in the state, U9, a DISCONNECT message is received by the UE.

10.1.3.3.4.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon the receipt of a DISCONNECT message without progress indicator information element or with progress indicator different from #8:

- stop all running call control timers;
- send a RELEASE message;
- start timer T308; and
- <u>enter the "release request" state.</u> A CC entity of the UE in CC state U9, "Mobile Terminating Call Confirmed", upon receipt of a DISCONNECT shall return a RELEASE message and enter the CC state U19, "Release Request".

References

TS 24.008 clause 5.4.4.1.2.1

10.1.3.3.4.3 Test purpose

To verify that a CC-entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed", upon receipt of a DISCONNECT returns a RELEASE message and enters the CC-state U19, "Release Request".

10.1.3.3.4.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U9 by using table 10.1.3/4.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U9. The SS sends a DISCONNECT message to the UE. The UE responds by sending a RELEASE message and enters state U19, release request. The SS verifies by using the status enquiry procedure that the UE has entered the correct state.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<-		DISCONNECT	
2	->		RELEASE	
3	<-		STATUS ENQUIRY	
4	->		STATUS	cause #30, state U19

Specific message contents:

None.

10.1.3.3.4.5 Test requirements

After step 1 a CC entity of the UE in CC state U9, "UE Terminating Call Confirmed", shall return a RELEASE message and enter the CC-state U19, "Release Request".

10.1.3.3.5 Incoming call / U9 mobile terminating call confirmed / RELEASE received

10.1.3.3.5.1 Definition

The call control entity of the UE being in the state, U9, a RELEASE message is received by the UE.

10.1.3.3.5.2 Conformance requirement

1)—The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release the MM connection; and return to the "null" state. A CC entity of the UE in CC state U9, "Mobile Terminating Call Confirmed", upon receipt of a RELEASE shall return a RELEASE COMPLETE and enter the CC state U0, "Null".

2) On returning to the idle mode the UE shall release the MM connection and the CC entities relating to the seven mobile terminating transaction identifiers shall be in CC state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.4.3.3

Conformance requirement 2: TS 24.008 clause 4.5.3, clause 5.5.3.2 and clause 8.3.1.

10.1.3.3.5.3 Test purpose

- 1) To verify that a CC-entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed", upon receipt of a RELEASE will return a RELEASE COMPLETE and enter the CC-state U0, "Null".
- 2) To verify that the UE on returning to the idle mode releases the MM-connection and that the CC-entities relating to the seven mobile terminating transaction identifiers are in CC-state U0, "Null".

10.1.3.3.5.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U9 by using table 10.1.3/4.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U9. The SS sends a RELEASE message to the UE. The UE responds by sending a RELEASE COMPLETE message and enters state U0, null. The SS verifies by using the status enquiry procedure that the UE has entered the correct state with the relevant transaction identifiers.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1 2	-	<- >	RELEASE RELEASE COMPLETE	with cause "Normal, unspecified"
3 4 5	<- -> SS		STATUS ENQUIRY RELEASE COMPLETE	cause #81 (invalid TI value) repeat steps 3-4 to cover all the
6		<-	RRC CONNECTION RELEASE Void	transaction identifiers from 000110 the main signalling link shall be released. The SS releases the RRC connection.
7	→		RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.3.3.5.5 Test requirements

After step 1 a CC entity of the UE in CC state U9, "UE Terminating Call Confirmed", shall return a RELEASE COMPLETE message.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.3.3.6 Incoming call / U9 mobile terminating call confirmed / lower layer failure

10.1.3.3.6.1 Definition

The call control entity of the UE being in the state, U9, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.3.3.6.2 Conformance requirement

1) The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. A CC entity of the UE in CC state U9, "Mobile Terminating Call Confirmed", having detected a lower layer failure shall return to idle mode, the CC entity is in state U0, "Null".

References

TS 24.008 clause 4.5.2.3, and clause 4.5.3, clause 5.5.3.2 and clause 8.3, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.3.3.6.3 Test purpose

To verify that a CC entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed", having detected a lower layer failure returns to idle mode, the CC entity is in state U0, "Null".

10.1.3.3.6.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U9 by using table 10.1.3/4.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The UE is brought to the state U9. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS remodifies the scrambling code of downlink transmission (DL DPCH) to generate a lower have the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	SS			SS modifies the scrambling code of DPCH for generating lower layer failure
2	->		CELL UPDATE	CCCH
3	<-		RRC CONNECTION RELEASE	CCCH
4	SS			SS re-modifies the scrambling code of DPCH to the original one.
5	SS			SS waits 60 s. UE shall send no message on the DCCH

Specific message contents:

None.

10.1.3.3.6.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.3.3.7 Incoming call / U9 mobile terminating call confirmed / unknown message received

10.1.3.3.7.1 Definition

The call control entity of the UE being in the state, U9, an unknown message is received by the UE.

10.1.3.3.7.2 Conformance requirement

1) If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC entity of the UE in CC state U9, "Mobile Terminating Call Confirmed" having received an unknown message from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.4.

10.1.3.3.7.3 Test purpose

To verify that a CC-entity of the UE in CC-state U9, "Mobile Terminating Call Confirmed" having received an unknown message from its peer entity returns a STATUS message.

10.1.3.3.7.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U9 by using table 10.1.3/4.

Test procedure

A MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U9. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<-		unknown message	message type not defined for PD
2	->			cause #97, state U9
3	<-		STATUS ENQUIRY	
4	->		STATUS	cause #30, state U9

Specific message contents:

None.

10.1.3.3.7.5 Test requirements

After step 1 a CC entity of the UE in CC state U9, "UE Terminating Call Confirmed", shall return a STATUS message.

- 10.1.3.4 Incoming call / U7 call received
- 10.1.3.4.1 Incoming call / U7 call received / call accepted
- 10.1.3.4.1.1 Definition

The call control entity of the UE being in the state, U7, a user accepts the incoming call.

10.1.3.4.1.2 Conformance requirement

1) In the "mobile terminating call confirmed" state or the "call received" state, the call control entity in the UE indicates acceptance of a mobile terminating call by:

- sending a CONNECT message to its peer entity in the network;
- starting Timer T313; and
- <u>entering the "connect request" state.</u> A CC entity of a UE in CC state U7, "Call Received", upon a user accepting the incoming call, shall send a CONNECT message to its peer entity and enter the CC state U8, "Connect Request".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.2.2.5.

10.1.3.4.1.3 Test purpose

To verify that a CC entity of a UE in CC-state U7, "Call Received", upon a user accepting the incoming call, shall send a CONNECT message to its peer entity and enter the CC-state U8, "Connect Request".

10.1.3.4.1.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U7. The user accepts the incoming call. The UE sends a CONNECT message. The SS checks by using the status enquiry procedure that the CC entity has entered state U8, connect request.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1				the UE is made to accept the call by the user
2	->		CONNECT	
3	<-		STATUS ENQUIRY	
4	-	>	STATUS	cause #30, state U8

Specific message contents:

None.

10.1.3.4.1.5 Test requirements

After step 1 a CC entity of a UE in CC state U7, "Call Received", shall send a CONNECT message and enter the CCstate U8, "Connect Request".

10.1.3.4.2 Incoming call / U7 call received / termination requested by the user

10.1.3.4.2.1 Definition

The call control entity of the UE being in the state, U7, a user requests to terminate incoming call.

10.1.3.4.2.2 Conformance requirement

1) Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and entering the "disconnect request" state. A CC entity of a UE in CC state U7, "Call Received", upon request by the user to terminate shall send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.4.3.1

10.1.3.4.2.3 Test purpose

To verify that a CC entity of a UE in CC-state U7, "Call Received", upon request by the user to terminate will send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.3.4.2.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U7. The user initiates clearing the incoming call. The UE sends a DISCONNECT message. The SS checks by using the status enquiry procedure that the CC entity has entered state U11, disconnect request.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1				the UE is made to terminate/reject the call
2	->		DISCONNECT	
3	<-		STATUS ENQUIRY	
4	-	>	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.3.4.2.5 Test requirements

After step 1 a CC entity of a UE in CC state U7, "Call Received", shall send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.3.4.3 Incoming call / U7 call received / DISCONNECT received

10.1.3.4.3.1 Definition

The call control entity of the UE being in the state, U7, a DISCONNECT message is received by the UE.

10.1.3.4.3.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon receipt of a DISCONNECT message with progress indicator #8:

i) if an appropriate speech traffic channel is not connected, continue clearing as defined in TS 24.008 clause 5.4.4.1.2.1 without connecting to the in-band tone/announcement;

ii) if an appropriate speech traffic channel is connected, attach the user connection for speech if it is not yet attached and enter the "disconnect indication" state. In that state, if upper layers request the clearing of the call, the call control entity of the UE shall proceed as defined in TS 24.008 clause 5.4.4.1.2.1. A CC entity of a UE in CC state U7, "Call Received", upon receipt of a DISCONNECT with a progress indicator indicating in band information from network, if a DTCH was not assigned, shall return a RELEASE message and enter the CC state U19, "Release Request".

References

TS 24.008 clause 5.4.4.1.1.1

10.1.3.4.3.3 Test purpose

To verify that a CC entity of a UE in CC-state U7, "Call Received", upon receipt of a DISCONNECT with a progress indicator indicating in-band information from network, if a DTCH was not assigned, returns a RELEASE message and enters the CC-state U19, "Release Request".

10.1.3.4.3.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U7. The SS sends a DISCONNECT message. The UE responds with a RELEASE message. The SS checks by using the status enquiry procedure that the CC entity has entered state U19, release request.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1	<-		DISCONNECT	(note)
2	->		RELEASE	
3	<-		STATUS ENQUIRY	
4	-	>	STATUS	cause #30, state U19

Specific message contents:

NOTE: With a progress indicator indicating in band information; the Progress Indicator, Progress Description value:

#8 in band information or appropriate pattern now available.

10.1.3.4.3.5 Test requirements

After step 1 a CC entity of a UE in CC state U7, "Call Received", if a DTCH was not assigned, shall return a RELEASE message and enter the CC-state U19, "Release Request".

10.1.3.4.4 Incoming call / U7 call received / RELEASE received

10.1.3.4.4.1 Definition

The call control entity of the UE being in the state, U7, a RELEASE message is received by the UE.

10.1.3.4.4.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release

the MM connection; and return to the "null" state. A CC entity of a UE in CC state U7, "Call Received", upon receipt of a RELEASE shall return a RELEASE COMPLETE and enter the CC state U0, "Null".

2) On returning to the idle mode the UE shall release the MM connection and the CC entities relating to the seven mobile terminating transaction identifiers shall be in CC state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.43.3

Conformance requirement 2: TS 24.008 clause 4.5.3, clause 5.5.3.2 and 8.3.1.

10.1.3.4.4.3 Test purpose

- 1) To verify that a CC entity of a UE in CC-state U7, "Call Received", upon receipt of a RELEASE will return a RELEASE COMPLETE and enter the CC-state U0, "Null".
- 2) To verify that the UE on returning to the idle mode releases the MM-connection and that the CC-entities relating to the seven mobile terminating transaction identifiers are in CC-state U0, "Null".

10.1.3.4.4.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U7. The SS sends a RELEASE message. The UE responds with a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity has entered state U0, null, with the relevant transaction identifiers.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1 2 3	<- ->		RELEASE RELEASE COMPLETE STATUS ENQUIRY	with cause "Normal, unspecified"
3 4 5	-	<- > SS	RELEASE COMPLETE	cause #81 (invalid TI value) repeat steps 3-4 to cover all the transaction identifiers from 000110
6		<-	RRC CONNECTION RELEASE Void	the main signalling link shall be released. The SS releases the RRC connection.
7	-	>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.3.4.4.5 Test requirements

After step 1 a CC entity of a UE in CC state U7, "Call Received", shall return a RELEASE COMPLETE message.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.3.4.5 Incoming call / U7 call received / lower layer failure

10.1.3.4.5.1 Definition

The call control entity of the UE being in the state, U7, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.3.4.5.2 Conformance requirement

1)—The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. A CC entity of a UE in CC state U7, "Call Received", having detected a lower layer failure shall return to idle mode, the CC entity is in state U0, "Null".

References

TS 24.008 clause 4.5.2.3, and clause 4.5.3, clause 5.5.3.2 and clause 8.3, TS 25.331 clause 8.3.1, and clause 8.5.6.

10.1.3.4.5.3 Test purpose

To verify that a CC entity of a UE in CC-state U7, "Call Received", having detected a lower layer failure returns to idle mode, the CC entity is in state U0, "Null".

10.1.3.4.5.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/2.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The UE is brought to the state U7. The SS modifies the scrambling code of downlink transmission(DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1 2 3 4 5	-> -> -> SS SS		CELL UPDATE RRC CONNECTION RELEASE	SS modifies the scrambling code of DPCH for generating lower layer failure CCCH SS re-modifies the scrambling code of DPCH to the original one. SS waits 60 s. UE shall send no message on the DCCH

Specific message contents:

None.

10.1.3.4.5.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.3.4.6 Incoming call / U7 call received / unknown message received

10.1.3.4.6.1 Definition

The call control entity of the UE being in the state, U7, an unknown message is received by the UE.

10.1.3.4.6.2 Conformance requirement

1) If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC entity of a UE in CC state U7, "Call Received", having received an unknown message from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.4.

Release 5

10.1.3.4.6.3 Test purpose

To verify that a CC entity of a UE in CC-state U7, "Call Received", having received an unknown message from its peer entity returns a STATUS message.

10.1.3.4.6.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U7. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
1	<-		unknown message	message type not defined for PD
2		>	STATUS	cause #97, state U7
3	<	<-	STATUS ENQUIRY	
4	-	>	STATUS	cause #30, state U7

Specific message contents:

None.

10.1.3.4.6.5 Test requirements

After step 1 a CC entity of a UE in CC state U7, "Call Received", shall return a STATUS message.

10.1.3.4.7 Incoming call / U7 call received / DTCH assignment

10.1.3.4.7.1 Definition

The call control entity of the UE being in the state, U7, a radio bearer establishment procedure is performed for traffic channel.

10.1.3.4.7.2 Conformance requirement

1) It is a network dependent decision when to initiate the assignment of a traffic channel during the mobile terminating call establishment phase.

Initiation of the assignment phase does not directly change the state of a CC entity nor affect any call control timer, but may have some secondary effects (see e.g. TS 24.008 clause 5.2.2.3.2). A CC entity of a UE in CC state U7, "Call Received", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, shall stay in CC state U7.

References

TS 25.331 clause 8.2.1, TS 24.008 clause 5.2.2.7.

10.1.3.4.7.3 Test purpose

To verify that a CC entity of a UE in CC-state U7, "Call Received", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, stays in CC-state U7.

10.1.3.4.7.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected basic service is telephony. If necessary, the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U7. The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The SS verifies by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

ſ	Step	Dire	ction	Message	Comments
		UE	SS		
1	1			Radio Bearer Setup Procedure	DTCH, See TS_34.108_clause 7.1.3
	2	<	<-	STATUS ENQUIRY	
	3	->		STATUS	cause #30, state U7

Specific message contents:

None.

10.1.3.4.7.5 Test requirements

After step 1 the CC state U7, "Call Received", shall remain unchanged.

10.1.3.4.8 Incoming call / U7 call received / RELEASE COMPLETE received

10.1.3.4.8.1 Definition

The call control entity of the UE being in the state, U7, the call is cleared by a RELEASE COMPLETE message sent by the SS.

10.1.3.4.8.2 Conformance requirement

- A call control entity shall accept an incoming RELEASE COMPLETE message used to initiate the call clearing even though the cause information element is not included. A CC entity of the UE in CC state U7, "call received", upon receipt of a RELEASE COMPLETE message with valid cause value, shall enter CC state U0, "Null".
- 2) A call control entity of the UE in any call control state shall, upon receipt of a RELEASE COMPLETE message from its peer entity in the network: stop all running call control timers ; release the MM connection; and return to the "null" state.On returning to idle mode, the CC entities relating to the seven mobile terminating transaction identifiers shall be in state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.2 and clause 5.4.4.

Conformance requirement 2: TS 24.008 clause 5.4.4.1.3 and 8.3.1.

10.1.3.4.8.3 Test purpose

- 1) To verify that a CC entity of the UE in CC-state U7, "Call received", upon receipt of a RELEASE COMPLETE message with valid cause value, enters CC state U0, "Null".
- 2) To verify that in returning to idle mode, the CC entities relating to the seven mobile terminating transaction identifiers are in state U0, "Null".

10.1.3.4.8.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U7 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE and for which the UE does not use immediate connection; if the UE supports MT telephony without immediate connection, the selected service is telephony. If necessary, the UE is configured for that basic service. The mobile terminated call is initiated. The CC

Release 5

1645

entity of the UE is brought to U7. The SS sends a RELEASE COMPLETE message to the UE. The SS checks by using the status enquiry procedure that the CC entity has entered the state U0 with all the relevant transaction identifiers.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<- <-		RELEASE COMPLETE STATUS ENQUIRY	note 1
3	->		RELEASE COMPLETE	cause #81 (invalid TI value), note 2
4	SS			repeat steps 2-3 to cover all the transaction identifiers from 000110
5	<-		RRC CONNECTION RELEASEVoid	the main signalling link shall be released. <u>The SS releases the RRC</u> connection.
6	-	>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

NOTE 1: With the cause value chosen arbitrarily.

NOTE 2: TI flag has the value indicating the SS as an originator of the call.

10.1.3.4.8.5 Test requirements

After step 2 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.3.5 Incoming call / U8 connect request

10.1.3.5.1 Incoming call / U8 connect request / CONNECT acknowledged

10.1.3.5.1.1 Definition

The call control entity of the UE being in the state, U8, a CONNECT ACKNOWLEDGE message is received by the UE.

10.1.3.5.1.2 Conformance requirement

In the "connect request" state, the call control entity of the UE shall, upon receipt of a CONNECT ACKNOWLEDGE message: stop timer T313 and enter the "active" state. A CC entity of a UE in CC state U8, "Connect Request", upon receipt of CONNECT ACKNOWLEDGE shall enter the CC state U10, "Call Active".

References

TS 24.008 clause 5.2.2.6.

10.1.3.5.1.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", upon receipt of CONNECT ACKNOWLEDGE shall enter the CC-state U10, "Call-Active".

10.1.3.5.1.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/2.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8 (if the UE uses immediate connection for the selected basic service then p = Y, otherwise p = N). The SS sends a CONNECT ACKNOWLEDGE message. The SS checks by using the status enquiry procedure that the CC entity of the UE has entered state U10, active.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
A1			Radio Bearer Setup Procedure	p = Y, See TS34.108
2	×	<-	CONNECT ACKNOWLEDGE	
3	<	<-	STATUS ENQUIRY	
4	-	>	STATUS	cause #30, state U10

Specific message contents:

None.

10.1.3.5.1.5 Test requirements

After step 2 a CC entity of a UE in CC state U8, "Connect Request", shall enter the CC-state U10, "Call-Active".

10.1.3.5.2 Incoming call / U8 connect request / timer T313 time-out

10.1.3.5.2.1 Definition

The call control entity of the UE being in the state, U8, if no response is then received from the SS, timer T313 expires at the UE side.

10.1.3.5.2.2 Conformance requirement

- 1) When timer T313 expires prior to the receipt of a CONNECT ACKNOWLEDGE message, the UE shall initiate clearing in accordance with clause 5.4.3. A CC entity of a UE in CC state U8, "Connect Request", having waited for a reasonable length of time (e.g. expiry of timer T313) without receiving the appropriate protocol message to complete the incoming call, shall initiate the clearing of that incoming call by sending the CC message DISCONNECT and enter the CC state U11, "Disconnect Request".
- If an UE disconnects too early then, in the case of very late assignment of a traffic channel, systematic waste of radio resources may occur.
- 2) Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and entering the "disconnect request" state.

References

Conformance requirement 1: TS 24.008 clause 5.2.2.6 and

Conformance requirement 2: TS 24.008 clause 5.4.3.1

10.1.3.5.2.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", having waited for a reasonable length of time (e.g. expiry of timer T313) without receiving the appropriate protocol message to complete the incoming call, shall initiate the clearing of that incoming call by sending the CC message DISCONNECT and enter the CC-state U11, "Disconnect Request".

10.1.3.5.2.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/2.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8 (if the UE uses immediate connection for the selected basic service then p = Y, otherwise p = N). The T313 expires at the UE and the UE sends a DISCONNECT message and enters state U11, disconnect request. The SS checks by using the status enquiry procedure that the UE has entered the correct state.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
A1	•		Radio Bearer Setup Procedure	p = Y, See TS34.108
2	->		DISCONNECT	Shall not be sent before 15 s after entry into state U8. But, shall be sent before 1,1 * T313 after entry into state U8.
3	<-		STATUS ENQUIRY	-
4	-	>	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.3.5.2.5 Test requirements

<u>After step A1</u> Upon expiry of timer T313 without receiving the appropriate protocol message to complete the incoming call a CC entity of a UE in CC state U8, "Connect Request", shall initiate the clearing of that incoming call by sending a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.3.5.3 Incoming call / U8 connect request / termination requested by the user

10.1.3.5.3.1 Definition

The call control entity of the UE being in the state, U8, the user requests for releasing of the call.

10.1.3.5.3.2 Conformance requirement

1) Apart from the exceptions identified in TS 24.008 clause 5.4.2, the call control entity of the UE shall initiate clearing by: stopping all running call control timers, sending a DISCONNECT message; starting timer T305; and entering the "disconnect request" state. A CC entity of a UE in CC state U8, "Connect Request", upon request by the user to terminate shall send a DISCONNECT message and enter the CC state U11, "Disconnect Request".

References

TS 24.007 clause 6.2.2, TS 24.008 clause 5.4.3.1

10.1.3.5.3.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", upon request by the user to terminate will send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.3.5.3.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services;
- MT circuit switched basic services for which immediate connect is not used.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/2.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8 (if the UE uses immediate connection for the selected basic service then p = Y, otherwise p = N). Then the user requests termination of the call. The UE sends a DISCONNECT message and enters state U11, disconnect request. The SS verifies by using the status enquiry procedure that the UE has entered the correct state.

Expected sequence

Step	Dire	ction	Message	Comments
	UE	SS		
A1			Radio Bearer Setup Procedure	p = Y, See TS34.108
2				the user requests to clear the call
3	->		DISCONNECT	
4	<-		STATUS ENQUIRY	
5	-	>	STATUS	cause #30, state U11

Specific message contents:

None.

10.1.3.5.3.5 Test requirements

After step 2 a CC entity of a UE in CC state U8, "Connect Request", shall send a DISCONNECT message and enter the CC-state U11, "Disconnect Request".

10.1.3.5.4 Incoming call / U8 connect request / DISCONNECT received with in-band information

10.1.3.5.4.1 Definition

The call control entity of the UE being in the state, U8, a DISCONNECT message indicating availability of in band information is received by the UE.

10.1.3.5.4.2 Conformance requirement

The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon receipt of a DISCONNECT message with progress indicator #8:

i) if an appropriate speech traffic channel is not connected, continue clearing as defined in TS 24.008 clause 5.4.4.1.2.1 without connecting to the in-band tone/announcement;

ii) if an appropriate speech traffic channel is connected, attach the user connection for speech if it is not yet attached and enter the "disconnect indication" state. In that state, if upper layers request the clearing of the call, the call control entity of the UE shall proceed as defined in TS 24.008 clause 5.4.4.1.2.1. A CC entity of a UE in CC state U8, "Connect Request", upon receipt of a DISCONNECT with progress indicator #8 shall enter CC state U12, if the traffic channel is in speech mode. If the DTCH is not in speech mode, the UE shall send a RELEASE message and enter CC state U19.

References

TS 24.008 clause 5.4.4.1.1.1 and clause 5.5.1.

10.1.3.5.4.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", upon receipt of a DISCONNECT with progress indicator #8 enters CC-state U12, if the traffic channel is in speech mode, and that the UE sends a RELEASE message and enters CC-state U19 if the DTCH is not in speech mode.

10.1.3.5.4.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/4.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8. The SS sends a DISCONNECT message containing indication of in-band information availability to the UE. If channel mode is speech, the UE enters state U12, disconnect indication. If channel mode is not speech, the UE sends a RELEASE message and enters state U19, release request.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<-		DISCONNECT	(note)
				DTCH in speech mode:
A2	<-		STATUS ENQUIRY	
A3	-	>	STATUS	cause #30, state U12
				DTCH is not in speech mode:
B2	->		RELEASE	
B3	<-		STATUS ENQUIRY	
B4	-	>	STATUS	cause #30, state U19

Specific message contents:

NOTE: With a progress indicator indicating in band information; the Progress Indicator, Progress description value:

#8 in band information or appropriate pattern now available.

10.1.3.5.4.5 Test requirements

After step 1 a CC entity of a UE in CC state U8, "Connect Request", shall enter CC-state U12, if the traffic channel is in speech mode. If the DTCH is not in speech mode, the UE shall send a RELEASE message and enter CC-state U19.

10.1.3.5.5 Incoming call / U8 connect request / DISCONNECT received without in-band information

10.1.3.5.5.1 Definition

The call control entity of the UE being in the state, U8, a DISCONNECT message is received by the UE. The DISCONNECT message does not contain indication of in-band information availability.

10.1.3.5.5.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state, the "disconnect indication" state, and the "release request" state, shall, upon the receipt of a DISCONNECT message without progress indicator information element or with progress indicator different from #8:

- stop all running call control timers;

- send a RELEASE message;

- start timer T308; and
- enter the "release request" state. A CC entity of a UE in CC state U8, "Connect Request", upon receipt of a DISCONNECT without progress indicator, shall return a RELEASE message and enter the CC state U19, "Release Request".

References

TS 24.008 clause 5.4.4 and clause 5.4.4.1.2.1

10.1.3.5.5.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", upon receipt of a DISCONNECT without progress indicator, returns a RELEASE message and enters the CC-state U19, "Release Request".

10.1.3.5.5.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/4.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8. The SS sends a DISCONNECT message not containing indication of in-band information availability to the UE. The UE shall respond with a RELEASE message. The SS checks by using the status enquiry procedure that the CC entity of the UE has entered the state U19, release request.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<	<-	DISCONNECT	without progress indicator(note)
2	-	>	RELEASE	
3	<	<-	STATUS ENQUIRY	
4	-	>	STATUS	cause #30, state U19

Specific message contents:

NOTE: Without a progress indicator indicating in band information.None.

10.1.3.5.5.5 Test requirements

After step 1 a CC entity of a UE in CC state U8, "Connect Request", shall return a RELEASE message and enter the CC-state U19, "Release Request".

1652

10.1.3.5.6 Incoming call / U8 connect request / RELEASE received

10.1.3.5.6.1 Definition

The call control entity of the UE being in the state, U8, a RELEASE message is received by the UE.

10.1.3.5.6.2 Conformance requirement

1) The call control entity of the UE in any state except the "null" state and the "release request" state, shall, upon receipt of a RELEASE message: stop all running call control timers; send a RELEASE COMPLETE message; release the MM connection; and return to the "null" state. A CC entity of a UE in CC state U8, "Connect Request", upon receipt of a RELEASE shall return a RELEASE COMPLETE and enter the CC state U0, "Null".

2) On returning to the idle mode the UE shall release the MM connection and the CC entities relating to the seven mobile terminating transaction identifiers shall be in CC state U0, "Null".

References

Conformance requirement 1: TS 24.008 clause 5.4.43.3

Conformance requirement 2: TS 24.008 clause 4.5.3 and clause 5.5.3.2 and 8.3.1.

10.1.3.5.6.3 Test purpose

- 1) To verify that a CC entity of a UE in CC-state U8, "Connect Request", upon receipt of a RELEASE will return a RELEASE COMPLETE and enter the CC-state U0, "Null".
- 2) To verify that the UE on returning to the idle mode releases the MM-connection and that the CC-entities relating to the seven mobile terminating transaction identifiers are in CC-state U0, "Null".

10.1.3.5.6.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8. The SS sends a RELEASE message. The UE responds with a RELEASE COMPLETE message. The SS checks by using the status enquiry procedure that the CC entity has entered state U0, null, with the relevant transaction identifiers.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1 2 3	-	<- >	RELEASE RELEASE COMPLETE STATUS ENQUIRY	with cause "Normal, unspecified"
3 4 5	<- -> SS		RELEASE COMPLETE	cause #81 (invalid TI value) repeat steps 3-4 to cover all the transaction identifiers from 000110
6		<-	RRC CONNECTION RELEASE Void	the main signalling link shall be released. The SS releases the RRC connection.
7	-	>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

None.

10.1.3.5.6.5 Test requirements

After step 1 a CC entity of a UE in CC state U8, "Connect Request", shall return a RELEASE COMPLETE message.

After step 3 CC entities relating to all mobile originating transaction identifiers shall send RELEASE COMPLETE messages with cause value #81 (invalid TI value).

10.1.3.5.7 Incoming call / U8 connect request / lower layer failure

10.1.3.5.7.1 Definition

The call control entity of the UE being in the state, U8, a lower layer failure is accomplished at the UE and consequently, communication at layer 3 level with the peer entity is terminated.

10.1.3.5.7.2 Conformance requirement

1)—The MM sublayer shall indicate to all CM entities associated with active MM connections that the MM connection is interrupted, the subsequent action of the MM sublayer (call re-establishment, see TS 24.008 clause 4.5.1.6, or local release) will then depend on the decisions by the CM entities. A CC entity of a UE in CC state U8, "Connect Request", having detected a lower layer failure shall return to idle mode, the CC entity is in state U0, "Null".

References

TS 24.008 clause 4.5.2.3, and clause 4.5.3 and clause 5.5.3.2, TS 25.331 clause 8.3.1 and clause 8.5.6.

10.1.3.5.7.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", having detected a lower layer failure returns to idle mode, the CC entity is in state U0, "Null".

10.1.3.5.7.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The UE is brought to the state U8. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE message as a response to the CELL UPDATE message from the UE. The SS re-modifies the scrambling code of downlink transmission (DL DPCH) to the original one and waits 60 s. The SS will check that the UE will not send any message during 60 s.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	S	S		SS modifies the scrambling code of DPCH for generating lower layer failure
2	->		CELL UPDATE	СССН
3	<	<-	RRC CONNECTION RELEASE	CCCH
4	S	S		SS re-modifies the scrambling code of DPCH to the original one.
5	SS			SS waits 60 s. UE shall send no message on the DCCH

Specific message contents:

None.

10.1.3.5.7.5 Test requirements

After step 4 the UE shall not send any message to the SS during 60 s.

10.1.3.5.8 Incoming call / U8 connect request / DTCH assignment

10.1.3.5.8.1 Definition

The call control entity of the UE being in the state, U8, a radio bearer establishment procedure is performed for traffic channel.

10.1.3.5.8.2 Conformance requirement

1) It is a network dependent decision when to initiate the assignment of a traffic channel during the mobile terminating call establishment phase.

Initiation of the assignment phase does not directly change the state of a CC entity nor affect any call control timer, but may have some secondary effects (see e.g. TS 24.008 clause 5.2.2.3.2 A CC entity of a UE in CC state U8, "Connect Request", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, shall stay in the CC state U8.

References

TS 25.331 clause 8.2.1, TS 24.008 clause 5.2.2.7.

10.1.3.5.8.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", when a traffic channel is allocated by the network performing the radio bearer establishment procedure, stays in the CC-state U8.

10.1.3.5.8.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8. The SS sends a RADIO BEARER SETUP for traffic channel to the UE. The UE shall respond with a RADIO BEARER SETUP COMPLETE message. The SS verifies by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Ī	Step	Direction		Message	Comments
		UE	SS		
	1			Radio Bearer Setup Procedure	DTCH, See TS_34.108 clause 7.1.3
	2	<	<-	STATUS ENQUIRY	
	3	->		STATUS	cause #30, state U8

Specific message contents:

None.

10.1.3.5.8.5 Test requirements

After step 1 the CC-state U8, "Connect Request", shall remain unchanged.

10.1.3.5.9 Incoming call / U8 connect request / unknown message received

10.1.3.5.9.1 Definition

The call control entity of the UE being in the state, U8, an unknown message is received by the UE.

10.1.3.5.9.2 Conformance requirement

1) If a UE receives an RR, MM or CC message with message type not defined for the PD or not implemented by the receiver in acknowledged mode, it shall return a status message (STATUS, MM STATUS depending on the protocol discriminator) with cause # 97 "message type non-existent or not implemented". A CC entity of a UE in CC state U8, "Connect Request", having received an unknown message from its peer entity shall return a STATUS message.

References

TS 24.008 clause 8.4.

10.1.3.5.9.3 Test purpose

To verify that a CC entity of a UE in CC-state U8, "Connect Request", having received an unknown message from its peer entity returns a STATUS message.

10.1.3.5.9.4 Method of test

Related ICS/IXIT statements

- supported MT circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U8 by using table 10.1.3/1.

Test procedure

An MT circuit switched basic service is selected that is supported by the UE; if the UE supports MT telephony, the selected basic service is telephony. If necessary the UE is configured for that basic service. Then a mobile terminated call is initiated. The CC entity of the UE is brought to the state U8. The SS sends a message with message type not defined for the protocol discriminator to the UE. The UE shall respond with a STATUS message, and finally the SS checks by using the status enquiry procedure that the state of the CC entity has remained unchanged.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<-		unknown message	message type not defined for PD
2	->		STATUS	cause #97, state U8
3	<-		STATUS ENQUIRY	
4	-	>	STATUS	cause #30, state U8

Specific message contents:

None.

10.1.3.5.9.5 Test requirements

After step 1 a CC entity of a UE in CC state U8, "Connect Request", shall return a STATUS message.

10.1.4 In call functions

- 10.1.4.1 In-call functions / DTMF information transfer
- 10.1.4.1.1 In-call functions / DTMF information transfer / basic procedures
- 10.1.4.1.1.1 Definition

Dual Tone Multi Frequency (DTMF) is an inband one out of four plus one out of four signalling system primarily used from terminal instruments in telecommunication networks.

- 10.1.4.1.1.2 Conformance requirement
- A user may cause a DTMF tone to be generated e.g. by depression of a key in the UE. The relevant action is interpreted by the UE as a requirement for a DTMF digit to be sent in a START DTMF message on an established FACCH. This message contains the value of the digit to be transmitted (0, 1, ..., 9, A, B, C, D, *, #).

Only a single digit will be transferred in each START DTMF message. An UE supporting the Mobile originating DTMF protocol control procedure, having a CC entity for speech in state U10, "Active": when made to send a DTMF tone, shall send a START DTMF message on the correct DCCH.

- 2) Upon receiving the START DTMF message the network will reconvert the received digit back into a DTMF tone which is applied toward the remote user and returns a START DTMF ACKNOWLEDGE message to the UE. This acknowledgement may be used in the UE to generate an indication as a feedback for a successful transmission. An UE supporting the Mobile originating DTMF protocol control procedure, having a CC entity for speech in state U10, "Active": when made to send a DTMF tone (the corresponding IA5 character being selected from among the ones supported), shall send a START DTMF message specifying the correct IA5 character in the "keypad information" field of the keypad facility information element.
- 3) When the user indicates that the DTMF sending should cease e.g. by releasing the key the UE will send a STOP DTMF message to the network.

References

Conformance requirement 1: TS 24.008 clause 5.5.7.1

Conformance requirement 2 and 4: TS 24.008 clause 5.5.7.2

Conformance requirement 3: TS 24.008 clause 5.5.7.3

10.1.4.1.1.3 Test purpose

- To verify that an UE supporting the Mobile originating DTMF protocol control procedure, having a CC entity for speech in state U10, "Active": when made to send a DTMF tone, sends a START DTMF message-on the correct DCCH.
- 2) To verify that an UE supporting the Mobile originating DTMF protocol control procedure, having a CC entity for speech in state U10, "Active": when made to send a DTMF tone (the corresponding IA5 character being selected from among the ones supported), sends a START DTMF message specifying the correct IA5 character in the "keypad information" field of the keypad facility information element- and to verify that acknowledgement send by the SS is used in the UE to generate a feedback indication for a successful transmission, if applicable.
- 3) To verify that the UE will send a STOP DTMF message to the network.
- 4) To verify that the state U10 of the UE CC entity has remained unchanged throughout the test procedure.

10.1.4.1.1.4 Method of test

Related ICS/IXIT statements

- supported teleservices;
- supported character set (e.g. 0-9, #, *, A, B, C, D);
- if and how DTMF tone is indicated to the user.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 "Active" by using table 10.1.2/1.for speech by using Generic call setup procedure for mobile originating circuit switched call defined in TS 34.108.

Test procedure

The UE being in the call-active state, a user causes a DTMF tone to be generated e.g. by depression of a key in the UE. A DTMF digit corresponding to the digit indicated by the user is sent in a START DTMF message by the UE. The SS will return a START DTMF ACKNOWLEDGE message to the UE. This acknowledgement may be used in the UE to generate an indication as a feedback for a successful transmission. Then the user indicates that the DTMF sending should cease e.g. by releasing the key. The UE will send a STOP DTMF message to the network which is acknowledged with STOP DTMF ACKNOWLEDGE by the SS.

The sequence described above is repeated for each of the applicable characters 0-9, #, *, A, B, C, and D.

Then a case of rejecting a DTMF tone is tested. and

<u>**t**</u>The state of the UE is verified throughout the test procedure.

Expected sequence

Step	Directi	on	Message	Comments
	UE S	SS		
1	SS			Request the user to cause a DTMF tone
				to be generated
	->		START DTMF	the SS will verify that the transmitted
				information corresponds to the digit
				pressed
2	<-		START DTMF ACKNOWLEDGE	possible indication of a DTMF tone
				depending the ICS/IXIT statements
3	<-		STATUS ENQUIRY	
4	->		STATUS	cause #30, state U10
5	->		STOP DTMF	
6	<-		STOP DTMF ACKNOWLEDGE	the DTMF tone indication shall be
				stopped
7				the steps 1-6 shall be repeated for each
				of the applicable characters 0-9, #, *, A,
				B, C, D.
8	<-		STATUS ENQUIRY	
9	->		STATUS	cause #30, state U10
10	SS			Request the user to cause a DTMF tone
				to be generated.
11	->		START DTMF	
12	<-		START DTMF REJECT	
13	<-		STATUS ENQUIRY	
14	->		STATUS	cause #30, state U10

Specific message contents:

None.

10.1.4.1.1.5 Test requirements

Upon a user making to send a DTMF tone the UE a CC entity for speech in the CC state U10, "Active", shall send a START DTMF message on the DCCH FACCH to SS.

The SS will verify that the transmitted information corresponds to the digit pressed in the UE.

After steps 2 and 7 (successful DTMF transmission) the CC-state U10, "Active", shall remain unchanged.

After step <u>11-12</u> (unsuccessful DTMF transmission) the CC-state U10, "Active", shall remain unchanged.

10.1.4.2 In-call functions / user notification

User notification procedure allows the network to notify a UE of any call related event during the "active" state of a call. It also may allow a UE to notify the remote user of any appropriate call related event during the "active" state of a call by sending a NOTIFY message containing a notification indicator to the network. No state change occurs at any of the interface sides during this procedure.

10.1.4.2.1 In-call functions / User notification / UE terminated

10.1.4.2.1.1 Definition

This is a case for testing user notification procedure terminated by the user equipment.

10.1.4.2.1.2 Conformance requirement

1) The mobile terminating user notification procedure allows the network to notify a mobile station of any appropriate call-related event during the "active" state of a call. The procedure consists in the network sending a NOTIFY message to the mobile station. No state change occurs at any of the interface sides following the sending or the receipt of this

message (but an appropriate indication may optionally be generated in the mobile station). A CC entity of a UE in CCstate U10, "active", upon receiving of a NOTIFY message shall remain in the active state.

References

TS 24.008 clause 5.3.1.

10.1.4.2.1.3 Test purpose

To verify that a CC entity of a UE in CC-state U10, "active", upon receiving of a NOTIFY message remains in the active state.

10.1.4.2.1.4 Method of test

Related ICS/IXIT statements

- supported circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 "Active" by using <u>table 10.1.2/1.Generic call setup procedure for mobile</u> originating circuit switched calls defined in TS 34.108.

Test procedure

The UE being in the call active state, the SS will send a NOTIFY message to the UE. The state of the UE is checked after that.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	<-	NOTIFY	
2	<-	STATUS ENQUIRY	
3	->	STATUS	cause #30, state U10

Specific message contents:

None.

10.1.4.2.1.5 Test requirements

After step 1 a CC entity of the UE in the CC-state U10, "active", shall remain unchangedin the active state.

10.1.4.3 In-call functions / channel changes

The two following test cases are for testing some elementary radio resource level procedures during an active state of a call to ensure call maintenance also during Hard handover.

10.1.4.3.1 In-call functions / channel changes / a successful channel change in active state/ Hard handover

10.1.4.3.1.1 Definition

This is a case to test a change of the frequency of a physical channel during active state of a call.

10.1.4.3.1.2 Conformance requirement

1) The UE being in the call active state after having successful completed a physical channel reconfiguration, shall remain in the call active state.

References

TS 24.008 clause 5.3.4.3.2, TS 25.331 clause 8.3.5.

10.1.4.3.1.3 Test purpose

To verify that the UE being in the call active state after having successful completed a physical channel reconfiguration remains in the call active state.

10.1.4.3.1.4 Method of test

Related ICS/IXIT statements

- supported circuit switched basic services;

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 "Active" by using <u>table 10.1.2/1. Generic call setup procedure for mobile</u> originating circuit switched calls defined in TS 34.108.

Test procedure

The UE being in the eall-active state, the SS initiated physical channel reconfiguration procedure causing an intracell change of channel by sending a PHYSICAL CHANNEL RECONFIGURATION message to the UE. The UE performs physical channel reconfiguration procedure and after the main signalling link is successfully established, the UE returns a PHYSICAL CHANNEL RECONFIGURATION COMPLETE message on the uplink DCCH using AM RLC. The state of the UE is then checked.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1	<-		PHYSICAL CHANNEL RECONFIGURATION	
2	->		PHYSICAL CHANNEL RECONFIGURATION	
			COMPLETE	
3	<	-	STATUS ENQUIRY	
4	-:	>	STATUS	cause #30, state U10

Specific message contents:

None.

10.1.4.3.1.5 Test requirements

After step 2 Tthe UE being in the call active state after having successful completed a physical channel reconfiguration, shall remain in the call active state.

10.1.4.3.2 In-call functions / channel changes / an unsuccessful channel change in active mode/Hard handover

10.1.4.3.2.1 Definition

This is a case to test an unsuccessful change of the frequency of a physical channel during active state of a call.

10.1.4.3.2.2 Conformance requirement

1) The UE, when returning to the old channel after physical channel reconfiguration failure, shall remain in the call active state.

References

TS 24.008 clause 5.3.4.3.

10.1.4.3.2.3 Test purpose

To verify that the UE, when returning to the old channel after physical channel reconfiguration failure, will remain in the call-active state.

10.1.4.3.2.4 Method of test

Related ICS/IXIT statements

- supported circuit switched basic services.

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in MM-state "idle, updated" with valid TMSI and CKSN.

The UE is brought into the state U10 "Active" by using <u>table 10.1.2/1.Generic call setup procedure for mobile</u> originating circuit switched calls defined in TS 34.108.

Test procedure

The SS sends a PHYSICAL CHANNEL RECONFIGURATION message, but does not activate the assigned physical channel. The UE shall attempt try to activate the new channel (this is not verified) and shall then reactivate the "old" channel. The UE shall send a PHYSICAL CHANNEL RECONFIGURATION FAILURE message on the DCCH using AM RLC and shall set the cause value in IE "failure cause" to "physical channel failure". The state of the UE is then checked.

Expected sequence

Step	Direction		Message	Comments	
	UE	SS			
1	<-		PHYSICAL CHANNEL RECONFIGURATION	The UE attempts and fails to re- configure the physical channel.	
2	-	>	PHYSICAL CHANNEL RECONFIGURATION FAILURE	NOTE	
3	<-		STATUS ENQUIRY		
4	-	>	STATUS	cause #30, state U10	
4	-	>	314103	cause #30, state 010	

Specific message contents:

NOTE: With the cause value "physical channel failure".

10.1.4.3.2.5 Test requirements

After step 2 Tthe UE being in the call active state after physical channel reconfiguration failure, shall remain in the call active state.

10.1.4.4 In-call functions / UE terminated in-call modification

10.1.4.4.1 In-call functions / UE terminated in-call modification / modify when new mode is not supported

This test is not applicable for R99.

10.1.4.5 In-call functions / UE originated in-call modification

10.1.4.5.1 In-call functions / UE originated in-call modification / a successful case of modifying

This test is not applicable for R99.

10.1.4.5.2 In-call functions / UE originated in-call modification / modify rejected

This test is not applicable for R99.

10.1.4.5.3 In-call functions / UE originated in-call modification / an abnormal case of acceptance

This test is not applicable for R99.

10.1.4.5.4 In-call functions / UE originated in-call modification / an abnormal case of rejection

This test is not applicable for R99.

10.1.4.5.5 In-call functions / UE originated in-call modification / time-out of timer T323 This test is not applicable for R99.

10.1.4.5.6 In-call functions / UE originated in-call modification / a successful channel change in state mobile originating modify

This test is not applicable for R99.

10.1.4.5.7 In-call functions / UE originated in-call modification / an unsuccessful channel change in state mobile originating modify

This test is not applicable for R99.

10.1.4.5.8 In-call functions / UE originated in-call modification / unknown message received

This test is not applicable for R99.

10.1.4.5.9 In-call functions / UE originated in-call modification / a release complete received

This test is not applicable for R99.

10.2 Call Re-establishment

10.2.1 Call Re-establishment/call present, re-establishment allowedVoid

10.2.1.1 Definition

This is to test a successful case of a call re establishment procedure.

10.2.1.2 Conformance requirement

 If the call is in the "active" state or "mobile originating modify" state, the indication from MM that reestablishment is possible shall cause call control to request re-establishment from the MM connection, suspend any further message to be sent and await the completion of the re-establishment procedure.

References

1) TS 24.008 clauses 4.5.1.6 and 5.5.4.2, TS 25.331 clause 8.3.1 and clause 8.5.6.

2) TS 24.008 clauses 4.5.1.6 and 5.5.4.3.

10.2.1.3 Test purpose

The purpose of this test is to verify that the UE can correctly perform a call re-establishment procedure.

10.2.1.4 Method of test

Related ICS/IXIT statements

-supported teleservices.

Initial conditions

System Simulator:

1cell, default parameters

User Equipment:

When the call control entity is notified that the MM-connection is re-established, it shall then resume the transmission of possibly suspended messages and resume user data exchange when an appropriate channel is available.

The UE is in MM state "idle, updated" with valid TMSI and CKSN.

Test procedure

The UE is brought into the state U10 "Active" by using Generic call setup procedure for mobile originating circuit switched calls defined in TS34.108. The SS modifies the scrambling code of downlink transmission (DL DPCH) to generate a lower layer failure at the UE. The SS waits long enough to enable the UE to perform cell update procedure. The SS sends RRC CONNECTION RELEASE as a response message to the CELL UPDATE message from the UE. The SS re modifies the scrambling code of downlink transmission (DL DPCH) to the original one. The UE shall re establish the call using CM RE ESTABLISHMENT message. The SS performs security mode control and radio bearer establishment procedures. The UE shall through-connect the appropriate bearer channel. Then, the call is cleared by the SS.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
<mark>1</mark>			The UE is brought into the state U10
			"Active" by using Generic call setup
			procedure for mobile originating circuit
			<mark>switched calls defined in TS34.108 (the</mark>
			appropriate bearer channel is through connected in both directions in DTCH)
_		_	connected in both directions in DTCH)
<mark>2</mark>	<mark>SS</mark>		SS modifies the scrambling code of
_			DPCH for generating lower layer failure.
<mark>3</mark>	<mark>-></mark>	CELL UPDATE	CCCH
<mark>4</mark>	<mark>≁</mark> ↓ <mark>\$\$</mark>	RRC CONNECTION RELEASE	CCCH
<mark>5</mark>	<mark>SS</mark>		SS re-modifies the scrambling code of
_			DPCH to the original one.
<mark>6</mark>	<mark>-></mark>	RRC CONNECTION REQUEST	
<mark>7</mark>	<u>↑</u> ↓ <u>↑</u> ↓ ↓ <mark>↑</mark>	RRC CONNECTION SETUP	
8 8	<mark>→</mark>	RRC CONNECTION SETUP COMPLETE	
<mark>8</mark>	<mark>-></mark>	CM REESTABLISHMENT REQUEST	<mark>note specific message contents</mark>
<mark>10</mark>	<mark>←</mark>	SECURITY MODE COMMAND	
<mark>11</mark>	<mark>-></mark>	SECURITY MODE COMPLETE	
10 11 1 2 1 3		Radio Bearer Setup Procedure	See TS34.108
<mark>13</mark>	UE		The appropriate bearer channel is
			through connected in both directions.
14 15 16 17	<mark>←</mark>	DISCONNECT	with cause value "Normal"
<mark>15</mark>	↓	RELEASE	
<mark>16</mark>	<mark>~-</mark>	RELEASE COMPLETE	
<mark>17</mark>	<mark>←</mark>	RRC CONNECTION RELEASE	
- <mark>18</mark>	<mark>→</mark>	RRC CONNECTION RELEASE COMPLETE	

Specific message contents:

CM RE ESTABLISHMENT REQUEST message contains Ciphering key sequence number IE with the value which the UE was allocated in .

10.2.1.5 Test requirements

After step 2 a CC entity of the UE in the "active" state, shall suspend any further message to be sent and await the completion of the re establishment procedure.

After step 12 the UE resume user data exchange when an appropriate channel is available.

10.3 User to user signalling

10.3.1 Definition

The "user to user" information element is used to convey information between the mobile user and a remote ISDN user.

Release 5

NOTE: There is no test for an UE originating call including a "user-user" information element since it is not a mandatory UE feature.

10.3.2 Conformance requirement

The inclusion of the "user-user" information element in downlink call control messages shall cause no adverse effects on the operation of the UE.

References

TS 24.008 clauses 5.2.2, 9.3.7, 9.3.23.1 and 10.5.4.25.

10.3.3 Purpose of the test

The purpose of this test is to verify that inclusion of the "user-user" information element in either of the down link messages, SETUP or DISCONNECT causes no adverse effects on the operation of the UE.

10.3.4 Method of test

Related ICS/IXIT statement(s)

- Supported MT circuit switched basic services.
- Support of user-user information element, and details of suitable codings.

Initial conditions.

System Simulator:

The SS simulates 1 cell, with default parameters.

User Equipment:

The UE is in MM-state "idle updated", with a valid TMSI and CKSN.

Test procedure

The SS attempts to set up a mobile terminated call, with one of the supported circuit switched basic services which has been arbitrarily chosen, the generic call set up procedures for mobile terminating circuit switched calls,(either speech or data) as specified in TS 34.108 clause 7. The default SETUP message contents are modified to include the user-user Information Element. The UE shall not respond adversely to the inclusion of the user-user information element.

After 30 s the SS sends a DISCONNECT message, again the UE shall not respond adversely to the inclusion of the user-user information element, but shall continue to clear down the call normally.

Expected sequence

Step	Direction		Message	Comments
	UE	SS		
1				Generic Call Setup procedure for mobile terminating circuit switched calls defined in TS 34.108, depending on choice of Bearer Capability. The SETUP message contains the user-user IE, see Specific message contents.
2				The SS waits 30 s.
3	<	-	DISCONNECT	Message contains the user-user IE, see Specific message contents
4	-;	>	RELEASE	
5	<	-	RELEASE COMPLETE	
6	<	-	RRC CONNECTION RELEASE Void	The SS releases the RRC connection.
7		>	RRC CONNECTION RELEASE COMPLETEVoid	

Specific message contents:

SETUP message contains user-user IE with the string coded in IA5 characters: for example "Call Setup".

DISCONNECT message contains user-user IE with the string coded in IA5 characters: for example "Call Disconnect". (The codings above are for example only. For the case of an UE which supports "user-user" signalling it may be necessary to add meaning to the data fields, see ICS/IXIT statement(s).)

NOTE: The codings above are for example only. For the case of an UE which supports "user-user" signalling it may be necessary to add meaning to the data fields, see ICS/IXIT statement(s).

10.3.5 Test requirements

After steps 1 and 3 Tthe inclusion of the "user-user" information element in downlink call control messages shall cause no adverse effects on the operation of the UE.

3GPP TSG-T1 Meeting #17 3GPP TSG-T1Sig Meeting #26 Luton, UK, 4 th – 8 th November							Тdoc ж T1-020792 Tdoc ж T1S020707					
										CR-Form-v7		
æ	34.	<mark>123-1</mark>	CR	343	ж rev	-	ж (Current vers	ion:	5.1.1	ж	
For <u>HELP</u> or	n using	g this for	m, see	e bottom of this	s page or	look a	at the	pop-up text	over	the	mbols.	
Proposed chang				ıpps ೫ <mark></mark>	ME	-		cess Networ				
Title:		ditorial o Package		ions in test cas	ses 11.1.1	l.1, 1 ⁻	1.3.2	(Package 1)	and	11.1.1.2.1	1	
Source:	ж <mark>N</mark>	EC Aus	tralia									
Work item code:	: ¥ T	EI						Date: ೫	4/1	1/2002		
Category:	De	F (con A (con B (add C (fun D (edi tailed exp	rection) respond lition of ctional torial m planatio	owing categories ds to a correctio feature), modification of t odification) ons of the above <u>TR 21.900</u> .	n in an eai eature)			Use <u>one</u> of 2 R96 R97 R98 R99 Rel-4 Rel-5	(GSM (Relea (Relea (Relea (Relea (Relea	-		

Reason for change: #	To allign the specification with 3G vocabulary.					
_						
Summary of change: #	• Test cases 11.1.1, 11.3.2 and 11.1.1.2.1: 'MS' replaced with 'UE'.					
	Test case 11.1.1.2.1: Reference to GSM specific behaviour deleted from Comformance requirement.					
Concernance if	Specification will not conform to 20 yearshulary					
Consequences if % not approved:	Specification will not conform to 3G vocabulary.					
Clauses affected: #	11.1.1, 11.1.2.1 and 11.3.2.					
Other specs भ affected:	Y N X Other core specifications # Test specifications # O&M Specifications #					

How to create CRs using this form:

Other comments: # Affects R99, Rel-4 and Rel-5

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.

Release 5

- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

<Start of modified section>

11.1.1.1 Attach initiated by context activation/QoS Offered by Network is the QoS Requested

11.1.1.1.1 Definition

11.1.1.1.2 Conformance requirement

SM procedures for identified access can only be performed if a GMM context has been established between the <u>MS-UE</u> and the network. If no GMM context has been established, the MM sublayer has to initiate the establishment of a GMM context by use of the GMM procedures as described in chapter 4, 3GPP TS 24.008. After GMM context establishment, SM uses services offered by GMM (see 3GPP TS 24.007). Ongoing SM procedures are suspended during GMM procedure execution.

In UMTS only, integrity protected signalling (see 3GPP TS 24.008 clause 4.1.1.1.1) and in general, see 3GPP TS 33.102) is mandatory. In UMTS only, all protocols shall use integrity protected signalling. Integrity protection of all SM signalling messages is the responsibility of lower layers. It is the network which activates integrity protection. This is done using the security mode control procedure (TS 25.331).

In order to request a PDP context activation, the UE sends an ACTIVATE PDP CONTEXT REQUEST message to the network, enters the state PDP-ACTIVE-PENDING and starts timer T3380. The message contains the selected NSAPI, PDP type, requested QoS and, if the UE requests a static address, the PDP address.

Upon receipt of an ACTIVATE PDP CONTEXT REQUEST message, the network selects a radio priority level based on the QoS negotiated and may reply with an ACTIVATE PDP CONTEXT ACCEPT message. Upon receipt of the message ACTIVATE PDP CONTEXT ACCEPT the UE shall stop timer T3380, shall enter the state PDP-ACTIVE.

If the QoS offered by the network is the same as the QoS requested by the UE, the UE shall accept the negotiated QoS.

In UMTS, both the network and the UE shall store the LLC SAPI and the radio priority in the PDP context. If a UMTS to GSM system change is performed, the new SGSN shall initiate establishment of the logical link using the negotiated QoS profile, the negotiated LLC SAPI, and selected radio priority level stored in the PDP context as in a GSM to GSM Routing Area Update.

A UE, which is capable of operating in both GSM and UMTS, shall use a valid LLC SAPI, while a UE which is capable of operating only in UMTS shall indicate the LLC SAPI value as "LLC SAPI not assigned" in order to avoid unnecessary value range checking and any other possible confusion in the network.

NOTE 1: The radio priority level and the LLC SAPI parameters, though not used in UMTS, shall be included in the messages, in order to support handover between UMTS and GSM networks.

Reference

3GPP TS 24.008 clauses 6.1.1 and 6.1.3.1.1.

11.1.1.1.3 Test purpose

To check that the UE initiates a PS attach, if one is not already active, when PDP context activation is requested.

To test the behaviour of the UE when SS responds to the PDP context activation request with the requested QoS.

11.1.1.1.4 Method of test

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in GMM-state "GMM-DEREGISTERED, normal service" with valid P-TMSI and CKSN.

Related ICS/IXIT statements

- PS Supported yes/no
- Auto Attach supported yes / no
- Method of context activation

Test procedure

If the UE is attached, then the Detach Request is originated from the UE indicating "GPRS detach without switching off". The SS responds with a Detach Accept after completing the security mode procedures. A PDP context activation is then requested by the user. The PS attach (ATTACH REQUEST) is then indirectly caused by a requested PDP context activation. The SS returns the ATTACH ACCEPT message to the UE. Now session management can proceed with PDP context activation.

On receipt of the ACTIVATE PDP CONTEXT REQUEST message an ACTIVATE PDP CONTEXT ACCEPT is returned by the SS with the same requested QoS. The contents of the ACTIVATE PDP CONTEXT REQUEST message shall then be checked. The SS then waits for T3380 seconds to ensure T3380 has been stopped and no more ACTIVATE PDP CONTEXT REQUEST messages are sent by the UE. The SS then sends a MODIFY PDP CONTEXT REQUEST message to which the UE shall reply with a MODIFY PDP CONTEXT ACCEPT message to ensure the context has been set up.

Expected sequence

Step	Direction	Message	Comments
Otop	UE SS	meeeage	Commonto
0	SS		SS checks that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Detach".
1	÷	DETACH REQUEST	Only sent if the UE attaches at power-up, if not go to step 3. Detach is performed by the UE using MMI or AT Commands
1a 2 2a 3	SS ← SS UE	DETACH ACCEPT	The SS starts integrity protection. SS sends Detach Accept message. The SS releases the RRC connection. Initiate a context activation
3a	SS		SS checks that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Registration".
4	\rightarrow	ATTACH REQUEST	Request attach with Follow-on request pending
4a	SS		The SS starts ciphering and integrity protection.
5	÷	ATTACH ACCEPT	Accept attach Negotiated Ready timer value IE should not be included Force to standby indicator set
6	\rightarrow	ACTIVATE PDP CONTEXT REQUEST	Request a PDP context activation (with static PDP address)
6a	SS		The SS establishes the RAB.
7	÷	ACTIVATE PDP CONTEXT	Accept the PDP context activation
8	SS		Wait for T3380 seconds to ensure no further activate request messages come from the UE
9	÷	MODIFY PDP CONTEXT REQUEST (NETWORK TO UE DIRECTION)	SS sends a modify request to UE for the activated context
10	\rightarrow	MODIFY PDP CONTEXT ACCEPT (UE TO NETWORK DIRECTION)	UE accepts the modification request from the SS to show context is activated
11	SS	,	SS releases the RRC connection due to inactivity (no user data transferred)

Specific message contents

None.

11.1.1.1.5 Test requirements

At step 0 the UE shall send an RRC CONNECTION REQUEST message with the IE Establishment cause set to "Detach".

At step 3a the UE shall send an RRC CONNECTION REQUEST message with the IE Establishment cause set to "Registration".

When requesting a PDP context activation, the UE shall:

- initiate a PS ATTACH if one is not already active;
- when the SS responds to a PDP context activation request, initiated by the UE, with the requested QoS, the UE shall complete the PDP context activation procedure. To check if the PDP context activation was successful, SS shall request PDP context modification and UE shall accept it.

11.1.1.2 QoS offered by the network is a lower QoS

- 11.1.1.2.1 QoS accepted by UE
- 11.1.1.2.1.1 Definition
- 11.1.1.2.1.2 Conformance requirement

In order to request a PDP context activation, the UE sends an ACTIVATE PDP CONTEXT REQUEST message to the network, enters the state PDP-ACTIVE-PENDING and starts timer T3380. If the QoS offered by the network is acceptable to UE, then upon receipt of the message ACTIVATE PDP CONTEXT ACCEPT, the UE shall stop timer T3380.

In GSM, the MS shall initiate establishment of the logical link for the LLC SAPI indicated by the network with the offered QoS and selected radio priority level, if no logical link has been already established for that SAPI.

In UMTS, both the network and the <u>MS-UE</u> shall store the LLC SAPI and the radio priority in the PDP context.

Reference

3GPP TS 24.008 clause 6.1.3.1.1.

11.1.1.2.1.3 Test purpose

To test the behaviour of the UE when the SS responds to a PDP context activation request with a lower QoS than that requested.

11.1.1.2.1.4 Method of test

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in GMM-state "GMM-REGISTERED, normal service" with valid P-TMSI and CKSN.

Related ICS/IXIT statements

- PS Supported yes/no
- User setting of Minimum QoS supported yes/no
- Method of setting minimum QoS
- Method of context activation

Test procedure

The requested QoS and Minimum QoS are set. A context activation is requested by the user. On receipt of the ACTIVATE PDP CONTEXT REQUEST message an ACTIVATE PDP CONTEXT ACCEPT is returned by the SS with QoS lower than the requested but higher than or equal to the minimum. The SS then sends a MODIFY PDP CONTEXT REQUEST message and the UE shall respond with a MODIFY PDP CONTEXT ACCEPT message to confirm the context is active.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	UE		Initiate a context activation
2	\rightarrow	ACTIVATE PDP CONTEXT REQUEST	Request a PDP context activation
3	÷	ACTIVATE PDP CONTEXT ACCEPT	Accept a PDP context activation
4	÷	MODIFY PDP CONTEXT REQUEST (NETWORK TO UE DIRECTION)	Send a modify request to UE for the activated context
5	÷	MODIFY PDP CONTEXT ACCEPT (UE TO NETWORK DIRECTION)	Accept the modification request from network to show context is activated

Specific message contents

None.

11.1.1.2.1.5 Test requirements

To pass the test UE shall:

- when the SS responds to a PDP context activation request, initiated by the UE, with the QoS lower than the requested but higher than or equal to the minimum, the UE shall complete the PDP context activation procedure.
- to see if the PDP context activation was successful, SS shall request PDP context modification and UE shall accept it.

<Next modified section>

11.3.2 PDP context deactivation initiated by the network

11.3.2.1 Definition

11.3.2.2 Conformance requirement

In order to deactivate a PDP context, the network sends a DEACTIVATE PDP CONTEXT REQUEST message to the <u>MS-UE</u> and starts timer T3395. The message contains the transaction identifier in use for the PDP context to be deactivated and a cause code that typically indicates one of the following causes:

- # 36: regular PDP context deactivation;
- # 38: network failure; or
- # 39: reactivation requested.

The UE shall, upon receipt of this message, reply with a DEACTIVATE PDP CONTEXT ACCEPT message. Upon receipt of the DEACTIVATE PDP CONTEXT ACCEPT message, the network shall stop the timer T3395. In UMTS, the network shall initiate the release of Radio Access Bearer associated with this PDP context.

Reference

3GPP TS 24.008 clauses 6.1.3.4, 6.1.3.4.2.

11.3.2.3 Test purpose

To test the behaviour of the UE upon receipt of a DEACTIVATE PDP CONTEXT REQUEST message from the SS.

11.3.2.4 Method of test

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in GMM-state "GMM-REGISTERED, normal service" with valid P-TMSI and CKSN.

Related ICS/IXIT statements

- PS Supported yes/no
- Method of activating a PDP context

Test procedure

A PDP context is activated by the user and accepted by the SS. A DEACTIVATE PDP CONTEXT REQUEST message is then sent by the SS. The UE shall reply with a DEACTIVATE PDP CONTEXT ACCEPT message. The SS shall then send a MODIFY PDP CONTEXT REQUEST for the deactivated context and the UE shall reply with an SM STATUS message with cause #81 'invalid transaction identifier value'.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1 1a	UE SS		Initiate a context activation SS checks that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to either Originating Conversational Call, Originating Streaming Call, Originating Interactive Call, Originating Background Call or Originating High Priority Signalling
1b	\rightarrow	SERVICE REQUEST	
1c	SS		The SS starts ciphering and integrity protection.
2	\rightarrow	ACTIVATE PDP CONTEXT REQUEST	Activate a PDP context
2a	SS		The SS establishes the RAB.
3	÷	ACTIVATE PDP CONTEXT ACCEPT	Accept the PDP context
4	÷	DEACTIVATE PDP CONTEXT REQUEST	Request a deactivation of a PDP context
5	\rightarrow	DEACTIVATE PDP CONTEXT	Accept the PDP context deactivation.
5a	SS		The SS releases the RAB.
6	÷	MODIFY PDP CONTEXT REQUEST (NETWORK TO UE DIRECTION)	Send a modify request to UE for the deactivated context.
7	\rightarrow	SM STATUS	Cause set to #81

Specific message contents

Steps 2 and 5. TI flag (bit 8) in TI IE is set to 0 (transaction initiated by the UE).

Steps 3, 4 and 6. TI flag in TI IE is set to 1.

Steps 2, 3, 4, 5 and 6. The value of TIO IE (bits 5-7) of the transaction identifier (TI) is the same in these test steps.

11.3.2.5 Test requirements

At step 1a the UE shall send an RRC CONNECTION REQUEST message with the IE Establishment cause set to one of the following causes:

- Originating Conversational Call;
- Originating Streaming Call;
- Originating Interactive Call;
- Originating Background Call or
- Originating High Priority Signalling.

Upon receipt of a request for deactivation of a PDP context from the SS, the UE shall deactivate PDP context. Then, upon modification procedure initiated by the network, for deactivated PDP context, UE shall reply with SM STATUS message with cause #81, as confirmation that previously SS requested PDP context deactivation was performed by the UE.

<End of modified section>

GPP TSG-T1 Meeting #17 GPP TSG-T1Sig Meeting #26 .uton, UK, 4 th – 8 th November						Тdoc ж T1-02079 Tdoc ж T1S02070					
			(CHANGE	REQ	UE	ST				CR-Form-v7
ж	3	<mark>4.123-1</mark>	CR	344 #	rev ،	-	ж Сі	urrent versi	ion: <mark>5</mark>	5.1.1	ж
For <mark>HELP</mark> of	on us	sing this for	m, see	e bottom of this p	bage or	look a	at the p	op-up text	over th	е Ж syı	nbols.
Proposed change affects: UICC apps # ME Radio Access Network Core Network											
Title:	ж	Extensior	of 'Te	est purpose' in te	st case	11.3.	1 (Pacl	kage 1 test	case).		
Source:	ж	NEC Aus	tralia								
Work item code	e: X	TEI						Date: ೫	4/11/2	2002	
Category:		F (con A (cor B (ada C (fun D (edi	rection) respon lition of ctional torial m olanatic	ds to a correction i f feature), modification of fea odification) ons of the above ca	ature)			Use <u>one</u> of 1 2 R96 R97 R98 R99 Rel-4 Rel-5	(GSM P (Releas (Releas (Releas (Releas (Releas (Releas	Phase 2) e 1996) e 1997) e 1998) e 1999) e 4) e 5)	
								Rel-6	(Releas	e 6)	
Reason for cha	nge	: ೫ ⁽ Test	purpo	se' sub-clause n	not in sy	nch w	/ith 'Co	nformance	require	ement' a	and 'Test

Consequences if # not approved:	Test purpose not clearly defined.
Summary of change: ೫	Second test purpose added, to reflect second conformance requirement (Ref 8.3.2 (b)) and steps 8 and 9 in 'Expected sequence'.
Reason for change.	procedure'.

Clauses affected:	¥ 11.3.1.3
Other specs	# X Other core specifications #
affected:	Test specifications O&M Specifications
Other comments:	# Affects R99, Rel-4 and Rel-5

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked **#** contain pop-up help information about the field that they are closest to.
- Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be

downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.

3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

<Start of modified section>

11.3.1 PDP context deactivation initiated by the UE

11.3.1.1 Definition

11.3.1.2 Conformance requirement

In order to deactivate a PDP context, the UE sends a DEACTIVATE PDP CONTEXT REQUEST message to the network, enters the state PDP-INACTIVE-PENDING and starts timer T3390. The message contains the transaction identifier (TI) in use for the PDP context to be deactivated and a cause code that typically indicates one of the following causes:

- *#*26: insufficient resources;
- #36: regular PDP context deactivation; or
- #37: QoS not accepted.

The network shall reply with the DEACTIVATE PDP CONTEXT ACCEPT message. Upon receipt of the DEACTIVATE PDP CONTEXT ACCEPT message, the UE shall stop timer T3390. In UMTS, the network shall initiate the release of Radio Access Bearer associated with this PDP context.

Whenever any session management message except REQUEST PDP CONTEXT ACTIVATION or SM-STATUS is received by the UE specifying a transaction identifier which is not recognised as relating to an active context or to a context that is in the process of activation or deactivation, the UE shall send a SM-STATUS message with cause #81 "invalid transaction identifier value" using the received transaction identifier value including the extension octet and remain in the PDP-INACTIVE state.

Reference

3GPP TS 24.008 clauses 6.1.3.4, 6.1.3.4.1 and 8.3.2 (b).

11.3.1.3 Test purpose

To test the behaviour of the UE upon receipt of a DEACTIVATE PDP CONTEXT ACCEPT message from the SS in PDP context deactivation procedure initiated by the UE.

To test the behaviour of the UE upon receipt of a session management message (except REQUEST PDP CONTEXT ACTIVATION or SM-STATUS) specifying a transaction identifier which is not recognised as relating to an active context or to a context that is in the process of activation or deactivation.

11.3.1.4 Method of test

Initial conditions

System Simulator:

1 cell, default parameters.

User Equipment:

The UE is in GMM-state "GMM-REGISTERED, normal service" with valid P-TMSI and CKSN.

Related ICS/IXIT statements

- PS Supported yes/no
- Method of activating a PDP context
- Method of deactivating the PDP context

Test procedure

A PDP context is activated by the user and accepted by the SS. PDP context deactivation is then requested by the user. The UE shall send a DEACTIVATE PDP CONTEXT REQUEST message to the SS. The SS shall then reply with a DEACTIVATE PDP CONTEXT ACCEPT message. The SS shall then wait for T3390 seconds to ensure T3390 has been stopped and that no further messages are sent from the UE. The SS shall then send a MODIFY PDP CONTEXT REQUEST for the deactivated context and the UE shall reply with an SM STATUS message with cause #81 'transaction identifier not known'.

Expected sequence

Step	Direction	Message	Comments
	UE SS		
1	UE		Initiate a context activation
1a	SS		SS checks that the IE "Establishment
			cause" in the received RRC CONNECTION
			REQUEST message is set to either Originating Conversational Call, Originating
			Streaming Call, Originating Interactive Call,
			Originating Background Call or Originating
			High Priority Signalling
1b	\rightarrow	SERVICE REQUEST	
1c	SS		The SS starts ciphering and integrity
2	\rightarrow	ACTIVATE PDP CONTEXT	protection. Activate a PDP context
2	7	REQUEST	Activate a PDP context
2a	SS		The SS establishes the RAB.
3	÷	ACTIVATE PDP CONTEXT	Accept the PDP context
		ACCEPT	
4	UE →	DEACTIVATE PDP CONTEXT	Initiate a context deactivation
5	7	REQUEST	Request a deactivation of a PDP context
6	←	DEACTIVATE PDP CONTEXT	SS accepts the PDP context deactivation
		ACCEPT	and starts waiting for 'T3390'.
6a	SS		The SS releases the RAB.
7	SS		SS waits for expiry of 'T3390' seconds to
			ensure no further deactivate request
8	←	MODIFY PDP CONTEXT	messages are sent Send a modify request to UE for the
0	ì	REQUEST (NETWORK TO UE	deactivated context.
		DIRECTION)	
9	\rightarrow	SM STATUŚ	Cause set to #81

Specific message contents

Steps 2 and 5. TI flag (bit 8) in TI IE is set to 0 (transaction initiated by the UE).

Step 3, 6 and 8. TI flag in TI IE is set to 1.

Steps 2, 3, 5, 6 and 8. The value of TIO IE (bits 5-7) of the transaction identifier (TI) is the same in these test steps.

11.3.1.5 Test requirements

At step 1a the UE shall send an RRC CONNECTION REQUEST message with the IE Establishment cause set to one of the following causes:

- Originating Conversational Call;
- Originating Streaming Call;
- Originating Interactive Call;
- Originating Background Call or
- Originating High Priority Signalling.

In PDP context deactivation procedure initiated by the UE, upon receipt of a DEACTIVATE PDP CONTEXT ACCEPT message from the SS, the UE shall deactivate PDP context associated with given PDP address and TI.

Then, upon modification procedure initiated by the network, for deactivated PDP context, UE shall reply with SM STATUS message with cause #81.

<End of modified section>

			C	CHANG	ERE	QUE	ST				CR-Form-v
H	34.1	<mark>23-1</mark>	CR	345	ж ге \	-	ж	Current	version:	5.1.1	ж
For <u>HELP</u>	on using	this for	m, see	bottom of t	his page	or look	at th	e pop-up	text ove	r the 🛱 sy	mbols.
Proposed cha	nge affec	<i>ts:</i> เ	JICC a	pps#	ME	X Ra	dio A	ccess Net	twork	Core N	etwork
Title:	¥ Mo	dificati	<mark>ons an</mark>	d correction	s of GMN	test c	ases				
Source:	ដ <mark>So</mark>	ny Cor	poratio	n, Ericsson							
Work item cod	le: ೫ TE	I						Date	e: ೫ <mark>05</mark>	/11/2002	
Category:	Deta	F (corr A (corr B (add C (fund D (edit ailed exp	rection) respond lition of ctional r torial mo blanatio	wing categor feature), nodification c odification) ns of the abo <u>R 21.900</u> .	tion in an e of feature)			2	<u>e</u> of the f (GS (Rel (Rel (Rel 4 (Rel 5 (Rel	el-5 ollowing re M Phase 2 ease 1996 ease 1997 ease 1999 ease 1999 ease 4) ease 5) ease 6))))
Reason for ch	ange: ೫	It is r - -	in TS to cha	ary; rect some to 24.008. ange some to ange some to ange some to	test cases	in orc	ler to	reflect the			•
Summary of cl	hange:	1.1	For su	ons of the te bclause 12.4 fication of th	4.1.1a"Ro	uting a	area u	updating /	-		
		(It is proof the	oposed to a UE attached ed in TS24.	dd the co d to PS de	nforma	ance i	requireme			
		(2		fication of th		rnose	the	Method of	test an	d the Test	·

In order to keep consistency with the above-mentioned modification, the Test purpose, the Method of test and the Test requirements in this test case are modified.

2. Corrections and modifications based on the changes in TS51.010.

It is proposed to change some test cases, based on the changes in TS51.010 that are informed by TSG-GERAN using the LS (T1S-020670).

requirements

2.1 For subclause 12.4.2.3 "Combined routing area updating / RA only accepted"

1) Correction of the Expected sequence.

In order to avoid an unnecessary location updating procedure before the UE performs a Routing Area Update request, the cell setting of cellB in the step34 is corrected.

2.2 For subclause 12.4.2.7 "Combined routing area updating / abnormal cases / attempt counter check / procedure timeout"

1) Modification of the Expected sequence.

Update type IE in the Step25 of the Expected sequence is modified because the message contents of Update type IE will be changed depending on whether the normal location updating procedure in step23a is performed or not (the step23a in the Expected sequence is an optional step.).

2.3 Modification of the test cases using a non-auto attaching UE

1) Modification of the Expected sequence

UE without automatic PS attach fails the current test cases, even if the UE is correctly implemented.

In order to avoid this problem, conditional steps for the UE are introduced to the Expected sequence.

12.2.2.7a, 12.3.2.5 & 12.4.2.4

2) Modification of Related ICS/IXIT statements in the method of test.

In relation to the above-mentioned modification, the Related ICS/IXIT statements in the method of test is modified.

3 Other changes

3.1 For subcluase 12.4.2.4, 12.4.2.5a (Change from T1S-020790)

In TS 23.122, section 3.1 it is stated: "The HPLMN shall not be stored on the list of forbidden PLMNs". It needs therefore to be ensured in TC 12.4.2.4 that the HPLMN does not send cause #11 'PLMN not allowed' in ROUTING AREA UPDATE REJECT.

RRC messages should in general not be shown in the NAS test cases. Which RRC procedures to be triggered by the SS should instead be indicated in the comments column.

The section 4.7.5.2.4 of TS 24.008 'Combined Routing Area Updating not accepted by the network' states that the MS upon receipt of a ROUTING AREA UPDATE REJECT with cause #13 'Roaming not allowed in this area' the MS shall change to state GMM-REGISTERED.LIMITED-SERVICE and the RAI, GPRS-CKSN, P-TMSI, P-TMSI signature and GPRS ciphering key sequence number shall not be deleted which indicate that a valid GMM context(s) still exists and is still attached to GPRS service.

(1) RRC messages have been removed and replaced with comments.

(2) In test case 12.4.2.4:

	The 'Conformance requirement' and the 'Method of test' are modified in order to prevent the PLMN sending cause #11 'PLMN not allowed' to be the same as the HPLMN.
	(3) In test case 12.4.2.5a: The Routing area update procedure is performed instead of the Attach procedure after reception of cause #13 'Roaming not allowed in this area'.
Consequences if solution consequences if solution consequences if a constant constant of the constant of the consequences of t	# In consistency with the core specification are left.
Clauses affected:	# 12.2.2.7a, 12.3.2.5, 12.4.1.1a, 12.4.2.3, 12.4.2.4, 12.4.2.5a, 12.4.2.7
Other specs	YN H Other core specifications #
Affected:	Test specifications O&M Specifications
Other comments:	₩

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

------<Start of modification>------

12.2.2.7a Combined PS attach / rejected / location area not allowed

12.2.2.7a.1 Definition

12.2.2.7a.2 Conformance requirement

- 1) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'location area not allowed' the User Equipment shall:
 - 1.1 not perform combined PS attach when in the same location area.
 - 1.2 delete the stored LAI, CKSN, TMSI, RAI, PS-CKSN, P-TMSI and P-TMSI signature.
 - 1.3 store the LA in the 'forbidden location areas for regional provision of service'.
 - 1.4 not delete the list of "equivalent PLMNs".
 - 1.5 perform a cell selection.
- 2) If the network rejects a combined PS attach procedure from the User Equipment with the cause 'location area not allowed' the User Equipment shall:
 - 2.1 perform combined PS attach when a new location area is entered.
 - 2.2 delete the list of forbidden LAs when power is switched off.

Reference

3GPP TS 24.008 clauses 4.7.3.2.

12.2.2.7a.3 Test purpose

To test the behaviour of the UE if the network rejects the combined PS attach procedure with the cause 'Location Area not allowed'.

To test that the UE deletes the list of forbidden LAs when power is switched off.

12.2.2.7a.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 (RAI-1), cell B in MCC1/MNC1/LAC1/RAC2 (RAI-4), cell C in MCC2/MNC1/LAC2/RAC1 (RAI-6). All cells are operating in network operation mode I.

The PLMN contains Cell C is equivalent to the PLMN that contains Cell A.

User Equipment:

The UE has a valid TMSI, P-TMSI and RAI.

Related ICS/IXIT statements

Support of PS serviceYes/NoUE operation mode AYes/NoSwitch off on buttonYes/NoAutomatic PS attach procedure at switch on or power onYes/NoPS attach attempted automatically by outstanding request Yes/No

Test procedure

The SS rejects a combined PS attach with the cause value 'Location Area not allowed'. The SS checks that the UE does not perform combined PS attach while in the location area, performs PS attach when a new location area is entered and deletes the list of forbidden LAs when switched off. CS services are not possible unless an IMSI attach procedure is performed.

Different types of UE may use different methods to periodically clear the list of forbidden location areas (e.g. every day at 12am). If the list is cleared while the test is being run, it may be necessary to re-run the test.

Expected Sequence

UE SS 1 SS 1 SS 1 SS 1 SS 2 UE 3 UE 3 UE 3a -> ATTACH REQUEST 3b <- ATTACH ACCEPT ATTACH ACCEPT ATTACH REQUEST C DETACH REQUEST C C C DETACH REQUEST ATTACH REQUEST The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. Attach type = 'Combined PS / IMSI attached" Mobile identity = P-TMSI-1 PoTMSI-1 Signature Mobile identity = PTMSI-1 P-TMSI-1 signature Mobile identity = TMSI-1 Routing area identity = RAI-1 Routing area identity = RAI-1 ATTACH REQUEST ATTACH REQUEST ATTACH REQUEST ATTACH REQUEST ATTACH REQUEST Attach type = 'Combined PS / IMSI atta''PS Attach while IMSI attached''' Mobile identity =	g cell". uitable uitable (see ind referred
1 SS received on cell A. 1 SS Set the cell type of cell A to the "Servin Set the cell type of cell B to the "Non-S cell". 2 UE Set the cell type of cell C to the "Non-S cell". 3 UE The UE is set in UE operation mode A ICS). 3 UE The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <- ATTACH ACCEPT Attach result = 'Combined PS / IMSI attached'' Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3c <- DETACH REQUEST DETACH REQUEST Detach type = re-attach required 3c <- DETACH REQUEST Detach type = 'Combined PS / IMSI attached'' Mobile identity = RAI-1 3d -> DETACH ACCEPT Attach type = 'Combined PS / IMSI attached'' Mobile identity = RAI-1 4 -> ATTACH REQUEST Detach type = 'Combined PS / IMSI attached'' Mobile identity = P-TMSI-1 4 -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached'' Mobile identity = P-TMSI-1	g cell". uitable uitable (see ind referred
1 SS Set the cell type of cell A to the "Servin Set the cell type of cell B to the "Non-S cell". 2 UE Set the cell type of cell C to the "Non-S cell". 3 UE The UE is set in UE operation mode A ICS). 3 UE The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	uitable uitable (see Ind referred
2 UE Set the cell type of cell B to the "Non-S cell". 3 UE Set the cell type of cell C to the "Non-S cell". 3 UE The UE is set in UE operation mode A ICS). 3a -> ATTACH REQUEST 3b <-	uitable uitable (see Ind referred
2 UE Cell". 3 UE The UE is set in UE operation mode A ICS). 3 UE The UE is set in UE operation mode A ICS). 3a -> ATTACH REQUEST 3b <-	uitable (see Ind referred
2 UE Set the cell type of cell C to the "Non-S cell". (see note) 3 UE The UE is set in UE operation mode A ICS). 3 UE The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	(see Ind referred
2 UE cell". 3 UE The UE is set in UE operation mode A ICS). 3 UE The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	(see Ind referred
2 UE (see note) 3 UE The UE is set in UE operation mode A ICS). 3 UE The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	ind referred
2 UE The UE is set in UE operation mode A ICS). 3 UE The UE is set in UE operation mode A ICS). 3a -> ATTACH REQUEST The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached" 3b <-	ind referred
3 UE ICS). 3a -> ATTACH REQUEST The UE is powered up or switched on a initiates an attach (see ICS). Cell A is p by the UE. 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	ind referred
3 UE 3a -> 3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	referred
3a -> ATTACH REQUEST initiates an attach (see ICS). Cell A is p by the UE. Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	referred
3a -> ATTACH REQUEST by the UE. 3b -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b <-	
3a -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta 3b -> ATTACH ACCEPT Mobile identity = P-TMSI-1 Routing area identity = RAI-1 3b ATTACH ACCEPT Attach result = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = TMSI-1 Routing area identity = RAI-1 Equivalent PLMNs = MCC2,MNC1 3c DETACH REQUEST DETACH ACCEPT Detach type = re-attach required 4 -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1	ch' or
3b<-	ch' or
3b <-	1
3b <-	
3b <-	
3b <-	
3c <-	ached'
3c DETACH REQUEST P-TMSI-1 signature 3d -> DETACH REQUEST Routing area identity = RAI-1 4 -> DETACH REQUEST Detach type = re-attach required 4 -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached'' Mobile identity = P-TMSI-1 Mobile identity = P-TMSI-1	
3c DETACH REQUEST Mobile identity = TMSI-1 3d -> DETACH REQUEST Detach type = re-attach required 4 -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached'' Mobile identity = TMSI-1 Attach type = 'Combined PS / IMSI attached''	
3c DETACH REQUEST Betach Required 3d -> DETACH ACCEPT Detach type = re-attach required 4 -> ATTACH REQUEST Attach type = 'Combined PS / IMSI attached'' Mobile identity = P-TMSI-1 P-TMSI-1	
3c DETACH REQUEST Detach type = re-attach required 3d > DETACH ACCEPT Attach type = 'Combined PS / IMSI attached'' 4 > ATTACH REQUEST Attach type = 'Combined PS / IMSI attached'' Mobile identity = P-TMSI-1 Detach type = P-TMSI-1	
3c DETACH REQUEST Detach type = re-attach required 3d > DETACH ACCEPT Attach type = 'Combined PS / IMSI atta 4 > ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1	
3d -> DETACH ACCEPT 4 -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1	
4 -> ATTACH REQUEST Attach type = 'Combined PS / IMSI atta "PS Attach while IMSI attached" Mobile identity = P-TMSI-1	
"PS Attach while IMSI attached" Mobile identity = P-TMSI-1	ch' or
Mobile identity = P-TMSI-1	
Routing area identity = RAI-1	
5 <- ATTACH REJECT GMM cause 'Location Area not allowed	
6 UE No LOCATION UPDATING REQ with t	/pe
'IMSI attach' is sent to the SS	
(SS waits 30 seconds).	
7 <- PAGING TYPE1 Mobile identity = TMSI	
Paging order is for CS services.	
8 UE The UE shall not initiate an RRC conne	ction.
This is checked during 3 seconds.	
9 <- PAGING TYPE1 Mobile identity = P-TMSI-1	
Paging order is for PS services.	
10 -> No response from the UE to the reques	t.
This is checked for 10 seconds	
The following messages are sent and s	hall be
received on cell B.	
11 SS Set the cell type of cell A to the "Non-S	uitable
cell".	
Set the cell type of cell B to the "Servin	g cell".
(see note)	
11a UE The UE performs cell selection.	
12 UE Cell B is preferred by the UE.	
13 UE No ATTACH REQUEST or LOCATION	
UPDATING REQ is sent to SS	
(SS waits 60 seconds)	
15 <- PAGING TYPE1 Mobile identity = P-TMSI-1	
Paging order is for PS services.	
16 UE No response from the UE to the request	t This
is checked for 10seconds.	
17 UE The UE initiates an attach by MMI or A	
command.	
	Г
checked for 10 seconds.	Г
The following messages are sent and s	T is
received on cell C.	T is

Step	Direction UE SS	Message	Comments
19	SS		Set the cell type of cell B to the "Non-Suitable
			cell".
			Set the cell type of cell C to the "Serving cell".
19a	UE		(see note) The UE performs cell selection
19a 20	UE		Cell C is preferred by the UE.
20	02		Step 20a and 20b areis only performed by an
			UE which will not initiate a PS attach
			automatically (see ICS)for non-auto attach UE
20a	UE	Pagistration on CS	and is optional.
conditio	UE	Registration on CS	Parameter Mobile identity is IMSI. See TS 34.108
nal			
20b	UE		UE initiates an attach automatically (see ICS)
conditio			via MMI or AT commands.
<u>nal</u>			
21	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
			TMSI status = no valid TMSI available
21a	<-	AUTHENTICATION AND	
		CIPHERING REQUEST	
21b	->	AUTHENTICATION AND	
01-	00	CIPHERING RESPONSE	The SS storte integrity protection
21c 22	SS <-	АТТАСН АССЕРТ	The SS starts integrity protection. Attach result = 'Combined PS / IMSI attached'
~~			Mobile identity = $P-TMSI1$
			P-TMSI-1 signature
			Mobile identity = TMSI-1
			Routing area identity = RAI-6
23	->	ATTACH COMPLETE	
24	<-	PAGING TYPE1	Mobile identity = TMSI-1 Paging order is for CS services.
25	->	RRC CONNECTION REQUEST	Faging order is for CS services.
26	<-	RRC CONNECTION SETUP	
27	->	RRC CONNECTION SETUP	
		COMPLETE	
28	->	PAGING RESPONSE	Mobile identity = TMSI-1
29	<-	RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
30	->	RRC CONNECTION RELEASE	
	-	COMPLETE	
31	<-	PAGING TYPE1	Mobile identity = P-TMSI-1
			Paging order is for PS services.
32	->	RRC CONNECTION REQUEST	
33 34	<- ->	RRC CONNECTION SETUP	
54	-	COMPLETE	
35	->	SERVICE REQUEST	Service type = "paging response"
36	<-	RRC CONNECTION RELEASE	
37	->	RRC CONNECTION RELEASE	
20		COMPLETE	The LIE is switched off or newer is removed
38	UE		The UE is switched off or power is removed (see ICS).
39	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined
			PS / IMSI detach'
			The following messages are sent and shall be
40			received on cell B.
40	UE		Set the cell type of cell B to the "Non-Suitable cell".
			Set the cell type of cell C to the "Serving cell".
			(see note)
			Cell B is preferred by the UE.
41	UE		The UE is powered up or switched on and
		1	initiates an attach (see ICS).

Step	Direction	Message	Comments
-	UE SS		
42			Step 43 is only performed for non-auto attach
10			
43	UE	Registration on CS	See TS 34.108
44	UE		UE initiates an attach automatically (see ICS),
45		ATTACH REQUEST	by MMI or AT commands. Attach type = 'Combined PS / IMSI attach' or
45	->	ATTACIT REQUEST	"PS Attach while IMSI attached"
			Mobile identity = P -TMSI-1
			Routing area identity = RAI-6
45a	<-	AUTHENTICATION AND	
		CIPHERING REQUEST	
45b	->	AUTHENTICATION AND	
		CIPHERING RESPONSE	
45c	SS		The SS starts integrity protection.
46	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Mobile identity = TMSI-2
47		ATTACH COMPLETE	Routing area identity = RAI-4
47	-> <-	PAGING TYPE1	Mobile identity = TMSI-2
-0	N ⁻		Paging order is for CS services.
49	->	RRC CONNECTION REQUEST	
50	<-	RRC CONNECTION SETUP	
51	->	RRC CONNECTION SETUP	
		COMPLETE	
52	->	PAGING RESPONSE	Mobile identity = TMSI-2
53	<-	RRC CONNECTION RELEASE	After sending of this message, the SS waits for
54			disconnection of the CS signalling link.
54	->	RRC CONNECTION RELEASE	
55		COMPLETE PAGING TYPE1	Mobile identity = P-TMSI-2
55	<-	FAGING TIFET	Paging order is for PS services.
56	->	RRC CONNECTION REQUEST	r aging order is for r o services.
57	<-	RRC CONNECTION SETUP	
58	->	RRC CONNECTION SETUP	
		COMPLETE	
59	->	SERVICE REQUEST	service type = "paging response"
60	<-	RRC CONNECTION RELEASE	
61	->	RRC CONNECTION RELEASE	
		COMPLETE	The UE is suitched off on power is not
62	UE		The UE is switched off or power is removed
63		DETACH REQUEST	(see ICS). Message not sent if power is removed.
03	->		Detach type = 'power switched off, combined
			PS / IMSI detach'
NOTE:	The definit	ions for "Non-Suitable cell" and "Se	rving cell" are specified in TS34.108 clause 6.1
		e Radio Conditions for signalling tes	
·			,

Specific message contents

None.

12.2.2.7a.5 Test requirements

At step4, when the UE is powered up or switched on, UE shall:

- initiate the combined PS attach procedure with the information elements specified in the above Expected Sequence

At step6, when the UE receives the ATTACH REJECT message with GMM cause = 'Location Area not allowed', UE shall:

- not initiate MM location updating procedure.

At step8, when the UE receives the paging message for CS domain, UE shall:

- not respond to the paging message for CS domain.

At step10 and 16, when the UE receives the paging message for PS domain, UE shall:

- not respond to the paging message for PS domain.

At step13 and 18, when the UE is in the same location area, UE shall:

- not perform PS attach procedure.

At step21, when the UE enters a new location area, UE shall

- perform the combined PS attach procedure.

At step28 and 52, when the UE receives the paging message for CS domain, UE shall:

- respond to the paging message for CS domain by sending the PAGING RESPONSE message.

At step35 and 59, when the UE receives the paging message for PS domain, UE shall:

- respond to the paging message for PS domain by sending the SERVICE REQUEST message.

At step45, when the UE is powered up or switched on, UE shall:

- perform the combined PS attach procedure.

12.3.2.5 PS detach / rejected / location area not allowed

12.3.2.5.1 Definition

12.3.2.5.2 Conformance requirement

- 1) If the network performs a PS detach procedure with the cause 'location area not allowed' the User Equipment shall:
 - 1.1 not perform combined PS attach when in the same location area.
 - 1.2 delete the stored LAI, CKSN, TMSI, RAI, PS-CKSN, P-TMSI and P-TMSI signature.
 - 1.3 store the LA in the 'forbidden location areas for regional provision of service'.
- 2) If the network performs a PS detach procedure with the cause 'location area not allowed' the User Equipment shall:
 - 2.1 perform combined PS attach when a new location area is entered.
 - 2.2 delete the list of forbidden LAs when power is switched off.

Reference

3GPP TS 24.008 clauses 4.7.4.2.

12.3.2.5.3 Test purpose

To test the behaviour of the UE if the network orders the PS detach procedure with the cause 'Location Area not allowed'.

To test that the UE deletes the list of forbidden LAs when power is switched off.

12.3.2.5.4 Method of test

Initial condition

System Simulator:

Three cells (not simultaneously activated), cell A in MCC2/MNC1/LAC1/RAC2 (RAI-2, Not HPLMN), cell B in MCC2/MNC1/LAC1/RAC2 (RAI-7, Not HPLMN), cell C in MCC2/MNC1/LAC2/RAC1 (RAI-6, Not HPLMN).

All cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No PS attach attempted automatically by outstanding request Yes/No

Test procedure

The SS orders a PS detach with the cause value 'Location Area not allowed'. The SS checks that the UE does not perform combined PS attach while in the location area, performs PS attach when a new location area is entered and deletes the list of forbidden LAs when switched off. CS services are not possible unless an IMSI attach procedure is performed.

Different types of UE may use different methods to periodically clear the list of forbidden location areas (e.g. every day at 12am). If the list is cleared while the test is being run, it may be necessary to re-run the test.

Expected Sequence

Step	Direction	Message	Comments
-	UE SS	_	
	SS		The following messages are sent and shall be
			received on cell A.
1	SS		Set the cell type of cell A to the "Serving cell". Set the cell type of cell B to the "Non-Suitable
			cell ".
			Set the cell type of cell C to the "Non-Suitable
			cell ".
2	UE		(see note)
2	UE		The UE is set in UE operation mode A (see ICS).
3	UE		The UE is powered up or switched on and
			initiates an attach (see ICS). Cell A is preferred
			by the UE.
4	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity = IMSI
			TMSI status = no valid TMSI available
4a	<-	AUTHENTICATION AND	
		CIPHERING REQUEST	
4b	->	AUTHENTICATION AND	
4c	SS	CIPHERING RESPONSE	The SS starts integrity protection.
40 5	<-	АТТАСН АССЕРТ	Attach result = 'Combined PS / IMSI attached'
-			Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Mobile identity = TMSI-1
6	->	ATTACH COMPLETE	Routing area identity = RAI-2
7	<-	DETACH REQUEST	Detach type = 're-attach not required'
			Cause 'Location Area not allowed'
8	->	DETACH ACCEPT	
9	UE		No LOCATION UPDATING REQ with type 'IMSI attach' is sent to the SS
			(SS waits 30 seconds).
10	<-	PAGING TYPE1	Mobile identity = TMSI-1
			Paging order is for CS services.
11	UE		The UE shall not initiate an RRC connection. This is checked during 3 seconds.
12	<-	PAGING TYPE1	Mobile identity = P -TMSI-1
.=			Paging order is for PS services.
13	UE		No response from the UE to the request.
			This is checked for 10 seconds
			The following messages are sent and shall be received on cell B.
14	SS		Set the cell type of cell A to the "Non-Suitable
			cell".
			Set the cell type of cell B to the "Serving cell".
15	UE		(see note) Cell B is preferred by the UE.
16	UE		The UE initiates an attach automatically, by
-			MMI or by AT command.
17	UE		No ATTACH REQUEST sent to SS
18	UE		(SS waits 30 seconds) No LOCATION UPDATING REQ with type
10			'IMSI attach' is sent to the SS
			(SS waits 30 seconds).
19	<-	PAGING TYPE1	Mobile identity = TMSI-1
20			Paging order is for CS services.
20	UE		The UE shall not initiate an RRC connection. This is checked during 3 seconds.
21	<-	PAGING TYPE1	Mobile identity = P -TMSI-1
			Paging order is for PS services.
22			No response from the UE to the request.
		l	This is checked for 10 seconds

Step	Direction	Message	Comments
	UE SS		
			The following messages are sent and shall be
	~~		received on cell C.
23	SS		Set the cell type of cell B to the "Non-Suitable
			cell".
			Set the cell type of cell C to the "Serving cell".
			(see note)
24	UE		Cell C is preferred by the UE.
			Step 25 and 26 areis only performed by an UE
			which will not initiate a PS attach automatically
			(see ICS)for non-auto attach UE.
25	UE	Registration on CS	See TS34.108
<u>condi</u> e			Parameter mobile identity is IMSI.
p tional			
26	UE		The UE initiates an attach automatically (See
conditio			ICS), by MMI or AT command.
nal			
27	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
			Mobile identity = IMSI
			TMSI status = no valid TMSI available
28	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI1
			P-TMSI-1 signature
			Mobile identity = TMSI-1
			Routing area identity = RAI-6
29	->	ATTACH COMPLETE	
30	<-	PAGING TYPE1	Mobile identity = TMSI-1
			Paging order is for CS services.
31	->	RRC CONNECTION REQUEST	
32	<-	RRC CONNECTION SETUP	
33	->	RRC CONNECTION SETUP	
	-	COMPLETE	
34	->	PAGING RESPONSE	Mobile identity = TMSI-1
35	<-	RRC CONNECTION RELEASE	After sending of this message, the SS waits for
00			disconnection of the CS signalling link.
36	->	RRC CONNECTION RELEASE	
50	-7	COMPLETE	
37	<-	PAGING TYPE1	Mobile identity = P-TMSI-1
07			Paging order is for PS services.
38	->	RRC CONNECTION REQUEST	
39		RRC CONNECTION SETUP	
	<-	RRC CONNECTION SETUP	
40	->	COMPLETE	
41			convice type - "pering response"
41 42	->		service type = "paging response"
	<-	RRC CONNECTION RELEASE	
43	->	RRC CONNECTION RELEASE	
14		COMPLETE	The LIE is ewitched off or new or is remained
44	UE		The UE is switched off or power is removed
45			(see ICS).
45	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined
			PS / IMSI detach'
			The following messages are sent and shall be
			received on cell B.
46	UE		Set the cell type of cell B to the "Serving cell".
			Set the cell type of cell C to the "Non-Suitable
			cell".
			(see note)
			Cell B is preferred by the UE.
47	UE		The UE is powered up or switched on and
			initiates an attach (see ICS).
			Step 48 is only performed for non-auto attach
			UE.
48	UE	Registration on CS	See TS34.108
			Parameter mobile identity is TMSI-1
40	UE		UE initiates an attach automatically (see ICS),
49			

Step	Direction	Message	Comments
	UE SS		
50 51	->	ATTACH REQUEST ATTACH ACCEPT	Attach type = 'Combined PS / IMSI attach' Mobile identity = P-TMSI-1 Routing area identity = RAI-6 TMSI status = valid TMSI available Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Mobile identity = TMSI-2
52	->	ATTACH COMPLETE	Routing area identity = RAI-7
53	<-	PAGING TYPE1	Mobile identity = TMSI-2 Paging order is for CS services.
54	->	RRC CONNECTION REQUEST	
55	<-	RRC CONNECTION SETUP	
56	->	RRC CONNECTION SETUP	
		COMPLETE	
57	->	PAGING RESPONSE	Mobile identity = TMSI-2
58	<-	RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.
59	->	RRC CONNECTION RELEASE	
60	<-	PAGING TYPE1	Mobile identity = P-TMSI-2 Paging order is for PS services.
61	->	RRC CONNECTION REQUEST	
62	<-	RRC CONNECTION SETUP	
63	->	RRC CONNECTION SETUP	
		COMPLETE	
64	->	SERVICE REQUEST	service type = "paging response"
65	<-	RRC CONNECTION RELEASE	
66	->	RRC CONNECTION RELEASE	
		COMPLETE	
67	UE		The UE is switched off or power is removed
			(see ICS).
68	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, combined PS / IMSI detach'
NOTE:	The definitions for "Non-Suitable cell" and "Serving cell" are specified in TS34.108 clause 6.1		
		e Radio Conditions for signalling test	

Specific message contents

None.

12.3.2.5.5 Test requirements

At step4, when the UE is powered up or switched on, UE shall:

- initiate the PS attach procedure with the information elements specified in the above Expected Sequence.

At step8, when the UE receive the DETACH REQUEST message (Detach type = 're-attach not required', Cause = 'Location Area not allowed') from SS, UE shall:

- send the DETACH ACCEPT message.

UE shall perform the following action depending on UE location.

1) UE is in the same location area.

- At step9 and 18, UE shall:
- not perform location updating procedure.

At step11 and 20, when the UE receives the paging message for CS domain, UE shall:

- not respond to the paging message for PS domain.

At step13 and 22, when the UE receives the paging message for PS domain, UE shall:

- not respond to the paging message for PS domain.
- At step17, UE shall;
- not perform PS attach procedure.
- 2) UE is in the new location area.
 - At step27, UE shall;
 - perform the combined PS attach procedure.
 - At step34, when the UE receives the paging message for CS domain with Mobile identity = IMSI, UE shall;
 - respond to the paging message for CS domain by sending the PAGING RESPONSE message.
 - At step41, when the UE receives the paging message for PS domain with Mobile identity = P-TMSI-1, UE shall:
 - respond to the paging message for PS domain by sending the SERVICE REQUEST message.

At step50, when the UE is powered up or switched on, UE shall:

- initiate the PS attach procedure with the information elements specified in the above Expected Sequence At step57, when the UE receives the paging message for CS domain with Mobile identity = IMSI, UE shall;
 - respond to the paging message for CS domain by sending the PAGING RESPONSE message.
- At step64, when the UE receives the paging message for PS domain with Mobile identity = P-TMSI-1, UE shall:
 - respond to the paging message for PS domain by sending the SERVICE REQUEST message.

----Next modification>------

12.4 Routing area updating procedure

This procedure is used to update the actual routing area of an UE in the network.

12.4.1 Normal routing area updating

The routing area updating procedure is a GMM procedure used by PS UEs of UE operation mode A or C that are IMSI attached for PS services only.

12.4.1.1a Routing area updating / accepted

- 12.4.1.1a.1 Definition
- 12.4.1.1a.2 Conformance requirement
 - 1) If the network accepts the routing area updating procedure and reallocates a P-TMSI, the UE shall acknowledge the new P-TMSI and continue communication with the new P-TMSI.
 - 2) If the network accepts the routing area updating procedure from the UE without reallocation of the old P-TMSI, the UE shall continue communication with the old P-TMSI.
 - 3) The routing area updating procedure shall also be used by a UE which is attached for PS services if a new PLMN is entered.

Reference

3GPP TS 24.008 clause <u>4.7.5, 4.7.5.1</u>.

12.4.1.1a.3 Test purpose

To test the behaviour of the UE if the network accepts the routing area updating procedure.

The following cases are identified:

- 1) P-TMSI / P-TMSI signature is reallocated.
- 2) Old P-TMSI / P-TMSI signature is not changed.

To test the behaviour of the UE if the UE enters the new PLMN.

12.4.1.1a.4 Method of test

Initial condition

System Simulator:

<u>ThreeTwo</u> cells, cell A in MCC1/MNC1/LAC1/RAC1 (RAI-1), cell B in MCC1/MNC1/LAC1/RAC2 (RAI-4), cell C in MCC2/MNC1/LAC1/RAC2 (RAI-7).

<u>All three</u>Both cells are operating in network operation mode II.

The PLMN contains cell C is equivalent to the PLMN that contains cell A.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS serviceYes/NoUE operation mode AYes/NoUE operation mode CYes/NoSwitch off on buttonYes/NoAutomatic PS attach procedure at switch on or power onYes/No

Test procedure

- The UE sends a ROUTING AREA UPDATE REQUEST message. The SS reallocates the P-TMSI and returns ROUTING AREA UPDATE ACCEPT message with a new P-TMSI. The UE acknowledge the new P-TMSI by sending ROUTING AREA UPDATE COMPLETE message. Further communication UE - SS is performed by the new P-TMSI. The UE will not answer signalling addressed to the old P-TMSI.
- The UE sends a ROUTING AREA UPDATE REQUEST message. The SS accepts the P-TMSI and returns ROUTING AREA UPDATE ACCEPT message without any P-TMSI. Further communication UE - SS is performed by the P-TMSI.

3) The UE sends a ROUTING AREA UPDATE REQUEST message. The SS reallocates the P-TMSI and returns ROUTING AREA UPDATE ACCEPT message with a new P-TMSI. The UE acknowledge the new P-TMSI by sending ROUTING AREA UPDATE COMPLETE message.

Step	Direction	Message	Comments
-	UE SS	_	
			The following messages are sent and shall be
1	SS		received on cell A. Set the cell type of cell A to the "Serving cell".
	00		Set the cell type of cell B to the "Suitable
			neighbour cell".
			Set the cell type of cell C to the "Suitable
			neighbour cell".
			(see note)
2	UE		The UE is set to attach to PS services only
			(see ICS). If that is not supported bytby the UE, goto step <u>32</u> 2.
3	UE		The UE is powered up or switched on and
Ū	02		initiates an attach (see ICS).
3a	SS		The SS checks that the IE "Establishment
			cause" in the received RRC CONNECTION
			REQUEST message is set to "Registration".
4	->	ATTACH REQUEST	Attach type = 'PS attach'
4a	<-	AUTHENTICATION AND	Mobile identity = IMSI
- 1 a	N -	CIPHERING REQUEST	
4b	->	AUTHENTICATION AND	
		CIPHERING RESPONSE	
4c	SS		The SS starts integrity protection.
5	<-	ATTACH ACCEPT	Attach result = 'PS only attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature Routing area identity = RAI-1
			Equivalent PLMN: $MCC = 2$, $MNC = 1$
6	->	ATTACH COMPLETE	$\frac{2}{2}$
6a	SS		The SS releases the RRC connection.
			The following messages are sent and shall be
_			received on cell B.
7	SS		Set the cell type of cell A to the "Suitable
			neighbour cell". Set the cell type of cell B to the "Serving cell".
			(see note)
7a	SS		The SS checks that the IE "Establishment
			cause" in the received RRC CONNECTION
			REQUEST message is set to "Registration".

Expected Sequence

Step	Direction UE SS	Message	Comments
8	->	ROUTING AREA UPDATE REQUEST	Update type = 'RA updating' P-TMSI-2 signature Routing area identity = RAI-1
8a 9	SS <-	ROUTING AREA UPDATE ACCEPT	The SS starts integrity protection. Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature
10	->	ROUTING AREA UPDATE COMPLETE	Routing area identity = RAI-4
11 11b 11c	SS	Void Void	The SS releases the RRC connection.
11d	<-	PAGING TYPE1	Mobile identity = P-TMSI-1 Paging order is for PS services.
11e	SS		SS verifies that the UE transmits an RRC CONNECTION REQUEST message. SS will reject this request. The IE "Establishment cause" is not checked.
12	<-	PAGING TYPE1	Mobile identity = P-TMSI-2 Paging order is for PS services.
13	UE		No response from the UE to the request. This is checked for 10 seconds.
14	SS		The following messages are sent and shall be received on cell A. Set the cell type of cell A to the "Serving cell". Set the cell type of cell B to the "Suitable neighbour cell".
15 15a	UE SS		(see note) Cell A is preferred by the UE. The SS checks that the IE "Establishment cause" in the received RRC CONNECTION
16	->	ROUTING AREA UPDATE REQUEST	REQUEST message is set to "Registration". Update type = 'RA updating' P-TMSI-1 signature Routing area identity = RAI-4
16a	SS		The SS starts integrity protection.
17	<-	ROUTING AREA UPDATE ACCEPT	No new mobile identity assigned. P-TMSI not included. Update result = 'RA updated' P-TMSI- <u>1</u> 2_signature Routing area identity = RAI-1
17a	SS		The SS releases the RRC connection.
18	<-	PAGING TYPE1	Mobile identity = P-TMSI-1 Paging order is for PS services. Paging cause = "Terminating interactive call".
18a	SS		The SS checks that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Terminating interactive call"
18b 18c 19	->	Void Void SERVICE REQUEST	service type = "paging response"
19aa 19a	SS SS		The SS starts integrity protection. The SS releases the RRC connection.
<u>20</u>	<u>SS</u>		The following messages are sent and shall be received on cell C. Set the cell type of cell A to the "Suitable neighbour cell". Set the cell type of cell C to the "Serving cell".
<u>21</u> <u>22</u>	UE SS		(see note) Cell C is preferred by the UE. The SS checks that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Registration".

Step	Direction	Message	Comments
	UE SS		
<u>23</u>	<u>-></u>	ROUTING AREA UPDATE	Update type = 'RA updating'
		REQUEST	P-TMSI-1 signature
			Routing area identity = RAI-1
<u>24</u> 25	<u>SS</u>		The SS starts integrity protection.
<u>25</u>	<-	ROUTING AREA UPDATE	Update result = 'RA updated'
		ACCEPT	$\frac{\text{Mobile identity} = P-TMSI-3}{2}$
			P-TMSI-3 signature
			Routing area identity = RAI-7
<u>26</u>	<u>-></u>	ROUTING AREA UPDATE	
07		COMPLETE	
<u>27</u>	<u>SS</u>		The SS releases the RRC connection.
<u>28</u> 20	UE		The UE is switched off or power is removed
			(see ICS).
<u>29</u> 20a	SS		The SS checks that the IE "Establishment
			cause" in the received RRC CONNECTION
			REQUEST message is set to "Detach".
<u>30</u> 21	->	DETACH REQUEST	Message not sent if power is removed.
			Detach type = 'power switched off, PS detach'
<u>31</u> 21a	SS		The SS releases the RRC connection.
<u>32</u> 22	UE		The UE is set to attach to both the PS and non-
			PS services (see ICS) and the test is repeated
			from step 3 to step <u>3121b</u> .
NOTE:			d "Serving cell" are specified in TS34.108 clause
	6.1 "Refere	ence Radio Conditions for signalling	test cases only".

Specific message contents

None.

12.4.1.1a.5 Test requirements

At step 3a, 7a, and 15a and 22 the UE shall send an RRC CONNECTION REQUEST message with the IE Establishment cause set to "Registration".

At step 18a the UE shall send an RRC CONNECTION REQUEST message with the IE Establishment cause set to "Terminating Interactive Call".

At step <u>29</u>20a the UE shall send an RRC CONNECTION REQUEST message with the IE Establishment cause set to "Detach".

At step4, when the UE is powered up or switched on, UE shall:

- initiate the PS attach procedure with the information elements specified in the above Expected Sequence.

At step8, UE shall;

- initiate the routing area updating procedure with the information elements specified in the above Expected Sequence.

At step13, when the UE receives the paging message for PS domain with Mobile identity = P-TMSI-2, UE shall:

- not respond to the paging message for PS domain.

At step16, UE shall;

- initiate the routing area updating procedure with the information elements specified in the above Expected Sequence.

At step19, when the UE receives the paging message for PS domain with Mobile identity = P-TMSI-1, UE shall:

- respond to the paging message for PS domain by sending the SERVICE REQUEST message.

At step23, UE shall;

- initiate the routing area updating procedure with the information elements specified in the above Expected <u>Sequence.</u>

----Next modification>------

12.4.2.3 Combined routing area updating / RA only accepted

12.4.2.3.1 Definition

12.4.2.3.2 Conformance requirement

- 1) If the network accepts the combined PS attach procedure, but GMM cause code 'IMSI unknown in HLR' is sent to the UE the User Equipment shall delete the stored TMSI, LAI and CKSN. The User Equipment shall consider USIM invalid for non-PS services until power is switched off or USIM is removed.
- If the network accepts the combined PS attach procedure, but GMM cause code 'MSC temporarily not reachable', 'Network failure' or 'Congestion' is sent to the UE, an UE operation mode A UE may perform an MM IMSI attach procedure.

Reference

3GPP TS 24.008 clause 4.7.5.2.

12.4.2.3.3 Test purpose

Test porpose1

To test the behaviour of the UE if the network accepts the routing area updating procedure with indication RA only, GMM cause 'IMSI unknown in HLR'.

Test porpose2

To test the behaviour of the UE if the network accepts the routing area updating procedure with indication RA only, GMM cause 'MSC temporarily not reachable', 'Network failure' or 'Congestion'.

12.4.2.3.4 Method of test

Test Procedure1

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 (RAI-1), cell B in MCC1/MNC1/LAC1/RAC2 (RAI-4). Both cells operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS serviceYes/NoUE operation mode AYes/NoSwitch off on buttonYes/NoAutomatic PS attach procedure at switch on or power onYes/No

Test procedure

After attach, the UE sends an ROUTING AREA UPDATE REQUEST message. The SS allocates a P-TMSI and returns ROUTING AREA UPDATE ACCEPT message with a P-TMSI. GMM cause 'IMSI unknown in HLR' is indicated from SS. Further communication UE - SS is performed by the P-TMSI. CS services are not possible.

Expected Sequence

Step	Direction	Message	Comments	
1	UE SS		Set the cell type of cell A to the "Serving cell"	
1	SS		Set the cell type of cell A to the "Serving cell". Set the cell type of cell B to the "Non-Suitable cell".	
1a	UE		(see note) The UE is set in UE operation mode A (see ICS).	
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).	
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI TMSI status = no valid TMSI available	
3a	<-	AUTHENTICATION AND CIPHERING REQUEST		
3b	->	AUTHENTICATION AND CIPHERING RESPONSE		
3c 4	SS <-	ATTACH ACCEPT	The SS starts integrity protection. Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature	
5	->	ATTACH COMPLETE	Routing area identity = RAI-1	
6	SS		The following messages are sent and shall be received on cell B. Set the cell type of cell A to the "Suitable neighbour cell". Set the cell type of cell B to the "Serving cell".	
7	->	ROUTING AREA UPDATE REQUEST	(see note) Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1	
8	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-4	
9	->	ROUTING AREA UPDATE COMPLETE	GMM cause = 'IMSI unknown in HLR'	
10	<-	PAGING TYPE1	Mobile identity = P-TMSI-1 Paging order is for PS services.	
10a 10b	->	RRC CONNECTION REQUEST RRC CONNECTION SETUP		
100 10c	<- ->	RRC CONNECTION SETUP RRC CONNECTION SETUP COMPLETE		
11	->	SERVICE REQUEST	service type = "paging response"	
11a 11b	<- ->	RRC CONNECTION RELEASE RRC CONNECTION RELEASE COMPLETE		
12	<-	PAGING TYPE1	Mobile identity = IMSI	
13	UE		Paging order is for CS services. The UE shall not initiate an RRC connection. This is checked during 3 seconds.	
14	UE		The UE is switched off or power is removed (see ICS).	
15	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'	
NOTE:				

Test Procedure2

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 (RAI-1), cell B in MCC1/MNC1/LAC1/RAC2 (RAI-4). Both cells operating in network operation mode I. T3212 is set to 6 minutes.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS service Yes/No UE operation mode A Yes/No Automatic MM IMSI attach procedure for UE operation mode A UE Yes/No Switch off on button Yes/No Automatic PS attach procedure at switch on or power on Yes/No

Test procedure

After attach, the UE sends an ROUTING AREA UPDATE REQUEST message . The SS allocates a new P-TMSI signature and returns ROUTING AREA UPDATE ACCEPT message. GMM cause 'MSC temporarily not reachable', 'Network failure' or 'Congestion' is indicated from SS. The cause code is arbitrarily chosen. This procedure is repeated until the routing area updating attempt counter is equal to five. An UE operation mode A UE may perform an MM IMSI attach procedure (according to the ICS statement). Further communication UE - SS is performed by the P-TMSI. The existence of a signalling channel is verified by a request for mobile identity. It is further verified that the UE after a successful IMSI attach procedure can perform CS services.

Expected Sequence

Dependent whether the option 'Automatic MM IMSI attach procedure for UE operation mode A UE' is not supported or not, the steps 1-13 or 14-35 apply depending on manufacturer (see ICS).

Step	Direction	Message	Comments
	UE SS		
1	SS		The following messages are sent and shall be received on cell A Set the cell type of cell A to the "Serving cell". Set the cell type of cell B to the "Non-Suitable cell".
1a	UE		(see note) The UE is set in UE operation mode A and no automatic MM IMSI attach procedure is indicated (see ICS).
2	UE		The UE is powered up or switched on and initiates an attach (see ICS).
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' or 'PS attach while IMSI attached' Mobile identity =IMSI TMSI status = no valid TMSI available
3a	<-	AUTHENTICATION AND CIPHERING REQUEST	
Зb	->	AUTHENTICATION AND CIPHERING RESPONSE	
3c	SS		The SS starts integrity protection.
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-1
5	->	ATTACH COMPLETE	

Step	Direction	Message	Comments
	UE SS		The following messages are sent and shall be
			received on cell B.
6	SS		Set the cell type of cell A to the "Suitable
			neighbour cell".
			Set the cell type of cell B to the "Serving cell".
-			(see note)
7	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature
		REQUEST	Routing area identity = RAI-1
			TMSI status = no valid TMSI available
8	<-	ROUTING AREA UPDATE	Update result = 'RA updated'
		ACCEPT	Mobile identity = P-TMSI-1P-TMSI-1 signature
			Routing area identity = RAI-4
			GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily
			chosen)
9	->	ROUTING AREA UPDATE	
-		COMPLETE	
10			The routing area updating attempt counter =1.
			The combined routing area updating procedure
4.4		ROUTING AREA UPDATE	is reinitialised at the expiry of T3311
11	->	REQUEST	Update type = 'Combined RA/LA updating□ with IMSI attach'
			P-TMSI-1 signature
			Routing area identity = $RAI-4$
			TMSI status = no valid TMSI available
12	<-	ROUTING AREA UPDATE	Update result = 'RA updated'
		ACCEPT	Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = RAI-4 GMM cause = 'MSC temporarily not reachable'.
			'Network failure' or 'Congestion' (arbitrarily
			chosen)
13	->	ROUTING AREA UPDATE	,
		COMPLETE	
14			The routing area updating attempt counter =2.
			The combined routing area updating procedure is reinitialised at the expiry of T3311
15	->	ROUTING AREA UPDATE	Update type = 'Combined RA/LA updating
		REQUEST	with IMSI attach'
			P-TMSI-1 signature
			Routing area identity = RAI-4
40			TMSI status = no valid TMSI available
16	<-	ROUTING AREA UPDATE	Update result = 'RA updated' Mobile identity = P-TMSI-1
			P-TMSI-1 signature
			Routing area identity = $RAI-4$
			GMM cause = 'MSC temporarily not reachable',
			'Network failure' or 'Congestion' (arbitrarily
47			chosen)
17	->	ROUTING AREA UPDATE	
18			The routing area updating attempt counter =3.
10			The combined routing area updating procedure
			is reinitialised at the expiry of T3311
19	->	ROUTING AREA UPDATE	Update type = 'Combined RA/LA updating
		REQUEST	with IMSI attach'
			P-TMSI-1 signature
			Routing area identity = RAI-4
	1	1	TMSI status = no valid TMSI available

ļ

Step	Direction UE SS	Message	Comments
20	<-	ROUTING AREA UPDATE ACCEPT	Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-4 GMM cause = 'MSC temporarily not reachable', 'Network failure' or 'Congestion' (arbitrarily
21	->	ROUTING AREA UPDATE	chosen)
22			The routing area updating attempt counter =4. The combined routing area updating procedure
23	->	ROUTING AREA UPDATE REQUEST	is reinitialised at the expiry of T3311 Update type = 'Combined RA/LA updating□ with IMSI attach' P-TMSI-1 signature Routing area identity = RAI-4
24	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-4 GMM cause = 'MSC temporarily not reachable' 'Network failure' or 'Congestion' (arbitrarily chosen)
25	->	ROUTING AREA UPDATE COMPLETE	
26			The routing area updating attempt counter =5. The combined routing area updating procedure is reinitialised at the expiry of T3311
27	UE		The UE is switched off or power is removed (see ICS).
28	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach' Stop the sequence.
			The following messages are sent and shall be received on cell B
29	UE		The UE is set in UE operation mode A and automatic MM IMSI attach procedure is indicated (see ICS).
30	UE		The UE is powered up or switched on and initiates an attach (see ICS).
31	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' or 'PS attach while IMSI attached' Mobile identity = IMSI
31a	<-		TMSI status = no valid TMSI available
31b	->	CIPHERING REQUEST AUTHENTICATION AND CIPHERING RESPONSE	
31c 32	SS <-	ATTACH ACCEPT	The SS starts integrity protection. Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-2 P-TMSI-2 signature Routing area identity = RAI-4
33	->	ATTACH COMPLETE	
34	SS		The following messages are sent and shall be received on cell A. Set the cell type of cell A to the "Serving cell". Set the cell type of cell B to the " <u>Suitable</u> <u>neighbour cell</u> <u>Non-Suitable cell</u> ". (see note)
35	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-4 TMSI status = no valid TMSI available

Step	Direction UE SS	Message	Comments
36	<-	ROUTING AREA UPDATE ACCEPT	Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable 'Network failure' or 'Congestion' (arbitrarily chosen)
37	->	ROUTING AREA UPDATE	
38			The routing area updating attempt counter =1. The combined routing area updating procedure is reinitialised at the expiry of T3311
39	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating□ with IMSI attach' P-TMSI-1 signature Routing area identity = RAI-1
40	<-	ROUTING AREA UPDATE ACCEPT	TMSI status = no valid TMSI available Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable 'Network failure' or 'Congestion' (arbitrarily chosen)
41	->	ROUTING AREA UPDATE	
42			The routing area updating attempt counter =2. The combined routing area updating procedure is reinitialised at the expiry of T3311
43	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating with IMSI attach' P-TMSI-1 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available
44	<-	ROUTING AREA UPDATE ACCEPT	Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable 'Network failure' or 'Congestion' (arbitrarily chosen)
45	->	ROUTING AREA UPDATE	
46			The routing area updating attempt counter =3. The combined routing area updating procedure is reinitialised at the expiry of T3311
47	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating□ with IMSI attach' P-TMSI-1 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available
48	<-	ROUTING AREA UPDATE ACCEPT	Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable 'Network failure' or 'Congestion' (arbitrarily chosen)
49	->	ROUTING AREA UPDATE	
50			The routing area updating attempt counter =4. The combined routing area updating procedure is reinitialised at the expiry of T3311

Step	Directio	n Message	Comments		
•	UE SS				
51	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating□with IMSI attach'		
52	<-	ROUTING AREA UPDATE ACCEPT	P-TMSI-1 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available Update result = 'RA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-1 GMM cause = 'MSC temporarily not reachable',		
			'Network failure' or 'Congestion' (arbitrarily chosen)		
53	->	ROUTING AREA UPDATE			
54			The routing area updating attempt counter =5.		
55	UE	Registration on CS	Optional step. See TS 34.108 This is applied only for UE in UE operation mode A. Parameter mobile identity is TMSI-1. Steps 56 - 62 are only performed if the UE has performed the Registration Procedure in step 55.		
56	<-	PAGING TYPE1	Mobile identity = TMSI-1 Paging order is for CS services.		
57	->	RRC CONNECTION REQUEST			
58	<-	RRC CONNECTION SETUP			
59	->	RRC CONNECTION SETUP COMPLETE			
60	->	PAGING RESPONSE	Mobile identity = TMSI-1		
61	<-	RRC CONNECTION RELEASE	After sending of this message, the SS waits for disconnection of the CS signalling link.		
62	->	RRC CONNECTION RELEASE			
63	UE		The UE is switched off or power is removed (see ICS).		
64	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, PS detach'		
NOTE:			le neighbour cell" and "Serving cell" are specified		
	in TS34.108 clause 6.1 "Reference Radio Conditions for signalling test cases only".				

Specific message contents

None.

12.4.2.3.5 Test requirements

Test requirements for Test Procedure1

At step3, when the UE is powered up or switched on, UE shall:

- initiate the combined PS attach procedure with information elements specified in the above Expected Sequence.

At step7, when the RF level of the attached cell is lower than the RF level of the new cell, UE shall:

- initiate the combined routing area updating procedure.

At step9, UE shall:

- acknowledge the new P-TMSI by sending the ROUTING AREA UPDATE COMPLETE message.

At step11, when the UE receives the paging message for PS domain, UE shall:

- respond to the paging message for PS domain by sending the SERVICE REQUEST message.

At step13, when the UE receives the paging message for CS domain, UE shall:

- not respond to the paging message for CS domain.
- Test requirements for Test Procedure2
- At step3 and 31, when the UE is powered up or switched on, UE shall:
 - initiate the combined PS attach procedure with information elements specified in the above Expected Sequence.
- At step6 and 35, when the RF level of the attached cell is lower than the RF level of the new cell, UE shall:
 - initiate the combined routing area updating procedure.

At step11, 15, 19 and 23, UE shall:

- re-initiate the combined routing area updating procedure.

At step39, 43, 47 and 51, UE shall:

- re-initiate the combined routing area updating procedure.

At step55, UE shall:

- perform MM location updating procedure.

At step60, when the UE receives the paging message for CS domain, UE shall:

- not respond to the paging message for CS domain.

12.4.2.4 Combined routing area updating / rejected / PLMN not allowed

12.4.2.4.1 Definition

12.4.2.4.2 Conformance requirement

- 1) If the network rejects a combined routing area updating procedure from the User Equipment with the cause 'PLMN not allowed' the User Equipment shall:
 - 1.1 not perform combined GPRA attach when switched on in the same location area or PLMN, except when the PLMN identity is equal to the HPLMN.
 - 1.2 delete the stored RAI, PS-CKSN, P-TMSI, P-TMSI signature, TMSI CKSN and LAI.
 - 1.3 store the PLMN in the 'forbidden PLMN list', except when the PLMN identity is equal to the HPLMN.
- 1) An MS that receives a ROUTING AREA UPDATE REJECT message stops timer T3330, enters state MM IDLE and for all causes except #12, #14 and #15 deletes the list of "equivalent PLMNs".

Reference

3GPP TS 24.008 clause 4.7.5.2.

3GPP TS 23.122 clause 3.1.

12.4.2.4.3 Test purpose

To test the behaviour of the UE if the network rejects the combined routing area updating procedure of the UE with the cause 'PLMN not allowed'.

12.4.2.4.4 Method of test

Initial condition

System Simulator:

Five cells (not simultaneously activated), cell A in MCC1/MNC2/LAC1/RAC1 (RAI-8), cell B in MCC1/MNC2/LAC1/RAC2 (RAI-10), cell C in MCC1/MNC2/LAC2/RAC1 (RAI-9) and cell D in MCC2/MNC1/LAC1/RAC1 (RAI-2), cell E in MCC1/MNC3/LAC1/RAC1 (RAI-11).

The PLMN containing Cell E is equivalent to the PLMN that contains Cell A. All five cells are operating in network operation mode I

The HPLMN is different from MCC1/MNC2.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS serviceYes/NoUE operation mode AYes/NoSwitch off on buttonYes/NoAutomatic PS attach procedure at switch on or power onYes/NoPS attach attempted automatically by outstanding request Yes/No

Test procedure

The SS rejects a combined routing area updating with the cause value 'PLMN not allowed'. The SS checks that the UE does not perform PS attach if activated in the same PLMN. The SS checks that the UE does not perform IMSI attach if activated in the same PLMN.

l

Expected Sequence

Step	Direction	Message	Comments
	UE SS		The following manages are cent and shall be
			The following messages are sent and shall be received on cell A.
1	SS		Set the cell type of cell A to the "Serving cell".
			Set the cell type of cell B to the "Non-Suitable
			cell".
			Set the cell type of cell C to the "Non-Suitable
			cell".
			Set the cell type of cell D to the "Non-Suitable cell".
			Set the cell type of cell E to the "Non-Suitable
			cell".
			(see note)
2	UE		The UE is powered up or switched on and
0.5			initiates an attach (see ICS.
<u>2a</u>	<u>SS</u>		The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION
			REQUEST message is set to "Registration".
3	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach'
-			Mobile identity =IMSI
			TMSI status = no valid TMSI available
3a	<-	AUTHENTICATION AND	
2 h			
3b	->	AUTHENTICATION AND CIPHERING RESPONSE	
3c	SS	CIFTIERING RESPONSE	The SS starts integrity protection.
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-8
			Mobile identity = TMSI-1
5	->	ATTACH COMPLETE	Equivalent PLMN: MCC = 1, MNC=3
<u>5a</u>	SS		The SS releases the RRC connection.
			The following messages are sent and shall be
			received on cell B and cell E.
_			
7	SS		Set the cell type of cell A to the "Suitable
7	SS		neighbour cell".
7	SS		neighbour cell". Set the cell type of cell B to the "Serving cell".
7	SS		neighbour cell".
			neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note)
8	UE		neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE.
			neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u>
8	UE		neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u>
8 <u>8a</u>	UE <u>SS</u>		neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u> REQUEST message is set to "Registration".
8	UE	ROUTING AREA UPDATE REQUEST	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u> <u>REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating'
8 <u>8a</u>	UE <u>SS</u>	ROUTING AREA UPDATE REQUEST	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u> <u>REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8
8 <u>8a</u> 9	UE <u>SS</u>	REQUEST	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available
8 <u>8a</u>	UE <u>SS</u>	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u> <u>REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8
8 <u>8a</u> 9	UE <u>SS</u> -> <-	REQUEST	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed'
8 <u>8a</u> 9 10 <u>10a</u>	UE <u>SS</u> -> <- <u>SS</u>	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u>
8 <u>8a</u> 9	UE <u>SS</u> -> <-	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u> The UE initiates an attach by MMI or AT
8 <u>8a</u> 9 10 <u>10a</u>	UE <u>SS</u> -> <- <u>SS</u>	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u>
8 <u>8a</u> 9 10 <u>10a</u> 11	UE <u>SS</u> -> <- <u>SS</u> UE	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u> The UE initiates an attach by MMI or AT command.
8 <u>8a</u> 9 10 <u>10a</u> 11	UE <u>SS</u> -> <- <u>SS</u> UE	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u> The UE initiates an attach by MMI or AT command. No ATTACH REQUEST sent to SS (SS waits 30 seconds). The SS deactivates cell E.
8 <u>8a</u> 9 10 <u>10a</u> 11 12	UE <u>SS</u> -> <- SS UE UE	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u> The UE initiates an attach by MMI or AT command. No ATTACH REQUEST sent to SS (SS waits 30 seconds). The SS deactivates cell E. Set the cell type of cell E to the "Non-Suitable
8 <u>8a</u> 9 10 <u>10a</u> 11 12 12a	UE SS -> VE UE SS	REQUEST ROUTING AREA UPDATE REJECT	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u> <u>REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u> The UE initiates an attach by MMI or AT command. No ATTACH REQUEST sent to SS (SS waits 30 seconds). The SS deactivates cell E. Set the cell type of cell E to the "Non-Suitable cell".
8 <u>8a</u> 9 10 <u>10a</u> 11 12	UE <u>SS</u> -> <- SS UE UE	REQUEST ROUTING AREA UPDATE	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u> <u>REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u> The UE initiates an attach by MMI or AT command. No ATTACH REQUEST sent to SS (SS waits 30 seconds). The SS deactivates cell E. Set the cell type of cell E to the "Non-Suitable cell". Mobile identity = P-TMSI-2
8 <u>8a</u> 9 10 <u>10a</u> 11 12 12a	UE SS -> VE UE SS	REQUEST ROUTING AREA UPDATE REJECT	neighbour cell". Set the cell type of cell B to the "Serving cell". Set the cell type of cell E to the "Suitable neighbour cell". (see note) Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> <u>cause" in the received RRC CONNECTION</u> <u>REQUEST message is set to "Registration".</u> Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-8 TMSI status = valid TMSI available GMM cause = 'PLMN not allowed' <u>The SS releases the RRC connection.</u> The UE initiates an attach by MMI or AT command. No ATTACH REQUEST sent to SS (SS waits 30 seconds). The SS deactivates cell E. Set the cell type of cell E to the "Non-Suitable cell".

l

Step	Direction	Message	Comments
15	UE SS		The following messages are sent and shall be received on cell C. Set the cell type of cell B to the "Non-Suitable
16	UE		cell". Set the cell type of cell C to the "Serving cell". (see note) Cell C is preferred by the UE.
17	UE		The UE initiates an attach by MMI or by AT command.
18	UE		No ATTACH REQUEST sent to SS (SS waits 30 seconds).
19	<-	PAGING TYPE1	Mobile identity = TMSI-1 Paging order is for CS services.
20	UE		The UE shall not initiate an RRC connection. This is checked during 3 seconds.
21	SS		The following messages are sent and shall be received on cell A. Set the cell type of cell A to the "Serving cell". Set the cell type of cell C to the "Non-Suitable cell".
22 23	UE UE		(see note) Cell A is preferred by the UE. The UE initiates an attach by MMI or by AT
24	UE		command. No ATTACH REQUEST sent to SS (SS waits 30 seconds).
25	<-	PAGING TYPE1	Mobile identity = P-TMSI-2 Paging order is for PS services.
26	UE		No response from the UE to the request. This is checked for 10 seconds.
27	SS		The following messages are sent and shall be received on cell D. Set the cell type of cell A to the "Non-Suitable cell".
28	UE		Set the cell type of cell D to the "Serving cell". (see note) Cell D is preferred by the UE. Step 28a and 29 are only performed by an UE
28a <u>condi</u> o p tional	UE	Registration on CS	which will not initiate a PS attach automatically (see ICS) See TS 34.108 This step is applied only for non-auto attach UE. Location Update Procedure initiated from the
29 <u>conditio</u>	UE		UE. The UE initiates an attach automatically (see ICS), by MMI or by AT command.
<u>nal</u> 29a	<u>SS</u>		The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Registration".
30	->	ATTACH REQUEST	Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI
<u>30a</u> 31	<u>SS</u> <-	ATTACH ACCEPT	TMSI status = no valid TMSI available <u>The SS starts integrity protection.</u> Attach result = 'Combined PS / IMSI attached' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-2
32	->	ATTACH COMPLETE	Mobile identity = IMSI
33 34	UE ->	DETACH REQUEST	The UE is switched off or power is removed (see ICS). Message not sent if power is removed. Detach type = 'power switched off, combined PS / IMSI detach'

Step	Direction	Message	Comments
	UE SS		
NOTE:	The definitions for "Non-Suitable cell", "Serving cell" and "Suitable neighbour cell" are specified		
	in TS34.10	8 clause 6.1 "Reference Radio Cond	ditions for signalling test cases only".

Specific message contents

None.

12.4.2.4.5 Test requirements

At step3, when the UE is powered up or switched on, UE shall:

- initiate the combined PS attach procedure with information elements specified in the above Expected Sequence.

At step9, when the RF level of the attached cell is lower than the RF level of the new cell, UE shall:

- - initiate the combined routing area update procedure(Update type = 'Combined RA/LA updating') with the information elements specified above Expected Sequence.

At step 10, the UE shall delete the equivalent PLMN list (MCC=1, MNC=3).

At step 12, the UE shall not initiate a PS attach procedure to cell E.

At step 18 and 24, UE shall:

- not initiate a PS attach procedure.

At step14, 20 and 26, when the UE receives the paging message for PS domain, UE shall:

- not respond to the paging message for PS domain.

At step20, when the UE receives the paging message for CS domain, UE shall:

- not respond to the paging message for CS domain.

At step30, UE shall:

- perform the PS attach procedure.

12.4.2.5a Combined routing area updating / rejected / roaming not allowed in this location area

12.4.2.5a.1 Definition

12.4.2.5a.2 Conformance requirement

- 1) If the network rejects a combined routing area updating procedure from the User Equipment with the cause 'roaming not allowed in this location area' the User Equipment:
 - 1.1 shall not perform combined PS attach when in the same location area.
 - 1.2 shall store the LA in the 'forbidden location areas for roaming'.
 - 1.3 may perform combined PS attach when a new location area is entered shall perform a routing area update when entering in a new location area if the LAI or the PLMN identity is not contained in any of the lists "forbidden LAs for roaming", "forbidden LAs for regional provision of service", "forbidden PLMNs for GPRS service" or "forbidden PLMNs" and the current update status is different from "IDLE NO IMSI".
- 2) The User Equipment shall reset the list of 'Forbidden location areas for roaming' when switched off or when the USIM is removed.

Reference

3GPP TS 24.008 clause 4.7.5.2.

<u>3GPP TS 23.122 clause 4.5.2.</u>

12.4.2.5a.3 Test purpose

Test purpose1

To test that on receipt of a rejection using the 'Roaming not allowed in this area' cause code, the UE ceases trying a routing area updating procedure on that location area. Successful combined routing area updating procedure is possible in other location areas.

Test purpose2

To test that if the UE is switched off or the USIM is removed the list of 'forbidden location areas for roaming' is cleared.

12.4.2.5a.4 Method of test

12.4.2.5a.4.1 Test procedure1

Initial condition

System Simulator:

Two cells, cell A in MCC2/MNC1/LAC1/RAC1 (RAI-2), cell B in MCC2/MNC1/LAC2/RAC1 (RAI-6). Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI.

Related ICS/IXIT statements

Support of PS serviceYes/NoUE operation mode AYes/NoSwitch off on buttonYes/NoAutomatic PS attach procedure at switch on or power onYes/No

Test procedure

The SS rejects a combined routing area updating with the cause value 'Roaming not allowed in this area'. A new attempt for a combined PS attach is not possible. Successful combined PS attachrouting area updating procedure is performed in another location area. The UE is moved back to the 1st location area. A combined routing area updating shall not be performed, as the LA is on the forbidden list.

I

Expected Sequence

Step	Direction	Message	Comments
	UE SS		The following measures are contand shall be
1	SS SS		The following messages are sent and shall be received on cell A. Set the cell type of cell A to the "Serving cell".
			Set the cell type of cell B to the "Suitable neighbour cell".
2	UE		(see note) The UE is powered up or switched on and initiates an attach (see ICS.
<u>2a</u>	<u>SS</u>		The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION
3	->	ATTACH REQUEST	REQUEST message is set to "Registration". Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI TMSI status = no valid TMSI available
3a	<-	AUTHENTICATION AND CIPHERING REQUEST	
3b	->	AUTHENTICATION AND CIPHERING RESPONSE	
3c 4	SS <-	ATTACH ACCEPT	The SS starts integrity protection. Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2 P-TMSI-2 signature
F			Routing area identity = RAI-2 Mobile identity = TMSI-1
5 <u>5a</u>	-> <u>SS</u>	ATTACH COMPLETE	The SS releases the RRC connection.
			The following messages are sent and shall be received on cell B.
7	SS		Set the cell type of cell A to the "Non-suitable cell".
			Set the cell type of cell B to the "Serving cell". (see note)
8 <u>8a</u>	UE <u>SS</u>		Cell B is preferred by the UE. <u>The SS verifies that the IE "Establishment</u> cause" in the received RRC CONNECTION
9	->	ROUTING AREA UPDATE REQUEST	REQUEST message is set to "Registration". Update type = 'Combined RA/LA updating' P-TMSI-2 signature
10	<-	ROUTING AREA UPDATE REJECT	Routing area identity = RAI-2 GMM cause = 'Roaming not allowed in this area'
<u>10a</u> 11	SS UE	Void	The SS releases the RRC connection. The UE initiates an attach by MMI or by AT
12	₩E	Void	command. No ATTACH REQUEST sont to SS
13	<-	PAGING TYPE1	(SS waits 30 seconds). Mobile identity = P-TMSI-2
14	UE		Paging order is for PS services. No response from the UE to the request. This is checked for 10 seconds.
15	<-	PAGING TYPE1	Mobile identity = TMSI-1 Paging order is for CS services.
16	UE		The UE shall not initiate an RRC connection. This is checked during 3 seconds.
17	SS		The following messages are sent and shall be received on cell A. Set the cell type of cell A to the "Serving cell". Set the cell type of cell B to the "Suitable neighbour cell".
18	UE		(see note) Cell A is preferred by the UE.

Step	Direction UE SS	Message	Comments
18a optiona I	UE	Registration on CSVoid	See TS 34.108 This step is applied only for non-auto attach UE.
19	₩E		Location Update Procedure initiated from the UE. The UE initiates an attach automatically (see
<u>19a</u>	<u>SS</u>		ICS), by MMI or by AT command. The SS verifies that the IE "Establishment
20		ATTACH REQUESTROUTING	cause" in the received RRC CONNECTION REQUEST message is set to "Registration". Attach type = 'Combined PS / IMSI attach'
	->	AREA UPDATE REQUEST	Mobile identity = P-TMSI-2 Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-2
<u>20a</u> 21	<u>SS</u> <-	ATTACH ACCEPT ROUTING AREA UPDATE ACCEPT	The SS starts integrity protection. Attach result = 'Combined PS / IMSI attached' Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-2 Mobile identity = TMSI-1
22	->	ATTACH COMPLETEROUTING AREA UPDATE COMPLETE	
<u>22a</u> 23	<u>SS</u> <-	PAGING TYPE1	The SS releases the RRC connection. Mobile identity = TMSI-1 Paging order is for CS services. Paging cause = "Terminating conversational call"
24	-> <u>SS</u>	RRC CONNECTION REQUEST	<u>The SS verifies that the IE "Establishment</u> cause" in the received RRC CONNECTION <u>REQUEST message is set to "Terminating</u> conversational call".
25 26	↔ →	VoidRRC CONNECTION SETUP VoidRRC CONNECTION SETUP COMPLETE	
27	->	PAGING RESPONSE	Mobile identity = TMSI-1
<u>27a</u> 28	<u>SS</u> ← <u>SS</u>	RRC CONNECTION RELEASE	<u>The SS starts integrity protection.</u> After sending of this message, the SS waits for disconnection of the CS signalling link <u>The SS</u> releases the RRC connection
29	→	VoidRRC CONNECTION RELEASE COMPLETE	
30	<-	PAGING TYPE1	Mobile identity = P-TMSI-1 Paging order is for PS services. Paging cause = "Terminating background call"
30a	-> <u>SS</u>	RRC CONNECTION REQUEST	The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Terminating background call".
30b 30c	↔ →	VoidRRC CONNECTION SETUP VoidRRC CONNECTION SETUP COMPLETE	
31	->	SERVICE REQUEST	service type = "paging response"
<u>31o</u> 31a 31b	<u>SS</u> <u>SS</u> ↓ ↑	RRC CONNECTION RELEASE VoidRRC CONNECTION RELEASE COMPLETE	The SS starts integrity protection. The SS releases the RRC connection.
32	SS		The following messages are sent and shall be received on cell B. Set the cell type of cell A to the "Suitable neighbour cell". Set the cell type of cell B to the "Serving cell". (see note)

Step	Direction	Message	Comments	
	UE SS			
33	UE		No ROUTING AREA UPDATE REQUEST sent to SS	
			(SS waits 30 seconds).	
34	<-	PAGING TYPE1	Mobile identity = P-TMSI-2	
			Paging order is for PS services.	
35	UE		No response from the UE to the request. This	
			is checked for 10 seconds.	
NOTE:	The definitions for "Suitable neighbour cell", "Non-suitable cell" and "Serving cell" are specified			
	in TS34.10	8 clause 6.1 "Reference Radio Cone	ditions for signalling test cases only".	

12.4.2.5a.4.2 Test procedure2

Initial condition

System Simulator:

Two cells, cell A in MCC2/MNC1/LAC1/RAC1 (RAI-2), cell B in MCC2/MNC1/LAC2/RAC1 (RAI-6). Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI. UE is Idle Updated on cell A.

Related ICS/IXIT statements

Support of PS serviceYes/NoUE operation mode AYes/NoUSIM removal possible without powering down Yes/NoSwitch off on buttonYes/NoAutomatic PS attach procedure at switch on or power onYes/No

Test procedure

The SS rejects a combined routing area updating with the cause value 'Roaming not allowed in this area'. The UE is switched off for 10 seconds and switched on again. The SS checks that a combined PS attach is possible on the cell on which the previous combined routing area updating had been rejected.

If USIM removal is possible without switching off:

The SS rejects a routing area updating with the cause value 'Roaming not allowed in this area'. The USIM is removed and inserted in the UE. The SS checks that a PS attach procedure and routing area updating procedure is possible on the cell on which the routing area updating had previously been rejected.

I

Expected Sequence

Step	Direction UE SS	Message	Comments
	SS SS		The following messages are sent and shall be
			received on cell A.
1	SS		Set the cell type of cell A to the "Serving cell".
			Set the cell type of cell B to the "Suitable
			neighbour cell". (see note)
2	UE		The UE is powered up or switched on and
			initiates an attach (see ICS.
<u>2a</u>	<u>SS</u>		The SS verifies that the IE "Establishment
			cause" in the received RRC CONNECTION
3	->	ATTACH REQUEST	REQUEST message is set to "Registration". Attach type = 'Combined PS / IMSI attach'
0	-		Mobile identity =IMSI
			TMSI status = no valid TMSI available
3a	<-	AUTHENTICATION AND	
		CIPHERING REQUEST	
3b	->		
3c	SS	CIPHERING RESPONSE	The SS starts integrity protection.
4	<-	ATTACH ACCEPT	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature
			Routing area identity = RAI-2
5	->	ATTACH COMPLETE	Mobile identity = TMSI-1
5a	SS		The SS releases the RRC connection.
			The following messages are sent and shall be
			received on cell B.
7	SS		Set the cell type of cell A to the "Suitable
			neighbour cell". Set the cell type of cell B to the "Serving cell".
			(see note)
8	UE		Cell B is preferred by the UE.
<u>8a</u>	SS		The SS verifies that the IE "Establishment
			cause" in the received RRC CONNECTION
9		ROUTING AREA UPDATE	REQUEST message is set to "Registration". Update type = 'Combined RA/LA updating'
9	->	REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-2
10	<-	ROUTING AREA UPDATE	GMM cause = 'Roaming not allowed in this
		REJECT	area'
<u>10a</u>	SS UE	Void	The SS releases the RRC connection. The UE initiates an attach by MMI or by AT
11		Void	command.
12	UE	Void	No ATTACH REQUEST sent to SS
			(SS waits 30 seconds).
13	<-	PAGING TYPE1	Mobile identity = P-TMSI-2
11			Paging order is for PS services.
14	UE		No response from the UE to the request. This is checked for 10 seconds.
15	<-	PAGING TYPE1	Mobile identity = TMSI-1
			Paging order is for CS services.
16	UE		The UE shall not initiate an RRC connection.
17	UE		This is checked during 3 seconds.
17	UE		If possible (see ICS) USIM removal is performed. Otherwise if possible (see ICS)
			switch off is performed. Otherwise the power is
			removed.
18	UE		The UE gets the USIM replaced, is powered up
			or switched on.

1

Step	Direction UE SS	Message	Comments
18a	UE	Registration on CS	See TS 34.108 This step is applied only for non-auto attach UE.
19	UE		Location Update Procedure initiated from the UE. The UE initiates an attach automatically (see ICS) by MMI or AT command.
<u>19a</u>	<u>SS</u>		The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION
20	->	ATTACH REQUEST	REQUEST message is set to "Registration". Attach type = 'Combined PS / IMSI attach' Mobile identity =IMSI TMSI status = no valid TMSI available
20a	<-	AUTHENTICATION AND CIPHERING REQUEST	
20b	->	AUTHENTICATION AND CIPHERING RESPONSE	
20c 21	SS <-	ATTACH ACCEPT ROUTING AREA UPDATE ACCEPT	The SS starts integrity protection. Attach result = 'Combined PS / IMSI attached' Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Routing area identity = RAI-6 Mobile identity = TMSI-1
22 <u>22a</u>	-> <u>SS</u>	ATTACH COMPLETE	The SS releases the RRC connection.
23	<u>-</u>	PAGING TYPE1	Mobile identity = TMSI-1 Paging order is for CS services. Paging cause = "Terminating conversational call"
24	-> <u>SS</u>	RRC CONNECTION REQUEST	The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Terminating conversational call".
25 26	- * +>	RRC CONNECTION SETUPVoid RRC CONNECTION SETUP COMPLETEVoid	
27 <u>27a</u> 28	-> <u>SS</u> - <- <u>SS</u>	PAGING RESPONSE RRC CONNECTION RELEASE	Mobile identity = TMSI-1 <u>The SS starts integrity protection.</u> After sending of this message, the SS waits for disconnection of the CS signaling link. <u>The SS</u> releases the BCC according
29	→	RRC CONNECTION RELEASE	releases the RRC connection.
30	<-	PAGING TYPE1	Mobile identity = P-TMSI-1 Paging cause = "Terminating background call"
30a	-> <u>SS</u>	RRC CONNECTION REQUEST	The SS verifies that the IE "Establishment cause" in the received RRC CONNECTION REQUEST message is set to "Terminating background call".
30b 30c	← →	RRC CONNECTION SETUPVoid RRC CONNECTION SETUP COMPLETEVoid	
31	->	SERVICE REQUEST	service type = "paging response"
<u>31o</u> 31a 31b	<u>SS</u> ≺SS →	RRC CONNECTION RELEASE RRC CONNECTION RELEASE COMPLETEVoid	<u>The SS starts integrity protection.</u> <u>The SS releases the RRC connection.</u>
32	UE		The UE is switched off or power is removed (see ICS).
33	->	DETACH REQUEST	Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach'

Step	Direc	ction	Message	Comments
	UE	SS		
NOTE:	The definitions for "Suitable neighbour cell" and "Serving cell" are specified in TS34.108 clause			
	6.1 '	Refere	ence Radio Conditions for signalling	test cases only".

Specific message contents

None.

12.4.2.5a.5 Test requirements

Test requirements for Test procedure1

At step3, when the UE is powered up or switched on, UE shall:

- initiate the combined PS attach procedure with information elements specified in the above Expected Sequence.

At step9, when the RF level of the attached cell is lower than the RF level of the new cell, UE shall:

- initiate the combined routing area update procedure(Update type = 'Combined RA/LA updating') with the information elements specified above Expected Sequence

At step12, when the SS rejects the combined routing area update procedure with GMM cause = 'Roaming not allowed in this area', UE shall:

- not initiate a PS attach procedure.

At step14, when the UE receives the paging message for PS domain, UE shall;

- not respond to the paging message for PS domain.

At step16, when the UE receives the paging message for CS domain, UE shall:

- not respond to the paging message for CS domain.

At step20, UE shall:

- initiate the combined <u>PS attachRA/LA updating</u> procedure.

At step27, when the UE receives the paging message for CS domain, UE shall;

- respond to the paging message for CS domain by sending the PAGING RESPONSE message.

At step31, when the UE receives the paging message for PS domain, UE shall:

- respond to the paging message for PS domain by sending the SERVICE REQUEST message.

At step35, when the UE receives the paging message for PS domain, UE shall;

- not respond to the paging message for PS domain.

Test requirements for Test procedure2

At step3, when the UE is powered up or switched on, UE shall:

- initiate the combined PS attach procedure with information elements specified in the above Expected Sequence.

At step9, UE shall:

- initiate the combined routing area update procedure(Update type = 'Combined RA/LA updating') with the information elements specified above Expected Sequence.

At step14, when the UE receives the paging message for PS domain, UE shall;

- not respond to the paging message for PS domain.

At step16, when the UE receives the paging message for CS domain, UE shall:

- not respond to the paging message for CS domain.

At step20, UE shall:

- initiate the combined PS attach procedure.

At step27, when the UE receives the paging message for CS domain, UE shall;

- respond to the paging message for CS domain by sending the PAGING RESPONSE message.

At step31, when the UE receives the paging message for PS domain, UE shall:

- respond to the paging message for PS domain by sending the SERVICE REQUEST message.

------<Next modification>-----

- 12.4.2.7 Combined routing area updating / abnormal cases / attempt counter check / procedure timeout
- 12.4.2.7.1 Definition

12.4.2.7.2 Conformance requirement

- 1) When a T3330 timeout has occurred during a routing area updating procedure, the UE shall repeat the routing area updating procedure after T3330 timeout until the procedure is repeated five times.
- 2) When a routing area updating procedure is repeated five times, the routing area updating attempt counter is incremented and five more routing area updating procedures are performed. This procedure is repeated until the routing area updating attempt counter is five, the UE shall then start timer T3302.
- 3) When the T3302 expire, a new routing area updating procedure shall be initiated.

Reference

3GPP TS 24.008 clause 4.7.5.2.

12.4.2.7.3 Test purpose

To test the behaviour of the UE with respect to the attempt counter.

12.4.2.7.4 Method of test

Initial condition

System Simulator:

Two cells (not simultaneously activated), cell A in MCC1/MNC1/LAC1/RAC1 (RAI-1), cell B in MCC1/MNC1/LAC1/RAC2 (RAI-4). Both cells are operating in network operation mode I.

User Equipment:

The UE has a valid IMSI. UE is Idle Updated on cell A.

Related ICS/IXIT statements

Support of PS serviceYes/NoUE operation mode AYes/NoSwitch off on buttonYes/NoAutomatic PS attach procedure at switch on or power onYes/No

Test procedure

The UE initiates a routing area updating procedure (routing area updating attempt counter zero). The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. The UE restarts the routing area updating procedure four times. The SS never answers with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. After five consecutive routing area update procedures, the routing area updating attempt counter is incremented and T3311 is started.

The UE initiates a new routing area updating procedure (routing area updating attempt counter one) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. The UE restarts the routing area updating procedure four times. The SS never answers with ROUTING AREA UPDATE

Release 5

ACCEPT message before T3330 timeout. After five consecutive routing area update procedures, the routing area updating attempt counter is incremented and T3311 is started.

The UE initiates a new routing area updating procedure (routing area updating attempt counter two) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. The UE restarts the routing area updating procedure four times. The SS never answers with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. After five consecutive routing area update procedures, the routing area updating attempt counter is incremented and T3311 is started.

The UE initiates a new routing area updating procedure (routing area updating attempt counter three) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. The UE restarts the routing area updating procedure four times. The SS never answers with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. After five consecutive routing area update procedures, the routing area updating attempt counter is incremented and T3311 is started.

The UE initiates a new routing area updating procedure (routing area updating attempt counter four) after T3311 expires. The SS does not answer with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. The UE restarts the routing area updating procedure four times. The SS never answers with ROUTING AREA UPDATE ACCEPT message before T3330 timeout. After five consecutive routing area update procedures, the routing area updating attempt counter is incremented and as the routing area updating attempt counter is five. T3302 is started.

The UE may perform a Location Update procedure.

The UE initiates a routing area updating procedure with routing area updating attempt counter zero after T3302 expires with the stored P-TMSI, P-TMSI signature, PS CKSN and RAI.

T3302; set to 12 minutes.

T3311; 15 seconds.

T3330; 15 seconds.

Expected Sequence

Step	Direction	Message	Comments
	UE SS		The following messages are sent and shall be
			received on cell A.
1	SS		Set the cell type of cell A to the "Serving cell".
			Set the cell type of cell B to the "Non-Suitable cell".
			(see note)
2	UE		The UE is powered up or switched on and
3	->	ATTACH REQUEST	initiates an attach (see ICS). Attach type = 'Combined PS / IMSI attach'
			Mobile identity =IMSI
3a	<-	AUTHENTICATION AND	TMSI status = no valid TMSI available
Ja	ζ-	CIPHERING REQUEST	
3b	->	AUTHENTICATION AND	
3c	SS	CIPHERING RESPONSE	The SS starts integrity protection.
4	<-	АТТАСН АССЕРТ	Attach result = 'Combined PS / IMSI attached'
			Mobile identity = P-TMSI-2
			P-TMSI-2 signature Routing area identity = RAI-1
			Mobile identity = IMSI
5	->	ATTACH COMPLETE	
			The following messages are sent and shall be received on cell B.
6	SS		Set the cell type of cell A to the "Non-Suitable
			cell".
			Set the cell type of cell B to the "Serving cell". (see note)
7	UE		Cell B is preferred by the UE.
0			K = 1.
8	->	ROUTING AREA UPDATE REQUEST	Update type = 'Combined RA/LA updating' P-TMSI-2 signature
			Routing area identity = RAI-1
			TMSI status = no valid TMSI available Routing area updating attempt counter = k (k is
			not visible. It is only used for clarifying the
			sequence.)
9	SS		Retransmission counter = 0 No response is given from the SS.
10	SS		The SS verifies that the time between the RA
11	->	ROUTING AREA UPDATE	update requests is T3330seconds Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature
			Routing area identity = RAI-1
			TMSI status = no valid TMSI available Routing area updating attempt counter = k
			Retransmission counter = 1
12 13	SS SS		No response is given from the SS. The SS verifies that the time between the RA
13			update requests is T3330seconds
14	->	ROUTING AREA UPDATE	Update type = 'Combined RA/LA updating'
		REQUEST	P-TMSI-2 signature Routing area identity = RAI-1
			TMSI status = no valid TMSI available
			Routing area updating attempt counter = k Retransmission counter = 2
15	SS		No response is given from the SS.
16	SS		The SS verifies that the time between the RA
I	ļ	l	update requests is T3330seconds

UE SS 17 -> ROUTING AREA UPDATE REQUEST Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available Routing area updating attempt counter = k Retransmission counter = 3 No response is given from the SS. The SS verifies that the time between the RA update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available Routing area identity = RAI-1 TMSI status = no valid TMSI available Routing area identity = RAI-1 TMSI status = no valid TMSI available Routing area updating attempt counter = k Retransmission counter = 4 No response is given from the SS. 22 SS The SS verifies that the time between the RA update requests is T3311 + T3330 seconds. Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 The UE may perform a normal location updating procedure. See TS 34.108 The SS verifies that the time between the RA update reguests is T3302 + T3330 seconds. Update type = -Combined RA/LA updating' Update type = -TMSI-1 TMSI status = no valid TMSI available Update result = Combined RA/LA updating' Update type = -TMSI-1 TMSI status = no valid TMSI available Update result = Combined RA/LA updated' Mobile identity = RAI-1 TMSI status = no valid TMSI available Update result = Combined RA/LA updated' Mobile identity = RAI-1 TMSI status = no valid TMSI available Update result = Combined RA/LA updated' Mobile identity = RAI-4 27 ->	Step	Direction	Message	Comments
REQUEST P-TMSI-2 signature Request Request 18 SS 19 SS 20 -> ROUTING AREA UPDATING No response is given from the SS. 20 -> REQUEST No response is given from the SS. 20 -> REQUEST Request is T3330seconds 20 -> ROUTING AREA UPDATING P-TMSI-2 signature Request is T3330seconds Update type = 'Combined RALA updating' PTMSI-2 signature Request is Ta330seconds 21 SS The SS verifies that the time between the RA update requests is T3311 + T330 seconds. 23 SS Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 23a UE Registration on CS The SS verifies that the time between the RA update requests is T3301 + T3300 seconds. 24 SS SS Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 25 -> ROUTING AREA UPDATE No deta type - 'Combined RALA updating' 10 It Step23a is not performed) -'combined RALA updating' 11 Combined RALA updating' If S		UE SS		
18 SS 19 SS 20 -> ROUTING AREA UPDATING REQUEST No response is given from the SS. 21 SS 22 SS 23 SS 24 SS 25 -> 26 -> 27 -> 28 -> 29 -> 20 -> 21 SS 22 SS 23 SS 24 SS 25 -> 26 -> 27 -> 28 VIE 29 -> 20 -> 26 <-	17	->		P-TMSI-2 signature
18 SS 19 SS 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 20 -> 21 SS 22 SS 23 SS 24 SS 23a UE optional I 24 SS 25 -> ROUTING AREA UPDATE REQUEST Set 53:108 26 -> 27 -> 28 UE 26 - 27 -> 28 UE 29 -> DETACH REQUEST <tr< td=""><td></td><td></td><td></td><td>TMSI status = no valid TMSI available Routing area updating attempt counter = k</td></tr<>				TMSI status = no valid TMSI available Routing area updating attempt counter = k
20 -> ROUTING AREA UPDATING REQUEST update requests is T3330seconds Update type = Combined RA/LA updating' P-TMSI-2 signature Routing area updating attempt counter = k Retransmission counter = 4 21 SS The SS verifies that the time between the RA update requests is T3311 + T3330 seconds. Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 23 SS The SS verifies that the time between the RA update requests is T3311 + T3330 seconds. Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 23a UE Registration on CS The UE may perform a normal location updating procedure. See TS 34.108 24 SS ROUTING AREA UPDATE REQUEST The SV erifies that the time between the RA update type = 'Combined RA/LA updating' Update result = 'Combined RA/LA updated' Mobie identity = P-TMSI-1 P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available Update result = 'Combined RA/LA updated' Mobie identity = P-TMSI-1 P-TMSI-1 signature Mobie identity = P-TMSI-1 P-TMSI-1 signature Mobie identity = RAI-4 27 -> ROUTING AREA UPDATE COMPLETE The UE is switched off or power is removed (see ICS). 29 -> DETACH REQUEST The UE is switched off or power is removed. Detach type = 'power switched off, combined PS/IMSI detach' </td <td>18</td> <td>SS</td> <td></td> <td></td>	18	SS		
20 -> ROUTING AREA UPDATING REQUEST Update type = 'Combined RA/LA updating' P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available Routing area updating attempt counter = k Retransmission counter = 4 21 SS The SS verifies that the time between the RA update requests is T3311 + T330 seconds. Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 23 SS The UE may perform a normal location update requests is T3302 + T3330 seconds. 24 SS The SS verifies that the time between the RA update requests is T3302 + T3330 seconds. 25 -> ROUTING AREA UPDATE REQUEST See TS 34.108 26 -> ROUTING AREA UPDATE REQUEST Update type = 'combined RA/LA updating' update requests is T3302 + T3330 seconds 26 <-	19	SS		
21 SS TMSI status = no valid TMSI available Routing area updating attempt counter = k Retransmission counter = 4 22 SS The SS verifies that the time between the RA update requests is T3311 + T3330 seconds. Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 23 UE optiona Registration on CS The UE may perform a normal location updating procedure. See TS 34.108 24 SS ROUTING AREA UPDATE REQUEST See TS 34.108 25 -> ROUTING AREA UPDATE REQUEST See TS 34.108 26 <-	20	->		Update type = 'Combined RA/LA updating' P-TMSI-2 signature
21 SS Retransmission counter = 4 22 SS No response is given from the SS. 23 SS The SS verifies that the time between the RA update requests is T3311 + T3330 seconds. Step 8 - 22 is repeated four times with k = 2, k = 3, k = 4 and k = 5 23a UE Registration on CS The UE may perform a normal location updating procedure. See TS 34.108 24 SS The SV verifies that the time between the RA update requests is T3302 + T3330 seconds 25 -> ROUTING AREA UPDATE REQUEST Update type = 'Combined RA/LA updating' Update type = 'Combined RA/LA updating' (If Step23a is not performed) - 'combined RA/LA updating' (If Step23a is not performed) P-TMSI-2 signature Routing area identity = RAI-1 26 <-				TMSI status = no valid TMSI available
22 SS 23 SS 23 SS 23a UE optiona I 1 SS 24 SS 25 -> ROUTING AREA UPDATE See TS 34.108 The SS verifies that the time between the RA updating procedure. See TS 34.108 25 -> ROUTING AREA UPDATE REQUEST -> ROUTING AREA UPDATE 26 -> ROUTING AREA UPDATE 26 -> ROUTING AREA UPDATE 27 -> ROUTING AREA UPDATE 28 UE -'combined RA/LA updating' 27 -> ROUTING AREA UPDATE 28 UE DETACH REQUEST 29 -> DETACH REQUEST 29 -> DETACH REQUEST NOTE: The definitions for "Non-Suitable cell" and "Serving cell" are specified in TS34.108 clause 6.1	04	00		Retransmission counter = 4
23 SS 23 SS 23a UE 24 SS 25 -> ROUTING AREA UPDATE See TS 34.108 REQUEST The UE may perform a normal location updating procedure. 25 -> ROUTING AREA UPDATE Update type = 'Combined RA/LA updating' Update type = 'Combined RA/LA updating' Update type = 'Combined RA/LA updating' (If Step23a is performed) - 'combined RA/LA updating' - 'Combined RA/LA updating' (If Step23a is not performed) - 'Combined RA/LA updating' -'Combined RA/LA updated' Mobile identity = Nasi Noter result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 -TMSI-1 signature Routing area identity = RAI-4 Nobile identity = RAI-4 26 <-				
23a UE Registration on CS = 3, k = 4 and k = 5 1 1 24 SS The UE may perform a normal location updating procedure. 24 SS SS The SS verifies that the time between the RA update requests is T3302 + T3330 seconds Update type = 'Combined RA/LA updating' 25 -> ROUTING AREA UPDATE Update type = 'Combined RA/LA updating' 26 -> ROUTING AREA UPDATE -'combined RA/LA updating' 26 ROUTING AREA UPDATE -'combined RA/LA updating' 26 ROUTING AREA UPDATE P-TMSI-2 signature 26 ROUTING AREA UPDATE Update result = 'Combined RA/LA updated' 27 -> ROUTING AREA UPDATE Update result = 'Combined RA/LA updated' 27 -> ROUTING AREA UPDATE The UE is switched off or power is removed (see ICS). 28 UE DETACH REQUEST The UE is switched off or power is removed. 29 -> DETACH REQUEST Message not sent if power is removed. 29 -> DETACH REQUEST Message not sent if power is removed. 29 -> DETACH REQUEST Message not sent if power is removed.	22			update requests is T3311 + T3330 seconds.
23a UE Registration on CS The UE may perform a normal location updating procedure. See TS 34.108 24 SS ROUTING AREA UPDATE See TS 34.108 25 -> ROUTING AREA UPDATE Update type = 'Combined RA/LA updating' Update type = 'Combined RA/LA updating' (If Step23a is not performed) 26 - ROUTING AREA UPDATE -'combined RA/LA updating' (If Step23a is not performed) 26 <-	23	SS		Step 8 – 22 is repeated four times with $k = 2, k$ = 3, $k = 4$ and $k = 5$
I 24 SS 25 -> ROUTING AREA UPDATE REQUEST See TS 34.108 The SS verifies that the time between the RA update requests is T3302 + T3330 seconds Update type = 'Combined RA/LA updating' Update type = 'Combined RA/LA updating' (If Step23a is not performed) - 'combined RA/LA updating' (If Step23a is not performed) P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI Routing area identity = RAI-4 26 <-		UE	Registration on CS	The UE may perform a normal location
24 SS 25 -> ROUTING AREA UPDATE REQUEST The SS verifies that the time between the RA update trequests is T3302 + T3330 seconds Update type = 'Combined RA/LA updating' Update type = ' 25 -> ROUTING AREA UPDATE REQUEST Update type = 'Combined RA/LA updating' Update type = ' 26 -> ROUTING AREA UPDATE ACCEPT 'If Step23a is not performed) - 'combined RA/LA updating' (If Step23a is not performed) P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available 26 -> ROUTING AREA UPDATE ACCEPT Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI Routing area identity = RAI-4 27 -> ROUTING AREA UPDATE COMPLETE The UE is switched off or power is removed (see ICS). 29 -> DETACH REQUEST Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach' NOTE: The definitions for "Non-Suitable cell" and "Serving cell" are specified in TS34.108 clause 6.1				
25 -> ROUTING AREA UPDATE REQUEST Update type = 'Combined RA/LA updating' Update type = 'combined RA/LA updating with IMSI attach' (If Step23a is performed) -'combined RA/LA updating' (If Step23a is not performed) P-TMSI-2 signature Routing area identity = RAI-1 TMSI status = no valid TMSI available Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature Mobile identity = IMSI Routing area identity = RAI-4 26 <-	24	SS		The SS verifies that the time between the RA
26 <-	25	->		Update type = 'Combined RA/LA updating'
26 <-				attach' (If Step23a is performed) - 'combined RA/LA updating' (If Step23a is not performed)
27 -> ROUTING AREA UPDATE COMPLETE Routing area identity = RAI-4 28 UE The UE is switched off or power is removed (see ICS). 29 -> DETACH REQUEST NOTE: The definitions for "Non-Suitable cell" and "Serving cell" are specified in TS34.108 clause 6.1	26	<-		Routing area identity = RAI-1 TMSI status = no valid TMSI available Update result = 'Combined RA/LA updated' Mobile identity = P-TMSI-1 P-TMSI-1 signature
28 UE COMPLETE The UE is switched off or power is removed (see ICS). 29 -> DETACH REQUEST Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach' NOTE: The definitions for "Non-Suitable cell" and "Serving cell" are specified in TS34.108 clause 6.1				
29 -> DETACH REQUEST (see ICS). Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach' NOTE: The definitions for "Non-Suitable cell" and "Serving cell" are specified in TS34.108 clause 6.1	27	->		
29 -> DETACH REQUEST Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach' NOTE: The definitions for "Non-Suitable cell" and "Serving cell" are specified in TS34.108 clause 6.1	28	UE		
	29			Message not sent if power is removed. Detach type = 'power switched off, combined PS/IMSI detach'
Nordenice Naulo Conditions for signalling test cases Unly .	NOTE:			ving cell" are specified in TS34.108 clause 6.1

Specific message contents

None.

12.4.2.7.5 Test requirements

At step3, when the UE is powered up or switched on, UE shall:

- initiate the combined PS attach procedure with information elements specified in the above Expected Sequence.

At step8, when the RF level of the attached cell is lower than the RF level of the new cell, UE shall:

- initiate the combined routing area updating procedure with information elements specified in the above Expected Sequence.

UE shall perform the following actions depending on the conditions described below.

Case 1) A timer T3330 timeout has occurred during a combined routing area updating procedure with the Routing area attempt counter less than five and the Retransmission counter less than five

At step11, 14, 17 and 20, UE shall:

- repeat the combined routing area updating procedure after the timer T3330 timeout

Case2) A timer T3330 timeout has occurred during a combined routing area updating procedure with the Routing area attempt counter less than five and the Retransmission counter five

At step 22, UE shall:

- not repeat the combined routing area updating procedure.

Case 3) A timer T3311 timeout has occurred and the Routing area attempt counter is less than five,

At step23, UE shall:

- repeat the combined routing area updating procedure

Case 4) A timer T3330 timeout has occurred during a combined routing area updating procedure with the Routing area attempt counter five and the Retransmission counter five.

At step24, UE shall:

- not initiate a routing area updating procedure.

Case5) The timer T3302 expires

At step25, UE shall:

- initiate the new routing area updating procedure

-----<End of modification>-----

3GPP TSG- T1 Meeting #17 Luton, UK, 4th – 8th November 2002

T1-020825

CR-Form-v6.1					R-Form-v6.1		
¥		CR 370 User Equipment (Part 1: Protocol ce		nance spec		^{ion:} 5.1.1	H H
For <u>HELP</u> of	n using this fc	orm, see bottom of t	his page or	look at the	pop-up text	over the X sym	nbols.
Proposed chang	ge affects: ¥	(U)SIM	ME/UE X	Radio Acc	ess Network	Core Net	twork
Title:	# Correction Release,	ns to 8.1.2 RRC Co TDD tests	nnection Est	tablishmen	t and 8.1.3 F	RRC Connection	n
Source:	ж Siemens	AG					
Work item code	: ೫ TEI				Date: ೫	28/10/2002	
Category:	F (co A (co B (ao C (fui D (co Detailed ex	the following categor rrection) rresponds to a correc dition of feature), nctional modification of litorial modification) splanations of the abo 3GPP <u>TR 21.900</u> .	ction in an ear of feature)	lier release)	2 R96 R97 R98 R99 REL-4	REL-5 the following relea (GSM Phase 2) (Release 1996) (Release 1997) (Release 1998) (Release 1999) (Release 4) (Release 5)	ases:
Reason for char	Reason for change: # Test cases are reviewed according to FDD changes. Some missing information is also added and some is corrected.					ation is	
Summary of cha	- For 8 - - For 8 - - -	 According to T15 According to T16 According to T17 	TION REQUI specification TION SETUR TION REQUE TION SETUR S-020730, SI f: Table 8.1.2. ext for IE "PI ION REQUE	EST (Step (TDD) EST (Step 2) (Step 3) (IB 11 adde 11 rimary CCF EST (Step 3	1) (TDD) 2) (TDD) TDD) d. PCH RSCP"		

Release 5

	- RRC CONNECTION SETUP (Step 4) (TDD)	
	For 8.1.3.6:	
	Values for TDD included in Table 8.1.3.6	
	MEASUREMENT CONTROL (Step 3) is not used according to T1S-020730	
	SIB type 11 included (TDD)	
	For 8.1.3.7:	
	Values for TDD included in Table 8.1.3.7	
Consequences if % not approved:	Cell configuration in each test case will not be consistent.	
0 1 1 0		
Clauses affected: #	8.1.2.1, 8.1.2.7, 8.1.2.10, 8.1.2.11, 8.1.3.6, 8.1.3.7	
Other specs भ affected:	Other core specifications # Test specifications # O&M Specifications •	
Other commontor	Affects BOO DEL 4 DEL 5	
Other comments: #	 Affects R99, REL-4, REL-5 T1S-020631 for FDD mode considered to be compatible (This document was provided to the reflector for email approval). 	

8.1.2.1 RRC Connection Establishment in CELL_DCH state: Success

8.1.2.1.1 Definition

8.1.2.1.2 Conformance requirement

The UE shall initiate the procedure when upper layers in the UE requests the establishment of a signalling connection and the UE is in idle mode (no RRC connection exists).

Upon initiation of the procedure, the UE shall:

•••

1> set the contents of the RRC CONNECTION REQUEST message according to TS 25.331 subclause 8.1.3.3;

- 1> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;
- 1> perform the mapping of the Access Class to an Access Service Class as specified in TS 25.331 subclause 8.5.13, and apply the given Access Service Class when accessing the RACH;
- 1> submit the RRC CONNECTION REQUEST message for transmission on the uplink CCCH;
- 1> set counter V300 to 1; and
- 1> start timer T300 when the MAC layer indicates success or failure to transmit the message;
- 1> select a Secondary CCPCH according to TS 25.304;
- 1> start receiving all FACH transport channels mapped on the selected Secondary CCPCH.

The UE shall, in the transmitted RRC CONNECTION REQUEST message:

- 1> set the IE "Establishment cause" to the value of the variable ESTABLISHMENT_CAUSE;
- 1> set the IE "Initial UE identity" to the value of the variable INITIAL_UE_IDENTITY;

...

The UE shall compare the value of the IE "Initial UE identity" in the received RRC CONNECTION SETUP message with the value of the variable INITIAL_UE_IDENTITY.

If the values are different, the UE shall:

1> ignore the rest of the message.

If the values are identical, the UE shall:

1> stop timer T300, and act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following:

2> if the UE will be in the CELL_FACH state at the conclusion of this procedure:

.

- 1> perform the physical layer synchronization procedure as specified in TS 25.214;
- 1> enter UTRA RRC connected mode, in a state according to TS 25.331 subclause 8.6.3.3;
- 1> submit an RRC CONNECTION SETUP COMPLETE message to the lower layers on the uplink DCCH after successful state transition per TS 25.331 subclause 8.6.3.3, with the contents set as specified below:

2> set the IE "RRC transaction identifier" to:

3> the value of "RRC transaction identifier" in the entry for the RRC CONNECTION SETUP message in the table "Accepted transactions" in the variable TRANSACTIONS; and

3> clear that entry.

. . .

- 2> retrieve its UTRA UE radio access capability information elements from variable UE_CAPABILITY_REQUESTED; and then
- 2> include this in IE "UE radio access capability" and IE "UE radio access capability extension", provided this IE is included in variable UE_CAPABILITY_REQUESTED;
- 2> retrieve its inter-RAT-specific UE radio access capability information elements from variable UE_CAPABILITY_REQUESTED; and then
- 2> include this in IE "UE system specific capability".

When the RRC CONNECTION SETUP COMPLETE message has been submitted to lower layers for transmission the UE shall:

1> consider the procedure to be successful;

And the procedure ends.

Reference

3GPP TS 25.331 clause 8.1.3.

8.1.2.1.3 Test purpose

- 1. To confirm that the UE leaves the Idle Mode and correctly establishes signalling radio bearers on the DCCH.
- 2. To confirm that the UE indicates the requested UE radio access capabilities and UE system specific capabilities (may be used by UTRAN e.g. to configure inter RAT- measurements).

8.1.2.1.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE.

Test Procedure

The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by attempting to make an outgoing call. After SS receives this message, it assigns the necessary radio resources and U-RNTI to be used by the UE. SS then transmits an RRC CONNECTION SETUP message containing an IE "Initial UE Identity" that does not match the IE "Initial UE Identity" in the most recent RRC CONNECTION REQUEST message sent by the UE. UE receives the RRC CONNECTION SETUP message before timer T300 expires but discards it due to a IE "Initial UE Identity" mismatch. UE shall wait for timer T300 to time out before re-transmitting a RRC CONNECTION REQUEST message to the SS. SS again assigns the necessary radio resources and U-RNTI. SS then transmits a RRC CONNECTION REQUEST message containing an IE "Initial UE Identity" that matches the IE "Initial UE Identity" in the most recent RRC CONNECTION REQUEST sent by the UE. SS then waits for the UE to transmit an RRC CONNECTION REQUEST sent by the UE. SS then waits for the UE to transmit an RRC CONNECTION SETUP COMPLETE message on the DCCH. SS calls for generic procedure C.3 to check that UE is in CELL_DCH state.

Release 5

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1	\rightarrow	RRC CONNECTION REQUEST	By outgoing call operation. See specific message contents.
2	÷	RRC CONNECTION SETUP	This message is not addressed to the UE. See specific message contents.
3	\rightarrow	RRC CONNECTION REQUEST	UE shall re-transmit the request message again after a time out of T300 from step 1.
4	÷	RRC CONNECTION SETUP	See specific message contents.
5			The UE configures the layer 2 and layer 1.
6	\rightarrow	RRC CONNECTION SETUP COMPLETE	See specific message contents.
7	\leftrightarrow	CALL C.3	If the test result of C.3 indicates that UE is in CELL_DCH state, the test passes, otherwise it fails.

Specific Message Content

System Information Block type 11 (FDD)

Use the default system information block with the same type specified in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
- Intra-frequency reporting quantity for RACH Reporting	
 SFN-SFN observed time difference reporting 	No report
indicator	
- CHOICE mode	
- FDD	
- Reporting quantity	CPICH Ec/N0
 Maximum number of reported cells on RACH 	current cell

System Information Block type 11 (TDD)

Use the default system information block with the same type specified in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
- Intra-frequency reporting quantity for RACH Reporting	
 SFN-SFN observed time difference reporting 	No report
indicator	
- CHOICE mode	TDD
- Reporting quantity list	
- Reporting quantity	P-CCPCH RSCP
- Maximum number of reported cells on RACH	current cell

RRC CONNECTION REQUEST (Step 1) (FDD)

Use the default message with the same message type specified in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
Measured results on RACH	Check to see if set in accordance with the IE "Intra-
	frequency reporting quantity for RACH Reporting"
	included in SYSTEM INFORMATION BLOCK Type 11
- Measurement result for current cell	
- CHOICE mode	
- FDD	
- CHOICE measurement quantity	
- CPICH Ec/N0	The actual reported value is not checked

RRC CONNECTION REQUEST (Step 1) (TDD)

Use the default message with the same message type specified in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
Measured results on RACH	Check to see if set in accordance with the IE "Intra-
	frequency reporting quantity for RACH Reporting"
	included in SYSTEM INFORMATION BLOCK Type 11
 Measurement result for current cell 	
- CHOICE mode	TDD
 <u>- CHOICE measurement quantity</u> 	
- P-CCPCH RSCP	The actual reported value is not checked

RRC CONNECTION SETUP (Step 2)

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark
Initial UE Identity	Set to the same type as in the RRC CONNECTION REQUEST message but with a different value

RRC CONNECTION SETUP (Step 4)

Use the default message with the same message type and covering the scenario used in this test (Transition to CELL_DCH) specified in clause 9 of TS 34.108.

RRC CONNECTION SETUP COMPLETE (Step 6)

Use the default message with the same message type specified in clause 9 of TS 34.108 with the following exception.

Information Element	Value/remark
UE Radio Access Capability	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.
UE radio access capability extension	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.
UE system specific Capability	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.

8.1.2.1.5 Test requirement

After step 2 the UE shall re-transmit the RRC CONNECTION REQUEST message again in order to continue the RRC connection establishment procedure.

After step 6 the UE shall establish an RRC connection and continue the procedure of the outgoing call on the DCCH.

<Next change>

- 8.1.2.7 RRC Connection Establishment in CELL_FACH state: Success
- 8.1.2.7.1 Definition

8.1.2.7.2 Conformance requirement

The UE shall initiate the procedure when upper layers in the UE requests the establishment of a signalling connection and the UE is in idle mode (no RRC connection exists).

Upon initiation of the procedure, the UE shall:

•••

- 1> set the contents of the RRC CONNECTION REQUEST message according to TS 25.331 subclause 8.1.3.3;
- 1> set CFN in relation to SFN of current cell according to TS 25.331 subclause 8.5.15;
- 1> perform the mapping of the Access Class to an Access Service Class as specified in TS 25.331 subclause 8.5.13, and apply the given Access Service Class when accessing the RACH;
- 1> submit the RRC CONNECTION REQUEST message for transmission on the uplink CCCH;
- 1> set counter V300 to 1; and
- 1> start timer T300 when the MAC layer indicates success or failure to transmit the message;
- 1> select a Secondary CCPCH according to TS 25.304;
- 1> start receiving all FACH transport channels mapped on the selected Secondary CCPCH.

The UE shall, in the transmitted RRC CONNECTION REQUEST message:

- 1> set the IE "Establishment cause" to the value of the variable ESTABLISHMENT_CAUSE;
- 1> set the IE "Initial UE identity" to the value of the variable INITIAL_UE_IDENTITY;

•••

The UE shall compare the value of the IE "Initial UE identity" in the received RRC CONNECTION SETUP message with the value of the variable INITIAL_UE_IDENTITY.

If the values are different, the UE shall:

1> ignore the rest of the message.

If the values are identical, the UE shall:

- 1> stop timer T300, and act upon all received information elements as specified in TS 25.331 subclause 8.6, unless specified otherwise in the following:
 - 2> if the UE will be in the CELL_FACH state at the conclusion of this procedure:
 - 3> if the IE "Frequency info" is included:
 - 4> select a suitable UTRA cell according to TS 25.304 on that frequency;
 - 3> select PRACH according to TS 25.331 subclause 8.5.17;
 - 3> select Secondary CCPCH according to TS 25.331 subclause 8.5.19;
 - 3> ignore the IE "UTRAN DRX cycle length coefficient" and stop using DRX.
- 1> perform the physical layer synchronization procedure as specified in TS 25.214 (FDD) or TS 25.224 (TDD);

- 1> enter UTRA RRC connected mode, in a state according to TS 25.331 subclause 8.6.3.3;
- 1> submit an RRC CONNECTION SETUP COMPLETE message to the lower layers on the uplink DCCH after successful state transition per TS 25.331 subclause 8.6.3.3, with the contents set as specified below:
 - 2> set the IE "RRC transaction identifier" to:
 - 3> the value of "RRC transaction identifier" in the entry for the RRC CONNECTION SETUP message in the table "Accepted transactions" in the variable TRANSACTIONS; and
 - 3> clear that entry.
 - 2> retrieve its UTRA UE radio access capability information elements from variable UE_CAPABILITY_REQUESTED; and then
 - 2> include this in IE "UE radio access capability" and IE "UE radio access capability extension", provided this IE is included in variable UE_CAPABILITY_REQUESTED;
 - 2> retrieve its inter-RAT-specific UE radio access capability information elements from variable UE_CAPABILITY_REQUESTED; and then
 - 2> include this in IE "UE system specific capability".

When the RRC CONNECTION SETUP COMPLETE message has been submitted to lower layers for transmission the UE shall:

- 1> if the UE has entered CELL_FACH state:
 - 2> start timer T305 using its initial value if periodical update has been configured by T305 in the IE "UE Timers and constants in connected mode" set to any other value than "infinity" in system information block type 1.
- 1> consider the procedure to be successful;

And the procedure ends.

Reference

3GPP TS 25.331 clause 8.1.3.

8.1.2.7.3 Test Purpose

- 1. To confirm that the UE is able to enter CELL_FACH state and setup signalling radio bearers using common physical channels.
- 2. To confirm that the UE indicates the requested UE radio access capabilities (used by UTRAN to decide which RAB to establish) and UE system specific capabilities (may be used by UTRAN to configure inter RAT-measurements).

8.1.2.7.4 Method of test

Initial Condition

System Simulator: 1 cell.

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE.

Test Procedure

The UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by attempting to make an outgoing call. After the SS receives this message, it assigns the necessary radio resources and U-RNTI to be used by the UE, and then transmits an RRC CONNECTION SETUP message to the UE within timer T300. SS then waits for the UE to transmit an RRC CONNECTION SETUP COMPLETE message on the DCCH. SS calls for generic procedure C.2 to check that UE is in CELL_FACH state.

Release 5

Expected sequence

Step	Step Direction		Message	Comment
	UE	SS	_	
1	\rightarrow		RRC CONNECTION REQUEST	Test operator is requested to make an outgoing call. The UE shall transmit this message, indicating the correct establishment cause. See specific message contents.
2	←		RRC CONNECTION SETUP	See specific message contents.
3				The UE shall configure the layer 2 and layer 1.
4	\rightarrow		RRC CONNECTION SETUP COMPLETE	UE shall send this message on the DCCH, carried by the assigned PRACH resources. See specific message contents. FDD or TDD
5	\leftrightarrow	>	CALL C.2	If the test result of C.2 indicates that UE is in CELL_FACH state, the test passes, otherwise it fails.

Specific Message Content

RRC CONNECTION REQUEST

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark
Initial UE identity	Same as the IMSI stored in the TEST USIM card, or the
	registered TMSI or P-TMSI
Establishment Cause	Originating Interactive Call or Originating Background
	Call or Originating Streaming Call

RRC CONNECTION SETUP (FDD)

For this message, the contents of the message to be used are basically identical to the message sub-type entitled "RRC CONNECTION SETUP message (Transition to CELL_FACH)" found in TS 34.108, clause 9 with the following exception:

Information Element	Value/remark
Capability update requirement	
UE radio access FDD capability update requirement	TRUE
UE radio access TDD capability update requirement	FALSE
System specific capability update requirement list	gsm

RRC CONNECTION SETUP (TDD)

For this message, the contents of the message to be used are basically identical to the message sub-type entitled "RRC CONNECTION SETUP message (Transition to CELL_FACH)" found in TS 34.108, clause 9 with the following exception:

Information Element	Value/remark
Capability update requirement	
UE radio access FDD capability update requirement	FALSE
UE radio access TDD capability update requirement	TRUE
System specific capability update requirement list	<u>gsm</u>

RRC CONNECTION SETUP COMPLETE

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark
UE Radio Access Capability	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.
UE radio access capability extension	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.
UE system specific Capability	Checked to see if compatible with the stated capability
	in PIXIT/PICS statements.

8.1.2.7.5 Test requirements

After step 3 the UE shall establish the RRC connection, and transmit RRC CONNECTION SETUP COMPLETE message on the DCCH using PRACH physical resource specified in system information block messages.

<Next change>

- 8.1.2.10 RRC connection establishment in CELL_DCH on another frequency
- 8.1.2.10.1 Definition

8.1.2.10.2 Conformance requirement

- 1. The UE shall, in the transmitted RRC CONNECTION REQUEST message:
 - set the IE "Establishment cause" to the value of the variable ESTABLISHMENT_CAUSE;
 - set the IE "Initial UE identity" to the value of the variable INITIAL_UE_IDENTITY;
 - set the IE "Protocol error indicator" to the value of the variable PROTOCOL_ERROR_INDICATOR;
 - include a measurement report in the IE "Measured results on RACH", as specified in the IE "Intra-frequency reporting quantity for RACH reporting" and the IE "Maximum number of reported cells on RACH" in System Information Block type 11.
- 2. The UE shall compare the value of the IE "Initial UE identity" in the received RRC CONNECTION SETUP message with the value of the variable INITIAL_UE_IDENTITY.

If the values are identical, the UE shall:

- perform the physical layer synchronization procedure

Reference

3GPP TS 25.331 clauses 8.3.1.3, 8.3.1.6

8.1.2.10.3 Test Purpose

To confirm that the UE manages to synchronize on another frequency when so required by UTRAN in the RRC CONNECTION SET UP message.

8.1.2.10.4 Method of test

Initial condition

System simulator: 2 cells – Cell 1 on UARFCN 1 and Cell 2 on UARFCN 2.

UE: "Registered idle mode on CS" (state 2) or "Registered idle mode on PS" (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE. If the UE supports both CS and PS domains, the initial state shall be "Registered idle mode on CS/PS" (state 7).

System Information Block type 11 (TDD)

Use same message sub-clause 6.1 of TS34.108, with following exception:

Information Element	Value/remark
- SIB12 indicator	FALSE
- Intra-frequency measurement system information	Not Present
- Inter-frequency measurement system information	
 Inter-frequency cell info list 	
 New inter-frequency cell id 	
- Inter frequency cell id	<u>4</u>
- Frequency info	
- CHOICE mode	
- UARFCN (Nt)	Reference to table 6.1 of TS34.108 for Cell 4
- Cell info	Not Present
 Cell individual offset Reference time difference to cell 	Not present
- Read SFN indicator	FALSE
- CHOICE mode	TDD
- Primary CCPCH info	Reference clause 6.1,TS34.108,Default settings for cell 4
- Cell parameters ID	Reference clause 6.1,TS34.108,Default settings for cell 4
- Primary CPICH TX power	Not present
- Timeslot list	Not present
 Cell Selection and Re-selection Info 	Not present
	For neigbouring cell, if HCS is not used and all the
	parameters in cell selection and re-selection info are
- Qoffset1 _{s.n}	Default value, this IE is absent. 0dB
- Maximum allowed UL TX power	Reference to table 6.1.6, TS 34.108
- HCS neighbouring cell information	Not present
- CHOICE mode	TDD
- Qrxlevmin	Reference to table 6.1.6, TS 34.108
- Cells for measurement	Not present

Test procedure

The UE is initially in idle mode and is camping on cell 1. SIB 11 is broadcast in cell 1, and the parameters used are as specified below.

SS prompts the operator to make an outgoing call of a supported traffic class. The UE shall transmit an RRC CONNECTION REQUEST on the CCCH, and SS replies with the RRC CONNECTION SETUP, in which the IEs are set as described below. The UE shall send the RRC CONNECTION SETUP COMPLETE back to SS in cell 2 on the DPCH described in the RRC CONNECTION SET UP messsage received from the SS. SS calls for generic procedure C.3 to check that UE is in CELL_DCH state.

Release 5

Expected sequence

Step	Direction		Message	Comment
	UE SS			
1	\rightarrow		RRC CONNECTION REQUEST	By outgoing call operation
2	÷		RRC CONNECTION SETUP	
3				The UE configures the layer 2 and layer 1.
4	<i>→</i>		RRC CONNECTION SETUP COMPLETE	This message is sent to on the frequency indicated in the RRC CONNECTION SETUP message
5	\leftrightarrow		CALL C.3	If the test result of C.3 indicates that UE is in CELL_DCH state, the test passes, otherwise it fails.

Specific message content

All messages indicated below shall use the same content as described in the default message content, with the following exceptions:

System Information Block type 11

Information Element	Value/Remark
SIB12 indicator	FALSE
FACH measurement occasion info	Not Present
Measurement control system information	
 Intra-frequency measurement system 	
information	
 Intra-frequency measurement identity 	1
 Intra-frequency cell info list 	
 CHOICE intra-frequency cell removal 	Remove no intra-frequency cells
 New intra-frequency info list 	
- Intra-frequency cell id	1
- Cell info	
- Cell individual offset	0 dB
 Reference time difference to cell 	256 chips
- Read SFN Indicator	FALSE
- CHOICE Mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	Set to same code as used for cell 1
- Primary CPICH TX power	Not Present
- TX Diversity Indicator	FALSE
- Cell selection and Re-selection	Not present
- Cell for measurement	Not present
 Intra-frequency measurement quantity 	Not present
- Intra-frequency measurement for RACH	
reporting	
SFN-SFN observed time difference	No reporCPICH Ec/No
- Reporting quantity	Current Cell
- Maximum number of reported cells on RACH	Not present
- Reporting information for state CELL_DCH	- -

RRC CONNECTION REQUEST (Step 2) (FDD)

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark	
Measured results on RACH	Check that the Ec/No for the cell 1 is reported.	

RRC CONNECTION REQUEST (Step 2) (TDD)

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Value/remark
the P-CCPCH RSCP for the cell 1 is
t

RRC CONNECTION SETUP (Step 3) (FDD)

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	UARFCN uplink of cell 2
- UARFCN downlink(Nd)	UARFCN downlink of cell 2

RRC CONNECTION SETUP (Step 3) (TDD)

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark
Frequency info	
- UARFCN(Nt)	UARFCN of the cell 4

8.1.2.10.5 Test requirement

In step 4, the UE shall send the RRC CONNECTION SETUP COMPLETE message on the frequency indicated in the RRC CONNECTION SETUP message.

<Next change>

- 8.1.2.11 RRC Connection Establishment in FACH state (Frequency band modification): Success
- 8.1.2.11.1 Definition
- 8.1.2.11.2 Conformance requirement

The UE shall compare the value of the IE "Initial UE identity" in the received RRC CONNECTION SETUP message with the value of the UE storing "Initial UE identity".

If the values are different, the UE shall:

1> ignore the rest of the message.

If the values are identical, the UE shall:

1> act upon all received information elements as specified in TS25.331 subclause 8.6, unless specified otherwise in the following:

2> if the UE will be in the CELL_FACH state at the conclusion of this procedure:

- 3> if the IE "Frequency info" is included:
 - 4> select a suitable UTRA cell according to TS25.304 on that frequency;
- 3> select PRACH according to TS25.331 subclause 8.5.17;
- 3> select Secondary CCPCH according to TS5.331 subclause 8.5.19;
- 3> ignore the IE "UTRAN DRX cycle length coefficient" and stop using DRX.
- 1> enter UTRA RRC connected mode, in a state according to TS25.331 subclause 8.6.3.3;
- 1> submit an RRC CONNECTION SETUP COMPLETE message to the lower layers on the uplink DCCH after successful state transition per TS25.331 subclause 8.6.3.3:

And the procedure ends.

Reference

3GPP TS 25.331 clause 8.1.3.

8.1.2.11.3 Test purpose

To confirm that the UE enters to CELL_FACH state and correctly establishes signalling radio bearers using common physical channels of a cell within the frequency band specified by SS in RRC CONNECTION SETUP message.

8.1.2.11.4 Method of test

Initial Condition

System Simulator: 2 cells–Cells 1 is active and cell 6 is inactive.

UE: Idle state (state 2 or state 3 or state 7) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE.

System Information Block type 11 (TDD)

Use same message sub-clause 6.1 of TS34.108, with following exception:

Information Element	Value/remark
- Intra-frequency reporting quantity for RACH	
Reporting	
-SFN-SFN observed time difference reporting	No report
indicator	
- CHOICE mode	TDD
- CHOICE measurement quantity	PCCPCH RSCP
- Maximum number of reported cells on RACH	current cell

Test Procedure

Parameter	Unit	Cell 1		Cell 6	
		T0	T1	T0	T1
UTRA RF Channel Number		Ch. 1		Ch. 2	
CPICH Ec (FDD)	dBm/ 3.84 MHz	-55	-72	Off	-72
P-CCPCH RSCP (TDD)	<u>dBm</u>	<u>-55</u>	<u>-72</u>	<u>Off</u>	<u>-72</u>

Table 8.1.2.11

Table 8.1.2.11 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. SS switches the power settings from columns "T0" to "T1", whenever the description in multi-cell condition specifies the transmission power settings for cell 1 and cell 6.

The UE is in Idle mode of cell 1 and the SS has configured its downlink transmission power setting according to columns "T0" in table 8.1.2.11. The SS switches its downlink transmission power settings to columns "T1" and the UE transmits an RRC CONNECTION REQUEST message to the SS on the uplink CCCH by attempting to make an outgoing call. After SS receives this message, it assigns the necessary radio resources and U-RNTI to be used by the UE. The SS then transmits an RRC CONNECTION SETUP message containing an IE "frequency info" IE "Frequency info" set to uplink/downlink UARFCN as used for cell 6 and IE "Primary CPICH info" set to Primary Scrambling Code assigned to P-CPICH of cell 6 for FDD mode, and UARFCN as used for cell 6 and IE "Primary CCPCH RSCP info" set as assigned in cell 6 for TDD mode. The SS monitors all uplink RACH channels of cell 6. The UE transmitting an RRC CONNECTION SETUP COMPLETE message on the DCCH (mapped onto RACH) of cell 6.

Expected sequence

Step	Dire	ction	Message	Comment
-	UE	SS		
1				The initial state of UE is in Idle mode of cell 1 and the SS has configured its downlink transmission power setting according to columns "T0" in table 8.1.2.11.
2				The SS switches its downlink transmission power settings to columns "T1" in table 8.1.2.11.
3	-	>	RRC CONNECTION REQUEST	Operator makes an outgoing call. The UE shall transmit this message, indicating the proper establishment cause.
4	•	÷	RRC CONNECTION SETUP	Including IE "Frequency info" set to frequency information of cell 6 and IE "Primary CPICH info" set to Primary Scrambling Code assigned to P-CPICH of cell 6 for FDD mode, and IE "Primary CCPCH RSCP info" set as assigned in cell 6 for TDD mode.
5	-	>	RRC CONNECTION SETUP COMPLETE	UE shall send this message on the DCCH, carried by the assigned PRACH resources in cell 6.

Specific Message Content

RRC CONNECTION REQUEST (Step 3) (FDD)

Use the same message type found in clause 9 of TS 34.108, with the following exception:

Information Element	Value/remark
Initial UE identity	Same as the IMSI stored in the TEST USIM card, or the
	registered TMSI or P-TMSI
Establishment Cause	Originating Interactive Call or Originating Background
	Call or Originating Streaming Call
Measured results on RACH	Check to see if set in accordance with the IE "Intra-
	frequency reporting quantity for RACH Reporting"
	included in SYSTEM INFORMATION BLOCK Type 511
- Measurement result for current cell	
- CHOICE mode	
- FDD	
 CHOICE measurement quantity 	
- CPICH Ec/N0	The actual reported value is not checked

RRC CONNECTION REQUEST (Step 3) (TDD)

Use the same message type found in clause 9 of TS 34.108, with the following exception:

Information Element	Value/remark
Initial UE identity	Same as the IMSI stored in the TEST USIM card, or the
	registered TMSI or P-TMSI
Establishment Cause	Originating Interactive Call or Originating Background
	Call or Originating Streaming Call
Measured results on RACH	Check to see if set in accordance with the IE "Intra-
	frequency reporting quantity for RACH Reporting"
	included in SYSTEM INFORMATION BLOCK Type 11
- Measurement result for current cell	
- CHOICE mode	TDD
 <u>- CHOICE measurement quantity</u> 	
- P-CCPCH RSCP	The actual reported value is not checked

RRC CONNECTION SETUP (Step 4) (FDD)

For this message, the contents of the message to be used are basically identical to the message sub-type entitled "RRC CONNECTION SETUP message (Transition to CELL_FACH)" found in [9] TS 34.108 clause 9 with the following exception:

Information Element	Value/remark
Frequency info	
- UARFCN uplink(Nu)	Same uplink UARFCN as used for cell 6
- UARFCN downlink(Nd)	Same downlink UARFCN as used for cell 6
Downlink information for each radio links	
- Primary CPICH info	
- Primary Scrambling Code	Set to same code as used for cell 6

RRC CONNECTION SETUP (Step 4) (TDD)

For this message, the contents of the message to be used are basically identical to the message sub-type entitled "RRC CONNECTION SETUP message (Transition to CELL_FACH)" found in [9] TS 34.108 clause 9 with the following exception:

Information Element	Value/remark
Frequency info	
- UARFCN (Nt)	Same UARFCN as used for cell 6
Downlink information for each radio links	
- Primary CCPCH info	
- Cell parameters ID	As used for cell 6

RRC CONNECTION SETUP COMPLETE (Step 5)

Use the same message type found in clause 9 of TS 34.108, with the following exception.

Information Element	Value/remark
UE Radio Access Capability	Checked to see if compatible with the stated capability in
	PIXIT/PICS statements.
UE radio access capability extension	Checked to see if compatible with the stated capability in
	PIXIT/PICS statements.
UE system specific Capability	Checked to see if compatible with the stated capability in
	PIXIT/PICS statements.

8.1.2.11.5 Test requirement

After step 5 the UE shall transmit RRC CONNECTION SETUP COMPLETE message on the uplink DCCH in cell 6.

<Next change>

- 8.1.3.6 RRC Connection Release in CELL_DCH state (Frequency band modification): Success
- 8.1.3.6.1 Definition
- 8.1.3.6.2 Conformance requirement

If the UE first receives an RRC CONNECTION RELEASE message in CELL_DCH state, it shall:

- initialize the counter V308 to zero;
- submit an RRC CONNECTION RELEASE COMPLETE message to the lower layers for transmission using UM RLC on the DCCH to the UTRAN;
- start timer T308 when the RRC CONNECTION RELEASE COMPLETE message is sent on the radio interface.

If the timer T308 expires, the UE shall:

- increment V308 by one;
- if V308 is equal to or smaller than N308:
 - retransmit the RRC CONNECTION RELEASE COMPLETE message;
- if V308 is greater than N308:
 - release all its radio resources;
 - enter idle mode;
 - perform cell-selection according to TS25.304;
 - procedure end;

Reference

3GPP TS 25.331 clause 8.1.4.

8.1.3.6.3 Test purpose

To confirm that when the UE receives an RRC CONNECTION RELEASE message the UE transmits N308+1 RRC CONNECTION RELEASE COMPLETE messages using UM on DCCH.

Release 5

To confirm that the UE enters into idle mode with performing cell-selection and selecting new cell configured by SS.

8.1.3.6.4 Method of test

Initial Condition

System Simulator: 2 cells-Cell 1 is active and cell 6 is inactive

UE: CS-DCCH+DTCH_DCH (state 6-9) or PS-DCCH+DTCH_DCH (state 6-10) as specified in clause 7.4 of TS 34.108, depending on the CN domain(s) supported by the UE

System Information Block type 11 (TDD)

Use same message sub-clause 6.1 of TS34.108, with following exception:

Information Element	Value/remark
 <u>- SIB12 indicator</u> <u>- Intra-frequency measurement system information</u> 	FALSE Not Present
 Inter-frequency measurement system information Inter-frequency cell info list 	
 <u>- New inter-frequency cell id</u> <u>- Inter frequency cell id</u> 	4
<u>- Frequency info</u> <u>- CHOICE mode</u> <u>- UARFCN (Nt)</u> - Cell info	TDD Reference to TS34.108 for Cell 4
- Cell individual offset - Reference time difference to cell - Read SFN indicator	Not Present Not present FALSE
- CHOICE mode - Primary CCPCH info	TDD Reference clause 6.1,TS34.108,Default settings for cell 4
- Cell parameters ID	Reference clause 6.1,TS34.108,Default settings for cell 4
- Primary CPICH TX power - Timeslot list	Not present Not present
<u>- Cell Selection and Re-selection Info</u>	Not present For neigbouring cell, if HCS is not used and all the parameters in cell selection and re-selection info are Default value, this IE is absent. OdB
 <u>Advisoring n</u> <u>- Maximum allowed UL TX power</u> <u>- HCS neighbouring cell information</u> 	Reference to table 6.1.6, TS34.108 Not present
- CHOICE mode - Qrxlevmin	TDD Reference to table 6.1.6, TS 34.108
<u>- Cells for measurement</u>	Not present

Test Procedure

Parameter	Unit	Cell 1		Cell 6	
		T0	T1	T0	T1
UTRA RF Channel Number		Ch. 1		Ch. 2	
CPICH Ec (FDD)	dBm/ 3.84 MHz	-55	-55	Off	-55
P-CCPCH RSCP (TDD)	<u>dBm</u>	<u>-55</u>	<u>-55</u>	<u>Off</u>	<u>-55</u>

Table 8.1.3.6

Table 8.1.3.6 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. SS switches the power settings from columns "T0" to "T1", whenever the description in multi-cell condition specifies the transmission power settings for cell 1 and cell 6.

The UE is in CELL_DCH state of cell 1 and the SS has configured its downlink transmission power setting according to columns "T0" in table 8.1.3.6. The SS switches its downlink transmission power settings to columns "T1" and transmits MEASUREMENT CONTROL message and add cell 6 into the IE "inter-frequency cell info". The SS modify contents of SIB3 in cell 1 and cell 6. The SS transmits an RRC CONNECTION RELEASE message. After the SS transmits an RRC CONNECTION RELEASE message. After the SS transmits an RRC CONNECTION RELEASE message to the UE, the SS waits for the UE to transmit RRC CONNECTION RELEASE messages using UM on DCCH and checks to see if N308+1 such messages has been received. The UE leaves connected mode and enters idle mode in cell 1. The UE shall perform cell reselection and camp on cell 6 after reading the system information. The SS calls for generic procedure C.3 to check that UE is in Idle state.

Expected sequence

Step	Direction	Message	Comment
	UE SS		
1			The UE is in the CELL_DCH
			state of cell 1 and the SS has
			configured its downlink
			transmission power setting
			according to columns "T0" in
			table 8.1.3.6.
2			The SS switches its downlink
			transmission power settings to
			columns "T1" in table 8.1.3.6.
3	÷	MEASUREMENT CONTROL	The SS specifies inter-frequency
			measurement for cell 6.
4	÷	System Information Block type 3	The SS modifies SIB 3 in cell 6.
5	÷	System Information Block type 3	The SS modifies SIB 3 in cell 1
			to indicate that the cell is barred.
6			The SS waits for 5 s.
7	÷	RRC CONNECTION RELEASE	
8	\rightarrow	RRC CONNECTION RELEASE	The SS waits for the arrival of
		COMPLETE	N308+1 such messages send on UM RLC.
9			The UE releases signalling radio
			bearer and dedicated resources.
			Then the UE goes to idle mode
			in cell 1.
10			The UE select s cell 6 and camp
			on it.
11			The SS waits for 15 s after
			receiving the last RRC
			RELEASE COMPLETE
			message.
12	\leftrightarrow	CALL C.1	If the test result of C.1 indicates
			that UE is in CELL_DCH state,
			the test passes, otherwise it fails.

Specific Message Content

MEASUREMENT CONTROL (Step 3)

Use the same message sub-type found in [9] TS 34.108 clause 9, with the following exceptions in the IE(s) concerned:

Information Element	Value/remark
Measurement Identity	15
Measurement Command	Setup
Measurement Reporting Mode	
- Measurement Reporting Transfer Mode	Acknowledged Mode RLC
- Periodic Reporting / Event Trigger Reporting Mode	Event Trigger
Additional measurements list	Not Present
CHOICE measurement type	Inter-frequency measurement
- Inter-frequency measurement object list	
- Inter-frequency cell info list	
- CHOICE inter-frequency cell removal	No inter-frequency cells removed
- New inter-frequency cells	
- Inter-frequency cell id	6
- Frequency info	
- UARFCN uplink (Nu)	UARFCN of the uplink frequency for cell 6
- UARFCN downlink (Nd)	UARFCN of the downlink frequency for cell 6
- Cell info	
- Cell individual offset	0 dB
 Reference time difference to cell 	0 chips
- Read SFN Indicator	
- CHOICE Mode	FDD
- Primary CPICH Info	
- Primary Scrambling Code	350
- Primary CPICH TX power	Not Present
- Primary CPICH TX power	
- TX Diversity Indicator	Not Present
- Cell for measurement	Not Present
 Inter-frequency measurement quantity 	
- CHOICE reporting criteria	Inter-frequency reporting criteria
- Filter Coefficient	0
- CHOICE Mode	FDD
- Measurement quantity for frequency quality	CPICH RSCP
estimate	
 Inter-frequency reporting quantity 	
- UTRA Carrier RSSI	FALSE
 Frequency quality estimate 	FALSE
 Non frequency related cell reporting quantities 	
- SFN-SFN observed time difference reporting	No report
indicator	
- Cell synchronisation information reporting	FALSE
indicator	
 Cell Identity reporting indicator 	TRUE
- COICE Mode	FDD
- CPICH Ec/No reporting indicator	FALSE
- CPICH RSCP reporting indicator	TRUE
- Pathloss reporting indicator	FALSE
- Reporting cell status	Not present
- CHOICE reported cell	Report cells within active and/or monitored set on used
	frequency or within active and/or monitored set on non-
	used frequency
- Maximum number of reported cells	2
- Measurement validity	
- UE state	CELL_DCH
- Inter-frequency set update	Not Present
- CHOICE report criteria	Inter-frequency measurement reporting criteria
- Parameters required for each event	
- Inter-frequency event identity	2c
- Threshold used frequency	Not present
- W used frequency	Not present
- Hysteresis	1.0 dB
- Time to trigger	10 [s]

 Reporting cell status CHOICH reported cell 	Report cells within active and/or monitored set on used frequency or within active and/or monitored set on non- used frequency
 Maximum number of reported cells 	2
- Parameters required for each non-used	
frequency	
 Threshold non used frequency 	-85dbm
- W non-used frequency	0.0

System Information Block type 3 (Step 4)

Use the same message type found in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
- Cell identity	0000 0000 0000 0000 0000 0000 0110B

System Information Block type 3 (Step 5)

Use the same message type found in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
- Cell Access Restriction	
- Cell barred	Barred
- Intra-frequency cell re-selection indicator	Not allowed
- T _{barred}	10[s]
- Cell Reserved for operator use	Not reserved
- Cell Reservation Extension	Not reserved
- Access Class Barred List	
- Access Class Barred0	barred
- Access Class Barred1	barred
- Access Class Barred2	barred
- Access Class Barred3	barred
- Access Class Barred4	barred
- Access Class Barred5	barred
- Access Class Barred6	barred
- Access Class Barred7	barred
- Access Class Barred10	barred
- Access Class Barred11	barred
- Access Class Barred12	barred
- Access Class Barred13	barred
- Access Class Barred14	barred
- Access Class Barred15	barred

RRC CONNECTION RELEASE (Step 6)

Use the same message type found in clause 9 of TS 34.108, with the following exceptions:

Information Element	Value/remark
N308	Arbitrarily chosen between 1 and 8

8.1.3.6.5 Test requirement

After step 6 the UE shall start to transmit N308 + 1 times RRC CONNECTION RELEASE COMPLETE messages using UM on DCCH.

After step 11 the UE shall be in Idle mode in cell 6.

Release 5

8.1.3.7 RRC Connection Release in CELL_FACH state (Frequency band modification): Success

- 8.1.3.7.1 Definition
- 8.1.3.7.2 Conformance requirement

When the UE receives the first RRC CONNECTION RELEASE message

the UE shall:

- 1> in state CELL_FACH:
 - 2> if the RRC CONNECTION RELEASE message was received on the DCCH:
 - 3> when the successful transmission of the RRC CONNECTION RELEASE COMPLETE message has been confirmed by the lower layers:
 - 4> release all its radio resources; and
 - 4> indicate the release of the established signalling connections ; and
 - 4> pass the value of the IE "Release cause" received in the RRC CONNECTION RELEASE message to upper layers;
 - 4> enter idle mode;
 - 4> perform the actions specified in TS25.331 subclause 8.5.2 when entering idle mode.
- a) 3> and the procedure ends.

Reference

3GPP TS 25.331 clause 8.1.4.

8.1.3.7.3 Test purpose

To confirm that when the UE receives an RRC CONNECTION RELEASE message, the UE releases signalling radio bearer and its radio resources and goes back to the idle.

To confirm that the UE enters into idle mode withby performing cell-selection and selecting other cell than the UE selecting cell in connected mode.

8.1.3.7.4 Method of test

Initial Condition

System Simulator: 2 cells - Cell 1 is active and cell 6 is inactive

UE: "Registered idle mode on CS" (state 2) or "Registered idle mode on PS" (state 3) in cell 1 as specified in clause 7.4 of TS 34.108, depending on the CN domain supported by the UE If the UE supports both CS and PS domains, the initial UE state shall be "Registered idle mode on CS/PS" (state 7).

Specific Message Content

For system information block 11 for Cell 1 (gives IE's which are different from defaults given in 34.108 sec 6.1) to be transmitted before idle update preamble.

System Information Block type 11

Use same message sub-clause 6.1 of TS34.108, with following exception:

Information Element	Value/remark
- SIB12 indicator	FALSE
 Intra-frequency measurement system information 	Not Present
 Inter-frequency measurement system information 	
 Inter-frequency cell info list 	
 New inter-frequency cell id 	
 Inter frequency cell id 	1
- Frequency info	
- UARFCN uplink(Nu)	Same uplink UARFCN as used for cell 6
- UARFCN downlink(Nd)	Same downlink UARFCN as used for cell 6
- Cell info	
- Cell individual offset	0dB
 Reference time difference to cell 	Not present
- Read SFN indicator	FALSE
- CHOICE mode	FDD
- Primary CPICH info	
- Primary scrambling code	Set to same code as used for cell 6
- Primary CPICH Tx power	Not present
- Cell Selection and Re-selection Info	
- Qoffset1 _{s,n}	0dB
- Qoffset2s,n	Not present
 Maximum allowed UL TX power 	Reference to table 6.1.1
 HCS neighbouring cell information 	Not present
- CHOICE mode	FDD
- Qqualmin	Reference to table 6.1.1
- Qrxlevmin	Reference to table 6.1.1
- Cell for measurement	Not present

Test Procedure

Parameter	Unit	Cell 1		Cell 6	
		T0	T1	T0	T1
UTRA RF Channel Number		Ch. 1		Ch. 2	
CPICH Ec (FDD)	dBm/ 3.84 MHz	-60	-60	off	-60
P-CCPCH RSCP (TDD)	<u>dBm</u>	<u>-60</u>	<u>-60</u>	<u>off</u>	<u>-60</u>

Table 8.1.3.7

Table 8.1.3.7 illustrates the downlink power to be applied for the 2 cells at various time instants of the test execution. SS switches the power settings from columns "T0" to "T1", whenever the description in multi-cell condition specifies the transmission power settings for cell 1 and cell 6.

The UE is in idle mode of cell 1 and the SS has configured its downlink transmission power setting according to columns "T0" in table 8.1.3.7. SS request operator to make an outgoing call. The SS and UE execute procedure P4 or P6. Next The SS and the UE execute procedure P8 or P10. The SS switches its downlink transmission power settings to columns "T1" and then modifies SIB 3 to indicate that cell 1 is barred. The SS transmits an RRC CONNECTION RELEASE message on DCCH. The UE shall transmit an RRC CONNECTION RELEASE COMPLETE message using UM on DCCH and try to enter idle mode state in cell 1. On selecting cell 1 the UE reads system information block 3 and is aware that cell 1 is barred cell. Hence the UE selects cell 6 and camp on cell 6. Upon completion of the procedure, the SS calls for generic procedure C.1 to check that UE is in idle mode.

Expected sequence

Step	Direction		Direction		Message	Comment
	UE	SS				
1				The UE is in the CELL_FACH state of cell 1 and the SS has configured its downlink transmission power setting according to columns "T0" in table 8.1.3.7. SS requests operator to make an outgoing call.		
2			SS executes procedure P4 (clause 7.4.2.1.2) or P6 (clause 7.4.2.2.2) specified in TS 34.108.			
3			SS executes procedure P8 (clause 7.4.2.3.2) or P10 (clause 7.4.2.4.2) specified in TS 34.108.			
4				The SS switches its downlink transmission power settings to columns "T1" in table 8.1.3.7.		
5			System Information Block type 3	The SS modifies SIB 3 in cell 1 to indicate that the cell is barred.		
6				The SS waits for 5 s		
7			RRC CONNECTION RELEASE			
8	÷	>	RRC CONNECTION RELEASE COMPLETE			
9				The SS waits for 5s		
10	÷	\rightarrow	CALL C.1	If the test result of C.1 indicates that UE is in idle mode, the test passes, otherwise it fails.		

Specific Message Content

System Information Block type 3 (Step 5)

Use the same message type found in clause 6 of TS 34.108, with the following exceptions:

Information Element	Value/remark
- Cell Access Restriction	
- Cell barred	Barred
 Intra-frequency cell re-selection indicator 	Not allowed
- T _{barred}	10[s]
- Cell Reserved for operator use	Not reserved
- Cell Reservation Extension	Not reserved
- Access Class Barred List	
- Access Class Barred0	barred
- Access Class Barred1	barred
- Access Class Barred2	barred
- Access Class Barred3	barred
- Access Class Barred4	barred
- Access Class Barred5	barred
- Access Class Barred6	barred
- Access Class Barred7	barred
- Access Class Barred10	barred
- Access Class Barred11	barred
- Access Class Barred12	barred
- Access Class Barred13	barred
- Access Class Barred14	barred
- Access Class Barred15	barred

8.1.3.7.5 Test requirement

After step 3 the UE shall transmit RRC CONNECTION RELEASE COMPLETE messages using AM on DCCH.

After step 9 the UE shall be in idle mode of cell 6.

Luton, UK, 4th – 8th November 2002 CR-Form-v7 CHANGE REQUEST ж Current version: ж 34.123-1 CR 382 ж жrev 5.1.1 For **HELP** on using this form, see bottom of this page or look at the pop-up text over the **#** symbols. ME X Radio Access Network Core Network UICC apps # Proposed change affects: +Title: **#** CR to 34.123-1 REL-5; Corrections to radio bearer test cases in clause 14.2 Source: Ж Ericsson Date: # 27/10/2002 Work item code: # TEI жF Category: Release: # REL-5 Use one of the following categories: Use one of the following releases: F (correction) (GSM Phase 2) 2 A (corresponds to a correction in an earlier release) R96 (Release 1996) B (addition of feature), R97 (Release 1997) **C** (functional modification of feature) R98 (Release 1998) **D** (editorial modification) R99 (Release 1999) Detailed explanations of the above categories can (Release 4) Rel-4 be found in 3GPP TR 21.900. (Release 5) Rel-5 Rel-6 (Release 6)

Reason for change: ೫	Test requirements incorrect for some radio bearer test cases Clarification to configuration under test needed for some radio bearer test cases.
Summary of change: #	 Clarified that main alternative for uplink TF0 is to be used for following test cases: 14.2.4, 14.2.4a, 14.2.5, 14.2.5a, 14.2.6, 14.2.7, 14.2.7a, 14.2.8, 14.2.9, 14.2.10, 14.2.11, 14.2.38.1, 14.2.38.2, 14.2.38a, 14.2.38b, 14.2.38c, 14.2.38d, 14.2.38e, 14.2.38f, 14.2.38g, 14.2.38h, 14.2.38i, 14.2.38j, 14.2.39.1, 14.2.39.2, 14.2.40, 14.2.41, 14.2.42.1, 14.2.42.2, 14.2.43.1, 14.2.43.2, 14.2.44.1, 14.2.44.2, 14.2.45, 14.2.46, 14.2.49.1, 14.2.49.2, 14.2.49a 14.2.38e: Corrected Downlink TFS for TF0 and RB5 to 1x0 according to transport channel parameters for Conversational / speech / DL: (12.2 7.95 5.9 4.75) kbps / CS RAB specified in 34.108 clause 6.10.2.4.1.4a.2.1.1. 14.2.38j: Corrected test requirement for sub-tests 25 to 29. The content of the returned SDU shall be equal to the first 1272 bits of the test data sent by the SS in downlink. 14.2.39.2: Corrected test requirement for sub-tests 9 to 14. The content of the returned SDU shall be equal to the first 632 bits of the test data
	sent by the SS in downlink.

Tdoc #T1-020840

Tdoc # T1S-020798

	 14.2.41: Corrected test requirement for sub-tests 9 to 11. The content of the returned SDU shall be equal to the first 952 bits of the test data sent by the SS in downlink. Corrected test requirement for sub-tests 12 to 14. The content of the returned SDU shall be equal to the first 1272 bits of the test data sent by the SS in downlink. 14.2.43.1: Corrected test requirement for sub-tests 12 to 17. The content of the returned SDU shall be equal to the first 632 bits of the test data sent by the SS in downlink. 14.2.43.1: Corrected test requirement for sub-tests 12 to 17. The content of the returned SDU shall be equal to the first 632 bits of the test data sent by the SS in downlink. 14.2.49a: Corrected Downlink TFS for TF0 and RB5 to 1x0 according to transport channel parameters for Conversational / speech / DL: (12.2 7.95 5.9 4.75) kbps / CS RAB specified in 34.108 clause 6.10.2.4.1.4a.2.1.1.
Consequences if % not approved:	Incorrect radio bearer test cases
Clauses affected: #	14.2
Other specs % affected:	YNXOther core specificationsXTest specificationsXO&M Specifications

How to create CRs using this form:

Comprehensive information and tips about how to create CRs can be found at <u>http://www.3gpp.org/specs/CR.htm</u>. Below is a brief summary:

- 1) Fill out the above form. The symbols above marked # contain pop-up help information about the field that they are closest to.
- 2) Obtain the latest version for the release of the specification to which the change is proposed. Use the MS Word "revision marks" feature (also known as "track changes") when making the changes. All 3GPP specifications can be downloaded from the 3GPP server under <u>ftp://ftp.3gpp.org/specs/</u> For the latest version, look for the directory name with the latest date e.g. 2001-03 contains the specifications resulting from the March 2001 TSG meetings.
- 3) With "track changes" disabled, paste the entire CR form (use CTRL-A to select it) into the specification just in front of the clause containing the first piece of changed text. Delete those parts of the specification which are not relevant to the change request.

14.2 Combinations on DPCH

14.2.1 Stand-alone UL:1.7 DL:1.7 kbps SRBs for DCCH

Test to verify establishment and signalling of stand-alone signalling reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.1.

The test case is performed by running test case 9.4.1 (Location updating / accepted) using the stand-alone signalling reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.1.

14.2.2 Stand-alone UL:3.4 DL:3.4 kbps SRBs for DCCH

Test to verify establishment and signalling of stand-alone signalling reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.2.

The test case is performed by running test case 9.4.1 (Location updating / accepted) using the stand-alone signalling reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.2.

14.2.3 Stand-alone UL:13.6 DL:13.6 kbps SRBs for DCCH

Impicitely tested.

NOTE The stand-alone UL:13.6 DL:13.6 kbps SRBs for DCCH in TS 34.108, clause 6.10.2.4.1.3 is the default signalling radio bearer used in the generic setup procedure as specified in TS 34.108 clasue 7.

14.2.4 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.4.1 Conformance requirement

The UE shall be able to establish the UTRAN requested radio bearers within the UE's signaled radio access capabilities.

The UE shall correctly transfer user data from peer to peer RLC entitities according to the requested radio bearer configuration.

Reference(s)

3GPP TS 25.331, clause 8.2.1

3GPP TS 25.2xx series (Physical Layer)

3GPP TS 25.321 (MAC)

3GPP TS 25.322 (RLC)

14.2.4.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.4.

14.2.4.3 Method of test

Uplink TFS:

1

	RB5RB6(RAB subflow #1)(RAB subflow #2)(RAE		RB7 (RAB subflow #3)	DCCH	
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x148
	TF2, bits	1x81	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, DCCH)				
UL_TFC0	(TF0, TF0, TF0, TF0)				
UL_TFC1	(TF1, TF0, TF0, TF0)				
UL_TFC2	(TF2, TF1, TF1, TF0)				
UL_TFC3	(TF0, TF0, TF0, TF1)				
UL_TFC4	(TF1, TF0, TF0, TF1)				
UL_TFC5	(TF2, TF1, TF1, TF1)				

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	DCCH
	TF0, bits	1x0	0x103	0x60	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x148
	TF2, bits	1x81	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, DCCH)				
DL_TFC0	(TF0, TF0, TF0, TF0)				
DL_TFC1	(TF1, TF0, TF0, TF0)				
DL_TFC2	(TF2, TF1, TF1, TF0)				
DL_TFC3	(TF0, TF0, TF0, TF1)				
DL_TFC4	(TF1, TF0, TF0, TF1)				
DL_TFC5	(TF2, TF1, TF1, TF1)				

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (note)	Test data size
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 103 bits RB7: 60 bits	RB5: 39 bits RB6: No data RB7: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5	RB5: 81 bits RB6: 103 bits RB7: 60 bits	RB5: 81 bits RB6: 103 bits RB7: 60 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loopl	back of RLC SE	Us.	

See 14.1.1 for test procedure.

14.2.4.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).

- for sub-test 2: RB5/TF2 (1x81); RB6/TF1 (1x103); and RB7/TF1 (1x60).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 or RB7.
 - for sub-test 2: an RLC SDU on each of RB5, RB6 and RB7 having the same content as sent by SS

14.2.4a Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL:(12.2 7.95 5.9 4.75) kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.4a.1 Conformance requirement

See clause 14.2.4.1.

14.2.4a.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.4a.

14.2.4a.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x148
TFS	TF1, bits	1x39	1x53	1x60	1x148
	TF2, bits	1x42	1x63	N/A	N/A
11-3	TF3, bits	1x55	1x84	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7,DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF0, TF0)
UL_TFC3	(TF3, TF2, TF0, TF0)
UL_TFC4	(TF4, TF3, TF0, TF0)
UL_TFC5	(TF5, TF4, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF0, TF1)
UL_TFC9	(TF3, TF2, TF0, TF1)
UL_TFC10	(TF4, TF3, TF0, TF1)
UL_TFC11	(TF5, TF4, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	рссн
	TF0, bits	1x0	0x103	0x60	0x148
	TF1, bits	1x39	1x53	1x60	1x148
TFS	TF2, bits	1x42	1x63	N/A	N/A
11-3	TF3, bits	1x55	1x84	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A

Release 5

Downlink TFCS:

TFCI	(RB5, RB6, RB7, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF0, TF0)
DL_TFC3	(TF3, TF2, TF0, TF0)
DL_TFC4	(TF4, TF3, TF0, TF0)
DL_TFC5	(TF5, TF4, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF0, TF1)
DL_TFC9	(TF3, TF2, TF0, TF1)
DL_TFC10	(TF4, TF3, TF0, TF1)
DL_TFC11	(TF5, TF4, TF1, TF1)

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7	RB5: 39 bits RB6: 103 bits RB7: 60 bits	RB5: 39 bits RB6: No data RB7: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 42 bits RB6: 53 bits RB7: 60 bits	RB5: 42 bits RB6: 53 bits RB7: No data
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 55 bits RB6: 63 bits RB7: 60 bits	RB5: 55 bits RB6: 63 bits RB7: No data
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC4, UL_TFC6, UL_TFC10	RB5: 75 bits RB6: 84 bits RB7: 60 bits	RB5: 75 bits RB6: 84 bits RB7: No data
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC11	RB5: 81 bits RB6: 103 bits RB7: 60 bits	RB5: 81 bits RB6: 103 bits RB7: 60 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE)Us.	

See clause 14.1.1 for test procedure.

14.2.4a.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x42) and RB6/TF1 (1x53)
 - for sub-test 3: RB5/TF3 (1x55) and RB6/TF2 (1x63)
 - for sub-test 4: RB5/TF4 (1x75) and RB6/TF3 (1x84)
 - for sub-test 5: RB5/TF5 (1x81), RB6/TF4 (1x103) and RB7/TF1 (1x60)

- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 or RB7.
 - for sub-test 2,3 and 4: an RLC SDU on RB5 and RB6 having the same content as sent by SS; and no data shall be received on RB7.
 - for sub-test 5: an RLC SDU on each of RB5, RB6 and RB7 having the same content as sent by SS.

14.2.5 Conversational / speech / UL:10.2 DL:10.2 kbps / CS RAB + UL:3.4 DL: 3.4 kbps SRBs for DCCH

14.2.5.1 Conformance requirement

See clause 14.2.4.1.

14.2.5.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.5.

14.2.5.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	DCCH
	TF0, bits	0x65 (alt. 1x0)	0x99	0x40	0x148
TFS	TF1, bits	1x39	1x99	1x40	1x148
	TF2, bits	1x65	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7,DCCH)		
UL_TFC0	(TF0, TF0, TF0, TF0)		
UL_TFC1	(TF1, TF0, TF0, TF0)		
UL_TFC2	(TF2, TF1, TF1, TF0)		
UL_TFC3	(TF0, TF0, TF0, TF1)		
UL_TFC4	(TF1, TF0, TF0, TF1)		
UL_TFC5	(TF2, TF1, TF1, TF1)		

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	DCCH
	TF0, bits	1x0	0x99	0x40	0x148
TFS	TF1, bits	1x39	1x99	1x40	1x148
	TF2, bits	1x65	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, DCCH)				
DL_TFC0	(TF0, TF0, TF0, TF0)				
DL_TFC1	(TF1, TF0, TF0, TF0)				
DL_TFC2	(TF2, TF1, TF1, TF0)				
DL_TFC3	(TF0, TF0, TF0, TF1)				
DL_TFC4	(TF1, TF0, TF0, TF1)				
DL_TFC5	(TF2, TF1, TF1, TF1)				

Release 5

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 99 bits RB7: 40 bits	RB5: 39 bits RB6: No data RB7: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5	RB5: 65 bits RB6: 99 bits RB7: 40 bits	RB5: 65 bits RB6: 99 bits RB7: 40 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	

See clause 14.1.1 for test procedure.

14.2.5.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x65); RB6/TF1 (1x99); and RB7/TF1 (1x40).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 or RB7.
 - for sub-test 2: an RLC SDU on each of RB5, RB6 and RB7 having the same content as sent by SS.

14.2.5a Conversational / speech / UL:(10.2, 6.7, 5.9, 4.75) DL:(10.2, 6.7, 5.9, 4.75) kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.5a.1 Conformance requirement

See clause 14.2.4.1.

14.2.5a.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.5a.

14.2.5a.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	DCCH
	TF0, bits	0x65 (alt. 1x0)	0x99	0x40	0x148
	TF1, bits	1x39	1x53	1x40	1x148
TFS	TF2, bits	1x42	1x63	N/A	N/A
115	TF3, bits	1x55	1x76	N/A	N/A
	TF4, bits	1x58	1x99	N/A	N/A
	TF5, bits	1x65	N/A	N/A	N/A

Release 5

Uplink TFCS:

TFCI	(RB5, RB6, RB7,DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF0, TF0)
UL_TFC3	(TF3, TF2, TF0, TF0)
UL_TFC4	(TF4, TF3, TF0, TF0)
UL_TFC5	(TF5, TF4, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF0, TF1)
UL_TFC9	(TF3, TF2, TF0, TF1)
UL_TFC10	(TF4, TF3, TF0, TF1)
UL_TFC11	(TF5, TF4, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	DCCH
	TF0, bits	1x0	0x99	0x40	0x148
	TF1, bits	1x39	1x53	1x40	1x148
TFS	TF2, bits	1x42	1x63	N/A	N/A
15	TF3, bits	1x55	1x76	N/A	N/A
	TF4, bits	1x58	1x99	N/A	N/A
	TF5, bits	1x65	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF0, TF0)
DL_TFC3	(TF3, TF2, TF0, TF0)
DL_TFC4	(TF4, TF3, TF0, TF0)
DL_TFC5	(TF5, TF4, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF0, TF1)
DL_TFC9	(TF3, TF2, TF0, TF1)
DL_TFC10	(TF4, TF3, TF0, TF1)
DL_TFC11	(TF5, TF4, TF1, TF1)

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
	under lest	Under lest			(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7	RB5: 39 bits RB6: 99 bits RB7: 40 bits	RB5: 39 bits RB6: No data RB7: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 42 bits RB6: 53 bits RB7: 40 bits	RB5: 42 bits RB6: 53 bits RB7: No data
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 55 bits RB6: 63 bits RB7: 40 bits	RB5: 55 bits RB6: 63 bits RB7: No data
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC4, UL_TFC6, UL_TFC10	RB5: 58 bits RB6: 76 bits RB7: 40 bits	RB5: 58 bits RB6: 76 bits RB7: No data
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC11	RB5: 65 bits RB6: 99 bits RB7: 40 bits	RB5: 65 bits RB6: 99 bits RB7: 40 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	

See clause 14.1.1 for test procedure.

14.2.5a.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x42) and RB6/TF1 (1x53)
 - for sub-test 3: RB5/TF3 (1x55) and RB6/TF2 (1x63)
 - for sub-test 4: RB5/TF4 (1x58) and RB6/TF3 (1x76)
 - for sub-test 5: RB5/TF5 (1x65), RB6/TF4 (1x99) and RB7/TF1 (1x40)
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by the SS; and no data shall be received on RB6 or RB7.
 - for sub-test 2, 3 and 4: an RLC SDU on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 5: an RLC SDU on each of RB5, RB6 and RB7 having the same content as sent by the SS.

14.2.6 Conversational / speech / UL:7.95 DL:7.95 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.6.1 Conformance requirement

See clause 14.2.4.1.

Release 5

14.2.6.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.6.

14.2.6.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	0x75 (alt. 1x0)	0x84	0x148
TFS	TF1, bits	1x39	1x84	1x148
	TF2, bits	1x75	N/A	N/A

Uplink TFCS:

TFCI		(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)	
UL_TFC1	(TF1, TF0, TF0)	
UL_TFC2	(TF2, TF1, TF0)	
UL_TFC3	(TF0, TF0, TF1)	
UL_TFC4	(TF1, TF0, TF1)	
UL_TFC5	(TF2, TF1, TF1)	

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	1x0	0x84	0x148
TFS	TF1, bits	1x39	1x84	1x148
	TF2, bits	1x75	N/A	N/A

Downlink TFCS:

TFCI		RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF2, TF1, TF0)	
DL_TFC3	(TF0, TF0, TF1)	
DL_TFC4	(TF1, TF0, TF1)	
DL_TFC5	(TF2, TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (note)	Test data size (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 84 bits	RB5: 39 bits RB6: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 75 bits RB6: 84 bits	RB5: 75 bits RB6: 84 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE)Us.	•

See clause 14.1.1 for test procedure.

14.2.6.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x75) and RB6/TF1 (1x84).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: an RLC SDU on each of RB5 and RB6 having the same content as sent by SS.

14.2.7 Conversational / speech / UL:7.4 DL:7.4 kbps / CS RAB+ UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.7.1 Conformance requirement

See clause 14.2.4.1.

14.2.7.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.7.

14.2.7.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	0x61 (alt. 1x0)	0x87	0x148
TFS	TF1, bits	1x39	1x87	1x148
	TF2, bits	1x61	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)				
UL_TFC0	(TF0, TF0, TF0)				
UL_TFC1	(TF1, TF0, TF0)				
UL_TFC2	(TF2, TF1, TF0)				
UL_TFC3	(TF0, TF0, TF1)				
UL_TFC4	(TF1, TF0, TF1)				
UL_TFC5	(TF2, TF1, TF1)				

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	1x0	0x87	0x148
TFS	TF1, bits	1x39	1x87	1x148
	TF2, bits	1x61	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)	
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF2, TF1, TF0)	
DL_TFC3	(TF0, TF0, TF1)	
DL_TFC4	(TF1, TF0, TF1)	
DL_TFC5	(TF2, TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (note)	Test data size
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 87 bits	RB5: 39 bits RB6: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 61 bits RB6: 87 bits	RB5: 61 bits RB6: 87 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	•

See clause 14.1.1 for test procedure.

14.2.7.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x61) and RB6/TF1 (1x87).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: an RLC SDU on each of RB5 and RB6 having the same content as sent by SS

14.2.7a Conversational / speech / UL:(7.4, 6.7, 5.9, 4.75) DL:(7.4, 6.7, 5.9, 4.75) kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.7a.1 Conformance requirement

See clause 14.2.4.1.

14.2.7a.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.7a.

14.2.7a.3 Method of test

Uplink TFS:

I

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	0x61 (alt. 1x0)	0x87	0x148
	TF1, bits	1x39	1x53	1x148
TFS	TF2, bits	1x42	1x63	N/A
15	TF3, bits	1x55	1x76	N/A
	TF4, bits	1x58	1x87	N/A
	TF5, bits	1x61	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7,DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0)
UL_TFC2	(TF2, TF1, TF0)
UL_TFC3	(TF3, TF2, TF0)
UL_TFC4	(TF4, TF3, TF0)
UL_TFC5	(TF5, TF4, TF0)
UL_TFC6	(TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF1)
UL_TFC8	(TF2, TF1, TF1)
UL_TFC9	(TF3, TF2, TF1)
UL_TFC10	(TF4, TF3, TF1)
UL_TFC11	(TF5, TF4, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	1x0	0x87	0x148
	TF1, bits	1x39	1x53	1x148
TFS	TF2, bits	1x42	1x63	N/A
15	TF3, bits	1x55	1x76	N/A
	TF4, bits	1x58	1x87	N/A
	TF5, bits	1x61	N/A	N/A

Downlink TFCS:

TFCI		(RB5, RB6, RB7, DCCH)
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF2, TF1, TF0)	
DL_TFC3	(TF3, TF2, TF0)	
DL_TFC4	(TF4, TF3, TF0)	
DL_TFC5	(TF5, TF4, TF0)	
DL_TFC6	(TF0, TF0, TF1)	
DL_TFC7	(TF1, TF0, TF1)	
DL_TFC8	(TF2, TF1, TF1)	
DL_TFC9	(TF3, TF2, TF1)	
DL_TFC10	(TF4, TF3, TF1)	
DL_TFC11	(TF5, TF4, TF1)	

Sub-tests:

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
	under test	Under test				
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
			UL_TFC6	UL_TFC1,	RB6: 87 bits	RB6: No data
				UL_TFC6,		
				UL_TFC7		
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
			UL_TFC6	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
				UL_TFC6,		
				UL_TFC8		
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
			UL_TFC6	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
				UL_TFC6,		
				UL_TFC9		
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 58 bits	RB5: 58 bits
			UL_TFC6	UL_TFC4,	RB6: 76 bits	RB6: 76 bits
				UL_TFC6,		
				UL_TFC10		
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 61 bits	RB5: 61 bits
			UL_TFC6	UL_TFC5,	RB6: 87 bits	RB6: 87 bits
				UL_TFC6,		
				UL_TFC11		
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE)Us.	

See clause 14.1.1 for test procedure.

14.2.7a.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.

- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x42) and RB6/TF1 (1x53)
 - for sub-test 3: RB5/TF3 (1x55) and RB6/TF2 (1x63)
 - for sub-test 4: RB5/TF4 (1x58) and RB6/TF3 (1x76)
 - for sub-test 5: RB5/TF5 (1x61) and RB6/TF4 (1x87)
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by the SS; and no data shall be received on RB6.
 - for sub-test 2 to 5: an RLC SDU on RB5 and RB6 having the same content as sent by the SS.

14.2.8 Conversational / speech / UL:6.7 DL:6.7 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.8.1 Conformance requirement

See clause 14.2.4.1.

14.2.8.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.8.

14.2.8.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	0x58 (alt. 1x0)	0x76	0x148
TFS	TF1, bits	1x39	1x76	1x148
	TF2, bits	1x58	N/A	N/A

Uplink TFCS:

TFCI		(RB5, RB6, DCCH)	
UL_TFC0	(TF0, TF0, TF0)		
UL_TFC1	(TF1, TF0, TF0)		
UL_TFC2	(TF2, TF1, TF0)		
UL_TFC3	(TF0, TF0, TF1)		
UL_TFC4	(TF1, TF0, TF1)		
UL_TFC5	(TF2, TF1, TF1)		

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	1x0	0x76	0x148
TFS	TF1, bits	1x39	1x76	1x148
	TF2, bits	1x58	N/A	N/A

Downlink TFCS:

TFCI		(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF2, TF1, TF0)	
DL_TFC3	(TF0, TF0, TF1)	
DL_TFC4	(TF1, TF0, TF1)	
DL_TFC5	(TF2, TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (note)	Test data size (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 76 bits	RB5: 39 bits RB6: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 58 bits RB6: 76 bits	RB5: 58 bits RB6: 76 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop		Us.	

See clause 14.1.1 for test procedure.

14.2.8.4 Test requirements

See clause 14.1.1 for definition of step 10 and step 15.

1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.

- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x58) and RB6/TF1 (1x76).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: an RLC SDU on each of RB5 and RB6 having the same content as sent by SS

14.2.9 Conversational / speech / UL:5.9 DL:5.9 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.9.1 Conformance requirement

See clause 14.2.4.1.

14.2.9.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.9.

14.2.9.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	0x55 (alt. 1x0)	0x63	0x148
TFS	TF1, bits	1x39	1x63	1x148
	TF2, bits	1x55	N/A	N/A

Uplink TFCS:

TFCI		(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)	
UL_TFC1	(TF1, TF0, TF0)	
UL_TFC2	(TF2, TF1, TF0)	
UL_TFC3	(TF0, TF0, TF1)	
UL_TFC4	(TF1, TF0, TF1)	
UL_TFC5	(TF2, TF1, TF1)	

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	1x0	0x63	0x148
TFS	TF1, bits	1x39	1x63	1x148
	TF2, bits	1x55	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)	
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF2, TF1, TF0)	
DL_TFC3	(TF0, TF0, TF1)	
DL_TFC4	(TF1, TF0, TF1)	
DL_TFC5	(TF2, TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (note)	Test data size
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 63 bits	RB5: 39 bits RB6: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 55 bits RB6: 63 bits	RB5: 55 bits RB6: 63 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loopl	back of RLC SE	Us.	•

See clause 14.1.1 for test procedure.

14.2.9.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x55) and RB6/TF1 (1x63).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: an RLC SDU on each of RB5 and RB6 having the same content as sent by SS

14.2.10 Conversational / speech / UL:5.15 DL:5.15 kbps / CS RAB + UL:1.7 DL:1.7 kbps SRBs for DCCH

14.2.10.1 Conformance requirement

See clause 14.2.4.1.

14.2.10.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.10.

14.2.10.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	0x49 (alt. 1x0)	0x54	0x148
TFS	TF1, bits	1x39	1x54	1x148
	TF2, bits	1x49	N/A	N/A

Uplink TFCS:

TFCI		(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)	
UL_TFC1	(TF1, TF0, TF0)	
UL_TFC2	(TF2, TF1, TF0)	
UL_TFC3	(TF0, TF0, TF1)	
UL_TFC4	(TF1, TF0, TF1)	
UL_TFC5	(TF2, TF1, TF1)	

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	1x0	0x54	0x148
TFS	TF1, bits	1x39	1x54	1x148
	TF2, bits	1x49	N/A	N/A

Downlink TFCS:

TFCI		(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF2, TF1, TF0)	
DL_TFC3	(TF0, TF0, TF1)	
DL_TFC4	(TF1, TF0, TF1)	
DL_TFC5	(TF2, TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 54 bits	RB5: 39 bits RB6: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 49 bits RB6: 54 bits	RB5: 49 bits RB6: 54 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	

See clause 14.1.1 for test procedure.

14.2.10.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).

- for sub-test 2: RB5/TF2 (1x49) and RB6/TF1 (1x54).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: an RLC SDU on each of RB5 and RB6 having the same content as sent by SS

14.2.11 Conversational / speech / UL:4.75 DL:4.75 kbps / CS RAB + UL:1.7 DL:1.7 kbps SRBs for DCCH

14.2.11.1 Conformance requirement

See clause 14.2.4.1.

14.2.11.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.11.

14.2.11.3 Method of test

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	0x49 (alt. 1x0)	0x53	0x148
TFS	TF1, bits	1x39	1x53	1x148
	TF2, bits	1x42	N/A	N/A

Uplink TFCS:

TFCI		(RB5, RB6, DCCH)	
UL_TFC0	(TF0, TF0, TF0)		
UL_TFC1	(TF1, TF0, TF0)		
UL_TFC2	(TF2, TF1, TF0)		
UL_TFC3	(TF0, TF0, TF1)		
UL_TFC4	(TF1, TF0, TF1)		
UL_TFC5	(TF2, TF1, TF1)		

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	DCCH
	TF0, bits	1x0	0x53	0x148
TFS	TF1, bits	1x39	1x53	1x148
	TF2, bits	1x42	N/A	N/A

Downlink TFCS:

TFCI		(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF2, TF1, TF0)	
DL_TFC3	(TF0, TF0, TF1)	
DL_TFC4	(TF1, TF0, TF1)	
DL_TFC5	(TF2, TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (note)	Test data size
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 39 bits RB6: 53 bits	RB5: 39 bits RB6: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 42 bits RB6: 53 bits	RB5: 42 bits RB6: 53 bits
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop		Us.	•

See clause 14.1.1 for test procedure.

14.2.11.4 Test requirements

See clause 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x39).
 - for sub-test 2: RB5/TF2 (1x42) and RB6/TF1 (1x53).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: an RLC SDU on each of RB5 and RB6 having the same content as sent by SS

14.2.12 Conversational / unknown / UL:28.8 DL:28.8 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.12.1 Conformance requirement

See 14.2.4.1.

14.2.12.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.12.

14.2.12.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RLC		
TM RLC		
Transmission RLC discard CHOICE SDU Discard Mode		
Timer based no explicit		
Timer_discard	100ms	
Segmentation indication	FALSE	
Downlink RLC		
TM RLC		
Segmentation indication	FALSE	
NOTE: 'Timer based discard without explicit sig uplink to secure that the UE will be able the case when the UE test loop function will not deliver the SDUs in one and the two subsequent TTIs.	to return data in uplink for n, due to processing delays,	

Uplink TFS:

	TFI	RB5 (28.8 kbps)	DCCH
	TF0, bits	0x576	0x148
TFS	TF1, bits	1x576	1x148
	TF2, bits	2x576	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)		
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF2, TF0)		
UL_TFC3	(TF0, TF1)		
UL_TFC4	(TF1, TF1)		
UL_TFC5	(TF2, TF1)		

Downlink TFS:

		RB5 (28.8 kbps)	DCCH
	TF0, bits	0x576	0x148
TFS	TF1, bits	1x576	1x148
	TF2, bits	2x576	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)		
DL_TFC0	(TF0, TF0)		
DL_TFC1	(TF1, TF0)		
DL_TFC2	(TF2, TF0)		
DL_TFC3	(TF0, TF1)		
DL_TFC4	(TF1, TF1)		
DL_TFC5	(TF2, TF1)		

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 576	RB5: 576
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 576	RB5: 2x576
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	

See clause 14.1.1 for test procedure.

14.2.12.4 Test requirements

See clause 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x576).
 - for sub-test 2: RB5/TF2 (2x576).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS.
 - for sub-test 2: two RLC SDUs on RB5 having the same content as sent by SS.

14.2.13 Conversational / unknown / UL:64 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.13.1 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI

14.2.13.1.1 Conformance requirement

See clause 14.2.4.1.

14.2.13.1.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.13 for the 20 ms TTI case.

14.2.13.1.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RL	C				
TM RL	_C				
Tra	ansmission RLC discard				
	CHOICE SDU Discard Mode				
	Timer based no explicit				
	Timer_discard	100ms			
Se	gmentation indication	FALSE			
Downlink	RLC				
TM RL	_C				
Se	gmentation indication	FALSE			
NOTE:	NOTE: Timer based discard without explicit signalling is used in uplink to				
	secure that the UE will be able to return data for the case when the				
	UE test loop function will not deliver all the SDUs in one and the				
	same TTI .				

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
TFS	TF0, bits	0x640	0x148
113	TF1, bits	2x640	1x148

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF0, TF1)	
UL_TFC3	(TF1, TF1)	

Downlink TFS:

	TFI	RB5 (64 kbps)	DCCH
TFS	TF0, bits	0x640	0x148
15	TF1, bits	2x640	1x148

Downlink TFCS:

TFCI	(RB5, DCCH)	
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF0, TF1)	
DL_TFC3	(TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 640	RB5: 2x640
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs.					

See clause 14.1.1 for test procedure.

14.2.13.1.4 Test requirements

See clause 14.1.1 for definition of step 10 and step 15.

1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.

- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (2x640).
- 3. At step 15 the UE shall return
 - for sub-test 1: two RLC SDUs on RB5 having the same content as sent by SS.

14.2.13.2 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 40 ms TTI

14.2.13.2.1 Conformance requirement

See clause 14.2.4.1.

14.2.13.2.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.13 for the 40 ms TTI case.

14.2.13.2.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RL	C		
TM RI	_C		
Tra	ansmission RLC discard		
	CHOICE SDU Discard Mode		
	Timer based no explicit		
	Timer_discard	100ms	
Se	gmentation indication	FALSE	
Downlink	RLC		
TM RI	_C		
Se	gmentation indication	FALSE	
NOTE:	Timer based discard without explicit sig	nalling is used in uplink to	
secure that the UE will be able to return data for the case when the			
UE test loop function will not deliver all the SDUs in one and the			
	same TTI .		

Uplink TFS:

	TFI	RB5 (64 kbps)	рссн
TFS	TF0, bits	0x640	0x148
15	TF1, bits	4x640	1x148

Uplink TFCS:

TFCI	(RB5, DCCH)		
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF0, TF1)		
UL_TFC3	(TF1, TF1)		

Downlink TFS:

	TFI	RB5 (64 kbps)	рссн
TFS	TF0, bits	0x640	0x148
	TF1, bits	4x640	1x148

Downlink TFCS:

TFCI	(RB5, DCCH)	
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF0, TF1)	
DL_TFC3	(TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 640	RB5: 4x640
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs.					

See 14.1.1 for test procedure.

14.2.13.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (4x640).
- 3. At step 15 the UE shall return
 - for sub-test 1: four RLC SDUs on RB5 having the same content as sent by SS.

14.2.14 Conversational / unknown / UL:32 DL:32 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.14.1 Conversational / unknown / UL:32 DL:32 kbps / CS RAB / 20 ms TTI

- 14.2.14.1.1 Conformance requirement
- See 14.2.4.1.
- 14.2.14.1.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.14 for the 20 ms TTI case.

14.2.14.1.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RLC	
TM RLC	
Segmentation indication	FALSE
Downlink RLC	
TM RLC	
Segmentation indication	FALSE

Uplink TFS:

	TFI	RB5 (32 kbps)	DCCH
TFS	TF0, bits	0x640	0x148
11-3	TF1, bits	1x640	1x148

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF0, TF1)	
UL_TFC3	(TF1, TF1)	

Downlink TFS:

	TFI	RB5 (32 kbps)	DCCH
TFS	TF0, bits	0x640	0x148
15	TF1, bits	1x640	1x148

Downlink TFCS:

TFCI	(RB5	, DCCH)
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF0, TF1)	
DL_TFC3	(TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 640	RB5: 640
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE)Us.	

See 14.1.1 for test procedure.

14.2.14.1.4 Test requirements

1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.

- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x640).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS.

14.2.14.2 Conversational / unknown / UL:32 DL:32 kbps / CS RAB / 40 ms TTI

14.2.14.2.1 Conformance requirement

See 14.2.4.1.

14.2.14.2.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.14 for the 40 ms TTI case.

14.2.14.2.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RLC	
TM RLC	
Transmission RLC discard	
CHOICE SDU Discard Mode	
Timer based no explicit	
Timer_discard	100ms
Segmentation indication	FALSE
Downlink RLC	
TM RLC	
Segmentation indication	FALSE
NOTE: Timer based discard without explicit sig	gnalling is used in uplink to
secure that the UE will be able to return	n data for the case when the
UE test loop function will not deliver all	the SDUs in one and the
same TTI .	

Uplink TFS:

	TFI	RB5 (32 kbps)	рссн
TFS	TF0, bits	0x640	0x148
15	TF1, bits	2x640	1x148

Uplink TFCS:

TFCI		(RB5, DCCH)	
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF0, TF1)		
UL_TFC3	(TF1, TF1)		

Downlink TFS:

	TFI	RB5 (32 kbps)	DCCH
TFS	TF0, bits	0x640	0x148
15	TF1, bits	2x640	1x148

Downlink TFCS:

TFCI	(RB5, DCCH)	
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF0, TF1)	
DL_TFC3	(TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)	
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 640	RB5: 2x640	
NOTE:							

See 14.1.1 for test procedure.

14.2.14.2.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (2x640).
- 3. At step 15 the UE shall return
 - for sub-test 1: two RLC SDUs on RB5 having the same content as sent by SS.

14.2.15 Streaming / unknown / UL:14.4/DL:14.4 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.15.1 Conformance requirement

See 14.2.4.1.

14.2.15.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.15.

14.2.15.3 Method of test

Uplink TFS:

	TFI	RB5 (14.4 kbps)	DCCH
TFS	TF0, bits	0x576	0x148
113	TF1, bits	1x576	1x148

Uplink TFCS:

TFCI		(RB5, DCCH)	
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF0, TF1)		
UL_TFC3	(TF1, TF1)		

Downlink TFS:

	TFI	RB5 (14.4 kbps)	DCCH
TFS	TF0, bits	0x576	0x148
11-3	TF1, bits	1x576	1x148

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF0, TF1)
DL_TFC3	(TF1, TF1)

Sub-tests:

	Under test	Under test			(bits) (note)	(note)
1 C	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 576	RB5: 576

See 14.1.1 for test procedure.

14.2.15.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x576).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS.

14.2.16 Streaming / unknown / UL:28.8/DL:28.8 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.16.1 Conformance requirement

See 14.2.4.1.

14.2.16.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.16.

14.2.16.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RL	С			
TM RI	_C			
Tra	ansmission RLC discard			
	CHOICE SDU Discard Mode			
	Timer based no explicit			
	Timer_discard	100ms		
Se	gmentation indication	FALSE		
Downlink	RLC			
TM RI	_C			
Se	gmentation indication	FALSE		
NOTE:	NOTE: Timer based discard without explicit signalling is used in uplink to			
	secure that the UE will be able to return data for the case when the			
	UE test loop function will not deliver all	the SDUs in one and the		
	same TTI .			

Uplink TFS:

	TFI	RB5 (28.8 kbps)	DCCH
	TF0, bits	0x576	0x148
TFS	TF1, bits	1x576	1x148
	TF2, bits	2x576	N/A

Uplink TFCS:

TFCI		(RB5, DCCH)	
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF2, TF0)		
UL_TFC3	(TF0, TF1)		
UL_TFC4	(TF1, TF1)		
UL_TFC5	(TF2, TF1)		

Downlink TFS:

		RB5 (28.8 kbps)	DCCH
	TF0, bits	0x576	0x148
TFS	TF1, bits	1x576	1x148
	TF2, bits	2x576	N/A

Downlink TFCS:

TFCI		(RB5, DCCH)
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF0, TF1)	
DL_TFC4	(TF1, TF1)	
DL_TFC5	(TF2, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 576	RB5: 576
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5,	RB5: 576	RB5: 2x576
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	•

See 14.1.1 for test procedure.

14.2.16.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x576).
 - for sub-test 2: RB5/TF2 (2x576).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS.
 - for sub-test 2: two RLC SDU on RB5 having the same content as sent by SS.

14.2.17 Streaming / unknown / UL:57.6/DL:57.6 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.17.1 Conformance requirement

See 14.2.4.1.

14.2.17.2 Test purpose

To verify radio bearer establishment and correct data transfer for reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.17.

14.2.17.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RLC			
TM RLC			
Transmission RLC discard			
CHOICE SDU Discard Mode			
Timer based no explicit			
Timer_discard	100ms		
Segmentation indication	FALSE		
Downlink RLC			
TM RLC			
Segmentation indication	FALSE		
NOTE: Timer based discard without explicit sig	gnalling is used in uplink to		
secure that the UE will be able to return data for the case when the			
UE test loop function will not deliver all the SDUs in one and the			
same TTI .			

Uplink TFS:

	TFI	RB5 (57.6 kbps)	DCCH
	TF0, bits	0x576	0x148
	TF1, bits	1x576	1x148
TFS	TF2, bits	2x576	N/A
	TF3, bits	3x576	N/A
	TF4, bits	4x576	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF0, TF1)
UL_TFC6	(TF1, TF1)
UL_TFC7	(TF2, TF1)
UL_TFC8	(TF3, TF1)
UL_TFC9	(TF4, TF1)

Downlink TFS:

	TFI	RB5 (57.6 kbps)	DCCH
	TF0, bits	0x576	0x148
	TF1, bits	1x576	1x148
TFS	TF2, bits	2x576	N/A
	TF3, bits	3x576	N/A
	TF4, bits	4x576	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF0, TF1)
DL_TFC6	(TF1, TF1)
DL_TFC7	(TF2, TF1)
DL_TFC8	(TF3, TF1)
DL_TFC9	(TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5,	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 576	RB5: 576
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 576	RB5: 2x576
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 576	RB5: 3x576
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 576	RB5: 4x576
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop		DUs.	1

See 14.1.1 for test procedure.

14.2.17.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x576).
 - for sub-test 2: RB5/TF2 (2x576).
 - for sub-test 3: RB5/TF3 (3x576).
 - for sub-test 4: RB5/TF4 (4x576).

- 3. At step 15 the UE shall return
 - for sub-test 1: one RLC SDU on RB5 having the same content as sent by SS.
 - for sub-test 2: two RLC SDU on RB5 having the same content as sent by SS.
 - for sub-test 3: three RLC SDU on RB5 having the same content as sent by SS.
 - for sub-test 4: four RLC SDU on RB5 having the same content as sent by SS.

14.2.18 Streaming / unknown / UL:0 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.18.1 Conformance requirement

See 14.2.4.1.

14.2.18.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.18.

To be able to test the downlink radio bearer using the UE loopback function for the reference radio bearer UL:0 DL: 64 kbps, the reference radio bearer configuration according to TS 34.108, clause 6.10.2.4.1.15.1 (Streaming/unknown/UL:14.4 kbps) is used in uplink.

14.2.18.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RL	С		
TM RL	_C		
Tra	ansmission RLC discard CHOICE SDU Discard Mode		
	Timer based no explicit		
	Timer_discard	100ms	
Se	gmentation indication	FALSE	
Downlink	RLC		
TM RL	.C		
Se	gmentation indication	FALSE	
NOTE:	NOTE: Timer based discard without explicit signalling is used in uplink to		
secure that the UE will be able to return data for the case when the			
UE test loop function will not deliver all the SDUs in one and the			
	same TTI .		

Uplink TFS:

	TFI	RB5 (14.4 kbps)	DCCH
TFS	TF0, bits	0x576	0x148
115	TF1, bits	1x576	1x148

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF0, TF1)	
UL_TFC3	(TF1, TF1)	

Release 5

Downlink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x320	0x148
	TF1, bits	1x320	1x148
TFS	TF2, bits	2x320	N/A
	TF3, bits	4x320	N/A
	TF4, bits	8x320	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)	
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF3, TF0)	
DL_TFC4	(TF4, TF0)	
DL_TFC5	(TF0, TF1)	
DL_TFC6	(TF1, TF1)	
DL_TFC7	(TF2, TF1)	
DL_TFC8	(TF3, TF1)	
DL_TFC9	(TF4, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note 1)	Test data size (bits) (note 1)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 576	RB5: 320
2	DL_TFC2	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 576	RB5: 2x320
3	DL_TFC3	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 576	RB5: 4x320
4	DL_TFC4	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 576	RB5: 8x320
NOTE	1: See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE)Us.	

See 14.1.1 for test procedure.

14.2.18.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 where the first 320 bits have the same content as the RLC SDU sent by the SS.
 - for sub-test 2 to 4: one or more RLC SDUs on RB5 where the first 320 bits have the same content as the RLC SDU sent by the SS.

14.2.19 Streaming / unknown / UL:64 DL:0 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.19.1 Conformance requirement

See 14.2.4.1.

14.2.19.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.19.

To be able to test the uplink radio bearer using the UE loopback function for the reference radio bearer UL:64 DL: 0 kbps, the reference radio bearer configuration according to TS 34.108, clause 6.10.2.4.1.15.2 (Streaming/unknown/DL:14.4 kbps) is used in downlink.

14.2.19.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS:

Uplink RLC	
TM RLC	
Segmentation indication	TRUE
Downlink RLC	
TM RLC	
Segmentation indication	TRUE

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x320	0x148
	TF1, bits	1x320	1x148
TFS	TF2, bits	2x320	N/A
	TF3, bits	4x320	N/A
	TF4, bits	8x320	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF0, TF1)
UL_TFC6	(TF1, TF1)
UL_TFC7	(TF2, TF1)
UL_TFC8	(TF3, TF1)
UL_TFC9	(TF4, TF1)

Downlink TFS:

	TFI	RB5 (14.4 kbps)	рссн
TFS	TF0, bits	0x576	0x148
15	TF1, bits	1x576	1x148

Downlink TFCS:

TFCI	(RB5, DCCH)		
DL_TFC0	(TF0, TF0)		
DL_TFC1	(TF1, TF0)		
DL_TFC2	(TF0, TF1)		
DL_TFC3	(TF1, TF1)		

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	(note 1) RB5: 320	(note 1) RB5: 576 (note 2)
2	DL_TFC1	UL_TFC2	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 640	RB5: 576 (note 3)
3	DL_TFC1	UL_TFC3	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1280	RB5: 576 (note 4)
4	DL_TFC1	UL_TFC4	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2560	RB5: 576 (note 5)
NOTE	 SS is using 320 bits of t SS is using 	a DL RLC SDU the test data. a DL RLC SDU	5.3.2.6.2 for details regarding loop J with 576 bits as test data (=DL RL J size of 576 bits as test data (=DL I	C PDU size for RLC PDU size f	DL/TF1). UE will for DL/TF1). UE v	vill return an

RLC SDU repeating the received DL RLC SDU two times (truncating the last one to fit the UL RLC SDU size of 640 bits).

NOTE 4: SS is using a DL RLC SDU size of 576 bits as test data (=DL RLC PDU size for DL/TF1). UE will return an RLC SDU repeating the received DL RLC SDU three times (truncating the last one to fit the UL RLC SDU size of 1280 bits).

NOTE 5: SS is using a DL RLC SDU size of 576 bits as test data (=DL RLC PDU size for DL/TF1). UE will return an RLC SDU repeating the received DL RLC SDU five times (truncating the last one to fit the UL RLC SDU size of 2560 bits).

See 14.1.1 for test procedure.

14.2.19.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x320).
 - for sub-test 2: RB5/TF2 (2x320).
 - for sub-test 3: RB5/TF3 (4x320).

- for sub-test 4: RB5/TF4 (8x320).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as the first 320 bits of the DL RLC SDU sent by the SS.
 - for sub-test 2: an RLC SDU on RB5 for which the first 576 bits are equal to the sent DL RLC SDU bit pattern and the remaining 64 bits are equal to the first 64 bits of the sent DL RLC SDU.
 - for sub-test 3: an RLC SDU on RB5 for which the first 1152 bits are equal to the sent DL RLC SDU bit pattern repeated twice and the remaining 128 bits are equal to the first 128 bits of the sent DL RLC SDU.
 - for sub-test 4: an RLC SDU on RB5 for which the first 2304 bits are equal to the sent DL RLC SDU bit pattern repeated four times and the remaining 256 bits are equal to the first 256 of the sent DL RLC SDU.
- 14.2.20 Void
- 14.2.21 Void
- 14.2.22 Void
- 14.2.23 Interactive or background / UL:32 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH
- 14.2.23.1 Interactive or background / UL:32 DL:8 kbps / PS RAB / (TC,10 ms TTI)
- 14.2.23.1.1 Conformance requirement
- See 14.2.4.1.
- 14.2.23.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23 for the turbo channel coding and uplink 10 ms TTI case.

14.2.23.1.3 Method of test

Uplink TFS:

	TFI	RB5 (32 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
113	TF1, bits	1x336	1x148

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF0, TF1)
UL_TFC3	(TF1, TF1)

Downlink TFS:

	TFI	RB5 (8 kbps)	рссн
TEO	TF0, bits	0x336	0x148
TFS	TF1, bits	1x336	1x148

Downlink TFCS:

TFCI		(RB5, DCCH)	
DL_TFC0	(TF0, TF0)		
DL_TFC1	(TF1, TF0)		
DL_TFC2	(TF0, TF1)		
DL_TFC3	(TF1, TF1)		

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 312	RB5: 312
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs.					

See 14.1.1 for test procedure.

14.2.23.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.23.2 Interactive or background / UL:32 DL:8 kbps / PS RAB / (TC, 20 ms TTI)

14.2.23.2.1 Conformance requirement

See 14.2.4.1.

14.2.23.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23 for the turbo channel coding and uplink 20 ms TTI case.

14.2.23.2.3 Method of test

Uplink TFS:

	TFI	RB5 (32 kbps)	DCCH
	TF0, bits	0x336	0x148
TFS	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A

Uplink TFCS:

TFCI		(RB5, DCCH)	
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF2, TF0)		
UL_TFC3	(TF0, TF1)		
UL_TFC4	(TF1, TF1)		
UL_TFC5	(TF2, TF1)		

Downlink TFS:

	TFI	RB5 (8 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF0, TF1)
DL_TFC3	(TF1, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 312	RB5: 312
2	DL_TFC1	UL_TFC2	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5	RB5: 632	RB5: 632
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	

See 14.1.1 for test procedure.

14.2.23.2.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).

- 3. At step 15 the UE shall return
 - for sub-test 1 and 2: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.23.3 Interactive or background / UL:32 DL:8 kbps / PS RAB / (CC, 10 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23 for the convolutional channel coding and uplink 10 ms TTI case.

See test case 14.2.23.1 for test procedure and test requirement.

14.2.23.4 Interactive or background / UL:32 DL:8 kbps / PS RAB / (CC, 20 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23 for the convolutional channel coding and uplink 20 ms TTI case.

See test case 14.2.23.2 for test procedure and test requirement.

14.2.23a Interactive or background / UL:8 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.23a.1 Conformance requirement

See 14.2.4.1.

14.2.23a.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23a.

14.2.23a.3 Method of test

See 14.1.1 for test procedure.

Uplink TFS:

	TFI	RB5 (8 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
113	TF1, bits	1x336	1x148

Uplink TFCS:

TFCI		(8 kbps RAB, DCCH)	
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF0, TF1)		
UL_TFC3	(TF1, TF1)		

Downlink TFS:

		RB5 (8 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
115	TF1, bits	1x336	1x148

Downlink TFCS:

TFCI	(8 kbps RAB, DCCH)	
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF0, TF1)	
DL_TFC3	(TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2,	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 312	RB5: 312
NOTE: RB5:						

14.2.23a.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.23b Interactive or background / UL:16 DL:16 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.23b.1 Conformance requirement

See clause 14.2.4.1.

14.2.23b.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23b.

14.2.23b.3 Method of test

Uplink TFS:

	TFI	RB5 (16 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF0, TF1)	
UL_TFC4	(TF1, TF1)	
UL_TFC5	(TF2, TF1)	

Downlink TFS:

	TFI	RB5 (32 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A

Downlink TFCS:

TFCI		(RB5, DCCH)
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF0, TF1)	
DL_TFC4	(TF1, TF1)	
DL_TFC5	(TF2, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, UL_TFC0 DL_TFC3,UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 312	RB5: 312
2	DL_TFC2	UL_TFC2	DL_TFC0, UL_TFC0 DL_TFC3 ,UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5	RB5: 632	RB5: 632
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB5: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).					

See 14.1.1 for test procedure.

14.2.23b.4 Test requirements

- 1. At step 10 the UE shall send a RADIO BEARER SETUP COMPLETE message.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).

- 3. At step 15 the UE shall return
 - for sub-test 1 and 2: an RLC SDU on RB5 having the same content as sent by the SS.

14.2.23c Interactive or background / UL:32 DL:32 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23c.

14.2.23c.1 Conformance requirement

See 14.2.4.1.

14.2.23c.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23c.

14.2.23c.3 Method of test

Uplink TFS:

	TFI	RB5 (32 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF0, TF1)
UL_TFC6	(TF1, TF1)
UL_TFC7	(TF2, TF1)
UL_TFC8	(TF3, TF1)
UL_TFC9	(TF4, TF1)

Downlink TFS:

	TFI	RB5 (32 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Release 5

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF0, TF1)
DL_TFC6	(TF1, TF1)
DL_TFC7	(TF2, TF1)
DL_TFC8	(TF3, TF1)
DL_TFC9	(TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, UL_TFC0	UL_TFC0,	RB5: 312	RB5: 312
			DL_TFC5,UL_TFC5	UL_TFC1,		
				UL_TFC5,		
				UL_TFC6		
2	DL_TFC2	UL_TFC2	DL_TFC0, UL_TFC0	UL_TFC0,	RB5: 632	RB5: 632
			DL_TFC5 ,UL_TFC5	UL_TFC2,		
				UL_TFC5,		
				UL_TFC7		
3	DL_TFC3	UL_TFC3	DL_TFC0, UL_TFC0	UL_TFC0,	RB5: 952	RB5: 952
			DL_TFC5 ,UL_TFC5	UL_TFC3,		
				UL_TFC5,		
				UL_TFC8		
4	DL_TFC4	UL_TFC4	DL_TFC0, UL_TFC0	UL_TFC0,	RB5: 1272	RB5: 1272
			DL_TFC5 ,UL_TFC5	UL_TFC4,		
				UL_TFC5,		
				UL_TFC9		
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. For RB5: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion					

See 14.1.1 for test procedure.

bit).

14.2.23c.4 Test requirements

- 1. At step 10 the UE shall send a RADIO BEARER SETUP COMPLETE message.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4: RB5/TF4 (4x336).

- 3. At step 15 the UE shall return
 - for sub-test 1 to 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.23d Interactive or background / UL:32 DL:32 kbps / PS RAB (20 ms TTI) + UL:3.4 DL:3.4 kbps SRBs for DCCH.

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.23d.

14.2.24 Interactive or background / UL:64 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.24.1 Interactive or background / UL:64 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH / TC
- 14.2.24.1.1 Conformance requirement

See 14.2.4.1.1.

14.2.24.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.24 for the downlink turbo coding case.

14.2.24.1.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI		(RB5, DCCH)
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (8 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
15	TF1, bits	1x336	1x148

Downlink TFCS:

TFCI	(RB5, DCCH)		
DL_TFC0	(TF0, TF0)		
DL_TFC1	(TF1, TF0)		
DL_TFC2	(TF0, TF1)		
DL_TFC3	(TF1, TF1)		

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312	RB5: 312
2	DL_TFC1	UL_TFC2	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632
3	DL_TFC1	UL_TFC3	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 952	RB5: 952
4	DL_TFC1	UL_TFC4	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 1272	RB5: 1272
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE	Us.	

See 14.1.1 for test procedure.

14.2.24.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.24.2 Interactive or background / UL:64 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH / CC

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.24 for the downlink convolutional channel coding case.

See test case 14.2.24.1 for test procedure and test requirement.

14.2.25 Interactive or background / UL:32 DL: 64 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.25.1 Interactive or background / UL:32 DL: 64 kbps / PS RAB / (TC, 10 ms TTI)
- 14.2.25.1.1 Conformance requirement
- See 14.2.4.1.
- 14.2.25.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.25 for the uplink turbo channel coding and 10 ms TTI case.

14.2.25.1.3 Method of test

Uplink TFS:

	TFI	RB5 (32 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
11-3	TF1, bits	1x336	1x148

Uplink TFCS:

TFCI	(RB5, DCCH)		
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF0, TF1)		
UL_TFC3	(TF1, TF1)		

Downlink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)	
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF3, TF0)	
DL_TFC4	(TF4, TF0)	
DL_TFC5	(TF0, TF1)	
DL_TFC6	(TF1, TF1)	
DL_TFC7	(TF2, TF1)	
DL_TFC8	(TF3, TF1)	
DL_TFC9	(TF4, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 312	RB5: 312
2	DL_TFC2	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 952	RB5: 952
4	DL_TFC4	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 1272	RB5: 1272

See 14.1.1 for test procedure.

14.2.25.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1 to 4: RB5/TF1 (1x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.25.2 Interactive or background / UL:32 DL: 64 kbps / PS RAB / (TC, 20 ms TTI)

14.2.25.2.1 Conformance requirement

See 14.2.4.1.

14.2.25.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.25 for the uplink turbo channel coding and 20 ms TTI case.

14.2.25.2.3 Method of test

Uplink TFS:

	TFI	RB5 (32 kbps)	DCCH
	TF0, bits	0x336	0x148
TFS	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A

Release 5

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF0, TF1)	
UL_TFC4	(TF1, TF1)	
UL_TFC5	(TF2, TF1)	

Downlink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF0, TF1)
DL_TFC6	(TF1, TF1)
DL_TFC7	(TF2, TF1)
DL_TFC8	(TF3, TF1)
DL_TFC9	(TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	(note) RB5: 312	(note) RB5: 312
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4	RB5: 952	RB5: 952
4	DL_TFC4	UL_TFC2	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC3	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5	RB5: 1272	RB5: 1272
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE)Us.	

See 14.1.1 for test procedure.

14.2.25.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF1 (1x336).
 - for sub-test 4: RB5/TF2 (2x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.25.3 Interactive or background / UL:32 DL:64 kbps / PS RAB / (CC, 10 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.25 for the uplink convolutional channel coding and 10 ms TTI case.

See test case 14.2.25.1 for test procedure and test requirement.

14.2.25.4 Interactive or background / UL:32 DL:64 kbps / PS RAB / (CC, 20 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.25 for the uplink convolutional channel coding and 20 ms TTI case.

See test case 14.2.25.2 for test procedure and test requirement.

14.2.26 Interactive or background / UL:64 DL: 64 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.26.1 Conformance requirement

See 14.2.4.1.

14.2.26.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.26.

14.2.26.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Release 5

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF0, TF1)
DL_TFC6	(TF1, TF1)
DL_TFC7	(TF2, TF1)
DL_TFC8	(TF3, TF1)
DL_TFC9	(TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5,	RB5: 312	RB5: 312
				UL_TFC6	DD5 000	DD5 000
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 952	RB5: 952
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 1272	RB5: 1272
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB5: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size paramater has been set to achieve verification of all test data sent by SS in downlink, i.e. UL RLC SDU size is set to nearest multiple of the payload size of the UL TF					

under test minus 8 bits (size of 7 bit length indicator and expansion bit) which is equal or bigger than the test

See 14.1.1 for test procedure.

data size.

14.2.26.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.27 Interactive or background / UL:64 DL:128 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.27.1 Conformance requirement

See 14.2.4.1.

14.2.27.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.27.

14.2.27.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	٦
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (128 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)	(RB5, DCCH)				
DL_TFC0	(TF0, TF0)					
DL_TFC1	(TF1, TF0)					
DL_TFC2	(TF2, TF0)					
DL_TFC3	(TF3, TF0)					
DL_TFC4	(TF4, TF0)					
DL_TFC5	(TF0, TF1)					
DL_TFC6	(TF1, TF1)					
DL_TFC7	(TF2, TF1)					
DL_TFC8	(TF3, TF1)					
DL_TFC9	(TF4, TF1)					

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)	
					(note)	(note)	
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312	RB5: 312	
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632	
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1912	RB5: 1272	
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2552	RB5: 2552	
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB5: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size paramater has been set to achieve verification of all test data sent by SS in downlink, i.e. UL RLC SDU size is set to nearest multiple of the payload size of the UL TF						

under test minus 8 bits (size of 7 bit length indicator and expansion bit) which is equal or bigger than the test

See 14.1.1 for test procedure.

data size.

14.2.27.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1, 2 and 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.
 - for sub-test 3: an RLC SDU on RB5 having the first 1272 bits equal to the content of the test data sent by the SS in downlink.

14.2.28 Interactive or background / UL:128 DL:128 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.28.1 Conformance requirement

See 14.2.4.1.

14.2.28.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.28.

14.2.28.3 Method of test

Uplink TFS:

	TFI	RB5 (128 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	(RB5, DCCH)			
UL_TFC0	(TF0, TF0)				
UL_TFC1	(TF1, TF0)				
UL_TFC2	(TF2, TF0)				
UL_TFC3	(TF3, TF0)				
UL_TFC4	(TF4, TF0)				
UL_TFC5	(TF0, TF1)				
UL_TFC6	(TF1, TF1)				
UL_TFC7	(TF2, TF1)				
UL_TFC8	(TF3, TF1)				
UL_TFC9	(TF4, TF1)				

Downlink TFS:

	TFI	RB5 (128 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)	(RB5, DCCH)				
DL_TFC0	(TF0, TF0)					
DL_TFC1	(TF1, TF0)					
DL_TFC2	(TF2, TF0)					
DL_TFC3	(TF3, TF0)					
DL_TFC4	(TF4, TF0)					
DL_TFC5	(TF0, TF1)					
DL_TFC6	(TF1, TF1)					
DL_TFC7	(TF2, TF1)					
DL_TFC8	(TF3, TF1)					
DL_TFC9	(TF4, TF1)					

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312	RB5: 312
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1272	RB5: 1272
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2552	RB5: 2552
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loop	back of RLC SE)Us.	

See 14.1.1 for test procedure.

14.2.28.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4: RB5/TF4 (8x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.29 Interactive or background / UL:64 DL:144 kbps / PS RAB + UL:3.4 DL: 3.4 kbps SRBs for DCCH

14.2.29.1 Conformance requirement

See 14.2.4.1.

14.2.29.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.29.

14.2.29.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (144 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
115	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	9x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF0, TF1)
DL_TFC7	(TF1, TF1)
DL_TFC8	(TF2, TF1)
DL_TFC9	(TF3, TF1)
DL_TFC10	(TF4, TF1)
DL_TFC11	(TF5, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 312	RB5: 312
			UL_TFC5	UL_TFC1,		
				UL_TFC5,		
_				UL_TFC6	DD5 000	
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 632	RB5: 632
			UL_TFC5	UL_TFC2,		
				UL_TFC5,		
-				UL_TFC7		
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 1912	RB5: 1272
			UL_TFC5	UL_TFC3,		
				UL_TFC5,		
-				UL_TFC8	DD5 0550	DD5 0550
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 2552	RB5: 2552
			UL_TFC5	UL_TFC4,		
				UL_TFC5,		
5				UL_TFC9	DDC: 0070	DDC: 0070
5	DL_TFC5	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 2872	RB5: 2872
			UL_TFC5	UL_TFC3,		
				UL_TFC5, UL_TFC8		
NOTE:	Soo TS 24	100 [10] dougo	5 2 2 6 2 for dotails regarding leanhs			
	RB5: Test of indicator ar data sent b	data size has be nd expansion bi y SS in downlin	 5.3.2.6.2 for details regarding loopbageen set to the payload size of the DL⁻ t). The UL RLC SDU size paramater hik, i.e. UL RLC SDU size is set to nea ze of 7 bit length indicator and expansional set to the set to	TF under test r nas been set to rest multiple of	ninus 8 bits (siz achieve verific the payload siz	ation of all test e of the UL TF

See 14.1.1 for test procedure.

data size.

14.2.29.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4: RB5/TF4 (8x336).
 - for sub-test 5: RB5/TF3 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 4 and 5: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.
 - for sub-test 3: an RLC SDU on RB5 having the first 1272 bits equal to the content of the test data sent by the SS in downlink.

14.2.30 Interactive or background / UL:144 DL:144 kbps / PS RAB + UL:3.4 DL: 3.4 kbps SRBs for DCCH

14.2.30.1 Conformance requirement

See 14.2.4.1.

Release 5

2061

14.2.30.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.30.

14.2.30.3 Method of test

Uplink TFS:

	TFI	RB5 (144 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	9x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF5, TF0)	
UL_TFC6	(TF0, TF1)	
UL_TFC7	(TF1, TF1)	
UL_TFC8	(TF2, TF1)	
UL_TFC9	(TF3, TF1)	
UL_TFC10	(TF4, TF1)	
UL_TFC11	(TF5, TF1)	

Downlink TFS:

	TFI	RB5 (144 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	9x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF0, TF1)
DL_TFC7	(TF1, TF1)
DL_TFC8	(TF2, TF1)
DL_TFC9	(TF3, TF1)
DL_TFC10	(TF4, TF1)
DL_TFC11	(TF5, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 312	RB5: 312
			UL_TFC6	UL_TFC1,		
				UL_TFC6,		
				UL_TFC7		
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 632	RB5: 632
			UL_TFC6	UL_TFC2,		
				UL_TFC6,		
				UL_TFC8		
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 1272	RB5: 1272
			UL_TFC6	UL_TFC3,		
				UL_TFC6,		
				UL_TFC9		
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 2552	RB5: 2552
			UL_TFC6	UL_TFC4,		
				UL_TFC6,		
				UL_TFC10		
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC6, UL_TFC0,	UL_TFC0,	RB5: 2872	RB5: 2872
			UL_TFC6	UL_TFC5,		
				UL_TFC6,		
				UL_TFC11		
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loopba	ack of RLC SD	Us.	

See 14.1.1 for test procedure.

14.2.30.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4: RB5/TF4 (8x336).
 - for sub-test 5: RB5/TF5 (9x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 5: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.31 Interactive or background / UL:64 DL:256 kbps / PS RAB + UL:3.4 DL: 3.4 kbps SRBs for DCCH

- 14.2.31.1 Interactive or background / UL:64 DL:256 kbps / PS RAB + UL:3.4 DL: 3.4 kbps SRBs for DCCH/ 10 ms TTI
- 14.2.31.1.1 Conformance requirement

See 14.2.4.1.

Release 5

2063

14.2.31.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.31 for the downlink 10 ms TTI case.

14.2.31.1.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(RB	5, DCCH)
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (256 kbps, 10ms)	рссн
TFS	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)	
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF3, TF0)	
DL_TFC4	(TF4, TF0)	
DL_TFC5	(TF0, TF1)	
DL_TFC6	(TF1, TF1)	
DL_TFC7	(TF2, TF1)	
DL_TFC8	(TF3, TF1)	
DL_TFC9	(TF4, TF1)	

Sub-tests:

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)		
	Under test	Under test			(bits)			
					(note)	(note)		
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC5, UL_TFC0,	UL_TFC0,	RB5: 312	RB5: 312		
			UL_TFC5	UL_TFC1,				
				UL_TFC5,				
				UL_TFC6				
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC5, UL_TFC0,	UL_TFC0,	RB5: 632	RB5: 632		
			UL_TFC5	UL_TFC2,				
				UL_TFC5,				
				UL_TFC7				
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC5, UL_TFC0,	UL_TFC0,	RB5: 1912	RB5: 1272		
			UL_TFC5	UL_TFC3,				
				UL_TFC5,				
				UL_TFC8				
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC5, UL_TFC0,	UL_TFC0,	RB5: 2552	RB5: 2552		
			UL_TFC5	UL_TFC4,				
				UL_TFC5,				
				UL_TFC9				
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs.							
	RB5: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length							
	indicator and expansion bit). The UL RLC SDU size paramater has been set to achieve verification of all test							
			k, i.e. UL RLC SDU size is set to nea					
	under test r	minus 8 bits (siz	ze of 7 bit length indicator and expans	sion bit) which i	s equal or bigge	er than the test		

See 14.1.1 for test procedure.

data size.

14.2.31.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1, 2 and 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.
 - for sub-test 3: an RLC SDU on RB5 having the first 1272 bits equal to the content of the test data sent by the SS in downlink.

14.2.31.2 Interactive or background / UL:64 DL:256 kbps / PS RAB / 20 ms TTI

14.2.31.2.1 Conformance requirement

See 14.2.4.1.

14.2.31.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.31 for the downlink 20 ms TTI case.

14.2.31.2.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (256 kbps, 20ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
TFS	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A
	TF6, bits	16x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF0, TF1)
DL_TFC8	(TF1, TF1)
DL_TFC9	(TF2, TF1)
DL_TFC10	(TF3, TF1)
DL_TFC11	(TF4, TF1)
DL_TFC12	(TF5, TF1)
DL_TFC13	(TF6, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC7, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1,	(note) RB5: 312	(note) RB5: 312
				UL_TFC5, UL_TFC6		
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC7, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC7, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1912	RB5: 1272
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC7, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2552	RB5: 2552
5	DL_TFC5	UL_TFC4	DL_TFC0, DL_TFC7, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 3832	RB5: 3832
6 NOTE:	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC7, UL_TFC0, UL_TFC5 9 5.3.2.6.2 for details regarding loopba	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 5112	RB5: 5112

See 14.1.1 for test procedure.

14.2.31.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4 to 6: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 6: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.32 Interactive or background / UL:64 DL:384 kbps / PS RAB + UL:3.4 DL: 3.4 kbps SRBs for DCCH

14.2.32.1 Interactive or background / UL:64 DL:384 kbps / PS RAB / 10 ms TTI

- 14.2.32.1.1 Conformance requirement
- See 14.2.4.1.

14.2.32.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.32 for the 10 ms TTI case.

14.2.32.1.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(1	RB5, DCCH)
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (384 kbps, 10ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
115	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A

Downlink TFCS:

TFCI		(RB5, DCCH)
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF3, TF0)	
DL_TFC4	(TF4, TF0)	
DL_TFC5	(TF5, TF0)	
DL_TFC6	(TF0, TF1)	
DL_TFC7	(TF1, TF1)	
DL_TFC8	(TF2, TF1)	
DL_TFC9	(TF3, TF1)	
DL_TFC10	(TF4, TF1)	
DL_TFC11	(TF5, TF1)	

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)	
					(note)	(note)	
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312	RB5: 312	
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632	
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1912	RB5: 1272	
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2552	RB5: 2552	
5	DL_TFC5	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 3832	RB5: 3832	
NOTE:	UL_TFC9 NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB5: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set to achieve verification of all test data sent by SS in downlink, i.e. UL RLC SDU size is set to nearest multiple of the payload size of the UL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit) which is equal or bigger than the test						

See 14.1.1 for test procedure.

data size.

14.2.32.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4 and 5: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 4 and 5: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.
 - for sub-test 3: an RLC SDU on RB5 having the first 1272 bits equal to the content of the test data sent by the SS in downlink.

14.2.32.2 Interactive or background / UL:64 DL:384 kbps / PS RAB / 20 ms TTI

14.2.32.2.1 Conformance requirement

See 14.2.4.1.

14.2.32.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.32 for the 20 ms TTI case.

14.2.32.2.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI		(RB5, DCCH)
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

	TFI	RB5 (384 kbps, 20ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
TFS	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A
	TF6, bits	16x336	N/A
	TF7, bits	20x336	N/A
	TF8, bits	24x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF0, TF1)
DL_TFC10	(TF1, TF1)
DL_TFC11	(TF2, TF1)
DL_TFC12	(TF3, TF1)
DL_TFC13	(TF4, TF1)
DL_TFC14	(TF5, TF1)
DL_TFC15	(TF6, TF1)
DL_TFC16	(TF7, TF1)
DL_TFC17	(TF8, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312	RB5: 312
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1912	RB5: 1272
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2552	RB5: 2552
5	DL_TFC5	UL_TFC4	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 3832	RB5: 3832
6	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 5112	RB5: 5112
7	DL_TFC7	UL_TFC4	DL_TFC0, DL_TFC9, , UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 6392	RB5: 6392
8	DL_TFC8	UL_TFC4	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 7672	RB5: 7672
NOTE:	See 13 34.	TOB [TO] CIAUSE	5.3.2.6.2 for details regarding loopba	AUK ULKLU SD	05.	

See 14.1.1 for test procedure.

14.2.32.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4 to 8: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 8: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.33 Interactive or background / UL:128 DL:384 kbps / PS RAB + UL:3.4 DL: 3.4 kbps SRBs for DCCH

- 14.2.33.1 Interactive or background / UL:128 DL:384 kbps / PS RAB / 10 ms TTI
- 14.2.33.1.1 Conformance requirement
- See 14.2.4.1.

14.2.33.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.33 for the 10 ms TTI case.

14.2.33.1.3 Method of test

Uplink TFS:

	TFI	RB5 (128 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF0, TF1)
UL_TFC6	(TF1, TF1)
UL_TFC7	(TF2, TF1)
UL_TFC8	(TF3, TF1)
UL_TFC9	(TF4, TF1)

Downlink TFS:

	TFI	RB5 (384 kbps, 10ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
115	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A

Downlink TFCS:

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF0, TF1)
DL_TFC7	(TF1, TF1)
DL_TFC8	(TF2, TF1)
DL_TFC9	(TF3, TF1)
DL_TFC10	(TF4, TF1)
DL_TFC11	(TF5, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312	RB5: 312
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1272	RB5: 1272
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2552	RB5: 2552
5	DL_TFC5	UL_TFC3	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 3832	RB5: 3832
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. The UL RLC SDU size have been choosen such that the UE will return all data received in downlink and that the UL RLC SDU will fill up the uplink transport format set under test over one or several transmission time intervals.					

See 14.1.1 for test procedure.

14.2.33.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4 and 5: RB5/TF4 (8x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 5: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.33.2 Interactive or background / UL:128 DL:384 kbps / PS RAB / 20 ms TTI

14.2.33.2.1 Conformance requirement

See 14.2.4.1.

14.2.33.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.33 for the 20 ms TTI case.

14.2.33.2.3 Method of test

Uplink TFS:

	TFI	RB5 (128 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF0, TF1)
UL_TFC6	(TF1, TF1)
UL_TFC7	(TF2, TF1)
UL_TFC8	(TF3, TF1)
UL_TFC9	(TF4, TF1)

Downlink TFS:

	TFI	RB5 (384 kbps, 20ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
TFS	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A
	TF6, bits	16x336	N/A
	TF7, bits	20x336	N/A
	TF8, bits	24x336	N/A

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF0, TF1)
DL_TFC10	(TF1, TF1)
DL_TFC11	(TF2, TF1)
DL_TFC12	(TF3, TF1)
DL_TFC13	(TF4, TF1)
DL_TFC14	(TF5, TF1)
DL_TFC15	(TF6, TF1)
DL_TFC16	(TF7, TF1)
DL_TFC17	(TF8, TF1)

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	onder test	onder test			(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312	RB5: 312
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632	RB5: 632
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 1272	RB5: 1272
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 2552	RB5: 2552
5	DL_TFC5	UL_TFC3	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5,	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 3832	RB5: 3832
6	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 5112	RB5: 5112
7	DL_TFC7	UL_TFC3	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 6392	RB5: 6392
8	DL_TFC8	UL_TFC4	DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4 UL_TFC5, UL_TFC9	RB5: 7672	RB5: 7672
NOTE:	The UL RLC and that the	C SDU size hav	5.3.2.6.2 for details regard ve been choosen such that will fill up the uplink transpo	the UE will return	all data receive	

See 14.1.1 for test procedure.

14.2.33.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4 to 8: RB5/TF4 (8x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 8: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.34 Interactive or background / UL:384 DL:384 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.34.1 Interactive or background / UL:384 DL:384 kbps / PS RAB / 10 ms TTI
- 14.2.34.1.1 Conformance requirement

See 14.2.4.1.

14.2.34.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.34 for the 10 ms TTI case.

14.2.34.1.3 Method of test

Uplink TFS:

	TFI	RB5 (384 kbps, 10ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
15	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF5, TF0)	
UL_TFC6	(TF0, TF1)	
UL_TFC7	(TF1, TF1)	
UL_TFC8	(TF2, TF1)	
UL_TFC9	(TF3, TF1)	
UL_TFC10	(TF4, TF1)	
UL_TFC11	(TF5, TF1)	

	TFI	RB5 (384 kbps, 10ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
113	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A

Downlink TFCS:

TEOL	
TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF0, TF1)
DL_TFC7	(TF1, TF1)
DL_TFC8	(TF2, TF1)
DL_TFC9	(TF3, TF1)
DL_TFC10	(TF4, TF1)
DL_TFC11	(TF5, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC7, UL_TFC0, UL_TFC7	UL_TFC0, UL_TFC1,	RB5: 312	RB5: 312
				UL_TFC7, UL_TFC8		
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC7, UL_TFC0,	UL_TFC0,	RB5: 632	RB5: 632
			UL_TFC7	UL_TFC2,		
				UL_TFC7,		
				UL_TFC9		
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC7, UL_TFC0,	UL_TFC0,	RB5: 1272	RB5: 1272
			UL_TFC7	UL_TFC3,		
				UL_TFC7,		
				UL_TFC10		
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC7, UL_TFC0,	UL_TFC0,	RB5: 2552	RB5: 2552
			UL_TFC7	UL_TFC4,		
				UL_TFC7,		
				UL_TFC11		
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC7, UL_TFC0,	UL_TFC0,	RB5: 3832	RB5: 3832
			UL_TFC7	UL_TFC5,		
				UL_TFC7,		
				UL_TFC12		
NOTE:	See TS 34.	109 [10] clause	5.3.2.6.2 for details regarding loopba	ack of RLC SD	Us.	

See 14.1.1 for test procedure.

14.2.34.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4: RB5/TF4 (8x336).
 - for sub-test 5: RB5/TF4 (12x336).

- 3. At step 15 the UE shall return
 - for sub-test 1 to 5: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.34.2 Interactive or background / UL:384 DL:384 kbps / PS RAB / 20 ms TTI

14.2.34.2.1 Conformance requirement

See 14.2.4.1.

14.2.34.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.34. for the 20 ms TTI case

14.2.34.2.3 Method of test

Uplink TFS:

	TFI	RB5 (384 kbps, 20ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
TFS	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A
	TF6, bits	16x336	N/A
	TF7, bits	20x336	N/A
	TF8, bits	24x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF5, TF0)
UL_TFC6	(TF6, TF0)
UL_TFC7	(TF7, TF0)
UL_TFC8	(TF8, TF0)
UL_TFC9	(TF0, TF1)
UL_TFC10	(TF1, TF1)
UL_TFC11	(TF2, TF1)
UL_TFC12	(TF3, TF1)
UL_TFC13	(TF4, TF1)
UL_TFC14	(TF5, TF1)
UL_TFC15	(TF6, TF1)
UL_TFC16	(TF7, TF1)
UL_TFC17	(TF8, TF1)

Downlink TFS:

	TFI	RB5 (384 kbps, 20ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
TFS	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A
	TF6, bits	16x336	N/A
	TF7, bits	20x336	N/A
	TF8, bits	24x336	N/A

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF0, TF1)
DL_TFC10	(TF1, TF1)
DL_TFC11	(TF2, TF1)
DL_TFC12	(TF3, TF1)
DL_TFC13	(TF4, TF1)
DL_TFC14	(TF5, TF1)
DL_TFC15	(TF6, TF1)
DL_TFC16	(TF7, TF1)
DL_TFC17	(TF8, TF1)

Under test Under test Under test (bits) (note) (note) (note) (note) 1 DL_TFC1 UL_TFC1 DL_TFC1 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC10 UL_TFC0, UL_TFC1, UL_TFC10 RB5: 312 RB5: 312 RB5: 312 2 DL_TFC2 UL_TFC2 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC10 UL_TFC0, UL_TFC9, UL_TFC10 UL_TFC2, UL_TFC9, UL_TFC10 RB5: 632 RB5: 632 3 DL_TFC3 UL_TFC3 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC10, UL_TFC4, UL_TFC4 UL_TFC3 RB5: 1272 RB5: 1272 4 DL_TFC4 UL_TFC5 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC13 UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC9, UL_TFC7, UL_TFC7, UL_TFC7, UL_TFC7, UL_TFC7, UL_TFC7, UL_TFC7, UL_TFC0, UL_TFC9, UL_TFC7, UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC7, UL_TFC6, UL_TFC7, UL_TFC7, UL_TFC6, UL_TFC7, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TF	Sub-	Downlink	Uplink	Implicitely tested	Restricted UL	UL RLC SDU	Test data size
Image: Construct of the system of t	test	TFCS	TFCS		TFCIs	size (bits)	(bits)
1 DL_TFC1 UL_TFC1 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC1, UL_TFC1, UL_TFC2, UL_TFC2 UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC2, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC1, UL_TFC1 RB5: 312 RB5: 312 RB5: 312 2 DL_TFC2 UL_TFC2 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC0, UL_TFC0, UL_TFC1, UL_TFC1, UL_TFC3, UL_TFC0, UL_TFC0, UL_TFC1 RB5: 632 RB5: 632 3 DL_TFC3 UL_TFC3 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC3, UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC0, UL_TFC4, UL_TFC3, UL_TFC0, UL_TFC4, UL_TFC3, UL_TFC0, UL_TFC4, UL_TFC3, UL_TFC0, UL_TFC4, UL_TFC3, UL_TFC0, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC0, UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC0, UL_TFC3, UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC3		Under test	Under lest			· · ·	(note)
Image: Second state of the second state of	1	DL TFC1	UL TFC1	DL TFC0, DL TFC9,	UL TFC0,		
Image: Constraint of the second sec				UL_TFC0, UL_TFC9	UL_TFC1,		
2 DL_TFC2 UL_TFC2 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC2, UL_TFC2, UL_TFC9, UL_TFC9, UL_TFC1 RB5: 632 RB5: 632 3 DL_TFC3 UL_TFC3 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC3, UL_TFC3, UL_TFC9, UL_TFC1 UL_TFC1 RB5: 1272 RB5: 1272 4 DL_TFC4 UL_TFC4 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC1 UL_TFC4, UL_TFC1 RB5: 2552 RB5: 2552 5 DL_TFC5 UL_TFC6, DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC4, UL_TFC1 UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC9, UL_TFC6, UL_TFC9, UL_TFC14 RB5: 3832 RB5: 3832 6 DL_TFC6 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC15 UL_TFC6, UL_TFC6, UL_TFC9, UL_TFC6, UL_TFC9, UL_TFC6, UL_TFC6, UL_TFC9, UL_TFC15 RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC0, DL_TFC0, DL_TFC9, UL_TFC16 UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC0, UL_TFC9, UL_TFC16 UL_TFC16 RB5: 7672 RB5: 7672							
Image: Second state UL_TFC0, UL_TFC9 UL_TFC2, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC12 RB5: 1272 RB5: 1272 3 DL_TFC4 UL_TFC3 DL_TFC0, UL_TFC9, UL_TFC9, UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC12 RB5: 1272 RB5: 1272 4 DL_TFC4 UL_TFC4 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC12 UL_TFC4, UL_TFC3 RB5: 2552 RB5: 2552 5 DL_TFC5 UL_TFC5 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC13 UL_TFC0, UL_TFC3 RB5: 3832 RB5: 3832 6 DL_TFC6 UL_TFC0 DL_TFC0, UL_TFC9, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC14 UL_TFC15 RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC15 UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC8 DL_TFC0, DL_TFC9, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC4, UL_TF							
Jule Jule <th< td=""><td>2</td><td>DL_TFC2</td><td>UL_TFC2</td><td></td><td></td><td>RB5: 632</td><td>RB5: 632</td></th<>	2	DL_TFC2	UL_TFC2			RB5: 632	RB5: 632
Image: Constraint of the second sec				UL_IFC0, UL_IFC9	_ ·		
3 DL_TFC3 UL_TFC3 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9 UL_TFC0, UL_TFC9, UL_TFC3, UL_TFC9, UL_TFC9, UL_TFC9, RB5: 1272 RB5: 1272 4 DL_TFC4 UL_TFC4 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9 UL_TFC0, UL_TFC9, UL_TFC4, UL_TFC9, UL_TFC9, RB5: 2552 RB5: 2552 5 DL_TFC5 UL_TFC5 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC13 UL_TFC6, UL_TFC9, UL_TFC14 RB5: 3832 RB5: 3832 6 DL_TFC6 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC15 UL_TFC6, UL_TFC16 RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC16 UL_TFC6, UL_TFC9, UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC16 UL_TFC0, UL_TFC9, UL_TFC16 RB5: 7672 RB5: 7672					_ ·		
Image: Second system UL_TFC3 UL_TFC3, UL_TFC3, UL_TFC9, UL_TFC9, UL_TFC12 4 DL_TFC4 UL_TFC4 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC0, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC9, UL_TFC4 RB5: 3832 RB5: 3832 5 DL_TFC6 UL_TFC6 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC0, UL_TFC6, UL_TFC9, UL_TFC14 RB5: 3832 RB5: 3832 6 DL_TFC6 UL_TFC0, DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC0, UL_TFC	3	DI TEC3	LII TEC3		_	RB5: 1272	RB5: 1272
UL_TFC9, UL_TFC12 UL_TFC9, UL_TFC12 4 DL_TFC4 UL_TFC4 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC13 RB5: 2552 RB5: 2552 5 DL_TFC5 UL_TFC5 DL_TFC5 UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC13 UL_TFC13 6 DL_TFC6 UL_TFC6 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC14 UL_TFC0, UL_TFC14 RB5: 5112 7 DL_TFC7 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC15 UL_TFC16 8 DL_TFC8 UL_TFC8 DL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC16 RB5: 7672 8 DL_TFC8 UL_TFC8 DL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC16 RB5: 7672	5	DL_1103	01_1103		_ ·	1105. 1272	1105. 1272
4 DL_TFC4 UL_TFC4 DL_TFC4 DL_TFC4 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC4, UL_TFC4, UL_TFC9, UL_TFC13 RB5: 2552 RB5: 2552 5 DL_TFC5 UL_TFC5 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC13 UL_TFC13 RB5: 3832 RB5: 3832 6 DL_TFC6 UL_TFC6 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC14 UL_TFC14 RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC16 UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC8 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC16 UL_TFC16 RB5: 7672 RB5: 7672					,		
4 DL_TFC4 UL_TFC4 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC0, UL_TFC4, UL_TFC4, UL_TFC3, UL_TFC3 RB5: 2552 RB5: 2552 5 DL_TFC5 UL_TFC5 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC4, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC4 UL_TFC13 RB5: 3832 RB5: 3832 6 DL_TFC6 UL_TFC6 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC14 UL_TFC14 RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC15 UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC8 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC16 UL_TFC16 RB5: 7672 RB5: 7672 8 DL_TFC8 UL_TFC8 DL_TFC0, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC17 UL_TFC17 RB5: 7672 RB5: 7672							
Image: Second	4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC9,		RB5: 2552	RB5: 2552
Image: Second				UL_TFC0, UL_TFC9			
5 DL_TFC5 UL_TFC5 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC14 RB5: 3832 RB5: 3832 6 DL_TFC6 UL_TFC6 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC15 UL_TFC14 RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC16 UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC8 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC16 UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC16 RB5: 7672 RB5: 7672							
Image: Second system UL_TFC8 UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC14 6 DL_TFC6 UL_TFC6 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC6, UL_TFC9, UL_TFC9, UL_TFC15 RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC7 DL_TFC0, DL_TFC0, UL_TFC9, UL_TFC9, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC7, UL_TFC0, UL_TFC0, UL_TFC9, UL_TFC16 UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC8 DL_TFC0, UL_TFC0, UL_TFC9, UL_TFC0,							
Image: Construct of the system Image:	5	DL_TFC5	UL_TFC5			RB5: 3832	RB5: 3832
Image: constraint of the state of				UL_IFC0, UL_IFC9	_ ·		
6 DL_TFC6 UL_TFC6 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC15 UL_TFC0, RB5: 5112 RB5: 5112 7 DL_TFC7 UL_TFC7 DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC16 UL_TFC3, UL_TFC9, UL_TFC9, UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC17 UL_TFC16 RB5: 7672 RB5: 7672							
Image: Second system Image: Second system <td< td=""><td>6</td><td>DI TEC6</td><td>LII TEC6</td><td></td><td></td><td>RB5: 5112</td><td>RB5: 5112</td></td<>	6	DI TEC6	LII TEC6			RB5: 5112	RB5: 5112
UL_TFC9, UL_TFC15 UL_TFC7 UL_TFC7 DL_TFC7 DL_TFC0, DL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC16 UL_TFC7, UL_TFC9, UL_TFC16 RB5: 6392 RB5: 6392 8 DL_TFC8 UL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC17 UL_TFC17 RB5: 7672 RB5: 7672	0	DL_11 00	02_1100			1120.0112	1100.0112
UL_TFC15 UL_TFC7 UL_TFC7 DL_TFC7 DL_TFC0, DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC16 UL_TFC16 RB5: 6392 RB5: 7672 RB5: 7672 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>							
UL_TFC0, UL_TFC9 UL_TFC7, UL_TFC9, UL_TFC9, UL_TFC16 8 DL_TFC8 UL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC17 RB5: 7672							
UL_TFC9, UL_TFC16 UL_TFC9, UL_TFC16 8 DL_TFC8 UL_TFC0, DL_TFC9, UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC17 RB5: 7672 RB5: 7672	7	DL_TFC7	UL_TFC7			RB5: 6392	RB5: 6392
UL_TFC16 UL_TFC36 8 DL_TFC8 UL_TFC0, DL_TFC9, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC8, UL_TFC9, UL_TFC9, UL_TFC17 RB5: 7672 RB5: 7672				UL_TFC0, UL_TFC9			
8 DL_TFC8 UL_TFC8 DL_TFC0, DL_TFC9, UL_TFC0, RB5: 7672 RB5: 7672 UL_TFC0, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC9, UL_TFC17							
UL_TFC0, UL_TFC9 UL_TFC9, UL_TFC17							
UL_TFC9, UL_TFC17	8	DL_TFC8	UL_TFC8	_ , _ ,	_ /	RB5: 7672	RB5: 7672
UL_TFC17				UL_1FC0, UL_1FC9	_ ·		
					_ ·		
	NOTE	·	100 [10] dougo	5 2 2 6 2 for dotails regard			

See 14.1.1 for test procedure.

14.2.34.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4: RB5/TF4 (8x336).
 - for sub-test 5: RB5/TF5 (12x336).
 - for sub-test 6: RB5/TF6 (16x336).
 - for sub-test 7: RB5/TF7 (20x336).
 - for sub-test 8: RB5/TF8 (24x336).

- 3. At step 15 the UE shall return
 - for sub-test 1 to 8: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.35 Interactive or background / UL:64 DL:2048 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.35.1 Interactive or background / UL:64 DL:2048 kbps / PS RAB / 10 ms TTI

14.2.35.1.1 Conformance requirement

See 14.2.4.1.

14.2.35.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.35 for the 10 ms TTI case.

14.2.35.1.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
TFS	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF0, TF1)
UL_TFC6	(TF1, TF1)
UL_TFC7	(TF2, TF1)
UL_TFC8	(TF3, TF1)
UL_TFC9	(TF4, TF1)

Downlink TFS:

	TFI	RB5 (2048 kbps, 10ms)	рссн
	TF0, bits	0x656	0x148
	TF1, bits	1x656	1x148
	TF2, bits	2x656	N/A
	TF3, bits	4x656	N/A
	TF4, bits	8x656	N/A
TFS	TF5, bits	12x656	N/A
	TF6, bits	16x656	N/A
	TF7, bits	20x656	N/A
	TF8, bits	24x656	N/A
	TF9, bits	28x656	N/A
	TF10, bits	32x656	N/A

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF9, TF0)
DL_TFC10	(TF10, TF0)
DL_TFC11	(TF0, TF1)
DL_TFC12	(TF1, TF1)
DL_TFC13	(TF2, TF1)
DL_TFC14	(TF3, TF1)
DL_TFC15	(TF4, TF1)
DL_TFC16	(TF5, TF1)
DL_TFC17	(TF6, TF1)
DL_TFC18	(TF7, TF1)
DL_TFC19	(TF8, TF1)
DL_TFC20	(TF9, TF1)
DL_TFC21	(TF10, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 632	RB5: 632
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 1272	RB5: 1272
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 2872	RB5: 2552
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 5112	RB5: 5112
5	DL_TFC5	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 7672	RB5: 7672
6	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 10232	RB5: 10232
7	DL_TFC7	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 12792	RB5: 12792
8	DL_TFC8	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 15352	RB5: 15352
9	DL_TFC9	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 17912	RB5: 17912
10	DL_TFC10	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 20472	RB5: 20472
NOTE:	The UL RL	C SDU size h	se 5.3.2.6.2 for details regarding loopb ave been choosen such that the UE wi up the uplink transport format set unde	ack of RLC SD ill return all data	received in dov	

See 14.1.1 for test procedure.

14.2.35.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).

- for sub-test 3: RB5/TF3 (3x336).
- for sub-test 4 to 10: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 10: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.35.2 Interactive or background / UL:64 DL:2048 kbps / PS RAB / 20 ms TTI

14.2.35.2.1 Conformance requirement

See 14.2.4.1.

14.2.35.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.35 for the 20 ms TTI case.

14.2.35.2.3 Method of test

Uplink TFS:

	TFI	RB5 (64 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	3x336	N/A
	TF4, bits	4x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (2048 kbps, 10ms)	DCCH
	TF0, bits	0x656	0x148
	TF1, bits	1x656	1x148
	TF2, bits	2x656	N/A
	TF3, bits	4x656	N/A
	TF4, bits	8x656	N/A
	TF5, bits	12x656	N/A
	TF6, bits	16x656	N/A
	TF7, bits	20x656	N/A
	TF8, bits	24x656	N/A
TFS	TF9, bits	28x656	N/A
	TF10, bits	32x656	N/A
	TF11, bits	36x656	N/A
	TF12, bits	40x656	N/A
	TF13, bits	44x656	N/A
	TF14, bits	48x656	N/A
	TF15, bits	52x656	N/A
	TF16, bits	56x656	N/A
	TF17, bits	60x656	N/A
	TF18, bits	64x656	N/A

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF9, TF0)
DL_TFC10	(TF10, TF0)
DL_TFC11	(TF11, TF0)
DL_TFC12	(TF12, TF0)
DL_TFC13	(TF13, TF0)
DL_TFC14	(TF14, TF0)
DL_TFC15	(TF15, TF0)
DL_TFC16	(TF16, TF0)
DL_TFC17	(TF17, TF0)
DL_TFC18	(TF18, TF0)
DL_TFC19	(TF0, TF1)
DL_TFC20	(TF1, TF1)
DL_TFC21	(TF2, TF1)
DL_TFC22	(TF3, TF1)
DL_TFC23	(TF4, TF1)
DL_TFC24	(TF5, TF1)
DL_TFC25	(TF6, TF1)
DL_TFC26	(TF7, TF1)
DL_TFC27	(TF8, TF1)
DL_TFC28	(TF9, TF1)
DL_TFC29	(TF10, TF1)
DL_TFC30	(TF11, TF1)
DL_TFC31	(TF12, TF1)
DL_TFC32	(TF13, TF1)
DL_TFC33	(TF14, TF1)
DL_TFC34	(TF15, TF1)

TFCI	(RB5, DCCH)
DL_TFC35	(TF16, TF1)
DL_TFC36	(TF17, TF1)
DL_TFC37	(TF18, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 632	RB5: 632
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 1272	RB5: 1272
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 2872	RB5: 2552
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 5112	RB5: 5112
5	DL_TFC5	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 7672	RB5: 7672
6	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 10232	RB5: 10232
7	DL_TFC7	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 12792	RB5: 12792
8	DL_TFC8	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 15352	RB5: 15352
9	DL_TFC9	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 17912	RB5: 17912
10	DL_TFC10	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 20472	RB5: 20472
11	DL_TFC11	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 23032	RB5: 23032
12	DL_TFC12	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 25592	RB5: 25592
13	DL_TFC13	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4 UL_TFC5, UL_TFC9	RB5: 28152	RB5: 28152

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)		
					(note)	(note)		
14	DL_TFC14	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 30712	RB5: 30712		
15	DL_TFC15	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 33272	RB5: 33272		
16	DL_TFC16	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 35832	RB5: 35832		
17	DL_TFC17	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4 UL_TFC5, UL_TFC9	RB5: 38392	RB5: 38392		
18	DL_TFC18	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 40952	RB5: 40952		
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. The UL RLC SDU size have been choosen such that the UE will return all data received in downlink and that the UL RLC SDU will fill up the uplink transport format set under test over one or several transmission time intervals.							

See 14.1.1 for test procedure.

14.2.35.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (3x336).
 - for sub-test 4 to 18: RB5/TF4 (4x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 18: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.36 Interactive or background / UL:128 DL:2048 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.36.1 Interactive or background / UL:128 DL:2048 kbps / PS RAB / 10 ms TTI

- 14.2.36.1.1 Conformance requirement
- See 14.2.4.1.

14.2.36.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.36 for the 10 ms TTI case.

14.2.36.1.3 Method of test

Uplink TFS:

	TFI	RB5 (128 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 (2048 kbps, 10ms)	DCCH
	TF0, bits	0x656	0x148
	TF1, bits	1x656	1x148
	TF2, bits	2x656	N/A
	TF3, bits	4x656	N/A
	TF4, bits	8x656	N/A
TFS	TF5, bits	12x656	N/A
	TF6, bits	16x656	N/A
	TF7, bits	20x656	N/A
	TF8, bits	24x656	N/A
	TF9, bits	28x656	N/A
	TF10, bits	32x656	N/A

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF9, TF0)
DL_TFC10	(TF10, TF0)

TFCI	(RB5, DCCH)
DL_TFC11	(TF0, TF1)
DL_TFC12	(TF1, TF1)
DL_TFC13	(TF2, TF1)
DL_TFC14	(TF3, TF1)
DL_TFC15	(TF4, TF1)
DL_TFC16	(TF5, TF1)
DL_TFC17	(TF6, TF1)
DL_TFC18	(TF7, TF1)
DL_TFC19	(TF8, TF1)
DL_TFC20	(TF9, TF1)
DL_TFC21	(TF10, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 632	RB5: 632
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 1272	RB5: 1272
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 2552	RB5: 2552
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 5112	RB5: 5112
5	DL_TFC5	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 7672	RB5: 7672
6	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 10232	RB5: 10232
7	DL_TFC7	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 12792	RB5: 12792
8	DL_TFC8	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 15352	RB5: 15352
9	DL_TFC9	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 17912	RB5: 17912
10	DL_TFC10	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 20472	RB5: 20472
NOTE:	The UL RL	C SDU size h	se 5.3.2.6.2 for details regarding loopba ave been choosen such that the UE wi up the uplink transport format set unde	Il return all data	a received in dov	

See 14.1.1 for test procedure.

14.2.36.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4 to 10: RB5/TF4 (8x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 10: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.36.2 Interactive or background / UL:128 DL:2048 kbps / PS RAB / 20 ms TTI

14.2.36.2.1 Conformance requirement

See 14.2.4.1.

14.2.36.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.36 for the 20 ms TTI case.

14.2.36.2.3 Method of test

Uplink TFS:

	TFI	RB5 (128 kbps)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF0, TF1)
UL_TFC6	(TF1, TF1)
UL_TFC7	(TF2, TF1)
UL_TFC8	(TF3, TF1)
UL_TFC9	(TF4, TF1)

Downlink TFS:

	TFI	RB5 (2048 kbps, 10ms)	DCCH
	TF0, bits	0x656	0x148
	TF1, bits	1x656	1x148
	TF2, bits	2x656	N/A
	TF3, bits	4x656	N/A
	TF4, bits	8x656	N/A
	TF5, bits	12x656	N/A
	TF6, bits	16x656	N/A
	TF7, bits	20x656	N/A
	TF8, bits	24x656	N/A
TFS	TF9, bits	28x656	N/A
	TF10, bits	32x656	N/A
	TF11, bits	36x656	N/A
	TF12, bits	40x656	N/A
	TF13, bits	44x656	N/A
	TF14, bits	48x656	N/A
	TF15, bits	52x656	N/A
	TF16, bits	56x656	N/A
	TF17, bits	60x656	N/A
	TF18, bits	64x656	N/A

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF9, TF0)
DL_TFC10	(TF10, TF0)
DL_TFC11	(TF11, TF0)
DL_TFC12	(TF12, TF0)
DL_TFC13	(TF13, TF0)
DL_TFC14	(TF14, TF0)
DL_TFC15	(TF15, TF0)
DL_TFC16	(TF16, TF0)
DL_TFC17	(TF17, TF0)
DL_TFC18	(TF18, TF0)
DL_TFC19	(TF0, TF1)
DL_TFC20	(TF1, TF1)
DL_TFC21	(TF2, TF1)
DL_TFC22	(TF3, TF1)
DL_TFC23	(TF4, TF1)
DL_TFC24	(TF5, TF1)
DL_TFC25	(TF6, TF1)
DL_TFC26	(TF7, TF1)
DL_TFC27	(TF8, TF1)
DL_TFC28	(TF9, TF1)
DL_TFC29	(TF10, TF1)
DL_TFC30	(TF11, TF1)
DL_TFC31	(TF12, TF1)
DL_TFC32	(TF13, TF1)
DL_TFC33	(TF14, TF1)
DL_TFC34	(TF15, TF1)

TFCI	(RB5, DCCH)
DL_TFC35	(TF16, TF1)
DL_TFC36	(TF17, TF1)
DL_TFC37	(TF18, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	(note) RB5: 632	(note) RB5: 632
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 1272	RB5: 1272
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 2552	RB5: 2552
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 5112	RB5: 5112
5	DL_TFC5	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 7672	RB5: 7672
6	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 10232	RB5: 10232
7	DL_TFC7	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 12792	RB5: 12792
8	DL_TFC8	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 15352	RB5: 15352
9	DL_TFC9	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 17912	RB5: 17912
10	DL_TFC10	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 20472	RB5: 20472
11	DL_TFC11	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 23032	RB5: 23032
12	DL_TFC12	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 25592	RB5: 25592
13	DL_TFC13	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 28152	RB5: 28152

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
14	DL_TFC14	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4,	RB5: 30712	RB5: 30712
				UL_TFC5, UL_TFC9		
15	DL_TFC15	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4,	RB5: 33272	RB5: 33272
				UL_TFC5, UL_TFC9		
16	DL_TFC16	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 35832	RB5: 35832
17	DL_TFC17	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 38392	RB5: 38392
18	DL_TFC18	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 40952	RB5: 40952
NOTE:	The UL RL	C SDU size h	se 5.3.2.6.2 for details regarding loopba ave been choosen such that the UE will up the uplink transport format set under	l return all data	a received in dow	

See 14.1.1 for test procedure.

intervals.

14.2.36.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4 to 18: RB5/TF4 (8x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 18: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.37 Interactive or background / UL:384 DL:2048 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.37.1 Interactive or background / UL:384 DL:2048 kbps / PS RAB / 10 ms TTI

- 14.2.37.1.1 Conformance requirement
- See 14.2.4.1.

14.2.37.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.37 for the 10 ms TTI case.

14.2.37.1.3 Method of test

Uplink TFS:

	TFI	RB5 (384 kbps, 10ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
TFS	TF2, bits	2x336	N/A
113	TF3, bits	4x336	N/A
	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF5, TF0)	
UL_TFC6	(TF0, TF1)	
UL_TFC7	(TF1, TF1)	
UL_TFC8	(TF2, TF1)	
UL_TFC9	(TF3, TF1)	
UL_TFC10	(TF4, TF1)	
UL_TFC11	(TF5, TF1)	

Downlink TFS:

	TFI	RB5 (2048 kbps, 10ms)	DCCH
	TF0, bits	0x656	0x148
	TF1, bits	1x656	1x148
	TF2, bits	2x656	N/A
	TF3, bits	4x656	N/A
	TF4, bits	8x656	N/A
TFS	TF5, bits	12x656	N/A
	TF6, bits	16x656	N/A
	TF7, bits	20x656	N/A
	TF8, bits	24x656	N/A
	TF9, bits	28x656	N/A
	TF10, bits	32x656	N/A

TFCI		(RB5, DCCH)
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF3, TF0)	
DL_TFC4	(TF4, TF0)	
DL_TFC5	(TF5, TF0)	
DL_TFC6	(TF6, TF0)	
DL_TFC7	(TF7, TF0)	

TFCI	(RB5, DCCH)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF9, TF0)
DL_TFC10	(TF10, TF0)
DL_TFC11	(TF0, TF1)
DL_TFC12	(TF1, TF1)
DL_TFC13	(TF2, TF1)
DL_TFC14	(TF3, TF1)
DL_TFC15	(TF4, TF1)
DL_TFC16	(TF5, TF1)
DL_TFC17	(TF6, TF1)
DL_TFC18	(TF7, TF1)
DL_TFC19	(TF8, TF1)
DL_TFC20	(TF9, TF1)
DL_TFC21	(TF10, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
					(note)	(note)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7	RB5: 632	RB5: 632
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 1272	RB5: 1272
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 2552	RB5: 2552
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC4, UL_TFC6, UL_TFC10	RB5: 5112	RB5: 5112
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC11	RB5: 7672	RB5: 7672
6	DL_TFC6	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC4, UL_TFC6, UL_TFC10	RB5: 10232	RB5: 10232
7	DL_TFC7	UL_TFC3	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 12792	RB5: 12792
8	DL_TFC8	UL_TFC5	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC11	RB5: 15352	RB5: 15352
9	DL_TFC9	UL_TFC3	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 17912	RB5: 17912
10	DL_TFC10	UL_TFC4	DL_TFC0, DL_TFC11, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC4, UL_TFC6, UL_TFC10	RB5: 20472	RB5: 20472
NOTE:	The UL RL	C SDU size h	se 5.3.2.6.2 for details regarding loopba ave been choosen such that the UE wil up the uplink transport format set under	I return all data	received in dov	

See 14.1.1 for test procedure.

14.2.37.1.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).

- for sub-test 4: RB5/TF3 (8x336).
- for sub-test 5 to 10: RB5/TF4 (12x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 10: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.37.2 Interactive or background / UL:384 DL:2048 kbps / PS RAB / 20 ms TTI

14.2.37.2.1 Conformance requirement

See 14.2.4.1.

14.2.37.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.37 for the 20 ms TTI case.

14.2.37.2.3 Method of test

Uplink TFS:

	TFI	RB5 (384 kbps, 20ms)	DCCH
	TF0, bits	0x336	0x148
	TF1, bits	1x336	1x148
	TF2, bits	2x336	N/A
	TF3, bits	4x336	N/A
TFS	TF4, bits	8x336	N/A
	TF5, bits	12x336	N/A
	TF6, bits	16x336	N/A
	TF7, bits	20x336	N/A
	TF8, bits	24x336	N/A

Uplink TFCS:

TFCI	(RB5, DCCH)
UL_TFC0	(TF0, TF0)
UL_TFC1	(TF1, TF0)
UL_TFC2	(TF2, TF0)
UL_TFC3	(TF3, TF0)
UL_TFC4	(TF4, TF0)
UL_TFC5	(TF5, TF0)
UL_TFC6	(TF6, TF0)
UL_TFC7	(TF7, TF0)
UL_TFC8	(TF8, TF0)
UL_TFC9	(TF0, TF1)
UL_TFC10	(TF1, TF1)
UL_TFC11	(TF2, TF1)
UL_TFC12	(TF3, TF1)
UL_TFC13	(TF4, TF1)
UL_TFC14	(TF5, TF1)
UL_TFC15	(TF6, TF1)
UL_TFC16	(TF7, TF1)
UL_TFC17	(TF8, TF1)

	TFI	RB5 (2048 kbps, 10ms)	DCCH
	TF0, bits	0x656	0x148
	TF1, bits	1x656	1x148
	TF2, bits	2x656	N/A
	TF3, bits	4x656	N/A
	TF4, bits	8x656	N/A
	TF5, bits	12x656	N/A
	TF6, bits	16x656	N/A
	TF7, bits	20x656	N/A
	TF8, bits	24x656	N/A
TFS	TF9, bits	28x656	N/A
	TF10, bits	32x656	N/A
	TF11, bits	36x656	N/A
	TF12, bits	40x656	N/A
	TF13, bits	44x656	N/A
	TF14, bits	48x656	N/A
	TF15, bits	52x656	N/A
	TF16, bits	56x656	N/A
	TF17, bits	60x656	N/A
	TF18, bits	64x656	N/A

TFCI	(RB5, DCCH)
DL_TFC0	(TF0, TF0)
DL_TFC1	(TF1, TF0)
DL_TFC2	(TF2, TF0)
DL_TFC3	(TF3, TF0)
DL_TFC4	(TF4, TF0)
DL_TFC5	(TF5, TF0)
DL_TFC6	(TF6, TF0)
DL_TFC7	(TF7, TF0)
DL_TFC8	(TF8, TF0)
DL_TFC9	(TF9, TF0)
DL_TFC10	(TF10, TF0)
DL_TFC11	(TF11, TF0)
DL_TFC12	(TF12, TF0)
DL_TFC13	(TF13, TF0)
DL_TFC14	(TF14, TF0)
DL_TFC15	(TF15, TF0)
DL_TFC16	(TF16, TF0)
DL_TFC17	(TF17, TF0)
DL_TFC18	(TF18, TF0)
DL_TFC19	(TF0, TF1)
DL_TFC20	(TF1, TF1)
DL_TFC21	(TF2, TF1)
DL_TFC22	(TF3, TF1)
DL_TFC23	(TF4, TF1)
DL_TFC24	(TF5, TF1)
DL_TFC25	(TF6, TF1)
DL_TFC26	(TF7, TF1)
DL_TFC27	(TF8, TF1)
DL_TFC28	(TF9, TF1)
DL_TFC29	(TF10, TF1)
DL_TFC30	(TF11, TF1)
DL_TFC31	(TF12, TF1)
DL_TFC32	(TF13, TF1)
DL_TFC33	(TF14, TF1)
DL_TFC34	(TF15, TF1)
DL_TFC35	(TF16, TF1)
DL_TFC36	(TF17, TF1)

TFCI		(RB5, DCCH)
DL TFC37	(TF18, TF1)	

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10	(note) RB5: 632	(note) RB5: 632
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11	RB5: 1272	RB5: 1272
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC3, UL_TFC9, UL_TFC12	RB5: 2552	RB5: 2552
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC4, UL_TFC9, UL_TFC13	RB5: 5112	RB5: 5112
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC5, UL_TFC9, UL_TFC14	RB5: 7672	RB5: 7672
6	DL_TFC6	UL_TFC6	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC6, UL_TFC9, UL_TFC15	RB5: 10232	RB5: 10232
7	DL_TFC7	UL_TFC7	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC7, UL_TFC9, UL_TFC16	RB5: 12792	RB5: 12792
8	DL_TFC8	UL_TFC8	DL_TFC0, DL_TFC19, , UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC8, UL_TFC9, UL_TFC17	RB5: 15352	RB5: 15352
9	DL_TFC9	UL_TFC3	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC3, UL_TFC9, UL_TFC12	RB5: 17912	RB5: 17912
10	DL_TFC10	UL_TFC6	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC6, UL_TFC9, UL_TFC15	RB5: 20472	RB5: 20472
11	DL_TFC11	UL_TFC3	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC3, UL_TFC9, UL_TFC12	RB5: 23032	RB5: 23032
12	DL_TFC12	UL_TFC7	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC7, UL_TFC9, UL_TFC16	RB5: 25592	RB5: 25592
13	DL_TFC13	UL_TFC3	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC3, UL_TFC9, UL_TFC12	RB5: 28152	RB5: 28152
14	DL_TFC14	UL_TFC8	DL_TFC0, DL_TFC19, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC8, UL_TFC9, UL_TFC17	RB5: 30712	RB5: 30712

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under Test	Under test			(bits)	
	Test	lesi			(note)	(note)
15	DL_TFC15	UL_TFC3	DL_TFC0, DL_TFC19, UL_TFC0,	UL_TFC0,	RB5: 33272	RB5: 33272
			UL_TFC9	UL_TFC3,		
				UL_TFC9,		
				UL_TFC12		
16	DL_TFC16	UL_TFC4	DL_TFC0, DL_TFC19, UL_TFC0,	UL_TFC0,	RB5: 35832	RB5: 35832
			UL_TFC9	UL_TFC4,		
				UL_TFC9,		
				UL_TFC13		
17	DL_TFC17	UL_TFC7	DL_TFC0, DL_TFC19, UL_TFC0,	UL_TFC0,	RB5: 38392	RB5: 38392
			UL_TFC9	UL_TFC7,		
				UL_TFC9,		
				UL_TFC16		
18	DL_TFC18	UL_TFC6	DL_TFC0, DL_TFC19, , UL_TFC0,	UL_TFC0,	RB5: 40952	RB5: 40952
			UL_TFC9	UL_TFC6,		
				UL_TFC9,		
				UL_TFC15		
NOTE:			se 5.3.2.6.2 for details regarding loopba			
			ave been choosen such that the UE will			
		SDU will fill	up the uplink transport format set under	test over one	or several trans	mission time
	intervals.					

See 14.1.1 for test procedure.

14.2.37.2.4 Test requirements

See 14.1.1 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15 the UE transmitted transport format shall be
 - for sub-test 1: RB5/TF1 (1x336).
 - for sub-test 2: RB5/TF2 (2x336).
 - for sub-test 3: RB5/TF3 (4x336).
 - for sub-test 4: RB5/TF4 (8336).
 - for sub-test 5: RB5/TF5 (12x336).
 - for sub-test 6: RB5/TF6 (16x336).
 - for sub-test 7: RB5/TF7 (20x336).
 - for sub-test 8 to 18: RB5/TF4 (24x336).
- 3. At step 15 the UE shall return
 - for sub-test 1 to 18: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS.

14.2.38 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.38.1 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:8 kbps / PS RAB / (TC, 20 ms TTI)
- 14.2.38.1.1 Conformance requirement

See 14.2.4.1.

14.2.38.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38 for the turbo channel coding and 20 ms TTI case.

14.2.38.1.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

_	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (32 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	2x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1, TF1)
UL_TFC11	(TF2, TF1, TF1, TF0, TF1)
UL_TFC12	(TF0, TF0, TF1, TF1)
UL_TFC13	(TF1, TF0, TF0, TF1, TF1)
UL_TFC14	(TF2, TF1, TF1, TF1, TF1)
UL_TFC15	(TF0, TF0, TF0, TF2, TF1)
UL_TFC16	(TF1, TF0, TF0, TF2, TF1)
UL_TFC17	(TF2, TF1, TF1, TF2, TF1)

		RB5	RB6	RB7	RB8	DCCH
		(RAB subflow #1)	(RAB subflow #2)	(RAB subflow #3)	(8 kbps)	
	TF0, bits	1x0	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Release 5

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF1, TF0, TF1)
DL_TFC9	(TF0, TF0, TF0, TF1, TF1)
DL_TFC10	(TF1, TF0, TF0, TF1, TF1)
DL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Sub-	Downlink	Uplink	Implicitely tested	Restricted UL	UL RLC	Test data size		
test	TFCS	TFCS		TFCIs	SDU size	(bits)		
	under	Under test			(bits)			
	test				(note)	(note)		
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 39	RB5: 39		
	DL_TFC7	UL_TFC10	UL_TFC0, UL_TFC9	UL_TFC1,	RB6: 103	RB6: No data		
				UL_TFC9,	RB7: 60	RB7: No data		
0				UL_TFC10	RB8: 632	RB8: No data		
2	DL_TFC2 DL_TFC8	UL_TFC2 UL_TFC11	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC2,	RB5: 81 RB6: 103	RB5: 81 RB6: 103		
	DL_IFC0		0L_1FC0, 0L_1FC9	UL_TFC9,	RB7: 60	RB0: 103 RB7: 60		
				UL_TFC11	RB8: 632	RB8: No data		
3	DL TFC3	UL_TFC3	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 39	RB5: No data		
U	DL_TFC9	UL_TFC12	UL_TFC0, UL_TFC9	UL_TFC3,	RB6: 103	RB6: No data		
	52_11 00	02_11 012		UL_TFC9,	RB7: 60	RB7: No data		
				UL_TFC12	RB8: 632	RB8: 312		
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 39	RB5: 39		
	DL_TFC10	UL_TFC13	UL_TFC0, UL_TFC9	UL_TFC1,	RB6: 103	RB6: No data		
				UL_TFC3,	RB7: 60	RB7: No data		
				UL_TFC4,	RB8: 632	RB8: 312		
				UL_TFC9,				
				UL_TFC10,				
				UL_TFC12,				
-				UL_TFC13				
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 81 RB6: 103	RB5: 81		
	DL_TFC11	UL_TFC14	UL_TFC0, UL_TFC9	UL_TFC2, UL_TFC3,	RB6: 103 RB7: 60	RB6: 103 RB7: 60		
				UL_TFC5,	RB8: 632	RB7: 00 RB8: 312		
				UL_TFC9,	ND0. 052	ND0. 312		
				UL_TFC11,				
				UL_TFC12,				
				UL_TFC14				
6	DL_TFC3	UL_TFC6	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 39	RB5: No data		
	DL_TFC9	UL_TFC15	UL_TFC0, UL_TFC9	UL_TFC6,	RB6: 103	RB6: No data		
				UL_TFC9,	RB7: 60	RB7: No data		
				UL_TFC15	RB8: 1272	RB8: 312		
7	DL_TFC4	UL_TFC7	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 39	RB5: 39		
	DL_TFC10	UL_TFC16	UL_TFC0, UL_TFC9	UL_TFC1,	RB6: 103	RB6: No data		
				UL_TFC6,	RB7: 60	RB7: No data		
				UL_TFC7, UL_TFC9,	RB8: 1272	RB8: 312		
				UL_TFC10,				
				UL_TFC15,				
				UL_TFC16				
8	DL_TFC5	UL_TFC8	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 81	RB5: 81		
	DL_TFC11	UL_TFC17	UL_TFC0, UL_TFC9	UL_TFC2,	RB6: 103	RB6: 103		
				UL_TFC6,	RB7: 60	RB7: 60		
				UL_TFC8,	RB8: 1272	RB8: 312		
				UL_TFC9,				
				UL_TFC11,				
				UL_TFC15,				
NOTE		100 [10] alar		UL_TFC17		<u> </u>		
NOTE:			se 5.3.2.6.2 for details regar been set to DL TFS size un			th indicator and		
			link TTI for RB8 is 20 ms wi					
			nk the size of the uplink RLC					
two subsequent TTIs, i.e. UL RLC SDU SIZE has been set to two times the uplink TFS size minus 8 (the size of a 7 bit length indicator and expansion bit)								

14.2.38.1.4 Test requirements

of a 7 bit length indicator and expansion bit).

See 14.1.2 for definition of step 10 and step 15.

1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.

- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub test.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3 and 6: an RLC SDU on RB8 having the first 312 bits equal to the content sent by the SS in the downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4 and 7: an RLC SDU on RB5 having the same content as sent by SS and on RB8 having the first 312 bits equal to the content sent by the SS in the downlink; and no data shall be received on RB6 and RB7.
 - for sub-test 5 and 8: an RLC SDU on RB5, RB6, RB7 having the same content as sent by SS and on RB8 having the first 312 bits equal to the content sent by the SS in the downlink.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38.2 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:8 kbps / PS RAB / (TC, 10 ms TTI)

14.2.38.2.1 Conformance requirement

See 14.2.4.1.

14.2.38.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38 for the turbo channel coding and 10 ms TTI case.

14.2.38.2.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (32 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF1, TF0, TF1)
UL_TFC9	(TF0, TF0, TF0, TF1, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1, TF1)
UL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (8 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

2105

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF1, TF0, TF1)
DL_TFC9	(TF0, TF0, TF0, TF1, TF1)
DL_TFC10	(TF1, TF0, TF0, TF1, TF1)
DL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Sub-tests:

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)	
	test				(note)	(note)	
1	DL_TFC1 DL_TFC7	UL_TFC1 UL_TFC7	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: No data	
2	DL_TFC2 DL_TFC8	UL_TFC2 UL_TFC8	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: No data	
3	DL_TFC3 DL_TFC9	UL_TFC3 UL_TFC9	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 312	
4	DL_TFC4 DL_TFC10	UL_TFC4 UL_TFC10	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 312	
5	DL_TFC5 DL_TFC11	UL_TFC5 UL_TFC11	DL_TFC0, DL_TFC6, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 312	
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the uplink TTI for RB8 is 10 ms while the downlink TTI is 40 ms then, to achieve continous data transmission in uplink the size of the uplink RLC SDU has been set such that it will be transmitted over four tsubsequent TTIs, i.e. UL RLC SDU SIZE has been set to four times the uplink TFS size minus 8 (the size of a 7 bit length indicator and expansion bit)							

a 7 bit length indicator and expansion bit).

Release 5

14.2.38.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub test.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the first 312 bits equal to the content sent by the SS in the downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB5 having the same content as sent by SS and on RB8 having the first 312 bits equal to the content sent by the SS in the downlink; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB5, RB6, RB7 having the same content as sent by SS and on RB8 having the first 312 bits equal to the content sent by the SS in the downlink.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38.3 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:8 kbps / PS RAB / (CC, 20 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38 for the convolutional channel coding and 20 ms TTI case.

See test case 14.2.38.1 for test procedure and test requirement.

14.2.38.4 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:8 kbps / PS RAB / (CC, 10 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38 for the convolutional channel coding and 10 ms TTI case.

See test case 14.2.38.2 for test procedure and test requirement.

14.2.38a Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:0 DL:0 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38a.1 Conformance requirement

See 14.2.4.1.

14.2.38a.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38a.

14.2.38a.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

1

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (0 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	N/A	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1)
UL_TFC4	(TF1, TF0, TF0, TF0, TF1)
UL_TFC5	(TF2, TF1, TF1, TF0, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (0 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	N/A	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1)
DL_TFC4	(TF1, TF0, TF0, TF0, TF1)
DL_TFC5	(TF2, TF1, TF1, TF0, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL TFC1	UL TFC1	DL TFC0, DL TFC3,	UL_TFC0,	RB5: 39	RB5: 39
•	DL_TFC4	UL_TFC4	UL_TFC0, UL_TFC3,	UL_TFC1,	RB6: 103	RB6: No data
				UL_TFC3,	RB7: 60	RB7: No data
				UL_TFC4	RB8: 0	RB8: No data
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC3,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC5	UL_TFC5	UL_TFC0, UL_TFC3,	UL_TFC2,	RB6: 103	RB6: 103
				UL_TFC3,	RB7: 60	RB7: 60
				UL_TFC5	RB8: 0	RB8: No data
NOTE:	See TS 34	.109 [10] claus	se 5.3.2.6.2 for details reg	garding loopback of	of RLC SDUs.	

14.2.38a.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return

- for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
- for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38b Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:8 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38b.1 Conformance requirement

See 14.2.4.1.

14.2.38b.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38b.

14.2.38b.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

_	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (8 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF1, TF0, TF1)
UL_TFC9	(TF0, TF0, TF0, TF1, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1, TF1)
UL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (8 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF1, TF0, TF1)
DL_TFC9	(TF0, TF0, TF0, TF1, TF1)
DL_TFC10	(TF1, TF0, TF0, TF1, TF1)
DL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Sub-	Downlink	Uplink	Implicitely tested	Restricted UL	UL RLC	Test data size		
test	TFCS Under	TFCS Under test		TFCIs	SDU size (bits)	(bits)		
	Test	Under lest			(note)	(note)		
1	DL_TFC1	UL_TFC1	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 39	RB5: 39		
	DL_TFC7	UL_TFC7	UL_TFC0, UL_TFC6	UL_TFC1,	RB6: 103	RB6: No data		
				UL_TFC6,	RB7: 60	RB7: No data		
				UL_TFC7	RB8: 312	RB8: No data		
2	DL_TFC2	UL_TFC2	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 81	RB5: 81		
	DL_TFC8	UL_TFC8	UL_TFC0, UL_TFC6	UL_TFC2,	RB6: 103	RB6: 103		
				UL_TFC6,	RB7: 60	RB7: 60		
				UL_TFC8	RB8: 312	RB8: No data		
3	DL_TFC3	UL_TFC3	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 81	RB5: No data		
	DL_TFC9	UL_TFC9	UL_TFC0, UL_TFC6	UL_TFC3,	RB6: 103	RB6: No data		
				UL_TFC6,	RB7: 60	RB7: No data		
				UL_TFC9	RB8: 312	RB8: 312		
4	DL_TFC4	UL_TFC4	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 39	RB5: 39		
	DL_TFC10	UL_TFC10	UL_TFC0, UL_TFC6	UL_TFC1,	RB6: 103	RB6: No data		
				UL_TFC3,	RB7: 60	RB7: No data		
				UL_TFC4,	RB8: 312	RB8: 312		
				UL_TFC6,				
				UL_TFC7,				
				UL_TFC9,				
				UL_TFC10				
5	DL_TFC5	UL_TFC5	DL_TFC0, DL_TFC6,	UL_TFC0,	RB5: 81	RB5: 81		
	DL_TFC11	UL_TFC11	UL_TFC0, UL_TFC6	UL_TFC2,	RB6: 103	RB6: 103		
				UL_TFC3,	RB7: 60	RB7: 60		
				UL_TFC5,	RB8: 312	RB8: 312		
				UL_TFC6,				
				UL_TFC8,				
				UL_TFC9,				
NOTE				UL_TFC11				
NOTE:			se 5.3.2.6.2 for details re			10 L - 01		
	RB8: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the TTI for RB8 is the same for both downlink and uplink then UL							
			set to achieve UE to retu					
			e uplink TFS size under to	est minus & dits (S	ize of 7 bit lengt	n indicator and		
	expansion	טונ).						

14.2.38b.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.

- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38c Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:32 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38c.1 Conformance requirement

See 14.2.4.1.

14.2.38c.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38c.

14.2.38c.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

		TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (32 kbps)	DCCH
		TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS		TF1, bits	1x39	1x103	1x60	1x336	1x148
	3	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A	
		TF4, bits	N/A	N/A	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Downlink TFS:

		RB5	RB6	RB7	RB8	DCCH
		(RAB subflow #1)	(RAB subflow #2)	(RAB subflow #3)	(32 kbps)	Doon
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A
	TF4, bits	N/A	N/A	N/A	4x336	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test				(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC16	UL_TFC16	DL_TFC15,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC16	RB8: 312	RB8: No data
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC17	UL_TFC17	DL_TFC15,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: 60
0			UL_TFC15	UL_TFC17	RB8: 312	RB8: No data
3	DL_TFC3, DL_TFC18	UL_TFC3,	DL_TFC0, DL_TFC15,	UL_TFC0, UL_TFC3,	RB5: 81 RB6: 103	RB5: No data RB6: No data
	DL_IFC10	UL_TFC18	UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC18	RB8: 312	RB8: 312
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC19	UL_TFC19	DL_TFC15,	UL_TFC1,	RB6: 103	RB6: No data
			DL_TFC0,	UL TFC3,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC4,	RB8: 312	RB8: 312
				UL_TFC15,		
				UL_TFC16,		
				UL_TFC18,		
				UL_TFC19		
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC20	UL_TFC20	DL_TFC15,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC5, UL_TFC15,	RB8: 312	RB8: 312
				UL_TFC17,		
				UL_TFC18,		
				UL_TFC20		
6	DL_TFC6,	UL_TFC6,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: No data
	DL_TFC21	UL_TFC21	DL_TFC15,	UL_TFC6,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC21	RB8: 632	RB8: 632
7	DL_TFC7,	UL_TFC7,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC22	UL_TFC22	DL_TFC15,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC7, UL_TFC15,	RB8: 632	RB8: 632
				UL_TFC16,		
				UL_TFC21,		
				UL_TFC22		
8	DL_TFC8,	UL_TFC8,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
-	DL_TFC23	UL_TFC23	DL_TFC15,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC8,	RB8: 632	RB8: 632
				UL_TFC15,		
				UL_TFC17,		
				UL_TFC21,		
				UL_TFC23		
9	DL_TFC9,	UL_TFC9,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: No data
	DL_TFC24	UL_TFC24	DL_TFC15,	UL_TFC9, UL_TFC15,	RB6: 103 RB7: 60	RB6: No data
			UL_TFC0, UL_TFC15	UL_TFC15, UL_TFC24	RB7: 60 RB8: 952	RB7: No data RB8: 952
10	DL_TFC10	UL_TFC10	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
			DL_TFC15,	UL_TFC1,	RB6: 103	RB6: No data
	, DL_TFC25	, UL_TFC25	UL_TFC0,	UL_TFC9,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC10,	RB8: 952	RB8: 952
				UL_TFC15,		
				UL_TFC16,		
				UL_TFC24,		
				UL_TFC25		

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)			
11	DL_TFC11 , DL_TFC26	UL_TFC11 , UL_TFC26	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 952	RB5: 81 RB6: 103 RB7: 60 RB8: 952			
12	DL_TFC12 , DL_TFC27	UL_TFC12 , UL_TFC27	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 1272			
13	DL_TFC13 , DL_TFC28	UL_TFC13 , UL_TFC28	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 1272			
14	DL_TFC14 , DL_TFC29	UL_TFC14 , UL_TFC29	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 1272			
NOTE:	UL_TFC29 NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).								

14.2.38c.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3, 6, 9 and 12: an RLC SDU on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4, 7, 10 and 13: an RLC SDU on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5, 8, 11 and 14: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38d Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:64 kbps / PS RAB + Interactive or background / UL:64 DL:64 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38d.1 Conformance requirement

See 14.2.4.1.

14.2.38d.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38d.

14.2.38d.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

_		TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 + RB9 (64 kbps, 20 ms TTI)	DCCH
Γ		TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x340	0x148
		TF1, bits	1x39	1x103	1x60	1x340	1x148
	TFS	TF2, bits	1x81	N/A	N/A	2x340	N/A
		TF3, bits	N/A	N/A	N/A	3x340	N/A
		TF4, bits	N/A	N/A	N/A	4x340	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8+RB9, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 + RB9 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x340	0x148
	TF1, bits	1x39	1x103	1x60	1x340	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x340	N/A
	TF3, bits	N/A	N/A	N/A	3x340	N/A
	TF4, bits	N/A	N/A	N/A	4x340	N/A

TFCI	(RB5, RB6, RB7, RB8+RB9, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under Test	Under test			(bits) Note 1	Note 1
1	DL_TFC1, DL_TFC16	UL_TFC1, UL_TFC16	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 312 RB9: 312	RB5: 39 RB6: No data RB7: No data RB8: No data RB9: No data
2	DL_TFC2, DL_TFC17	UL_TFC2, UL_TFC17	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 312 RB9: 312	RB5: 81 RB6: 103 RB7: 60 RB8: No data RB9: No data
3	DL_TFC3, DL_TFC18	UL_TFC3, UL_TFC18	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC3, UL_TFC15, UL_TFC18	RB5: 39 RB6: 103 RB7: 60 RB8: 312 RB9: 312	RB5: No data RB6: No data RB7: No data RB8: 312 RB9: No data
4	DL_TFC4, DL_TFC19	UL_TFC4, UL_TFC19	DL_TFC0, DL_TFC15, DUL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC19	RB5: 39 RB6: 103 RB7: 60 RB8: 312 RB9: 312	RB5: 39 RB6: No data RB7: No data RB8: 312 RB9: No data
5	DL_TFC5, DL_TFC20	UL_TFC5, UL_TFC20	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC15, UL_TFC17, UL_TFC18, UL_TFC20	RB5: 81 RB6: 103 RB7: 60 RB8: 312 RB9: 312	RB5: 81 RB6: 103 RB7: 60 RB8: 312 RB9: No data
6	DL_TFC6, DL_TFC21	UL_TFC6, UL_TFC21	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC6, UL_TFC15, UL_TFC21	RB5: 39 RB6: 103 RB7: 60 RB8: 632 RB9: 632	RB5: No data RB6: No data RB7: No data RB8: 632 RB9: No data
7	DL_TFC7, DL_TFC22	UL_TFC7, UL_TFC22	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC22	RB5: 39 RB6: 103 RB7: 60 RB8: 632 RB9: 632	RB5: 39 RB6: No data RB7: No data RB8: 632 RB9: No data
8	DL_TFC8, DL_TFC23	UL_TFC8, UL_TFC23	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC21, UL_TFC23	RB5: 81 RB6: 103 RB7: 60 RB8: 632 RB9: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 632 RB9: No data
9	DL_TFC9, DL_TFC24	UL_TFC9, UL_TFC24	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC9, UL_TFC15, UL_TFC24	RB5: 39 RB6: 103 RB7: 60 RB8: 952 RB9: 952	RB5: No data RB6: No data RB7: No data RB8: 952 RB9: No data

Sub-	Downlink	Uplink	Implicitely	Restricted	UL RLC	Test data size
test	TFCS Under	TFCS Under test	tested	UL TFCIs	SDU size (bits)	(bits)
	Test				Note 1	Note 1
10	DL_TFC10, DL_TFC25	UL_TFC10, UL_TFC25	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC25	RB5: 39 RB6: 103 RB7: 60 RB8: 952 RB9: 952	RB5: 39 RB6: No data RB7: No data RB8: 952 RB9: No data
11	DL_TFC11, DL_TFC26	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 952 RB9: 952	RB5: 81 RB6: 103 RB7: 60 RB8: 952 RB9: No data
12	DL_TFC12, DL_TFC27	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272 RB9: 1272	RB5: No data RB6: No data RB7: No data RB8: 1272 RB9: No data
13	DL_TFC13, DL_TFC28	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272 RB9: 1272	RB5: 39 RB6: No data RB7: No data RB8: 1272 RB9: No data
14	DL_TFC14, DL_TFC29	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272 RB9: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 1272 RB9: No data
15	DL_TFC14, DL_TFC29	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272 RB9: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: No data RB9: 1272
			2.6.2 for details r			
						nder test minus 8 bits een set equal to the
	size of the payle					ength indicator and
	expansion bit).					

14.2.38d.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return

- for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7, RB8 or RB9.
- for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8 or RB9.
- for sub-test 3, 6, 9 and 12: an RLC SDU on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6, RB7 or RB9.
- for sub-test 4, 7, 10 and 13: an RLC SDU on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6, RB7 or RB9.
- for sub-test 5, 8, 11 and 14: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS. No data shall be received on RB9.
- for sub-test 15: an RLC SDU on RB5, RB6, RB7 and RB9 having the same content as sent by SS; and no data shall be received on RB8.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38e Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL:(12.2 7.95 5.9 4.75) kbps / CS RAB + Interactive or background / UL:0 DL:0 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38e.1 Conformance requirement

See 14.2.4.1.

14.2.38e.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38e.

14.2.38e.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (0 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	N/A	1x148
TFS	TF2, bits	1x42	1x63	N/A	N/A	N/A
113	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF0, TF0, TF0)
UL_TFC3	(TF3, TF2, TF0, TF0, TF0)
UL_TFC4	(TF4, TF3, TF0, TF0, TF0)
UL_TFC5	(TF5, TF1, TF1, TF0, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF0, TF0, TF1)
UL_TFC9	(TF3, TF2, TF0, TF0, TF1)
UL_TFC10	(TF4, TF3, TF0, TF0, TF1)
UL_TFC11	(TF5, TF1, TF1, TF0, TF1)

Downlink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (0 kbps)	DCCH
	TF0, bits	<u>1x0</u> 0x81(alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	N/A	1x148
TFS	TF2, bits	1x42	1x63	N/A	N/A	N/A
15	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF0, TF0, TF0)
DL_TFC3	(TF3, TF2, TF0, TF0, TF0)
DL_TFC4	(TF4, TF3, TF0, TF0, TF0)
DL_TFC5	(TF5, TF1, TF1, TF0, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF0, TF0, TF1)
DL_TFC9	(TF3, TF2, TF0, TF0, TF1)
DL_TFC10	(TF4, TF3, TF0, TF0, TF1)
DL_TFC11	(TF5, TF1, TF1, TF0, TF1)

Sub-	Downlink	Uplink	Implicitely tested	Restricted UL	UL RLC SDU	Test data size	
test	TFCS	TFCS		TFCIs	size	(bits)	
	Under	Under test			(bits)		
	Test				(note)	(note)	
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits	
	DL_TFC7	UL_TFC7	DL_TFC6,	UL_TFC1,	RB6: 103 bits	RB6: No data	
			UL_TFC0,	UL_TFC6,	RB7: 60 bits	RB7: No data	
			UL_TFC6	UL_TFC7	RB8: 0 bits	RB8: No data	
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits	
	DL_TFC8	UL_TFC8	DL_TFC6,	UL_TFC2,	RB6: 53 bits	RB6: 53 bits	
			UL_TFC0,	UL_TFC6,	RB7: 60 bits	RB7: No data	
			UL_TFC6	UL_TFC8	RB8: 0 bits	RB8: No data	
3	DL_TFC3,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits	
	DL_TFC9	UL_TFC9	DL_TFC6,	UL_TFC3,	RB6: 63 bits	RB6: 63 bits	
			UL_TFC0,	UL_TFC6,	RB7: 60 bits	RB7: No data	
			UL_TFC6	UL_TFC9	RB8: 0 bits	RB8: No data	
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits	
	DL_TFC10	UL_TFC10	DL_TFC6,	UL_TFC4,	RB6: 84 bits	RB6: 84 bits	
			UL_TFC0,	UL_TFC6,	RB7: 60 bits	RB7: No data	
			UL_TFC6	UL_TFC10	RB8: 0 bits	RB8: No data	
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits	
	DL_TFC11	UL_TFC11	DL_TFC6,	UL_TFC5,	RB6: 103 bits	RB6: 103 bits	
			UL_TFC0,	UL_TFC6,	RB7: 60 bits	RB7: 60 bits	
			UL_TFC6	UL_TFC11	RB8: 0 bits	RB8: No data	
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs.						

14.2.38e.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2,3 and 4: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
 - for sub-test 5: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38f Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL:(12.2 7.95 5.9 4.75) kbps / CS RAB + Interactive or background / UL:8 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38f.1 Conformance requirement

See clause 14.2.4.1.

14.2.38f.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38f.

14.2.38f.3 Method of test

See clause 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (8 kbps, 40 ms TTI)	DCCH
TFS	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
	TF2, bits	1x42	1x63	N/A	N/A	N/A
	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7,RB8,DCCH)
UL_TFC0	(TF0,TF0,TF0,TF0,TF0)
UL_TFC1	(TF1,TF0,TF0,TF0,TF0)
UL_TFC2	(TF2,TF1,TF0,TF0,TF0)
UL_TFC3	(TF3,TF2,TF0,TF0,TF0)
UL_TFC4	(TF4,TF3,TF0,TF0,TF0)
UL_TFC5	(TF5,TF4,TF1,TF0,TF0)
UL_TFC6	(TF0,TF0,TF0,TF1,TF0)
UL_TFC7	(TF1,TF0,TF0,TF1,TF0)
UL_TFC8	(TF2,TF1,TF0,TF1,TF0)
UL_TFC9	(TF3,TF2,TF0,TF1,TF0)
UL_TFC10	(TF4,TF3,TF0,TF1,TF0)
UL_TFC11	(TF5,TF4,TF1,TF1,TF0)
UL_TFC12	(TF0,TF0,TF0,TF0,TF1)
UL_TFC13	(TF1,TF0,TF0,TF0,TF1)
UL_TFC14	(TF2,TF1,TF0,TF0,TF1)
UL_TFC15	(TF3,TF2,TF0,TF0,TF1)
UL_TFC16	(TF4,TF3,TF0,TF0,TF1)
UL_TFC17	(TF5,TF4,TF1,TF0,TF1)
UL_TFC18	(TF0,TF0,TF0,TF1,TF1)
UL_TFC19	(TF1,TF0,TF0,TF1,TF1)
UL_TFC20	(TF2,TF1,TF0,TF1,TF1)
UL_TFC21	(TF3,TF2,TF0,TF1,TF1)
UL_TFC22	(TF4,TF3,TF0,TF1,TF1)
UL_TFC23	(TF5,TF4,TF1,TF1,TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (8 kbps, 40 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
TFS	TF2, bits	1x42	1x63	N/A	N/A	N/A
115	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0,TF0,TF0,TF0,TF0)
DL_TFC1	(TF1,TF0,TF0,TF0,TF0)
DL_TFC2	(TF2,TF1,TF0,TF0,TF0)
DL_TFC3	(TF3,TF2,TF0,TF0,TF0)
DL_TFC4	(TF4,TF3,TF0,TF0,TF0)
DL_TFC5	(TF5,TF4,TF1,TF0,TF0)
DL_TFC6	(TF0,TF0,TF0,TF1,TF0)
DL_TFC7	(TF1,TF0,TF0,TF1,TF0)
DL_TFC8	(TF2,TF1,TF0,TF1,TF0)
DL_TFC9	(TF3,TF2,TF0,TF1,TF0)
DL_TFC10	(TF4,TF3,TF0,TF1,TF0)
DL_TFC11	(TF5,TF4,TF1,TF1,TF0)
DL_TFC12	(TF0,TF0,TF0,TF0,TF1)
DL_TFC13	(TF1,TF0,TF0,TF0,TF1)
DL_TFC14	(TF2,TF1,TF0,TF0,TF1)
DL_TFC15	(TF3,TF2,TF0,TF0,TF1)
DL_TFC16	(TF4,TF3,TF0,TF0,TF1)
DL_TFC17	(TF5,TF4,TF1,TF0,TF1)
DL_TFC18	(TF0,TF0,TF0,TF1,TF1)
DL_TFC19	(TF1,TF0,TF0,TF1,TF1)
DL_TFC20	(TF2,TF1,TF0,TF1,TF1)
DL_TFC21	(TF3,TF2,TF0,TF1,TF1)
DL_TFC22	(TF4,TF3,TF0,TF1,TF1)
DL_TFC23	(TF5,TF4,TF1,TF1,TF1)

under test Under test (note) (note) (note) 1 DL TFC1, DL TFC13 UL TFC1, UL TFC13 UL TFC0, DL TFC12, UL TFC14 UL TFC1, UL TFC14 RB5: 30 bits RB5:	Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
1 DL TFC1. DL TFC13 UL TFC14. UL TFC3 DL TFC0. DL TFC12. UL TFC4. UL TFC14. UL TFC6. UL TFC14. RB5: 39 bits RB5: 39 bits RB7: No dat RB7:	lesi				OL IFCIS		
DL_TFC13 UL_TFC12 UL_TFC12 UL_TFC14 RB6: No date UL_TFC14 RB6: No date RB7: No date UL_TFC14 2 DL_TFC14 UL_TFC2, UL_TFC14 UL_TFC3, UL_TFC14 UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC5, UL_TFC4, UL	_						
2 DL_TFC12, RB7: No dati RB8: No dati UL_TFC13 RB7: No dati RB8: No dati RB8: No dati RB8: No dati RB8: So bits RB7: No dati RB8: No dati RB8: So bits 2 DL_TFC14 UL_TFC2, UL_TFC14 DL_TFC12, UL_TFC14 UL_TFC2, RB7: So bits RB8: A2 bits RB8: No dati RB8: No dati RB8: No dati UL_TFC14 RB8: So bits RB8: So bits RB8: No dati RB8: No dati UL_TFC14 3 DL_TFC15 UL_TFC3, UL_TFC15 DL_TFC15, UL_TFC16 UL_TFC3, RB7: No dati RB8: No dati UL_TFC17 RB7: No dati RB8: No dati UL_TFC17 RB8: So bits RB8: So bits RB8: So bits RB8: No dati RB8: No dati UL_TFC17 RB7: No dati RB8: No dati UL_TFC17 RB7: No dati RB8: No dati UL_TFC17 RB8: So bits RB8: R0 dati RB8: No dati RB8: No dati UL_TFC17 RB8: R0 bits RB8: R0	1						
L		DL_IFC13	UL_IFC13	UL_1FC0, UL_1FC12			
2 DL TFC2, DL TFC14 UL TFC2, UL TFC14 DL TFC0, DL TFC12, UL TFC12, UL TFC12, UL TFC14 UL TFC0, RB5: 42 bits RB7: No dati UL TFC14, RB7: No dati RB7: No dati RB7: No dati RB7: No dati RB7: No dati RB7: No dati RB5: 55 bits RB7: No dati RB7: No dati RB5: 55 bits RB7: No dati RB7:							
DL_TFC14 UL_TFC12 UL_TFC12 Refs. 53 bits UL_TFC12 Refs. 53 bits Refs. 78 otas Refs. 78 otas R	0						
Image: Second state UL_TFC12 RB7: 60 bits UL_TFC14 RB7: 80 dats RB6: 63 bits UL_TFC15 RB6: 85 bits RB7: 80 dats UL_TFC12 RB7: 60 bits UL_TFC3 RB7: 80 dats RB7: 80 dats UL_TFC15 4 DL_TFC4, DL_TFC15 UL_TFC4, UL_TFC16 UL_TFC4, UL_TFC16 UL_TFC4, UL_TFC16 RB6: 75 bits RB7: 80 dats UL_TFC16 RB6: 75 bits RB7: 80 dats RB7: 80 dats UL_TFC17 RB6: 76 bits RB7: 80 dats RB7: 80 dats RB	2						
Image: Construction of the section of the sectin of the section of the section of the section of the se		DL_IFC14	UL_IFC14	UL_1FC0, UL_1FC12			
3 DL_TFC3, DL_TFC15 UL_TFC15 DL_TFC0, UL_TFC12, UL_TFC0, UL_TFC3, DL_TFC16 UL_TFC3, RB7: 60 bits UL_TFC17 RB5: 55 bits RB7: No dat UL_TFC17 RB5: 75 bits RB7: No dat RB7: No dat RB7: No dat UL_TFC14 RB5: 75 bits RB7: No dat UL_TFC14 RB5: 75 bits RB7: No dat RB7: No dat RB7: No dat RB7: No dat UL_TFC14 RB5: 75 bits RB7: No dat RB7: No dat RB7: No dat RB7: No dat UL_TFC14 RB7: No dat RB7: S0							
DL_TFC15 UL_TFC3 UL_TFC12 UL_TFC32 RB6: 63 bits RB7: No dats UL_TFC14 RB6: 63 bits RB7: No dats RB7: No dats UL_TFC14 RB6: 63 bits RB7: No dats RB7: No dats UL_TFC17 RB6: 63 bits RB7: No dats RB7: No d	0						
4 DL_TFC4, DL_TFC16 UL_TFC4, UL_TFC16 UL_TFC4, UL_TFC16 UL_TFC4, UL_TFC16 UL_TFC4, UL_TFC16 UL_TFC4, UL_TFC16 RB: 78 bits UL_TFC17 RB: 78 bits UL_TFC17 RB: 78 bits RB: 88 bits RB: 88 bits RB: 88 bits RB: 88 bits RB: 88 bits RB: 88 bits RB: 81 bits RB	3			_ / _ /			
4 DL_TFC4, DL_TFC16 UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC7, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC7, UL_TFC6, UL_TFC6, UL_TFC7, UL_TFC2, UL_TFC2, UL_TFC7, UL_TFC7, UL_TFC7, UL_TFC1, UL_TFC1, UL_TFC1		DL_IFC15	UL_IFCI5				
4 DL_TFC4, DL_TFC16 UL_TFC4, UL_TFC16 UL_TFC4, UL_TFC17 UL_TFC4, UL_TFC12 RBs: 75 bits, RBs: 84 bits, RBs: 80 bits, RBs: 81 bits, RB							
DL_TFC16 UL_TFC16 UL_TFC16 UL_TFC12 UL_TFC14, UL_TFC14, RB7: 60 bits RB7: No dats UL_TFC17 RB7: No dats RB7: No dats R	4						
Image: Construct of the system of t	4						
- UL_TFC16 RB3: No dat 5 DL_TFC3, DL_TFC17 UL_TFC4, UL_TFC17 UL_TFC4, UL_TFC17 UL_TFC4, UL_TFC17 RB3: No dat 6 DL_TFC6, DL_TFC18 UL_TFC7, UL_TFC19 DL_TFC6, UL_TFC18 UL_TFC7, UL_TFC19 UL_TFC19 RB3: No dat 8 DL_TFC19 UL_TFC7, UL_TFC19 UL_TFC7, UL_TFC19 UL_TFC19 RB3: No dat RB3: No dat 7 DL_TFC7, DL_TFC19 UL_TFC7, UL_TFC19 UL_TFC19 UL_TFC10, RB3: S12 bits RB5: S3 bits RB5: S3 bits 8 DL_TFC8, DL_TFC20 UL_TFC4, UL_TFC20 UL_TFC19 RB5: 42 bits RB5: 42 bits RB5: 42 bits 9 DL_TFC9, DL_TFC21 UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC24 UL_TFC20, UL_TFC12, UL_TFC20, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC24 RB5: 75 bits RB5: 75 bits RB5: 75 bits 9 DL_TFC10, DL_TFC21 UL_TFC20, UL_TFC12, UL_TFC2, UL_TFC24, UL_TFC24, UL_TFC14, UL_TFC24, U			UL_IFC10				
5 DL_TFC5, DL_TFC17 UL_TFC6, UL_TFC17 DL_TFC6, UL_TFC17 DL_TFC6, UL_TFC17 DL_TFC6, UL_TFC17 DL_TFC6, UL_TFC17 DL_TFC6, UL_TFC17 DL_TFC6, UL_TFC18 DL_TFC6, UL_TFC17 DL_TFC6, UL_TFC18 DL_TFC6, UL_TFC17 DL_TFC6, UL_TFC17 DL_TFC7, UL_TFC18 DL_TFC7, UL_TFC19 DL_TFC7, UL_TFC19 DL_TFC7, UL_TFC19 DL_TFC7, UL_TFC19 DL_TFC0, UL_TFC19 DL_TFC0, UL_TFC19 DL_TFC0, UL_TFC19 DL_TFC0, UL_TFC19 DL_TFC0, UL_TFC19 DL_TFC0, UL_TFC19 DL_TFC0, UL_TFC19 RB5: 312 bits UL_TFC18, UL_TFC18, UL_TFC18, RB5: 34 bits RB5: 34 bits RB5: 34 bits RB5: 34 bits RB5: 312 bits RB5: 34 bits 8 DL_TFC8, DL_TFC20 UL_TFC7, UL_TFC20 UL_TFC8, UL_TFC0, UL_TFC18, UL_TFC18, UL_TFC18, RB5: 42 bits RB5: 42 bits 8 DL_TFC9, DL_TFC20 UL_TFC20 DL_TFC0, UL_TFC12, UL_TFC0, UL_TFC18, UL_TFC0, UL_TFC18, RB5: 55 bits RB5: 55 bits 9 DL_TFC10, DL_TFC21 UL_TFC20, UL_TFC12, UL_TFC20, UL_TFC12, UL_TFC18, UL_TFC14, UL_TFC18, UL_TFC18, RB5: 55 bits RB5: 55 bits 10 DL_TFC10, DL_TFC21 UL_TFC0, UL_TFC12, UL_TFC18, UL_TFC20, UL_TFC18, UL_TFC18, RB5: 75 bits RB5: 75 bits							
DL_TFC17 UL_TFC17 UL_TFC17 UL_TFC12 UL_TFC5; UL_TFC17 RB6: 103 bits RB7: 60 bits RB8: 312 bits RB6: 103 bits RB7: 60 bits RB8: N0 dat 6 DL_TFC6 UL_TFC6, UL_TFC18 UL_TFC6, UL_TFC19 UL_TFC6, UL_TFC19 UL_TFC6, UL_TFC19 RB6: 103 bits RB8: N0 dat RB6: 103 bits RB8: N0 dat 7 DL_TFC7, DL_TFC19 UL_TFC17 DL_TFC19 UL_TFC19 UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12 UL_TFC1, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC18, RB6: 30 bits RB8: N0 dat RB6: 30 bits RB7: N0 dat 8 DL_TFC8, DL_TFC29 UL_TFC8, UL_TFC20 DL_TFC0, DL_TFC12, UL_TFC14, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC20 RB6: 42 bits RB7: N0 dat RB6: 53 bits RB7: N0 dat 9 DL_TFC9, DL_TFC21 UL_TFC20, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC14, UL_TFC20, RB6: 55 bits RB7: N0 dat RB6: 55 bits RB7: N0 dat 9 DL_TFC9, DL_TFC21 UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC14, UL_TFC20, RB6: 55 bits RB7: N0 dat RB6: 55 bits RB7: N0 dat 10 DL_TFC10, DL_TFC22 UL_TFC10, UL_TFC20, UL_TFC12, UL_TFC16, UL_TFC18, UL_TFC20, UL_TFC20, UL_TFC12, UL_TFC14, RB6: 75 bits RB7: 76 bits RB7: 76 bits RB7: 76 bits RB7:	5						
Image: Construct of the system of t	5				- /		
Image: Construct of the system of t				0L_1100, 0L_1F012			
6 DL_TFC6, DL_TFC18 UL_TFC6, UL_TFC18 UL_TFC6, UL_TFC18 UL_TFC6, UL_TFC0, UL_TFC12 UL_TFC6, UL_TFC12 UL_TFC6, UL_TFC18 UL_TFC7, UL_TFC19 RBS: 103 bits RBS: 103 bits RBS: 103 bits RBS: 104 bits							
DL_TFC18 UL_TFC18 UL_TFC18 UL_TFC12 UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC14 RB6: 103 bits RB7: No dat RB7: No dat RB8: 312 bits RB6: No dat RB7: No dat RB7: No dat RB8: 312 bits 7 DL_TFC7, DL_TFC19 UL_TFC7, UL_TFC19 UL_TFC0, UL_TFC0, UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC20 RB5: 30 bits RB7: No dat RB7: No dat	6	DI TECA	LII TECA	DI TECO DI TEC12			
Image: Construct of the system of t	0				- /		
Image: Construct of the system of t			0				
7 DL_TFC7, DL_TFC19 UL_TFC7, UL_TFC19 UL_TFC7, UL_TFC19 DL_TFC0, UL_TFC19 DL_TFC0, UL_TFC19 UL_TFC0, UL_TFC19 RB5: 39 bits RB6: 3103 bits RB7: 60 bits RB7: 75 bits RB7: 75 bits RB7: 60 bits RB7: 75 bits RB7: 60 bits RB7: 75 bit							
DL_TFC19 UL_TFC19 UL_TFC19 UL_TFC0, UL_TFC12 UL_TFC6, UL_TFC7, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC20 RB5: 40 bits RB3: 312 bits RB6: No data RB7: 00 bits RB6: 312 bits 8 DL_TFC8, DL_TFC20 UL_TFC8, UL_TFC20 UL_TFC7, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC7, UL_TFC6, UL_TFC7, UL_TFC6, UL_TFC7, UL_TFC6, UL_TFC7, UL_TFC14, UL_TFC7, UL_TFC7, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC21 RB5: 55 bits RB6: 55 bits RB6: 55 bits RB6: 55 bits RB7: No dat RB8: 312 bits RB5: 55 bits RB6: 55 bits RB7: No dat RB8: 312 bits 9 DL_TFC9, DL_TFC21 UL_TFC0, DL_TFC12, UL_TFC14, UL_TFC14, UL_TFC21 UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC15, UL_TFC15, UL_TFC14, UL_TFC15, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC23 RB5: 55 bits RB6: RB7: No dat RB8: 312 bits RB5: 55 bits RB6: 63 bits RB7: No dat RB8: 312 bits RB5: 55 bits RB6: 63 bits RB7: No dat RB8: 312 bits 10 DL_TFC10, DL_TFC10, UL_TFC10, UL_TFC22 UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC24 RB5: 75 bits RB6: 84 bits RB7: No dat RB8: 312 bits RB5: 75 bits RB6: 84 bits RB7: No dat RB8: 312 bits 11 DL_TFC11, DL_TFC23 UL_TFC11, UL_TFC24 UL_TFC20, UL_TFC12, UL_TFC26, UL_TFC14, UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC24 RB5: 81 bits RB6: 103 bits RB7: No dat RB8: 312 bit RB7	7	DI TEC7	UL TEC7	DI TECO DI TEC12			
B DL_TFC8, DL_TFC13, UL_TFC14, UL_TFC13, UL_TFC18, DL_TFC20 UL_TFC8, UL_TFC20 UL_TFC8, UL_TFC20 UL_TFC8, UL_TFC20 RB5: 42 bits RB5: 42 bits RB5: 312 bits 8 DL_TFC20 UL_TFC20 UL_TFC20 UL_TFC20 UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC21 RB5: 42 bits RB5: 42 bits RB5: 312 bits 9 DL_TFC9, DL_TFC21 UL_TFC9, UL_TFC21 UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC15, UL_TFC23 RB5: 55 bits RB5: 55 bits RB6: 63 bits RB6: 63 bits 9 DL_TFC21 UL_TFC21 UL_TFC0, UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC15, UL_TFC23 UL_TFC21 RB5: 55 bits RB6: 63 bits RB6: 63 bits 10 DL_TFC10, DL_TFC22 UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC15, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC23 RB5: 75 bits RB6: 75 bits RB6: 75 bits 11 DL_TFC11, DL_TFC23 UL_TFC11, UL_TFC23 DL_TFC0, UL_TFC12, UL_TFC10, UL_TFC2, UL_TFC24, UL_TFC14, UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC23 RB5: 81 bits RB5: 81 bits RB6: 103							
B DL_TFC1, DL_TFC2, DL_TFC2, DL_TFC2, RBS: 312 bits UL_TFC12, UL_TFC12, UL_TFC18, UL_TFC18, RBS: 312 bits RBS: 312 bits RBS: 312 bits RBS: 312 bits 8 DL_TFC8, DL_TFC20 UL_TFC8, UL_TFC20 DL_TFC9, UL_TFC20 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12 UL_TFC2, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC20 RBS: 42 bits RBS: 42 bits RBS: 53 bits 9 DL_TFC21 UL_TFC9, UL_TFC21 DL_TFC0, DL_TFC12, UL_TFC0, DL_TFC12, UL_TFC14, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC21 RBS: 55 bits RBS: 55 bits RBS: 55 bits 9 DL_TFC21 UL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC14, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC21 RBS: 55 bits RBS: 55 bits RBS: 55 bits 10 DL_TFC10, DL_TFC22 UL_TFC10, UL_TFC22 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC18, UL_TFC23 RBS: 75 bits RBS: 75 bits RBS: 75 bits 11 DL_TFC11, DL_TFC23 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC12, UL_TFC12, UL_TFC14, UL			0				
Image: Second							
B DL_TFC13, UL_TFC18, UL_TFC18, DL_TFC20 UL_TFC3, UL_TFC20 UL_TFC3, UL_TFC20 RB5: 42 bits RB5: 53 bits UL_TFC2, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC20 RB5: 42 bits RB5: 53 bits RB7: No date RB3: 312 bits 9 DL_TFC9, DL_TFC21 UL_TFC9, UL_TFC21 UL_TFC9, UL_TFC21 UL_TFC9, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC20 RB5: 55 bits RB5: 55 bits RB5: 63 bits RB6: 63 bits RB6: 63 bits RB7: No date UL_TFC20 RB5: 55 bits RB6: 63 bits RB7: 06 bits RB7: 06 bits RB7: 06 bits RB7: 06 bits RB7: 06 bits RB7: No date UL_TFC15, UL_TFC12 RB5: 75 bits RB7: 60 bits RB7: 60 bits RB7: No date UL_TFC15, UL_TFC16, UL_TFC21 RB5: 75 bits RB6: 75 bits RB7: No date RB7: 06 bits RB7: 06 bits RB7: No date UL_TFC14, UL_TFC15, UL_TFC14, UL_TFC14, UL_TFC22 RB5: 75 bits RB7: No date RB7: 06 bits RB7: 06 bits RB7: 06 bits RB7: No date RB7: 06 bits RB7: No date RB7: 06 bits RB7: 06 bit							
B DL_TFC8, DL_TFC20 UL_TFC18, UL_TFC20 UL_TFC8, UL_TFC20 DL_TFC9, UL_TFC20 DL_TFC9, UL_TFC20 DL_TFC9, UL_TFC20 RB5: 42 bits RB5: 53 bits RB5: 53 bits RB5: 42 bits RB5: 53 bits 9 DL_TFC9, DL_TFC21 UL_TFC9, UL_TFC21 UL_TFC9, UL_TFC21 DL_TFC9, UL_TFC21 DL_TFC9, UL_TFC21 DL_TFC9, UL_TFC21 RB5: 55 bits RB5: 55 bits RB6: 63 bits RB6: 63 bits RB6: 63 bits RB6: 63 bits RB7: No dat 9 DL_TFC21 UL_TFC21 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC6, UL_TFC21 UL_TFC21 RB5: 55 bits RB6: 63 bits RB7: No dat 10 DL_TFC10, DL_TFC12, UL_TFC12, UL_TFC22 UL_TFC10, UL_TFC10, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC6, UL_TFC6, UL_TFC14, UL_TFC14, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC23 DL_TFC11, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC0, RB6: 103 bits RB6							
Image: system in the							
DL_TFC20 UL_TFC20 UL_TFC0, UL_TFC12 UL_TFC2, UL_TFC2, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC4, UL_TFC14, UL_TFC12, UL_TFC14, UL_TFC12, UL_TFC21 RB6: 53 bits RB7: No data RB8: 312 bits RB6: 55 bits RB7: No data RB8: 312 bits 9 DL_TFC21 UL_TFC9, UL_TFC21 DL_TFC0, DL_TFC12, UL_TFC12, UL_TFC14, UL_TFC3, UL_TFC20 UL_TFC20 RB6: 55 bits RB7: No data RB8: 312 bits RB6: 55 bits RB7: No data RB8: 312 bits 9 DL_TFC21 UL_TFC21, UL_TFC21 DL_TFC0, UL_TFC12, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC5, UL_TFC5, UL_TFC22 RB6: 55 bits RB7: No data RB8: 312 bits RB6: 55 bits RB7: No data RB7: No data RB8: 312 bits 10 DL_TFC10, UL_TFC10, UL_TFC10, UL_TFC0, UL_TFC12, UL_TFC18, UL_TFC6, RB7: 60 bits RB7: No data RB8: 312 bits RB6: 75 bits RB7: No data RB8: 312 bits RB6: 75 bits RB7: No data RB8: 312 bits 11 DL_TFC11, UL_TFC11, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC23 UL_TFC0, UL_TFC12, UL_TFC5, RB6: 103 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits RB6: 103 bits RB7							
Image: second systemUL_TFC1, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2,RB7: 60 bits RB8: 312 bitsRB7: No dat RB8: 312 bits9DL_TFC9, DL_TFC21UL_TFC9, UL_TFC21DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC15, UL_TFC14, UL_TFC16, UL_TFC17,RB5: 55 bits RB5: 63 bits RB7: No dat RB6: 63 bits RB7: No dat RB7: 60 bits RB7: 60 bits R	8		UL_TFC8,	DL_TFC0, DL_TFC12,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
Image: system in the system is a system is a system is a system in the system is a system is a system is a system in the system is a system i		DL_TFC20	UL_TFC20	UL_TFC0, UL_TFC12	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
9DL_TFC9, DL_TFC21UL_TFC9, UL_TFC21DL_TFC9, UL_TFC21DL_TFC9, UL_TFC21DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC6, UL_TFC6, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC21RB5: 55 bits RB6: 63 bits RB7: 00 bits RB7: 00 bits RB8: 312 bitsRB5: 55 bits RB6: 63 bits RB7: 00 bits RB8: 312 bits10DL_TFC10, DL_TFC22DL_TFC10, UL_TFC22DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC18, UL_TFC14, UL_TFC14, UL_TFC16, UL_TFC10, UL_TFC10, UL_TFC22RB5: 75 bits RB7: No dat RB8: 312 bits RB8: 312 bitsRB5: 75 bits RB7: No dat RB8: 312 bits10DL_TFC10, UL_TFC22DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC14, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17,RB5: 75 bits RB7: No dat RB8: 312 bitsRB5: 75 bits RB6: 84 bits RB7: No dat RB8: 312 bits11DL_TFC11, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC24, UL_TFC26, UL_TFC26, RB7: 60 bits RB7:							RB7: No data
9DL_TFC9, DL_TFC21UL_TFC19, UL_TFC21DL_TFC0, DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC3, UL_TFC3, RB6: 63 bitsRB5: 55 bits RB6: 63 bits RB7: 60 bits RB7: 60 bits RB8: 312 bits10DL_TFC10, DL_TFC22UL_TFC10, UL_TFC22DL_TFC10, UL_TFC10, UL_TFC10, UL_TFC22DL_TFC0, DL_TFC12, UL_TFC10, UL_TFC10, UL_TFC21UL_TFC4, UL_TFC13, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC18, UL_TFC18, UL_TFC10, UL_TFC13,RB5: 75 bits RB7: No dat RB8: 312 bits10DL_TFC10, UL_TFC22UL_TFC10, UL_TFC22DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC10, UL_TFC18, UL_TFC18, UL_TFC13, UL_TFC22RB5: 75 bits RB7: No dat RB7: No dat RB8: 312 bits11DL_TFC11, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12UL_TFC00, UL_TFC5, RB6: 103 bits RB7: 60 bits RB8: 312 bit						RB8: 312 bits	RB8: 312 bits
9 DL_TFC9, DL_TFC21 UL_TFC9, UL_TFC21 DL_TFC9, UL_TFC21 DL_TFC0, UL_TFC0, UL_TFC12 UL_TFC0, UL_TFC12 UL_TFC0, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC15, UL_TFC15, UL_TFC14, RB5: 55 bits RB6: 63 bits RB7: No dats RB7: No dats RB8: 312 bits RB5: 75 bits RB8: 312 bits RB5: 75 bits RB8: 312 bits 10 DL_TFC10, DL_TFC22 UL_TFC10, UL_TFC12, UL_TFC22 DL_TFC10, UL_TFC22 DL_TFC10, UL_TFC12, UL_TFC12, UL_TFC14, RB5: 75 bits UL_TFC16, UL_TFC16, UL_TFC10, UL_TFC10, UL_TFC16, UL_TFC16, UL_TFC12, RB5: 75 bits RB8: 312 bits RB5: 75 bits RB6: 84 bits RB6: 84 bits 11 DL_TFC11, DL_TFC23 UL_TFC0, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12, UL_TFC10, UL_TFC12, UL_TFC12, UL_TFC11, DL_TFC23 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC13, UL_TFC13, UL_TFC14, UL_TFC							
9 DL_TFC9, DL_TFC21 UL_TFC9, UL_TFC21 DL_TFC9, UL_TFC21 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12 UL_TFC0, UL_TFC3, UL_TFC9, UL_TFC9, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC14, RB5: 55 bits RB6: 63 bits RB6: 63 bits RB7: No dat RB8: 312 bits RB5: 55 bits RB7: No dat RB8: 312 bits 10 DL_TFC10, DL_TFC22 UL_TFC10, UL_TFC22 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC14, UL_TFC0, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC12 RB5: 75 bits RB6: 84 bits RB6: 84 bits RB5: 75 bits RB6: 84 bits RB7: No dat RB8: 312 bits 11 DL_TFC11, DL_TFC23 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12 UL_TFC0, UL_TFC6, UL_TFC6, UL_TFC16, UL_TFC24 RB5: 81 bits RB6: 103 bits RB5: 81 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits 11 DL_TFC23 UL_TFC0, UL_TFC0, UL_TFC17, UL_TFC24 DL_TFC17, UL_TFC17, UL_TFC17, RB6: 312 bits RB5: 81 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits							
9 DL_TFC9, DL_TFC21 UL_TFC9, UL_TFC21 DL_TFC0, UL_TFC0, UL_TFC12 UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC14, UL_TFC15, UL_TFC15, UL_TFC22 RB5: 55 bits RB6: 63 bits RB7: No dat RB8: 312 bits RB5: 55 bits RB6: 63 bits RB7: No dat RB8: 312 bits 10 DL_TFC10, DL_TFC22 UL_TFC10, UL_TFC22 DL_TFC10, UL_TFC22 DL_TFC10, UL_TFC22 DL_TFC10, UL_TFC22 RB5: 75 bits UL_TFC0, UL_TFC12 RB5: 75 bits UL_TFC4, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC22 RB5: 81 bits RB8: 312 bits RB5: 81 bits RB8: 312 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits 11 DL_TFC11, DL_TFC23 UL_TFC11, UL_TFC23 DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12 UL_TFC0, UL_TFC16, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC17, RB5: 81 bits RB6: 103 bits RB7: 60 bits RB5: 81 bits RB7: 80 bits RB5: 81 bits RB7: 80 bits							
DL_TFC21UL_TFC21UL_TFC0, UL_TFC12UL_TFC3, UL_TFC9, UL_TFC9, UL_TFC12, UL_TFC12, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC21RB6: 63 bits RB7: 60 bits RB7: 10 bits RB8: 312 bitsRB6: 63 bits RB7: No dat RB8: 312 bits10DL_TFC10, DL_TFC22UL_TFC10, UL_TFC22DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC12, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC23DL_TFC0, UL_TFC12, UL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC12, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC23RB5: 81 bits RB5: 81 bits RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7:							
UL_TFC1, DL_TFC2UL_TFC1, UL_TFC2UL_TFC1, UL_TFC2RB5: 75 bits RB5: 84 bits RB7: No dat RB5: 84 bits RB7: No dat RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 6	9						
UL_TFC9, UL_TFC18, UL_TFC18, UL_TFC22RB8: 312 bitsRB8: 312 bitsRB8: 312 bits10DL_TFC10, DL_TFC22UL_TFC10, UL_TFC22DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC1, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC4, UL_TFC6, UL_TFC16, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC22RB5: 75 bits RB6: 84 bits RB6: 84 bits RB7: 00 bits RB8: 312 bitsRB5: 75 bits RB6: 84 bits RB6: 84 bits RB7: No dat RB8: 312 bits11DL_TFC11, DL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12, UL_TFC12, UL_TFC16, UL_TFC0, UL_TFC23RB5: 81 bits RB5: 81 bits UL_TFC0, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC17,RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 312 bits		DL_IFC21	UL_IFC21	UL_IFC0, UL_TFC12			
Image: Line of the system UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC15, UL_TFC18, UL_TFC18, UL_TFC21 UL_TFC10, UL_TFC10, DL_TFC10, DL_TFC12, UL_TFC21 UL_TFC0, UL_TFC2, UL_TFC22 RB5: 75 bits RB5: 75 bits RB6: 84 bits RB6: 84 bits RB6: 84 bits RB7: No data 10 DL_TFC22 UL_TFC22 UL_TFC0, UL_TFC12 UL_TFC6, UL_TFC6, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC16, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC22 RB8: 312 bits RB8: 312 bits RB5: 81 bits 11 DL_TFC11, UL_TFC11, UL_TFC23 DL_TFC0, DL_TFC12, UL_TFC12, UL_TFC5, RB6: 103 bits RB5: 81 bits RB5: 81 bits RB6: 103 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits RB6: 103 bits RB7: 60 bits RB6: 103 bits RB8: 312 bits RB8: 312 bits RB8: 312 bits RB8: 312 bits R							
UL_TFC15, UL_TFC10, DL_TFC22UL_TFC10, UL_TFC22UL_TFC10, UL_TFC22DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC4, UL_TFC4, UL_TFC6, UL_TFC10, UL_TFC12, UL_TFC12, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC22RB5: 75 bits RB5: 75 bits RB6: 84 bits RB7: No data RB8: 312 bits11DL_TFC11, DL_TFC23UL_TFC11, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, DL_TFC12, UL_TFC12UL_TFC0, UL_TFC12, UL_TFC5, UL_TFC6, UL_TFC6, RB7: 60 bitsRB5: 81 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits11DL_TFC23UL_TFC11, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC5, UL_TFC6, RB7: 60 bitsRB5: 81 bits RB6: 103 bits RB7: 60 bits						KB8: 312 bits	RB8: 312 bits
Image: constraint of the constra					- '		
Image: construct of the construction of the constr							
10DL_TFC10, DL_TFC22UL_TFC10, UL_TFC22DL_TFC0, UL_TFC22DL_TFC0, UL_TFC0, UL_TFC0,UL_TFC0, RB5: 75 bits RB6: 84 bits UL_TFC6, UL_TFC10, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC12, UL_TFC16, UL_TFC13, UL_TFC22UL_TFC1, RB8: 312 bitsRB5: 75 bits RB6: 84 bits RB7: No data RB8: 312 bits11DL_TFC11, DL_TFC13, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL							
DL_TFC22UL_TFC22UL_TFC22UL_TFC0, UL_TFC12UL_TFC4, UL_TFC6, UL_TFC10, UL_TFC12, UL_TFC12, UL_TFC13, UL_TFC23RB6: 84 bits RB7: 80 bits RB8: 312 bitsRB6: 84 bits RB7: 80 bits RB8: 312 bits11DL_TFC11, DL_TFC23UL_TFC11, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC12RB5: 81 bits RB6: 103 bits RB6: 103 bits RB7: 60 bitsRB5: 81 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits	10					DB5. 75 hito	PB5: 75 hito
UL_TFC6, UL_TFC10, UL_TFC11, DL_TFC23RB7: 60 bits RB7: 80 dataRB7: No data RB8: 312 bits11DL_TFC11, DL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC5, UL_TFC5, UL_TFC6, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC23RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 312 bits	10						
UL_TFC10, UL_TFC12, UL_TFC13,RB8: 312 bitsRB8: 312 bits11DL_TFC11, DL_TFC23UL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC11, UL_TFC11, UL_TFC12,RB5: 81 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB8: 312 bitsRB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 312 bits				0L_1FC0, 0L_1FC12			
11DL_TFC11, DL_TFC23UL_TFC11, UL_TFC23DL_TFC0, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC12, UL_TFC5, UL_TFC6, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL							
UL_TFC16, UL_TFC11,UL_TFC11, UL_TFC23DL_TFC0, DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC6, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC17,RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB8: 312 bits							1100. 012 0113
UL_TFC18, UL_TFC21UL_TFC11, UL_TFC23UL_TFC11, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC11, UL_TFC12, UL_TFC17,RB5: 81 bits RB5: 81 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits RB8: 312 bits							
Image: Constraint of the constrated of the constraint of the constraint of the constraint of the							
11DL_TFC11, DL_TFC23UL_TFC11, UL_TFC23DL_TFC0, DL_TFC12, UL_TFC0, UL_TFC12UL_TFC0, UL_TFC5, UL_TFC6, UL_TFC11, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC17,RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 312 bitsRB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 312 bits					UL TFC22		
DL_TFC23 UL_TFC23 UL_TFC0, UL_TFC12 UL_TFC5, UL_TFC6, UL_TFC11, UL_TFC12, UL_TFC12, UL_TFC17, RB6: 103 bits RB7: 60 bits RB6: 103 bits	11	DL TFC11	UL TFC11	DL_TFC0_DL_TFC12		RB5: 81 bits	RB5: 81 bits
UL_TFC6, RB7: 60 bits UL_TFC11, RB8: 312 bits UL_TFC12, UL_TFC17, UL_TFC17,							
UL_TFC11, RB8: 312 bits RB8: 312 bit UL_TFC12, UL_TFC17,							
UL_TFC12, UL_TFC17,							RB8: 312 bits
UL_TFC17,							
					UL_TFC18,		
UL_TFC23							

NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).

14.2.38f.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2,3 and 4: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
 - for sub-test 5: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
 - for sub-test 6: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 7: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 8,9 and 10: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 11: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38g Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL:(12.2 7.95 5.9 4.75) kbps / CS RAB + Interactive or background / UL:16 DL:16 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38g.1 Conformance requirement

See clause 14.2.4.1.

14.2.38g.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38g.

14.2.38g.3 Method of test

See clause 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (16 kbps, 40 ms TTI)	DCCH
TFS	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
	TF2, bits	1x42	1x63	N/A	2x336	N/A
	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI (RB5, RB6, RB7, RB8, DCCH) UL_TFC0 (TF0, TF0, TF0, TF0, TF0) UL_TFC1 (TF1, TF0, TF0, TF0, TF0)
UL_TFC1 (TF1,TF0,TF0,TF0)
UL_TFC2 (TF2,TF1,TF0,TF0,TF0)
UL_TFC3 (TF3,TF2,TF0,TF0,TF0)
UL_TFC4 (TF4,TF3,TF0,TF0,TF0)
UL_TFC5 (TF5,TF4,TF1,TF0,TF0)
UL_TFC6 (TF0,TF0,TF1,TF0)
UL_TFC7 (TF1,TF0,TF1,TF0)
UL_TFC8 (TF3,TF2,TF0,TF1,TF0)
UL_TFC9 (TF4,TF3,TF0,TF1,TF0)
UL_TFC10 (TF5,TF4,TF1,TF1,TF0)
UL_TFC11 (TF0,TF0,TF0,TF2,TF0)
UL_TFC12 (TF1,TF0,TF0,TF2,TF0)
UL_TFC13 (TF3,TF2,TF0,TF2,TF0)
UL_TFC14 (TF4,TF3,TF0,TF2,TF0)
UL_TFC15 (TF5,TF4,TF1,TF2,TF0)
UL_TFC16 (TF0,TF0,TF0,TF1)
UL_TFC17 (TF1,TF0,TF0,TF0,TF1)
UL_TFC18 (TF2,TF1,TF0,TF0,TF1)
UL_TFC19 (TF3,TF2,TF0,TF0,TF1)
UL_TFC20 (TF4,TF3,TF0,TF0,TF1)
UL_TFC21 (TF5,TF4,TF1,TF0,TF1)
UL_TFC22 (TF0,TF0,TF1,TF1)
UL_TFC23 (TF1,TF0,TF1,TF1)
UL_TFC24 (TF3,TF2,TF0,TF1,TF1)
UL_TFC25 (TF4,TF3,TF0,TF1,TF1)
UL_TFC26 (TF5,TF4,TF1,TF1,TF1)
UL_TFC27 (TF0,TF0,TF2,TF1)
UL_TFC28 (TF1,TF0,TF0,TF2,TF1)
UL_TFC29 (TF3,TF2,TF0,TF2,TF1)
UL_TFC30 (TF4,TF3,TF0,TF2,TF1)
UL_TFC31 (TF5,TF4,TF1,TF2,TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (16 kbps, 40 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
TFS	TF2, bits	1x42	1x63	N/A	2x336	N/A
15	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0,TF0,TF0,TF0,TF0)
DL_TFC1	(TF1,TF0,TF0,TF0,TF0)
DL_TFC2	(TF2,TF1,TF0,TF0,TF0)
DL_TFC3	(TF3,TF2,TF0,TF0,TF0)
DL_TFC4	(TF4,TF3,TF0,TF0,TF0)
DL_TFC5	(TF5,TF4,TF1,TF0,TF0)
DL_TFC6	(TF0,TF0,TF0,TF1,TF0)
DL_TFC7	(TF1,TF0,TF0,TF1,TF0)
DL_TFC8	(TF2,TF1,TF0,TF1,TF0)
DL_TFC9	(TF3,TF2,TF0,TF1,TF0)
DL_TFC10	(TF4,TF3,TF0,TF1,TF0)
DL_TFC11	(TF5,TF4,TF1,TF1,TF0)
DL_TFC12	(TF0,TF0,TF0,TF2,TF0)
DL_TFC13	(TF1,TF0,TF0,TF2,TF0)
DL_TFC14	(TF2,TF1,TF0,TF2,TF0)
DL_TFC15	(TF3,TF2,TF0,TF2,TF0)
DL_TFC16	(TF4,TF3,TF0,TF2,TF0)
DL_TFC17	(TF5,TF4,TF1,TF2,TF0)
DL_TFC18	(TF0,TF0,TF0,TF1)
DL_TFC19	(TF1,TF0,TF0,TF0,TF1)
DL_TFC20	(TF2,TF1,TF0,TF0,TF1)
DL_TFC21	(TF3,TF2,TF0,TF0,TF1)
DL_TFC22	(TF4,TF3,TF0,TF0,TF1)
DL_TFC23	(TF5,TF4,TF1,TF0,TF1)
DL_TFC24	(TF0,TF0,TF0,TF1,TF1)
DL_TFC25	(TF1,TF0,TF0,TF1,TF1)
DL_TFC26	(TF2,TF1,TF0,TF1,TF1)
DL_TFC27	(TF3,TF2,TF0,TF1,TF1)
DL_TFC28	(TF4,TF3,TF0,TF1,TF1)
DL_TFC29	(TF5,TF4,TF1,TF1,TF1)
DL_TFC30	(TF0,TF0,TF0,TF2,TF1)
DL_TFC31	(TF1,TF0,TF0,TF2,TF1)
DL_TFC32	(TF2,TF1,TF0,TF2,TF1)
DL_TFC33	(TF3,TF2,TF0,TF2,TF1)
DL_TFC34	(TF4,TF3,TF0,TF2,TF1)
DL_TFC35	(TF5,TF4,TF1,TF2,TF1)

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
	under test	Under test			(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0, DL_TFC18,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
	DL_TFC19	UL_TFC17	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC16,	RB7: 60 bits	RB7: No data
2	DL_TFC2,	UL_TFC2,	DL_TFC0, DL_TFC18,	UL_TFC17 UL_TFC0,	RB8: 312 bits RB5: 42 bits	RB8: No data RB5: 42 bits
2	DL_TFC20	UL TFC18	UL_TFC0, UL_TFC16	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
				UL_TFC16,	RB7: 60 bits	RB7: No data
				UL_TFC18	RB8: 312 bits	RB8: No data
3	DL_TFC3,	UL_TFC3,	DL_TFC0, DL_TFC18,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
	DL_TFC21	UL_TFC19	UL_TFC0, UL_TFC16	UL_TFC3, UL_TFC16,	RB6: 63 bits RB7: 60 bits	RB6: 63 bits RB7: No data
				UL_TFC19	RB8: 312 bits	RB8: No data
4	DL_TFC4,	UL_TFC4,	DL_TFC0, DL_TFC18,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC22	UL_TFC20	UL_TFC0, UL_TFC16	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
				UL_TFC16,	RB7: 60 bits	RB7: No data
				UL_TFC20	RB8: 312 bits	RB8: No data
5	DL_TFC5, DL_TFC23	UL_TFC5, UL_TFC21	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16	UL_TFC0, UL_TFC5,	RB5: 81 bits RB6: 103 bits	RB5: 81 bits RB6: 103 bits
	DL_IF023	UL_IFC21		UL_TFC16,	RB7: 60 bits	RB7: 60 bits
				UL_TFC21	RB8: 312 bits	RB8: No data
6	DL_TFC6,	UL_TFC6,	DL_TFC0, DL_TFC18,	UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC24	UL_TFC22	UL_TFC0, UL_TFC16	UL_TFC6,	RB6: 103 bits	RB6: No data
				UL_TFC16,	RB7: 60 bits	RB7: No data
7				UL_TFC22	RB8: 312 bits	RB8: 312 bits
7	DL_TFC7, DL_TFC25	UL_TFC7, UL_TFC23	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16	UL_TFC0, UL_TFC1,	RB5: 39 bits RB6: 103 bits	RB5: 39 bits RB6: No data
	DL_11023	0L_11023		UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC7,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC22,		
8	DL_TFC8,	UL TFC8,	DL TFC0, DL TFC18,	UL_TFC23 UL_TFC0,	RB5: 55 bits	RB5: 42 bits
0	DL_TFC26	UL TFC24	UL_TFC0, UL_TFC16	UL_TFC3,	RB6: 63 bits	RB6: 53 bits
				UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC8,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16,		
				UL_TFC19,		
				UL_TFC22, UL_TFC24		
9	DL_TFC9,	UL_TFC8,	DL_TFC0, DL_TFC18,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
	DL_TFC27	UL_TFC24	UL_TFC0, UL_TFC16	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
				UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC8,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16,		
				UL_TFC19, UL_TFC22,		
				UL_TFC24		
10	DL_TFC10,	UL_TFC9,	DL_TFC0, DL_TFC18,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC28	UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
				UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16, UL_TFC20,		
				UL_TFC22,		
				UL_TFC25		
11	DL_TFC11,	UL_TFC10,	DL_TFC0, DL_TFC18,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC29	UL_TFC26	UL_TFC0, UL_TFC16	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC6,	RB7: 60 bits	RB7: 60 bits
				UL_TFC10, UL_TFC16,	RB8: 312 bits	RB8: 312 bits
				UL_TFC21,		
				UL_TFC22,		
				UL_TFC26		
1	1	1	1		1	1

RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload	10								
Image: Note of the second se	12	·							
Image: Construct of the second seco		DL_IFC30	UL_IFC2/	UL_IFC0, UL_IFC16					
13 DL_TFC13, DL_TFC31 UL_TFC12, UL_TFC28 UL_TFC0, UL_TFC16, UL_TFC16 UL_TFC0, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC23 RB5: 39 bits RB5: 30 bits RB5: No data RB7: No d									
DL_TFC31 UL_TFC28 UL_TFC0, UL_TFC16 UL_TFC1, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC27, UL_TFC32 RB6: 103 bits RB7: No data RB7: S5 bits RB7: No data RB7: S5 bits RB7: S5 bits RB7: No data RB7: S5 bits RB7: S5 bits RB7: No data RB7: No data RB7: No data RB7: S5 bits RB7: No data RB7:	10								
14 DL_TFC14, DL_TFC32 UL_TFC13, UL_TFC32 UL_TFC13, UL_TFC32 UL_TFC13, UL_TFC32 UL_TFC3, UL_TFC32 DL_TFC13, UL_TFC32 DL_TFC14, UL_TFC33 UL_TFC3, UL_TFC33 RB5: 55 bits UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC33 RB5: 55 bits RB5: 632 bits RB5: 632 bits RB5: 632 bits RB5: 55 bits RB5: 632 bits RB5: 632 bits RB5: 632 bits RB5: 632 bits RB5: 75 bits RB5: 632 bits RB5: 75 bits RB5: 632 bits RB5: 75 bits RB5: 75 bits RB5: 75 bits RB5: 75 bits RB5: 632 bits RB5: 75 bits RB5: 75 bits RB5: 75 bits RB5: 75 bits RB5: 75 bits RB5: 632 bits RB5: 75 bits RB5: 75 bits RB5: 75 bits RB5: 632 bits RB5: 75 bits RB5: 632 bits RB5: 75 bits RB5: 632 bits	13								
Image: Section of the section of t		DL_IFC31	UL_IFC28	UL_IFC0, UL_IFC16					
14 DL_TFC14, DL_TFC32 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC16, UL_TFC0, UL_TFC16, UL_TFC11, DL_TFC32 DL_TFC13, UL_TFC29 DL_TFC0, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC13, DL_TFC33 RB5: 55 bits RB6: 63 bits RB7: No data RB6: 63 bits RB7: No data 15 DL_TFC15, DL_TFC33 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC33 RB5: 55 bits RB6: 63 bits RB7: No data RB5: 55 bits RB7: No data 16 DL_TFC16, DL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC27, UL_TFC29 RB5: 75 bits RB7: No data 16 DL_TFC16, DL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC20, UL_TFC20, UL_TFC30 RB5: 75 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC31 UL_TFC35 RB5: 81 bits RB7: No data 17 DL_TFC17, DL_TFC35 DL_TFC3, UL_TFC3 DL_TFC0, DL_TFC18, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC21, UL_TFC31 RB5: 81 bits RB7: No data 17 DL_TFC17, DL_TFC35 DL_TFC3, UL_TFC31 DL_TFC31, UL_TFC31 RB5: 81 bits RB7: No data NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL T runder test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL_RCSDU size parameter has been set eet minus 8 bits (size of 7 bit length indicator					UL_IFC11,				
14 DL_TFC14, DL_TFC32 UL_TFC13, UL_TFC32 DL_TFC13, UL_TFC32 DL_TFC13, UL_TFC32 RB5: 55 bits RB5: 42 bits RB5: 55 bits RB5: 55 bits RB7: No data RB7: 60 bits RB7: 60 bits RB						RB8: 632 bits	RB8: 632 bits		
14 DL_TFC14, DL_TFC32 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16 UL_TFC0, UL_TFC17, UL_TFC11, UL_TFC11, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC29 RB5: 55 bits RB5: 63 bits RB5: 63 bits RB6: 63 bits RB7: No data 15 DL_TFC15, DL_TFC33 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC16, UL_TFC29 UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC34 DL_TFC14, UL_TFC30 DL_TFC0, DL_TFC16, UL_TFC14, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC34 RB5: 75 bits RB6: 75 bits RB6: 75 bits RB6: 84 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC16, UL_TFC0, UL_TFC16, UL_TFC30 UL_TFC0, DL_TFC16, UL_TFC3, UL_TFC3, UL_TFC30 RB5: 75 bits RB7: No data RB5: 75 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC16, UL_TFC0, UL_TFC16, UL_TFC30 UL_TFC0, RB5: 81 bits RB7: No data RB5: 81 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC16, UL_TFC16, UL_TFC20, UL_TFC21, UL_TFC31 RB5: 81 bits RB7: No data RB5: 81 bits RB7: No data NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indcator and expansion bit). The UL_RCSDU Size paramater has been set equal to the size of the payload <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>									
Image: Construct of the system of t									
14 DL_TFC14, DL_TFC32 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC17, UL_TFC11, UL_TFC11, UL_TFC11, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC14, UL_TFC13, UL_TFC29 RB5: 55 bits RB8: 632 bits RB8: 632 bits RB5: 55 bits RB8: 632 bits 15 DL_TFC15, DL_TFC33 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC10, UL_TFC29, UL_TFC11, UL_TFC11, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC3 RB5: 55 bits RB7: No data RB5: 55 bits RB7: No data 16 DL_TFC16, DL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC29 UL_TFC0, RB5: 75 bits RB7: No data RB8: 632 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC16, UL_TFC10, UL_TFC16, UL_TFC27, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC21, UL_									
DL_TFC32 UL_TFC29 UL_TFC0, UL_TFC16 UL_TFC11, UL_TFC13, UL_TFC13, UL_TFC14, UL_TFC19, UL_TFC29 RB6: 63 bits RB7: 60 bits R									
Image: space of the payload size of	14				_ ·				
UL_TFC13, UL_TFC16, UL_TFC33RB8: 632 bitsRB8: 632 bits15DL_TFC15, UL_TFC33UL_TFC13, UL_TFC29DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16UL_TFC0, UL_TFC3, UL_TFC3, UL_TFC11, UL_TFC11, RB7: 60 bits RB6: 63 bits RB7: 60 bits RB8: 632 bitsRB5: 55 bits RB6: 63 bits RB6: 63 bits RB7: No data RB8: 632 bits RB7: No data RB8: 632 bits16DL_TFC16, UL_TFC34UL_TFC14, UL_TFC34DL_TFC0, DL_TFC18, UL_TFC30UL_TFC19, UL_TFC27, UL_TFC16, UL_TFC16, UL_TFC14, UL_TFC27,RB5: 75 bits RB7: No data RB7: No data RB7: No data RB7: No data RB7: No data RB7: No data17DL_TFC17, UL_TFC35UL_TFC31, UL_TFC31DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC27, UL_TFC30,UL_TFC30, RB7: No data RB7:		DL_TFC32	UL_TFC29	UL_TFC0, UL_TFC16					
15 DL_TFC15, DL_TFC33 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC0, RB5: 55 bits UL_TFC33 RB5: 55 bits UL_TFC29 RB5: 55 bits UL_TFC0, UL_TFC16, UL_TFC13, UL_TFC13, UL_TFC13, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC29 RB5: 55 bits RB6: 63 bits RB7: No data RB8: 632 bits 16 DL_TFC16, UL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC16, UL_TFC20, UL_TFC16, UL_TFC27, UL_TFC29 UL_TFC14, RB7: 60 bits RB6: 84 bits RB5: 75 bits RB6: 84 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC16, UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC27, UL_TFC27, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC21, UL_TFC30 RB5: 81 bits RB6: 103 bits RB6: 103 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits RB7: 80 bits RB7: 80 bits RB7: 80 bits RB6: 103 bits RB7: 80 bits RB7: 60 b									
15 DL_TFC15, DL_TFC33 UL_TFC13, UL_TFC29 DL_TFC16, UL_TFC33 DL_TFC13, UL_TFC29 DL_TFC13, UL_TFC33 RB5: 55 bits RB6: 63 bits UL_TFC11, UL_TFC14, UL_TFC14, UL_TFC16, UL_TFC19, UL_TFC16, UL_TFC14, UL_TFC34 RB5: 55 bits RB7: No data RB7: No data RB8: 632 bits 16 DL_TFC16, UL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC18, UL_TFC16, UL_TFC16, UL_TFC14, UL_TFC14, UL_TFC29 RB5: 75 bits RB6: 84 bits RB7: No data RB8: 632 bits 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC17, DL_TFC35 DL_TFC15, UL_TFC31 RB5: 81 bits UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, DL_TFC35 RB5: 81 bits RB6: 103 bits RB6: 103 bits RB6: 103 bits RB7: 60 bits R					·	RB8: 632 bits	RB8: 632 bits		
15 DL_TFC15, DL_TFC33 UL_TFC13, UL_TFC29 DL_TFC13, UL_TFC29 DL_TFC13, UL_TFC13, UL_TFC14, UL_TFC13, UL_TFC14, UL_TFC14, DL_TFC34 RB5: 55 bits RB6: 63 bits RB7: No data RB7: No data RB8: 632 bits 16 DL_TFC16, DL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC18, UL_TFC0, DL_TFC16, UL_TFC29 UL_TFC29 16 DL_TFC34 UL_TFC30 DL_TFC0, DL_TFC16, UL_TFC30 UL_TFC20, UL_TFC30 RB5: 75 bits RB6: 84 bits RB6: 84 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC20, UL_TFC27, UL_TFC30 UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC20, UL_TFC27, UL_TFC30 RB5: 81 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC16, UL_TFC20, UL_TFC16, UL_TFC20, UL_TFC16, UL_TFC27, UL_TFC30 RB5: 81 bits RB7: 60 bit									
Image: construct of the systemUL_TFC13, UL_TFC13, UL_TFC13, UL_TFC0, UL_TFC16, UL_TFC3, UL_TFC3, UL_TFC33UL_TFC13, UL_TFC29RB5: 55 bits RB5: 55 bits RB5: 55 bits RB5: 55 bits RB7: No data UL_TFC13, UL_TFC13, UL_TFC19, UL_TFC19, UL_TFC19, UL_TFC19, UL_TFC19, UL_TFC27, UL_TFC27, UL_TFC29RB8: 632 bits RB8: 632 bits									
15 DL_TFC15, DL_TFC33 UL_TFC13, UL_TFC29 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16 UL_TFC0, UL_TFC3, UL_TFC13, UL_TFC11, DL_TFC13, UL_TFC16, UL_TFC16, UL_TFC19, UL_TFC27, UL_TFC29 RB5: 55 bits RB6: 63 bits RB7: No data RB8: 632 bits 16 DL_TFC16, DL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC14, DL_TFC30 UL_TFC14, UL_TFC30 DL_TFC30 RB5: 75 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC27, UL_TFC30 UL_TFC31 DL_TFC31 DL_TFC0, DL_TFC18, UL_TFC16, UL_TFC27, UL_TFC27, UL_TFC16, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC35 RB5: 81 bits RB7: No data 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC26, UL_TFC16, UL_TFC16, UL_TFC27, UL_TFC16, UL_TFC27, UL_TFC31 RB5: 81 bits RB7: 60 bits RB7: 60 bits RB7: 60 bits RB8: 632 bits									
DL_TFC33 UL_TFC29 UL_TFC0, UL_TFC16 UL_TFC3, UL_TFC3, UL_TFC1, RB6: 63 bits RB7: No data UL_TFC11, RB8: 632 bits RB6: 63 bits RB7: No data RB8: 632 bits 16 DL_TFC16, UL_TFC14, UL_TFC3, UL_TFC0, DL_TFC18, UL_TFC29 UL_TFC29 RB6: 63 bits RB7: No data 16 DL_TFC34 UL_TFC30 UL_TFC0, DL_TFC16, UL_TFC4, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC14, UL_TFC14, RB8: 632 bits RB5: 75 bits RB7: No data 17 DL_TFC17, UL_TFC15, UL_TFC3 DL_TFC0, DL_TFC16, UL_TFC16, UL_TFC5, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC5, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC5, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC5, UL_TFC4, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TF									
Image: here is the second se	15								
UL_TFC13, UL_TFC16, UL_TFC27, UL_TFC34RB8: 632 bitsRB8: 632 bits16DL_TFC16, UL_TFC34UL_TFC14, UL_TFC30DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16UL_TFC2, UL_TFC4, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC34DL_TFC30RB5: 75 bits RB6: 84 bits RB6: 84 bits RB7: No data RB8: 632 bits17DL_TFC17, DL_TFC35UL_TFC15, UL_TFC31DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16UL_TFC20, UL_TFC16, UL_TFC27, UL_TFC5, UL_TFC5, UL_TFC35RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 632 bitsNOTE:See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload		DL_TFC33	UL_TFC29	UL_TFC0, UL_TFC16					
UL_TFC16, UL_TFC27, UL_TFC34UL_TFC14, UL_TFC30DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16UL_TFC2916DL_TFC34UL_TFC14, UL_TFC30DL_TFC0, DL_TFC18, UL_TFC30UL_TFC4, UL_TFC11, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC16, UL_TFC27, UL_TFC30RB5: 75 bits RB6: 84 bits RB7: No data RB8: 632 bits17DL_TFC17, DL_TFC35UL_TFC15, UL_TFC31DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC35UL_TFC15, UL_TFC31RB5: 81 bits RB6: 103 bits RB7: No data RB7: 60 bits RB7: 60 bits RB7: 60 bits RB7: 60 bits RB8: 632 bitsRB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB7: 60 bits RB8: 632 bitsNOTE:See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
Image: Section of the section of th						RB8: 632 bits	RB8: 632 bits		
UL_TFC27, UL_TFC16,UL_TFC14, UL_TFC30DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC0, UL_TFC0, UL_TFC4, UL_TFC11, UL_TFC14, UL_TFC14, RB7: 60 bits RB7: 60 bits RB7: No data RB8: 632 bitsRB5: 75 bits RB6: 84 bits RB7: No data RB7: No data RB8: 632 bits RB8: 632 bits17DL_TFC17, DL_TFC35UL_TFC15, UL_TFC31DL_TFC0, DL_TFC18, UL_TFC0, DL_TFC18, UL_TFC16, UL_TFC27, UL_TFC11, UL_TFC230RB5: 81 bits RB7: No data RB8: 632 bits17DL_TFC17, DL_TFC35UL_TFC31DL_TFC0, DL_TFC18, UL_TFC31UL_TFC5, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC231RB5: 81 bits RB7: 60 bits RB8: 632 bitsNOTE:See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
Image: Construct of the payload size of the payloadUL_TFC19, UL_TFC19, RB5: 75 bits, RB5: 75 bits, RB6: 84 bits, RB6: 84 bits, RB6: 84 bits, RB7: No data, RB6: 84 bits, RB7: No data, UL_TFC14, UL_TFC14, RB8: 632 bits, RB7: No data, UL_TFC20, UL_TFC16, UL_TFC20, UL_TFC27, UL_TFC31UL_TFC0, UL_TFC18, UL_TFC10, RB5: 81 bits, RB5: 81 bits, RB6: 103 bits, RB6: 103 bits, RB7: 60 b									
16 DL_TFC16, DL_TFC34 UL_TFC14, UL_TFC30 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16 UL_TFC0, UL_TFC4, UL_TFC11, UL_TFC14, UL_TFC14, UL_TFC16, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC30 RB5: 75 bits RB6: 84 bits RB7: No data RB8: 632 bits 17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16, UL_TFC0, UL_TFC16, UL_TFC11, UL_TFC11, UL_TFC15, UL_TFC11, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC31 RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 632 bits NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
DL_TFC34 UL_TFC30 UL_TFC0, UL_TFC16 UL_TFC4, UL_TFC4, UL_TFC14, UL_TFC11, UL_TFC11, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC16, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC30 RB8: 632 bits RB8: 632 bits 17 DL_TFC17, UL_TFC15, UL_TFC31 DL_TFC0, UL_TFC16, UL_TFC16, UL_TFC5, UL_TFC5, UL_TFC31 UL_TFC31 UL_TFC16, UL_TFC5, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC21, UL_TFC31 RB8: 632 bits RB8: 632 bits RB8: 632 bits NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
Image: space of the payload size of	16				_ ·				
UL_TFC14, UL_TFC20, UL_TFC27, UL_TFC35RB8: 632 bitsRB8: 632 bits17DL_TFC17, DL_TFC35UL_TFC15, UL_TFC31DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16UL_TFC0, UL_TFC36RB5: 81 bits RB6: 103 bits RB6: 103 bits UL_TFC11, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC31RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB8: 632 bitsNOTE:See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload		DL_TFC34	UL_TFC30	UL_TFC0, UL_TFC16					
Image: space of the payload size of the payload									
17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16 UL_TFC0, UL_TFC5, UL_TFC1, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC31 RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB8: 632 bits NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload						RB8: 632 bits	RB8: 632 bits		
17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16 UL_TFC0, UL_TFC5, UL_TFC11, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21 RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB8: 632 bits NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16 UL_TFC0, UL_TFC5, UL_TFC11, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC31 RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB8: 632 bits NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
17 DL_TFC17, DL_TFC35 UL_TFC15, UL_TFC31 DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC16 UL_TFC0, UL_TFC5, UL_TFC11, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21 RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 632 bits NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
DL_TFC35 UL_TFC31 UL_TFC0, UL_TFC16 UL_TFC5, UL_TFC1, UL_TFC11, UL_TFC11, UL_TFC15, UL_TFC16, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC31 RB6: 103 bits RB7: 60 bits RB7: 60 bits RB8: 632 bits NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload RB7: 60 bits RB7: 60 bits RB8: 632 bits RB7: 60 bits RB8: 632 bits	17				_ ·				
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload		DL_TFC35	UL_TFC31	UL_TFC0, UL_TFC16					
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload					UL_TFC15,	RB8: 632 bits	RB8: 632 bits		
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload					UL_TFC16,				
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload					UL_TFC21,				
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload									
RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload	NOTE:								
	RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length								
size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).									
	size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).								

14.2.38g.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.

- 3. At step 15 the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2,3 and 4: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
 - for sub-test 5: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
 - for sub-test 6: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 7: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 8: RLC SDUs on RB5 having the first 42 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB6 having the first 53 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 9 and 10: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 11: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 12: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 13: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 14: RLC SDUs on RB5 having the first 42 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB6 having the first 53 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 15 and 16: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 17: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38h Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL:(12.2 7.95 5.9 4.75) kbps / CS RAB + Interactive or background / UL:32 DL:32 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38h.1 Conformance requirement

See clause 14.2.4.1.

14.2.38h.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38h.

14.2.38h.3 Method of test

See clause 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (32 kbps, 40 ms TTI)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
TFS	TF2, bits	1x42	1x63	N/A	2x336	N/A
115	TF3, bits	1x55	1x84	N/A	3x336	N/A
	TF4, bits	1x75	1x103	N/A	4x336	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0,TF0,TF0,TF0,TF0)
UL_TFC1	(TF0,TF0,TF0,TF1,TF0)
UL_TFC2	(TF0,TF0,TF0,TF2,TF0)
UL_TFC3	(TF0,TF0,TF0,TF4,TF0)
UL TFC4	(TF5,TF4,TF1,TF0,TF0)
UL_TFC5	(TF5,TF4,TF1,TF1,TF0)
UL_TFC6	(TF5,TF4,TF1,TF2,TF0)
UL_TFC7	(TF5,TF4,TF1,TF4,TF0)
UL_TFC8	(TF4,TF3,TF0,TF0,TF0)
UL_TFC9 UL_TFC10	(TF4,TF3,TF0,TF1,TF0)
UL_TFC10	(TF3,TF2,TF0,TF0,TF0)
	(TF2,TF1,TF0,TF0,TF0)
UL_TFC12	(TF1,TF0,TF0,TF0,TF0)
UL_TFC13	(TF1,TF0,TF0,TF1,TF0)
UL_TFC14	(TF1,TF0,TF0,TF2,TF0)
UL_TFC15	(TF1,TF0,TF0,TF4,TF0)
UL_TFC16	(TF0,TF0,TF0,TF0,TF1)
UL_TFC17	(TF0,TF0,TF1,TF1)
UL_TFC18	(TF0,TF0,TF0,TF2,TF1)
UL_TFC19	(TF0,TF0,TF0,TF4,TF1)
UL_TFC20	(TF5,TF4,TF1,TF0,TF1)
UL_TFC21	(TF5,TF4,TF1,TF1,TF1)
UL_TFC22	(TF5,TF4,TF1,TF2,TF1)
UL_TFC23	(TF5,TF4,TF1,TF4,TF1)
UL_TFC24	(TF4,TF3,TF0,TF0,TF1)
UL_TFC25	(TF4,TF3,TF0,TF1,TF1)
UL_TFC26	(TF3,TF2,TF0,TF0,TF1)
UL_TFC27	(TF2,TF1,TF0,TF0,TF1)
UL_TFC28	(TF1,TF0,TF0,TF1)
UL_TFC29	(TF1,TF0,TF0,TF1,TF1)
UL_TFC30	(TF1,TF0,TF0,TF2,TF1)
UL_TFC31	(TF1,TF0,TF0,TF4,TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (32 kbps, 40 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
TFS	TF2, bits	1x42	1x63	N/A	2x336	N/A
15	TF3, bits	1x55	1x84	N/A	3x336	N/A
	TF4, bits	1x75	1x103	N/A	4x336	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

TEOL	
TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0,TF0,TF0,TF0,TF0)
DL_TFC1	(TF0,TF0,TF0,TF1,TF0)
DL_TFC2	(TF0,TF0,TF0,TF2,TF0)
DL_TFC3	(TF0,TF0,TF0,TF4,TF0)
DL_TFC4	(TF5,TF4,TF1,TF0,TF0)
DL_TFC5	(TF5,TF4,TF1,TF1,TF0)
DL_TFC6	(TF5,TF4,TF1,TF2,TF0)
DL_TFC7	(TF5,TF4,TF1,TF4,TF0)
DL_TFC8	(TF4,TF3,TF0,TF0,TF0)
DL_TFC9	(TF4,TF3,TF0,TF1,TF0)
DL_TFC10	(TF4,TF3,TF0,TF2,TF0)
DL_TFC11	(TF4,TF3,TF0,TF4,TF0)
DL_TFC12	(TF3,TF2,TF0,TF0,TF0)
DL_TFC13	(TF3,TF2,TF0,TF1,TF0)
DL_TFC14	(TF3,TF2,TF0,TF2,TF0)
DL_TFC15	(TF3,TF2,TF0,TF4,TF0)
DL_TFC16	(TF2,TF1,TF0,TF0,TF0)
DL_TFC17	(TF2,TF1,TF0,TF1,TF0)
DL_TFC18	(TF2,TF1,TF0,TF2,TF0)
DL_TFC19	(TF2,TF1,TF0,TF4,TF0)
DL_TFC20	(TF1,TF0,TF0,TF0,TF0)
DL_TFC21	(TF1,TF0,TF0,TF1,TF0)
DL_TFC22	(TF1,TF0,TF0,TF2,TF0)
DL_TFC23	(TF1,TF0,TF0,TF4,TF0)
DL_TFC24	(TF0,TF0,TF0,TF1)
DL_TFC25	(TF0,TF0,TF0,TF1,TF1)
DL_TFC26	(TF0,TF0,TF0,TF2,TF1)
DL_TFC27	(TF0,TF0,TF0,TF4,TF1)
DL_TFC28	(TF5,TF4,TF1,TF0,TF1)
DL_TFC29	(TF5,TF4,TF1,TF1,TF1)
DL_TFC30	(TF5,TF4,TF1,TF2,TF1)
DL_TFC31	(TF5,TF4,TF1,TF4,TF1)
DL_TFC32	(TF4,TF3,TF0,TF0,TF1)
DL_TFC33	(TF4,TF3,TF0,TF1,TF1)
DL_TFC34	(TF4,TF3,TF0,TF2,TF1)
DL_TFC35	(TF4,TF3,TF0,TF4,TF1)
DL_TFC36	(TF3,TF2,TF0,TF0,TF1)
DL_TFC37	(TF3,TF2,TF0,TF1,TF1)
DL_TFC38	(TF3,TF2,TF0,TF2,TF1)
DL_TFC39	(TF3,TF2,TF0,TF4,TF1)
DL_TFC40	(TF2,TF1,TF0,TF0,TF1)
DL_TFC41	(TF2,TF1,TF0,TF1,TF1)
DL_TFC42	(TF2,TF1,TF0,TF2,TF1)
DL_TFC43	(TF2,TF1,TF0,TF4,TF1)
DL_TFC44	(TF1,TF0,TF0,TF0,TF1)
DL_TFC45	(TF1,TF0,TF0,TF1,TF1)
DL_TFC46	(TF1,TF0,TF0,TF2,TF1)
DL_TFC47	(TF1,TF0,TF0,TF4,TF1)

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
	under test	Under test			(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC25	UL_TFC17	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC16,	RB7: 60 bits	RB7: No data
_	DI 7700		D : TC D : TC C C	UL_TFC17	RB8: 312 bits	RB8: 312
2	DL_TFC2,	UL_TFC2,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC26	UL_TFC18	UL_TFC0, UL_TFC16	UL_TFC2, UL_TFC16,	RB6: 103 bits RB7: 60 bits	RB6: No data RB7: No data
				UL_TFC18	RB8: 632 bits	RB8: 632
3	DL_TFC3,	UL_TFC3,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC27	UL_TFC19	UL_TFC0, UL_TFC16	UL_TFC3,	RB6: 103 bits	RB6: No data
				UL_TFC16,	RB7: 60 bits	RB7: No data
				UL_TFC19	RB8: 1272 bits	RB8: 1272
4	DL_TFC4,	UL_TFC4,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC28	UL_TFC20	UL_TFC0, UL_TFC16	UL_TFC4, UL_TFC16,	RB6: 103 bits RB7: 60 bits	RB6: 103 bits RB7: 60 bits
				UL_TFC20	RB8: 312 bits	RB8: No data
5	DL_TFC5,	UL_TFC5,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC29	UL_TFC21	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 103 bits	RB6: 103 bits
				UL_TFC4,	RB7: 60 bits	RB7: 60 bits
				UL_TFC5,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16,		
				UL_TFC17, UL_TFC20,		
				UL_TFC20,		
6	DL_TFC6,	UL_TFC6,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
Ũ	DL_TFC30	UL_TFC22	UL_TFC0, UL_TFC16	UL_TFC2,	RB6: 103 bits	RB6: 103 bits
	_	_		UL_TFC4,	RB7: 60 bits	RB7: 60 bits
				UL_TFC6,	RB8: 632 bits	RB8: 632 bits
				UL_TFC16,		
				UL_TFC18, UL_TFC20,		
				UL_TFC22		
7	DL_TFC7,	UL_TFC7,	DL_TFC0, DL_TFC24,	UL TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC31	UL_TFC23	UL_TFC0, UL_TFC16	UL_TFC3,	RB6: 103 bits	RB6: 103 bits
				UL_TFC4,	RB7: 60 bits	RB7: 60 bits
				UL_TFC7,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC16,		
				UL_TFC19, UL_TFC20,		
				UL_TFC23		
8	DL_TFC8,	UL_TFC8,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC32	UL_TFC24	UL_TFC0, UL_TFC16	UL_TFC8,	RB6: 84 bits	RB6: 84 bits
				UL_TFC16,	RB7: 60 bits	RB7: No data
				UL_TFC24	RB8: 312 bits	RB8: No data
9	DL_TFC9, DL_TFC33	UL_TFC9, UL_TFC25	DL_TFC0, DL_TFC24, UL_TFC0, UL_TFC16	UL_TFC0, UL_TFC1,	RB5: 75 bits RB6: 84 bits	RB5: 75 bits RB6: 84 bits
	DL_11033	0L_1F020		UL_TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC24,		
10		UL_TFC9,		UL_TFC25	DDE: 75 hito	DDE: 75 hito
10	DL_TFC10, DL_TFC34	UL_TFC9, UL_TFC25	DL_TFC0, DL_TFC24, UL_TFC0, UL_TFC16	UL_TFC0, UL_TFC1,	RB5: 75 bits RB6: 84 bits	RB5: 75 bits RB6: 84 bits
	DL_11034	01_11020	0L_1100, 0L_1F010	UL_TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 632 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC24,		
				UL_TFC25		

					DD2 / :	
11	DL_TFC11,	UL_TFC9,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC35	UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 84 bits	RB6: 84 bits
				UL_TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 1272 bits
				UL_TFC16, UL_TFC17,		
				UL_TFC17,		
				UL_TFC24,		
12	DL_TFC12,	UL_TFC10,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
12	DL_TFC36	UL_TFC26	UL_TFC0, UL_TFC16	UL_TFC10,	RB6: 63 bits	RB6: 63 bits
	22_11 000	02_11 020		UL_TFC16,	RB7: 60 bits	RB7: No data
				UL_TFC26	RB8: 312 bits	RB8: No data
13	DL_TFC13,	UL_TFC9,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 75 bits	RB5: 55 bits
10	DL_TFC37	UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 84 bits	RB6: 63 bits
				UL_TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC24,		
L				UL_TFC25		
14	DL_TFC14,	UL_TFC9,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 75 bits	RB5: 55 bits
	DL_TFC38	UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 84 bits	RB6: 63 bits
				UL_TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 632 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC24,		
				UL_TFC25		
15	DL_TFC15,	UL_TFC9,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 75 bits	RB5: 55 bits
	DL_TFC39	UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 84 bits	RB6: 63 bits
				UL_TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 1272 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC24,		
10				UL_TFC25		
16	DL_TFC16,	UL_TFC11,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
	DL_TFC40	UL_TFC27	UL_TFC0, UL_TFC16	UL_TFC11,	RB6: 53 bits	RB6: 53 bits
				UL_TFC16, UL_TFC27	RB7: 60 bits	RB7: No data
17	DL_TFC17,	UL_TFC9,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB8: 312 bits RB5: 75 bits	RB8: No data RB5: 42 bits
17	DL_TFC17, DL_TFC41	UL_TFC9, UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC0,	RB5: 75 bits RB6: 84 bits	RB5: 42 bits RB6: 53 bits
		0L_1F020		UL_TFC1,	RB7: 60 bits	RB7: No data
				UL_TFC8,	RB8: 312 bits	RB8: 312 bits
				UL_TFC16,		100.012010
				UL_TFC10,		
				UL_TFC24,		
				UL_TFC25		
18	DL_TFC18,	UL_TFC9,	DL TFC0, DL TFC24,	UL_TFC0,	RB5: 75 bits	RB5: 42 bits
	DL_TFC42	UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 84 bits	RB6: 53 bits
	.			UL TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 632 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC24,		
				UL_TFC25		
19	DL_TFC19,	UL_TFC9,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 75 bits	RB5: 42 bits
	DL_TFC43	UL_TFC25	UL_TFC0, UL_TFC16	UL_TFC1,	RB6: 84 bits	RB6: 53 bits
				UL_TFC8,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 1272 bits
				UL_TFC16,		
				UL_TFC17,		
				UL_TFC24,		
L				UL_TFC25		
20	DL_TFC20,	UL_TFC12,	DL_TFC0, DL_TFC24,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
	DL_TFC44	UL_TFC28	UL_TFC0, UL_TFC16	UL_TFC12,	RB6: 103 bits	RB6: No data
				UL_TFC16,	RB7: 60 bits	RB7: No data
				UL_TFC28	RB8: 312 bits	RB8: No data

21	DL_TFC21, DL_TFC45	UL_TFC13, UL_TFC29	DL_TFC0, DL_TFC24, UL_TFC0, UL_TFC16	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC16, UL_TFC17, UL_TFC28, UL_TFC29	RB5: 39 bits RB6: 103 bits RB7: 60 bits RB8: 312 bits	RB5: 39 bits RB6: No data RB7: No data RB8: 312 bits	
22	DL_TFC22, DL_TFC46	UL_TFC14, UL_TFC30	DL_TFC0, DL_TFC24, UL_TFC0, UL_TFC16	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC16, UL_TFC18, UL_TFC28, UL_TFC30	RB5: 39 bits RB6: 103 bits RB7: 60 bits RB8: 632 bits	RB5: 39 bits RB6: No data RB7: No data RB8: 632 bits	
23	DL_TFC23, DL_TFC47	UL_TFC15, UL_TFC31	DL_TFC0, DL_TFC24, UL_TFC0, UL_TFC16	UL_TFC0, UL_TFC3, UL_TFC12, UL_TFC15, UL_TFC16, UL_TFC19, UL_TFC28, UL_TFC31	RB5: 39 bits RB6: 103 bits RB7: 60 bits RB8: 1272 bits	RB5: 39 bits RB6: No data RB7: No data RB8: 1272 bits	
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload						
			st minus 8 bits (the size of 7 bi				

14.2. 38h.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1,2 and 3: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
 - for sub-test 5, 6 and 7: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 8: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
 - for sub-test 9: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 10 and 11: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; RLC SDUs on RB8 having the same content as the first 312 bits of the test data sent by the SS in downlink; and no data shall be received on RB7.

- for sub-test 12: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
- for sub-test 13: RLC SDUs on RB5 having the first 55 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB6 having the first 63 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB7.
- for sub-test 14 and 15: RLC SDUs on RB5 having the first 55 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB6 having the first 63 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB8 having the same content as the first 312 bits of the test data sent by the SS in downlink; and no data shall be received on RB7.
- for sub-test 16: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
- for sub-test 17: RLC SDUs on RB5 having the first 42 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB6 having the first 53 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB7.
- for sub-test 18 and 19: RLC SDUs on RB5 having the first 42 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB6 having the first 53 bits equal to the content of the test data sent by the SS in downlink; RLC SDUs on RB8 having the same content as the first 312 bits of the test data sent by the SS in downlink; and no data shall be received on RB7.
- for sub-test 20: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
- for sub-test 21, 22 and 23: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38i Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL:(12.2 7.95 5.9 4.75) kbps / CS RAB + Interactive or background / UL:64 DL:64 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38i.1 Conformance requirement

See clause 14.2.4.1.

14.2.38i.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38i.

14.2.38i.3 Method of test

See clause 14.1.2 for test procedure.

Uplink TFS:

		TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
		TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
		TF1, bits	1x39	1x53	1x60	1x336	1x148
Ι.	TFS	TF2, bits	1x42	1x63	N/A	2x336	N/A
'	гэ	TF3, bits	1x55	1x84	N/A	3x336	N/A
		TF4, bits	1x75	1x103	N/A	4x336	N/A
		TF5, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0,TF0,TF0,TF0,TF0)
UL_TFC1	(TF1,TF0,TF0,TF0,TF0)
UL_TFC2	(TF2,TF1,TF0,TF0,TF0)
UL_TFC3	(TF3,TF2,TF0,TF0,TF0)
UL_TFC4	(TF4,TF3,TF0,TF0,TF0)
UL_TFC5	(TF5,TF4,TF1,TF0,TF0)
UL_TFC6	(TF0,TF0,TF0,TF1,TF0)
UL_TFC7	(TF1,TF0,TF0,TF1,TF0)
UL_TFC8	(TF2,TF1,TF0,TF1,TF0)
UL_TFC9	(TF3,TF2,TF0,TF1,TF0)
UL_TFC10	(TF4,TF3,TF0,TF1,TF0)
UL_TFC11	(TF5,TF4,TF1,TF1,TF0)
UL_TFC12	(TF0,TF0,TF0,TF2,TF0)
UL_TFC13	(TF1,TF0,TF0,TF2,TF0)
UL_TFC14	(TF2,TF1,TF0,TF2,TF0)
UL_TFC15	(TF3,TF2,TF0,TF2,TF0)
UL_TFC16	(TF4,TF3,TF0,TF2,TF0)
UL_TFC17	(TF5,TF4,TF1,TF2,TF0)
UL_TFC18	(TF0,TF0,TF0,TF4,TF0)
UL_TFC19	(TF1,TF0,TF0,TF4,TF0)
UL_TFC20	(TF2,TF1,TF0,TF4,TF0)
UL_TFC21	(TF3,TF2,TF0,TF4,TF0)
UL_TFC22	(TF4,TF3,TF0,TF4,TF0)
UL_TFC23	(TF5,TF4,TF1,TF4,TF0)
UL_TFC24	(TF0,TF0,TF0,TF1)
UL_TFC25	(TF1,TF0,TF0,TF0,TF1)
UL_TFC26	(TF2,TF1,TF0,TF0,TF1)
UL_TFC27	(TF3,TF2,TF0,TF0,TF1)
UL_TFC28	(TF4,TF3,TF0,TF0,TF1)
UL_TFC29	(TF5,TF4,TF1,TF0,TF1)
UL_TFC30	(TF0,TF0,TF0,TF1,TF1)
UL_TFC31	(TF1,TF0,TF0,TF1,TF1)
UL_TFC32	(TF2,TF1,TF0,TF1,TF1)
UL_TFC33	(TF3,TF2,TF0,TF1,TF1)
UL_TFC34 UL_TFC35	(TF4,TF3,TF0,TF1,TF1) (TF5,TF4,TF1,TF1,TF1)
UL_TFC36	(TF0,TF0,TF0,TF2,TF1)
UL TFC36	(TF1,TF0,TF0,TF2,TF1)
UL_TFC38	(TF2,TF1,TF0,TF2,TF1)
UL_TFC39	(TF3,TF2,TF0,TF2,TF1)
UL_TFC40	(TF4,TF3,TF0,TF2,TF1)
UL_TFC41	(TF5,TF4,TF1,TF2,TF1)
UL_TFC42	(TF0,TF0,TF0,TF4,TF1)
UL_TFC43	(TF1,TF0,TF0,TF4,TF1)
UL_TFC44	(TF2,TF1,TF0,TF4,TF1)
UL_TFC45	(TF3,TF2,TF0,TF4,TF1)
UL_TFC46	(TF4,TF3,TF0,TF4,TF1)
UL_TFC47	(TF5,TF4,TF1,TF4,TF1)
0L_1104/	(, , , , , , , , , , , , , , , , , , ,

Release 5

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
TFS	TF2, bits	1x42	1x63	N/A	2x336	N/A
115	TF3, bits	1x55	1x84	N/A	3x336	N/A
	TF4, bits	1x75	1x103	N/A	4x336	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0,TF0,TF0,TF0,TF0)
DL_TFC1	(TF1,TF0,TF0,TF0,TF0)
DL_TFC2	(TF2,TF1,TF0,TF0,TF0)
DL_TFC3	(TF3,TF2,TF0,TF0,TF0)
DL_TFC4	(TF4,TF3,TF0,TF0,TF0)
DL_TFC5	(TF5,TF4,TF1,TF0,TF0)
DL_TFC6	(TF0,TF0,TF0,TF1,TF0)
DL_TFC7	(TF1,TF0,TF0,TF1,TF0)
DL_TFC8	(TF2,TF1,TF0,TF1,TF0)
DL_TFC9	(TF3,TF2,TF0,TF1,TF0)
DL_TFC10	(TF4,TF3,TF0,TF1,TF0)
DL_TFC11	(TF5,TF4,TF1,TF1,TF0)
DL_TFC12 DL_TFC13	(TF0,TF0,TF0,TF2,TF0)
	(TF1,TF0,TF0,TF2,TF0)
DL_TFC14	(TF2,TF1,TF0,TF2,TF0)
DL_TFC15	(TF3,TF2,TF0,TF2,TF0)
DL_TFC16	(TF4,TF3,TF0,TF2,TF0)
DL_TFC17	(TF5,TF4,TF1,TF2,TF0)
DL_TFC18 DL_TFC19	(TF0,TF0,TF0,TF3,TF0) (TE1,TE0,TE0,TE3,TE0)
DL_TFC19 DL_TFC20	(TF1,TF0,TF0,TF3,TF0) (TF2,TF1,TF0,TF3,TF0)
	(TF2,TF1,TF0,TF3,TF0)
DL_TFC21	(TF3,TF2,TF0,TF3,TF0)
DL_TFC22 DL_TFC23	(TF4,TF3,TF0,TF3,TF0) (TF5,TF4,TF1,TF3,TF0)
DL_TFC23	(TF0,TF0,TF0,TF4,TF0)
DL_TFC24	(TF1,TF0,TF0,TF4,TF0)
DL_TFC25	(TF2,TF1,TF0,TF4,TF0)
DL_TFC20	(TF3,TF2,TF0,TF4,TF0)
DL_TFC27	(TF4,TF3,TF0,TF4,TF0)
-	(TF5,TF4,TF1,TF4,TF0)
DL_TFC29 DL_TFC30	(TF0,TF0,TF0,TF0,TF1)
DL_TFC31	(TF1,TF0,TF0,TF0,TF1)
DL_TFC32	(TF2,TF1,TF0,TF0,TF1)
DL_TFC33	(TF3,TF2,TF0,TF0,TF1)
DL_TFC34	(TF4,TF3,TF0,TF0,TF1)
DL_TFC35	(TF5,TF4,TF1,TF0,TF1)
DL_TFC36	(TF0,TF0,TF0,TF1,TF1)
DL TFC37	(TF1,TF0,TF0,TF1,TF1)
DL_TFC38	(TF2,TF1,TF0,TF1,TF1)
DL_TFC39	(TF3,TF2,TF0,TF1,TF1)
DL_TFC40	(TF4,TF3,TF0,TF1,TF1)
DL_TFC41	(TF5,TF4,TF1,TF1,TF1)
DL_TFC42	(TF0,TF0,TF0,TF2,TF1)
DL_TFC43	(TF1,TF0,TF0,TF2,TF1)
DL_TFC44	(TF2,TF1,TF0,TF2,TF1)
DL_TFC45	(TF3,TF2,TF0,TF2,TF1)
DL_TFC46	(TF4,TF3,TF0,TF2,TF1)
DL TFC47	(TF5,TF4,TF1,TF2,TF1)
DL_TFC48	(TF0,TF0,TF0,TF3,TF1)
DL_TFC49	(TF1,TF0,TF0,TF3,TF1)
DL_TFC50	(TF2,TF1,TF0,TF3,TF1)
DL_TFC51	(TF3,TF2,TF0,TF3,TF1)
DL_TFC52	(TF4,TF3,TF0,TF3,TF1)
DL_TFC53	(TF5,TF4,TF1,TF3,TF1)
DL_TFC54	(TF0,TF0,TF0,TF4,TF1)
DL_TFC55	(TF1,TF0,TF0,TF4,TF1)
DL_TFC56	(TF2,TF1,TF0,TF4,TF1)
DL_TFC57	(TF3,TF2,TF0,TF4,TF1)
DL_TFC58	(TF4,TF3,TF0,TF4,TF1)
DL_TFC59	(TF5,TF4,TF1,TF4,TF1)

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
					(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
	DL_TFC31	UL_TFC25	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC24,	RB7: 60 bits	RB7: No data
2	DL_TFC2,	UL_TFC2,	DL_TFC0, DL_TFC30,	UL_TFC25 UL_TFC0,	RB8: 312 bits RB5: 42 bits	RB8: No data RB5: 42 bits
2	DL_TFC2, DL_TFC32	UL_TFC26	UL_TFC0, UL_TFC24	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
	DL_11002	02_11 020		UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC26	RB8: 312 bits	RB8: No data
3	DL_TFC3,	UL_TFC3,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
	DL_TFC33	UL_TFC27	UL_TFC0, UL_TFC24	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
				UL_TFC24,	RB7: 60 bits	RB7: No data
4	DL_TFC4,	UL_TFC4,	DL_TFC0, DL_TFC30,	UL_TFC27 UL_TFC0,	RB8: 312 bits RB5: 75 bits	RB8: No data RB5: 75 bits
4	DL_TFC4, DL_TFC34	UL_TFC28	UL_TFC0, UL_TFC24	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
	DL_11 001	02_11 020		UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC28	RB8: 312 bits	RB8: No data
5	DL_TFC5,	UL_TFC5,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC35	UL_TFC29	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC24,	RB7: 60 bits	RB7: 60 bits
6	DL_TFC6,	UL_TFC6,	DL_TFC0, DL_TFC30,	UL_TFC29 UL_TFC0,	RB8: 312 bits RB5: 81 bits	RB8: No data RB5: No data
	DL_TFC36	UL_TFC30	UL_TFC0, UL_TFC24	UL TFC6,	RB6: 103 bits	RB6: No data
			,	UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC30	RB8: 312 bits	RB8: 312 bits
7	DL_TFC7,	UL_TFC7,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
	DL_TFC37	UL_TFC31	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC6, UL_TFC7,	RB7: 60 bits RB8: 312 bits	RB7: No data RB8: 312 bits
				UL_TFC24,	RDO. 312 DIIS	RDO. 312 DIIS
				UL_TFC25,		
				UL_TFC30,		
				UL_TFC31		
8	DL_TFC8,	UL_TFC8,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
	DL_TFC38	UL_TFC32	UL_TFC0, UL_TFC24	UL_TFC2, UL_TFC6,	RB6: 53 bits RB7: 60 bits	RB6: 53 bits RB7: No data
				UL_TFC8,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24,		
				UL_TFC26,		
				UL_TFC30,		
				UL_TFC32		
9	DL_TFC9, DL_TFC39	UL_TFC9, UL_TFC33	DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24	UL_TFC0, UL_TFC3,	RB5: 55 bits RB6: 63 bits	RB5: 55 bits RB6: 63 bits
	DL_11039	02_11033	0L_1100, 0L_1F024	UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24,		
				UL_TFC27,		
				UL_TFC30,		
10	DL_TFC10,	UL_TFC10,	DL_TFC0, DL_TFC30,	UL_TFC33 UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC40	UL_TFC34	UL TFC0, UL TFC24	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
				UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC10,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24,		
				UL_TFC28,		
				UL_TFC30, UL_TFC34		
11	DL_TFC11,	UL_TFC11,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC41	UL_TFC35	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC6,	RB7: 60 bits	RB7: 60 bits
				UL_TFC11,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24,		
				UL_TFC29, UL_TFC30,		
				UL_TFC35		
L	1	L	1	02_11000	1	l

Release 5

DL.TFC42 UL_TFC36 UL_TFC36 UL_TFC37 UL_TFC36 RB6: 103 bits RB7: No data RB7: No	4.0						
UL_TFC24, UL_TFC36 RB7: 80 bits RB5: 82 bits RB7: No data RB5: 83 bits RB5: 80 bits RB7: No data RB7: No	12	DL_TFC12,	UL_TFC12,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: No data
Image: constraint of the system is a system in the system is a syst		DL_IFC42	UL_IFC36	UL_1FC0, UL_1FC24			
13 DL_TFC13, DL_TFC43 UL_TFC17, UL_TFC37 DL_TFC0, DL_TFC30, UL_TFC4 UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC38, RBS: 39 bits RBS: 00 b							
DL_TFC43 UL_TFC37 UL_TFC0, UL_TFC24 UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC34, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC37, UL_TFC38, UL_TFC	12						
UL, TFC12, UL, TFC13, UL, TFC38 RB7: 60 bits RB8: 632 bits RB7: No data RB8: 632 bits 14 DL, TFC14, DL, TFC44 UL, TFC14, UL, TFC38 UL, TFC30, UL, TFC30, UL, TFC24, UL, TFC44 UL, TFC14, UL, TFC44, UL, TF	13						
14 DL_TFC14, UL_TFC24, UL_TFC36, UL_TFC36, UL_TFC37 RB8: 632 bits RB3: 52 bits 14 DL_TFC14, DL_TFC44 UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC36, UL_TFC12, UL_TFC14, DL_TFC44 UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC37, UL_TFC37, UL_TFC38, UL_TFC39, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TF		DL_11043	02_11037	0L_1100, 0L_11024			
14 DL_TFC14, DL_TFC34, UL_TFC36, UL_TFC36, UL_TFC34 UL_TFC14, UL_TFC38, UL_TFC38, UL_TFC34 UL_TFC37, UL_TFC36, UL_TFC24 RB5: 42 bits RB5: 55 bits RB7: No data 14 DL_TFC14, DL_TFC44 UL_TFC38, UL_TFC38 UL_TFC30, UL_TFC24 UL_TFC27, UL_TFC24, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC39 RB5: 55 bits RB7: No data RB5: 55 bits RB6: 63 bits RB7: No data 15 DL_TFC15, DL_TFC45 UL_TFC15, UL_TFC39 UL_TFC0, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC38, UL_TFC39, UL_TFC39, UL_TFC39, UL_TFC39, UL_TFC48 RB5: 81 bits RB6: 103 bits RB7: No data 17 D_TFC48, UL_TFC48, UL_TFC48, UL_TFC44,							
14 DL_TFC14, DL_TFC34, UL_TFC38, UL_TFC34, DL_TFC44 UL_TFC14, UL_TFC38, UL_TFC37, UL_TFC37, UL_TFC37, UL_TFC37, UL_TFC38, UL_TFC34, UL_TFC34, UL_TFC34, UL_TFC34, UL_TFC34, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC37, UL_TFC37, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC37, UL_TFC36, UL_TFC37, UL_TFC36, UL_TFC37, UL_TFC37, UL_TFC36, UL_TFC37, UL_TFC38, UL_TFC3						1100. 032 0113	ND0. 052 bits
14 DL_TFC14, DL_TFC44 UL_TFC17, UL_TFC38 UL_TFC0, DL_TFC30, UL_TFC24 UL_TFC37, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC26, UL_TFC26, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC38, UL_TFC39 RB5: 42 bits RB7: No data RB5: 55 bits RB7: No data RB5: 55 bits RB7: No data RB5: 63 bits RB7: No data RB5: 75 bits RB7: No data RB5: 75 bits RB7: No data RB5: 63 bits RB7: No data RB5: 75 bits RB7: No data RB5: 63 bits RB5: 75 bits RB7: No data RB5: 63 bits RB5: 75 bits RB5: 63 bits RB5: 60 bits RB5: 60 bits RB5: 60 bits RB5: 76 bits RB5: 75 bits RB5							
Ide DL_TFC14, DL_TFC4, UL_TFC37 UL_TFC37 RB5: 42 bits RB5: 63 bits RB5: 63 bits RB7: 80							
14 DL_TFC14, DL_TFC44 UL_TFC14, UL_TFC38 UL_TFC0, UL_TFC2, UL_TFC24 RB5: 42 bits RB5: 53 bits UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC26, UL_TFC26, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC37, UL_TFC45 RB5: 55 bits RB5: 63 bits RB5: 75 bits RB5: 81 bits RB5: 82 bits RB5: 81 bits RB5: 81 bits RB5: 81 bits RB5: 81 bits RB5: 81 bits RB5: 82 bits RB5: 81 bits RB5: 82 bits RB5: 82 bits RB5: 81 b							
DL_TFC14 UL_TFC138 UL_TFC0, UL_TFC24 UL_TFC12, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC14, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC37, UL_TFC34, UL_TFC36, UL_TFC37, UL_TFC44, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC46, UL_TFC37, UL_TFC46, UL_TFC47, UL_TFC46, UL_TFC47, UL_TFC46, UL_TFC47, UL_TFC47, UL_TFC47, UL_TFC47, UL_TFC44, UL_TFC36, UL_TFC36, UL_TFC47, UL_TFC41, UL_TFC44, UL_TFC36, UL_TFC44, UL_TFC36, UL_TFC44, UL_TFC36, UL_TFC44, UL_TFC46, UL_TFC46, UL_TFC44, UL_TF	14	DI TEC14	UL TEC14	DI TECO DI TEC30		RB5: 42 bits	RB5: 42 hits
Image: Line and the system of the s							
Image: Non-State State UL_TFC14, UL_TFC24, UL_TFC34, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC37, UL_TFC39, UL_TFC39, UL_TFC39, UL_TFC39, UL_TFC39, UL_TFC39, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC30, UL_TFC41 RB5: 81 bits RB5: 81 bits RB5: 103 bits RB3: 632 bits RB7: 60 bits			02_11 000				
Image:							
Image:							
Image: Construct of the system of t							
Image: Construct of the system of t							
15 DL_TFC15, DL_TFC45 UL_TFC16, UL_TFC39 DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24 UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC24, UL_TFC24, UL_TFC38 RB5: 55 bits RB7: 80 bits RB7: 80 bits RB7: No data 16 DL_TFC16, DL_TFC46 UL_TFC17, UL_TFC46 UL_TFC17, UL_TFC40 DL_TFC40, UL_TFC39 RB5: 75 bits RB7: 80 bits RB							
DL_TFC45 UL_TFC39 UL_TFC0, UL_TFC24 UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC27, UL_TFC27, UL_TFC27, UL_TFC28, UL_TFC27, UL_TFC39 RB6: 63 bits RB7: 60 bits RB6: 63 bits RB6: 63 bits RB7: No data 16 DL_TFC16, DL_TFC46 UL_TFC16, UL_TFC46 UL_TFC16, UL_TFC0, UL_TFC30, UL_TFC24, UL_TFC12, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC47 RB5: 75 bits RB6: 75 bits RB7: No data 17 DL_TFC17, DL_TFC47 UL_TFC17, UL_TFC41 DL_TFC0, DL_TFC30, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC47 RB5: 81 bits RB6: 103 bits RB7: No data RB7: No data RB7: No data RB7: No data RB6: 102 bits RB7: No data RB6: 102 bits RB6: 102 bits RB6	15	DL_TFC15,	UL_TFC15,		UL_TFC0,	RB5: 55 bits	RB5: 55 bits
Image: Large							RB6: 63 bits
Image:							RB7: No data
Image:						RB8: 632 bits	RB8: 632 bits
16 DL_TFC16, DL_TFC46 UL_TFC16, UL_TFC46 UL_TFC16, UL_TFC40 DL_TFC0, DL_TFC30, UL_TFC24 UL_TFC4, UL_TFC4, UL_TFC1, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC26, UL_TFC26, UL_TFC26, UL_TFC26, UL_TFC27, DL_TFC47 RB5: 75 bits RB6: 84 bits RB7: No data 17 DL_TFC17, DL_TFC47 UL_TFC17, UL_TFC41 DL_TFC0, DL_TFC30, UL_TFC20, UL_TFC24 UL_TFC40, UL_TFC24, UL_TFC27, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 70 data RB7: 70 dat							
Image: Construct of the system of t							
16 DL_TFC16, DL_TFC46 UL_TFC16, UL_TFC40 DL_TFC0, UL_TFC0, UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC36, UL_TFC36, UL_TFC47 RB5: 75 bits RB6: 84 bits RB7: 80 bits RB7: 80 bits RB7: 80 bits RB8: 83 bits 17 DL_TFC17, DL_TFC47 UL_TFC17, UL_TFC47 DL_TFC17, UL_TFC47 DL_TFC17, UL_TFC47 DL_TFC17, UL_TFC47 DL_TFC17, UL_TFC47 RB5: 81 bits UL_TFC47 RB5: 81 bits UL_TFC47 RB5: 81 bits UL_TFC47 RB5: 81 bits RB7: 60 bits RB7: 60 bits RB5: 81 bits RB7: 60 bits RB5: 81 bits RB7: 60 bits RB5: 81 bits RB7: 60 bits 18 DL_TFC18, DL_TFC48 UL_TFC18, UL_TFC49 DL_TFC19, UL_TFC0, UL_TFC24 UL_TFC0, UL_TFC44 RB5: 81 bits RB7: 60 bits RB5: No data RB7: No data 19 DL_TFC19, DL_TFC49 UL_TFC19, UL_TFC43 DL_TFC0, UL_TFC30, UL_TFC44 UL_TFC18, UL_TFC44 RB6: 103 bits RB7: No data RB6: 103 bits RB7: No data 20 DL_TFC20, UL_TFC49 UL_TFC44, UL_TFC44 UL_TFC30, UL_TFC44 RB5: 42 bits UL_TFC44 RB5: 42 bits RB7: No data 20 DL_TFC20, UL_TFC49 UL_TFC44, UL_							
DL_TFC46 UL_TFC40 UL_TFC0, UL_TFC24 UL_TFC4, UL_TFC12, UL_TFC12, UL_TFC24, UL_TFC24, UL_TFC26, UL_TFC36, UL_TFC36, UL_TFC47 RB6: 84 bits RB7: 60 bits RB6: 84 bits RB7: 60 bits 17 DL_TFC17, DL_TFC47 UL_TFC17, UL_TFC41 DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24 UL_TFC0, UL_TFC12, UL_TFC12, UL_TFC41 RB5: 81 bits RB5: 81 bits RB6: 103 bits RB7: 60 bits 18 DL_TFC18, DL_TFC48 UL_TFC42 UL_TFC0, DL_TFC30, UL_TFC24, UL_TFC41 UL_TFC41 RB5: 81 bits RB5: 81 bits RB7: 60 bits RB7: 70 data RB7: 70 data							
UL_TFC12, UL_TFC36, UL_TFC36, UL_TFC36, UL_TFC47RB7: 60 bits RB8: 632 bitsRB7: No data RB8: 632 bits17DL_TFC17, DL_TFC47UL_TFC17, UL_TFC41DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC40UL_TFC36, UL_TFC17, UL_TFC17, UL_TFC41UL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, RB5: 81 bits RB7: 60 bits RB7: 80 bits <td>16</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	16						
Image: Large stateUL_TFC16, UL_TFC24, UL_TFC24, UL_TFC28, UL_TFC28, UL_TFC28, UL_TFC30, UL_TFC30, UL_TFC40RB8: 632 bitsRB8: 632 bits17DL_TFC17, DL_TFC17, UL_TFC17, UL_TFC41UL_TFC0, DL_TFC30, UL_TFC24, UL_TFC40UL_TFC0, RB5: 81 bits, UL_TFC17, RB7: 60 bits, UL_TFC17, UL_TFC41RB5: 81 bits, RB7: 60 bits, RB7: 60 bits, RB8: 632 bitsRB5: 81 bits, RB7: 60 bits, RB7: 60 bits, RB8: 632 bits18DL_TFC18, UL_TFC18, UL_TFC48UL_TFC0, DL_TFC0, DL_TFC30, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC41UL_TFC41RB5: 81 bits, RB7: 60 bits, RB8: 632 bits19DL_TFC19, UL_TFC49UL_TFC43DL_TFC0, DL_TFC30, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC25, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC25, UL_TFC24, UL_TF		DL_TFC46	UL_TFC40	UL_TFC0, UL_TFC24			
UL_TFC24, UL_TFC28, UL_TFC40 UL_TFC24, UL_TFC36, UL_TFC40 RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 80 data RB7: No data RB7: N							
Image: Line of the system of the sy						RB8: 632 bits	RB8: 632 bits
Image: Instant of the systemUL_TFC20, UL_TFC30, UL_TFC30, UL_TFC30, UL_TFC47UL_TFC17, UL_TFC47UL_TFC17, UL_TFC41DL_TFC0, UL_TFC30, UL_TFC24UL_TFC5, UL_TFC17, UL_TFC41RB5: 81 bits, RB6: 103 bits, RB6: 103 bits, RB7: 60 bits, UL_TFC29, UL_TFC24, UL_TFC29, UL_TFC24, UL_TFC29, UL_TFC24, UL_TFC24, UL_TFC41RB3: 632 bits, RB8: 632 bits,							
Image: Construct of the system of t							
17 DL_TFC17, DL_TFC47 UL_TFC17, UL_TFC41 DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24 UL_TFC0, UL_TFC17, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC24, UL_TFC24, UL_TFC24, RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 632 bits RB5: 81 bits RB7: 60 bits RB7: 60 bits RB8: 632 bits 18 DL_TFC18, DL_TFC48 UL_TFC18, UL_TFC48 UL_TFC18, UL_TFC42 DL_TFC0, DL_TFC30, UL_TFC24 UL_TFC41 RB5: 81 bits RB8: 632 bits RB5: No data RB6: 103 bits RB8: 632 bits 19 DL_TFC19, DL_TFC49 UL_TFC19, UL_TFC43 DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24 UL_TFC1, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC42, UL_TFC43 RB5: 39 bits RB6: 103 bits RB7: No data RB7: No data RB8: 1272 bits 20 DL_TFC20, DL_TFC20, UL_TFC20, UL_TFC43 DL_TFC20, UL_TFC43 DL_TFC20, UL_TFC43 RB5: 42 bits RB7: No data RB7: No data RB7: No data RB7: No data RB7: No data RB7: No data RB8: 1272 bits							
DL_TFC47UL_TFC41UL_TFC0, UL_TFC24UL_TFC2, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC12, UL_TFC24, UL_TFC24, UL_TFC28, UL_TFC28, UL_TFC48RB6: 103 bits RB7: 60 bits RB8: 632 bitsRB6: 103 bits RB7: 60 bits RB8: 632 bits18DL_TFC18, DL_TFC48UL_TFC18, UL_TFC48DL_TFC0, DL_TFC30, UL_TFC42UL_TFC18, UL_TFC48, UL_TFC42UL_TFC48, UL_TFC48, UL_TFC48, UL_TFC48, UL_TFC49DL_TFC19, UL_TFC43DL_TFC0, DL_TFC30, UL_TFC0, DL_TFC30, UL_TFC43UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC19, UL_TFC43RB5: No data RB6: 103 bits RB6: 103 bits RB6: 103 bits RB7: No data RB7: No data20DL_TFC20, UL_TFC20, UL_TFC20, UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, DL_TFC30, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24,RB5: 42 bits RB6: 103 bits RB7: No data RB7: No data RB7: No data RB7: No data RB8: 1272 bits20DL_TFC20, UL_TFC20, UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, RB8: 1272 bitsRB5: 42 bits RB6: 53 bits RB7: No data RB7: No data RB7: No data RB7: No data RB7: No data RB8: 952 bits	17					DD5: 91 bito	DD5: 91 hito
UL_TFC2, UL_TFC4RB7: 60 bits RB8: 632 bitsRB7: 60 bits RB8: 632 bitsRB7: 60 bits RB8: 632 bits18DL_TFC18, DL_TFC48UL_TFC18, UL_TFC48UL_TFC18, UL_TFC42DL_TFC0, DL_TFC30, UL_TFC41UL_TFC41RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 60 bits RB7: 80 data RB7: 80 data RB8: 952 bits20DL_TFC20, UL_TFC20, UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC24,UL_TFC0, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC18, RB7: 80 bits RB7: 80 bits RB7: 80 data RB7: 80 data RB8: 952 bits<	17						
UL_TFC17, UL_TFC24, UL_TFC36, UL_TFC41RB8: 632 bitsRB8: 632 bits18DL_TFC18, DL_TFC48UL_TFC18, UL_TFC42DL_TFC0, DL_TFC30, UL_TFC4, UL_TFC0, UL_TFC24UL_TFC4, UL_TFC41RB5: 81 bits RB6: 103 bits RB7: 60 bits RB7: 80 data RB7: 80 data RB7: 80 data RB7: 80 data RB7: 80 data RB7: 80 bits RB5: 39 bits RB5: 39 bits RB5: 1272 bits RB5: 12			00_11041	0L_1100, 0L_11024			
UL_TFC24, UL_TFC36, UL_TFC48UL_TFC18, UL_TFC48UL_TFC18, UL_TFC42UL_TFC0, DL_TFC30, UL_TFC41UL_TFC41RB5: 81 bits RB6: 103 bits RB7: No data RB7: No data RB8: 952 bits20DL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24, UL_TFC18, UL_TFC18, UL_TFC18, RB7: 60 bits RB7: No data RB8: 952 bitsRB5: 42 bits RB7: No data RB8: 952 bits							
Image: log bit stateUL_TFC20, UL_TFC23, UL_TFC36, UL_TFC41UL_TFC18, UL_TFC18, UL_TFC42UL_TFC41, UL_TFC41RB5: 81 bits RB5: No data18DL_TFC48UL_TFC42UL_TFC0, UL_TFC0, UL_TFC24UL_TFC18, UL_TFC18, UL_TFC42RB6: 103 bits RB6: No data19DL_TFC19, UL_TFC19, UL_TFC43UL_TFC0, UL_TFC0, UL_TFC30, UL_TFC1, UL_TFC14, UL_TFC24, RB8: 1272 bitsRB5: 42 bits RB5: 42 bits RB6: 53 bits RB7: No data RB7: S0 bits RB6: 53 bits RB6: 53 bits RB6: 53 bits RB7: No data RB7: No data RB7: S0 bits RB7: S0 bit							100.002 010
Image: line line line line line line line line							
18DL_TFC18, DL_TFC48UL_TFC18, UL_TFC42DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC18, UL_TFC18, UL_TFC24,RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 1272 bitsRB5: No data RB6: No data RB7: No data RB7: No data RB7: 80 bits RB7: 8					UL TFC36		
18DL_TFC18, DL_TFC48UL_TFC18, UL_TFC42DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC18, UL_TFC42RB5: 81 bits RB6: 103 bits RB7: 60 bits RB8: 1272 bitsRB5: No data RB6: No data RB7: No data RB8: 952 bits19DL_TFC19, DL_TFC49UL_TFC19, UL_TFC43DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC24RB5: 39 bits RB6: 103 bits RB8: 1272 bitsRB5: 39 bits RB8: 952 bits19DL_TFC49UL_TFC43DL_TFC0, UL_TFC20, UL_TFC43UL_TFC0, UL_TFC10, UL_TFC24UL_TFC0, UL_TFC18, UL_TFC24, UL_TFC24,RB6: 103 bits RB6: 103 bits RB6: 103 bits RB6: No data RB7: No data RB7: No data RB7: No data20DL_TFC20, UL_TFC20, UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24,UL_TFC0, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, RB6: 53 bits RB7: 60 bits RB6: 53 bits RB7: No data RB7: No data RB7: No data RB7: No data RB8: 952 bits							
DL_TFC48UL_TFC42UL_TFC0, UL_TFC24UL_TFC18, UL_TFC42RB6: 103 bits RB7: 60 bits RB7: 100 bits RB8: 1272 bitsRB6: No data RB7: No data RB8: 952 bits19DL_TFC19, DL_TFC49UL_TFC19, UL_TFC43DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC18, UL_TFC19, UL_TFC19, UL_TFC19, UL_TFC19, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC25, UL_TFC43RB6: 103 bits RB5: 39 bits RB6: No data RB7: No data RB7: No data RB7: No data RB7: No data RB7: No data RB7: No data20DL_TFC20, DL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24, UL_TFC0, UL_TFC24, UL_TFC24, UL_TFC18, UL_TFC20, UL_TFC18, UL_TFC20, UL_TFC24,RB5: 42 bits RB5: 42 bits RB6: 53 bits RB7: No data RB7: No data RB7: No data RB8: 952 bits	18	DL_TFC18.	UL_TFC18.	DL_TFC0, DL_TFC30.		RB5: 81 bits	RB5: No data
Image: constraint of the constra	-						
Image: constraint of the constra							RB7: No data
19DL_TFC19, DL_TFC49UL_TFC19, UL_TFC43DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC24RB5: 39 bits RB6: 103 bits RB7: 60 bits UL_TFC19, UL_TFC19, UL_TFC24, UL_TFC24, UL_TFC25, UL_TFC43RB5: 39 bits RB6: 103 bits RB7: No data RB8: 1272 bits20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC24, UL_TFC24, UL_TFC26, UL_TFC24,RB5: 42 bits RB5: 42 bits RB5: 42 bits RB6: 53 bits RB6: 53 bits RB7: No data					UL_TFC42		
DL_TFC49UL_TFC43UL_TFC0, UL_TFC24UL_TFC1, UL_TFC19, UL_TFC19, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC25, UL_TFC43RB6: 103 bits RB7: 60 bits RB8: 1272 bitsRB6: No data RB7: No data RB8: 952 bits20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC28, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC20, UL_TFC20, UL_TFC24,RB5: 42 bits RB5: 42 bits RB6: 53 bits RB6: 53 bits RB7: No data RB6: 53 bits RB7: No data RB7: No data RB8: 952 bits	19				UL_TFC0,	RB5: 39 bits	
UL_TFC19, UL_TFC24, UL_TFC25, UL_TFC44RB8: 1272 bitsRB8: 952 bits20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC24, UL_TFC18, UL_TFC20, UL_TFC24,RB5: 42 bits RB6: 53 bits RB7: 60 bits RB7: 60 bits RB8: 1272 bitsRB5: 42 bits RB6: 53 bits RB7: No data RB8: 952 bits			UL_TFC43	UL_TFC0, UL_TFC24		RB6: 103 bits	RB6: No data
20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC43RB5: 42 bits RB6: 53 bits RB6: 53 bits RB7: 60 bits RB7: No data RB8: 1272 bitsRB5: 42 bits RB7: No data RB8: 952 bits						RB7: 60 bits	RB7: No data
20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC24RB5: 42 bits RB6: 53 bits RB6: 53 bits RB7: 60 bits RB7: No data RB8: 1272 bitsRB5: 42 bits RB6: 53 bits RB7: No data RB8: 952 bits						RB8: 1272 bits	RB8: 952 bits
20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, UL_TFC0, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC2, UL_TFC18, UL_TFC20, UL_TFC24,RB5: 42 bits RB6: 53 bits RB6: 53 bits RB7: 00 bits RB7: No data RB8: 1272 bits20DL_TFC50UL_TFC44DL_TFC0, UL_TFC0, UL_TFC0, UL_TFC2, UL_TFC24UL_TFC0, RB6: 53 bits RB7: 60 bits RB8: 1272 bitsRB5: 42 bits RB6: 53 bits RB7: No data RB8: 952 bits							
20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC24RB5: 42 bits RB6: 53 bitsRB5: 42 bits RB6: 53 bits20DL_TFC50UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC18, UL_TFC20, UL_TFC20, UL_TFC24,RB5: 42 bits RB6: 53 bitsRB5: 42 bits RB6: 53 bits							
20DL_TFC20, DL_TFC50UL_TFC20, UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24UL_TFC0, UL_TFC24UL_TFC0, UL_TFC18, UL_TFC20, UL_TFC24,RB5: 42 bits RB6: 53 bitsRB5: 42 bits RB6: 53 bits20DL_TFC50UL_TFC44DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC2, UL_TFC24,UL_TFC0, RB7: 60 bits RB8: 1272 bitsRB5: 42 bits RB6: 53 bits							
DL_TFC50UL_TFC44UL_TFC0, UL_TFC24UL_TFC2, UL_TFC18, UL_TFC20, UL_TFC24,RB6: 53 bits RB7: 60 bits RB8: 1272 bitsRB6: 53 bits RB7: No data RB8: 952 bits							
UL_TFC18, RB7: 60 bits RB7: No data UL_TFC20, RB8: 1272 bits RB8: 952 bits UL_TFC24,	20						
UL_TFC20, RB8: 1272 bits RB8: 952 bits UL_TFC24,		DL_TFC50	UL_TFC44	UL_IFC0, UL_TFC24			
UL_TFC24,							
						RB8: 1272 bits	RB8: 952 bits
					UL_TFC26,		
UL_TFC42,	•	1					
UL_TFC44							

04						
21	DL_TFC21, DL_TFC51	UL_TFC21, UL_TFC45	DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24	UL_TFC0, UL_TFC3,	RB5: 55 bits RB6: 63 bits	RB5: 55 bits RB6: 63 bits
			0L_1F00, 0L_1F024	UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC21,	RB8: 1272 bits	RB8: 952 bits
				UL_TFC24,	1.00. 1212 010	1.20.002.010
				UL_TFC27,		
				UL_TFC42,		
				UL_TFC45		
22	DL_TFC22,	UL_TFC22,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC52	UL_TFC46	UL_TFC0, UL_TFC24	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC22,	RB8: 1272 bits	RB8: 952 bits
				UL_TFC24,		
				UL_TFC28, UL_TFC42,		
				UL_TFC42,		
23	DL_TFC23,	UL_TFC23,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC53	UL_TFC47	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC18,	RB7: 60 bits	RB7: 60 bits
				UL_TFC23,	RB8: 1272 bits	RB8: 952 bits
				UL_TFC24,		
				UL_TFC29,		
				UL_TFC42,		
				UL_TFC47		
24	DL_TFC24,	UL_TFC18,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC54	UL_TFC42	UL_TFC0, UL_TFC24	UL_TFC18,	RB6: 103 bits	RB6: No data
				UL_TFC24, UL_TFC42	RB7: 60 bits RB8: 1272 bits	RB7: No data RB8: 1272 bits
25	DL_TFC25,	UL_TFC19,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
23	DL_TFC55	UL_TFC43	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
			01_11 00, 01_11 024	UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC19,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24,		
				UL_TFC25,		
				UL_TFC42,		
				UL_TFC43		
26	DL_TFC26,	UL_TFC20,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
	DL_TFC56	UL_TFC44	UL_TFC0, UL_TFC24	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC20,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24, UL_TFC26,		
				UL_TFC42,		
				UL_TFC44		
27	DL_TFC27,	UL_TFC21,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
	DL_TFC57	UL_TFC45	UL_TFC0, UL_TFC24	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC21,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24,		
				UL_TFC27,		
				UL_TFC42,		
				UL_TFC45		
28	DL_TFC28,	UL_TFC22,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC58	UL_TFC46	UL_TFC0, UL_TFC24	UL_TFC4, UL_TFC18,	RB6: 84 bits RB7: 60 bits	RB6: 84 bits RB7: No data
				UL_TFC18, UL_TFC22,	RB7: 60 bits RB8: 1272 bits	RB8: 1272 bits
				UL_TFC22,	1100. 1212 0115	1100. 1212 DILS
				UL_TFC28,		
				UL_TFC42,		
				UL_TFC46		
29	DL_TFC29,	UL_TFC23,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC59	UL_TFC47	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC18,	RB7: 60 bits	RB7: 60 bits
				UL_TFC23,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24,		
				UL_TFC29,		
				UL_TFC42, UL_TFC47		

NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).

14.2.38i.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2,3 and 4: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
 - for sub-test 5: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
 - for sub-test 6: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 7: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 8,9 and 10: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 11: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 12: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 13: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 14,15 and 16: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 17: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 18: RLC SDUs on RB8 having the first 952 bits equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 19: RLC SDUs on RB5 having the same content as sent by the SS; RLC SDUs on RB8 having the first 952 bits equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
 - for sub-test 20,21 and 22: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; RLC SDUs on RB8 having the first 952 bits equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB7.
 - for sub-test 23: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and RLC SDUs on RB8 having the first 952 bits equal to the content of the test data sent by the SS in downlink.

- for sub-test 24: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 25: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
- for sub-test 26,27 and 28: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
- for sub-test 29: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.38j Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL:(12.2 7.95 5.9 4.75) kbps / CS RAB + Interactive or background / UL:64 DL:128 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.38j.1 Conformance requirement

See clause 14.2.4.1.

14.2.38j.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.38j.

14.2.38j.3 Method of test

See clause 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
TFS	TF2, bits	1x42	1x63	N/A	2x336	N/A
115	TF3, bits	1x55	1x84	N/A	3x336	N/A
	TF4, bits	1x75	1x103	N/A	4x336	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

-	
TFCI	(RB5, RB6, RB7,RB8,DCCH)
UL_TFC0	(TF0,TF0,TF0,TF0,TF0)
UL_TFC1	(TF1,TF0,TF0,TF0,TF0)
UL_TFC2	(TF2,TF1,TF0,TF0,TF0)
UL_TFC3	(TF3,TF2,TF0,TF0,TF0)
UL_TFC4	(TF4,TF3,TF0,TF0,TF0)
UL_TFC5	(TF5,TF4,TF1,TF0,TF0)
UL_TFC6	(TF0,TF0,TF0,TF1,TF0)
UL_TFC7	(TF1,TF0,TF0,TF1,TF0)
UL_TFC8	(TF2,TF1,TF0,TF1,TF0)
UL_TFC9	(TF3,TF2,TF0,TF1,TF0)
UL_TFC10	(TF4,TF3,TF0,TF1,TF0)
UL_TFC11	(TF5,TF4,TF1,TF1,TF0)
UL_TFC12	(TF0,TF0,TF0,TF2,TF0)
UL_TFC13	(TF1,TF0,TF0,TF2,TF0)
UL_TFC14	(TF2,TF1,TF0,TF2,TF0)
UL_TFC15	(TF3,TF2,TF0,TF2,TF0)
UL_TFC16	(TF4,TF3,TF0,TF2,TF0)
UL_TFC17	(TF5,TF4,TF1,TF2,TF0)
UL_TFC18	(TF0,TF0,TF0,TF4,TF0)
UL_TFC19	(TF1,TF0,TF0,TF4,TF0)
UL_TFC20	(TF2,TF1,TF0,TF4,TF0)
UL_TFC21	(TF3,TF2,TF0,TF4,TF0)
UL_TFC22 UL_TFC23	(TF4,TF3,TF0,TF4,TF0)
	(TF5,TF4,TF1,TF4,TF0)
UL_TFC24	(TF0,TF0,TF0,TF0,TF1) (TE1,TE0,TE0,TE0,TE1)
UL_TFC25 UL_TFC26	(TF1,TF0,TF0,TF0,TF1) (TF2,TF1,TF0,TF0,TF1)
UL_TFC27	(TF3,TF2,TF0,TF0,TF1)
UL_TFC28	(TF4,TF3,TF0,TF0,TF1)
UL_TFC29	(TF5,TF4,TF1,TF0,TF1)
UL_TFC30	(TF0,TF0,TF0,TF1,TF1)
UL_TFC31	(TF1,TF0,TF0,TF1,TF1)
UL_TFC32	(TF2,TF1,TF0,TF1,TF1)
UL_TFC33	(TF3,TF2,TF0,TF1,TF1)
UL_TFC34	(TF4,TF3,TF0,TF1,TF1)
UL_TFC35	(TF5,TF4,TF1,TF1,TF1)
UL TFC36	(TF0.TF0.TF0.TF2.TF1)
UL_TFC37	(TF1,TF0,TF0,TF2,TF1)
UL_TFC38	(TF2,TF1,TF0,TF2,TF1)
UL_TFC39	(TF3,TF2,TF0,TF2,TF1)
UL_TFC40	(TF4,TF3,TF0,TF2,TF1)
UL_TFC41	(TF5,TF4,TF1,TF2,TF1)
UL_TFC42	(TF0,TF0,TF0,TF4,TF1)
UL_TFC43	(TF1,TF0,TF0,TF4,TF1)
UL_TFC44	(TF2,TF1,TF0,TF4,TF1)
UL_TFC45	(TF3,TF2,TF0,TF4,TF1)
UL_TFC46	(TF4,TF3,TF0,TF4,TF1)
UL TFC47	(TF5.TF4.TF1.TF4.TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (128 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x53	1x60	1x336	1x148
TFS	TF2, bits	1x42	1x63	N/A	2x336	N/A
115	TF3, bits	1x55	1x84	N/A	4x336	N/A
	TF4, bits	1x75	1x103	N/A	8x336	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0,TF0,TF0,TF0,TF0)
DL_TFC1	(TF1,TF0,TF0,TF0,TF0)
DL_TFC2	(TF2,TF1,TF0,TF0,TF0)
DL_TFC3	(TF3,TF2,TF0,TF0,TF0)
DL_TFC4	(TF4,TF3,TF0,TF0,TF0)
DL_TFC5	(TF5,TF4,TF1,TF0,TF0)
DL_TFC6	(TF0,TF0,TF0,TF1,TF0)
DL_TFC7	(TF1,TF0,TF0,TF1,TF0)
DL_TFC8	(TF2,TF1,TF0,TF1,TF0)
DL_TFC9	(TF3,TF2,TF0,TF1,TF0)
DL_TFC10	(TF4,TF3,TF0,TF1,TF0)
DL_TFC11	(TF5,TF4,TF1,TF1,TF0)
DL_TFC12 DL_TFC13	(TF0,TF0,TF0,TF2,TF0)
	(TF1,TF0,TF0,TF2,TF0)
DL_TFC14	(TF2,TF1,TF0,TF2,TF0)
DL_TFC15	(TF3,TF2,TF0,TF2,TF0)
DL_TFC16	(TF4,TF3,TF0,TF2,TF0)
DL_TFC17	(TF5,TF4,TF1,TF2,TF0)
DL_TFC18 DL_TFC19	(TF0,TF0,TF0,TF3,TF0) (TE1,TE0,TE0,TE3,TE0)
DL_TFC19 DL_TFC20	(TF1,TF0,TF0,TF3,TF0) (TF2,TF1,TF0,TF3,TF0)
	(TF2,TF1,TF0,TF3,TF0)
DL_TFC21	(TF3,TF2,TF0,TF3,TF0)
DL_TFC22 DL_TFC23	(TF4,TF3,TF0,TF3,TF0) (TF5,TF4,TF1,TF3,TF0)
DL_TFC23	(TF0,TF0,TF0,TF4,TF0)
DL_TFC24	(TF1,TF0,TF0,TF4,TF0)
DL_TFC25	(TF2,TF1,TF0,TF4,TF0)
DL_TFC20	(TF3,TF2,TF0,TF4,TF0)
DL_TFC28	(TF4,TF3,TF0,TF4,TF0)
-	(TF5,TF4,TF1,TF4,TF0)
DL_TFC29 DL_TFC30	(TF0,TF0,TF0,TF0,TF1)
DL_TFC31	(TF1,TF0,TF0,TF0,TF1)
DL_TFC32	(TF2,TF1,TF0,TF0,TF1)
DL_TFC33	(TF3,TF2,TF0,TF0,TF1)
DL_TFC34	(TF4,TF3,TF0,TF0,TF1)
DL_TFC35	(TF5,TF4,TF1,TF0,TF1)
DL_TFC36	(TF0,TF0,TF0,TF1,TF1)
DL TFC37	(TF1,TF0,TF0,TF1,TF1)
DL_TFC38	(TF2,TF1,TF0,TF1,TF1)
DL_TFC39	(TF3,TF2,TF0,TF1,TF1)
DL_TFC40	(TF4,TF3,TF0,TF1,TF1)
DL_TFC41	(TF5,TF4,TF1,TF1,TF1)
DL_TFC42	(TF0,TF0,TF0,TF2,TF1)
DL_TFC43	(TF1,TF0,TF0,TF2,TF1)
DL_TFC44	(TF2,TF1,TF0,TF2,TF1)
DL_TFC45	(TF3,TF2,TF0,TF2,TF1)
DL_TFC46	(TF4,TF3,TF0,TF2,TF1)
DL TFC47	(TF5,TF4,TF1,TF2,TF1)
DL_TFC48	(TF0,TF0,TF0,TF3,TF1)
DL_TFC49	(TF1,TF0,TF0,TF3,TF1)
DL_TFC50	(TF2,TF1,TF0,TF3,TF1)
DL_TFC51	(TF3,TF2,TF0,TF3,TF1)
DL_TFC52	(TF4,TF3,TF0,TF3,TF1)
DL_TFC53	(TF5,TF4,TF1,TF3,TF1)
DL_TFC54	(TF0,TF0,TF0,TF4,TF1)
DL_TFC55	(TF1,TF0,TF0,TF4,TF1)
DL_TFC56	(TF2,TF1,TF0,TF4,TF1)
DL_TFC57	(TF3,TF2,TF0,TF4,TF1)
DL_TFC58	(TF4,TF3,TF0,TF4,TF1)
DL_TFC59	(TF5,TF4,TF1,TF4,TF1)

Sub-tests:

Sub- test	Downlink TFCS under test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size
					(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
	DL_TFC31	UL_TFC25	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC25	RB8: 312 bits	RB8: No data
2	DL_TFC2,	UL_TFC2,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
	DL_TFC32	UL_TFC26	UL_TFC0, UL_TFC24	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
				UL_TFC24, UL_TFC26	RB7: 60 bits RB8: 312 bits	RB7: No data RB8: No data
3	DL_TFC3,	UL_TFC3,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
3	DL_TFC33	UL_TFC27	UL_TFC0, UL_TFC24	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
	DL_11000	02_11 027	01_11 00, 01_11 024	UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC27	RB8: 312 bits	RB8: No data
4	DL_TFC4,	UL_TFC4,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC34	UL_TFC28	UL_TFC0, UL_TFC24	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
				UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC28	RB8: 312 bits	RB8: No data
5	DL_TFC5,	UL_TFC5,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC35	UL_TFC29	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC24,	RB7: 60 bits	RB7: 60 bits
_	DI 770			UL_TFC29	RB8: 312 bits	RB8: No data
6	DL_TFC6,	UL_TFC6,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC36	UL_TFC30	UL_TFC0, UL_TFC24	UL_TFC6,	RB6: 103 bits	RB6: No data RB7: No data
				UL_TFC24, UL_TFC30	RB7: 60 bits RB8: 312 bits	RB8: 312 bits
7	DL_TFC7,	UL_TFC7,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
'	DL_TFC37	UL_TFC31	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
	DL_11007	02_11001		UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC7,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24,		
				UL_TFC25,		
				UL_TFC30,		
				UL_TFC31		
8	DL_TFC8,	UL_TFC8,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
	DL_TFC38	UL_TFC32	UL_TFC0, UL_TFC24	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
				UL_TFC6, UL_TFC8,	RB7: 60 bits RB8: 312 bits	RB7: No data RB8: 312 bits
				UL_TFC24,	KB0. 312 DIIS	ND0. 312 DIIS
				UL_TFC26,		
				UL_TFC30,		
				UL_TFC32		
9	DL_TFC9,	UL_TFC9,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
	DL_TFC39	UL_TFC33	UL_TFC0, UL_TFC24	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
				UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC9,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24,		
				UL_TFC27,		
				UL_TFC30,		
10	DL_TFC10,	UL_TFC10,	DL_TFC0, DL_TFC30,	UL_TFC33 UL_TFC0,	RB5: 75 bits	RB5: 75 bits
10	DL_TFC10, DL_TFC40	UL_TFC10,	UL_TFC0, UL_TFC30,	UL_TFC0, UL_TFC4,	RB6: 84 bits	RB6: 84 bits
	DL_11040	01_11 034	0L_1100, 0L_1F024	UL_TFC6,	RB7: 60 bits	RB7: No data
				UL_TFC10,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24,		
				UL_TFC28,		
				UL_TFC30,		
				UL_TFC34		
11	DL_TFC11,	UL_TFC11,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC41	UL_TFC35	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC6,	RB7: 60 bits	RB7: 60 bits
				UL_TFC11,	RB8: 312 bits	RB8: 312 bits
				UL_TFC24, UL_TFC29,		
				UL_TFC30,		
				UL_TFC35		
L		1	•		1	

Release 5

40						
12	DL_TFC12, DL_TFC42	UL_TFC12, UL_TFC36	DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24	UL_TFC0, UL_TFC12,	RB5: 81 bits RB6: 103 bits	RB5: No data RB6: No data
	DL_11042	02_11030	0L_1100, 0L_11024	UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC36	RB8: 632 bits	RB8: 632 bits
13	DL_TFC13,	UL_TFC13,	DL_TFC0, DL_TFC30,	UL TFC0,	RB5: 39 bits	RB5: 39 bits
_	DL_TFC43	UL_TFC37	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC12,	RB7: 60 bits	RB7: No data
				UL_TFC13,	RB8: 632 bits	RB8: 632 bits
				UL_TFC24,		
				UL_TFC25,		
				UL_TFC36,		
4.4				UL_TFC37		RB5: 42 bits
14	DL_TFC14, DL_TFC44	UL_TFC14, UL_TFC38	DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24	UL_TFC0, UL_TFC2,	RB5: 42 bits RB6: 53 bits	RB6: 53 bits
		02_11000	02_1100, 02_11024	UL_TFC12,	RB7: 60 bits	RB7: No data
				UL_TFC14,	RB8: 632 bits	RB8: 632 bits
				UL_TFC24,		
				UL_TFC26,		
				UL_TFC36,		
		·		UL_TFC38		
15	DL_TFC15,	UL_TFC15,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
	DL_TFC45	UL_TFC39	UL_TFC0, UL_TFC24	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
				UL_TFC12, UL_TFC15,	RB7: 60 bits RB8: 632 bits	RB7: No data RB8: 632 bits
				UL_TFC15,	ND0. 032 DIIS	NDO. 032 DIIS
				UL_TFC27,		
				UL_TFC36,		
				UL_TFC39		
16	DL_TFC16,	UL_TFC16,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC46	UL_TFC40	UL_TFC0, UL_TFC24	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
				UL_TFC12,	RB7: 60 bits	RB7: No data
				UL_TFC16,	RB8: 632 bits	RB8: 632 bits
				UL_TFC24,		
				UL_TFC28, UL_TFC36,		
				UL_TFC40		
17	DL_TFC17,	UL_TFC17,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC47	UL_TFC41	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC12,	RB7: 60 bits	RB7: 60 bits
				UL_TFC17,	RB8: 632 bits	RB8: 632 bits
				UL_TFC24,		
				UL_TFC29,		
				UL_TFC36,		
18	DL_TFC18,	UL_TFC18,	DL_TFC0, DL_TFC30,	UL_TFC41 UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC18, DL_TFC48	UL_TFC42	UL_TFC0, UL_TFC30,	UL_TFC18,	RB6: 103 bits	RB6: No data
				UL_TFC24,	RB7: 60 bits	RB7: No data
				UL_TFC42	RB8: 1272 bits	RB8: 1272 bits
19	DL_TFC19,	UL_TFC19,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
	DL_TFC49	UL_TFC43	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC19,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24, UL_TFC25,		
				UL_TFC25, UL_TFC42,		
				UL_TFC43		
20	DL_TFC20,	UL_TFC20,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
-	DL_TFC50	UL_TFC44	UL_TFC0, UL_TFC24	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC20,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24,		
				UL_TFC26,		
				UL_TFC42,		
L	1	L		UL_TFC44		

04						
21	DL_TFC21, DL_TFC51	UL_TFC21, UL_TFC45	DL_TFC0, DL_TFC30, UL_TFC0, UL_TFC24	UL_TFC0, UL_TFC3,	RB5: 55 bits RB6: 63 bits	RB5: 55 bits RB6: 63 bits
			0L_1F00, 0L_1F024	UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC21,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24,	1.00. 1212 010	
				UL_TFC27,		
				UL_TFC42,		
				UL_TFC45		
22	DL_TFC22,	UL_TFC22,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC52	UL_TFC46	UL_TFC0, UL_TFC24	UL_TFC4,	RB6: 84 bits	RB6: 84 bits
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC22,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24,		
				UL_TFC28, UL_TFC42,		
				UL_TFC42,		
23	DL_TFC23,	UL_TFC23,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC53	UL_TFC47	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC18,	RB7: 60 bits	RB7: 60 bits
				UL_TFC23,	RB8: 1272 bits	RB8: 1272 bits
				UL_TFC24,		
				UL_TFC29,		
				UL_TFC42,		
				UL_TFC47		
24	DL_TFC24,	UL_TFC18,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: No data
	DL_TFC54	UL_TFC42	UL_TFC0, UL_TFC24	UL_TFC18,	RB6: 103 bits	RB6: No data
				UL_TFC24, UL_TFC42	RB7: 60 bits RB8: 1272 bits	RB7: No data RB8: 2552 bits
25	DL_TFC25,	UL_TFC19,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 39 bits	RB5: 39 bits
23	DL_TFC55	UL_TFC43	UL_TFC0, UL_TFC24	UL_TFC1,	RB6: 103 bits	RB6: No data
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC19,	RB8: 1272 bits	RB8: 2552 bits
				UL_TFC24,		
				UL_TFC25,		
				UL_TFC42,		
		· · · · · · · · · · · · · · · · · · ·		UL_TFC43		
26	DL_TFC26,	UL_TFC20,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 42 bits	RB5: 42 bits
	DL_TFC56	UL_TFC44	UL_TFC0, UL_TFC24	UL_TFC2,	RB6: 53 bits	RB6: 53 bits
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC20,	RB8: 1272 bits	RB8: 2552 bits
				UL_TFC24, UL_TFC26,		
				UL_TFC42,		
				UL_TFC44		
27	DL_TFC27,	UL_TFC21,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 55 bits	RB5: 55 bits
	DL_TFC57	UL_TFC45	UL_TFC0, UL_TFC24	UL_TFC3,	RB6: 63 bits	RB6: 63 bits
				UL_TFC18,	RB7: 60 bits	RB7: No data
				UL_TFC21,	RB8: 1272 bits	RB8: 2552 bits
				UL_TFC24,		
				UL_TFC27,		
				UL_TFC42,		
				UL_TFC45		
28	DL_TFC28,	UL_TFC22,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 75 bits	RB5: 75 bits
	DL_TFC58	UL_TFC46	UL_TFC0, UL_TFC24	UL_TFC4, UL_TFC18,	RB6: 84 bits RB7: 60 bits	RB6: 84 bits RB7: No data
				UL_TFC22,	RB7: 60 bits RB8: 1272 bits	RB8: 2552 bits
				UL_TFC24,	120. 1212 0115	1100. 2002 DIIO
				UL_TFC28,		
				UL_TFC42,		
				UL_TFC46		
29	DL_TFC29,	UL_TFC23,	DL_TFC0, DL_TFC30,	UL_TFC0,	RB5: 81 bits	RB5: 81 bits
	DL_TFC59	UL_TFC47	UL_TFC0, UL_TFC24	UL_TFC5,	RB6: 103 bits	RB6: 103 bits
				UL_TFC18,	RB7: 60 bits	RB7: 60 bits
				UL_TFC23,	RB8: 1272 bits	RB8: 2552 bits
				UL_TFC24,		
				UL_TFC29,		
				UL_TFC42,		
			l	UL_TFC47		

NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size parameter has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).

14.2.38j.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2,3 and 4: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; and no data shall be received on RB7 and RB8.
 - for sub-test 5: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
 - for sub-test 6: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 7: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 8,9 and 10: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 11: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 12: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 13: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 14,15 and 16: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 17: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 18: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 19: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 20,21 and 22: RLC SDUs on RB5, RB6 and RB8 having the same content as sent by the SS; and no data shall be received on RB7.
 - for sub-test 23: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 24: RLC SDUs on RB8 having the first 1272 bits equal to the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 25: RLC SDUs on RB5 having the same content as sent by the SS; RLC SDUs on RB8 having the first 1272 bits equal to the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.

- for sub-test 26,27 and 28: RLC SDUs on RB5 and RB6 having the same content as sent by the SS; RLC SDUs on RB8 having the first 1272 bits equal to the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB7.
- for sub-test 29: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and RLC SDUs on RB8 having the first 1272 bits equal to the content equal to the first 1272 bits of the test data sent by the SS in downlink.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.39 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:64 kbps / PS RAB+ UL:3.4 DL: 3.4 kbps SRBs for DCCH

- 14.2.39.1 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:64 kbps / PS RAB / (TC, 10 ms TTI)
- 14.2.39.1.1 Conformance requirement

See 14.2.4.1.

14.2.39.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.39 for the uplink turbo channel coding and 10 ms TTI case.

14.2.39.1.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (32 kbps, 10 ms TTI)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF1, TF1)
UL_TFC8	(TF2, TF1, TF1, TF0, TF1)
UL_TFC9	(TF0, TF0, TF0, TF1, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1, TF1)
UL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A
	TF4, bits	N/A	N/A	N/A	4x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, DL_TFC16	UL_TFC1 UL_TFC7	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: No data
2	DL_TFC2, DL_TFC17	UL_TFC2, UL_TFC8	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: No data
3	DL_TFC3, DL_TFC18	UL_TFC3, UL_TFC9	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 312
4	DL_TFC4, DL_TFC19	UL_TFC4, UL_TFC10	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 312
5	DL_TFC5, DL_TFC20	UL_TFC5 UL_TFC11	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 312
6	DL_TFC6, DL_TFC21	UL_TFC3, UL_TFC9	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 632
7	DL_TFC7, DL_TFC22	UL_TFC4, UL_TFC10	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 632
8	DL_TFC8, DL_TFC23	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 632
9	DL_TFC9, DL_TFC24	UL_TFC3, UL_TFC9	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 952
10	DL_TFC10, DL_TFC25	UL_TFC4, UL_TFC10	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 952

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)			
11	DL_TFC11, DL_TFC26	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 952			
12	DL_TFC12, DL_TFC27	UL_TFC3, UL_TFC9	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 1272			
13	DL_TFC13, DL_TFC28	UL_TFC4, UL_TFC10	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 1272			
14	DL_TFC14, DL_TFC29	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 1272			
NOTE:	UL_TFC11								

of 7 bit length indicator and expansion bit).

14.2.39.1.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.

- for sub-test 6: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 7: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 8: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- for sub-test 9: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 10: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 11: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 12: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 13: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.39.2 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:64 kbps / PS RAB / (TC, 20 ms TTI)

- 14.2.39.2.1 Conformance requirement
- See 14.2.4.1.

14.2.39.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.39 for the uplink turbo channel coding and 20 ms TTI case.

14.2.39.2.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (32 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	2x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1)
UL_TFC11	(TF2, TF1, TF1, TF0, TF1)
UL_TFC12	(TF0, TF0, TF0, TF1, TF1)
UL_TFC13	(TF1, TF0, TF0, TF1, TF1)
UL_TFC14	(TF2, TF1, TF1, TF1, TF1)
UL_TFC15	(TF0, TF0, TF0, TF2, TF1)
UL_TFC16	(TF1, TF0, TF0, TF2, TF1)
UL_TFC17	(TF2, TF1, TF1, TF2, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A
	TF4, bits	N/A	N/A	N/A	4x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, DL_TFC16	UL_TFC1, UL_TFC10	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: No data
2	DL_TFC2, DL_TFC17	UL_TFC2, UL_TFC11	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: No data
3	DL_TFC3, DL_TFC18	UL_TFC3, UL_TFC12	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC3, UL_TFC9, UL_TFC12	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: No data RB6: No data RB7: No data RB8: 312
4	DL_TFC4, DL_TFC19	UL_TFC4, UL_TFC13	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC9, UL_TFC10, UL_TFC12, UL_TFC13	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: 312
5	DL_TFC5, DL_TFC20	UL_TFC5, UL_TFC14	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC9, UL_TFC11, UL_TFC12, UL_TFC14	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: 312
6	DL_TFC6, DL_TFC21	UL_TFC6, UL_TFC15	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC6, UL_TFC9, UL_TFC15	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 632
7	DL_TFC7, DL_TFC22	UL_TFC7, UL_TFC16	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 632
8	DL_TFC8, DL_TFC23	UL_TFC8, UL_TFC17	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 632
9	DL_TFC9, DL_TFC24	UL_TFC6, UL_TFC15	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC6, UL_TFC9, UL_TFC15	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 952
10	DL_TFC10, DL_TFC25	UL_TFC7, UL_TFC16	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 952

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)			
11	DL_TFC11, DL_TFC26	UL_TFC8, UL_TFC17	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 952			
12	DL_TFC12, DL_TFC27	UL_TFC6, UL_TFC15	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC6, UL_TFC9, UL_TFC15	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 1272			
13	DL_TFC13, DL_TFC28	UL_TFC7, UL_TFC16	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 1272			
14	DL_TFC14, DL_TFC29	UL_TFC8, UL_TFC17	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC9	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 1272			
NOTE:									

14.2.39.2.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
 - for sub-test 6: an RLC SDU on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.

- for sub-test 7: an RLC SDU on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 8: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- for sub-test 9: an RLC SDU on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 10: an RLC SDU on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 11: an RLC SDU on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 12: an RLC SDU on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 13: an RLC SDU on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: an RLC SDU on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.39.3 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:64 kbps / PS RAB / (CC, 10 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.39 for the uplink convolutional channel coding and 10 ms TTI case.

See test case 14.2.39.1 for test procedure and test requirement.

14.2.39.4 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:32 DL:64 kbps / PS RAB / (CC, 20 ms TTI)

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.39 for the uplink convolutional channel coding and 20 ms TTI case.

See test case 14.2.39.2 for test procedure and test requirement.

14.2.40 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:64 kbps / PS RAB+ UL:3.4 DL: 3.4 kbps SRBs for DCCH

14.2.40.1 Conformance requirement

See 14.2.4.1.

14.2.40.2Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.40.

14.2.40.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A
	TF4, bits	N/A	N/A	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A
	TF4, bits	N/A	N/A	N/A	4x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub-tests:

- UL_TFC0, UL_TFC15 UL_TFC16, UL_TFC16 RB7: 60 a RB5: 81 a RB5: 80 a RB5	Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
DL_TFC16 UL_TFC16 DL_TFC15 UL_TFC15 RB6: No 36 RB7: No 48 2 DL_TFC2, DL_TFC17 UL_TFC17 UL_TFC15 UL_TFC16 RB8: S12 RB6: No 36 3 DL_TFC3, DL_TFC18 UL_TFC16, UL_TFC16 UL_TFC17 RB6: No 36 RB7: 60 RB7: 60 4 DL_TFC17 UL_TFC18 DL_TFC17 RB6: No 36 RB7: No 48 5 DL_TFC18 UL_TFC16 UL_TFC16 RB7: No 48 RB7: No 48 4 DL_TFC4, DL_TFC19 UL_TFC17 DL_TFC16 RB7: No 48 RB7: No 48 6 DL_TFC5, DL_TFC19 UL_TFC2, UL_TFC19 UL_TFC16 RB7: No 48 RB7: No 48 7 DL_TFC5, DL_TFC20 UL_TFC2, UL_TFC15 UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC21 RB5: 312 RB6: 103 RB6: 103 6 DL_TFC6, DL_TFC7, DL_TFC21 UL_TFC16, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24 RB6: 103 RB6: 103 7 DL_TFC7, DL_TFC24 UL_TFC24, UL_TFC25, UL_TFC15, UL_TFC15, UL_TFC26, UL_TFC24 <td< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th></td<>							
- - - U_TFC0, U_LTFC15 U_LTFC16 RBS:12 RBS:No da RBS:No da RBS:No da RBS:103 2 DL_TFC17 UL_TFC17 U_LTFC17 U_LTFC15, U_LTFC15, U_LTFC15, U_LTFC15, U_LTFC15, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC17, U_LTFC18, U_LTFC19 RBS:103 RBS:No da RBS:No da RBS:No da RBS:No da RBS:103 RBS:103 RBS:103 RBS:103 RBS:No da RBS:No da RBS:104 4 DL_TFC4, DL_TFC19 U_LTFC15, U_LTFC15, U_LTFC19 U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC17, U_LTFC19 RBS:103 RBS:103 RBS:102 RBS:112 RBS:112 5 DL_TFC20 U_LTFC20, U_LTFC20 U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC16, U_LTFC20 RBS:103 RBS:103 RBS:103 RBS:103 RBS:103 RBS:104 RBS:103 RBS:103 RBS:104 RBS:104 6 D_LTFC21 U_LTFC20, U_LTFC21 U_LTFC20, U_LTFC15, U_LTFC15, U_LTFC16, U_LTFC26, RBS:39 RB5:103 RBS:103 RBS:104 RBS:104 RBS:103 RBS:104 RBS:104 RBS:103 RBS:104 RBS:104 RBS:104 RBS:104 RBS:104 RBS:105 7 D_LTFC21 U_LTFC21, U_LTFC24, D_LTFC24 D_LTFC15, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC27, U_LTFC27, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC26, U_LTFC27, U_LTFC2	1						
- - U_TFC15 U_TFC16 RBS: 312 RBS: 61 RBS: 61 2 DL_TFC2, DL_TFC17 UL_TFC17 DL_TFC3, UL_TFC15, UL_TFC15 UL_TFC2, UL_TFC16, UL_TFC17 RBS: 103 RBS: 103 RBS: 103 RBS: 103 3 D_TFC18, D_TFC18 UL_TFC17 DL_TFC3, UL_TFC16, UL_TFC15 UL_TFC3, UL_TFC16, UL_TFC16 RBS: 103 RBS: 103 RBS: 104 4 D_TFC19 UL_TFC4, D_TFC19 UL_TFC4, UL_TFC16 UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17 RBS: 103 RBS: 103 5 D_TFC20 UL_TFC20, UL_TFC20 D_TFC10, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC17, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC16, UL_TFC21 RBS: 103 RBS: 103 RBS: 104 RBS: 103 RBS: 105 RBS: 104 6 D_TFC20 UL_TFC26, UL_TFC21, UL_TFC21, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RBS: 103 RBS: 103 RBS: 104 RBS: 103 RBS: 104 RBS: 104 R		DL_TFC16	UL_TFC16				RB6: No data
2 DL_TFC2, DL_TFC17 UL_TFC2, UL_TFC17 UL_TFC0, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17 RBS: 81 RBS: 103 RBS: 103 RBS: 103 RBS: 103 RBS: 103 RBS: 103 RBS: 104 RBS: 104 3 DL_TFC3, UL_TFC18 UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC15, UL_TFC16, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC19 UL_TFC3, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC19 RBS: 103 RBS: 103 RBS: 103 RBS: 104 RBS: 104 RBS: 312 RBS: 312 RBS: 312 5 DL_TFC5, UL_TFC20 UL_TFC15, UL_TFC20, UL_TFC20 DL_TFC16, UL_TFC16, UL_TFC16, UL_TFC20, UL_TFC20 RBS: 81 UL_TFC18, UL_TFC18, UL_TFC20, UL_TFC20 RBS: 81 RBS: 103 RBS: 103 RBS: 103 RBS: 103 RBS: 104 RBS: 103 RBS: 104 RBS: 103 RBS: 104 RBS: 104 RBS: 312 RBS: 81 RBS: 81 RBS: 81 RBS: 103 RBS: 104 RBS: 103 RBS: 104 RBS: 104 RBS							
DL_TFC17 UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC15 UL_TFC2, UL_TFC15, UL_TFC17 RB6: 103 RB5: 39 RB5: No da RB7: No da R	0				UL_IFC16		
UL_TFC0, UL_TFC15 UL_TFC17 RBF: 301 RBS: No da RBS: N	2						
- - UL_TFC15 UL_TFC17 RBS: 122 RBS: No da RBS:		DL_IFCI/					
3 DL_TFC3, DL_TFC18 UL_TFC3, UL_TFC18 DL_TFC4, DL_TFC18 UL_TFC3, UL_TFC5, UL_TFC5, DL_TFC4, DL_TFC4 UL_TFC4, DL_TFC4 UL_TFC4, DL_TFC19 UL_TFC5, UL_TFC15, UL_TFC15, UL_TFC6, DL_TFC6, DL_TFC6, UL_TFC6, UL_TFC16, UL_TFC16, UL_TFC6, RB5: 39 RB5: 30 RB5: 30 RB6: 103 RB6: 103 RB6: 103 RB6: 103 5 DL_TFC5, DL_TFC20 UL_TFC5, UL_TFC15, UL_TFC6, UL_TFC15, UL_TFC6, UL_TFC16, RB6: 103 UL_TFC16, RB6: 103 RB6: 103 RB6: 103 RB6: 103 6 DL_TFC6, DL_TFC20 UL_TFC20, DL_TFC15, UL_TFC15, UL_TFC6, UL_TFC16, RB6: 312 RB6: 103 RB7: 60 7 DL_TFC6, DL_TFC7, UL_TFC21 UL_TFC2, UL_TFC15, UL_TFC16, UL_TFC16, RB6: 103 RB6: No da RB6: No da 7 DL_TFC7, DL_TFC7, UL_TFC2, UL_TFC2, UL_TFC15, UL_TFC1, UL_TFC1, RB6: 103 RB6: No da RB6: No da 8 DL_TFC7, DL_TFC7, UL_TFC2, UL_TFC3, UL_TFC15, UL_TFC1, UL_TFC1, RB6: 103 RB6: No da RB6: No da 7 DL_TFC7, DL_TFC2, UL_TFC2, UL_TFC2, UL_TFC3, UL_TFC1, RB6: 103 RB6: No da RB6: No da 8 DL_TFC2, DL_TFC2, UL_TFC3, UL_TFC15, <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>							
DL_TFC18 UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC19, UL_TFC10, UL_TFC21, UL_TFC21, UL_TFC20, UL_TFC10, UL_TFC2	3	DI TEC3	LIL TEC3				
Image: Construct of the second seco	Ũ		_ /				RB6: No data
- UL_TFC15 UL_TFC18 RBS: 312 RBS: 312 4 DL_TFC19 UL_TFC4, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC3, UL_TFC3, UL_TFC4,		52_11 010	02_11 010				RB7: No data
4 DL_TFC4, DL_TFC19 UL_TFC4, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC4, UL_							
DL_TFC19 UL_TFC19 DL_TFC15, DUL_TFC0, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC17, UL_TFC19 UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC19 RB5: 103 RB5: 103 RB5: 312 RB5: 10 and RB5: 312 5 DL_TFC5, DL_TFC20 UL_TFC3, UL_TFC16, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC17, DL_TFC21 DL_TFC0, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC21 RB5: 81 RB5: 103 UL_TFC17, UL_TFC17, UL_TFC16, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC21 RB5: 81 RB5: 39 RB5: 30 RB5: 30 RB5	4	DL TFC4.	UL TFC4.				
Image: Second state UL_TFC3 UL_TFC4, UL_TFC4, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC18, UL_TFC18, UL_TFC20 RB8: 312 RB8: 312 5 DL_TFC5, DL_TFC5, UL_TFC0, UL_TFC1, UL_TFC1, UL_TFC16, UL_TFC3, RB7: 60 RB7: 60 RB7: 60 RB7: 80 RB7: 80 RB8: 312 6 DL_TFC6, DL_TFC1, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC3, UL_TFC20 UL_TFC16, UL_TFC3, UL_TFC3, UL_TFC3, RB7: 80 RB6: No da RB7: No							RB6: No data
Image: Second state UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC18, UL_TFC19 UL_TFC15, UL_TFC18, UL_TFC18, UL_TFC19 RB5: 81 RB5: 81 5 DL_TFC20 UL_TFC20, UL_TFC20 DL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC18, UL_TFC18, UL_TFC21 UL_TFC20, UL_TFC16, UL_TFC17, UL_TFC16, UL_TFC21 RB5: 81 RB5: 103 RB6: 103 6 DL_TFC21 UL_TFC6, UL_TFC21 DL_TFC21, UL_TFC21 DL_TFC21, UL_TFC21 DL_TFC21, UL_TFC21 UL_TFC20, UL_TFC21, UL_TFC21 UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC22 RB5: 39 RB5: No da RB6: No da RB7: 60 7 DL_TFC21, DL_TFC22 UL_TFC7, UL_TFC22 UL_TFC7, UL_TFC23 UL_TFC2, UL_TFC23 UL_TFC2, UL_TFC23 UL_TFC2, UL_TFC23 UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC23, UL_TFC23 RB5: 81 RB6: 103 RB7: 60 RB7: 80 RB7: 80 8 DL_TFC8, DL_TFC23 UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC43 UL_TFC3, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC4, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6				DUL_TFC0,	UL_TFC3,		RB7: No data
5 DL_TFC5, DL_TFC20 UL_TFC6, UL_TFC40 UL_TFC16, UL_TFC18, UL_TFC17, UL_TFC3, UL_TFC3, UL_TFC4, UL_TF				UL_TFC15		RB8: 312	RB8: 312
Image: Second state in the second state in							
5 DL_TFC5, DL_TFC20 UL_TFC5, UL_TFC20 DL_TFC5, UL_TFC20 DL_TFC6, UL_TFC20 UL_TFC6, UL_TFC20 UL_TFC6, UL_TFC15, UL_TFC15, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC15, DL_TFC21 UL_TFC6, UL_TFC17, UL_TFC17, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC21 RB5: 81 RB6: 103 RB7: No da RB6: 802 6 DL_TFC7, DL_TFC7, UL_TFC21 UL_TFC6, UL_TFC21, UL_TFC21, UL_TFC22 DL_TFC7, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC14, UL_TFC21, UL_TFC21, UL_TFC23, UL_TFC23 RB5: 81 RB6: 802 RB6: 103 RB7: No da RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: No da RB7: 60 RB7: No da RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: 60 RB7: No da RB7: No da R							
DL_TFC20 UL_TFC20 DL_TFC15, UL_TFC0, UL_TFC0, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC18, UL_TFC20 RBS: 103 RB7: 60 RB3: 312 RBS: 103 RB7: 60 RB3: 312 6 DL_TFC6, DL_TFC21 UL_TFC6, UL_TFC21 UL_TFC0, UL_TFC15, UL_TFC16, UL_TFC21 UL_TFC6, RB5: 39 RB5: No da RB7: No da R	_						DD5 04
Image: Second state UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC15, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC20 RB8: 312 RB8: 312 RB8: 312 6 DL_TFC6, DL_TFC6, DL_TFC0, UL_TFC15, UL_TFC22 RB5: 39 RB5: 39 8 DL_TFC8, DL_TFC23 UL_TFC3, UL_TFC15, UL_TFC6, RB7: 60 RB7: 80 RB8: 632 RB8: 632 8 DL_TFC8, DL_TFC23 UL_TFC3, UL_TFC0, UL_TFC6, RB7: 60 RB7: 80 RB7: 60 RB7: 80 9 DL_TFC3, DL_TFC4 UL_TFC15, UL_TFC3, UL_TFC15, UL_TFC17, UL_TFC17, UL_TFC17, UL_TFC14, UL_TFC23 RB6: 103 RB6: 103 RB6: 103 10 DL_TFC24 UL_TFC24 DL_TFC10, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC15, UL_TFC4, RB8: 952 RB8: 952 RB8: 952 RB8: 952 10 DL_TFC10, UL_TFC4,	5						
Image: Second state Image: Second state <thimage: second="" state<="" th=""> Image: Second state</thimage:>		DL_TFC20	UL_IFC20				
Image: Second							
Image: Second state UL_TFC3, UL_TFC17, UL_TFC20 UL_TFC20 RB5: 39 RB5: No da 6 DL_TFC21 UL_TFC21 DL_TFC15, UL_TFC6, UL_TFC6, UL_TFC6, RB7: 60 RB6: 103 RB6: No da 7 DL_TFC7, DL_TFC22 UL_TFC7, UL_TFC7, UL_TFC22, UL_TFC22, UL_TFC15, UL_TFC15, UL_TFC6, RB7: 60 RB7: No da RB7: No da 7 DL_TFC22 UL_TFC7, UL_TFC7, UL_TFC7, UL_TFC15, UL_TFC6, RB7: 60 RB7: 60 RB7: No da 8 DL_TFC23 UL_TFC8, UL_TFC23 DL_TFC7, UL_TFC15, UL_TFC16, UL_TFC2, RB8: 632 RB6: 103 RB6: 103 8 DL_TFC23 UL_TFC8, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC2, RB8: 60 RB7: 60 RB7: 60 9 DL_TFC24 UL_TFC24 DL_TFC15, UL_TFC15, UL_TFC3, UL_TFC15, UL_TFC23, RB8: 632 RB6: 103 10 DL_TFC9, DL_TFC10, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC24 RB6: 103 RB6: No da 110 DL_TFC10, UL_TFC15, UL_TFC15, UL_TFC24, RB8: 952 RB5: 39 RB5: 39 RB5: 39 110 DL_TFC10, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC24 RB6: No da <td></td> <td></td> <td></td> <td>UL_IFCI5</td> <td></td> <td>RD0. 312</td> <td>RD0. 312</td>				UL_IFCI5		RD0. 312	RD0. 312
Image: Second							
Image: Construct of the system of t							
6 DL_TFC6, DL_TFC21 UL_TFC6, UL_TFC21 DL_TFC6, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC6, UL_TFC15, UL_TFC15, UL_TFC21 RB5: 39 RB5: No da RB6: No da RB7: No da R							
Image: Construct of the system UL_TFC9, UL_TFC0, UL_TFC15, UL_TFC15, UL_TFC21 RB7: 60 RB7: No da RB8: 632 7 DL_TFC7, DL_TFC7, UL_TFC7, DL_TFC0, UL_TFC0, UL_TFC1, RB6: 103 RB6: 39 RB5: 39 RB5: 39 7 DL_TFC22 UL_TFC22 DL_TFC15, UL_TFC1, UL_TFC1, RB6: 103 RB6: No da RB7: No da UL_TFC24 9 DL_TFC9, DL_TFC10, DL_TFC0, UL_TFC0, UL_TFC0, UL_TFC15, UL_TFC15, UL_TFC14, RB8: 632 RB5: No da RB7: No da RB7: No da UL_TFC14, UL_TFC15, RB7: 60 RB7: No da UL_TFC15, UL_TFC15, RB7: 60 RB7: No da UL_TFC15, UL_TFC14, RB8: 952 10 DL_TFC10, DL_TFC10, UL_TFC15, UL_TFC0, UL_TFC14, RB8: 952 DL_TFC25 DL_TFC15, UL_TFC14, RB8: 952 RB6: No da RB7: No da RB7: No da RB7: No da UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC14, RB8: 952 RB8: No da RB7: No da UL_TFC15, UL_TFC15, UL_TFC14, RB8: 952 RB8: 952	6	DL_TFC6,	UL_TFC6,	DL_TFC0,		RB5: 39	RB5: No data
Image: Construct of the system UL_TFC1 UL_TFC21 RB8: 632 RB8: 632 7 DL_TFC7, DL_TFC22 UL_TFC7, UL_TFC22 DL_TFC0, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC2, UL_TFC1, UL_TFC2, UL_TFC1, UL_TFC2, UL_TFC2, UL_TFC1, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC1, UL_TFC2, UL_TFC1, UL_TFC2, UL_TFC1, UL_TFC2, RB8: 952 RB5: 632 RB5: 632 9 DL_TFC9, DL_TFC2, UL_TFC2, UL_TFC1, UL_TFC0, UL_TFC1, UL_TFC0, UL_TFC1, UL_TFC1, UL_TFC1, UL_TFC1, RB6: 103 RB5: No da RB5: No da RB7: No da RB6: No da RB7: No da RB6: No da UL_TFC1, UL_TFC1, RB6: 103 RB6: No da RB7:		DL_TFC21	UL_TFC21	DL_TFC15,	UL_TFC6,	RB6: 103	RB6: No data
7 DL_TFC7, DL_TFC22 UL_TFC7, UL_TFC22 DL_TFC7, UL_TFC22 DL_TFC7, DL_TFC15, UL_TFC15, UL_TFC6, UL_TFC6, UL_TFC7, UL_TFC16, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC22 RB5: 39 RB5: 39 RB6: 103 RB7: No da RB7: No da RB7: No da RB8: 632 8 DL_TFC8, DL_TFC23 UL_TFC8, UL_TFC23 DL_TFC8, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC17, UL_TFC17, UL_TFC24 RB5: 81 RB5: 81 RB5: 81 RB5: 81 RB5: 81 9 DL_TFC9, DL_TFC24 UL_TFC9, UL_TFC24 DL_TFC9, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC14, UL_TFC14, UL_TFC15, UL_TFC15, UL_TFC14, UL_TFC15, UL_TFC14, RB5: 39 RB5: No da RB5: No da RB5: No da 10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC15, UL_TFC25 DL_TFC10, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RB5: 39 RB5: 39						RB7: 60	RB7: No data
DL_TFC22 UL_TFC22 DL_TFC15, UL_TFC0, UL_TFC15 UL_TFC1, UL_TFC6, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC22 RB6: 103 RB7: 60 RB7: 60 RB8: 632 RB6: No da RB7: No da RB7: No da 8 DL_TFC8, DL_TFC23 UL_TFC8, UL_TFC23 DL_TFC0, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC23 UL_TFC8, DL_TFC23 DL_TFC0, UL_TFC15, UL_TFC15, UL_TFC15 RB5: 81 UL_TFC2, UL_TFC2, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC17, UL_TFC23 RB5: 81 RB5: 81 RB5: 81 RB5: 81 9 DL_TFC9, DL_TFC24 UL_TFC9, UL_TFC24 DL_TFC0, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC10, DL_TFC24 RB5: 39 RB5: 39 RB5: 39 RB5: 39 RB5: 39 RB5: 39 RB5: 39 RB5: 39 RB5: No da RB6: 103 RB6: 103 RB6: 103 RB6: No da UL_TFC15, UL_TFC15, UL_TFC24, RB8: 952 10 DL_TFC10, DL_TFC25 DL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RB5: 39 RB5: 39 RB5: 39 RB5: No da RB6: No da RB7: No da RB6: No da RB7: No da							
Image: second	7						
Image: Second		DL_IFC22	UL_IFC22				
Image: Second system UL_TFC16, UL_TFC15, UL_TFC21, UL_TFC21, UL_TFC21, UL_TFC22, UL_TFC23, UL_TFC23 UL_TFC8, UL_TFC8, UL_TFC6, UL_TFC0, UL_TFC2, RB5: 81 RB5: 81 RB5: 81 8 DL_TFC23 UL_TFC23 DL_TFC15, UL_TFC2, UL_TFC2, UL_TFC2, RB6: 103 RB6: 103 RB6: 103 9 DL_TFC9, UL_TFC9, UL_TFC9, UL_TFC4 UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC2, UL_TFC24 DL_TFC16, UL_TFC2, RB8: 632 RB5: No da 9 DL_TFC10, UL_TFC24, UL_TFC24, UL_TFC15, UL_TFC3, UL_TFC3, UL_TFC3, UL_TFC4, UL_TFC4, RB6: 103 RB6: No da 10 DL_TFC10, UL_TFC10, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC24, RB8: 952 RB5: 39 RB5: 39 10 DL_TFC10, UL_TFC10, UL_TFC15, UL_TFC4, RB6: 103 RB6: No da RB6: No da 01_TFC25 UL_TFC10, UL_TFC15, UL_TFC14, RB6: 103 RB6: No da 01_TFC24 UL_TFC15, UL_TFC4, RB6: 103 RB6: No da 01_TFC25 UL_TFC10, UL_TFC15, UL_TFC1, RB6: 103 RB6: No da 01_TFC24 UL_TFC25 UL_TFC15, UL_TFC1, RB6: 103 RB6: No da 01_TFC24 UL_TFC24 RB8: 952 RB8: 952 RB8: 952							
Image: Second system UL_TFC16, UL_TFC21, UL_TFC22 Image: Second system RB5: 81 RB5: 103 RB6: 632 RB5: 82 RB5: 82 RB5: 82 RB5: 82 RB5: 82 RB5: 83 RB5: 83 RB5: 83 RB5: 83 RB5: 83 RB6: No da 9 DL_TFC24 UL_TFC24 DL_TFC15, UL_TFC0, UL_TFC3, RB6: No da RB7: No da 10 DL_TFC10, UL_TFC25 UL_TFC25 DL_TFC0, UL_TFC1, RB6: 103 RB6: No da 11 DL_TFC26, UL_TFC25 DL_TFC25 UL_TFC0, UL_TFC1, RB6: 103 RB5: 39 <td></td> <td></td> <td></td> <td>UL_IFC15</td> <td></td> <td>RB8: 632</td> <td>RB8: 632</td>				UL_IFC15		RB8: 632	RB8: 632
Image: system of the							
Image: system in the							
8 DL_TFC8, DL_TFC23 UL_TFC8, UL_TFC23 DL_TFC8, DL_TFC23 DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC0, UL_TFC15 UL_TFC0, UL_TFC6, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC17, UL_TFC21, UL_TFC21, UL_TFC23 RB5: 81 RB6: 103 RB7: 60 RB8: 632 RB5: 81 RB6: 103 RB7: 60 RB8: 632 9 DL_TFC9, DL_TFC24 UL_TFC9, UL_TFC24 DL_TFC0, DL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RB5: 39 RB5: 39 RB5: 39							
DL_TFC23 UL_TFC23 DL_TFC15, UL_TFC0, UL_TFC15 UL_TFC2, UL_TFC6, UL_TFC6, UL_TFC3, UL_TFC15, UL_TFC15, UL_TFC17, UL_TFC21, UL_TFC23 RB6: 103 RB7: 60 RB8: 632 RB6: 103 RB7: 60 RB8: 632 9 DL_TFC9, DL_TFC24 UL_TFC9, UL_TFC24 DL_TFC9, DL_TFC24 DL_TFC0, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC24 RB5: 39 RB5: No da RB6: 103 RB6: No da UL_TFC15, UL_TFC24 RB5: 39 RB5: 39 RB5: 39 RB5: No da RB6: No da RB7: No da RB7: No da RB7: No da RB7: No da UL_TFC15, UL_TFC10, DL_TFC15, UL_TFC10, UL_TFC10, UL_TFC15, UL_TFC10, UL_TFC15, UL_TFC10, UL_TFC15, UL_TFC10, UL_TFC15, UL_TFC10, UL_TFC10, UL_TFC15, UL_TFC10, UL_TFC15, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC10, UL_TFC15, UL_TFC10	8	DL_TEC8.	UL_TEC8.	DL_TEC0.		RB5: 81	RB5: 81
9 DL_TFC9, DL_TFC24 UL_TFC9, UL_TFC24 DL_TFC9, DL_TFC24 UL_TFC9, DL_TFC24 DL_TFC9, DL_TFC24 DL_TFC9, DL_TFC24 DL_TFC24, DL_TFC15, UL_TFC15, DL_TFC15, DL_TFC0, UL_TFC15, UL_TFC15, UL_TFC9, UL_TFC9, UL_TFC15, RB5: 39 UL_TFC9, UL_TFC16, RB5: 39 UL_TFC15, RB5: No da RB6: No da RB7: No da RB8: 952	•						
9 DL_TFC24 UL_TFC15, UL_TFC24 UL_TFC24, UL_TFC25, UL_TFC15, UL_TFC25 UL_TFC10, UL_TFC24 DL_TFC10, UL_TFC24 DL_TFC10, UL_TFC24 RB5: 39 RB5: No da 10 DL_TFC25, DL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RB5: 39 RB5: 39 RB5: No da 10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, UL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RB6: 103 RB6: No da							
Image: system Image: s							
Image: system of the					UL_TFC15,		
9 DL_TFC9, DL_TFC24 UL_TFC9, UL_TFC24 UL_TFC9, UL_TFC24 DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15 UL_TFC9, UL_TFC15, UL_TFC24 RB5: 39 RB6: 103 RB6: No da RB7: No da RB7: No da 10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC15, UL_TFC15, UL_TFC15 UL_TFC1, UL_TFC10, UL_TFC16, UL_TFC16, UL_TFC24, RB6: 103 RB6: No da							
9 DL_TFC9, DL_TFC24 UL_TFC9, UL_TFC24 UL_TFC9, DL_TFC24 DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15 UL_TFC9, UL_TFC15, UL_TFC15, UL_TFC24 RB5: 39 RB5: No da RB6: No da RB7: No da RB8: 952 10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, DL_TFC25 DL_TFC10, DL_TFC25 DL_TFC10, DL_TFC15, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RB5: 39 RB5: 39 RB5			1				
DL_TFC24 UL_TFC24 DL_TFC15, UL_TFC0, UL_TFC15 UL_TFC9, UL_TFC15, UL_TFC15, UL_TFC24 RB6: 103 RB7: 60 RB7: No da RB8: 952 RB6: No da RB7: No da RB8: 952 10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC0, UL_TFC0, UL_TFC15, UL_TFC15 UL_TFC1, UL_TFC10, UL_TFC10, UL_TFC16, UL_TFC16, UL_TFC24, RB6: 103 RB6: No da							
UL_TFC0, UL_TFC15 UL_TFC15, UL_TFC24 RB7: 60 RB8: 952 RB7: No da RB8: 952 10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC0, UL_TFC16, UL_TFC16, UL_TFC16, UL_TFC24, RB7: 00 RB7: No da	9						RB5: No data
Image: box with the system UL_TFC15 UL_TFC24 RB8: 952 RB8: 952 10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, DL_TFC15, UL_TFC15, UL_TFC0, UL_TFC10, UL_TFC10, UL_TFC15, RB6: 103 RB6: No da 0 UL_TFC25 UL_TFC25 UL_TFC15, UL_TFC15, UL_TFC10, UL_TFC16, UL_TFC16, RB8: 952 RB8: 952		DL_TFC24	UL_TFC24				RB6: No data
10 DL_TFC10, DL_TFC25 UL_TFC10, UL_TFC25 DL_TFC10, UL_TFC25 DL_TFC10, DL_TFC15, UL_TFC0, UL_TFC0, UL_TFC15, UL_TFC1, UL_TFC10, RB5: 39 RB6: 103 RB5: 39 RB6: No da 10 DL_TFC25 DL_TFC10, UL_TFC15, UL_TFC1, UL_TFC10, RB6: 103 RB6: No da 10 UL_TFC15, UL_TFC10, UL_TFC10, RB8: 952 RB8: 952 10 UL_TFC16, UL_TFC24, UL_TFC24, UL_TFC24, UL_TFC24,							
DL_TFC25 UL_TFC25 DL_TFC15, UL_TFC0, UL_TFC15 UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC15, UL_TFC16, UL_TFC24, RB6: 103 RB7: 60 RB7: No da RB6: No da	10						
UL_TFC0, UL_TFC9, RB7: 60 RB7: No da UL_TFC15 UL_TFC10, RB8: 952 RB8: 952 UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC24,	10						
UL_TFC15 UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC24, RB8: 952 RB8: 952		DL_IFC25	0L_1F625				
UL_TFC15, UL_TFC16, UL_TFC24,			1				
UL_TFC16, UL_TFC24,						1100. 302	1100. 332
UL_TFC24,			1				
UL TFC25					UL_TFC25		

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)		
	Test				(note)	(note)		
11	DL_TFC11, DL_TFC26	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 952	RB5: 81 RB6: 103 RB7: 60 RB8: 952		
12	DL_TFC12, DL_TFC27	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 1272		
13	DL_TFC13, DL_TFC28	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 1272		
14	DL_TFC14, DL_TFC29	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 1272		
NOTE:								

indicator and expansion bit).

14.2.40.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3, 6, 9 and 12: RLC SDUs on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4, 7, 10 and 13: RLC SDUs on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5, 8, 11 and 14: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.41 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:128 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.41.1 Conformance requirement

See 14.2.4.1.

14.2.41.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.41.

14.2.41.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

_	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A
	TF4, bits	N/A	N/A	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (128 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	4x336	N/A
	TF4, bits	N/A	N/A	N/A	8x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test		DI 7700		(note)	(note)
1	DL_TFC1, DL_TFC16	UL_TFC1, UL_TFC16	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: No data
2	DL_TFC2, DL_TFC17	UL_TFC2, UL_TFC17	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: No data
3	DL_TFC3, DL_TFC18	UL_TFC3, UL_TFC18	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC3, UL_TFC15, UL_TFC18	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: No data RB6: No data RB7: No data RB8: 312
4	DL_TFC4, DL_TFC19	UL_TFC4, UL_TFC19	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC3 UL_TFC4, UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC19	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: 312
5	DL_TFC5, DL_TFC20	UL_TFC5, UL_TFC20	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC15, UL_TFC17, UL_TCF18, UL_TFC20	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: 312
6	DL_TFC6, DL_TFC21	UL_TFC6, UL_TFC21	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC6, UL_TFC15, UL_TFC21	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 632
7	DL_TFC7, DL_TFC22	UL_TFC7, UL_TFC22	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC22	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 632
8	DL_TFC8, DL_TFC23	UL_TFC8, UL_TFC23	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC21, UL_TFC23	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 632
9	DL_TFC9, DL_TFC24	UL_TFC9, UL_TFC24	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC9, UL_TFC15, UL_TFC24	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: No data RB6: No data RB7: No data RB8: 1272
10	DL_TFC10, DL_TFC25	UL_TFC10, UL_TFC25	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC25	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: 39 RB6: No data RB7: No data RB8: 1272

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)			
11	DL_TFC11, DL_TFC26	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 952	RB5: 81 RB6: 103 RB7: 60 RB8: 1272			
12	DL_TFC12, DL_TFC27	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 2552			
13	DL_TFC13, DL_TFC28	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC15, , UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 2552			
14	DL_TFC14, DL_TFC29	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 2552			
NOTE:									

length indicator and expansion bit).

14.2.41.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: RLC SDUs on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: RLC SDUs on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
 - for sub-test 6: RLC SDUs on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.

- for sub-test 7: RLC SDUs on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 8: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- for sub-test 9: RLC SDUs on RB8 having the first 952 bits equal to the content equal to the first 952 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 10: RLC SDUs on RB8 having the first 952 bits equal to the content equal to the first 952 bits of the test data sent by the SS in downlink; RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 11: RLC SDUs on RB8 having the first 952 bits equal to the content equal to the first 952 bits of the test data sent by the SS in downlink; RLC SDUs on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 12: RLC SDUs on RB8 having the first 1272 bits equal to the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 13: RLC SDUs on RB8 having the first 1272 bits equal to the content equal to the first 1272 bits of the test data sent by the SS in downlink; RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: RLC SDUs on RB8 having the first 1272 bits equal to the content equal to the first 1272 bits of the test data sent by the SS in downlink; RLC SDUs on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.42 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:256 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.42.1 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:256 kbps / PS RAB / 10 ms TTI
- 14.2.42.1.1 Conformance requirement

See 14.2.4.1.

14.2.42.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.42 for the downlink 10 ms TTI case.

14.2.42.1.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

		TFI	RB5	RB6	RB7	RB8	DCCH
_			(RAB subflow #1)	(RAB subflow #2)	(RAB subflow #3)	(64 kbps)	
	TFS	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
		TF1, bits	1x39	1x103	1x60	1x336	1x148
		TF2, bits	1x81	N/A	N/A	2x336	N/A
		TF3, bits	N/A	N/A	N/A	3x336	N/A
L		TF4, bits	N/A	N/A	N/A	4x336	N/A

Release 5

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (256 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	4x336	N/A
	TF4, bits	N/A	N/A	N/A	8x336	N/A

TFCI	
DL TFC0	(RB5, RB6, RB7, RB8, DCCH)
	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, DL_TFC16	UL_TFC1, DL_TFC16	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: No data
2	DL_TFC2, DL_TFC17	UL_TFC2, DL_TFC17	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: No data
3	DL_TFC3, DL_TFC18	UL_TFC3, DL_TFC18	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC3, UL_TFC15, UL_TFC18	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: No data RB6: No data RB7: No data RB8: 312
4	DL_TFC4, DL_TFC19	UL_TFC4, DL_TFC19	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC19	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: 312
5	DL_TFC5, DL_TFC20	UL_TFC5, DL_TFC20	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC15, UL_TFC17, UL_TFC18, UL_TFC20	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: 312
6	DL_TFC6, DL_TFC21	UL_TFC6, DL_TFC21	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC6, UL_TFC15, UL_TFC21	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 632
7	DL_TFC7, DL_TFC22	UL_TFC7, DL_TFC22	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC22	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 632
8	DL_TFC8, DL_TFC23	UL_TFC8, DL_TFC23	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC21, UL_TFC23	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 632
9	DL_TFC9, DL_TFC24	UL_TFC9, DL_TFC24	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC9, UL_TFC15, UL_TFC24	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: No data RB6: No data RB7: No data RB8: 1272
10	DL_TFC10, DL_TFC25	UL_TFC10, DL_TFC25	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC25	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: 39 RB6: No data RB7: No data RB8: 1272

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)		
11	DL_TFC11, DL_TFC26	UL_TFC11, DL_TFC26	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 952	RB5: 81 RB6: 103 RB7: 60 RB8: 1272		
12	DL_TFC12, DL_TFC27	UL_TFC12, DL_TFC27	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 2552		
13	DL_TFC13, DL_TFC28	UL_TFC13, DL_TFC28	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 2552		
14	DL_TFC14, DL_TFC29	UL_TFC14, DL_TFC29	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 2552		
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the uplink TTI for RB8 is 20 ms while the downlink TTI is 10 ms then, to achieve continous data transmission in uplink the size of the uplink RLC SDU has been set such that it will be transmitted over each TTIs, i.e. UL RLC SDU SIZE has been set to the uplink TFS size under test minus							

8 (the size of a 7 bit length indicator and expansion bit).

14.2.42.1.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.

- for sub-test 6: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 7: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 8: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- for sub-test 9: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 10: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 11: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 12: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 13: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.42.2 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:256 kbps / PS RAB / 20 ms TTI

14.2.42.2.1 Conformance requirement

See 14.2.4.1.

14.2.42.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.42 for the downlink 20 ms TTI case.

14.2.42.2.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

		TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps)	DCCH
		TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
		TF1, bits	1x39	1x103	1x60	1x336	1x148
	TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
		TF3, bits	N/A	N/A	N/A	3x336	N/A
		TF4, bits	N/A	N/A	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (256 kbps, 20 ms)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	2x336	N/A
TFS	TF3, bits	N/A	N/A	N/A	4x336	N/A
	TF4, bits	N/A	N/A	N/A	8x336	N/A
	TF5, bits	N/A	N/A	N/A	12x336	N/A
	TF6, bits	N/A	N/A	N/A	16x336	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF5, TF0)
DL_TFC16	(TF1, TF0, TF0, TF5, TF0)
DL_TFC17	(TF2, TF1, TF1, TF5, TF0)
DL_TFC18	(TF0, TF0, TF0, TF6, TF0)
DL_TFC19	(TF1, TF0, TF0, TF6, TF0)
DL_TFC20	(TF2, TF1, TF1, TF6, TF0)
DL_TFC21	(TF0, TF0, TF0, TF0, TF1)
DL_TFC22	(TF1, TF0, TF0, TF0, TF1)
DL_TFC23	(TF2, TF1, TF1, TF0, TF1)
DL_TFC24	(TF0, TF0, TF0, TF1, TF1)
DL_TFC25	(TF1, TF0, TF0, TF1, TF1)
DL_TFC26	(TF2, TF1, TF1, TF1, TF1)
DL_TFC27	(TF0, TF0, TF0, TF2, TF1)
DL_TFC28	(TF1, TF0, TF0, TF2, TF1)
DL_TFC29	(TF2, TF1, TF1, TF2, TF1)
DL_TFC30	(TF0, TF0, TF0, TF3, TF1)
DL_TFC31	(TF1, TF0, TF0, TF3, TF1)
DL_TFC32	(TF2, TF1, TF1, TF3, TF1)
DL_TFC33	(TF0, TF0, TF0, TF4, TF1)
DL_TFC34	(TF1, TF0, TF0, TF4, TF1)
DL_TFC35	(TF2, TF1, TF1, TF4, TF1)
DL_TFC36 DL_TFC37	(TF0, TF0, TF0, TF5, TF1) (TF1, TF0, TF0, TF5, TF1)
DL_TFC38 DL_TFC39	(TF2, TF1, TF1, TF5, TF1) (TF0, TF0, TF0, TF6, TF1)
DL_TFC39	(TF1, TF0, TF0, TF6, TF1)
DL_TFC40 DL_TFC41	(TF2, TF1, TF1, TF6, TF1)
	(172, 171, 171, 170, 171)

Sub-tests:

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
1	Test DL_TFC1, DL_TFC22	UL_TFC1, UL_TFC16	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC15, UL_TFC16	(note) RB5: 39 RB6: 103 RB7: 60 RB8: 312	(note) RB5: 39 RB6: No data RB7: No data RB8: No data
2	DL_TFC2, DL_TFC23	UL_TFC2, UL_TFC17	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: No data
3	DL_TFC3, DL_TFC24	UL_TFC3, UL_TFC18	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC3, UL_TFC15, UL_TFC18	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: No data RB6: No data RB7: No data RB8: 312
4	DL_TFC4, DL_TFC25	UL_TFC4, UL_TFC19	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC19	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: 312
5	DL_TFC5, DL_TFC26	UL_TFC5, UL_TFC20	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC15, UL_TFC17, UL_TFC18, UL_TFC20	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: 312
6	DL_TFC6, DL_TFC27	UL_TFC6, UL_TFC21	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC6, UL_TFC15, UL_TFC21	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 632
7	DL_TFC7, DL_TFC28	UL_TFC7, UL_TFC22	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC22	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 632
8	DL_TFC8, DL_TFC29	UL_TFC8, UL_TFC23	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC21 UL_TFC23	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 632
9	DL_TFC9, DL_TFC30	UL_TFC9, UL_TFC24	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC9, UL_TFC15, UL_TFC24	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: No data RB6: No data RB7: No data RB8: 1272
10	DL_TFC10, DL_TFC31	UL_TFC10, UL_TFC25	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC25	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: 39 RB6: No data RB7: No data RB8: 1272

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test				(note)	(note)
11	DL_TFC11, DL_TFC32	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 952	RB5: 81 RB6: 103 RB7: 60 RB8: 1272
12	DL_TFC12, DL_TFC33	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 2552
13	DL_TFC13, DL_TFC34	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27 UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 2552
14	DL_TFC14, DL_TFC35	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 2552
15	DL_TFC15, DL_TFC36	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 3832
16	DL_TFC16, DL_TFC37	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27 UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 3832
17	DL_TFC17, DL_TFC38	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 3832
18	DL_TFC18, DL_TFC39	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 5112
19	DL_TFC19, DL_TFC40	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC21, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 5112

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)		
	Under	Under test			(bits)	<i>i</i>		
	Test				(note)	(note)		
20	DL_TFC20,	UL_TFC14,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81		
	DL_TFC41	UL_TFC29	DL_TFC21,	UL_TFC2,	RB6: 103	RB6: 103		
			UL_TFC0,	UL_TFC12,	RB7: 60	RB7: 60		
			UL_TFC15	UL_TFC14,	RB8: 1272	RB8: 5112		
				UL_TFC15,				
				UL_TFC17,				
				UL_TFC27,				
				UL_TFC29				
NOTE:	See TS 34.1	09 [10] clause 5	5.3.2.6.2 for details r	egarding loopback	of RLC SDUs.			
	RB8: Test da	ata size has bee	n set to DL TFS size	e under test minus 8	B bits (size of 7 bit le	ength indicator		
	and expansion bit). As the TTI for RB8 is the same for both downlink and uplink then UL RLC SDU size							
	has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size has been set equal to							
					ator and expansion			

14.2.42.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB5 and RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
 - for sub-test 6: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 7: an RLC SDU on RB5 and RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
 - for sub-test 8: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
 - for sub-test 9: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 10: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 11: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
 - for sub-test 12: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.

- for sub-test 13: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 15: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 16: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 17: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 18: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 19: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 20: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.43 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:384 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.43.1 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:384 kbps / PS RAB / 10 ms TTI
- 14.2.43.1.1 Conformance requirement

See 14.2.4.1.

14.2.43.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.43 for the downlink 10 ms TTI case.

14.2.43.1.3 Method of test

See 14.1.2 for test procedure.

For the PS DL:384/UL:64 kbps radio bearer the downlink TTI is 10ms while the uplink TTI is 20ms. As the SS will send one DL SDU every 10 ms then the UE test loop function will return 2 UL SDUs per uplink TTI. To not cause uplink transmission buffer overflow then the UL RLC SDU size should be chosen such that the UE will transmit 2 RLC SDUs per uplink TTI. For the case when the transport format under test does not allow for 2 SDUs to fit into the transport format size without requiring concatenation then the UL RLC SDU size shall be chosen such that one SDU is returned per uplink TTI.

The following RLC parameter value is used in the RADIO BEARER SETUP message used to setup the PS DL:384/UL:64 kbps radio bearer:

Uplink RLC	
Transmission window size	512

Release 5

NOTE The transmission window size value have been chosen to avoid that UE transmission buffer becomes full during the test.

Uplink TFS:

		TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
TFS		TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
		TF1, bits	1x39	1x103	1x60	1x336	1x148
	TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A	
		TF4, bits	N/A	N/A	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (384 kbps, 10 ms)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
15	TF3, bits	N/A	N/A	N/A	4x336	N/A
	TF4, bits	N/A	N/A	N/A	8x336	N/A
	TF5, bits	N/A	N/A	N/A	12x336	N/A

TECI	
TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF5, TF0)
DL_TFC16	(TF1, TF0, TF0, TF5, TF0)
DL_TFC17	(TF2, TF1, TF1, TF5, TF0)
DL_TFC18	(TF0, TF0, TF0, TF1)
DL_TFC19	(TF1, TF0, TF0, TF0, TF1)
DL_TFC20	(TF2, TF1, TF1, TF0, TF1)
DL_TFC21	(TF0, TF0, TF0, TF1, TF1)
DL_TFC22	(TF1, TF0, TF0, TF1, TF1)
DL_TFC23	(TF2, TF1, TF1, TF1, TF1)
DL_TFC24	(TF0, TF0, TF0, TF2, TF1)
DL_TFC25	(TF1, TF0, TF0, TF2, TF1)
DL_TFC26	(TF2, TF1, TF1, TF2, TF1)
DL_TFC27	(TF0, TF0, TF0, TF3, TF1)
DL_TFC28	(TF1, TF0, TF0, TF3, TF1)
DL_TFC29	(TF2, TF1, TF1, TF3, TF1)
DL_TFC30	(TF0, TF0, TF0, TF4, TF1)
DL_TFC31	(TF1, TF0, TF0, TF4, TF1)
DL_TFC32	(TF2, TF1, TF1, TF4, TF1)
DL_TFC33	(TF0, TF0, TF0, TF5, TF1)
DL_TFC34	(TF1, TF0, TF0, TF5, TF1)
DL_TFC35	(TF2, TF1, TF1, TF5, TF1)

Sub-tests:

Sub-	Downlink	Uplink	Implicitely	Restricted UL	UL RLC SDU	Test data size
test	TFCS Under	TFCS Under test	tested	TFCIs	size (bits)	(bits)
	Test				(note 1)	(note 1)
1	DL_TFC1, DL_TFC19	UL_TFC1, UL_TFC16	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: No data
2	DL_TFC2, DL_TFC20	UL_TFC2, UL_TFC17	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: No data
3	DL_TFC3, DL_TFC21	UL_TFC3, UL_TFC18	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC3, UL_TFC15, UL_TFC18	RB5: 39 RB6: 103 RB7: 60 RB8: 312 (note 2)	RB5: No data RB6: No data RB7: No data RB8: 312
4	DL_TFC4, DL_TFC22	UL_TFC4, UL_TFC19	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC19	RB5: 39 RB6: 103 RB7: 60 RB8: 312 (note 2)	RB5: 39 RB6: No data RB7: No data RB8: 312
5	DL_TFC5, DL_TFC23	UL_TFC5, UL_TFC20	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC5, UL_TFC17, UL_TFC18, UL_TFC15, UL_TFC20	RB5: 81 RB6: 103 RB7: 60 RB8: 312 (note 2)	RB5: 81 RB6: 103 RB7: 60 RB8: 312
6	DL_TFC6, DL_TFC24	UL_TFC6, UL_TFC21	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC6, UL_TFC15, UL_TFC21	RB5: 39 RB6: 103 RB7: 60 RB8: 312 (note 3)	RB5: No data RB6: No data RB7: No data RB8: 632
7	DL_TFC7, DL_TFC25	UL_TFC7, UL_TFC22	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC22	RB5: 39 RB6: 103 RB7: 60 RB8: 312 (note 3)	RB5: 39 RB6: No data RB7: No data RB8: 632
8	DL_TFC8, DL_TFC26	UL_TFC8, UL_TFC23	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC21, UL_TFC23	RB5: 81 RB6: 103 RB7: 60 RB8: 312 (note 3)	RB5: 81 RB6: 103 RB7: 60 RB8: 632
9	DL_TFC9, DL_TFC27	UL_TFC9, UL_TFC24	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC9, UL_TFC15, UL_TFC24	RB5: 39 RB6: 103 RB7: 60 RB8: 1272 (note 2)	RB5: No data RB6: No data RB7: No data RB8: 1272

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
lest	Under Test	Under test	lesteu	TFCIS	(bits) (note 1)	(note 1)
10	DL_TFC10, DL_TFC28	UL_TFC10, UL_TFC25	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC25	RB5: 39 RB6: 103 RB7: 60 RB8: 1272 (note 2)	RB5: 39 RB6: No data RB7: No data RB8: 1272
11	DL_TFC11, DL_TFC29	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 1272 (note 2)	RB5: 81 RB6: 103 RB7: 60 RB8: 1272
12	DL_TFC12, DL_TFC30	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 632 (note 3)	RB5: No data RB6: No data RB7: No data RB8: 2552
13	DL_TFC13, DL_TFC31	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 632 (note 3)	RB5: 39 RB6: No data RB7: No data RB8: 2552
14	DL_TFC14, DL_TFC32	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 632 (note 3)	RB5: 81 RB6: 103 RB7: 60 RB8: 2552
15	DL_TFC15, DL_TFC33	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 632 (note 3)	RB5: No data RB6: No data RB7: No data RB8: 3832
16	DL_TFC16, DL_TFC34	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC18, , UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 632 (note 3)	RB5: 39 RB6: No data RB7: No data RB8: 3832
17	DL_TFC17, DL_TFC35	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC18, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 632 (note 3)	RB5: 81 RB6: 103 RB7: 60 RB8: 3832

Sub-	Downlink	Uplink	Implicitely	Restricted UL	UL RLC SDU	Test data size			
test	TFCS	TFCS	tested	TFCIs	size	(bits)			
	Under	Under test			(bits)				
	Test				(note 1)	(note 1)			
NOTE 1				egarding loopback of					
	RB8: Test da	ata size has bee	n set to the payload	I size of the DL TF un	der test minus 8 b	its (size of 7 bit			
	length indica	ator and expansi	on bit).						
NOTE 2	: RB8 (TF1/TI	-3): For sub-test	s where uplink trans	sport format TF1 (1x3	36) or TF3 (3x336) are used then			
				10 ms) and uplink TTI					
	require the l	JE to concatena	e 2 SDUs into one	PDU for TF1; or into t	hree PDUs for TF	For these			
	sub-tests the	e UL RLC SDU s	ize is set equal to the	he payload size of the	UL TF under test	minus 8 bits			
	(the size of 7	7 bit length indica	ator and expansion	bit).					
NOTE 3	: RB8 (TF2/TI	F4): For sub-test	s where uplink trans	sport formats TF2 (2x	336) or TF4 (4x33	6) is used then			
				s) and uplink TTI (20m					
		been chosen such that 2 SDUs will be returned per uplink TTI. I.e. the UL RLC SDU size is set equal to							
	half the payl	half the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and							
	expansion b	it).							

14.2.43.1.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
 - for sub-test 3: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 6: RLC SDUs on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 7: RLC SDUs on RB5 having the same content as sent by the SS; RLC SDUs on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
 - for sub-test 8: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and RLC SDUs on RB8 having the first 312 bits equal to the content of the test data sent by the SS in downlink.
 - for sub-test 9: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 10: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 11: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
 - for sub-test 12: RLC SDUs on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.

- for sub-test 13: RLC SDUs on RB5 having the same content as sent by the SS; RLC SDUs on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
- for sub-test 14: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and RLC SDUs on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink.
- for sub-test 15: RLC SDUs on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 16: RLC SDUs on RB5 having the same content as sent by the SS; RLC SDUs on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
- for sub-test 17: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the the SS; and RLC SDUs on RB8 having the first 632 bits equal to the content equal to the first 632 bits of the test data sent by the SS in downlink.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.43.2 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:64 DL:384 kbps / PS RAB / 20 ms TTI

14.2.43.2.1 Conformance requirement

See 14.2.4.1.

14.2.43.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.43 for the downlink 20 ms TTI case.

14.2.43.2.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

		TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps)	DCCH
		TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
TFS		TF1, bits	1x39	1x103	1x60	1x336	1x148
	TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	3x336	N/A	
		TF4, bits	N/A	N/A	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (384 kbps, 20 ms)	DCCH
	TF0, bits	1x0	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	4x336	N/A
TFS	TF4, bits	N/A	N/A	N/A	8x336	N/A
	TF5, bits	N/A	N/A	N/A	12x336	N/A
	TF6, bits	N/A	N/A	N/A	16x336	N/A
	TF7, bits	N/A	N/A	N/A	20x336	N/A
	TF8, bits	N/A	N/A	N/A	24x336	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF5, TF0)
DL_TFC16	(TF1, TF0, TF0, TF5, TF0)
DL_TFC17	(TF2, TF1, TF1, TF5, TF0)
DL_TFC18	(TF0, TF0, TF0, TF6, TF0)
DL_TFC19	(TF1, TF0, TF0, TF6, TF0)
DL_TFC20	(TF2, TF1, TF1, TF6, TF0)
DL_TFC21	(TF0, TF0, TF0, TF7, TF0)
DL_TFC22	(TF1, TF0, TF0, TF7, TF0)
DL_TFC23	(TF2, TF1, TF1, TF7, TF0)
DL_TFC24	(TF0, TF0, TF0, TF8, TF0)
DL_TFC25	(TF1, TF0, TF0, TF8, TF0)
DL_TFC26	(TF2, TF1, TF1, TF8, TF0)
DL_TFC27	(TF0, TF0, TF0, TF1)
DL_TFC28	(TF1, TF0, TF0, TF0, TF1)
DL_TFC29	(TF2, TF1, TF1, TF0, TF1)
DL_TFC30	(TF0, TF0, TF0, TF1, TF1)
DL_TFC31	(TF1, TF0, TF0, TF1, TF1)
DL_TFC32	(TF2, TF1, TF1, TF1, TF1)
DL_TFC33	(TF0, TF0, TF0, TF2, TF1)
DL_TFC34	(TF1, TF0, TF0, TF2, TF1)
DL_TFC35	(TF2, TF1, TF1, TF2, TF1)
DL_TFC36	(TF0, TF0, TF0, TF3, TF1)
DL_TFC37	(TF1, TF0, TF0, TF3, TF1)
DL_TFC38	(TF2, TF1, TF1, TF3, TF1)
DL_TFC39	(TF0, TF0, TF0, TF4, TF1)
DL_TFC40	(TF1, TF0, TF0, TF4, TF1)
DL_TFC41	(TF2, TF1, TF1, TF4, TF1)
DL_TFC42	(TF0, TF0, TF0, TF5, TF1)
DL_TFC43	(TF1, TF0, TF0, TF5, TF1)
DL_TFC44	(TF2, TF1, TF1, TF5, TF1)
DL_TFC45	(TF0, TF0, TF0, TF6, TF1)
DL_TFC46	(TF1, TF0, TF0, TF6, TF1)
DL_TFC47	(TF2, TF1, TF1, TF6, TF1)
DL_TFC48	(TF0, TF0, TF0, TF7, TF1)
DL_TFC49	(TF1, TF0, TF0, TF7, TF1)
DL_TFC50	(TF2, TF1, TF1, TF7, TF1)
DL_TFC51	(TF0, TF0, TF0, TF8, TF1)
DL_TFC52	(TF1, TF0, TF0, TF8, TF1)
DL_TFC53	(TF2, TF1, TF1, TF8, TF1)

Sub-tests:

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under	Under test			(bits)	
	Test				(note)	(note)
1	DL_TFC1,	UL_TFC1,U	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC28	L_TFC16	DL_TFC27,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC16	RB8: 312	RB8: No data
2	DL_TFC2,	UL_TFC2,U	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC29	L_TFC17	DL_TFC27,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC17	RB8: 312	RB8: No data
3	DL_TFC3,	UL_TFC3,U	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC30	L_TFC19	DL_TFC27,	UL_TFC3,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC18	RB8: 312	RB8: 312

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test				(note)	(note)
4	DL_TFC4, DL_TFC31	UL_TFC4,U L_TFC19	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15,	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC19	RB5: 39 RB6: 103 RB7: 60 RB8: 312	RB5: 39 RB6: No data RB7: No data RB8: 312
5	DL_TFC5, DL_TFC32	UL_TFC5,U L_TFC20	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC15, UL_TFC17, UL_TFC18 UL_TFC20	RB5: 81 RB6: 103 RB7: 60 RB8: 312	RB5: 81 RB6: 103 RB7: 60 RB8: 312
6	DL_TFC6, DL_TFC33	UL_TFC6,U L_TFC21	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC6, UL_TFC15, UL_TFC21	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: No data RB6: No data RB7: No data RB8: 632
7	DL_TFC7, DL_TFC34	UL_TFC7,U L_TFC22	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC22	RB5: 39 RB6: 103 RB7: 60 RB8: 632	RB5: 39 RB6: No data RB7: No data RB8: 632
8	DL_TFC8, DL_TFC35	UL_TFC8,U L_TFC23	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC21, UL_TFC23	RB5: 81 RB6: 103 RB7: 60 RB8: 632	RB5: 81 RB6: 103 RB7: 60 RB8: 632
9	DL_TFC9, DL_TFC36	UL_TFC9,U L_TFC24	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC9, UL_TFC15, UL_TFC24	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: No data RB6: No data RB7: No data RB8: 1272
10	DL_TFC10, DL_TFC37	UL_TFC10, UL_TFC25	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24 UL_TFC25	RB5: 39 RB6: 103 RB7: 60 RB8: 952	RB5: 39 RB6: No data RB7: No data RB8: 1272
11	DL_TFC11, DL_TFC38	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 952	RB5: 81 RB6: 103 RB7: 60 RB8: 1272
12	DL_TFC12, DL_TFC39	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 2552

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test				(note)	(note)
13	DL_TFC13, DL_TFC40	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 2552
14	DL_TFC14, DL_TFC41	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 2552
15	DL_TFC15, DL_TFC42	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 3832
16	DL_TFC16, DL_TFC43	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC27, , UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 3832
17	DL_TFC17, DL_TFC44	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 3832
18	DL_TFC18, DL_TFC45	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 5112
19	DL_TFC19, DL_TFC46	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 5112
20	DL_TFC20, DL_TFC47	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 5112
21	DL_TFC21, DL_TFC48	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 6392

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
22	DL_TFC22, DL_TFC49	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 6392
23	DL_TFC23, DL_TFC50	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 6392
24	DL_TFC24, DL_TFC51	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: No data RB6: No data RB7: No data RB8: 7672
25	DL_TFC25, DL_TFC52	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 1272	RB5: 39 RB6: No data RB7: No data RB8: 7672
26	DL_TFC26, DL_TFC53	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC27, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 7672
NOTE:	RB8: Test da and expansi	ata size has bee on bit). As the T	n set to DL TFS s TI for RB8 is the s	s regarding loopback of ize under test minus & ame for both downlinl SDU per TTI, i.e. the	3 bits (size of 7 bit k and uplink then I	JL ŘLC SDU

14.2.43.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.

equal to the uplink TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit).

- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.

- for sub-test 4: an RLC SDU on RB5 and RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
- for sub-test 5: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- for sub-test 6: an RLC SDU on RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 7: an RLC SDU on RB5 and RB8 having the content equal to the content of the test data sent by the SS in downlink; and no data shall be received on RB6 and RB7.
- for sub-test 8: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- for sub-test 9: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 10: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 11: an RLC SDU on RB8 having the content equal to the first 952 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 12: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 13: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 15: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 16: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 17: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 18: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 19: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 20: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 21: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 22: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 23: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 24: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.

- for sub-test 25: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 26: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.44 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:128 DL:2048 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.44.1 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:128 DL:2048 kbps / PS RAB / 10 ms TTI
- 14.2.44.1.1 Conformance requirement

See 14.2.4.1.

14.2.44.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.44 for the downlink 10 ms TTI case.

14.2.44.1.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (128 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
	TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A
	TF3, bits	N/A	N/A	N/A	4x336	N/A
	TF4, bits	N/A	N/A	N/A	8x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (2048 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x656	0x148
	TF1, bits	1x39	1x103	1x60	1x656	1x148
	TF2, bits	1x81	N/A	N/A	2x656	N/A
	TF3, bits	N/A	N/A	N/A	4x656	N/A
	TF4, bits	N/A	N/A	N/A	8x656	N/A
TFS	TF5, bits	N/A	N/A	N/A	12x656	N/A
	TF6, bits	N/A	N/A	N/A	16x656	N/A
	TF7, bits	N/A	N/A	N/A	20x656	N/A
	TF8, bits	N/A	N/A	N/A	24x656	N/A
	TF9, bits	N/A	N/A	N/A	28x656	N/A
	TF10, bits	N/A	N/A	N/A	32x656	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF5, TF0)
DL TFC16	(TF1, TF0, TF0, TF5, TF0)
DL_TFC17	(TF2, TF1, TF1, TF5, TF0)
DL_TFC18	(TF0, TF0, TF0, TF6, TF0)
DL_TFC19	(TF1, TF0, TF0, TF6, TF0)
DL_TFC20	(TF2, TF1, TF1, TF6, TF0)
DL_TFC21	(TF0, TF0, TF0, TF7, TF0)
DL TFC22	(TF1, TF0, TF0, TF7, TF0)
DL TFC23	(TF2, TF1, TF1, TF7, TF0)
DL_TFC24	(TF0, TF0, TF0, TF8, TF0)
DL_TFC25	(TF1, TF0, TF0, TF8, TF0)
DL_TFC26	(TF2, TF1, TF1, TF8, TF0)
DL_TFC27	(TF0, TF0, TF0, TF9, TF0)
DL TFC28	(TF1, TF0, TF0, TF9, TF0)
DL_TFC29	(TF2, TF1, TF1, TF9, TF0)
DL TFC30	(TF0, TF0, TF10, TF10, TF0)
DL_TFC31	(TF1, TF0, TF0, TF10, TF0)
DL_TFC32	(TF2, TF1, TF1, TF10, TF0)
DL_TFC33	(TF0, TF0, TF0, TF1)
DL_TFC34	(TF1, TF0, TF0, TF0, TF1)
DL_TFC35	(TF2, TF1, TF1, TF0, TF1)
DL_TFC36	(TF0, TF0, TF1, TF1)
DL_TFC37	(TF1, TF0, TF0, TF1, TF1)
DL_TFC38	(TF2, TF1, TF1, TF1, TF1)
DL_TFC39	(TF0, TF0, TF0, TF2, TF1)
DL_TFC40	(TF1, TF0, TF0, TF2, TF1)
DL_TFC41	(TF2, TF1, TF1, TF2, TF1)
DL_TFC42	(TF0, TF0, TF0, TF3, TF1)
DL_TFC43	(TF1, TF0, TF0, TF3, TF1)
DL_TFC44	(TF2, TF1, TF1, TF3, TF1)
DL_TFC45	(TF0, TF0, TF0, TF4, TF1)
DL_TFC46	(TF1, TF0, TF0, TF4, TF1)
DL_TFC47	(TF2, TF1, TF1, TF4, TF1)
DL_TFC48	(TF0, TF0, TF0, TF5, TF1)
DL_TFC49	(TF1, TF0, TF0, TF5, TF1)
DL_TFC50	(TF2, TF1, TF1, TF5, TF1)
DL_TFC51	(TF0, TF0, TF0, TF6, TF1)
DL_TFC52	(TF1, TF0, TF0, TF6, TF1)
DL_TFC53	(TF2, TF1, TF1, TF6, TF1)
DL_TFC54	(TF0, TF0, TF0, TF7, TF1)
DL_TFC55	(TF1, TF0, TF0, TF7, TF1)
DL_TFC56	(TF2, TF1, TF1, TF7, TF1)
DL_TFC57	(TF0, TF0, TF0, TF8, TF1)
DL_TFC58	(TF1, TF0, TF0, TF8, TF1)
DL_TFC59	(TF2, TF1, TF1, TF8, TF1)
DL_TFC60	(TF0, TF0, TF0, TF9, TF1)
DL_TFC61	(TF1, TF0, TF0, TF9, TF1)
DL_TFC62	(TF2, TF1, TF1, TF9, TF1)
DL_TFC63	(TF0, TF0, TF10, TF1)
DL_TFC64	(TF1, TF0, TF0, TF10, TF1)
DL_TFC65	(TF2, TF1, TF1, TF10, TF1)

Sub-tests:

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	test				(note)	(note)

Sub-	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL	UL RLC SDU size	Test data size
test	under	Under test	tested	TFCIs	(bits)	(bits)
4	test	UL_TFC1,			(note)	(note)
1	DL_TFC1,	UL_TFC1, UL_TFC16	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC34	UL_IFC10	DL_TFC33, UL_TFC0,	UL_TFC1, UL_TFC15,	RB6: 103 RB7: 60	RB6: No data RB7: No data
			UL_TFC15	UL_TFC16	RB8: 632	
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 81	RB8: No data RB5: 81
2	DL_TFC35	UL_TFC17	DL_TFC33,	UL_TFC2,	RB6: 103	RB6: 103
	DL_11033		UL_TFC0,	UL TFC15,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC17	RB8: 632	RB8: No data
3	DL_TFC3,	UL_TFC3,	DL TFC0,	UL_TFC0,	RB5: 39	RB5: No data
Ŭ	DL_TFC36	UL_TFC18	DL_TFC33,	UL_TFC3,	RB6: 103	RB6: No data
	DL_11 000	02_11010	UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC18	RB8: 312	RB8: 632
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC37	UL_TFC19	DL_TFC33,	UL_TFC1,	RB6: 103	RB6: No data
		• <u>-</u> •.•	UL_TFC0,	UL_TFC3,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC4,	RB8: 312	RB8: 632
				UL_TFC15,		
				UL_TFC16,		
				UL_TFC18,		
				UL_TFC19		
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC38	UL_TFC20	DL_TFC33,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC5,	RB8: 632	RB8: 632
				UL_TFC15,		
				UL_TFC17,		
				UL_TFC18,		
				UL_TFC20		
6	DL_TFC6,	UL_TFC6,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC39	UL_TFC21	DL_TFC33,	UL_TFC6,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC21	RB8: 632	RB8: 1272
7	DL_TFC7,	UL_TFC7,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC40	UL_TFC22	DL_TFC33,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC7,	RB8: 632	RB8: 1272
				UL_TFC15, UL_TFC16,		
				UL_TFC16, UL_TFC21,		
				UL_TFC22		
8	DL_TFC8,	UL_TFC8,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
Ŭ	DL_TFC41	UL_TFC23	DL_TFC33,	UL_TFC2,	RB6: 103	RB6: 103
		51_11 025	UL TFC0,	UL TFC6,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC8,	RB8: 632	RB8: 1272
			020.0	UL_TFC15,		
				UL_TFC17,		
				UL_TFC21,		
				UL_TFC23		
9	DL_TFC9,	UL_TFC9,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC42	UL_TFC24	DL_TFC33,	UL_TFC9,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC24	RB8: 1272	RB8: 2552
10	DL_TFC10,	UL_TFC10,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC43	UL_TFC25	DL_TFC33,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC9,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC10,	RB8: 1272	RB8: 2552
				UL_TFC15,		
				UL_TFC16,		
				UL_TFC24,		
				UL_TFC25		

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
11	test DL_TFC11, DL_TFC44	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	(note) RB5: 81 RB6: 103 RB7: 60 RB8: 1272	(note) RB5: 81 RB6: 103 RB7: 60 RB8: 2552
12	DL_TFC12, DL_TFC45	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 5112
13	DL_TFC13, DL_TFC46	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 5112
14	DL_TFC14, DL_TFC47	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27 UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 5112
15	DL_TFC15, DL_TFC48	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 7672
16	DL_TFC16, DL_TFC49	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 7672
17	DL_TFC17, DL_TFC50	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 7672
18	DL_TFC18, DL_TFC51	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 10232
19	DL_TFC19, DL_TFC52	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 10232

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	under test	Under test			(bits) (note)	(note)
20	DL_TFC20, DL_TFC53	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 10232
21	DL_TFC21, DL_TFC54	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 12792
22	DL_TFC22, DL_TFC55	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 12792
23	DL_TFC23, DL_TFC56	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 12792
24	DL_TFC24, DL_TFC57	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 15352
25	DL_TFC25, DL_TFC58	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 15352
26	DL_TFC26, DL_TFC59	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 15352
27	DL_TFC27, DL_TFC60	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 17912
28	DL_TFC28, DL_TFC61	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 17912

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	test				(note)	(note)
29	DL_TFC29, DL_TFC62	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27 UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 17912
30	DL_TFC30, DL_TFC63	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 20472
31	DL_TFC31, DL_TFC64	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 20472
32	DL_TFC32, DL_TFC65	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC33, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27 UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 20472
NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB8: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the uplink TTI for RB8 is 20 ms while the downlink TTI is 10 ms then, to achieve continous data transmission in uplink the size of the uplink RLC SDU has been set such that it will be transmitted over each TTI, ie the uplink TFS size minus 8 bits (size of 7 bit length indicator and expansion bit).						

14.2.44.1.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
 - 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the content equal to the first 312 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB8 having the content equal to the first 312 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB8 having the content equal to the first 312 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.

- for sub-test 6: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 7: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 8: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 9: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 10: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 11: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 12: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 13: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 15: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 16: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 17: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 18: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 19: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 20: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 21: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 22: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 23: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 24: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 25: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.

- for sub-test 26: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 27: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 28: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 29: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 30: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 31: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 32: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.44.2 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Interactive or background / UL:128 DL:2048 kbps / PS RAB / 20 ms TTI

14.2.44.2.1 Conformance requirement

See 14.2.4.1.

14.2.44.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.44 for the downlink 20 ms TTI case.

14.2.44.2.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

_		TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (128 kbps)	DCCH
ſ		TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x336	0x148
		TF1, bits	1x39	1x103	1x60	1x336	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x336	N/A	
		TF3, bits	N/A	N/A	N/A	4x336	N/A
		TF4, bits	N/A	N/A	N/A	8x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (2048 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x656	0x148
	TF1, bits	1x39	1x103	1x60	1x656	1x148
	TF2, bits	1x81	N/A	N/A	2x656	N/A
	TF3, bits	N/A	N/A	N/A	4x656	N/A
	TF4, bits	N/A	N/A	N/A	8x656	N/A
	TF5, bits	N/A	N/A	N/A	12x656	N/A
	TF6, bits	N/A	N/A	N/A	16x656	N/A
	TF7, bits	N/A	N/A	N/A	20x656	N/A
	TF8, bits	N/A	N/A	N/A	24x656	N/A
TFS	TF9, bits	N/A	N/A	N/A	28x656	N/A
	TF10, bits	N/A	N/A	N/A	32x656	N/A
	TF11, bits	N/A	N/A	N/A	36x656	N/A
	TF12, bits	N/A	N/A	N/A	40x656	N/A
	TF13, bits	N/A	N/A	N/A	44x656	N/A
	TF14, bits	N/A	N/A	N/A	48x656	N/A
	TF15, bits	N/A	N/A	N/A	52x656	N/A
	TF16, bits	N/A	N/A	N/A	56x656	N/A
	TF17, bits	N/A	N/A	N/A	60x656	N/A
	TF18, bits	N/A	N/A	N/A	64x656	N/A

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL TFC0	(TF0, TF0, TF0, TF0, TF0)
DL TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12 DL_TFC13	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0) (TF2, TF1, TF1, TF4, TF0)
DL_TFC14	(TF0, TF0, TF0, TF5, TF0)
DL_TFC16	(TF1, TF0, TF0, TF5, TF0)
DL_TFC17	(TF2, TF1, TF1, TF5, TF0)
DL TFC18	(TF0, TF0, TF0, TF6, TF0)
DL_TFC19	(TF1, TF0, TF0, TF6, TF0)
DL_TFC20	(TF2, TF1, TF1, TF6, TF0)
DL_TFC21	(TF0, TF0, TF0, TF7, TF0)
DL_TFC22	(TF1, TF0, TF0, TF7, TF0)
DL_TFC23	(TF2, TF1, TF1, TF7, TF0)
DL_TFC24	(TF0, TF0, TF0, TF8, TF0)
DL_TFC25	(TF1, TF0, TF0, TF8, TF0)
DL_TFC26	(TF2, TF1, TF1, TF8, TF0)
DL_TFC27	(TF0, TF0, TF0, TF9, TF0)
DL_TFC28	(TF1, TF0, TF0, TF9, TF0)
DL_TFC29 DL_TFC30	(TF2, TF1, TF1, TF9, TF0) (TF0, TF0, TF0, TF10, TF0)
DL_TFC30	(TF1, TF0, TF0, TF10, TF0)
DL_TFC32	(TF2, TF1, TF1, TF10, TF0)
DL_TFC33	(TF0, TF0, TF1, TF0)
DL_TFC34	(TF1, TF0, TF1, TF0)
DL_TFC35	(TF2, TF1, TF1, TF11, TF0)
DL_TFC36	(TF0, TF0, TF0, TF12, TF0)
DL_TFC37	(TF1, TF0, TF0, TF12, TF0)
DL_TFC38	(TF2, TF1, TF1, TF12, TF0)
DL_TFC39	(TF0, TF0, TF0, TF13, TF0)
DL_TFC40	(TF1, TF0, TF0, TF13, TF0)
DL_TFC41	(TF2, TF1, TF1, TF13, TF0)
DL_TFC42	(TF0, TF0, TF0, TF14, TF0)
DL_TFC43	(TF1, TF0, TF0, TF14, TF0)
DL_TFC44 DL_TFC45	(TF2, TF1, TF1, TF14, TF0)
DL_TFC45	(TF0, TF0, TF0, TF15, TF0) (TF1, TF0, TF0, TF15, TF0)
DL_TFC46	(TF2, TF1, TF1, TF15, TF0)
DL_TFC48	(TF0, TF0, TF0, TF16, TF0)
DL_TFC49	(TF1, TF0, TF0, TF16, TF0)
DL_TFC50	(TF2, TF1, TF1, TF16, TF0)
DL_TFC51	(TF0, TF0, TF17, TF0)
DL_TFC52	(TF1, TF0, TF0, TF17, TF0)
DL_TFC53	(TF2, TF1, TF1, TF17, TF0)
DL_TFC54	(TF0, TF0, TF0, TF18, TF0)
DL_TFC55	(TF1, TF0, TF0, TF18, TF0)
DL_TFC56	(TF2, TF1, TF1, TF18, TF0)
DL_TFC57	(TF0, TF0, TF0, TF1)
DL_TFC58	(TF1, TF0, TF0, TF0, TF1)
DL_TFC59	(TF2, TF1, TF1, TF0, TF1)
DL_TFC60	(TF0, TF0, TF0, TF1, TF1)

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC61	(TF1, TF0, TF0, TF1, TF1)
DL_TFC61	(TF2, TF1, TF1, TF1, TF1)
DL_TFC63	(TF0, TF0, TF0, TF2, TF1)
DL_TFC64	(TF1, TF0, TF0, TF2, TF1)
DL_TFC65	(TF2, TF1, TF1, TF2, TF1)
DL_TFC66	(TF0, TF0, TF0, TF3, TF1)
DL_TFC67	(TF1, TF0, TF0, TF3, TF1)
DL_TFC68	(TF2, TF1, TF1, TF3, TF1)
DL_TFC69	(TF0, TF0, TF0, TF4, TF1)
DL_TFC70	(TF1, TF0, TF0, TF4, TF1)
DL_TFC71	(TF2, TF1, TF1, TF4, TF1)
DL_TFC72	(TF0, TF0, TF0, TF5, TF1)
DL_TFC73	(TF1, TF0, TF0, TF5, TF1)
DL_TFC74	(TF2, TF1, TF1, TF5, TF1)
DL_TFC75	(TF0, TF0, TF0, TF6, TF1)
DL_TFC76	(TF1, TF0, TF0, TF6, TF1)
DL_TFC77 DL_TFC78	(TF2, TF1, TF1, TF6, TF1)
DL_TFC78 DL_TFC79	(TF0, TF0, TF0, TF7, TF1) (TF1, TF0, TF0, TF7, TF1)
DL_TFC80	(TF2, TF1, TF1, TF7, TF1)
DL TFC81	(TF0, TF0, TF0, TF8, TF1)
DL_TFC82	(TF1, TF0, TF0, TF8, TF1)
DL_TFC83	(TF2, TF1, TF1, TF8, TF1)
DL TFC84	(TF0, TF0, TF0, TF1)
DL_TFC85	(TF1, TF0, TF0, TF9, TF1)
DL_TFC86	(TF2, TF1, TF1, TF9, TF1)
DL_TFC87	(TF0, TF0, TF0, TF10, TF1)
DL_TFC88	(TF1, TF0, TF0, TF10, TF1)
DL_TFC89	(TF2, TF1, TF1, TF10, TF1)
DL_TFC90	(TF0, TF0, TF0, TF11, TF1)
DL_TFC91	(TF1, TF0, TF0, TF11, TF1)
DL_TFC92	(TF2, TF1, TF1, TF11, TF1)
DL_TFC93	(TF0, TF0, TF12, TF1)
DL_TFC94	(TF1, TF0, TF0, TF12, TF1)
DL_TFC95	(TF2, TF1, TF1, TF12, TF1)
DL_TFC96	(TF0, TF0, TF0, TF13, TF1)
DL_TFC97	(TF1, TF0, TF0, TF13, TF1)
DL_TFC98 DL_TFC99	(TF2, TF1, TF1, TF13, TF1)
DL_TFC100	(TF0, TF0, TF0, TF14, TF1) (TF1, TF0, TF0, TF14, TF1)
DL_TFC100	(TF2, TF1, TF1, TF14, TF1)
DL_TFC102	(TF0, TF0, TF0, TF15, TF1)
DL TFC103	(TF1, TF0, TF0, TF15, TF1)
DL TFC104	(TF2, TF1, TF1, TF15, TF1)
DL_TFC105	(TF0, TF0, TF16, TF1)
DL_TFC106	(TF1, TF0, TF0, TF16, TF1)
DL_TFC107	(TF2, TF1, TF1, TF16, TF1)
DL_TFC108	(TF0, TF0, TF0, TF17, TF1)
DL_TFC109	(TF1, TF0, TF0, TF17, TF1)
DL_TFC110	(TF2, TF1, TF1, TF17, TF1)
DL_TFC111	(TF0, TF0, TF0, TF18, TF1)
DL_TFC112	(TF1, TF0, TF0, TF18, TF1)
DL_TFC113	(TF2, TF1, TF1, TF18, TF1)

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
1	test DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	(note) RB5: 39	(note) RB5: 39
1	DL_TFC58	UL_TFC16	DL_TFC57,	UL_TFC1,	RB6: 103	RB5: 39 RB6: No data
	DL_11030	02_11010	UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC16	RB8: 632	RB8: No data
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
-	DL_TFC59	UL_TFC17	DL_TFC57,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC17	RB8: 632	RB8: No data
3	DL_TFC3,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC60	UL_TFC18	DL_TFC57,	UL_TFC3,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC18	RB8: 312	RB8: 632
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC61	UL_TFC19	DL_TFC57,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC4,	RB8: 312	RB8: 632
				UL_TFC15, UL_TFC16,		
				UL_TFC18		
				UL_TFC19		
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
C	DL_TFC62	UL_TFC20	DL_TFC57,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC5,	RB8: 312	RB8: 632
				UL_TFC15,		
				UL_TFC17,		
				UL_TFC18,		
				UL_TFC20		
6	DL_TFC6,	UL_TFC6,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC63	UL_TFC21	DL_TFC57,	UL_TFC6,	RB6: 103	RB6: No data
			UL_TFC0, UL_TFC15	UL_TFC15, UL_TFC21	RB7: 60 RB8: 632	RB7: No data RB8: 1272
7	DL_TFC7,	UL_TFC7,	DL_TFC0,	UL TFC0,	RB5: 39	RB5: 39
'	DL_TFC64	UL_TFC22	DL_TFC57,	UL_TFC1,	RB6: 103	RB6: No data
	52	02_11 022	UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC7,	RB8: 632	RB8: 1272
			_	UL_TFC15,		
				UL_TFC16,		
				UL_TFC21,		
				UL_TFC22		
8	DL_TFC8,	UL_TFC8,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC65	UL_TFC23	DL_TFC57,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC8, UL_TFC15,	RB8: 632	RB8: 1272
				UL_TFC17,		
				UL_TFC21,		
				UL TFC23		
9	DL_TFC9,	UL_TFC9,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
-	DL_TFC66	UL_TFC24	DL_TFC57,	UL_TFC9,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC15,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC24	RB8: 1272	RB8: 2552
10	DL_TFC10,	UL_TFC10,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC67	UL_TFC25	DL_TFC57,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC9,	RB7: 60	RB7: No data
			UL_TFC15	UL_TFC10,	RB8: 1272	RB8: 2552
				UL_TFC15,		
				UL_TFC16,		
				UL_TFC24,		
				UL_TFC25		

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	test	Under test			(note)	(note)
11	DL_TFC11, DL_TFC68	UL_TFC11, UL_TFC26	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9 UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 1272	RB5: 81 RB6: 103 RB7: 60 RB8: 2552
12	DL_TFC12, DL_TFC69	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 5112
13	DL_TFC13, DL_TFC70	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 5112
14	DL_TFC14, DL_TFC71	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 5112
15	DL_TFC15, DL_TFC72	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 7672
16	DL_TFC16, DL_TFC73	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 7672
17	DL_TFC17, DL_TFC74	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27 UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 7672
18	DL_TFC18, DL_TFC75	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 10232
19	DL_TFC19, DL_TFC76	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 10232

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	under test	Under test			(bits) (note)	(note)
20	DL_TFC20, DL_TFC77	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 10232
21	DL_TFC21, DL_TFC78	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 12792
22	DL_TFC22, DL_TFC79	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 12792
23	DL_TFC23, DL_TFC80	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 12792
24	DL_TFC24, DL_TFC81	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 15352
25	DL_TFC25, DL_TFC82	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 15352
26	DL_TFC26, DL_TFC83	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 15352
27	DL_TFC27, DL_TFC84	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 17912
28	DL_TFC28, DL_TFC85	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 17912

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	test				(note)	(note)
29	DL_TFC29, DL_TFC86	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 17912
30	DL_TFC30, DL_TFC87	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 20472
31	DL_TFC31, DL_TFC88	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 20472
32	DL_TFC32, DL_TFC89	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 20472
33	DL_TFC33, DL_TFC90	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 23032
34	DL_TFC34, DL_TFC91	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 23032
35	DL_TFC35, DL_TFC92	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 23032
36	DL_TFC36, DL_TFC93	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 25592
37	DL_TFC37, DL_TFC94	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 25592

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
38	test DL_TFC38, DL_TFC95	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	(note) RB5: 81 RB6: 103 RB7: 60 RB8: 2552	(note) RB5: 81 RB6: 103 RB7: 60 RB8: 25592
39	DL_TFC39, DL_TFC96	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 28152
40	DL_TFC40, DL_TFC97	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 28152
41	DL_TFC41, DL_TFC98	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 28152
42	DL_TFC42, DL_TFC99	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 30712
43	DL_TFC43, DL_TFC100	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 30712
44	DL_TFC44, DL_TFC101	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 30712
45	DL_TFC45, DL_TFC102	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 33272
46	DL_TFC46, DL_TFC103	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 33272

Sub- test	Downlink TFCS under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	test				(note)	(note)
47	DL_TFC47, DL_TFC104	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 33272
48	DL_TFC48, DL_TFC105	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 35832
49	DL_TFC49, DL_TFC106	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 35832
50	DL_TFC50, DL_TFC107	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 35832
51	DL_TFC51, DL_TFC108	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 38392
52	DL_TFC52, DL_TFC109	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 38392
53	DL_TFC53, DL_TFC110	UL_TFC14, UL_TFC29	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2552	RB5: 81 RB6: 103 RB7: 60 RB8: 38392
54	DL_TFC54, DL_TFC111	UL_TFC12, UL_TFC27	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC27	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: No data RB6: No data RB7: No data RB8: 40952
55	DL_TFC55, DL_TFC112	UL_TFC13, UL_TFC28	DL_TFC0, DL_TFC57, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2552	RB5: 39 RB6: No data RB7: No data RB8: 40952

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	under	Under test			(bits)	(1.5.1.5)
	test				(note)	(note)
56	DL_TFC56,	UL_TFC14,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC113	UL_TFC29	DL_TFC57,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC12,	RB7: 60	RB7: 60
			UL_TFC15	UL_TFC14,	RB8: 2552	RB8: 40952
				UL_TFC15,		
				UL TFC17,		
				UL_TFC27,		
				UL_TFC29		
NOTE:	See TS 34.1	09 [10] clause 5	5.3.2.6.2 for deta	ils regarding loopback	of RLC SDUs.	
	RB8: Test da	ata size has bee	n set to DL TFS	size under test minus	8 bits (size of 7	' bit length
	indicator and	d expansion bit).	As the TTI for R	B8 is the same for bo	th downlink and	I uplink then UL
				o return one SDU per		
		t equal to the up		der test minus 8 bits		

14.2.44.2.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: an RLC SDU on RB8 having the content equal to the first 312 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB8 having the content equal to the first 312 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: an RLC SDU on RB8 having the content equal to the first 312 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
 - for sub-test 6: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 7: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 8: an RLC SDU on RB8 having the content equal to the first 632 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
 - for sub-test 9: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 10: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 11: an RLC SDU on RB8 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.

- for sub-test 12: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 13: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 14: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 15: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 16: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 17: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 18: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 19: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 20: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 21: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 22: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 23: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 24: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 25: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 26: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 27: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 28: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 29: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 30: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 31: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.

- for sub-test 32: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 33: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 34: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 35: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 36: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 37: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 38: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 39: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 40: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 41: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 42: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 43: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 44: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 45: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 46: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 47: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 48: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 49: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 50: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 51: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.

- for sub-test 52: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 53: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- for sub-test 54: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; and no data shall be received on RB5, RB6 and RB7.
- for sub-test 55: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
- for sub-test 56: an RLC SDU on RB8 having the content equal to the first 2552 bits of the test data sent by the SS in downlink; an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.45 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Streaming / unknown / UL:57.6 DL:57.6 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.45.1 Conformance requirement

See 14.2.4.1.

14.2.45.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.45.

14.2.45.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (57.6 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x576	0x148
	TF1, bits	1x39	1x103	1x60	1x576	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x576	N/A
	TF3, bits	N/A	N/A	N/A	3x576	N/A
	TF4, bits	N/A	N/A	N/A	4x576	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF2, TF0)
UL_TFC7	(TF1, TF0, TF0, TF2, TF0)
UL_TFC8	(TF2, TF1, TF1, TF2, TF0)
UL_TFC9	(TF0, TF0, TF0, TF3, TF0)
UL_TFC10	(TF1, TF0, TF0, TF3, TF0)
UL_TFC11	(TF2, TF1, TF1, TF3, TF0)
UL_TFC12	(TF0, TF0, TF0, TF4, TF0)
UL_TFC13	(TF1, TF0, TF0, TF4, TF0)
UL_TFC14	(TF2, TF1, TF1, TF4, TF0)
UL_TFC15	(TF0, TF0, TF0, TF1)
UL_TFC16	(TF1, TF0, TF0, TF0, TF1)
UL_TFC17	(TF2, TF1, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF1, TF1, TF1)
UL_TFC21	(TF0, TF0, TF0, TF2, TF1)
UL_TFC22	(TF1, TF0, TF0, TF2, TF1)
UL_TFC23	(TF2, TF1, TF1, TF2, TF1)
UL_TFC24	(TF0, TF0, TF0, TF3, TF1)
UL_TFC25	(TF1, TF0, TF0, TF3, TF1)
UL_TFC26	(TF2, TF1, TF1, TF3, TF1)
UL_TFC27	(TF0, TF0, TF0, TF4, TF1)
UL_TFC28	(TF1, TF0, TF0, TF4, TF1)
UL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (57.6 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x576	0x148
	TF1, bits	1x39	1x103	1x60	1x576	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x576	N/A
	TF3, bits	N/A	N/A	N/A	3x576	N/A
	TF4, bits	N/A	N/A	N/A	4x576	N/A

Downlink TFCS:

TFCI	
DL TFC0	(RB5, RB6, RB7, RB8, DCCH)
	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, DL_TFC16	UL_TFC1, DL_TFC16	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC15, UL_TFC16	RB5: 39 RB6: 103 RB7: 60 RB8: 576	RB5: 39 RB6: No data RB7: No data RB8: No data
2	DL_TFC2, DL_TFC17	UL_TFC2, DL_TFC17	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC15, UL_TFC17	RB5: 81 RB6: 103 RB7: 60 RB8: 576	RB5: 81 RB6: 103 RB7: 60 RB8: No data
3	DL_TFC3, DL_TFC18	UL_TFC3,U L_TFC18	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC3, UL_TFC15, UL_TFC18	RB5: 39 RB6: 103 RB7: 60 RB8: 576	RB5: No data RB6: No data RB7: No data RB8: 576
4	DL_TFC4, DL_TFC19	UL_TFC4, DL_TFC19	DL_TFC0, DL_TFC15, , UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC3 UL_TFC4, UL_TFC15, UL_TFC16, UL_TFC18, UL_TFC19	RB5: 39 RB6: 103 RB7: 60 RB8: 576	RB5: 39 RB6: No data RB7: No data RB8: 576
5	DL_TFC5, DL_TFC20	UL_TFC5, DL_TFC20	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC15, UL_TFC17, UL_TFC18, UL_TFC20	RB5: 81 RB6: 103 RB7: 60 RB8: 576	RB5: 81 RB6: 103 RB7: 60 RB8: 576
6	DL_TFC6, DL_TFC21	UL_TFC6, DL_TFC21	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC6, UL_TFC15, UL_TFC21	RB5: 39 RB6: 103 RB7: 60 RB8: 1152	RB5: No data RB6: No data RB7: No data RB8: 1152
7	DL_TFC7, DL_TFC22	UL_TFC7, DL_TFC22	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7, UL_TFC15, UL_TFC16, UL_TFC21, UL_TFC22	RB5: 39 RB6: 103 RB7: 60 RB8: 1152	RB5: 39 RB6: No data RB7: No data RB8: 1152
8	DL_TFC8, DL_TFC23	UL_TFC8, DL_TFC23	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8, UL_TFC15, UL_TFC17, UL_TFC21, UL_TFC23	RB5: 81 RB6: 103 RB7: 60 RB8: 1152	RB5: 81 RB6: 103 RB7: 60 RB8: 1152
9	DL_TFC9, DL_TFC24	UL_TFC9, DL_TFC24	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC9, UL_TFC15, UL_TFC24	RB5: 39 RB6: 103 RB7: 60 RB8: 1728	RB5: No data RB6: No data RB7: No data RB8: 1728
10	DL_TFC10, DL_TFC25	UL_TFC10, UL_TFC25	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC9, UL_TFC10, UL_TFC15, UL_TFC16, UL_TFC24, UL_TFC25	RB5: 39 RB6: 103 RB7: 60 RB8: 1728	RB5: 39 RB6: No data RB7: No data RB8: 1728

Sub-	Downlink	Uplink	Implicitely	Restricted UL	UL RLC SDU	Test data size
test	TFCS Under	TFCS Under test	tested	TFCIs	size (bits)	(bits)
	Test	onder test			(note)	(note)
11	DL_TFC11, DL_TFC26	UL_TFC11, DL_TFC26	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC9, UL_TFC11, UL_TFC15, UL_TFC17, UL_TFC24, UL_TFC26	RB5: 81 RB6: 103 RB7: 60 RB8: 1728	RB5: 81 RB6: 103 RB7: 60 RB8: 1728
12	DL_TFC12, DL_TFC27	UL_TFC12, DL_TFC27	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC12, UL_TFC15, UL_TFC26	RB5: 39 RB6: 103 RB7: 60 RB8: 2304	RB5: No data RB6: No data RB7: No data RB8: 2304
13	DL_TFC13, DL_TFC28	UL_TFC13, DL_TFC28	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC1, UL_TFC12, UL_TFC13, UL_TFC15, UL_TFC16, UL_TFC27, UL_TFC28	RB5: 39 RB6: 103 RB7: 60 RB8: 2304	RB5: 39 RB6: No data RB7: No data RB8: 2304
14	DL_TFC14, DL_TFC29	UL_TFC14, DL_TFC29	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC15	UL_TFC0, UL_TFC2, UL_TFC12, UL_TFC14, UL_TFC15, UL_TFC17, UL_TFC27, UL_TFC29	RB5: 81 RB6: 103 RB7: 60 RB8: 2304	RB5: 81 RB6: 103 RB7: 60 RB8: 2304
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. As the TTI for RB8 is the same for both downlink and uplink then UL RLC SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size has been set equal to the uplink TFS size under test.					

14.2.45.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified in the actual sub test.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3, 6, 9 and 12: an RLC SDU on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4, 7, 10 and 13: an RLC SDU on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5, 8, 11 and 14: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.46 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Streaming / unknown / UL:0 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.46.1 Conformance requirement

See 14.2.4.1.

14.2.46.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.46.

To be able to test the downlink radio bearer using the UE loopback function for the reference radio bearer UL:0 DL: 64 kbps,, the reference radio bearer configuration according to TS 34.108, clause 6.10.2.4.1.15.1 (Streaming/unknown/UL:14.4 kbps) is used in uplink.

14.2.46.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (14.4 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x576	0x148
TFS	TF1, bits	1x39	1x103	1x60	1x576	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF1, TF0, TF1)
UL_TFC9	(TF0, TF0, TF0, TF1, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1, TF1)
UL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x320	0x148
	TF1, bits	1x39	1x103	1x60	1x320	1x148
TFS	TF2, bits	1x81	N/A	N/A	2x320	N/A
	TF3, bits	N/A	N/A	N/A	4x320	N/A
	TF4, bits	N/A	N/A	N/A	8x320	N/A

Downlink TFCS:

TECI	
TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF2, TF0)
DL_TFC7	(TF1, TF0, TF0, TF2, TF0)
DL_TFC8	(TF2, TF1, TF1, TF2, TF0)
DL_TFC9	(TF0, TF0, TF0, TF3, TF0)
DL_TFC10	(TF1, TF0, TF0, TF3, TF0)
DL_TFC11	(TF2, TF1, TF1, TF3, TF0)
DL_TFC12	(TF0, TF0, TF0, TF4, TF0)
DL_TFC13	(TF1, TF0, TF0, TF4, TF0)
DL_TFC14	(TF2, TF1, TF1, TF4, TF0)
DL_TFC15	(TF0, TF0, TF0, TF0, TF1)
DL_TFC16	(TF1, TF0, TF0, TF0, TF1)
DL_TFC17	(TF2, TF1, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF1, TF1, TF1)
DL_TFC21	(TF0, TF0, TF0, TF2, TF1)
DL_TFC22	(TF1, TF0, TF0, TF2, TF1)
DL_TFC23	(TF2, TF1, TF1, TF2, TF1)
DL_TFC24	(TF0, TF0, TF0, TF3, TF1)
DL_TFC25	(TF1, TF0, TF0, TF3, TF1)
DL_TFC26	(TF2, TF1, TF1, TF3, TF1)
DL_TFC27	(TF0, TF0, TF0, TF4, TF1)
DL_TFC28	(TF1, TF0, TF0, TF4, TF1)
DL_TFC29	(TF2, TF1, TF1, TF4, TF1)

Sub-	Downlink	Uplink	Implicitely	Restricted UL	UL RLC	Test data size
test	TFCS	TFCS	tested	TFCIs	SDU size	(bits)
	Under Test	Under test			(bits) (note 1)	(note 1)
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC16	UL_TFC7	DL_TFC15,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC7	RB8: 576	RB8: No data
2	DL_TFC2, DL_TFC17	UL_TFC2, UL_TFC8	DL_TFC0, DL_TFC15,	UL_TFC0, UL_TFC2,	RB5: 81 RB6: 103	RB5: 81 RB6: 103
	DL_IFCI/	UL_IFC0	UL TFC0,	UL_TFC6,	RB0: 103 RB7: 60	RB7: 60
			UL_TFC6	UL_TFC8	RB8: 576	RB8: No data
3	DL_TFC3,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC18	UL_TFC9	DL_TFC15,	UL_TFC3,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC9	RB8: 576	RB8: 320 (note 2)
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC19	UL_TFC10	DL_TFC15,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC4,	RB8: 576	RB8: 320
				UL_TFC6, UL_TFC7,		(note 2)
				UL_TFC9,		
				UL_TFC10		
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC20	UL_TFC11	DL_TFC15,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0, UL_TFC6	UL_TFC3, UL_TFC5,	RB7: 60 RB8: 576	RB7: 60 RB8: 320
			02_1100	UL_TFC6,	1100.070	(note 2)
				UL_TFC8,		(
				UL_TFC9,		
6		UL_TFC3,		UL_TFC11	RB5: 39	RB5: No data
6	DL_TFC6, DL_TFC21	UL_TFC9	DL_TFC0, DL_TFC15,	UL_TFC0, UL_TFC3,	RB5: 39 RB6: 103	RB5: No data
	0202.	0200	UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC9	RB8: 576	RB8: 640
						(note 3)
7	DL_TFC7,	UL_TFC4,	DL_TFC0,	UL_TFC0, UL_TFC1,	RB5: 39	RB5: 39
	DL_TFC22	UL_TFC10	DL_TFC15, UL_TFC0,	UL_TFC3,	RB6: 103 RB7: 60	RB6: No data RB7: No data
			UL_TFC6	UL_TFC4,	RB8: 576	RB8: 640
				UL_TFC6,		(note 3)
				UL_TFC7,		
				UL_TFC9,		
8	DL_TFC8,	UL_TFC5,	DL_TFC0,	UL_TFC10 UL_TFC0,	RB5: 81	RB5: 81
Ĩ	DL_TFC23	UL_TFC11	DL_TFC15,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: 60
			UL_TFC6	UL_TFC5,	RB8: 576	RB8: 640
				UL_TFC6,		(note 3)
				UL_TFC8, UL_TFC9,		
				UL_TFC11		
9	DL_TFC9,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC24	UL_TFC9	DL_TFC15,	UL_TFC3,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC9	RB8: 576	RB8: 1280 (note 4)
		L	1			

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
10	Test DL_TFC10, DL_TFC25	UL_TFC4, UL_TFC10	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9 UL_TFC10	(note 1) RB5: 39 RB6: 103 RB7: 60 RB8: 576	(note 1) RB5: 39 RB6: No data RB7: No data RB8: 1280 (note 4)
11	DL_TFC11, DL_TFC26	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 576	RB5: 81 RB6: 103 RB7: 60 RB8: 1280 (note 4)
12	DL_TFC12, DL_TFC27	UL_TFC3 UL_TFC9	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 39 RB6: 103 RB7: 60 RB8: 576	RB5: No data RB6: No data RB7: No data RB8: 2560 (note 5)
13	DL_TFC13, DL_TFC28	UL_TFC4, UL_TFC10	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10	RB5: 39 RB6: 103 RB7: 60 RB8: 576	RB5: 39 RB6: No data RB7: No data RB8: 2560 (note 5)
14	DL_TFC14, DL_TFC29	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC15, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 81 RB6: 103 RB7: 60 RB8: 576	RB5: 81 RB6: 103 RB7: 60 RB8: 2560 (note 5)
			ODI	ils regarding loopback		r DI /TE1) LIE will
	return one R	LC PDU. SS cre	eates an UL RLC	ts as test data (=DL F C SDU from the first 3	20 bits of the red	ceived RLC PDU.
NOTE	NOTE 3: RB8: SS is using a DL RLC SDU size of 640 bits as test data (=DL RLC PDU size for DL/TF2). UE will return one RLC PDU. SS creates an UL RLC SDU from the received RLC PDU.					
NOTE	NOTE 4: RB8: SS is using a DL RLC SDU size of 1280 bits as test data (=DL RLC PDU size for DL/TF3). UE					
NOTE	will return one RLC PDU. SS creates an UL RLC SDU from the received RLC PDU. NOTE 5: RB8: SS is using a DL RLC SDU size of 2560 bits as test data (=DL RLC PDU size for DL/TF4). UE					
	will return or TTI for RB8 is t	he RLC PDU. SS he same for bot	S creates an UL h downlink and ι	RLC SDU from the re uplink then UL RLC SI LC SDU size has been	ceived RLC PD DU size has bee	U. en set to achieve

14.2.46.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified in the actual sub test.
- 3. At step 15 the UE shall return
 - for sub-test 3, 6, 9 and 12: no data on RB5, RB6 and RB7.

- for sub-test 1, 4, 7, 10 and 13: an RLC SDU on RB5 having the same content as sent by the SS; and no data shall be received on RB6 or RB7.
- for sub-test 2, 5, 8, 11 and 14: an RLC SDU on each of RB5, RB6 and RB7 having the same content as sent by the SS.
- for sub-test 1 to 2: no data on RB8.
- for sub-test 3 to 5: an RLC SDU on RB8 having the same content as sent by the SS.
- for sub-test 6 to 14: an RLC SDU on RB5 having the same content as the first 576 bits of the RLC SDU sent by the SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.47 Void

- 14.2.48 Void
- 14.2.49 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Conversational / unknown / UL:64 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH
- 14.2.49.1 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI
- 14.2.49.1.1 Conformance requirement
- See 14.2.4.1.
- 14.2.49.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.49 for the 20 ms TTI case.

14.2.49.1.3 Method of test

See 14.1.2 for test procedure.

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB8):

Uplink RL	С		
TM RL	_C		
Tra	ansmission RLC discard		
	CHOICE SDU Discard Mode		
	Timer based no explicit		
	Timer_discard	100ms	
Se	gmentation indication	FALSE	
Downlink	RLC		
TM RL	_C		
Se	gmentation indication	FALSE	
NOTE:	NOTE: Timer based discard without explicit signalling is used in uplink to		
	secure that the UE will be able to return	data for the case when the	
	UE test loop function will not deliver all	the SDUs in one and the	
	same TTI.		

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x640	0x148
TFS	TF1, bits	1x39	1x103	1x60	2x640	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF1, TF0, TF1)
UL_TFC9	(TF0, TF0, TF0, TF1, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1, TF1)
UL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps, 20 ms TTI)	DCCH
	TF0, bits	1x0	0x103	0x60	0x640	0x148
TFS	TF1, bits	1x39	1x103	1x60	2x640	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF1, TF0, TF1)
DL_TFC9	(TF0, TF0, TF0, TF1, TF1)
DL_TFC10	(TF1, TF0, TF0, TF1, TF1)
DL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
lesi	Under	Under test	lesieu	11 013	(bits)	(DILS)
	Test				(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC7	DL_TFC7	DL_TFC6,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC7	RB8: 640	RB8: No data
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC8	DL_TFC8	DL_TFC6,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: 60
			UL_TFC6	UL_TFC8	RB8: 640	RB8: No data
3	DL_TFC3,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC9	DL_TFC9	DL_TFC6,	UL_TFC3,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC9	RB8: 640	RB8: 2x640
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC10	UL_TFC10	DL_TFC6,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC4,	RB8: 640	RB8: 2x640
				UL_TFC6,		
				UL_TFC7,		
				UL_TFC9,		
				UL_TFC10		
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC11	UL_TFC11	DL_TFC6,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: 60
			UL_TFC6	UL_TFC5,	RB8: 640	RB8: 2x640
				UL_TFC6,		
				UL_TFC8,		
				UL_TFC9,		
				UL_TFC11		
NOTE:	See TS 34	.109 [10] claus	se 5.3.2.6.2 for deta	ils regarding loop	back of RLC SDUs	3.

See 14.1.1 for test procedure.

14.2.49.1.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: RLC SDUs on RB5, RB6 and RB7 having the same content as sent by the SS; and no data shall be received on RB8.
 - for sub-test 3: RLC SDUs on RB8 having the same content as sent by the SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: RLC SDUs on RB5 and RB8 having the same content as sent by the SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5: RLC SDUs on RB5, RB6, RB7 and RB8 having the same content as sent by the SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

2229

14.2.49.2 Conversational / speech / UL:12.2 DL:12.2 kbps / CS RAB + Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 40 ms TTI

14.2.49.2.1 Conformance requirement

See 14.2.4.1.

14.2.49.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.49 for the 40 ms TTI case.

14.2.49.2.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB8):

Liplink DL	<u>^</u>	
Uplink RL		
TM RL	.C	
Se	gmentation indication	FALSE
Tra	ansmission RLC discard	
	CHOICE SDU Discard Mode	
	Timer based no explicit	
	Timer_discard	100ms
Downlink	RLC	
TM RL	.C	
Se	gmentation indication	FALSE
NOTE:	Timer based discard without explicit sig	nalling is used in uplink to
	secure that the UE will be able to return	
	UE test loop function will not deliver all	the SDLIs in one and the
	same TTI .	

See 14.1.2 for test procedure.

Uplink TFS:

_	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x640	0x148
TFS	TF1, bits	1x39	1x103	1x60	4x640	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF1, TF0, TF0)
UL_TFC3	(TF0, TF0, TF0, TF1, TF0)
UL_TFC4	(TF1, TF0, TF0, TF1, TF0)
UL_TFC5	(TF2, TF1, TF1, TF1, TF0)
UL_TFC6	(TF0, TF0, TF0, TF0, TF1)
UL_TFC7	(TF1, TF0, TF0, TF0, TF1)
UL_TFC8	(TF2, TF1, TF1, TF0, TF1)
UL_TFC9	(TF0, TF0, TF0, TF1, TF1)
UL_TFC10	(TF1, TF0, TF0, TF1, TF1)
UL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Downlink TFS:

		RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps)	DCCH
	TF0, bits	1x0	0x103	0x60	0x640	0x148
TFS	TF1, bits	1x39	1x103	1x60	4x640	1x148
	TF2, bits	1x81	N/A	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, RB7, RB8, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF1, TF0, TF0)
DL_TFC3	(TF0, TF0, TF0, TF1, TF0)
DL_TFC4	(TF1, TF0, TF0, TF1, TF0)
DL_TFC5	(TF2, TF1, TF1, TF1, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1)
DL_TFC7	(TF1, TF0, TF0, TF0, TF1)
DL_TFC8	(TF2, TF1, TF1, TF0, TF1)
DL_TFC9	(TF0, TF0, TF0, TF1, TF1)
DL_TFC10	(TF1, TF0, TF0, TF1, TF1)
DL_TFC11	(TF2, TF1, TF1, TF1, TF1)

Sub-	Downlink	Uplink	Implicitely tested	Restricted UL	UL RLC SDU	Test data size
test	TFCS	TFCS	implicitory tootou	TFCIs	size	(bits)
	Under	Under test			(bits)	(2.10)
	Test				(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC7	DL_TFC7	DL_TFC6,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC7	RB8: 640	RB8: No data
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC8	DL_TFC8	DL_TFC6,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: 60
			UL_TFC6	UL_TFC8	RB8: 640	RB8: No data
3	DL_TFC3,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: No data
	DL_TFC9	DL_TFC9	DL_TFC6,	UL_TFC3,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC6,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC9	RB8: 640	RB8: 4x640
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC10	DL_TFC10	DL_TFC6,	UL_TFC1,	RB6: 103	RB6: No data
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: No data
			UL_TFC6	UL_TFC4,	RB8: 640	RB8: 4x640
				UL_TFC6,		
				UL_TFC7,		
				UL_TFC9,		
				UL_TFC10		
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC11	DL_TFC11	DL_TFC6,	UL_TFC2,	RB6: 103	RB6: 103
			UL_TFC0,	UL_TFC3,	RB7: 60	RB7: 60
			UL_TFC6	UL_TFC5,	RB8: 640	RB8: 4x640
				UL_TFC6,		
				UL_TFC8,		
				UL_TFC9,		
				UL_TFC11		
NOTE:			se 5.3.2.6.2 for details			
			same for both downli			
		E to return one	SDU per TTI, i.e. the	UL RLC SDU siz	e has been set eq	ual to the uplink
	TB size.					

Release 5

14.2.49.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3: four RLC SDUs on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4: an RLC SDU on RB5 and four RLC SDUs on RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - - for sub-test 5: an RLC SDU on RB5, RB6 and RB7; and four RLC SDUs on RB8 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.49a Conversational / speech / UL:(12.2 7.95 5.9 4.75) DL(12.2 7.95 5.9 4.75) kbps / CS RAB + Conversational / unknown / UL:64 DL:64 kbps / CS RAB+ UL:3.4 DL: 3.4 kbps SRBs for DCCH

14.2.49a.1 Conformance requirement

See 14.2.4.1.

14.2.49a.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.49a.

14.2.49a.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps)	DCCH
	TF0, bits	0x81 (alt. 1x0)	0x103	0x60	0x640	0x148 (alt. 1x0)
TFS	TF1, bits	1x39	1x53	1x60	2x640 (alt. 4x640)	1x148
	TF2, bits	1x42	1x63	N/A	N/A	N/A
	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, RB7, 64 kbps RAB, DCCH)
UL_TFC0	(TF0, TF0, TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0, TF0, TF0)
UL_TFC2	(TF2, TF1, TF0, TF0, TF0)
UL_TFC3	(TF3, TF2, TF0, TF0, TF0)
UL_TFC4	(TF4, TF3, TF0, TF0, TF0)
UL_TFC5	(TF5, TF4, TF1, TF0, TF0)
UL_TFC6	(TF0, TF0, TF0, TF1, TF0)
UL_TFC7	(TF1, TF0, TF0, TF1, TF0)
UL_TFC8	(TF2, TF1, TF0, TF1, TF0)
UL_TFC9	(TF3, TF2, TF0, TF1, TF0)
UL_TFC10	(TF4, TF3, TF0, TF1, TF0)
UL_TFC11	(TF5, TF4, TF1, TF1, TF0)
UL_TFC12	(TF0, TF0, TF0, TF1)
UL_TFC13	(TF1, TF0, TF0, TF0, TF1)
UL_TFC14	(TF2, TF1, TF0, TF0, TF1)
UL_TFC15	(TF3, TF2, TF0, TF0, TF1)
UL_TFC16	(TF4, TF3, TF0, TF0, TF1)
UL_TFC17	(TF5, TF4, TF1, TF0, TF1)
UL_TFC18	(TF0, TF0, TF0, TF1, TF1)
UL_TFC19	(TF1, TF0, TF0, TF1, TF1)
UL_TFC20	(TF2, TF1, TF0, TF1, TF1)
UL_TFC21	(TF3, TF2, TF0, TF1, TF1)
UL_TFC22	(TF4, TF3, TF0, TF1, TF1)
UL_TFC23	(TF5, TF4, TF1, TF1, TF1)

Downlink TFS:

I

	TFI	RB5 (RAB subflow #1)	RB6 (RAB subflow #2)	RB7 (RAB subflow #3)	RB8 (64 kbps)	DCCH
	TF0, bits	<u>1x0</u> 0x81 (alt. 1x0)	0x103	0x60	0x640	0x148 (alt. 1x0)
TFS	TF1, bits	1x39	1x53	1x60	2x640 (alt. 4x640)	1x148
	TF2, bits	1x42	1x63	N/A	N/A	N/A
	TF3, bits	1x55	1x84	N/A	N/A	N/A
	TF4, bits	1x75	1x103	N/A	N/A	N/A
	TF5, bits	1x81	N/A	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB2, RB3, RB4, 64 kbps RAB, DCCH)
DL_TFC0	(TF0, TF0, TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0, TF0, TF0)
DL_TFC2	(TF2, TF1, TF0, TF0, TF0)
DL_TFC3	(TF3, TF2, TF0, TF0, TF0)
DL_TFC4	(TF4, TF3, TF0, TF0, TF0)
DL_TFC5	(TF5, TF4, TF1, TF0, TF0)
DL_TFC6	(TF0, TF0, TF0, TF1, TF0)
DL_TFC7	(TF1, TF0, TF0, TF1, TF0)
DL_TFC8	(TF2, TF1, TF0, TF1, TF0)
DL_TFC9	(TF3, TF2, TF0, TF1, TF0)
DL_TFC10	(TF4, TF3, TF0, TF1, TF0)
DL_TFC11	(TF5, TF4, TF1, TF1, TF0)
DL_TFC12	(TF0, TF0, TF0, TF1)
DL_TFC13	(TF1, TF0, TF0, TF0, TF1)
DL_TFC14	(TF2, TF1, TF0, TF0, TF1)
DL_TFC15	(TF3, TF2, TF0, TF0, TF1)
DL_TFC16	(TF4, TF3, TF0, TF0, TF1)
DL_TFC17	(TF5, TF4, TF1, TF0, TF1)
DL_TFC18	(TF0, TF0, TF0, TF1, TF1)
DL_TFC19	(TF1, TF0, TF0, TF1, TF1)
DL_TFC20	(TF2, TF1, TF0, TF1, TF1)
DL_TFC21	(TF3, TF2, TF0, TF1, TF1)
DL_TFC22	(TF4, TF3, TF0, TF1, TF1)
DL_TFC23	(TF5, TF4, TF1, TF1, TF1)

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under	Under test			(bits)	
	Test				(note)	(note)
1	DL_TFC0,	UL_TFC0,		UL_TFC0,	RB5: 81	RB5: No data
	DL_TFC12	UL_TFC12		UL_TFC12	RB6: 103	RB6: No data
					RB7: 60	RB7: No data
					RB8: 640	RB8: No data
2	DL_TFC1,	UL_TFC1,		UL_TFC0,	RB5: 39	RB5: 39
	DL_TFC13	UL_TFC13		UL_TFC1,	RB6: 103	RB6:No data
				UL_TFC12,	RB7: 60	RB7:No data
				UL_TFC13	RB8: 640	RB8:No data
3	DL_TFC2,	UL_TFC2,		UL_TFC0,	RB5: 42	RB5: 42
	DL_TFC14	UL_TFC14		UL_TFC2,	RB6: 53	RB6: 53
				UL_TFC12,	RB7: 60	RB7:No data
				UL_TFC14	RB8: 640	RB8:No data
4	DL_TFC3,	UL_TFC3,		UL_TFC0,	RB5: 55	RB5: 55
	DL_TFC15	UL_TFC15		UL_TFC3,	RB6: 63	RB6: 63
				UL_TFC12,	RB7: 60	RB7:No data
				UL_TFC15	RB8: 640	RB8:No data
5	DL_TFC4,	UL_TFC4,		UL_TFC0,	RB5: 75	RB5: 75
	DL_TFC16	UL_TFC16		UL_TFC4,	RB6: 84	RB6: 84
				UL_TFC12,	RB7: 60	RB7:No data
				UL_TFC16	RB8: 640	RB8:No data
6	DL_TFC5,	UL_TFC5,		UL_TFC0,	RB5: 81	RB5: 81
	DL_TFC17	UL_TFC17		UL_TFC5,	RB6: 103	RB6: 103
				UL_TFC12,	RB7: 60	RB7: 60
				UL_TFC17	RB8: 640	RB8:No data
7	DL_TFC6,	UL_TFC6,		UL_TFC0,	RB5:81	RB5:No data
	DL_TFC18	UL_TFC18		UL_TFC6,	RB6:103	RB6:No data
				UL_TFC12,	RB7: 60	RB7:No data
				UL_TFC18	RB8: 1280	RB8: 1280 (alt.
					(alt. 2560)	2560)
NOTE:	See TS 34	.109 [10] claus	e 5.3.2.6.2 for details r	egarding loopback	of RLC SDUs.	

Release 5

14.2.49a.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6, RB7 and RB8.
 - for sub-test 2: an RLC SDU on RB5, RB6 and RB7 having the same content as sent by SS; and no data shall be received on RB8.
 - for sub-test 3, 6, 9 and 12: an RLC SDU on RB8 having the same content as sent by SS; and no data shall be received on RB5, RB6 and RB7.
 - for sub-test 4, 7, 10 and 13: an RLC SDU on RB5 and RB8 having the same content as sent by SS; and no data shall be received on RB6 and RB7.
 - for sub-test 5, 8, 11 and 14: an RLC SDU on RB5, RB6, RB7 and RB8 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.50 Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Conversational / unknown / UL:64 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.50.1 Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Conversational / unknown / UL:64 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH / 20 ms TTI
- 14.2.50.1.1 Conformance requirement

See 14.2.4.1.

14.2.50.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.50 for the 20 ms TTI case.

14.2.50.1.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5 and RB6):

	RB5	RB6			
	(64 kbps)	(64 kbps)			
Uplink RLC					
TM RLC					
Segmentation indication	FALSE	FALSE			
Transmission RLC discard					
CHOICE SDU Discard Mode					
Timer based no explicit					
Timer_discard	100ms	100ms			
Downlink RLC					
TM RLC					
Segmentation indication	FALSE	FALSE			
NOTE: Timer based discard without explicit sig	NOTE: Timer based discard without explicit signalling is used in uplink to				
secure that the UE will be able to return data for the case when					
the UE test loop function will not deliver all the SDUs in one and					
the same TTI.					

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (64 kbps)	RB6 (64 kbps)	DCCH
	TF0, bits	0x640	0x640	0x148
TFS	TF1, bits	2x640	2x640	1x148
	TF2, bits	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0)
UL_TFC2	(TF0, TF1, TF0)
UL_TFC3	(TF1, TF1, TF0)
UL_TFC4	(TF0, TF0, TF1)
UL_TFC5	(TF1, TF0, TF1)
UL_TFC6	(TF0, TF1, TF1)
UL_TFC7	(TF1, TF1, TF1)

Downlink TFS:

	TFI	RB5 (64 kbps)	RB6 (64 kbps)	DCCH
	TF0, bits	0x640	0x640	0x148
TFS	TF1, bits	2x640	2x640	1x148
	TF2, bits	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0)
DL_TFC2	(TF0, TF1, TF0)
DL_TFC3	(TF1, TF1, TF0)
DL_TFC4	(TF0, TF0, TF1)
DL_TFC5	(TF1, TF0, TF1)
DL_TFC6	(TF0, TF1, TF1)
DL_TFC7	(TF1, TF1, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, DL_TFC5	UL_TFC1, DL_TFC5	DL_TFC0, DL_TFC4, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC4, UL_TFC5	RB5: 640 RB6: 640	RB5: 2x640 RB6: No data
2	DL_TFC2, DL_TFC6	UL_TFC2, DL_TFC6	DL_TFC0, DL_TFC4, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC2, UL_TFC4, UL_TFC6	RB5: 640 RB6: 640	RB5: No data RB6: 2x640
3	DL_TFC3, DL_TFC7	UL_TFC3, DL_TFC7	DL_TFC0, DL_TFC4, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC6, UL_TFC7	RB5: 640 RB6: 640	RB5: 2x640 RB6: 2x640
	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. As the TTI for RB5 and RB6 is the same for both downlink and uplink then UL RLC SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size has been set equal to the uplink TB size.					

14.2.50.1.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1: two RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: two RLC SDUs on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 3: two RLC SDUs on RB5 and RB6 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

Release 5

14.2.50.2 Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Conversational / unknown / UL:64 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH / 40 ms TTI

14.2.50.2.1 Conformance requirement

See 14.2.4.1.

14.2.50.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.50 for the 40 ms TTI case.

14.2.50.2.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5 and RB6):

	RB5	RB6		
	(64 kbps)	(64 kbps)		
Uplink RLC				
TM RLC				
Segmentation indication	FALSE	FALSE		
Transmission RLC discard				
CHOICE SDU Discard Mode				
Timer based no explicit				
Timer_discard	100ms	100ms		
Downlink RLC				
TM RLC				
Segmentation indication	FALSE	FALSE		
NOTE: Timer based discard without explicit sig	nalling is used	t in uplink to		
secure that the UE will be able to return data for the case when				
the UE test loop function will not deliver all the SDUs in one and				
the same TTI.				

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (64 kbps)	RB6 (64 kbps)	DCCH
	TF0, bits	0x640	0x640	0x148
TFS	TF1, bits	4x640	4x640	1x148
	TF2, bits	N/A	N/A	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0)
UL_TFC2	(TF0, TF1, TF0)
UL_TFC3	(TF1, TF1, TF0)
UL_TFC4	(TF0, TF0, TF1)
UL_TFC5	(TF1, TF0, TF1)
UL_TFC6	(TF0, TF1, TF1)
UL_TFC7	(TF1, TF1, TF1)

Downlink TFS:

	TFI	RB5 (64 kbps)	RB6 (64 kbps)	DCCH
	TF0, bits	0x640	0x640	0x148
TFS	TF1, bits	4x640	4x640	1x148
	TF2, bits	N/A	N/A	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)	
DL_TFC0	(TF0, TF0, TF0)	
DL_TFC1	(TF1, TF0, TF0)	
DL_TFC2	(TF0, TF1, TF0)	
DL_TFC3	(TF1, TF1, TF0)	
DL_TFC4	(TF0, TF0, TF1)	
DL_TFC5	(TF1, TF0, TF1)	
DL_TFC6	(TF0, TF1, TF1)	
DL_TFC7	(TF1, TF1, TF1)	

Sub-tests:

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under	Under test			(bits)	
	Test				(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: 4x640
	DL_TFC5	DL_TFC5	DL_TFC4,	UL_TFC1,	RB6: 640	RB6: No data
			UL_TFC0,	UL_TFC4,		
			UL_TFC4	UL_TFC5		
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data
	DL_TFC6	DL_TFC6	DL_TFC4,	UL_TFC2,	RB6: 640	RB6: 4x640
			UL_TFC0,	UL_TFC4,		
			UL_TFC4	UL_TFC6		
3	DL_TFC3,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: 4x640
	DL_TFC7	DL_TFC7	DL_TFC4,	UL_TFC1,	RB6: 640	RB6: 4x640
			UL_TFC0,	UL_TFC2,		
			UL_TFC4	UL_TFC3,		
				UL_TFC4,		
				UL_TFC5,		
				UL_TFC6,		
				UL_TFC7		
NOTE:			se 5.3.2.6.2 for deta			
As the			ame for both downl			
			ne SDU per TTI, i.e	. the UL RLC SDL	J size has been se	et equal to the
	uplink TB s	size.				

14.2.50.2.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1: four RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: four RLC SDUs on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 3: four RLC SDUs on RB5 and RB6 having the same content as sent by SS.

- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.
- 14.2.51 Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Interactive or background / UL:64 DL:64 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH
- 14.2.51.1 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI + Interactive or background / UL:64 DL:64 kbps / PS RAB
- 14.2.51.1.1 Conformance requirement

See 14.2.4.1.

14.2.51.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.51 for the 20 ms TTI case.

14.2.51.1.3 Method of test

See 14.1.2 for test procedure.

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5):

		RB5 (Conv. 64 kbps)
Uplink RL	C	
TM RI	_C	
Se	gmentation indication	FALSE
Tra	ansmission RLC discard	
	CHOICE SDU Discard Mode	
Timer_discard		100ms
Downlink RLC		
TM RI	TM RLC	
Se	Segmentation indication	
NOTE:	Timer based discard without explicit sig	nalling is used
	in uplink to secure that the UE will be a	ble to return
	data for the case when the UE test loop	function will
	not deliver all the SDUs in one and the	same TTI .

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps, 20 ms TTI)	RB6 (I/B 64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	2x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF0, TF3, TF0)
UL_TFC4	(TF0, TF4, TF0)
UL_TFC5	(TF1, TF0, TF0)
UL_TFC6	(TF1, TF1, TF0)
UL_TFC7	(TF1, TF2, TF0)
UL_TFC8	(TF1, TF3, TF0)
UL_TFC9	(TF1, TF4, TF0)
UL_TFC10	(TF0, TF0, TF1)
UL_TFC11	(TF0, TF1, TF1)
UL_TFC12	(TF0, TF2, TF1)
UL_TFC13	(TF0, TF3, TF1)
UL_TFC14	(TF0, TF4, TF1)
UL_TFC15	(TF1, TF0, TF1)
UL_TFC16	(TF1, TF1, TF1)
UL_TFC17	(TF1, TF2, TF1)
UL_TFC18	(TF1, TF3, TF1)
UL_TFC19	(TF1, TF4, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps, 20 ms TTI)	RB6 (I/B 64 kbps, 20 ms TTI)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	2x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub-	Downlink	Uplink	Implicitely	Restricted UL	UL RLC	Test data size
test	TFCS Under	TFCS Under test	tested	TFCIs	SDU size (bits)	(bits)
	Test		DI 7700		(note)	(note)
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data
	DL_TFC11	UL_TFC11	DL_TFC10, UL_TFC0,	UL_TFC1, UL_TFC10,	RB6: 312	RB6: 312
			UL_TFC10	UL_TFC11		
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data
	DL_TFC12	UL_TFC12	DL_TFC10,	UL_TFC2,	RB6: 632	RB6: 632
	_	_	UL_TFC0,	UL_TFC10,		
			UL_TFC10	UL_TFC12		
3	DL_TFC3,	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data
	DL_TFC13	UL_TFC13	DL_TFC10,	UL_TFC3,	RB6: 952	RB6: 952
			UL_TFC0, UL_TFC10	UL_TFC10, UL_TFC13		
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data
-	DL_TFC14	UL_TFC14	DL_TFC10,	UL_TFC4,	RB6: 1272	RB6: 1272
		· · · · · · · · · · · · · · · · · · ·	UL_TFC0,	UL_TFC10,		
			UL_TFC10	UL_TFC14		
5	DL_TFC5,	UL_TFC5,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: 2x640
	DL_TFC15	UL_TFC15	DL_TFC10,	UL_TFC5,	RB6: 312	RB6: No data
			UL_TFC0,	UL_TFC10,		
6	DL_TFC6,	UL TFC6,	UL_TFC10 DL_TFC0,	UL_TFC15 UL_TFC0,		RB5: 2x640
6	DL_TFC6, DL_TFC16	UL_TFC6, UL_TFC16	DL_TFC10,	UL_TFC1,	RB5: 640 RB6: 312	RB5: 2x640 RB6: 312
	DL_11010	02_11010	UL_TFC0,	UL TFC5,	1100.012	1100.012
			UL_TFC10	UL_TFC6,		
				UL_TFC10,		
				UL_TFC11,		
				UL_TFC15,		
7	DL_TFC7,	UL_TFC7,	DL_TFC0,	UL_TFC16 UL_TFC0,	RB5: 640	RB5: 2x640
'	DL_TFC17	UL_TFC17	DL_TFC10,	UL_TFC2,	RB6: 632	RB6: 632
		· · · · · · · · · · · · · · · · · · ·	UL_TFC0,	UL_TFC5,		
			UL_TFC10	UL_TFC7,		
				UL_TFC10,		
				UL_TFC12,		
				UL_TFC15,		
8	DL_TFC8,	UL_TFC8,	DL_TFC0,	UL_TFC17 UL_TFC0,	RB5: 640	RB5: 2x640
0	DL_TFC8, DL_TFC18	UL_TFC8,	DL_TFC10,	UL_TFC3,	RB5: 040 RB6: 952	RB5: 2X040 RB6: 952
	52	02_11010	UL_TFC0,	UL_TFC5,	1120.002	1120.002
			UL_TFC10	UL_TFC8,		
				UL_TFC10,		
				UL_TFC13,		
				UL_TFC15,		
9				UL_TFC18	RB5: 640	PB5: 2x640
3	DL_TFC9, DL_TFC19	UL_TFC9, UL_TFC19	DL_TFC0, DL_TFC10,	UL_TFC0, UL_TFC4,	RB5: 640 RB6: 1272	RB5: 2x640 RB6: 1272
			UL_TFC0,	UL_TFC5,	120. 1212	1.00.1272
			UL_TFC10	UL_TFC9,		
				UL_TFC10,		
				UL_TFC14,		
				UL_TFC15,		
NOTE		100 [10] alar		UL_TFC19		
NOTE:				ils regarding loopba		
				The UL RLC SDU size		
				under test minus 8 b		
		nd expansion				Ŭ

14.2.51.1.4 Test requirements

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1, 2, 3, 4: RLC SDUs on RB6 having the same content as sent by the SS; and no data shall be received on RB5.
 - for sub-test 5: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6.
 - for sub-test 6, 7, 8 and 9: RLC SDUs on RB5 and RB6 having the same content as sent by the SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.51.2 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 40 ms TTI + Interactive or background / UL:64 DL:64 kbps / PS RAB

14.2.51.2.1 Conformance requirement

See 14.2.4.1.

14.2.51.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.51 for the 40 ms TTI case.

14.2.51.2.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5):

		RB5 (Conv. 64 kbps)
Uplink RLC		
TM RLC Segmentation indication	n	FALSE
Transmission RLC dis		FALSE
CHOICE SDU Dis		
Timer based no	o explicit	
Timer_disc	ard	100ms
Downlink RLC		
TM RLC		
Segmentation indication	on	FALSE
NOTE: Timer based disca	rd without explicit sign	nalling is used
in uplink to secure	that the UE will be ab	ole to return
	when the UE test loop	
not deliver all the	SDUs in one and the s	same TTI .

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 64 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	4x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF0, TF3, TF0)
UL_TFC4	(TF0, TF4, TF0)
UL_TFC5	(TF1, TF0, TF0)
UL_TFC6	(TF1, TF1, TF0)
UL_TFC7	(TF1, TF2, TF0)
UL_TFC8	(TF1, TF3, TF0)
UL_TFC9	(TF1, TF4, TF0)
UL_TFC10	(TF0, TF0, TF1)
UL_TFC11	(TF0, TF1, TF1)
UL_TFC12	(TF0, TF2, TF1)
UL_TFC13	(TF0, TF3, TF1)
UL_TFC14	(TF0, TF4, TF1)
UL_TFC15	(TF1, TF0, TF1)
UL_TFC16	(TF1, TF1, TF1)
UL_TFC17	(TF1, TF2, TF1)
UL_TFC18	(TF1, TF3, TF1)
UL_TFC19	(TF1, TF4, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 64 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	4x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)

TFCI	(RB5, RB6, DCCH)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub-	Downlink	Uplink	Implicitely	Restricted UL	UL RLC SDU	Test data size		
test	TFCS	TFCS	tested	TFCIs	size	(bits)		
	Under	Under test			(bits)	()		
	Test				(note)	(note)		
1	DL_TFC1,	UL_TFC1,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data		
	DL_TFC11	DL_TFC11	DL_TFC10,	UL_TFC1,	RB6: 312	RB6: 312		
			UL_TFC0, UL_TFC10	UL_TFC10, UL_TFC11				
2	DL_TFC2,	UL_TFC2,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data		
_	DL_TFC12	DL_TFC12	DL_TFC10,	UL_TFC2,	RB6: 632	RB6: 632		
			UL_TFC0,	UL_TFC10,				
-			UL_TFC10	UL_TFC12				
3	DL_TFC3, DL_TFC13	UL_TFC3,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data		
	DL_IFCI3	DL_TFC13	DL_TFC10, UL_TFC0,	UL_TFC3, UL_TFC10,	RB6: 952	RB6: 952		
			UL_TFC10	UL_TFC13				
4	DL_TFC4,	UL_TFC4,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: No data		
	DL_TFC14	DL_TFC14	DL_TFC10,	UL_TFC4,	RB6: 1272	RB6: 1272		
			UL_TFC0,	UL_TFC10,				
5		UL_TFC5,	UL_TFC10	UL_TFC14 UL_TFC0,	DD5: 640	RB5: 4x640		
5	DL_TFC5, DL_TFC15	DL_TFC5, DL_TFC15	DL_TFC0, DL_TFC10,	UL_TFC0,	RB5: 640 RB6: 312	RB5: 4x640 RB6: No data		
	52_11015	52_11015	UL_TFC0,	UL TFC5,		1.50. NO Gala		
			UL_TFC10	UL_TFC5,				
				UL_TFC10,				
				UL_TFC11,				
				UL_TFC15, UL_TFC15				
6	DL_TFC6,	UL_TFC6,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: 4x640		
U	DL_TFC16	DL_TFC16	DL_TFC10,	UL_TFC6,	RB6: 312	RB6: 312		
	—	_	UL_TFC0,	UL_TFC10,				
			UL_TFC10	UL_TFC16				
7	DL_TFC7,	UL_TFC7,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: 4x640		
	DL_TFC17	DL_TFC17	DL_TFC10, UL_TFC0,	UL_TFC2, UL_TFC5,	RB6: 632	RB6: 632		
			UL_TFC10	UL_TFC7,				
				UL_TFC10,				
				UL_TFC12,				
				UL_TFC15,				
8	DL_TFC8,	UL_TFC8,	DL_TFC0,	UL_TFC17 UL_TFC0,	RB5: 640	RB5: 4x640		
0		DL_TFC8, DL_TFC18	DL_TFC0, DL_TFC10,	UL_TFC3,	RB5: 040 RB6: 952	RB6: 952		
			UL_TFC0,	UL_TFC5,				
			UL_TFC10	UL_TFC8,				
				UL_TFC10,				
				UL_TFC13, UL_TFC15,				
				UL_TFC18				
9	DL_TFC9,	UL_TFC9,	DL_TFC0,	UL_TFC0,	RB5: 640	RB5: 4x640		
	DL_TFC19	DL_TFC19	DL_TFC10,	UL_TFC4,	RB6: 1272	RB6: 1272		
			UL_TFC0,	UL_TFC5,				
			UL_TFC10	UL_TFC9,				
				UL_TFC10, UL_TFC14,				
				UL_TFC15,				
				UL_TFC19				
NOTE:				ils regarding loopba				
RB6: T				r test minus 8 bits (s				
				is the same for both				
	SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size for RB6 has been set equal to the uplink TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit).and .the UL RLC SDU size for RB5 has been set equal to the uplink TB size .							

14.2.51.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 3, 4: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 5: two RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 6, 7, 8 and 9: two RLC SDUs on RB5 and one RLC SDU on RB6 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.51a Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Interactive or Background / UL:8 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

- 14.2.51a.1 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI + Interactive or background / UL:8 DL:8 kbps / PS RAB
- 14.2.51a.1.1 Conformance requirement

See 14.2.4.1.

14.2.51a.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.51a for the 20 ms TTI case.

14.2.51a.1.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 8 kbps)	DCCH
TFS	TF0, bits	0x640	0x336	0x148
15	TF1, bits	2x640	1x336	1x148

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF1, TF0, TF0)
UL_TFC3	(TF1, TF1, TF0)
UL_TFC4	(TF0, TF0, TF1)
UL_TFC5	(TF0, TF1, TF1)
UL_TFC6	(TF1, TF0, TF1)
UL_TFC7	(TF1, TF1, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 8kbps)	DCCH
TFS	TF0, bits	0x640	0x336	0x148
113	TF1, bits	2x640	1x336	1x148

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF1, TF0, TF0)
DL_TFC3	(TF1, TF1, TF0)
DL_TFC4	(TF0, TF0, TF1)
DL_TFC5	(TF0, TF1, TF1)
DL_TFC6	(TF1, TF0, TF1)
DL_TFC7	(TF1, TF1, TF1)

Sub-tests:

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) Note 1	Test data size (bits) Note 1
1	DL_TFC1 DL_TFC5	UL_TFC1 UL_TFC5	DL_TFC0, DL_TFC4, UL_TFC0, UL_TFC4,	UL_TFC0, UL_TFC1, UL_TFC4, UL_TFC5	RB5: 1280 RB6: 312	RB5: No data RB6: 312
2	DL_TFC2 DL_TFC6	UL_TFC2 UL_TFC6	DL_TFC0, DL_TFC4, UL_TFC0, UL_TFC4,	UL_TFC0, UL_TFC2, UL_TFC4, UL_TFC6	RB5: 1280 RB6: 312	RB5: 1280 RB6: No data
3	DL_TFC3 DL_TFC7	UL_TFC3 UL_TFC7	DL_TFC0, DL_TFC4, UL_TFC0, UL_TFC4,	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC6, UL_TFC7	RB5: 1280 RB6: 312	RB5: 1280 RB6: 312
			.2.6.2 for details regardir	ig loopback of RLC S		star and

RB6: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the TTI for RB8 is the same for both downlink and uplink then UL RLC SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size has been set equal to the uplink TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit).

14.2.51a.1.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 2: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 3: an RLC SDU on RB5 and RB6 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.51a.2 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 40 ms TTI + Interactive or background / UL:8 DL:8 kbps / PS RAB

14.2.51a.2.1 Conformance requirement

See 14.2.4.1.

14.2.51a.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.51a for the 40 ms TTI case.

14.2.51a.2.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 8 kbps)	DCCH
TFS	TF0, bits	0x640	0x336	0x148
115	TF1, bits	4x640	1x336	1x148

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)	
UL_TFC0	(TF0, TF0, TF0)	
UL_TFC1	(TF0, TF1, TF0)	
UL_TFC2	(TF1, TF0, TF0)	
UL_TFC3	(TF1, TF1, TF0)	
UL_TFC4	(TF0, TF0, TF1)	
UL_TFC5	(TF0, TF1, TF1)	
UL_TFC6	(TF1, TF0, TF1)	
UL_TFC7	(TF1, TF1, TF1)	

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 64 kbps)	DCCH
TFS	TF0, bits	0x640	0x336	0x148
115	TF1, bits	4x640	1x336	1x148

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF1, TF0, TF0)
DL_TFC3	(TF1, TF1, TF0)
DL_TFC4	(TF0, TF0, TF1)
DL_TFC5	(TF0, TF1, TF1)
DL_TFC6	(TF1, TF0, TF1)
DL_TFC7	(TF1, TF1, TF1)

Sub-tests:

DL_TFC5, UL_TFC5 UL_TFC0, UL_TFC4, UL_TFC1, RB6: 312 RB6: 312 2 DL_TFC2 UL_TFC2 DL_TFC0, DL_TFC4, UL_TFC0, RB5: 2560 RB5: 2560	Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) Note 1	Test data size (bits) Note 1
DL_TFC5, UL_TFC6 UL_TFC0, UL_TFC4, UL_TFC2, RB6: 312 RB6: No data 3 DL_TFC3, UL_TFC3 DL_TFC0, DL_TFC4, UL_TFC0, UL_TFC0, RB5: 2560 RB5: 2560 RB6: 312 RB6: 312	1		_	_ / _ /	UL_TFC1, UL_TFC4,		RB5: No data RB6: 312
DL_TFC5, UL_TFC7 UL_TFC0, UL_TFC4, UL_TFC1, RB6: 312 RB6: 312 UL_TFC3, UL_TFC4, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC5, UL_TFC6, UL_TFC6,	2	_	_	_ / _ /	UL_TFC2, UL_TFC4,		RB5: 2560 RB6: No data
	3	_	_		UL_TFC1, UL_TFC2, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC6,		

expansion bit). As the TTI for RB8 is the same for both downlink and uplink then UL RLC SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size has been set equal to the uplink TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit).

14.2.51a.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 2: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 3: an RLC SDU on RB5 and RB6 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.51b Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Interactive or Background / UL:16 DL:64 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

- 14.2.51b.1 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI + Interactive or background / UL:16 DL:64 kbps / PS RAB
- 14.2.51b.1.1 Conformance requirement

See 14.2.4.1.

14.2.51b.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.51b for the 20 ms TTI case.

14.2.51b.1.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 16 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
TFS	TF1, bits	2x640	1x336	1x148
	TF2, bits	N/A	2x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF1, TF0, TF0)
UL_TFC4	(TF1, TF1, TF0)
UL_TFC5	(TF1, TF2, TF0)
UL_TFC6	(TF0, TF0, TF1)
UL_TFC7	(TF0, TF1, TF1)
UL_TFC8	(TF0, TF2, TF1)
UL_TFC9	(TF1, TF0, TF1)
UL_TFC10	(TF1, TF1, TF1)
UL_TFC11	(TF1, TF2, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 64 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	2x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)					
	Under Test	Under test			(bits) Note 1	Note 1					
1	DL_TFC1, DL_TFC11	UL_TFC1, UL_TFC7	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC6, UL_TFC7	RB5: 1280 RB6: 312	RB5: No data RB6: 312					
2	DL_TFC2, DL_TFC12	UL_TFC2 ,UL_TFC8	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 1280 RB6: 632	RB5: No data RB6: 632					
3	DL_TFC3, DL_TFC13	UL_TFC2, UL_TFC8	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 1280 RB6: 632	RB5: No data RB6: 952					
4	DL_TFC4, DL_TFC14	UL_TFC2 ,UL_TFC8	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC6, UL_TFC8	RB5: 1280 RB6: 632	RB5: No data RB6: 1272					
5	DL_TFC5, DL_TFC15	UL_TFC3 ,UL_TFC9	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC3, UL_TFC6, UL_TFC9	RB5: 1280 RB6: 312	RB5: 1280 RB6: No data					
6	DL_TFC6, DL_TFC16	UL_TFC4, UL_TFC10	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC1, UL_TFC3, UL_TFC4, UL_TFC6, UL_TFC7, UL_TFC9, UL_TFC10	RB5: 1280 RB6: 312	RB5: 1280 RB6: 312					
7	DL_TFC7, DL_TFC17	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 1280 RB6: 632	RB5: 1280 RB6: 632					
8	DL_TFC8, DL_TFC18	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 1280 RB6: 632	RB5: 1280 RB6: 952					
9	DL_TFC9, DL_TFC19	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2, UL_TFC3, UL_TFC5, UL_TFC6, UL_TFC8, UL_TFC9, UL_TFC11	RB5: 1280 RB6: 632	RB5: 1280 RB6: 1272					
	est data size h As the TTI achieve UI size under	has been set to for RB5 and F E to return one test minus 8 b	DL TFS size under test minus 8 bits RB6 is the same for both downlink an SDU per TTI, i.e. the UL RLC SDU s	(size of 7 bit lengt d uplink then UL R size for RB6 has be	UL_TFC11 NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB6: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the TTI for RB5 and RB6 is the same for both downlink and uplink then UL RLC SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size for RB6 has been set equal to the uplink TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit).and .the UL RLC SDU size for RB5						

has been set equal to the uplink TB size .

14.2.51b.1.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1, 2: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 3, 4: an RLC SDU on RB5 having the same content as sent by SS and on RB6 having the content equal to the first 632 bits of the test data sent by the SS in downlink
 - for sub-test 5: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 6, 7 : an RLC SDU on RB5 and RB6 having the same content as sent by SS
 - for sub-test 8, 9: on RB5 an RLC SDU having the same content as sent by SS and RB6 having the content equal to the first 632 bits of the test data sent by the SS in downlink
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.51b.2 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI + Interactive or background / UL:16 DL:64 kbps / PS RAB

14.2.51b.2.1 Conformance requirement

See 14.2.4.1.

14.2.51b.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.51b for the 40 ms TTI case.

14.2.51b.2.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 16 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
TFS	TF1, bits	4x640	1x336	1x148
	TF2, bits	N/A	2x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF1, TF0, TF0)
UL_TFC4	(TF1, TF1, TF0)
UL_TFC5	(TF1, TF2, TF0)
UL_TFC6	(TF0, TF0, TF1)
UL_TFC7	(TF0, TF1, TF1)
UL_TFC8	(TF0, TF2, TF1)
UL_TFC9	(TF1, TF0, TF1)
UL_TFC10	(TF1, TF1, TF1)
UL_TFC11	(TF1, TF2, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 64 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	4x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
1031	Under Test	Under test			(bits) Note 1	Note 1
1	DL_TFC1,	UL_TFC1,	DL_TFC0, DL_TFC10,	UL_TFC0,	RB5: 2560	RB5: No data
	DL_TFC11	UL_TFC7	UL_TFC0, UL_TFC6	UL_TFC1,	RB6: 312	RB6: 312
				UL_TFC6, UL_TFC7		
2	DL_TFC2,	UL_TFC2,	DL_TFC0, DL_TFC10,	UL_TFC0,	RB5: 2560	RB5: No data
	DL_TFC12	UL_TFC8	UL_TFC0, UL_TFC6	UL_TFC2,	RB6: 632	RB6: 632
				UL_TFC6, UL_TFC8		
3	DL_TFC3,	UL_TFC2,	DL_TFC0, DL_TFC10,	UL_TFC0,	RB5: 2560	RB5: No data
	DL_TFC13	UL_TFC8	UL_TFC0, UL_TFC6	UL_TFC2, UL_TFC6,	RB6: 632	RB6: 952
				UL_TFC8		
4	DL_TFC4,	UL_TFC2,	DL_TFC0, DL_TFC10,	UL_TFC0,	RB5: 2560	RB5: No data
	DL_TFC14	UL_TFC8	UL_TFC0, UL_TFC6	UL_TFC2, UL_TFC6,	RB6: 632	RB6: 1272
				UL_TFC8		
5	DL_TFC5,	UL_TFC3,	DL_TFC0, DL_TFC10,	UL_TFC0,	RB5: 2560	RB5: 2560
	DL_TFC15	UL_TFC9	UL_TFC0, UL_TFC6	UL_TFC3, UL_TFC6,	RB6: 312	RB6: No data
				UL_TFC9		
6	DL_TFC6,	UL_TFC4,	DL_TFC0, DL_TFC10,	UL_TFC0,	RB5: 2560	RB5: 2560
	DL_TFC16	UL_TFC10	UL_TFC0, UL_TFC6	UL_TFC1, UL_TFC3,	RB6: 312	RB6: 312
				UL_TFC4,		
				UL_TFC6,		
				UL_TFC7, UL_TFC9,		
				UL_TFC10		
7	DL_TFC7, DL_TFC17	UL_TFC5, UL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC6	UL_TFC0, UL_TFC2,	RB5: 2560 RB6: 632	RB5: 2560 RB6: 632
		02_11011		UL_TFC3,	1100. 002	1100.002
				UL_TFC5,		
				UL_TFC6, UL_TFC8,		
				UL_TFC9,		
8	DL_TFC8,	UL_TFC5,	DL_TFC0, DL_TFC10,	UL_TFC11 UL_TFC0,	RB5: 2560	RB5: 2560
0	DL_TFC8, DL_TFC18	UL_TFC11	UL_TFC0, UL_TFC6	UL_TFC0,	RB5: 2300 RB6: 632	RB6: 952
				UL_TFC3,		
				UL_TFC5, UL_TFC6,		
				UL_TFC8,		
				UL_TFC9,		
9	DL_TFC9,	UL_TFC5,	DL_TFC0, DL_TFC10,	UL_TFC11 UL_TFC0,	RB5: 2560	RB5: 2560
-	DL_TFC19	UL_TFC11	UL_TFC0, UL_TFC6	UL_TFC2,	RB6: 632	RB6: 1272
				UL_TFC3, UL_TFC5,		
				UL_TFC5,		
				UL_TFC8,		
				UL_TFC9, UL_TFC11		
NOTE:			se 5.3.2.6.2 for details regarding loop	back of RLC SDU		1
RB6:			set to DL TFS size under test minus			
			nd RB6 is the same for both downlir rn one SDU per TTI, i.e. the UL RLC			
	uplink TFS	size under te	st minus 8 bits (size of 7 bit length inc			
	size for RB5 has been set equal to the uplink TB size .					

14.2.51b.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1, 2: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 3, 4: an RLC SDU on RB5 having the same content as sent by SS and on RB6 having the content equal to the first 632 bits of the test data sent by the SS in downlink
 - for sub-test 5: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 6, 7 : an RLC SDU on RB5 and RB6 having the same content as sent by SS
 - for sub-test 8, 9: on RB5 an RLC SDU having the same content as sent by SS and RB6 having the content equal to the first 632 bits of the test data sent by the SS in downlink
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.52 Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Interactive or background / UL:64 DL:128 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.52.1 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI + Interactive or background / UL:64 DL:128 kbps / PS RAB
- 14.2.52.1.1 Conformance requirement
- See 14.2.4.1.

14.2.52.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.52 for the 20 ms TTI case.

14.2.52.1.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5):

		RB5 (Conv. 64 kbps)		
Uplink RL	С			
TM RI	_C			
Se	gmentation indication	FALSE		
Tra	ansmission RLC discard			
	CHOICE SDU Discard Mode			
	Timer based no explicit			
	Timer_discard	100ms		
Downlink	RLC			
TM RI	_C			
Se	gmentation indication	FALSE		
NOTE:	Timer based discard without explicit sig			
in uplink to secure that the UE will be able to return				
data for the case when the UE test loop function will				
	not deliver all the SDUs in one and the	same III.		

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 64 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	2x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF0, TF3, TF0)
UL_TFC4	(TF0, TF4, TF0)
UL_TFC5	(TF1, TF0, TF0)
UL_TFC6	(TF1, TF1, TF0)
UL_TFC7	(TF1, TF2, TF0)
UL_TFC8	(TF1, TF3, TF0)
UL_TFC9	(TF1, TF4, TF0)
UL_TFC10	(TF0, TF0, TF1)
UL_TFC11	(TF0, TF1, TF1)
UL_TFC12	(TF0, TF2, TF1)
UL_TFC13	(TF0, TF3, TF1)
UL_TFC14	(TF0, TF4, TF1)
UL_TFC15	(TF1, TF0, TF1)
UL_TFC16	(TF1, TF1, TF1)
UL_TFC17	(TF1, TF2, TF1)
UL_TFC18	(TF1, TF3, TF1)
UL_TFC19	(TF1, TF4, TF1)

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 128 kbps)	DCCH
TFS	TF0, bits	0x640	0x336	0x148
	TF1, bits	2x640	1x336	1x148
	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	4x336	N/A
	TF4, bits	N/A	8x336	N/A

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test				(note)	(note)
1	DL_TFC1, DL_TFC11	UL_TFC1, DL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC10, UL_TFC11	RB5: 640 RB6: 312	RB5: No data RB6: 312
2	DL_TFC2, DL_TFC12	UL_TFC2, DL_TFC12	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC10, UL_TFC12	RB5: 640 RB6: 632	RB5: No data RB6: 632
3	DL_TFC3, DL_TFC13	UL_TFC3, DL_TFC13	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC10, UL_TFC13	RB5: 640 RB6: 952	RB5: No data RB6: 1272
4	DL_TFC4, DL_TFC14	UL_TFC4, DL_TFC14	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC10, UL_TFC10, UL_TFC14	RB5: 640 RB6: 1272	RB5: No data RB6: 2552
5	DL_TFC5, DL_TFC15	UL_TFC5, DL_TFC15	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 640 RB6: 312	RB5: 2x640 RB6: No data
6	DL_TFC6, DL_TFC16	UL_TFC6, DL_TFC16	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC15,	RB5: 640 RB6: 312	RB5: 2x640 RB6: 312
7	DL_TFC7, DL_TFC17	UL_TFC7, DL_TFC17	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC17	RB5: 640 RB6: 632	RB5: 2x640 RB6: 632
8	DL_TFC8, DL_TFC18	UL_TFC8, DL_TFC18	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC15,	RB5: 640 RB6: 952	RB5: 2x640 RB6: 1272
9	DL_TFC9, DL_TFC19	UL_TFC9, DL_TFC19	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC15,	RB5: 640 RB6: 1272	RB5: 2x640 RB6: 2552
NOTE: RB6: T	est data size f expansion RLC SDU for RB6 ha	has been set to bit). As the TT size has been as been set eq	DL TFS size und I for RB5 and RB set to achieve UE ual to the uplink T	tails regarding loo ler test minus 8 bit 6 is the same for to return one SD FS size under test	s (size of 7 bit len both downlink and U per TTI, i.e. the minus 8 bits (size	gth indicator and I uplink then UL UL RLC SDU size

14.2.52.1.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 3, 4: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 5: two RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 6: two RLC SDUs on RB5 and one RLC SDU on RB6 having the same content as sent by SS.
 - For sub-test 3: RLC SDU on RB6 having the content equal to the first 952 bits of the test data sent by the SS in downlink;
 - For sub-test 4: RLC SDU on RB6 having the content equal to the first 1272 bits of the test data sent by the SS in downlink;
 - for sub-test 8: an RLC SDU on RB6 having the content equal to the first 952 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS.
 - for sub-test 9: an RLC SDU on RB6 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.52.2 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 40 ms TTI + Interactive or background / UL:64 DL:128 kbps / PS RAB

- 14.2.52.2.1 Conformance requirement
- See 14.2.4.1.

14.2.52.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.51 for the 40 ms TTI case.

14.2.52.2.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5):

		RB5 (Conv. 64 kbps)			
Uplink RL	С				
TM RI	_C				
Se	gmentation indication	FALSE			
Tra	ansmission RLC discard				
	CHOICE SDU Discard Mode				
	Timer based no explicit				
	Timer_discard				
Downlink	RLC				
TM RI	_C				
Se	gmentation indication	FALSE			
NOTE:					
in uplink to secure that the UE will be able to return					
data for the case when the UE test loop function will					
	not deliver all the SDUs in one and the	same III.			

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 64 kbps)	DCCH
TFS	TF0, bits	0x640	0x336	0x148
	TF1, bits	4x640	1x336	1x148
	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	3x336	N/A
	TF4, bits	N/A	4x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF0, TF3, TF0)
UL_TFC4	(TF0, TF4, TF0)
UL_TFC5	(TF1, TF0, TF0)
UL_TFC6	(TF1, TF1, TF0)
UL_TFC7	(TF1, TF2, TF0)
UL_TFC8	(TF1, TF3, TF0)
UL_TFC9	(TF1, TF4, TF0)
UL_TFC10	(TF0, TF0, TF1)
UL_TFC11	(TF0, TF1, TF1)
UL_TFC12	(TF0, TF2, TF1)
UL_TFC13	(TF0, TF3, TF1)
UL_TFC14	(TF0, TF4, TF1)
UL_TFC15	(TF1, TF0, TF1)
UL_TFC16	(TF1, TF1, TF1)
UL_TFC17	(TF1, TF2, TF1)
UL_TFC18	(TF1, TF3, TF1)
UL_TFC19	(TF1, TF4, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 128 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	4x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	4x336	N/A
	TF4, bits	N/A	8x336	N/A

Downlink TFCS:

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits)
1	DL_TFC1, DL_TFC11	UL_TFC1, DL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC10, UL_TFC11	RB5: 640 RB6: 312	RB5: No data RB6: 312
2	DL_TFC2, DL_TFC12	UL_TFC2, DL_TFC12	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC10, UL_TFC12	RB5: 640 RB6: 632	RB5: No data RB6: 632
3	DL_TFC3, DL_TFC13	UL_TFC3, DL_TFC13	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC10, UL_TFC13	RB5: 640 RB6: 952	RB5: No data RB6: 1272
4	DL_TFC4, DL_TFC14	UL_TFC4, DL_TFC14	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC10, UL_TFC14	RB5: 640 RB6: 1272	RB5: No data RB6: 2552
5	DL_TFC5, DL_TFC15	UL_TFC5, DL_TFC15	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 640 RB6: 312	RB5: 4x640 RB6: No data

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under	Under test			(bits)	
	Test				(note)	(note)
6	DL_TFC6, DL_TFC16	UL_TFC6, DL_TFC16	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC16	RB5: 640 RB6: 312	RB5: 4x640 RB6: 312
7	DL_TFC7, DL_TFC17	UL_TFC7, DL_TFC17	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC17	RB5: 640 RB6: 632	RB5: 4x640 RB6: 632
8	DL_TFC8, DL_TFC18	UL_TFC8, DL_TFC18	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC18	RB5: 640 RB6: 952	RB5: 4x640 RB6: 1272
9	DL_TFC9, DL_TFC19	UL_TFC9, DL_TFC19	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC15, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC19	RB5: 640 RB6: 1272	RB5: 4x640 RB6: 2552
	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB6: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the TTI for RB5 and RB6 is the same for both downlink and uplink then UL RLC SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size for RB6 has been set equal to the uplink TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit).and .the UL RLC SDU size for RB5 has been set equal to the uplink TB size .					

14.2.52.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 3, 4: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 5: four RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 6: four RLC SDUs on RB5 and one RLC SDU on RB6 having the same content as sent by SS.
 - For sub-test 3: RLC SDU on RB6 having the content equal to the first 652 bits of the test data sent by the SS in downlink;

- For sub-test 4: RLC SDU on RB6 having the content equal to the first 1272 bits of the test data sent by the SS in downlink;
- for sub-test 8: an RLC SDU on RB6 having the content equal to the first 652 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS.
- for sub-test 9: an RLC SDU on RB6 having the content equal to the first 1272 bits of the test data sent by the SS in downlink; an RLC SDU on RB5 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.53 Conversational / unknown / UL:64 DL:64 kbps / CS RAB + Interactive or background / UL:128 DL:128 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

- 14.2.53.1 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 20 ms TTI + Interactive or background / UL:128 DL:128 kbps / PS RAB
- 14.2.53.1.1 Conformance requirement

See 14.2.4.1.

14.2.53.1.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.53 for the 20 ms TTI case.

14.2.53.1.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5):

	RB5 (Conv. 64 kbps)				
Uplink RLC					
TM RLC					
Segmentation indication	FALSE				
Transmission RLC discard					
CHOICE SDU Discard Mode					
Timer based no explicit					
Timer_discard	100ms				
Downlink RLC					
TM RLC					
Segmentation indication	FALSE				
NOTE: Timer based discard without explicit signalling is used					
in uplink to secure that the UE will be able to return					
data for the case when the UE test loop function will					
not deliver all the SDUs in one and the	same TTI .				

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 128 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
TFS	TF1, bits	2x640	1x336	1x148
	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	4x336	N/A
	TF4, bits	N/A	8x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF0, TF3, TF0)
UL_TFC4	(TF0, TF4, TF0)
UL_TFC5	(TF1, TF0, TF0)
UL_TFC6	(TF1, TF1, TF0)
UL_TFC7	(TF1, TF2, TF0)
UL_TFC8	(TF1, TF3, TF0)
UL_TFC9	(TF1, TF4, TF0)
UL_TFC10	(TF0, TF0, TF1)
UL_TFC11	(TF0, TF1, TF1)
UL_TFC12	(TF0, TF2, TF1)
UL_TFC13	(TF0, TF3, TF1)
UL_TFC14	(TF0, TF4, TF1)
UL_TFC15	(TF1, TF0, TF1)
UL_TFC16	(TF1, TF1, TF1)
UL_TFC17	(TF1, TF2, TF1)
UL_TFC18	(TF1, TF3, TF1)
UL_TFC19	(TF1, TF4, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 128 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	2x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	4x336	N/A
	TF4, bits	N/A	8x336	N/A

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)

TFCI	(RB5, RB6, DCCH)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test				(note)	(note)
1	DL_TFC1, DL_TFC11	UL_TFC1, DL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC10, UL_TFC11	RB5: 640 RB6: 312	RB5: No data RB6: 312
2	DL_TFC2, DL_TFC12	UL_TFC2, DL_TFC12	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC10, UL_TFC12	RB5: 640 RB6: 632	RB5: No data RB6: 632
3	DL_TFC3, DL_TFC13	UL_TFC3, DL_TFC13	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC10, UL_TFC13	RB5: 640 RB6: 1272	RB5: No data RB6: 1272
4	DL_TFC4, DL_TFC14	UL_TFC4, DL_TFC14	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC10, UL_TFC14	RB5: 640 RB6: 2552	RB5: No data RB6: 2552
5	DL_TFC5, DL_TFC15	UL_TFC5, DL_TFC15	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 640 RB6: 312	RB5: 2x640 RB6: No data
6	DL_TFC6, DL_TFC16	UL_TFC6, DL_TFC16	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC16	RB5: 640 RB6: 312	RB5: 2x640 RB6: 312
7	DL_TFC7, DL_TFC17	UL_TFC7, DL_TFC17	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC17	RB5: 640 RB6: 632	RB5: 2x640 RB6: 632
8	DL_TFC8, DL_TFC18	UL_TFC8, DL_TFC18	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC18	RB5: 640 RB6: 1272	RB5: 2x640 RB6: 1272
9	DL_TFC9, DL_TFC19	UL_TFC9, DL_TFC19	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC19	RB5: 640 RB6: 2552	RB5: 2x640 RB6: 2552
NOTE: RB6: T	est data size h and expan then UL R RLC SDU (size of 7 b	has been set to sion bit). As th LC SDU size h size for RB6 h bit length indica	se 5.3.2.6.2 for deta DL TFS size unde e TTI for RB5 and l as been set to achi as been set equal t ator and expansion TS size under test.	r test minus 8 bit RB6 is the same eve UE to return o the uplink TFS	s (size of 7 bit le for both downli one SDU per T size under test	ength indicator nk and uplink TI, i.e. the UL minus 8 bits

14.2.53.1.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 3, 4: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 5: two RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 6, 7, 8 and 9: two RLC SDUs on RB5 and one RLC SDU on RB6 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.53.2 Conversational / unknown / UL:64 DL:64 kbps / CS RAB / 40 ms TTI + Interactive or background / UL:128 DL:128 kbps / PS RAB

14.2.53.2.1 Conformance requirement

See 14.2.4.1.

14.2.53.2.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.53 for the 40 ms TTI case.

14.2.53.2.3 Method of test

Initial Conditions

The following RLC Info parameter values shall be set by the SS for the Conversational / unknown / UL:64 DL:64 kbps / CS RAB (RB5):

	RB5 (Conv. 64 kbps)			
Uplink RLC				
TM RLC				
Segmentation indication	FALSE			
Transmission RLC discard				
CHOICE SDU Discard Mode				
Timer based no explicit				
Timer_discard	100ms			
Downlink RLC				
TM RLC				
Segmentation indication	FALSE			
NOTE: Timer based discard without explicit sig	nalling is used			
in uplink to secure that the UE will be able to return				
data for the case when the UE test loop function will				
not deliver all the SDUs in one and the	same TTI .			

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 128 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	4x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	4x336	N/A
	TF4, bits	N/A	8x336	N/A

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF0, TF1, TF0)
UL_TFC2	(TF0, TF2, TF0)
UL_TFC3	(TF0, TF3, TF0)
UL_TFC4	(TF0, TF4, TF0)
UL_TFC5	(TF1, TF0, TF0)
UL_TFC6	(TF1, TF1, TF0)
UL_TFC7	(TF1, TF2, TF0)
UL_TFC8	(TF1, TF3, TF0)
UL_TFC9	(TF1, TF4, TF0)
UL_TFC10	(TF0, TF0, TF1)
UL_TFC11	(TF0, TF1, TF1)
UL_TFC12	(TF0, TF2, TF1)
UL_TFC13	(TF0, TF3, TF1)
UL_TFC14	(TF0, TF4, TF1)
UL_TFC15	(TF1, TF0, TF1)
UL_TFC16	(TF1, TF1, TF1)
UL_TFC17	(TF1, TF2, TF1)
UL_TFC18	(TF1, TF3, TF1)
UL_TFC19	(TF1, TF4, TF1)

Downlink TFS:

	TFI	RB5 (Conv. 64 kbps)	RB6 (I/B 128 kbps)	DCCH
	TF0, bits	0x640	0x336	0x148
	TF1, bits	4x640	1x336	1x148
TFS	TF2, bits	N/A	2x336	N/A
	TF3, bits	N/A	4x336	N/A
	TF4, bits	N/A	8x336	N/A

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF0, TF1, TF0)
DL_TFC2	(TF0, TF2, TF0)
DL_TFC3	(TF0, TF3, TF0)
DL_TFC4	(TF0, TF4, TF0)
DL_TFC5	(TF1, TF0, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF1, TF2, TF0)
DL_TFC8	(TF1, TF3, TF0)
DL_TFC9	(TF1, TF4, TF0)
DL_TFC10	(TF0, TF0, TF1)
DL_TFC11	(TF0, TF1, TF1)
DL_TFC12	(TF0, TF2, TF1)
DL_TFC13	(TF0, TF3, TF1)
DL_TFC14	(TF0, TF4, TF1)
DL_TFC15	(TF1, TF0, TF1)
DL_TFC16	(TF1, TF1, TF1)
DL_TFC17	(TF1, TF2, TF1)
DL_TFC18	(TF1, TF3, TF1)
DL_TFC19	(TF1, TF4, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, UL_TFC11	UL_TFC1, UL_TFC11	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC10, UL_TFC11	RB5: 640 RB6: 312	RB5: No data RB6: 312
2	DL_TFC2, UL_TFC12	UL_TFC2, UL_TFC12	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC10, UL_TFC12	RB5: 640 RB6: 632	RB5: No data RB6: 632
3	DL_TFC3, UL_TFC13	UL_TFC3, UL_TFC13	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC10, UL_TFC13	RB5: 640 RB6: 1272	RB5: No data RB6: 1272
4	DL_TFC4, UL_TFC14	UL_TFC4, UL_TFC14	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC10, UL_TFC14	RB5: 640 RB6: 2552	RB5: No data RB6: 2552
5	DL_TFC5, UL_TFC15	UL_TFC5, UL_TFC15	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 640 RB6: 312	RB5: 4x640 RB6: No data
6	DL_TFC6, UL_TFC16	UL_TFC6, UL_TFC16	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5 UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC15, UL_TFC16	RB5: 640 RB6: 312	RB5: 4x640 RB6: 312
7	DL_TFC7, UL_TFC17	UL_TFC7, UL_TFC17	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC17	RB5: 640 RB6: 632	RB5: 4x640 RB6: 632
8	DL_TFC8, UL_TFC18	UL_TFC8, UL_TFC18	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC15, UL_TFC18	RB5: 4x640 RB6: 1272	RB5: 4x640 RB6: 1272
9	DL_TFC9, UL_TFC19	UL_TFC9, UL_TFC19	DL_TFC0, DL_TFC10, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC19	RB5: 640 RB6: 2552	RB5: 4x640 RB6: 2552
NOTE: RB6: T	est data size h expansion RLC SDU for RB6 ha	has been set to bit). As the TT size has been as been set equind expansion	DL TFS size und I for RB5 and RB set to achieve UE ual to the uplink T	tails regarding loopba er test minus 8 bits (s 6 is the same for both to return one SDU p FS size under test mi LC SDU size for RB5	size of 7 bit lengt downlink and u er TTI, i.e. the U nus 8 bits (size o	th indicator and plink then UL IL RLC SDU size of 7 bit length

2271

14.2.53.2.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 3, 4: an RLC SDU on RB6 having the same content as sent by SS; and no data shall be received on RB5.
 - for sub-test 5: four RLC SDUs on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 6, 7, 8 and 9: four RLC SDUs on RB5 and one RLC SDU on RB6 having the same content as sent by SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.54 Interactive or background / UL:64 DL:128 kbps / PS RAB + Streaming / unknown / UL:0 DL:64 kbps / CS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH

14.2.54.1 Conformance requirement

See 14.2.4.1.

14.2.54.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.54.

To be able to test the downlink radio bearer using the UE loopback function for the reference radio bearer UL:0 DL: 64 kbps, the reference radio bearer configuration according to TS 34.108, clause 6.10.2.4.1.15.1 (Streaming/unknown/UL:14.4 kbps) is used in uplink.

14.2.54.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 (I/B 64 kbps)	RB6 (Str. 14.4 kbps)	DCCH
	TF0, bits	0x336	0x576	0x148
	TF1, bits	1x336	1x576	1x148
TFS	TF2, bits	2x336	N/A	N/A
	TF3, bits	3x336	N/A	N/A
	TF4, bits	4x336	N/A	N/A

2273

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0)
UL_TFC2	(TF2, TF0, TF0)
UL_TFC3	(TF3, TF0, TF0)
UL_TFC4	(TF4, TF0, TF0)
UL_TFC5	(TF0, TF1, TF0)
UL_TFC6	(TF1, TF1, TF0)
UL_TFC7	(TF2, TF1, TF0)
UL_TFC8	(TF3, TF1, TF0)
UL_TFC9	(TF4, TF1, TF0)
UL_TFC10	(TF0, TF0, TF1)
UL_TFC11	(TF1, TF0, TF1)
UL_TFC12	(TF2, TF0, TF1)
UL_TFC13	(TF3, TF0, TF1)
UL_TFC14	(TF4, TF0, TF1)
UL_TFC15	(TF0, TF1, TF1)
UL_TFC16	(TF1, TF1, TF1)
UL_TFC17	(TF2, TF1, TF1)
UL_TFC18	(TF3, TF1, TF1)
UL_TFC19	(TF4, TF1, TF1)

Downlink TFS:

	TFI	RB5 (I/B 128 kbps)	RB6 (Str. 64 kbps)	DCCH
	TF0, bits	0x336	0x320	0x148
	TF1, bits	1x336	1x320	1x148
TFS	TF2, bits	2x336	2x320	N/A
	TF3, bits	4x336	4x320	N/A
	TF4, bits	8x336	8x320	N/A

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0)
DL_TFC2	(TF2, TF0, TF0)
DL_TFC3	(TF3, TF0, TF0)
DL_TFC4	(TF4, TF0, TF0)
DL_TFC5	(TF0, TF1, TF0)
DL_TFC6	(TF1, TF1, TF0)
DL_TFC7	(TF2, TF1, TF0)
DL_TFC8	(TF3, TF1, TF0)
DL_TFC9	(TF4, TF1, TF0)
DL_TFC10	(TF0, TF2, TF0)
DL_TFC11	(TF1, TF2, TF0)
DL_TFC12	(TF2, TF2, TF0)
DL_TFC13	(TF3, TF2, TF0)
DL_TFC14	(TF4, TF2, TF0)
DL_TFC15	(TF0, TF3, TF0)
DL_TFC16	(TF1, TF3, TF0)
DL_TFC17	(TF2, TF3, TF0)
DL_TFC18	(TF3, TF3, TF0)
DL_TFC19	(TF4, TF3, TF0)
DL_TFC20	(TF0, TF4, TF0)
DL_TFC21	(TF1, TF4, TF0)
DL_TFC22	(TF2, TF4, TF0)
DL_TFC23	(TF3, TF4, TF0)
DL_TFC24	(TF4, TF4, TF0)

TFCI	(RB5, RB6, DCCH)
DL_TFC25	(TF0, TF0, TF1)
DL_TFC26	(TF1, TF0, TF1)
DL_TFC27	(TF2, TF0, TF1)
DL_TFC28	(TF3, TF0, TF1)
DL_TFC29	(TF4, TF0, TF1)
DL_TFC30	(TF0, TF1, TF1)
DL_TFC31	(TF1, TF1, TF1)
DL_TFC32	(TF2, TF1, TF1)
DL_TFC33	(TF3, TF1, TF1)
DL_TFC34	(TF4, TF1, TF1)
DL_TFC35	(TF0, TF2, TF1)
DL_TFC36	(TF1, TF2, TF1)
DL_TFC37	(TF2, TF2, TF1)
DL_TFC38	(TF3, TF2, TF1)
DL_TFC39	(TF4, TF2, TF1)
DL_TFC40	(TF0, TF3, TF1)
DL_TFC41	(TF1, TF3, TF1)
DL_TFC42	(TF2, TF3, TF1)
DL_TFC43	(TF3, TF3, TF1)
DL_TFC44	(TF4, TF3, TF1)
DL_TFC45	(TF0, TF4, TF1)
DL_TFC46	(TF1, TF4, TF1)
DL_TFC47	(TF2, TF4, TF1)
DL_TFC48	(TF3, TF4, TF1)
DL_TFC49	(TF4, TF4, TF1)

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
	Test	onder test			(note 1)	(note 1)
1	DL_TFC1, DL_TFC26	UL_TFC1, UL_TFC11	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC10, UL_TFC11	RB5: 312 RB6: 576	RB5: 312 RB6: No data
2	DL_TFC2, DL_TFC27	UL_TFC2, UL_TFC12	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC10, UL_TFC12	RB5: 632 RB6: 576	RB5: 632 RB6: No data
3	DL_TFC3, DL_TFC28	UL_TFC3, UL_TFC13	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC10, UL_TFC13	RB5: 952 RB6: 576	RB5: 1272 RB6: No data
4	DL_TFC4, DL_TFC29	UL_TFC4, UL_TFC14	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC10, UL_TFC14	RB5: 1272 RB6: 576	RB5: 2552 RB6: No data
5	DL_TFC5, DL_TFC30	UL_TFC5, UL_TFC15	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 312 RB6: 576	RB5: No data RB6: 320 (note 2)
6	DL_TFC6, DL_TFC31	UL_TFC6, UL_TFC16	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC16	RB5: 312 RB6: 576	RB5: 312 RB6: 320 (note 2)

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under Test	Under test			(bits) (note 1)	(note 1)
7	DL_TFC7, DL_TFC32	UL_TFC7, UL_TFC17	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5 UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC17	RB5: 632 RB6: 576	RB5: 632 RB6: 320 (note 2)
8	DL_TFC8, DL_TFC33	UL_TFC8, UL_TFC18	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5 UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC15, UL_TFC18	RB5: 952 RB6: 576	RB5: 1272 RB6: 320 (note 2)
9	DL_TFC9, DL_TFC34	UL_TFC9, UL_TFC19	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC15, UL_TFC19	RB5: 1272 RB6: 576	RB5: 2552 RB6: 320 (note 2)
10	DL_TFC10, DL_TFC35	UL_TFC5, UL_TFC15	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 312 RB6: 576	RB5: No data RB6: 640 (note 3)
11	DL_TFC11, DL_TFC36	UL_TFC6, UL_TFC16	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC15, UL_TFC16	RB5: 312 RB6: 576	RB5: 312 RB6: 640 (note 3)
12	DL_TFC12, DL_TFC37	UL_TFC7, UL_TFC17	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5 UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC15, UL_TFC17	RB5: 632 RB6: 576	RB5: 632 RB6: 640 (note 3)
13	DL_TFC13, DL_TFC38	UL_TFC8, UL_TFC18	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC15, UL_TFC18	RB5: 952 RB6: 576	RB5: 1272 RB6: 640 (note 3)
14	DL_TFC14, DL_TFC39	UL_TFC9, UL_TFC19	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC19	RB5: 1272 RB6: 576	RB5: 2552 RB6: 640 (note 3)
15	DL_TFC15, DL_TFC40	UL_TFC5, UL_TFC15	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 312 RB6: 576	RB5: No data RB6: 1280 (note 4)

Sub- test	Downlink TFCS Under	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits)	Test data size (bits)
16	Test DL_TFC16, DL_TFC41	UL_TFC6, UL_TFC16	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC15, UL_TFC16	(note 1) RB5: 312 RB6: 576	(note 1) RB5: 312 RB6: 1280 (note 4)
17	DL_TFC17, DL_TFC42	UL_TFC7, UL_TFC17	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC17	RB5: 632 RB6: 576	RB5: 632 RB6: 1280 (note 4)
18	DL_TFC18, DL_TFC43	UL_TFC8, UL_TFC18	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC15,	RB5: 952 RB6: 576	RB5: 1272 RB6: 1280 (note 4)
19	DL_TFC19, DL_TFC44	UL_TFC9, UL_TFC19	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC15, UL_TFC19	RB5: 1272 RB6: 576	RB5: 2552 RB6: 1280 (note 4)
20	DL_TFC20, DL_TFC45	UL_TFC5, UL_TFC15	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC5, UL_TFC10, UL_TFC15	RB5: 312 RB6: 576	RB5: No data RB6: 2560 (note 5)
21	DL_TFC21, DL_TFC46	UL_TFC6, UL_TFC16	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6, UL_TFC10, UL_TFC11, UL_TFC15, UL_TFC16	RB5: 312 RB6: 576	RB5: 312 RB6: 2560 (note 5)
22	DL_TFC22, DL_TFC47	UL_TFC7, UL_TFC17	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7, UL_TFC10, UL_TFC12, UL_TFC15, UL_TFC17	RB5: 632 RB6: 576	RB5: 632 RB6: 2560 (note 5)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note 1)	Test data size (bits) (note 1)
23	DL_TFC23, DL_TFC48	UL_TFC8, UL_TFC18	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC3, UL_TFC5 UL_TFC8, UL_TFC10, UL_TFC13, UL_TFC15, UL_TFC15, UL_TFC18	RB5: 952 RB6: 576	RB5: 1272 RB6: 2560 (note 5)
24	DL_TFC24, DL_TFC49	UL_TFC9, UL_TFC19	DL_TFC0, DL_TFC25, UL_TFC0, UL_TFC10	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9, UL_TFC10, UL_TFC14, UL_TFC15, UL_TFC15, UL_TFC19	RB5: 1272 RB6: 576	RB5: 2552 RB6: 2560 (note 5)
NOTE	2: RB6: SS is u UE will retur received RL	using a DL RLC n one RLC PDU C PDU.	SDU with 320 bi . SS creates an	ils regarding loopt ts as test data (=E UL RLC SDU fron	DL RLC PDU size n the first 320 bit	e for DL/TF1). ts of the
NOTE	UE will retur	n one RLC PDU	. SS creates an	bits as test data (UL RLC SDU from	n the received R	LC PDU.
NOTE	DL/TF3). UE PDU.	will return one	RLC PDU. SS ci	0 bits as test data reates an UL RLC	SDU from the re	eceived RLC
NOTE	NOTE 5: RB6: SS is using a DL RLC SDU size of 2560 bits as test data (=DL RLC PDU size for DL/TF4). UE will return one RLC PDU. SS creates an UL RLC SDU from the received RLC PDU.					
RB5: T	RB5: Test data size has been set to DL TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit). As the TTI for RB5 and RB6 is the same for both downlink and uplink then UL RLC SDU size has been set to achieve UE to return one SDU per TTI, i.e. the UL RLC SDU size for RB5 has been set equal to the uplink TFS size under test minus 8 bits (size of 7 bit length indicator and expansion bit).and .the UL RLC SDU size for RB6 has been set equal to the uplink TFS size under test.					

14.2.54.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual subtest.
- 3. At step 15 the UE shall return
 - for sub-test 1, 2, 6, 7, 11, 12, 16, 17, 21, 22: an RLC SDU on RB5 having the same content as sent by the SS.
 - for sub-test 5, 10, 15 and 20: no data shall be received on RB5.
 - for sub-test 1 to 4: no data shall be received on RB6.
 - for sub-test 5 to 9: an RLC SDU on RB6 having the same content as sent by the SS.

- for sub-test 10, 11, 12, 15, 16, 17, 20, 21 and 22: an RLC SDU on RB5 having the same content as the first 576 bits of the RLC SDU sent by the SS.

- For sub-test 3,8,13,18,23: an RLC SDU on RB5 having the content equal to the first 952 bits of the test data sent by the SS in downlink;
- For sub-test 4,9,14,19,24: an RLC SDU on RB5 having the content equal to the first 1272 bits of the test data sent by the SS in downlink;

4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.55 Void

14.2.56 Interactive or background / UL:8 DL:8 kbps / PS RAB + Interactive or background / UL:8 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.56.1 Conformance requirement

See 14.2.4.1.

14.2.56.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.56.

14.2.56.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 + RB6 (2x8 kbps)	DCCH
TFS	TF0, bits	0x340	0x148
113	TF1, bits	1x340	1x148

Uplink TFCS:

TFCI		(RB5 + RB6, DCCH)	
UL_TFC0	(TF0, TF0)		
UL_TFC1	(TF1, TF0)		
UL_TFC2	(TF0, TF1)		
UL_TFC3	(TF1, TF1)		

Downlink TFS:

		RB5 + RB6 (2x8 kbps)	DCCH
TFS	TF0, bits	0x340	0x148
15	TF1, bits	1x340	1x148

TFCI		(RB5+RB6, DCCH)	
DL_TFC0	(TF0, TF0)		
DL_TFC1	(TF1, TF0)		
DL_TFC2	(TF0, TF1)		
DL_TFC3	(TF1, TF1)		

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, DL_TFC3	UL_TFC1, UL_TFC3	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 312 RB6: 312	RB5: 312 RB6: no data
2	DL_TFC1, DL_TFC3	UL_TFC1, UL_TFC3	DL_TFC0, DL_TFC2, UL_TFC0, UL_TFC2	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3	RB5: 312 RB6: 312	RB5: no data RB6: 312
NOTE:	NOTE: See TS 34.109 [10] clause 5.3.2.6.2 for details regarding loopback of RLC SDUs. RB5 and RB6: Test data size has been set to the payload size of the DL TF under test minus 8 bits (size of 7 bit length indicator and expansion bit). The UL RLC SDU size has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).					

14.2.56.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: an RLC SDU on RB5 having the same content as sent by SS; and no data shall be received on RB6.
 - for sub-test 2: an RLC SDU on RB6; and no data shall be received on RB5.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.57 Interactive or background / UL:64 DL:64 kbps / PS RAB + Interactive or background / UL:64 DL:64 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.57.1 Conformance requirement

See 14.2.4.1.

14.2.57.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.57.

14.2.57.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 + RB6 (64 kbps RAB, 20 ms TTI)	DCCH
	TF0, bits	0x340	0x148
	TF1, bits	1x340	1x148
TFS	TF2, bits	2x340	N/A
	TF3, bits	3x340	N/A
	TF4, bits	4x340	N/A

Uplink TFCS:

TFCI	(RB5 + RB6, DCCH)	
UL_TFC0	(TF0, TF0)	
UL_TFC1	(TF1, TF0)	
UL_TFC2	(TF2, TF0)	
UL_TFC3	(TF3, TF0)	
UL_TFC4	(TF4, TF0)	
UL_TFC5	(TF0, TF1)	
UL_TFC6	(TF1, TF1)	
UL_TFC7	(TF2, TF1)	
UL_TFC8	(TF3, TF1)	
UL_TFC9	(TF4, TF1)	

Downlink TFS:

	TFI	RB5 + RB6 (64 kbps RAB, 20 ms TTI)	DCCH
	TF0, bits	0x340	0x148
	TF1, bits	1x340	1x148
TFS	TF2, bits	2x340	N/A
	TF3, bits	3x340	N/A
	TF4, bits	4x340	N/A

Downlink TFCS:

TFCI	(RB5 + RB6, D0	CCH)
DL_TFC0	(TF0, TF0)	
DL_TFC1	(TF1, TF0)	
DL_TFC2	(TF2, TF0)	
DL_TFC3	(TF3, TF0)	
DL_TFC4	(TF4, TF0)	
DL_TFC5	(TF0, TF1)	
DL_TFC6	(TF1, TF1)	
DL_TFC7	(TF2, TF1)	
DL_TFC8	(TF3, TF1)	
DL_TFC9	(TF4, TF1)	

Sub- test	Downlink TFCS	Uplink TFCS	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size	Test data size (bits)
	Under test	Under test			(bits) Note 1	Note 1
1	DL_TFC1 DL_TFC6	UL_TFC1 DL_TFC6	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC1, UL_TFC5, UL_TFC6	RB5: 312 RB6: 312	RB5: 312 RB6: No data
2	DL_TFC2 DL_TFC7	UL_TFC2 DL_TFC7	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC2, UL_TFC5, UL_TFC7	RB5: 632 RB6: 632	RB5: 632 RB6: No data
3	DL_TFC3 DL_TFC8	UL_TFC3 DL_TFC8	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC3, UL_TFC5, UL_TFC8	RB5: 952 RB6: 952	RB5: 952 RB6: No data
4	DL_TFC4 DL_TFC9	UL_TFC4 DL_TFC9	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 1272 RB6: 1272	RB5: 1272 RB6: No data
5	DL_TFC4 DL_TFC9	UL_TFC4 DL_TFC9	DL_TFC0, DL_TFC5, UL_TFC0, UL_TFC5	UL_TFC0, UL_TFC4, UL_TFC5, UL_TFC9	RB5: 1272 RB5: 1272	RB5: No data RB6: 1272
	RB5 and RB6:	Test data size	3.2.6.2 for details reg has been set to the p	bayload size of th		

of 7 bit length indicator and expansion bit). The UL RLC SDU size has been set equal to the size of the payload size of the UL TF under test minus 8 bits (the size of 7 bit length indicator and expansion bit).

14.2.57.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.
- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1 to 4: an RLC SDU on RB5 having the same content as the DL RLC SDU sent by the SS in RB5.
 - for sub-test 5: an RLC SDU on RB6 having the same content as the DL RLC SDU sent by the SS in RB6.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.

14.2.58 Streaming / unknown / UL:16 DL:64 kbps / PS RAB + Interactive or background / UL:8 DL:8 kbps / PS RAB + UL:3.4 DL:3.4 kbps SRBs for DCCH.

14.2.58.1 Conformance requirement

See 14.2.4.1.

14.2.58.2 Test purpose

Test to verify establishment and data transfer of reference radio bearer configuration as specified in TS 34.108, clause 6.10.2.4.1.58.

14.2.58.3 Method of test

See 14.1.2 for test procedure.

Uplink TFS:

	TFI	RB5 uplink (16 kbps, 20 ms TTI)	RB6 (8 kbps)	DCCH
TFS	TF0, bits	0x336	0x336	0x148
113	TF1, bits	1x336	1x336	1x148

Uplink TFCS:

TFCI	(RB5, RB6, DCCH)
UL_TFC0	(TF0, TF0, TF0)
UL_TFC1	(TF1, TF0, TF0)
UL_TFC2	(TF0, TF1, TF0)
UL_TFC3	(TF1, TF1, TF0)
UL_TFC4	(TF0, TF0, TF1)
UL_TFC5	(TF1, TF0, TF1)
UL_TFC6	(TF0, TF1, TF1)
UL_TFC7	(TF1, TF1, TF1)

Downlink TFS:

	TFI	RB5 (64 kbps, 40 ms TTI)	RB6 (8 kbps)	DCCH
	TF0, bits	0x656	0x336	0x148
TFS	TF1, bits	1x656	1x336	1x148
11-3	TF2, bits	2x656	N/A	N/A
	TF3, bits	4x656	N/A	N/A

TFCI	(RB5, RB6, DCCH)
DL_TFC0	(TF0, TF0, TF0)
DL_TFC1	(TF1, TF0, TF0)
DL_TFC2	(TF2, TF0, TF0)
DL_TFC3	(TF3, TF0, TF0)
DL_TFC4	(TF0, TF1, TF0)
DL_TFC5	(TF1, TF1, TF0)
DL_TFC6	(TF2, TF1, TF0)
DL_TFC7	(TF3, TF1, TF0)
DL_TFC8	(TF0, TF0, TF1)
DL_TFC9	(TF1, TF0, TF1)
DL_TFC10	(TF2, TF0, TF1)
DL_TFC11	(TF3, TF0, TF1)
DL_TFC12	(TF0, TF1, TF1)
DL_TFC13	(TF1, TF1, TF1)
DL_TFC14	(TF2, TF1, TF1)
DL_TFC15	(TF3, TF1, TF1)

Sub- test	Downlink TFCS Under Test	Uplink TFCS Under test	Implicitely tested	Restricted UL TFCIs	UL RLC SDU size (bits) (note)	Test data size (bits) (note)
1	DL_TFC1, DL_TFC9	UL_TFC1, UL_TFC5	DL_TFC0, DL_TFC8, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC4, UL_TFC5	RB5: 632 RB6: 312	RB5: 632 RB6: no data
2	DL_TFC2, DL_TFC10	UL_TFC1, UL_TFC5	DL_TFC0, DL_TFC8, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC4, UL_TFC5	RB5: 632 RB6: 312	RB5: 1272 RB6: no data
3	DL_TFC3, DL_TFC11	UL_TFC1, UL_TFC5	DL_TFC0, DL_TFC8, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC4, UL_TFC5	RB5: 632 RB6: 312	RB5: 2552 RB6: no data
4	DL_TFC4, DL_TFC12	UL_TFC2, UL_TFC6	DL_TFC0, DL_TFC8, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC2, UL_TFC4, UL_TFC6	RB5: 632 RB6: 312	RB5: no data RB6: 312
5	DL_TFC5, DL_TFC13	UL_TFC3, UL_TFC7	DL_TFC0, DL_TFC8, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC6	RB5: 632 RB6: 312	RB5: 632 RB6: 312
6	DL_TFC6, DL_TFC14	UL_TFC3, UL_TFC7	DL_TFC0, DL_TFC8, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC6	RB5: 632 RB6: 312	RB5: 1272 RB6: 312
7	DL_TFC7, DL_TFC15	UL_TFC3, UL_TFC7	DL_TFC0, DL_TFC8, UL_TFC0, UL_TFC4	UL_TFC0, UL_TFC1, UL_TFC2, UL_TFC3, UL_TFC4, UL_TFC5, UL_TFC6, UL_TFC7	RB5: 632 RB6: 312	RB5: 2552 RB6: 312
NOTE:	RB5: Test bit length in 40 ms ther been set s set to two t indicator a RB6: Test bit length in	data size has ndicator and e n, to achieve c uch that it will times the paylo nd expansion l data size has ndicator and e	se 5.3.2.6.2 for details in been set to the payload xpansion bit). As the up ontinous data transmiss be transmitted over two bad size of the UL TF u bit). been set to the payload xpansion bit). The UL F e of a 7 bit length indica	regarding loopback d size of the DL TF u blink TTI for RB5 is a sion in uplink the siz o subsequent TTIs, i under test minus 8 b d size of the DL TF u RLC SDU size has b	under test minus 20 ms while the ce of the uplink F i.e. UL RLC SDU its (the size of a under test minus been set equal to	downlink TTI is RLC SDU has J size has been 7 bit length 8 bits (size of 7

14.2.58.4 Test requirements

See 14.1.2 for definition of step 10 and step 15.

- 1. At step 10 the UE shall send RADIO BEARER SETUP COMPLETE.
- 2. At step 15a and step 15b the UE transmitted transport format shall be within the set of restricted TFCIs as specified for the actual sub-test.

- 3. At step 15a and step 15b the UE shall return
 - for sub-test 1: RLC SDUs on RB5 having the same content as sent by the SS; and no data shall be received on RB6.
 - for sub-test 2 and 3: RLC SDUs on RB5 having the first 632 bits equal to the content of the test data sent by theSS in downlink; and no data shall be received on RB6.
 - for sub-test 4: RLC SDUs on RB6 having the same content as sent by the SS; and no data shall be received on RB5.
 - for sub-test 5: RLC SDUs on RB5 and RB6 having the same content as sent by the SS.
 - for sub-test 6 and 7: RLC SDUs on RB5 having the first 632 bits equal to the content of the test data sent by theSS in downlink; and RLC SDUs on RB6 having the same content as sent by the SS.
- 4. At step 15b the UE shall send at least one MEASUREMENT REPORT message.